PATTERNS AND PROCESSES OF SPECIES DIVERSITY IN FRAGMENTED NORTHERN HARDWOOD FORESTS

BY

WILLIAM BRUCE DRAPER

A THESIS SUBMITTED IN CONFORMITY WITH THE REQUIREMENTS FOR THE DEGREE OF MASTER OF SCIENCE GRADUATE DEPARTMENT OF BOTANY, UNIVERSITY OF TORONTO

© WILLIAM BRUCE DRAPER 2001



National Library of Canada

Acquisitions and Bibliographic Services

395 Wellington Street Ottawa ON K1A 0N4 Canada Bibliothèque nationale du Canada

Acquisitions et services bibliographiques

395, rue Wellington Ottawa ON K1A 0N4 Canada

Your Se Voint attinued

Our No. Notre reference

The author has granted a nonexclusive licence allowing the National Library of Canada to reproduce, loan, distribute or sell copies of this thesis in microform, paper or electronic formats.

The author retains ownership of the copyright in this thesis. Neither the thesis nor substantial extracts from it may be printed or otherwise reproduced without the author's permission.

L'auteur a accordé une licence non exclusive permettant à la Bibliothèque nationale du Canada de reproduire, prêter, distribuer ou vendre des copies de cette thèse sous la forme de microfiche/film, de reproduction sur papier ou sur format électronique.

L'auteur conserve la propriété du droit d'auteur qui protège cette thèse. Ni la thèse ni des extraits substantiels de celle-ci ne doivent être imprimés ou autrement reproduits sans son autorisation.

0-612-58736-3

Canadä

PATTERNS AND PROCESSES OF SPECIES DIVERSITY IN FRAGMENTED NORTHERN HARDWOOD FORESTS

William Bruce Draper, Department of Botany, University of Toronto

Master of Science, 2001

ABSTRACT

Twenty-four upland hardwood forests were examined to determine the influence of plant dispersal and environmental heterogeneity on the composition and richness of species in the forest understory. Patterns in the dispersal attributes of established plants were evaluated in relation to associated plant traits, microhabitats on the forest floor, and measured or inferred gradients in the availability of moisture, nutrients and light.

Plant attributes that maximally explained differences in species richness were mode of dispersal, habitat affinity. life form, shade tolerance, and moisture affinity. Environmental variables that maximally explained species richness were stand structure, soil moisture and soil parent material. Species richness was strongly influenced by sugar maple abundance and declined sharply on mesic, calcium-rich soils.

In this study, modes of dispersal were strongly correlated with plant traits and habitat factors that govern germination, establishment and persistence. The contribution of dispersal and environmental processes to species richness could not be distinguished, statistically.

ACKNOWLEDGEMENTS

I wish to thank the members of my supervisory committee. Dr. D.A. Wedin. Dr. T.J. Carleton, and Dr. R.L. Jefferies. Thank you, Dave, for your willingness to oversee this study during the initial stages and for your encouragement during the darker hours and financial support. Thank you, Terry, for your interest, unfailing good advice, and the many hours of exchange with you and members of your lab. Thank you, Bob, for supervising this study during the final stages and for your sound advice and encouragement throughout this study.

I especially wish to thank the following faculty members of the Department of Botany, University of Toronto, for their generous assistance and advice with various technical analyses: Drs. John Williams. Mobasher Kahn, and Nancy Dengler for lipid analyses related to ant dispersal; Drs. Victor Timmer and Y. Teng for the analysis of base cations: and, Dr. Roger Hansell, Institute for Environmental Studies, for identifying a graphical method for comparing the degree to which elements in linear regression models contribute to an explanation of variance.

I also wish to acknowledge, and thank. Dr. James E. Eckenwalder for the many discussions regarding the nature of plant dispersal and for comments received on the first draft of the introduction to Chapter 3.

I have also benefitted from the generous assistance of specialists in the Ontario flora in the identification of difficult taxa and plant specimens. *Carex* and Poaceace identifications were reviewed by M.J. Oldham. Natural Heritage Information Centre, Ontario Ministry of Natural Resources; *Carex*. Section Ovales, identifications were reviewed by Dr. P.W. Ball, Erindale College, University of Toronto; selected grass specimens were identified by Dr. S.J. Darbyshire. Agriculture and Agri-Food Canada; selected fern identifications were reviewed by Dr. D.M. Britton. Professor Emeritus, University of Guelph.

I wish also to thank Dr. T.A. Dickinson for generous access to the collections at the Vascular Plant Herbarium, Royal Ontario Museum, Toronto, Ontario. I especially wish to acknowledge, and thank. Donna Stulgys and Indira Pillai, Inter-Library Loan Service, Gerstein Science Information Centre, University of Toronto, for their creativity and tireless efforts in obtaining texts and photocopies from other institutions.

I wish also to thank Deborah Tam, Department of Botany, for encouragement and support, and. John McCarron and Ian Kennedy. Department of Forestry, for generous access to field equipment and computer resources.

I wish also to thank Richard Joos. Department of Forestry, and David Kubien, Department of Botany, for listening and for their advice.

I especially wish to thank Jane Bowles, University of Western Ontario, for providing rich opportunities to learn and for asking me to participate in the field studies that sparked my interest in the questions examined in this thesis.

This thesis was funded in part by a PGS A scholarship from the Natural Sciences and Engineering Research Council.

I dedicate this thesis to my life-long partner, and friend. Linda Galen,

TABLE OF CONTENTS

		Pa	ige
Abstra	ct		ïí
Ackno	wledger	ments	tii
Table o	of Conte	ents	. v
Append	dices		'iii
List of	Tables		ix
List of	Figures	\$	xi
1.0	NOR	NUCTION	
1.0	INTRO		. 1
2.0	ENVIE	RONMENTAL PATTERNS	. 9
2.1	Introdu	uction	. 9
2.2	Metho	ds	12
	2.2.1	Selection of Study Sites	12
	2.2.2	Sampling Methods	14
	2.2.3	Environmental Variables	15
		2.2.3.1 Soil Variables	15
		2.2.3.2 Forest Stand Structure Variables	22
		2.2.3.3 Vegetation Variables	24
		2.2.3.4 Plant Attribute Variables	25
		2.2.3.5 Microhabitat Variables	26
		2.2.3.6 Landscape Variables	32
		2.2.3.7 Supplementary Data	33
	2.2.4	Analytical Methods	33
		2.2.4.1 Overview	33
		2.2.4.2 Distribution of Species on Environmental Gradients	35
		2.2.4.3 Distribution of Species in Forest Microhabitats	37
		2.2.4.4 Distribution of Sugar Maple on a Moisture-Fertility Gradient	39
		2.2.4.5 Distribution of Plant Attributes on Primary Gradients	41
	2.3	Results	42
		2.3.1 Vegetation Survey	42
		2.3.2 Relationships Among Environmental Variables	55
		2.3.3 Species Response to Environmental Gradients	63
		2.3.4 Species Response to Forest Microhabitats	71
		2.3.5 Response of Sugar Maple and Understory Herbs to Available	
		Calcium	79
		2.3.6 Plant Attributes	87
	2.4	Discussion	95
	2.5	Principal Findings 1	08
	NOR		
3.0	DISPE	KSAL PATTERNS	13
5.1	Introd		ć I .

TABLE OF CONTENTS

	Mathe	1.	Page
3.2		Classific start of Discourse has dee	107
	1.2.د	Classification of Dispersal Modes	123
		3.2.1.1. Overview	122
		S.2.1.2. Classification Criteria and Related Considerations	121
	3.2.2	Classification of Other Plant Traits	121
	د.۷.ډ	Identification of Patient	122
		3.2.3.1 Pattern in Relation to Plant Attributes	201
		3.2.3.2 Pattern in Relation to Environmental Variables	
		3.2.3.3 Pattern in Relation to Abundance Variables	
• •	• L	3.2.3.4 Pattern in Kelation to Spatial Scale	136
5.3	Result	IS,	136
	3.3.1	Pattern in Relation to Plant Attributes	136
		3.3.1.1 Life Form	137
		3.3.1.2 Life History	139
		3.3.1.3 Provenance	139
		3.3.1.4 Modality	, . 144
		3.3.1.5 Fruit Type	. 144
		3.3.1.6 Taxonomic Rank	.,149
	3.3.2	Pattern in Relation to Environmental Variables	, . 149
		3.3.2.1 Habitat Affinity	. 149
		3.3.2.2 Environmental Gradients	154
		3.3.2.3 Microhabitats	169
	3.3.3	Pattern in Relation to Abundance Variables	184
		3.3.3.1 Frequency Class	184
		3.3.3.2 Cover Class	184
		3.3.3.3 Species Richness Class	187
	3.3.4	Pattern in Relation to Spatial Scale	187
		3.3.4.1 Patch Size	189
		3.3.4.2 Patch Isolation	189
3.4	Discu	ssion	192
3.5	Princi	pal Findings	203
40	ратт	TERNS OF SPECIES RICHNESS	206
4.0	Introd	luction	706
4.2	Study	Mathode	
⊣. ≟	4.2.1	Environmental Correlates of Dickness	212
	4.2.L	4.2.1.1 Concretized Litera Regression Models	212
		4.2.1.1 Generalized Linear Regression Models	
		4.2.1.2 Contribution of Porest Stand Structure	112 م. مرد
		4.2.1.5 Contribution of SOIF Perfility	
		4.2.1.4. Contribution of Patch Isolation and Patch Size	
		4.2.1.5 Contribution of Microhabitats	

TABLE OF CONTENTS

			Page
	4.2.2	Plant Trait Correlates of Richness	218
	4.2.3	Comparison of Alternative Models of Species Richness	219
	4,2,4	Contribution of Phylogeny	220
4.3	Study	Results	221
	4.3.1.	Environmental Correlates of Richness	221
		4.3.1.1 Generalized Linear Regression Models	221
		4.3.1.2 Contribution of Forest Stand Structure	236
		4.3.1.3 Contribution of Soil Fertility	220
		4.3.1.4 Contribution of Patch Isolation and Patch Size	248
		4.3.1.5 Contribution of Microhabitats	749
	4.3.2	Plant Trait Correlates of Richness	・・・ユ キ ノ うら7
		4 3 2 1 Generalized Linear Regression Models	757
	433	Comparison of Alternative Models of Spacies Richness	270
	4.2.2	1331 Generalized Linear Parametrica Modela	±70 270
		4.3.2.2 Graphical Evolution of Alternative Medals of Species Diskeys	
	4.2.4	4.5.5.2 Graphical Evaluation of Alternative Models of Species Rienness	
	4.3.4	Contribution of Phylogeny	
4.4	Discu	SSION	
4.5	Princi	pal Findings	301
5.0	GENE	ERAL CONCLUSIONS	306
6.0	LITE	RATURE CITED	322

APPENDICES

	Page
Appendix 1	Location of study sites
Appendix 2	Summary of plant attributes by species
Appendix 3	Species codes
Appendix 4	Distribution of species by soil parent material, soil order, soil moisture
	and canopy closure
Appendix 5	Listing of species by microhabitat
Appendix 6	Distribution of species by microhabitat: closed canopy
Appendix 7	Distribution of species by microhabitat: open canopy
Appendix 8	Environmental data summary by quadrat (1)
Appendix 9	Environmental data summary by quadrat (II)
Appendix 10	Environmental data summary by quadrat (III)
Appendix 11	Representative seed dispersal distances of native and alien species
	in North America
Appendix 12	Principal frugivores of eastern North America
Appendix 13	Known myrmecochores in the U.S. Northeast
Appendix 14	Species prevalent in the herb layer in the Maple-Basswood Forest Region
	in southern Wisconsin and present in sugar maple dominated stands in the
	vicinity of Peterborough. Ontario

LIST OF TABLES

Table 1.1 Comparison of dispersal spectra in plant communities of broadly similar structure in eastern North America 4 Table 2.1 Size distribution of sampled forest patches 13 Table 2.2 Summary of environmental variables 16 Table 2.3 Definition and areal extent of forest microhabitats 27 Table 2.4 Summary of surveyed taxa by taxonomic rank. life form. life history and provenance 43 Table 2.5 Cover class of surveyed taxa by faving (alphabetical order) 46 Table 2.6 Check list of surveyed taxa by faving (alphabetical order) 46 Table 2.7 Distribution of selected environmental variables by soil parent material 56 Table 2.8 Distribution of selected environmental variables by soil moisture class 58 Table 2.10 Distribution of selected environmental variables by soil order 57 Table 2.11 Attributes of forest stand structure by forest cover type 60 Table 2.12 Distribution of selected environmental variables by patch size 61 Table 2.13 Relative importance of environmental variables in species ordination (CCA) 68 Table 2.14 Attributes of forest stand structure by tariables in species cover type 60 Table 2.15		Page
structure in eastern North America 4 Table 2.1 Size distribution of sampled forest patches 13 Table 2.2 Summary of environmental variables 16 Table 2.3 Definition and areal extent of forest microhabitats 27 Table 2.4 Summary of surveyed taxa by taxonomic rank. life form. life history 43 Table 2.5 Cover class of surveyed taxa by family (alphabetical order) 46 Table 2.6 Check list of surveyed taxa by family (alphabetical order) 46 Table 2.7 Distribution of selected environmental variables by soil parent material 56 Table 2.7 Distribution of selected environmental variables by soil order 57 Table 2.9 Distribution of selected environmental variables by forest cover type 50 Table 2.1 Distribution of selected environmental variables by forest cover type 60 Table 2.1 Distribution of selected environmental variables in species ordination (CCA) 68 Table 2.1 Relative importance of environmental variables in species ordination (CCA) 68 Table 2.11 Number of species with a restricted and unrestricted spatial distribution in relation to soil parent material. soil order. soil moisture. forest cover type 70 Table 2.15 <td< td=""><td>Table 1.4</td><td>Comparison of dispersal spectra in plant communities of broadly similar</td></td<>	Table 1.4	Comparison of dispersal spectra in plant communities of broadly similar
Table 2.1 Size distribution of sampled forest patches 13 Table 2.2 Summary of environmental variables 16 Table 2.3 Definition and areal extent of forest microhabitats 27 Table 2.4 Summary of surveyed taxa by taxonomic rank. life form. life history and provenance 43 Table 2.5 Cover class of surveyed taxa in herb layer inf0m x 10m quadrats by life form. life history, provenance, and habitat affinity 45 Table 2.6 Check list of surveyed taxa by family (alphabetical order) 46 Table 2.7 Distribution of selected environmental variables by soil order 57 Table 2.8 Distribution of selected environmental variables by soil order 57 Table 2.10 Distribution of selected environmental variables by soil order 59 Table 2.11 Attributes of forest stand structure by forest cover type 60 Table 2.12 Distribution of selected environmental variables by patch size 61 Table 2.13 Relative importance of environmental variables in species ordination (CCA) 68 Table 2.14 Number of species with a restricted and uncestricted spatial distribution in relation to soil parent material, soil order, soil moisture, forest cover type and canopy closure 70 Table 2.15 Number of microhabitats o		structure in eastern North America
Table 2.2 Summary of environmental variables 16 Table 2.3 Definition and areal extent of forest microhabitats 27 Table 2.4 Summary of surveyed taxa by taxonomic rank. life form, life history and provenance 43 Table 2.5 Cover class of surveyed taxa by family (alphabetical order) 45 Table 2.6 Check list of surveyed taxa by family (alphabetical order) 46 Table 2.6 Check list of surveyed taxa by family (alphabetical order) 46 Table 2.6 Distribution of selected environmental variables by soil parent material 56 Table 2.6 Distribution of selected environmental variables by soil order 57 Table 2.10 Distribution of selected environmental variables by forest cover type 60 Table 2.11 Attributes of forest stand structure by forest cover type 60 Table 2.12 Distribution of selected environmental variables in species ordination (CCA) 68 61 Table 2.13 Relative importance of environmental variables in species ordination (CCA) 68 75 Table 2.14 Number of species with a restricted and unrestricted spatial distribution in relation to soil parent material, soil order, soil moisture, forest cover type and canopy closure 75 Table 2.15 Number of surveyed taxa by life form </td <td>Table 2.1</td> <td>Size distribution of sampled forest patches</td>	Table 2.1	Size distribution of sampled forest patches
Table 2.3 Definition and areal extent of forest microhabitats 27 Table 2.4 Summary of surveyed taxa by taxonomic rank. life form. life history and provenance 43 Table 2.5 Cover class of surveyed taxa in herb layer in10m x 10m quadrats by life form. life history, provenance, and habitat affinity 45 Table 2.6 Check list of surveyed taxa by family (alphabetical order)	Table 2.2	Summary of environmental variables
Table 2.4 Summary of surveyed taxa by taxonomic rank, life form, life history and provenance 43 Table 2.5 Cover class of surveyed taxa in herb layer in 10m x 10m quadrats by life form, life history, provenance, and habitat affinity 45 Table 2.6 Check list of surveyed taxa by family (alphabetical order) 46 Table 2.7 Distribution of selected environmental variables by soil order 57 Table 2.8 Distribution of selected environmental variables by soil moisture class 58 Table 2.9 Distribution of selected environmental variables by forest cover type 59 Table 2.10 Distribution of selected environmental variables by protest cover type 60 Table 2.11 Attributes of forest stand structure by forest cover type 61 Table 2.12 Distribution of selected environmental variables in species ordination (CCA) 68 Table 2.14 Number of species with a restricted and uncestricted spatial distribution in relation to soil parent material, soil order, soil moisture, forest cover type and canopy closure 70 Table 2.16 Proportion of total inertia in the dispersion of species scores (CCA) explained by functional attributes of sampled flora by life form 91 Table 2.19 Shade tolerance of sampled flora by life form 138 Table 2.11 Moisture tolerance of s	Table 2.3	Definition and areal extent of forest microhabitats
and provenance 43 Table 2.5 Cover class of surveyed taxa in herb layer in10m x 10m quadrats by life form. life history, provenance, and habitat affinity 45 Table 2.6 Check list of surveyed taxa by family (alphabetical order) 46 Table 2.7 Distribution of selected environmental variables by soil order 57 Table 2.8 Distribution of selected environmental variables by soil moisture class 58 Table 2.10 Distribution of selected environmental variables by forest cover type 60 Table 2.11 Attributes of forest stand structure by torest cover type 61 Table 2.12 Distribution of selected environmental variables by patch size 61 Table 2.13 Relative importance of environmental variables in species ordination (CCA) 68 Table 2.14 Relative importance of environmental variables in species ordination (CCA) 68 Table 2.15 Number of species with a restricted and unrestricted spatial distribution in relation to soil parent material, soil order, soil moisture, forest cover type 70 Table 2.16 Proportion of total inertia in the dispersion of species scores (CCA) 88 Table 2.17 Shade tolerance of sampled flora by life form 91 Table 2.18 Moisture tolerance of sampled flora by life form	Table 2.4	Summary of surveyed taxa by taxonomic rank, life form, life history
Table 2.5 Cover class of surveyed taxa in herb layer in 10m x 10m quadrats by life form. life history, provenance, and habitat affinity 45 Table 2.6 Check list of surveyed taxa by family (alphabetical order) 46 Table 2.7 Distribution of selected environmental variables by soil parent material 56 Table 2.7 Distribution of selected environmental variables by soil moisture class 58 Table 2.9 Distribution of selected environmental variables by forest cover type 59 Table 2.10 Distribution of selected environmental variables by patch size 61 Table 2.11 Attributes of forest stand structure by torest cover type 60 Table 2.12 Distribution of selected environmental variables in species ordination (CCA) 68 Table 2.13 Relative importance of environmental variables in species ordination (CCA) 68 Table 2.14 Number of species with a restricted and unrestricted spatial distribution in relation to soil parent material, soil order, soil moisture, forest cover type and canopy closure 70 Table 2.15 Number of microhabitats occupied by surveyed taxa 75 Table 2.16 Proportion of total inertia in the dispersion of species scores (CCA) 88 Table 2.17 Shade tolerance of sampled flora by life form 90		and provenance
life form. life history, provenance, and habitat affinity 45 Table 2.6 Check list of surveyed taxa by family (alphabetical order) 46 Table 2.7 Distribution of selected environmental variables by soil parent material 56 Table 2.8 Distribution of selected environmental variables by soil moisture class 58 Table 2.9 Distribution of selected environmental variables by soil moisture class 58 Table 2.10 Distribution of selected environmental variables by patch size 61 Table 2.11 Attributes of forest stand structure by forest cover type 59 Table 2.12 Distribution of selected environmental variables by patch size 61 Table 2.13 Relative importance of environmental variables in species ordination (CCA) 68 Table 2.14 Number of species with a restricted and unrestricted spatial distribution in relation to soil parent material, soil order, soil moisture, forest cover type and canopy closure 70 Table 2.15 Number of microhabitats occupied by surveyed taxa 75 Table 2.16 Proportion of total inertia in the dispersion of species scores (CCA) explained by functional attributes of sampled plants 88 Table 2.17 Shade tolerance of sampled flora by life form 90 Table 2.18 Moisture tolerance	Table 2.5	Cover class of surveyed taxa in herb layer in10m x 10m quadrats by
Table 2.6Check list of surveyed taxa by family (alphabetical order)46Table 2.7Distribution of selected environmental variables by soil parent material56Table 2.8Distribution of selected environmental variables by soil moisture class58Table 2.9Distribution of selected environmental variables by forest cover type59Table 2.11Attributes of forest stand structure by torest cover type60Table 2.12Distribution of selected environmental variables by patch size61Table 2.13Relative importance of environmental variables in species ordination (CCA)68Table 2.14Number of species with a restricted and unrestricted spatial distribution in relation to soil parent material, soil order, soil moisture, forest cover type and canopy closure70Table 2.15Number of microhabitats occupied by surveyed taxa75Table 2.16Proportion of total inertia in the dispersion of species scores (CCA) explained by functional attributes of sampled plants88Table 2.19Percentage of classified taxa in surveyed microhabitats by provenance, habitat affinity, and moisture affinity92Table 3.1Dispersal modes of surveyed taxa by dispersal mode and life form140Table 3.2Life history of surveyed taxa by dispersal mode and life form142Table 3.4Modality of surveyed taxa by dispersal mode and life form142Table 3.5Fruit type of surveyed taxa by dispersal mode and life form142Table 3.6Distribution of dispersal modes by taxonomic rank I (genus)151Table 3.7Distribution o		life form, life history, provenance, and habitat affinity
Table 2.7Distribution of selected environmental variables by soil parent material56Table 2.8Distribution of selected environmental variables by soil moisture class57Table 2.10Distribution of selected environmental variables by soil moisture class58Table 2.11Attributes of forest stand structure by torest cover type60Table 2.12Distribution of selected environmental variables by patch size61Table 2.13Relative importance of environmental variables in species ordination (CCA)68Table 2.14Number of species with a restricted and unrestricted spatial distribution in relation to soil parent material, soil order, soil moisture, forest cover type and canopy closure70Table 2.15Number of microhabitats occupied by surveyed taxa75Table 2.16Proportion of total inertia in the dispersion of species scores (CCA) explained by functional attributes of sampled flora by life form91Table 2.19Moisture tolerance of sampled flora by life form91Table 2.19Dispersal modes of surveyed taxa by dispersal mode and life form92Table 3.1Dispersal modes of surveyed taxa by dispersal mode and life form142Table 3.2Fruit type of surveyed taxa by dispersal mode and life form142Table 3.3Provenance of surveyed taxa by dispersal mode and life form142Table 3.4Modality of surveyed taxa by dispersal mode and life form142Table 3.5Fruit type of surveyed taxa by dispersal mode and life form142Table 3.6Distribution of dispersal modes by taxonomic rank II (family	Table 2.6	Check list of surveyed taxa by family (alphabetical order)
Table 2.8Distribution of selected environmental variables by soil order57Table 2.9Distribution of selected environmental variables by forest cover type58Table 2.10Distribution of selected environmental variables by forest cover type60Table 2.11Attributes of forest stand structure by forest cover type60Table 2.12Distribution of selected environmental variables by patch size61Table 2.13Relative importance of environmental variables in species ordination (CCA)68Table 2.14Number of species with a restricted and unrestricted spatial distribution in relation to soil parent material, soil order, soil moisture, forest cover type and canopy closure70Table 2.15Number of microhabitats occupied by surveyed taxa75Table 2.16Proportion of total inertia in the dispersion of species scores (CCA) explained by functional attributes of sampled flora by life form91Table 2.17Shade tolerance of sampled flora by life form91Table 2.18Moisture tolerance of sampled flora by life form92Table 2.19Percentage of classified taxa in surveyed microhabitats by provenance, habitat affinity, and moisture affinity92Table 3.1Dispersal modes of surveyed taxa by dispersal mode and life form142Table 3.2Fruit type of surveyed taxa by dispersal mode and life form142Table 3.4Modality of surveyed taxa by dispersal mode147Table 3.5Fruit type of surveyed taxa by dispersal mode142Table 3.6Distribution of dispersal modes by taxonomic rank II (family)	Table 2.7	Distribution of selected environmental variables by soil parent material 56
Table 2.9Distribution of selected environmental variables by soil moisture class58Table 2.10Distribution of selected environmental variables by forest cover type59Table 2.11Attributes of forest stand structure by forest cover type60Table 2.12Distribution of selected environmental variables by patch size61Table 2.13Relative importance of environmental variables in species ordination (CCA)68Table 2.14Number of species with a restricted and unrestricted spatial distribution in relation to soil parent material, soil order, soil moisture, forest cover type and canopy closure70Table 2.15Number of microhabitats occupied by surveyed taxa75Table 2.16Proportion of total inertia in the dispersion of species scores (CCA) explained by functional attributes of sampled plants88Table 2.19Percentage of classified taxa in surveyed microhabitats by provenance, habitat affinity, and moisture affinity92Table 3.1Dispersal modes of surveyed taxa by dispersal mode and life form140Table 3.2Life history of surveyed taxa by dispersal mode and life form142Table 3.4Modality of surveyed taxa by dispersal mode and life form142Table 3.5Fruit type of surveyed taxa by dispersal mode and life form143Table 3.6Distribution of dispersal modes by taxonomic rank II (family)151Table 3.7Distribution of dispersal modes by taxonomic rank III (order)152Table 3.8Distribution of dispersal modes by taxonomic rank III (order)152Table 3.9Dispersal modes o	Table 2.8	Distribution of selected environmental variables by soil order
Table 2.10Distribution of selected environmental variables by forest cover type59Table 2.11Attributes of forest stand structure by forest cover type60Table 2.12Distribution of selected environmental variables in species ordination (CCA)68Table 2.13Relative importance of environmental variables in species ordination (CCA)68Table 2.14Number of species with a restricted and unrestricted spatial distribution in relation to soil parent material, soil order, soil moisture, forest cover type and canopy closure70Table 2.15Number of microhabitats occupied by surveyed taxa75Table 2.16Proportion of total inertia in the dispersion of species scores (CCA) explained by functional attributes of sampled plants88Table 2.19Moisture tolerance of sampled flora by life form90Table 2.19Percentage of classified taxa in surveyed microhabitats by provenance, 	Table 2.9	Distribution of selected environmental variables by soil moisture class 58
Table 2.11Attributes of forest stand structure by forest cover type60Table 2.12Distribution of selected environmental variables by patch size61Table 2.13Relative importance of environmental variables in species ordination (CCA)68Table 2.14Number of species with a restricted and unrestricted spatial distribution in relation to soil parent material, soil order, soil moisture, forest cover type and canopy closure70Table 2.15Number of microhabitats occupied by surveyed taxa75Table 2.16Proportion of total inertia in the dispersion of species scores (CCA) explained by functional attributes of sampled plants88Table 2.19Shade tolerance of sampled flora by life form90Table 2.19Percentage of classified taxa in surveyed microhabitats by provenance, habitat affinity, and moisture affinity92Table 3.1Dispersal modes of surveyed taxa by dispersal mode and life form142Table 3.2Fruit type of surveyed taxa by dispersal mode and life form142Table 3.4Modality of surveyed taxa by dispersal mode and life form142Table 3.5Fruit type of surveyed herbs by dispersal mode147Table 3.6Distribution of dispersal modes by taxonomic rank I (genus)151Table 3.7Dispersal modes of herbs by habitat affinity153Table 3.8Distribution of dispersal modes by taxonomic rank II (order)152Table 3.9Dispersal modes of herbs by habitat affinity153Table 3.1Dispersal modes of herbs by habitat affinity153Table 3.2Dispersal	Table 2.10	Distribution of selected environmental variables by forest cover type
Table 2.12Distribution of selected environmental variables by patch size61Table 2.13Relative importance of environmental variables in species ordination (CCA)68Table 2.14Number of species with a restricted and unrestricted spatial distribution in relation to soil parent material, soil order, soil moisture, forest cover type and canopy closure70Table 2.15Number of microhabitats occupied by surveyed taxa75Table 2.16Proportion of total inertia in the dispersion of species scores (CCA) explained by functional attributes of sampled plants88Table 2.17Shade tolerance of sampled flora by life form90Table 2.18Moisture tolerance of sampled flora by life form91Table 2.19Percentage of classified taxa in surveyed microhabitats by provenance, habitat affinity, and moisture affinity92Table 3.1Dispersal modes of surveyed taxa by dispersal mode and life form140Table 3.2Life history of surveyed taxa by dispersal mode and life form142Table 3.4Modality of surveyed herbs by dispersal mode and life form142Table 3.5Fruit type of surveyed herbs by dispersal mode and life form145Table 3.6Distribution of dispersal modes by taxonomic rank I (genus)151Table 3.7Dispersal modes of herbs by habitat affinity152Table 3.8Dispersal modes of herbs by habitat affinity152Table 3.9Dispersal modes of herbs by habitat affinity151Table 3.4Modality of surveyed herbs by habitat affinity151Table 3.5Fruit type of su	Table 2.11	Attributes of forest stand structure by forest cover type
Table 2.13Relative importance of environmental variables in species ordination (CCA).68Table 2.14Number of species with a restricted and unrestricted spatial distribution in relation to soil parent material, soil order, soil moisture, forest cover type and canopy closure	Table 2.12	Distribution of selected environmental variables by patch size
Table 2.14Number of species with a restricted and unrestricted spatial distribution in relation to soil parent material, soil order, soil moisture, forest cover type and canopy closure	Table 2.13	Relative importance of environmental variables in species ordination (CCA) 68
relation to soil parent material, soil order, soil moisture, forest cover type and canopy closure	Table 2.14	Number of species with a restricted and unrestricted spatial distribution in
and canopy closure70Table 2.15Number of microhabitats occupied by surveyed taxa75Table 2.16Proportion of total inertia in the dispersion of species scores (CCA)88Table 2.17Shade tolerance of sampled flora by life form90Table 2.18Moisture tolerance of sampled flora by life form91Table 2.19Percentage of classified taxa in surveyed microhabitats by provenance,91Table 3.1Dispersal modes of surveyed taxa by life form92Table 3.2Life history of surveyed taxa by dispersal mode and life form140Table 3.3Provenance of surveyed taxa by dispersal mode and life form142Table 3.4Modality of surveyed taxa by dispersal mode and life form142Table 3.5Fruit type of surveyed taxa by dispersal mode and life form145Table 3.6Distribution of dispersal modes by taxonomic rank I (genus)150Table 3.7Distribution of dispersal modes by taxonomic rank II (amily)151Table 3.8Dispersal modes of herbs by habitat affinity153Table 3.9Dispersal modes of herbs by shade tolerance156Table 3.10Dispersal modes of herbs by shade tolerance156Table 3.12Dispersal modes of herbs by shade tolerance156Table 3.13Number of observed and expected herbs in surveyed microhabitats by158		relation to soil parent material, soil order, soil moisture, forest cover type
Table 2.15Number of microhabitats occupied by surveyed taxa75Table 2.16Proportion of total inertia in the dispersion of species scores (CCA)explained by functional attributes of sampled plants88Table 2.17Shade tolerance of sampled flora by life form90Table 2.18Moisture tolerance of sampled flora by life form91Table 2.19Percentage of classified taxa in surveyed microhabitats by provenance.92Table 3.1Dispersal modes of surveyed taxa by life form138Table 3.2Life history of surveyed taxa by life form140Table 3.3Provenance of surveyed taxa by dispersal mode and life form142Table 3.4Modality of surveyed taxa by dispersal mode and life form142Table 3.5Fruit type of surveyed herbs by dispersal mode and life form145Table 3.6Distribution of dispersal modes by taxonomic rank I (genus)151Table 3.7Distribution of dispersal modes by taxonomic rank III (order)152Table 3.8Dispersal modes of herbs by habitat affinity153Table 3.10Dispersal modes of herbs by shade tolerance156Table 3.12Dispersal modes of herbs in 10m x 10m quadrats by environmental attribute158Table 3.13Number of observed and expected herbs in surveyed microhabitats by158		and canopy closure
Table 2.16Proportion of total inertia in the dispersion of species scores (CCA) explained by functional attributes of sampled plants88Table 2.17Shade tolerance of sampled flora by life form90Table 2.18Moisture tolerance of sampled flora by life form91Table 2.19Percentage of classified taxa in surveyed microhabitats by provenance, habitat affinity, and moisture affinity92Table 3.1Dispersal modes of surveyed taxa by life form138Table 3.2Life history of surveyed taxa by dispersal mode and life form140Table 3.3Provenance of surveyed taxa by dispersal mode and life form142Table 3.4Modality of surveyed taxa by dispersal mode and life form142Table 3.5Fruit type of surveyed herbs by dispersal mode and life form145Table 3.6Distribution of dispersal modes by taxonomic rank I (genus)151Table 3.7Distribution of dispersal modes by taxonomic rank II (order)152Table 3.8Dispersal modes of herbs by habitat affinity153Table 3.9Dispersal modes of herbs by shade tolerance156Table 3.10Dispersal modes of herbs by shade tolerance156Table 3.12Distribution of herbs in 10m x 10m quadrats by environmental attribute and dispersal mode158Table 3.13Number of observed and expected herbs in surveyed microhabitats bv158	Table 2.15	Number of microhabitats occupied by surveyed taxa
explained by functional attributes of sampled plants88Table 2.17Shade tolerance of sampled flora by life form90Table 2.18Moisture tolerance of sampled flora by life form91Table 2.19Percentage of classified taxa in surveyed microhabitats by provenance, habitat affinity, and moisture affinity92Table 3.1Dispersal modes of surveyed taxa by life form138Table 3.2Life history of surveyed taxa by life form138Table 3.3Provenance of surveyed taxa by dispersal mode and life form142Table 3.4Modality of surveyed taxa by dispersal mode and life form142Table 3.5Fruit type of surveyed herbs by dispersal mode and life form145Table 3.6Distribution of dispersal modes by taxonomic rank I (genus)150Table 3.7Distribution of dispersal modes by taxonomic rank II (family)151Table 3.8Distribution of dispersal modes by taxonomic rank III (order)152Table 3.9Dispersal modes of herbs by habitat affinity153Table 3.10Dispersal modes of herbs by shade tolerance156Table 3.12Distribution of herbs in 10m x 10m quadrats by environmental attribute and dispersal mode158Table 3.13Number of observed and expected herbs in surveyed microhabitats by158	Table 2.16	Proportion of total inertia in the dispersion of species scores (CCA)
Table 2.17Shade tolerance of sampled flora by life form90Table 2.18Moisture tolerance of sampled flora by life form91Table 2.19Percentage of classified taxa in surveyed microhabitats by provenance, habitat affinity, and moisture affinity92Table 3.1Dispersal modes of surveyed taxa by life form138Table 3.2Life history of surveyed taxa by dispersal mode and life form140Table 3.3Provenance of surveyed taxa by dispersal mode and life form142Table 3.4Modality of surveyed taxa by dispersal mode and life form145Table 3.5Fruit type of surveyed herbs by dispersal mode and life form145Table 3.6Distribution of dispersal modes by taxonomic rank I (genus)150Table 3.7Distribution of dispersal modes by taxonomic rank II (family)151Table 3.8Dispersal modes of herbs by habitat affinity153Table 3.10Dispersal modes of herbs by moisture affinity153Table 3.11Dispersal modes of herbs by shade tolerance156Table 3.12Distribution of herbs in 10m x 10m quadrats by environmental attribute and dispersal mode158Table 3.13Number of observed and expected herbs in surveyed microhabitats by158		explained by functional attributes of sampled plants
Table 2.18Moisture tolerance of sampled flora by life form91Table 2.19Percentage of classified taxa in surveyed microhabitats by provenance, habitat affinity, and moisture affinity92Table 3.1Dispersal modes of surveyed taxa by life form138Table 3.2Life history of surveyed taxa by dispersal mode and life form140Table 3.3Provenance of surveyed taxa by dispersal mode and life form142Table 3.4Modality of surveyed taxa by dispersal mode and life form142Table 3.5Fruit type of surveyed herbs by dispersal mode and life form145Table 3.6Distribution of dispersal modes by taxonomic rank I (genus)150Table 3.7Distribution of dispersal modes by taxonomic rank II (family)151Table 3.8Distribution of dispersal modes by taxonomic rank III (order)152Table 3.9Dispersal modes of herbs by moisture affinity153Table 3.10Dispersal modes of herbs by shade tolerance156Table 3.12Distribution of herbs in 10m x 10m quadrats by environmental attribute and dispersal mode158Table 3.13Number of observed and expected herbs in surveyed microhabitats by158	Table 2.17	Shade tolerance of sampled flora by life form
Table 2.19Percentage of classified taxa in surveyed microhabitats by provenance, habitat affinity, and moisture affinity92Table 3.1Dispersal modes of surveyed taxa by life form138Table 3.2Life history of surveyed taxa by dispersal mode and life form140Table 3.3Provenance of surveyed taxa by dispersal mode and life form142Table 3.4Modality of surveyed taxa by dispersal mode and life form142Table 3.5Fruit type of surveyed taxa by dispersal mode and life form145Table 3.6Distribution of dispersal modes by dispersal mode147Table 3.6Distribution of dispersal modes by taxonomic rank I (genus)150Table 3.7Distribution of dispersal modes by taxonomic rank II (family)151Table 3.8Distribution of dispersal modes by taxonomic rank III (order)152Table 3.9Dispersal modes of herbs by moisture affinity153Table 3.10Dispersal modes of herbs by shade tolerance156Table 3.12Distribution of herbs in 10m x 10m quadrats by environmental attribute and dispersal mode158Table 3.13Number of observed and expected herbs in surveyed microhabitats bv158	Table 2.18	Moisture tolerance of sampled flora by life form
habitat affinity, and moisture affinity92Table 3.1Dispersal modes of surveyed taxa by life form138Table 3.2Life history of surveyed taxa by dispersal mode and life form140Table 3.3Provenance of surveyed taxa by dispersal mode and life form142Table 3.4Modality of surveyed taxa by dispersal mode and life form142Table 3.5Fruit type of surveyed taxa by dispersal mode and life form145Table 3.6Distribution of dispersal modes by taxonomic rank I (genus)150Table 3.7Distribution of dispersal modes by taxonomic rank II (family)151Table 3.8Distribution of dispersal modes by taxonomic rank III (order)152Table 3.9Dispersal modes of herbs by habitat affinity153Table 3.10Dispersal modes of herbs by shade tolerance156Table 3.11Dispersal modes of herbs by shade tolerance156Table 3.12Distribution of herbs in 10m x 10m quadrats by environmental attribute and dispersal mode158Table 3.13Number of observed and expected herbs in surveyed microhabitats by158	Table 2.19	Percentage of classified taxa in surveyed microhabitats by provenance,
Table 3.1Dispersal modes of surveyed taxa by life form138Table 3.2Life history of surveyed taxa by dispersal mode and life form140Table 3.3Provenance of surveyed taxa by dispersal mode and life form142Table 3.4Modality of surveyed taxa by dispersal mode and life form145Table 3.5Fruit type of surveyed herbs by dispersal mode147Table 3.6Distribution of dispersal modes by taxonomic rank I (genus)150Table 3.7Distribution of dispersal modes by taxonomic rank II (family)151Table 3.8Distribution of dispersal modes by taxonomic rank III (order)152Table 3.9Dispersal modes of herbs by moisture affinity153Table 3.10Dispersal modes of herbs by shade tolerance156Table 3.12Distribution of herbs in 10m x 10m quadrats by environmental attribute and dispersal mode158Table 3.13Number of observed and expected herbs in surveyed microhabitats bv		habitat affinity, and moisture affinity
Table 3.2Life history of surveyed taxa by dispersal mode and life form140Table 3.3Provenance of surveyed taxa by dispersal mode and life form142Table 3.4Modality of surveyed taxa by dispersal mode and life form145Table 3.5Fruit type of surveyed herbs by dispersal mode147Table 3.6Distribution of dispersal modes by taxonomic rank I (genus)150Table 3.7Distribution of dispersal modes by taxonomic rank II (family)151Table 3.8Distribution of dispersal modes by taxonomic rank III (order)152Table 3.9Dispersal modes of herbs by habitat affinity153Table 3.10Dispersal modes of herbs by shade tolerance156Table 3.11Dispersal modes in 10m x 10m quadrats by environmental attribute and dispersal mode158Table 3.13Number of observed and expected herbs in surveyed microhabitats by	Table 3.1	Dispersal modes of surveyed taxa by life form
Table 3.3Provenance of surveyed taxa by dispersal mode and life form142Table 3.4Modality of surveyed taxa by dispersal mode and life form145Table 3.5Fruit type of surveyed herbs by dispersal mode147Table 3.6Distribution of dispersal modes by taxonomic rank I (genus)150Table 3.7Distribution of dispersal modes by taxonomic rank II (family)151Table 3.8Distribution of dispersal modes by taxonomic rank III (order)152Table 3.9Dispersal modes of herbs by habitat affinity153Table 3.10Dispersal modes of herbs by shade tolerance156Table 3.12Distribution of herbs in 10m x 10m quadrats by environmental attribute158Table 3.13Number of observed and expected herbs in surveyed microhabitats bv158	Table 3.2	Life history of surveyed taxa by dispersal mode and life form
Table 3.4Modality of surveyed taxa by dispersal mode and life form145Table 3.5Fruit type of surveyed herbs by dispersal mode147Table 3.6Distribution of dispersal modes by taxonomic rank I (genus)150Table 3.7Distribution of dispersal modes by taxonomic rank II (family)151Table 3.8Distribution of dispersal modes by taxonomic rank III (order)152Table 3.8Distribution of dispersal modes by taxonomic rank III (order)152Table 3.9Dispersal modes of herbs by habitat affinity153Table 3.10Dispersal modes of herbs by moisture affinity155Table 3.11Dispersal modes of herbs by shade tolerance156Table 3.12Distribution of herbs in 10m x 10m quadrats by environmental attribute and dispersal mode158Table 3.13Number of observed and expected herbs in surveyed microhabitats by158	Table 3.3	Provenance of surveyed taxa by dispersal mode and life form
Table 3.5Fruit type of surveyed herbs by dispersal mode147Table 3.6Distribution of dispersal modes by taxonomic rank I (genus)150Table 3.7Distribution of dispersal modes by taxonomic rank II (family)151Table 3.8Distribution of dispersal modes by taxonomic rank III (order)152Table 3.9Dispersal modes of herbs by habitat affinity153Table 3.10Dispersal modes of herbs by moisture affinity155Table 3.11Dispersal modes of herbs by shade tolerance156Table 3.12Distribution of herbs in 10m x 10m quadrats by environmental attribute158Table 3.13Number of observed and expected herbs in surveyed microhabitats by158	Table 3.4	Modality of surveyed taxa by dispersal mode and life form
Table 3.6Distribution of dispersal modes by taxonomic rank I (genus)150Table 3.7Distribution of dispersal modes by taxonomic rank II (family)151Table 3.8Distribution of dispersal modes by taxonomic rank III (order)152Table 3.9Dispersal modes of herbs by habitat affinity153Table 3.10Dispersal modes of herbs by moisture affinity155Table 3.11Dispersal modes of herbs by shade tolerance156Table 3.12Distribution of herbs in 10m x 10m quadrats by environmental attribute158Table 3.13Number of observed and expected herbs in surveyed microhabitats by158	Table 3.5	Fruit type of surveyed herbs by dispersal mode
Table 3.7Distribution of dispersal modes by taxonomic rank II (family)151Table 3.8Distribution of dispersal modes by taxonomic rank III (order)152Table 3.9Dispersal modes of herbs by habitat affinity153Table 3.10Dispersal modes of herbs by moisture affinity155Table 3.11Dispersal modes of herbs by shade tolerance156Table 3.12Distribution of herbs in 10m x 10m quadrats by environmental attribute158Table 3.13Number of observed and expected herbs in surveyed microhabitats by	Table 3.6	Distribution of dispersal modes by taxonomic rank I (genus) 150
Table 3.8Distribution of dispersal modes by taxonomic rank III (order)152Table 3.9Dispersal modes of herbs by habitat affinity153Table 3.10Dispersal modes of herbs by moisture affinity155Table 3.11Dispersal modes of herbs by shade tolerance156Table 3.12Distribution of herbs in 10m x 10m quadrats by environmental attribute158Table 3.13Number of observed and expected herbs in surveyed microhabitats by	Table 3.7	Distribution of dispersal modes by taxonomic rank II (family)
Table 3.9Dispersal modes of herbs by habitat affinity153Table 3.10Dispersal modes of herbs by moisture affinity155Table 3.11Dispersal modes of herbs by shade tolerance156Table 3.12Distribution of herbs in 10m x 10m quadrats by environmental attribute158Table 3.13Number of observed and expected herbs in surveyed microhabitats by	Table 3.8	Distribution of dispersal modes by taxonomic rank III (order)
Table 3.10Dispersal modes of herbs by moisture affinity155Table 3.11Dispersal modes of herbs by shade tolerance156Table 3.12Distribution of herbs in 10m x 10m quadrats by environmental attribute and dispersal mode158Table 3.13Number of observed and expected herbs in surveyed microhabitats by	Table 3.9	Dispersal modes of herbs by habitat affinity
Table 3.11Dispersal modes of herbs by shade tolerance156Table 3.12Distribution of herbs in 10m x 10m quadrats by environmental attribute and dispersal mode158Table 3.13Number of observed and expected herbs in surveyed microhabitats by	Table 3.10	Dispersal modes of herbs by moisture affinity
Table 3.12Distribution of herbs in 10m x 10m quadrats by environmental attribute and dispersal mode	Table 3.11	Dispersal modes of herbs by shade tolerance
and dispersal mode158Table 3.13Number of observed and expected herbs in surveyed microhabitats by	Table 3.12	Distribution of herbs in 10m x 10m quadrats by environmental attribute
Table 3.13 Number of observed and expected herbs in surveyed microhabitats by		and dispersal mode
	Table 3.13	Number of observed and expected herbs in surveyed microhabitats by

LIST OF TABLES

	Page
	dispersal mode
Table 3.14	Difference in mean percent of herbs (n=234) by mode of dispersal in
	contrasting microhabitats within 10m x 10m quadrats: paired samples 177
Table 3.15	Mean percent of herbs (n=234) by mode of dispersal in contrasting
	microhabitats within 10m x 10m quadrats: independent samples
Table 3.16	Dispersal modes of herbs (n=252) by frequency class
Table 3.17	Dispersal modes of herbs (n=252) by cover class
Table 3.18	Dispersal modes of herbs (n=252) by species richness class
Table 3.19	Dispersal modes of herbs (n=252) by patch size class
Table 3.20	Dispersal modes of herbs (n=252) by patch isolation class
Table 4.1.	Environmental correlates of species richness in 10m x 10m quadrats
Table 4.2	Species richness (10m x10m quadrats) of soil parent materials, soil
	orders, soil moisture classes, forest cover types and disturbance classes 231
Table 4.3	Selected correlations involving # tree species. % stems sugar maple.
	and # tree stems 0-4 cm dbh 237
Table 4.4	Contribution of microhabitats to species richness in 10m x 10m quadrats 250
Table 4.5	Contribution of microhabitats to species richness in surveyed forest patches 252
Table 4,6	Contribution of microhabitats to species richness at the landscape scale 254
Table 4.7	Comparison of microhabitats by moisture and canopy closure class
Table 4.8	Plant trait correlates of species richness in 10m x 10m quadrats
Table 4.9	Comparison of selected GLM models of species richness
Table 4.10	Summary of graphical evaluation of leading models of species richness 275
Table 4.11	Explanatory variables included in leading models of species richness
Table 4.12	Contribution of phylogeny to superior models of species richness
Table 4.13	Proportion of taxa with selected plant attributes at progressively more
	inclusive taxonomic ranks

LIST OF FIGURES

	Page
Figure 2.1	Distribution of forest patches (N=24) in relation to DCA axes 1 and 2 64
Figure 2.2	Species ordination (CCA) constrained by environmental variables
Figure 2.3	Partial decomposition of variance in CCA species ordination: local
	versus regional processes
Figure 2.4	Distribution of microhabitats in relation to DCA axes 1 and 2
Figure 2.5	DCA ordination of open, seasonally dry, canopy gaps
Figure 2.6	Distribution of species among microhabitats
Figure 2.7	Sugar maple abundance versus calcium in upper 15 cm of soil profile on
	Brunisolic and Luvisolic soils overlying calcareous till
Figure 2.8	Sugar maple abundance versus calcium in upper 15 cm of soil profile on
	Brunisolic soils overlying calcareous till
Figure 2.9	Herb response to increasing sugar maple abundance and available
	calcium on Brunisolic and Luvisolic soils overlying calcareous till
Figure 2.10	Mean calcium affinity and shade tolerance of understory plants in relation to
	flowering phenology
Figure 3.1	Distribution of herbs dispersed by animal ingestion and by animal adhesion
	in relation to DCA axes 1 and 2
Figure 3.2	Distribution of herbs dispersed by ants and wind in relation to DCA axes
	1 and 2
Figure 3.3	Distribution of herbs dispersed by prolonged dormancy in the soil and by
-	mechanical expulsion in relation to DCA axes 1 and 2
Figure 3.4	Distribution of herbs dispersed by unassisted means and by multiple modes
-	in relation to DCA axes 1 and 2
Figure 3.5	Distribution of herbs dispersed by vegetative expansion and by selected
-	animal agents in relation to DCA axes 1 and 2
Figure 3.6	Distribution of dispersal modes of herbs in relation to environmental
-	variables
Figure 4.1	Scatter plots of selected correlates of species richness in 10m x 10m
-	quadrats I
Figure 4.2	Scatter plots of selected correlates of species richness in 10m x 10m
-	quadrats []
Figure 4.3	Scatter plots of selected correlates of species richness in 10m x 10m
-	quadrats III
Figure 4.4	Scatter plots of selected correlates of species richness in 10m x 10m
•	quadrats IV
Figure 4.5	Species richness, sugar maple abundance, and available calcium, in 10m
~	x 10m guadrats on Brunisolic and Luvisolic soils overlying calcareous till 241
Figure 4.6	Species richness, sugar maple abundance, and available calcium, in
<u>.</u>	10m x 10m quadrats on Brunisolic soils overlying calcareous till
Figure 4.7	Comparison of mean plant response in 10m x 10m quadrats on soils
	of contrasting fertility I

LIST OF FIGURES

		Page
Figure 4.8	Comparison of mean plant response in 10m x 10m quadrats on soils	
	of contrasting fertility II.	. 245
Figure 4.9	Comparison of mean plant response in 10m x 10m quadrats on soils	
	of contrasting fertility III.	246
Figure 4.10	Scatter plots of dispersal correlates of species richness in 10m x 10m	
	quadrats I	264
Figure 4.11	Scatter plots of dispersal correlates of species richness in 10m x 10m	
	quadrats []	265
Figure 4.12	Representative results from graphical evaluation of leading GLM Models	
	of species richness	277
Figure 4.13	Graphical evaluation of superior GLM models of species richness	278

1.0 INTRODUCTION

The question, "Why do some places have more species than others?", has attracted attention throughout the history of ecology. One reason for this is that the answer one gives both reflects, and informs, one's understanding of the fundamental processes that govern the distribution and abundance of organisms. A core issue since the late 1950's has been the identification of processes that allow numerous species to coexist in the same environment. Classical models of species interactions (Grinnell 1904, Volterra 1926, Gause 1934) predict that two similar species cannot coexist indefinitely on a single limiting resource in a uniform environment. In the presence of such conditions, one species should eventually displace the other and the assemblage should be reduced to a single species (Hutchinson 1957, 1959; Hardin 1960). How is it, then, that so many natural habitats are species-rich?

One productive approach to this question has been to examine the assumptions of the classical models: what happens if species are not "similar", if interactions do not proceed to equilibrium, if there is more than one limiting resource, or, if the environment is not spatially or temporally uniform. This research effort has generated a vast literature and many alternative explanations (see Chapter 4 and reviews by Connell 1978, Huston 1979, Pickett 1980, Sousa 1984, Petraitis *et al.* 1989, Hart and Horowitz 1991, Tilman and Pacala 1993, Ricklefs and Schluter 1993, Huston 1994, Palmer 1994, Heywood 1995, Grace 1999). Despite this effort, a synthetic understanding that applies at all spatial and temporal scales has not been achieved.

In terrestrial plant communities, a broad consensus has emerged that the environment is not spatially and temporally uniform. Under these conditions, heterogeneity in limiting resources is expected to promote species coexistence by increasing the probability that there will be some place, or time, where one's competitors perform poorly or do not survive and where populations of one's own kind may expand (Hutchinson 1961, Levin 1974, Warner and Chesson 1985, Comins and Noble 1985, Hurtt and Pacala 1995). Implicit in this perspective is the expectation that differences in plant traits will lead to pattern in the distribution of species and variability in the composition and dynamics of plant assemblages (Whittaker 1956, Grubb 1977, Tilman 1982, Chesson 1986). The spatial and temporal scales at which heterogeneity contributes to species diversity, and to species coexistence, have not been resolved (cf. Pacala and Silander 1985, Ricklefs 1987, Cornell and Lawton 1992, Holmes and Willson 1998, Cain *et al.* 2000). A long-standing presumption has been that interactions among local species and the physical environment are the principal means by which plant and animal assemblages are structured (Ricklefs 1987). If true, then the composition and richness of a given assemblage may be explained solely by reference to processes operating within the local patch. The diversity of species, however, often fails to converge under similar conditions, suggesting that regional and historical processes, as well as unique events and circumstances, are important contributors to community structure (Ricklefs 1987, Cornell and Lawton 1992, Ricklefs and Schluter 1993).

One spatial process that is expected to influence the composition and richness of plant assemblages is dispersal. Dispersal is predicted to have profound consequences for populations and communities since it governs the size and composition of the seed rain (Clark and Yi 1995), affects the probability that diaspore will land in a site suitable for germination (Harper 1977, Sorensen 1978, Venable and Levin 1985), determines the initial conditions that seeds and seedlings must contront (Schupp and Fuentes 1995), affects the initial spatial array of individuals in a population (Thiede and Augspurger 1995), determines who interacts with whom and with what intensity (Schmida and Ellner 1984, Pacala 1986, Silander and Pacala 1990, Rees 1996, Rees *et al.* 1996), influences local extinction rates by affecting the probability that declining or extirpated populations are rescued (Brown and Kodric-Brown 1977, Holt 1993), influences the rate at which plants colonize new habitat (Halpern *et al.* 1990, Matlack 1994, Kotanen 1997, Brunet and von Oheimb 1998) and the sequence in which they arrive (Drake 1991, Fastic 1995), and, influences the fevel of gene flow within and between populations and thus the degree to which neighboring plants are related (Williams and Guries 1994) and genetic variation is structured spatially (Levin 1981, Hamrick and Godt 1997, Hamrick *et al.* 1993).

The contribution of dispersal to observed differences in the composition and richness of individual habitats and plant assemblages, however, is poorly understood. One reason for this is the logistic

challenge of monitoring the dispersal of seeds and spores. Diaspores are typically released through time and often by means and over distances that cannot be readily observed. Despite a concerted effort to determine the dispersal reach of species (Appendix 11), uncertainty remains regarding the proportion of diaspores that land beyond the immediate vicinity of the maternal plant (Portnoy and Willson 1993) and regarding the frequency and importance of longer-distance dispersal events (Cain *et al.* 2000).

One way forward has been the use of indirect measures, such as the proportion of established taxa dispersed by a given mode, to characterize the dispersal spectra associated with particular habitats (Dansereau and Lems 1957, Frenkel 1970, Pojar 1974, Lufensteiner 1979, Ellner and Shmida 1981, Hoehne 1981, Morton and Hogg 1989, Willson *et al.* 1990) and plant traits (Westoby *et al.* 1990, Hughes *et al.* 1994, Leishman *et al.* 1995, Mabry *et al.* 2000). The data from these studies, while limited, have found pattern in the relative frequency of dispersal modes and broad similarities in the dispersal spectra of similar habitats. In the temperate forests of eastern North America, dispersal by animal ingestion, unassisted means, and wind is typically more frequent than dispersal by ants, animal adhesion and mechanical expulsion (Table 1.1.). The taxa of wetland and disturbed habitats, in contrast, may typically be dispersed by the wind.

The use of indirect measures, however, makes it difficult to determine whether the pattern in the distribution and composition of species is due to the failure of seeds to land there (dispersal limitation), germinate there (recruitment limitation), or, persist there (survival limitation) (Schupp and Fuentes 1995). This creates a measure of uncertainty with respect to causation and ambiguity with respect to the contribution of plant traits that may be correlated with dispersal. Pattern in the dispersal attributes of established plant assemblages, therefore, may be inherently ambiguous with respect to the mechanisms that give rise to it.

Those who wish to examine the contribution of dispersal at the scale of habitats and landscapes reluctantly accept such ambiguity in the expectation that pattern revealed through the use of indirect methods will provide a starting point for future research with more revealing methods.

Table 1.) Comparison of dispersal spectra in plant communities of broadly similar structure in eastern North America. Cell entries = mean percent of flora (references 1, 2, 3, this study); median percent of flora (reference 4). n = number of stands. Modes of dispersal: Al=animal ingestion, AA=animal adhesion, AC=animal conveyance (ants), A(s.i.)=animal dispersed; MI(=mechanical expulsion, U=unassisted, O=other (man). Sources: 1. Dansereau and Lems 1957; 2. Hoebne 1981; 3. Morton and Hogg 1989; 4. Willson *et al.* 1990. See original sources for details regarding size and attributes of the sampled flora.

	Mode of Dispersal								
Community	AI	AA	AC	A (s.l.)	Wind	ME	U	Water	0
Deciduous Forests:									
Illinois, Michigan, Tennessee4 (n=4)	29	7	10	46	18	3	36	-	-
Quebec: climax' (n=3)	-	-	-	39	43	-	17	-	-
Quebec: disturbed ¹ (n=3)	-	-	-	26	62	-	12	-	-
Wisconsin: forest islands 1951 ² (n=4)	23	15	3	4]	15	4	19	5	-
Wisconsin: forest islands 1975 ¹ (n-4)	26	9	11	46	3	3	35	6	-
This Study (n=24)	17	13	10	40	31	5	22	-	-
Conifer Forests:									
Minnesota ¹ (n=1)	22	3	7	32	31	2	34	-	-
Quebec ¹ (n=1)	-	-	-	29	71	-	-	-	-
Island Flora:		·	•	•					
Ontario ² (n=1)	42	24	-	64	23		-	63	6

		•
a (cont'd).		
rth Americ		
častern Nor	ıl	
aucture in 1	of Disperse	
r similar str	Mode	
s of broadly		
ommunities		
t in plant c		
Table 1.1 Comparison of dispersal spectre		Community

Community AI AA Old Field - - Quebec ¹ (n=1) - -	al spectra in plant communities							
Community AI AA Old Field - - Quebec ¹ (n=1) - -			Mode	of Dispersa	II			
Old Field Old Field - Quebee' (n=1) - Bluegrass Meadow - Quebee' (n=1) - Alder Thicket - Quebee' (n=1) -	VV IV	АС	A (X,L)	Wind	ME	n	Water	0
Quebec ¹ (n=1) - - - Bluegrass Meadow - - - Quebec ¹ (n=1) - - - Alder Thicket - - - Quebec ¹ (n=1) - - -								
Bluegrass Meadow Quebee ¹ (n=1)	•	ı	17	83	-	·	1	,
Quebee ¹ (n=1) · · · · · · · · · · · · · · · · · · ·					:			
Alder Thicket Quebec' (n=1) · · ·		-	13	87	-		ı	•
Quebec' (n=1)								
	•	1	40	60	-		'	,
Fen								
Quebee ¹ {13=1}	•	ı	2	83	•	<u>۰</u>	-	,

The principal motivation for this thesis has been to gain a clearer understanding of the roles of dispersal and environmental heterogeneity in structuring the composition and richness of species in the forest understory. The forests in this study are isolated fragments of the pre-settlement forest of the Huron-Ontario Section of the Great Lakes – St. Lawrence Forest Region (Rowe 1972). The effects of forest fragmentation are considered only briefly, however, since the methods of the thesis cannot readily distinguish their effects from the effects of within-patch processes.

Objectives of the Thesis

- To characterize the dispersal profile of established plants in the understory of sampled forest patches;
- To identify the environmental factors and plant traits that best explain observed differences in the composition and richness of established plants in the understory of sampled forest patches;
- iii) To compare the degree to which environmental factors and plant traits explain observed differences in species richness in the understory of sampled forest patches.

General Approach

In keeping with recent studies, the principal method for investigating the influence of dispersal on plant assemblages will be the comparison of the proportion of taxa dispersed by a given mode in contrasting habitats and environmental states. Ambiguity with respect to causal factors remains but has been minimized by recording the distribution of species in relation to uniform microhabitats and by considering the contribution of associated plant traits that may independently influence the distribution of species or constrain the mode of dispersal. The latter method draws on long-standing initiatives to identify and evaluate pattern in relation to species groups with similar structural or functional traits (e.g. Raunkiaer 1934, Root 1967, MacArthur and Wilson 1967, Grime 1977, Noble and Slayter 1980, Willson *et al.* 1990, Leishman and Westoby 1992, Smith *et al.* 1997).

The first objective of the thesis was achieved by conducting a spring and summer survey of twentyfour forest patches in the general vicinity of Peterborough. Ontario. and. by characterizing the apparent mode of dispersal of 413 species of vascular plants in relation to the morphology and known properties of the diaspore. Plant traits that may independently influence the distribution of species or modes of dispersal were also characterized with reference to published sources.

The vegetation survey was conducted in relation to a field-based classification of microhabitats that characterized uniform conditions within each 10m x 10m quadrat with respect to canopy closure, soil moisture, substrate, and disturbance. This approach provided insight into the scale of spatial heterogeneity within the forest and the degree to which this heterogeneity was associated with differences in species composition, richness, and functional traits.

The second objective of the thesis was achieved by collecting soil samples and field data to characterize the moisture and fertility of forest soils, the composition and structure of forest trees, and site disturbance. The base cation status, soil pH and percent soil organic matter were determined by laboratory analysis; percent canopy closure was determined by hemispherical photography. Pattern in the composition and distribution of species and plant traits was evaluated by detrended correspondence analysis (DCA) and canonical correspondence analysis (CCA). The degree to which environmental variables and plant traits explained differences in species richness in 10m x 10m quadrats was determined by simple linear regression.

The third objective of the thesis was achieved by multiple linear regression and detrended correspondence analysis (DCA). The latter analysis, suggested by Dr. R. Hansell. Institute of Environmental Studies. University of Toronto, provided a graphical method for determining the degree to which environmental and dispersal variables accounted for the same fraction of variance summarized by multiple linear regression.

Structure of the Thesis

Chapter 2 describes the sampling protocol and methods associated with the vegetation survey and

environmental inventory; summarizes the relationships among key environmental variables; and, identifies the environmental factors and plant traits that influence the distribution and composition of species in the forest understory.

Chapter 3 describes the procedure for classifying modes of dispersal and plant traits; and, identifies patterns in the distribution of dispersal modes in relation to life form. life history, provenance, modality, fruit type, taxonomic rank, habitat affinity, environmental gradients, microhabitats, species frequency class, species richness class, plant cover class, patch size, and, patch isolation.

Chapter 4 describes the methods for identifying and comparing the contribution of variables to species richness; and, identifies the degree to which environmental variables and plant traits individually, and collectively, explain observed differences in species richness in 10m x 10m quadrats.

Chapter 5 presents the principal findings and conclusions of the thesis.

Chapter 6 summarizes the cited literature.

Appendices: Supporting information original to the thesis is presented in Appendices 1 to 10. Contextual information compiled from secondary sources is presented in Appendices 11 to 14.

2.0 ENVIRONMENTAL PATTERNS

2.1 Introduction

Spatial and temporal heterogeneity in conditions that govern the germination, establishment, growth and reproduction of plants is a striking feature of natural environments. In the presence of heterogeneity, differences in plant traits are expected to create pattern in the distribution of species and variability in the composition and dynamics of plant assemblages (Whittaker 1956, Levin 1974, Grubb 1977, Tilman 1982, Chesson 1986). Variation in the availability of limiting resources is expected to promote species coexistence by increasing the probability that there will be some place, or time, where competitors perform poorly or do not survive and where populations of low abundance may expand (Hutchinson 1961, Levin 1974, Warner and Chesson 1985, Comins and Noble 1985, Hurtt and Pacala 1995). Coexistence under these circumstances requires that species be ecologically distinct since the populations of rare species cannot otherwise expand in the presence of species that are more abundant (Chesson 1991, Pacala and Tilman 1994).

In the temperate forests of eastern North America, environmental heterogeneity may be created at a variety of spatial and temporal scales. Influential processes that operate at large spatial or temporal scales include glaciation (Hills 1952. Peet and Christensen 1980, Howell and Vankat 1981), pedogenesis (McFadden *et al.* 1994. Reich *et al.* 1997), plant migration (Davis 1981a), succession (Glenn-Lewin *et al.* 1992), forest stand development (Oliver and Larson 1996), periodic outbreaks of pathogens (Davis 1981b, Perry and Moore 1987), periodic storm events (Oliver and Stephens 1977, Frelich and Lorimer 1991), periodic extremes in precipitation and temperature (Tilman and Downing 1994), periodic fire events (Nowacki *et al.* 1990, Reich *et al.* 1990), chance colonization events and sequences (Drake 1991, Fastie 1995), vegetation and soil feed-back processes (Watt 1947, Fox 1977, Pastor *et al.* 1984, Frelich *et al.* 1993), and, past land use practices (Pettit *et al.* 1995, Motzin *et al.* 1996).

In contrast, processes that operate at small spatial scales, and over short time frames, are typically plant-based processes such as the uptake of moisture and nutrients (Tilman 1982, 1988; Roberston *et al.* 1988), dispersal (Levin 1976, Roughgarden 1977, Pacala 1987, Portnoy and Willson 1993).

germination (Grubb 1977, 1986), recruitment (Hurtt and Pacala 1995), the creation of space through plant death (Chesson and Warner 1981, Tilman 1994), competitive interactions (Goldberg and Barton 1992, Gurevitch *et al.* 1992, Bengtsson *et al.* 1994), and, herbivory (Whitney 1984).

In the northern hardwood forest, an important source of heterogeneity is the forest canopy. Canopy trees influence the quantity, spectral quality and phenology of light received at the forest floor (Minkler and Woerhiede 1965, Horn 1971, 1975, Brewer 1980, Messier and Bellefleur 1988, Canham *et al.* 1990, Canham and Burbank 1994), the availability of nutrients and moisture in the rooting zone of forest soils (Aber *et al.* 1991, Pastor and Post 1986, Zinke 1962, Crozier and Boerner 1984, 1986, Leininger and Winner 1988, Boerner and Koslowsky 1989), the quantity and quality of coarse woody debris on the forest floor (Harmon *et al.* 1986, Hale and Pastor 1998), the probability and size of tree pits created during wind-throw events (Putz *et al.* 1983), and the timing and size of gaps in the forest canopy (Lorimer *et al.* 1988, Lorimer 1989, Frelich and Lorimer 1991). Mechanisms that contribute to these effects include differences in canopy architecture (Horn1971), seasonal patterns of development and senescence (Brewer 1980), tissue and leaf litter chemistry (Vitousek 1982, Melillo *et al.* 1982), mechanical firmness (Mergen 1954), response to wind-throw (Beatty and Stone 1986), and life span (Lorimer 1989).

The spatial scale at which heterogeneity contributes to species coexistence is influenced by the dispersal properties of potentially competing plants. For most plants, the fraction of propagules dispersed beyond the immediate vicinity of the parent declines sharply with distance, creating seed dispersal curves with long tails (Portnoy and Willson 1993). Most propagules travel only metres to tens of metres from the parent plant and the most important dispersal outcome may be achieved within 1-2 canopy diameters of the maternal parent (Appendix 11: Hughes *et al.* 1994). At this distance, dispersal is a non-limiting process and pattern in the distribution and composition of species is governed primarily by factors governing germination, establishment and persistence.

Recent studies suggest that dispersal over this distance may be an important precondition for the coexistence of species (Atkinson and Shorrocks 1981: Shmida and Ellner 1984; Pacala 1986, 1987: Pacala and Silander 1987; Rees *et al.* 1996). Rather than interact with a number of individuals in

a plant assemblage, and thereby experience the average density of the population, plants interact primarily with individuals that lie within a canopy or root crown diameter (Harper 1977, Pacala and Silander 1985. Venable and Brown 1993). In the forest understory, this means that the distance over which most plants competitively interact is on the order of centimetres to metres. When the dispersal distance of plants is short, siblings tend to aggregate into monospecific clumps and competing species become segregated spatially. Under these circumstances, individuals tend to compete more with their own kind than with others and thus create the primary conditions for coexistence(Schmida and Ellner 1984, Pacala 1987, Lavorel *et al.* 1994).

When the dispersal reach of plants extends beyond the neighborhood in which plants compete, an increasing number of seeds land in environments that are less favorable than the home patch, and, a declining number of conspecific seeds land in close proximity to one another. These conditions reduce the tendency for monospecific clumping by favoring small founding populations and low fecundity. Under these circumstances, coexistence is facilitated by the spatial segregation of competitors when seeds land in empty but suitable environments (Hurtt and Pacala 1995, Holmes and Wilson 1998), and, by reversals in relative competitive strength when more widely dispersed seeds germinate next to weaker competitors or to stronger competitors weakened by less favorable settings (Chesson and Case 1986).

Pattern in the response of species to environmental variation has been examined at a variety of spatial and temporal scales. One pattern that has attracted increasing attention in recent years is the response of forest plants on contrasting geological substrates (Curtis 1959, Buell *et al.* 1966, Peet and Christensen 1980, Howell and Vankat 1981, Pregitzer and Barnes 1982, Pastor *et al.* 1984, Zak *et al.* 1986, Host *et al.* 1988, Palmer 1990, Whitney 1991, Host and Pregitzer 1992, McFadden *et al.* 1994. Kobe *et al.* 1995, Kobe 1996, Reich *et al.* 1997, van Breeman *et al.* 1997, Hutchinson *et al.* 1999). Marked contrasts in the composition and richness of species observed within the forest canopy, and on the forest floor, have typically been attributed to gradients in moisture and fertility associated with differences in the bedrock, soil parent material, and degree of soil weathering (soil order).

The relative importance of moisture and nutrients to differences in forest composition and productivity is a long-standing issue that has been critically re-examined in several recent papers (cf. Pastor *et al.* 1984, Pastor and Post 1986; Reich *et al.* 1997; Kobe *et al.* 1995, Kobe 1996, van Breeman *et al.* 1997). This research has contributed to a rethinking of the relative importance of plant and soil based processes to nutrient dynamics in north temperate forests, and, has stimulated research into the relative importance of nitrogen and base cations to the distribution and relative abundance of important canopy trees such as sugar maple.

The calcium-based explanations proposed by Kobe *et al.* (1995) and Kobe (1996) to account for the marked decline in juvenile sugar maple mortality on calcium-rich soils in northwestern Connecticut (see Section 2.4) have broad phytogeographic and ecological importance since they provide a mechanistic basis for the predominance of sugar maple in the Maple-Basswood, and Beech-Maple, forest regions (Braun 1950), and, for the greater shade tolerance of sugar maple, and red spruce (McLaughlin *et al.* 1991), on calcium-rich soils.

Study Objectives: The principal objective of this chapter is to identify the environmental factors that influence the distribution and composition of plants in the understory of sampled forest patches. Plant response is examined in relation to inferred gradients in limiting resources and to apparent microhabitats in the forest understory. Pattern in the distribution of dispersal modes in relation to microhabitats and environmental gradients will be examined in Chapter 3. The contribution of environmental and dispersal factors to species diversity will be compared in Chapter 4.

2.2 Methods

2.2.1 Selection of Study Sites

Twenty-four forest patches (Appendix 1) were chosen to reflect a range of patch sizes in two landscapes of contrasting forest fragmentation (Table 2.1). The forest patches were second-growth deciduous stands in the vicinity of Peterborough. Ontario. Sampled areas within patches were standardized, where possible, with respect to topographic relief, soil texture, soil moisture, forest cover type, and disturbance. The intent was to sample upland sugar maple stands on sites with low

	Landscape [•• -	Landscape II				
(159	% forest cover)	a	(35% forest cover) ¹				
Size Class (ha) ²	# Patches	% (N=12)	Size Class (ha) ² # Patches % (N=12)				
10-20	1	8.3	10-20	2	16.7		
21-50	3	25.0	21-50	i	8.3		
51-100	5	41.7	51-100	3	25.0		
101-500	3	25.0	101-500	5	41.7		
>500	0	0	>500	1	8.3		
Total	12	100.0	Total	12	100.0		

Table 2.1. Size distribution of sampled forest patches.

Notes:

1. M. Austen, Long Point Bird Observatory, pers. com. Digitized forest cover map derived from LANDSAT TM satellite imagery.

 Source: Forest stand maps (1:10.000 scale). Forest Resource Inventory. Ontario Ministry of Natural Resources (1978, 1979, 1980). topographic relief (< 2%), mesic soils, sand or loam textures, and low disturbance. In practice, the moisture regime of sampled areas varied with soil parent material, landscape position, and microtopography, and therefore ranged from dry-mesic to wet.

The chosen patches, with one exception, were selected from among sixty-three forest patches previously surveyed by the author (1995) for the Woodlands Biodiversity Project. Long Point Bird Observatory (Austen *et al.*, unpublished). The study sites established by the Long Point Bird Observatory (one circular quadrat, 35.4 m radius, per patch) were accepted for this study. The exceptions were study sites O9 and GG1, which were moved to more level terrain to satisfy the relief criterion for this study, and study site DD1, which was established *de novo* to balance the sampling design for soil texture. The protocol for establishing quadrats within study sites is presented below (Section 2.2.2).

The study sites are situated within the Huron-Ontario Section of the Great Lakes-St. Lawrence Forest Region (Rowe 1972), also known as the Great Lake Section of the Hemlock-White Pine-Northern Hardwoods Forest Region (Braun 1950). The forest patches have developed on sand and foam soils of the Brunisolic, Luvisolic and Gleysolic orders, overlying glacio-fluvial, calcareous till, calcareous outwash, or lacustrine parent materials, on Ordovician limestone of the Trenton formation (Weber and Morwick 1946, Gillespie and Acton 1981, Hoffman and Acton 1974). The climate of the region is humid, mesothermal, with little or no water deficiency (Energy, Mines and Resources Canada 1990a,b). The mean annual temperature (1964-1990) at Peterborough is 6.0 °C, with a mean daily temperature of -9.4 °C in January and 20.0 °C in July. The mean annual precipitation is 882.2 mm. (Environment Canada 1993).

2.2.2 Sampling Methods

Eight 10m x 10m quadrats were located within each 0.4 ha study site in relation to a stratified random sampling design. At each study site, a sketch map was prepared showing the location and estimated area of each microhabitat (see Microhabitat Variables, Section 2.2.3.5). Quadrats were located at random within each microhabitat in relation to the following sampling design: for

extensive microhabitats, such as closed, seasonally dry. forest floors, quadrats were located with reference to coordinates (bearing and pacing from the site centre) drawn from a random number table; for small microhabitats (features less than 10m in each direction), such as tip-up mounds, tree pits, seeps, and most canopy gaps, quadrats were centred on the feature which was chosen at random from a numbered set of such features; for linear microhabitats (features longer than 10m in one direction), such as seepage tracks, access lanes and associated ditches, quadrats were centred on the feature at a random number of paces initiated from the feature's edge. Stumps and logs were sampled passively in relation to the microhabitats in which they were found. The number of quadrats located in a given microhabitat was proportional to its area within the quadrat, subject to the constraint that each microhabitat type be sampled at least once.

Quadrats were located within study sites (rather than within forest patches) in order to clarify the contribution of environmental heterogeneity to species richness within a sampling space of arbitrary but constant area (0.4 ha). A sample area of constant size was used to avoid passive sampling effects which inflate species richness when the sampled area is proportional to patch size (Connor and McCoy 1979).

2.2.3 Environmental Variables

The environmental variables for this study are summarized in Table 2.2. Brief descriptions of the rationale for inclusion, criteria for classification, field or laboratory methods, and sources for published data sets, are presented below. Data collected in the field but not included in analyses are summarized in Section 2.2.3.7.

2.2.3.1 Soil Variables

Soil Parent Material: an indicator of the composition, sorting, and stratification of soil materials (Flint 1971; Bloom 1978); in this study, an indicator of regional differences in the inherent moisture retention and fertility of sampled soils. The inferred order of moisture retention capacity was lacustrine>calcareous till>calcareous outwash>glacio-fluvial, based on the declining abundance of fine particles. This ordering of parent materials also reflected inherent differences in soil fertility.

Variable	Attribute	Source
soil parent material	 glacio-fluvial calcareous tili calcareous outwash lacustrine 	Weber and Morwick (1946) Gillespie and Acton (1981) Hoffman and Acton (1974)
soil order	 brunisol gleyed brunisol luvisol gleyed tuvisol gleysol 	Weber and Morwick (1946) Gillespie and Acton (1981) Hoffman and Acton (1974) field observation
soil moisture class	 seasonally wet depressions seasonally moist depressions seasonally dry depressions 	tield observation
soil organic matter	continuous variable	laboratory observation
soil pH	continuous variable	laboratory observation
base cations	 available calcium calcium:magnesium ratio potassium:magnesium ratio 	laboratory observation
tree diameter class, breast height	 1. 0-30 cm: 2 cm increments 2. > 30.0 cm: continuous values measured to nearest 0.1 cm 	field observation
tree height class	1. < 1 m 2. 1-3 m 3. 3-10 m 4. 10-15 m 5. 15-25 m 6. > 25 m	field observation
forest stratum	1. herb layer (<1 m) 2. shrub layer (1-3 m) 3. subcanopy (3-15m) 4. canopy (>15 m)	field observation

Table 2.2. Summary of environmental variables.

Variable	Attribute	Source
canopy cover type	 oak, no sugar maple oak + sugar maple sugar maple sugar maple + wet-mesic or wet tree taxa wet-mesic + wet tree taxa 	field observation
canopy closure	continuous variable (%)	field observation
cover class (herb layer)	 solitary individual (~) 2-5 individuals (r) <1% 1-5% 6-15% 16-25% 26-50% 51-75% 76-100% 	field observation
habitat affinity	1. forest 2. forest – open 3. open + forest 4. open	Voss (1972, 1985, 1996)
coefficient of wetness	 -5 obligate wetland -4 facultative wetland -3 facultative wetland -2 facultative wetland -1 facultative 0 facultative +1 facultative +2 facultative upland +3 facultative upland +4 facultative upland +5 obligate upland 	Oldham <i>et al.</i> (1995)
microhabitat type	see Table 2.4	field observation
# microhabitats	continuous variable	field observation

Table 2.2 Summary of environmental variables (cont'd).

Variable	Attribute	Source
patch interior area class	1. 0-5 ha 2. 6-10 ha 3. 11-20 ha 4. 21-50 ha 5. 51-100 ha 6. 101-200 ha 7. 201-500 ha 8. 501-1000 ha	M. Austen, pers. com.
patch area	continuous variable	Ontario Ministry of Natural Resources (1978, 1979, 1980)
patch isolation	continuous variable	NTS map sheets (1:50.000)
forest cover	continuous variable	NTS map sheets (1:50,000)

Table 2.2 Summary of environmental variables (cont^{*}d).

based on an inferred decline in cation exchange capacity (Brady 1990) and soil pH (Dancer *et al.* 1993). However, soils on the lacustrine parent materials were saturated seasonally and thus subject to periodic denitrification (Ponnamperuma 1972) and to lower rates of ammonification (Patrick and Mahaptera 1968, Ponnamperuma 1972). The presumed order of fertility, based on the joint influence of contributing factors, was calcareous till ≥calcareous outwash>lacustrine>glacio-fluvial, Quadrats were classified in relation to published soil survey maps and reports (Weber and Morwick 1946, Gillespie and Acton 1981, Hoffman and Acton 1974).

Soil Order: an indicator of the effects of the dominant soil-forming processes on soil properties (Canada Soil Survey Committee 1978): in this study, an indicator of regional differences in the inherent fertility of sampled soils. The inferred order of soil fertility was Luvisolic>Brunisolic> Gleysolic, based on the presence/absence of a Bt horizon and the degree of periodic or prolonged reducing conditions associated with saturated soils. Soils in quadrats with seasonally moist or wet depressions were classified as gleyed Brunisolic, and gleyed Luvisolic, if mottling observed during a previous field assessment (1995) was observed within 50 cm of the mineral surface (Canada Soil Survey Committee 1978). Quadrats were classified in relation to published soil survey maps and reports (Weber and Morwick 1946, Gillespie and Acton 1981, Hoffman and Acton 1974).

Soil Moisture Regime: in this study, an indicator of the duration and intensity of moisture deficit in the soil (Patterson 1978). In contrast to measures of soil moisture, soil moisture regime is an indicator of variations in the soil moisture supply through time (= 100 years) (Pierpoint 1978). The former is a direct measure of soil moisture content, whereas the fatter is assessed in relation to physical soil properties and soil profile characteristics (Ontario Institute of Pedology 1985).

In 1995, the moisture regime at the centre of each study site was determined with reference to the depth and properties of soil mottles, and, the texture of mineral soil in the upper B horizon (Ontario Institute of Pedology 1985). This method was not used for the 1996 field season owing to difficulties in determining the depth to mottles during the late August assessment. The moisture regime of quadrats was therefore classified in relation to the moisture status of forest depressions. a supplementary, planned, assessment that was initiated at the beginning of the spring survey.

Quadrats with standing water in depressions during the spring and/or summer vegetation survey (see Section 2.2.3.3) were classified as "seasonally wet". Quadrats with moist depressions (soil dark and moist to the touch when compared to the adjacent forest floor) during the spring and/or summer vegetation survey were classified as "seasonally moist". Quadrats with dry depressions during both the spring and summer vegetation survey were classified as "seasonally dry". This method, while lacking the rigor of the Ontario Institute of Pedology procedure, was deemed sufficient to characterize major differences in soil moisture within and between forest patches, when applied with reference to the results of the 1995 survey.

Soil Organic Matter (SOM): a general indicator of soil quality but not useful for predicting nitrogen mineralization (Nadelhoffer *et al.* 1983); in this study, an indicator of soil moisture retention capacity (Brady 1990), and, of seasonally saturated soils when organic matter content was greater than 30% (Canada Soil Survey Committee 1978).

Ten soil cores to 15 cm depth were removed with a slot-tube soil sampler from each 10m x 10m quadrat in relation to a stratified sampling design and placed in plastic freezer bag. The cores were taken in a hap-hazard manner in proportion to the type and area of microhabitat present. Loose leaf litter was removed prior to taking the core; litter ramified with fungal mycelia was left undisturbed and included in the core. Each sample was air-dried prior to storage and subsequently mortared and pestled, passed through a 4.00 mm sieve, placed in a drier oven, and dried overnight at 105°C (David 1988, Karam 1993). Approximately 10-15 grams of dried soil, from which macro charcoal and large roots had been removed, were placed in a crucible of known weight and subsequently ashed in a muffle furnace for 12 hours at 430° C (Nelson and Sommers 1982, Davies 1974) (550° C for soils with high apparent soil organic matter: Karam 1993). The furnace temperature was increased to the loss-on-ignition temperature over a three hour period (four hours for organic soils) to avoid flash combustion of organic matter and to minimize transient surges due to thermal lag. Following cooling (approximately 8 hours), samples were removed and weighed on an OHAUS Precision Plus digital balance. Percent soil organic matter was calculated as:

% SOM = [("soil" - "ash") / "soil"] x 100

Replicates were not taken owing to the loss of sixty soil samples during storage.

Soil pH: in this study, an indicator of conditions favoring nitrification (Dancer *et al.* 1973). Five grams of soil (previously dried at 105° C and stored 3.5 years), and 15 ml of de-ionized water, were placed in a small plastic beaker, stirred vigorously for 60 seconds on a mechanical stirrer, and let stand for at least 15 minutes. The 1:3 ratio of soil to de-ionized water was increased to 1:6 for soils with > 30% soil organic matter to achieve wet slurry conditions. The pH_{water} reading was taken with a glass electrode. Corning pH Meter, Model 7, calibrated to buffer pH 7.0. The pH of soil samples was determined for quadrats 1-24, 34-40, 49-56, 61-192, only, since the original soil samples for quadrats 1-60 were lost during storage, and, changes in land use prevented replicates from being taken at each location.

Base Cations: in this study, a measure of the concentration of available calcium, exchangeable magnesium, and exchangeable potassium, in the upper 15 cm of the soil profile. Free calcium carbonate was not purged from soil samples, in order to determine the concentration of calcium "available" for plant uptake. Results were therefore reported as "available calcium" since the recorded concentration will include both "exchangeable" calcium, and free calcium carbonate, when the latter is present in the soil profile.

Twenty grams of soil, passed through a 1mm sieve, and one teaspoon of silica, were placed in a plexiglass leaching tube containing approximately 1 cm of tightly packed glass wool, a one-hole rubber stopper, and rubber tubing. After adding a teaspoon of silica to cover the soil surface, the leaching tube was suspended between a 300 ml flask containing 250 ml of 1 N, pH 7.0, ammonium acetate, and, a labeled, empty 300 ml flask. The outflow rate of leachate was subsequently adjusted to a slow drip, approximately 10 drops per 60 seconds, so that the leaching process would last at least 4 hours.

The concentration of calcium, magnesium, and potassium cations was subsequently determined by atomic adsorption spectrophotometry in a Perkin Elmer Atomic Absorption Spectrometer, Model 3100. The leachate was diluted, as required, with a 3% solution of lanthamum chloride, with the aid

of a Nichiryo AutoDilutor. The following equation was used to calculate the concentration of base cations:

 $\frac{\text{spectrometer reading x 50 (dilution factor) x 250 (ml leachate)}}{20 (grams of soil)} = ppm \text{ base cations in soil}$

The concentration of cations was subsequently expressed in centimoles (10⁻² moles) per kg of soil (cmol/kg). The atomic adsorption spectrophotometry was conducted by Dr.Teng, Department of Forestry, University of Toronto.

The calcium:magnesium ratio, and, potassium:magnesium ratio, were also calculated to clarify the degree to which cation uptake may be affected by an imbalance among base cations (Ouimet and Camire 1995). Soil samples were reacted with 0.1N hydrochloric acid to assess the degree to which free calcium carbonates was present. Soils that "fizzed" upon reaction were classified as "reactive".

2.2.3.2 Forest Stand Structure Variables

Tree Diameter Breast Height (dbh): in this study, an indicator of the time since last disturbance (Henry and Swan 1974, Oliver and Stephens 1977, Lorimer 1980). The diameter of live and standing dead stems > I m tall was measured with a diameter tape and recorded in 2 cm increments. for stems 0-30 cm dbh.: stems > 30 cm dbh were measured to the nearest 0.1 cm in continuous increments. The diameter of stems in the 0-2 cm and 2-4 cm size classes was estimated visually in stands with many stems. Stems < 1 m tall were recorded as seedlings and assigned to one of eight cover classes.

Tree Height Class: in this study, an indicator of forest stand development (Oliver and Larson 1996). Tree height was measured during leaf flush with the aid of a Suunto clinometer and 30 m tape. Stems were assigned to the following height classes: < 1 m, 1-3 m, 3-10 m, 10-15 m, 15-25 m, and >25 m. Stems <1 m tall were recorded as seedlings and assigned to one of eight cover classes. Representative stems in each height class were measured and subsequently used as visual standards for assigning stems to the appropriate size class. **Forest Stratum:** in this study, an indicator of forest stand development (Oliver and Larson 1996), and, a datum used to classify forest cover type and selected microhabitat types. Four strata were recognized in this study: herb layer (<1 m), shrub layer (1-3 m), subcanopy (3-15 m), and canopy (>15 m). The threshold values for each stratum were measured for representative trees with the aid of Suunto clinometer and a 30 m tape.

Forest Cover Type: in this study, an indicator of recurring tree assemblages, soil moisture regime (Maycock 1963), available light (Messier and Bellefleur 1988, Canham *et al.* 1990), and litter quality (Pastor *et al.* 1984, Reich *et al.* 1997). Five cover types were recognized, based on their association with increasing soil moisture, and, the presence/absence of sugar maple: i) red or white oak (no sugar maple, no wet-mesic or wet taxa); ii) sugar maple + red or white oak; iii) sugar maple (no red or white oak, no wet-mesic or wet taxa); iv) sugar maple = silver maple, black ash, red ash, balsam poplar, or American elm; v) silver maple, black ash, red ash, balsam poplar, or American elm; v) silver maple, black ash, red ash, balsam poplar, or American elm (no sugar maple, no red or white oak). Cover types were assigned to quadrats based on the presence/absence of indicator species in the shrub, subcanopy or canopy layer. This approach accommodated variability in species composition and stand structure caused by recent disturbance, and, revealed differences in moisture conditions more accurately than a classification based solely on the predominant species in the canopy layer.

Canopy Closure: in this study, an indicator of the quantity and quality of solar radiation received at the forest floor (Messier and Belletleur 1988, Canham *et al.* 1990). Canopy closure was defined as the proportion of the sky hemisphere covered by leaves, branches and stems. High canopy closure therefore corresponds to low levels of red-shifted solar radiation received at the forest floor. Percent canopy closure was calculated by "HEMIPHOT" (ter Steege 1994) from digitized, hemispherical, 35 mm photographs taken with a Nikon F3 camera that was fitted with a Nikkor 8 mm *f*2.8 "fisheye" lens, and, a Nikon DR-3, right-angle view finder. A high contrast black and white film, Kodalith High Contrast Ortho Film, ASA 12, Eastman Kodak, was used to maximize the contrast between sky and canopy (Chan *et al.* 1986).

A bemispherical photograph of the forest canopy was taken from the centre of each quadrat. The
camera was mounted on a Manfrodo tripod, set at 1.0 m above ground level, oriented to 0° north, and leveled in the vertical and horizontal planes with the aid of pocket spirit level. An exposure setting was chosen that would maximize the contrast between sky and canopy. The canopy and visible sky were metered with the aid of a Minolta M spotmeter and an *f*-stop was selected that was at least two stops higher than the value for open sky. Brightly illuminated spots in the canopy were metered to ensure that the sky value was at least two stops greater than the highest value recorded for the canopy. For greater security, this exposure value was bracketed by one stop. Inspection of the developed images revealed that the middle exposure best represented the status of small and large canopy openings.

This approach was field tested to confirm that canopy vegetation was not mis-classified as "sky". Field testing revealed that under high canopy closure conditions (low light), the edge of the exposed image could not be differentiated from the film base. Experimentation subsequently determined that by taping a white plastic arrow to the rim of the lens at each cardinal direction, sufficient light would be reflected to the film plane to reveal the reveal the edge of the exposed image. The north arrow was made larger than the other arrows to ensure proper orientation of the digitized image. The arrows were cut from a white plastic container. The projecting portion of the E. S. and W arrows was 8 mm wide x 5 mm high: the projecting portion of the N arrow was 14 mm wide x 8 mm high.

The developed images (5 cm x 5 cm) were digitized with a SCANJET IIC. Hewlett Packard, scanner. Images were scanned at 200 DPI at contrast setting 250 and brightness setting 125. The digitized images were processed in HEMIPHOT with a fixed radius of 195 units.

The radius over which light was collected in each image varied with canopy height and was typically on the order 20-25 metres.

2.2.3.3 Vegetation Variables

Species Composition: The plant inventory was conducted 7 May 1996 to14 June 1996 (spring survey) and 20 June 1996 to 25 July 1996 (summer survey). The boundary of each 10m x 10m

quadrat was marked by yellow propylene rope and sampled intensively by walking close interval (approximately 1.5 m) line transects. Taxa were recorded by microhabitat to facilitate analyses of compositional trends in relation to environmental factors. Known taxa were identified to species rank in the field with the aid of 10x hand lens: voucher specimens were collected for unknown taxa. and, for taxa requiring more than 10x magnification of diagnostic characters. Voucher specimens were identified with the aid of a dissecting microscope and with reference to the following authorities: trees (Barnes and Wagner 1981; Farrar 1995), willows (Argus 1992), shrubs (Voss 1985, Soper and Heimburger 1982), ferns and fern allies (Cody and Britton 1989), grasses (Voss 1972, Dore and McNeill 1980), sedges (Voss 1972, Gleason 1952, Webber and Ball 1984), herbs (Voss 1972, 1985, 1996; Gleason 1952, Gleason and Cronquist 1991), Asteraceae (Fisher 1988, Sempleand Ringius 1983, Semple and Heard 1987). Identifications of difficult taxa were reviewed by specialists in the Ontario flora. Curex and Poaceae identifications were reviewed by M.J. Oldham. Natural Heritage Information Center, Ontario Ministry of Natural Resources; Carex, Section Ovales, identifications were reviewed by Dr. P.W. Ball, Erindale College, University of Toronto: selected grass specimens were identified by Dr. S.J. Darbyshire, Agriculture and Agri-Food Canada; selected fern identifications were reviewed by Dr. D.M. Britton. Professor Emeritus. University of Guelph. Twenty-five specimens could not be identified to species.

Cover Class: in this study, an indicator of plant abundance. The abundance of each species, and overall plant cover, within a 10 m x 10 m quadrat was estimated visually with the aid of one-metre swag sticks and marked quadrat diagonals. Plant abundance was recorded in relation to eight cover classes: i) solitary individual (+), 2-5 individuals (r); ii) < 1%; iii) 1-5%; iv) 6-15%; v) 16-25%; vi) 26-50%; vii) 51-75%; viii) 76-100%.

2.2.3.4 Plant Attribute Variables

Habitat Affinity: in this study, an indicator of the habitats in which a species is typically found and thus an indicator of transient and permanent residents in the forest understory. Four habitat types were recognized based on the apparent affinity for closed vs open canopy conditions: i) "forest": species found exclusively in closed forest habitats: ii) "forest + open": species found primarily in

closed forest habitats but also present in forest openings and thickets; iii) "open + forest": species found primarily in open habitats such as marshes, old fields, thickets, forest edges, but also present in closed forest habitats; iv) "open": species found exclusively in open habitats or disturbed sites in forest habitats. Species were assigned to one habitat type based on the habitat descriptions summarized by Voss (1972, 1985, 1996) for the State of Michigan, U.S.A.

Cnefficient of Wetness: the probability of finding a species in a wetland habitat (Oldham *et al.* 1995); in this study, an indicator of the moisture affinity of surveyed species. Plants in southern Ontario were classified by Oldham *et al.* (1995) in relation to the following categories: i) "obligate wetland": species almost always occurs in wetlands under natural conditions, >99% probability of being found in a wetland habitat: ii) "facultative wetland": species usually occurs in wetlands but occasionally found in non-wetland habitats. 67-99% probability of being found in a wetland habitat: 67-99% probability of being found in a wetland habitat: iii) "facultative upland": species equally likely to occur in wetlands or non-wetlands, 34-66% probability of being found in a wetland habitat: iv) "facultative upland": species occasionally occurs in wetlands but usually occurs in non-wetland habitats, 1-33% probability of being found in a wetland habitat: v) "obligate upland": species almost never occurs in wetlands under natural conditions. <1%

2.2.3.5 Microhabitat Variables

Microhabitat: in this study, a dimension of environmental heterogeneity, and, an indicator of uniform conditions, at the quadrat scale, with respect to canopy closure, soil moisture, substrate, and disturbance. Thirty-nine microhabitats were recognized (Table 2.3). The core features in the classification are forest floors, depressions, tip-up mounds, tree pits, stumps, logs, canopy gaps, seeps, raised root mats, and, features created by human disturbance, such as lanes, access roads, ditches, and regenerating fields. The moisture status for selected features is not recorded in Table 2.3 to reduce the number of elements in the classification. The moisture status of seep, tiparian marsh, and riparian thicket is "seasonally wet": the moisture status of riparian meadow is "seasonally moist"; the moisture status of mounds and regenerating fields is "seasonally dry". The moisture status of stumps, logs, raised root mats, and lane/access roads varies from seasonally dry to

Table 2.3. Definition and areal extent of forest microhabitats. Notes: "seasonally dry" = forest floor or depression-free of standing water or moist soil during spring and summer survey; "seasonally moist" – forest floor or depression has moist soil during spring or summer survey; "seasonally wet" = forest floor or depression has standing water during spring or summer survey; "forest floors, *sensn stricto*" = that portion of the forest floor free of seasonally moist or wet depressions, tip-up mounds, tree pits, logs, stumps, raised root mats, seeps, lane/access roads, regenerating fields, riparian meadows, riparian marshes and riparian thickets; "-cc" – feature under a closed canopy of trees, saplings or tall shrubs; "-oc" = feature under an open canopy (feature has a clear view to the sky); % SA – % of sampled area (19,200 m²).

Microhabitat	% SA	Description
seasonally dry forest floors-ce	54.3	forest floors, sensu stricto, that are seasonally dry and under a closed canopy
seasonally dry forest floors-oc	6,8	forest floors, scosn stricto, that are seasonally dry and under an open canopy
seasonally moist forest floors-ee	3.0	forest floors, sensu stricto, that are seasonally moist and under a closed canopy
seasonally moist forest floors-oc	3.0	forest floors, seusn surieto, that are seasonally moist and under an open canopy
seasonally moist forest depressions-ce	4.1	forest depressions (depression in forest floor +20 cm deep) that are seasonally moist and under a closed canopy
seasonally moist forest depressions-oc	0.2	forest depressions (depression in forest floor >20 cm deep) that are seasonally moist and under an open canopy
seasonally wet forest floors-cc	0.6	forest floors, sensu stricto, that are seasonally wet and under a closed canopy
seasonally wet forest depressions-ce	4.0	forest depressions (depression in forest floor >20 cm deep) that are seasonally wet and under a closed canopy
seasonally wel forest depressions-oc	1.0	forest depressions (depression in forest floor >20 cm deep) that are seasonally wet and under an open canopy

Table 2.3. Definition and areal extent of forest microhabitats (cont¹d).

38

Microhabitat	% SA	Description
meand-soc	0.4	a raised mass of cards that has been deposited by soil eroding from the roots of a wind- thrown tree, and, under an open canopy, <i>sensu stricto</i> , the portion of a pit-mound complex fying above the plane of the forest floor
seasonally dry pit-ce	ī.l	a depression in the forest floor that has been created by the uprooting of a wind-drown tree, and, that is seasonally dry and under a closed canopy; <i>season stricto</i> , the portion of a pit-mound complex lying below the plane of the forest floor.
seasonally dry pit-oc	0.1	a depression in the forest flour that has been created by the uprooting of a wind-thrown tree, and, that is seasonally dry and under an open canopy, <i>sensu strictu</i> , the portion of a pit-mound complex fying below the plane of the forest floor.
scasonally moist pit-cc	0.3	a depression in the forest floor that has been created by the uprooting of a wind-thrown tree, and, that is seasonally moist and under a closed canopy, <i>sense stricto</i> , the portion of a pit-mound complex lying below the plane of the forest floor.
seasonally wet pil-cc	0.2	a depression in the forest floor that has been created by the uproofing of a wind-thrown tree, and, that is seasonally wet and under a closed canopy; <i>sensu writeto</i> , the portion of a pit-mound complex lying below the plane of the forest floor.
seasonally wet pit-oc	0.1	a depression in the forest theor that has been created by the uproofing of a wind-thrown tree, and, that is seasonally wet and under an open camppy, <i>seasus stricto</i> , the portion of a pit-mound complex tying below the plane of the forest floor.
log-cc	0.5	a tree bole (> 4 cm dbh) that is in contact with the earth and under a closed campy
ဂမ္မာတင္	2.0	a tree bole (> 4 cm dbh) that is in contact with the carth and under an open cartopy

Table 2.3. Definition and areal extent of forest microhabitats (cont'd).

62

Microhabitat	%	Description
stump-cc	0,2	the upright remnant of a cut or wind-thrown tree that is under a closed canopy
stump-oc	0.1	the upright remnant of a cut or wind-thrown tree that is under an open canopy
raised root mat-cc	<u> </u>	the root mat of a partially wind-thrown tree, typically associated with seasonally moist or wet soils, and, a deep (>15 cm) humus layer, or, the elevated ball of soil and roots at the base of shrubs on seasonally moist or wet soils; campy closed (defined here as $\pm60\%$ campy closed (defined here as $\pm60\%$
raised root mat-oc	0.6	the root mat of a partially wind-thrown tree, typically associated with seasonally moist or wet suils, and, a deep (>15 cm) humus fayer; or, the clevated ball of soil and roots at the hase of shrubs on seasonally moist or wet soils, canopy open (defined here as < 60% closure)
slone-ec	<0.1	exposed. limestone, boulders, under a closed cauopy
hune/road-ce	2.3	a farm lane, or forest access road, under a closed canopy
lane/road-oc	1.3	a farm lane, or forest access road, under an open canopy
ditch-cc	0.3	a ditch associated with a farm tane or forest access road, under a closed canopy
ditch-oc	1.()	a ditch associated with a farm lane or forest access road, under an open canopy
regenerating field-cc	£ .1	a plowed, farm field that is undergoing forest succession and presently is under a closed canopy of saplings or tall shrubs
regenerating field-oc	6.1	a plowed, farm field that is undergoing forest succession and presently is in the "old field" stage, under an open canopy

Table 2.3. Definition and area) extent of forest microhabitats (cont'd).

ŝ

Description	a seasonally flooded, riparian meadow, on mineral soil, under an open canopy that has been maintained by periodic cuting	a seasonally flooded, shallow marsh, on organic soil	a seasonally flooded thicket swamp, on mineral soil, under an open canopy that has been maintained by periodic cutting
% SA	0.7	0,4	0.1
Microhabitat	riparian meadow-oc	ripatjan mersh-oc	riparian thicket-oc

Table 2.3. Definition and areal extent of forest microhabitats (cont'd).

seasonally wet depending on the feature and the quadrat surveyed. Forest depressions were defined as depressions in the forest floor >20 cm deep and were measured with the aid of painted swag sticks (alternating black and tan bands 20 centimetres wide) and a pocket spirit level.

Microhabitats: in this study, an indicator of environmental heterogeneity. The value represents the number of recognized habitats in a 10 m x 10 m quadrat.

2.2.3.6 Landscape Variables

Patch Interior Area: in this study, an indicator of forest fragmentation: the area (ha) of continuous forest that is more than 100 m from the forest edge (Austen *et al.* unpublished). Interior area was calculated from digitized LANDSAT TM satellite imagery by the geographic information system, ARC-INFO (S. Hounsell, Ontario Hydro, *pers. com.*). This data set is the property of the Long Point Bird Observatory (LPBO) and has been used in the thesis with their permission (M. Austen, *pers. com.*). Forest patches were assigned by the LPBO to one of eight area classes: 0-5 ha, 6-10 ha, 11-20 ha, 21-50 ha, 51-100 ha, 101-200 ha, 201-500 ha, 501-1000 ha.

Patch Area: in this study, an indicator of forest fragmentation: the area (ha) of continuous forest within a forest patch. Patch area was calculated from the stand data recorded on forest stand maps, 1:10.000 scale (Ontario Ministry of Natural Resources 1978, 1979, 1980). In contrast to the previous data set, patch area is a continuous variable that accounts for the total area of continuous forest in the patch.

Patch Isolation: in this study, an indicator of forest fragmentation. Patch isolation was defined as the mean distance (m) to the nearest forest patch in eight 45° arcs radiating from the site centre. The distance from patch edge to patch edge was measured to the nearest 50 metres on a 1:50,000 scale National Topographic Service (NTS) map sheet with the aid of a Staedtler-Mars ratio scale.

Forest Cover: in this study, an indicator of forest fragmentation. Forest cover was defined as the percent forest cover within a 5 km x 5 km square centred on the study site. In contrast to the

previous data set, this approach includes forest cover within the sampled patch and thus accounts for total forest cover within the vicinity of the vegetation sample. The forest cover on a 1:50,000 scale National Topographic Service (NTS) map sheet was calculated with the aid of a dot grid.

2.2.3.7 Supplementary Data

The following data was recorded in the field but not used in analyses reported in this thesis.

Decomposition Status: an indicator of time since death. The status of logs and standing dead stems was recorded in relation to the following classification: i) bark firm; ii) bark loose; iii) stem hare, firm: iv) stem mossy. firm: v) stem soft, bare: vi) stem soft, mossy.

Reproductive Status: The reproductive status of species was recorded in relation to the following classification: i) vegetative; ii) flowering (flower in bud or open); iii) fruiting.

Phenology: a datum that enables the reproductive status of plants to be correlated with canopy conditions. Reproductive status at the time of sampling was recorded in relation to the following classification: i) trees in bud or leaf flush: ii) trees with immature leaves, iii) trees with mature leaves; iv) trees with senescing leaves.

Herbivory Status: the herbivory status of plants was recorded in relation to the following classification: i) leaf tip bitten: ii) leaf with holes: iii) leaf with ragged edge; iv) leaf with fungal dots: v) leaf with egg cases; vi) plant decapitated. The intensity of herbivory was recorded in relation to the following classification: i) "low":one to few leaves per plant. <1% of population: ii) "intermediate": few to many leaves per plant. 1-10% of population: iii) "high": few to many leaves. >10 % of population.

2.2.4 Analytical Methods

2.2.4.1 Overview

The objectives of this thesis have been achieved by means of a series of comparative, mensurative,

experiments (Hurlburt 1984) in which a system property (species richness, species composition, species attribute) has been measured repeatedly in treatment space (environmental states, environmental gradients, microhabitats, forest patches). The experimental units in each experiment were the physical locations at which the system property was measured (Hurlburt 1984).

In mensurative experiments, pseudoreplication may arise when the sample space in which system properties are measured is smaller than the inference space implicit in the hypothesis being tested (Hurlburt 1984). In this chapter, the inference space is second-growth northern hardwood stands on Ordovician limestone in the vicinity of Peterborough, Ontario. Sample space is commensurate with this inference, since system properties were measured in treatment space across the region.

The degrees of freedom for determining a treatment effect varied with the experimental treatment since treatment conditions were rarely present in each quadrat or forest patch, or, were not statistically independent: n < 192 for system properties measured in relation to environmental states, gradients and microhabitats; n < 24 for system properties measured in relation to forest patches.

The statistical independence of samples was more apparent for some analyses than for others. While it is clear that replication within patches should not contribute degrees of freedom when testing for a treatment effect among patches, the statistical independence of two or more quadrats within a patch is less transparent when testing for a treatment effect among environmental states, gradients, or microhabitats. At issue is whether differences in disturbance history at the micro-scale sufficiently influence the composition of the seed rain, germination success, resource availability, competitive interactions, or plant persistence, for quadrats to be regarded as independent samples. Since this could not be determined, *a priori*, each quadrat was permitted to contribute one degree of freedom when testing for a treatment effect among environmental states, gradients. The effects of patch membership, if present, were subsequently removed by treating patch membership as a co-variable in multivariate analyses.

2.2.4.2 Distribution of Species on Environmental Gradients

Analysis #1: The tendency for the composition of species to be more similar within, than among, forest patches was investigated in order to evaluate the relative influence of local processes, such as short-distance dispersal and competitive exclusion, and regional processes, such as glaciation and pedogenesis.

Similarities in species composition were investigated by detrended correspondence analysis (DCA) (Jongerman *et al.* 1987). The degree to which quadrats were clustered within patches was assessed visually. The contribution of patch membership to the dispersion of quadrats was evaluated in canonical correspondence analysis (CCA) by the Monte Carlo permutation test (n=1000 permutations), after fitting all remaining environmental variables as co-variables. DCA was the preferred method for the pattern analysis since the dispersion of quadrats is governed by species relations with the underlying environment rather than with the set of variables chosen for study. DCA was preferred to correspondence analysis (CA) since the latter ordination displayed a pronounced arch that distorted the order of quadrats along the second axis. All analyses were performed in CANOCO Version 3.12 (ter Braak 1991). The DCA ordination diagram is presented Figure 2.1.

Analysis # 2: The distribution of species on environmental gradients was examined by canonical correspondence analysis (CCA) in order to determine which environmental variables exerted the greatest influence on the composition of herb assemblages. Patch membership was treated as a covariable in this analysis to minimize the contribution of replicate samples within patches when testing for an overall treatment effect on the dispersion of species in ordination space. The variable "number of live tree stems >1m" was omitted from the analysis due to high collinearity (variance inflation factor =36.6) with the variable "number of stems 0-4 cm dbh". Ordination scores were scaled in relation to scaling mode 2 (i.e. species scores are weighted mean sample scores) since the primary interest was the dispersion of species rather than quadrats (ter Braak 1994). Species in ordination space are therefore situated at the centroid of the quadrats (not shown) in which they occur. The relative importance of each environmental variable is represented by an arrow that points

in the direction of maximum influence and that is scaled to reflect the strength of the correlation between the variable and the fitted abundance of the plotted species (ter Braak 1994). The scaled co-ordinates for the head of each arrow are the biplot scores for the first and second axis (ter Braak 1994). An overall test of significance of the species ordination was determined by the Monte Carlo permutation test (n=9.999 permutations). All analyses were performed in CANOCO version 3.12 (ter Braak 1991). The scatter plot was created in S-Plus. Version 4.5 (Mathsoft Inc.1998) and annotated in Publisher 98 (Microsoft Corporation 1998). The CCA ordination diagram is presented in Figure 2.2.

Analysis #3: The relative importance of environmental variables was examined further by determining the percentage of the total inertia explained by each variable. The percent inertia explained was computed as the (Σ canonical eigenvalues for each variable + Σ unconstrained eigenvalues for the ordination) x 100. Since most environmental variables interacted with at least one other variable, the fraction of inertia that was uniquely explained by each variable was also determined by fitting all remaining variables as co-variables. The significance of the latter value was determined by the Monte Carlo permutation test (n=1,000 permutations), after Bonferroni correction for n=16 tests (p<0.003). All analyses were performed in CANOCO, Version 3.12 (ter Braak 1991). The results of this analysis are presented in Table 2.13.

Analysis #4: The distribution of species on the principal environmental gradients was examined further to determine if species were restricted to a particular edaphic condition. forest cover type, or disturbance state. This analysis was conducted in relation to variables that had the highest inter-set correlations with the first or second axis of the CCA ordination. The percentage of taxa restricted to a given soil parent material, soil order and forest cover type was evaluated at the quadrat spatial scale: the percentage of taxa restricted to a given moisture condition and disturbance class was evaluated at the quadrat scale. The analysis was performed in JMP Version 3.2.2. (SAS Institute Inc. 1997): results are presented in Table 2.14.

Analysis #5: The variance explained by environmental variables in Analysis # 2 was decomposed.

(Quinghong and Brakenhielm 1995) to determine the relative contribution of edaphic variables, patch variables, and landscape variables, to the dispersion of species scores in ordination space. This was undertaken to clarify the relative importance of local versus regional processes (*sensu* Ricklefs and Schluter 1993) in the distribution of understory herbs.

The method of Quinghong and Brakenhielm (1995) enables one to determine the degree of interaction among variables, and, to determine the independent (unique) contribution of each variable. For this analysis, regional processes such as glaciation and pedogenesis are considered to be the primary contributors to properties of the edaphic variables (soil parent material, soil order, soil moisture class, percent soil organic matter, and soil pH), and, plant migration is considered to be the principal means by which propagules are distributed across the landscape. Recruitment limitation is considered to be a consequence of regional processes when due to soil based factors. Local processes, such as short-distance dispersal, competitive interactions, and heterogeneity arising from the death or removal of canopy trees, are considered to be the primary contributors to patch dynamics and to local species composition. The "patch scale" variables for this analysis were forest cover type, percent canopy closure, stem diameter class, number of microhabitats, open microhabitats, disturbed microhabitats, and patch membership. Landscape variables (patch isolation and patch size) were considered to reflect the degree of migration constraint inherent in the spatial configuration of forest patches in the present-day landscape.

The results of this analysis were summarized in a Venn diagram that displays the percentage of the explained variance uniquely, and jointly, accounted for by edaphic, patch, and matrix (landscape) variables. The analysis was performed in CANOCO Version 3.12 (ter Braak 1991). The Venn diagram, presented in Figure 2.3, was created in Publisher 98 (Microsoft Corporation 1998).

2.2.4.3 Distribution of Species in Forest Microhabitats

Analysis #1: The response of species to the pragmatic classification of microhabitats was investigated by detrended correspondence analysis (DCA). At issue was the degree to which the

composition of species differed among habitat types. The significance of the ordination was assessed in relation to the criterion that habitats separated by 4 or more standard deviation units on the first or second axis should have few if any species in common (ter Braak 1987). The Monte Carlo permutation test cannot be performed on DCA scores in CANOCO Version 3.12. CCA was not suitable for this analysis since the habitat categories represented a nested set of environmental states within microhabitats rather than discrete or continuous variables within quadrats. The analysis was performed in CANOCO Version 3.12 (ter Braak 1991). The DCA ordination diagram is presented in Figure 2.4: the number of habitats occupied by surveyed taxa is presented in Table 2.15.

Analysis #2: The species composition of canopy gaps was investigated further to determine the influence of canopy-opening size on species composition. Ten size classes were established for the ordination analysis: $0-10 \text{ m}^2$, $11-20 \text{ m}^2$, $21-30 \text{ m}^2$, $31-40 \text{ m}^2$, $41-50 \text{ m}^2$, $51-60 \text{ m}^2$, $61-70 \text{ m}^2$, $71-80 \text{ m}^2$, $81-90 \text{ m}^2$, $91-100 \text{ m}^2$. The species composition of small gaps was expected to be similar to that of closed, seasonally dry, forest floors. If gap size were the primary determinant of species composition in canopy openings, then gaps of a given size should cluster in ordination space, and, gaps of increasing size should cluster at increasing distance from the reference condition. Gaps that were separated by 4 or more standard deviation units were expected to have few if any species in common (ter Braak 1987). The analysis was performed in CANOCO Version 3.12 (ter Braak 1991). The DCA diagram is presented in Figure 2.5.

Analysis #3: The distribution of species across habitats was assessed to determine the degree to which environmental heterogeneity in the forest understory has been utilized by plants. A restricted distribution pattern was considered support for the view that heterogeneity increases the species richness of forest patches primarily through the provision of novel resources, whereas, a pattern of widespread use was considered support for the view that heterogeneity maintains the richness of forest patches primarily through the species that heterogeneity maintains the richness of forest patches primarily through the species of competing species.

Four broad habitat categories were established for the analysis: closed dry forest floors. *sensu stricto*; natural disturbance features (canopy gaps, tree pits, tip-up mounds, stumps, logs); human disturbance

(regenerating fields, lanes, ditches); and moist or wet habitats (forest floors, depressions, seeps, riparian meadow, riparian marsh and riparian thicket). The distinction between natural and human disturbance is somewhat forced since many of the stumps and canopy gaps in this study were created by selective logging. The effect on the forest canopy, however, is similar to natural tree fall. The analysis was performed in JMP Version 3.2.2. (SAS Institute Inc. 1997) and EXCEL Version 7.0a (Microsoft Corporation 1996). The results of this analysis are presented in Figure 2.6.

2.2.4.4 Distribution of Sugar Maple on a Moisture-Fertility Gradient

Analysis #1: Species richness in sampled forest stands was inversely correlated with sugar maple abundance (Chapter 4.0). This pattern was especially evident on mesic soils overlying calcareous till where sugar maple abundance explained 63.0% of the variance in species richness in undisturbed stands. The distribution and abundance of sugar maple at small spatial scales has recently been attributed to differences in the availability of calcium ions in the soil profile (Kobe *et al.* 1995, Kobe 1996). The structure of sugar maple stands in this study was subsequently examined on a gradient of increasing calcium availability to determine if there was evidence of increased sugar maple survivorship on calcium rich soils.

The analysis was initially restricted to undisturbed stands on Bransolic and Luvisolic soils, overlying calcareous till, in order to standardize samples with respect to soil parent material, soil moisture, and recent site disturbance. Soils with free calcium carbonate in the upper 15 cm of the soil profile (positive reaction to 0.1N HC1) were excluded from this analysis in order to standardize samples with respect to exchangeable calcium (see contrasting treatment in Analysis 2). The analysis was further restricted to quadrats with forest cover type 2 (sugar maple + red or white oak) and forest cover type 3 (sugar maple, no red or white oak, no wet-mesic, wet, trees) in order to standardize samples with respect to forest cover. N = 29 10m x 10m quadrats in 7 forest patches.

Stand structure was evaluated in relation to four size classes: 0-4 cm, 4-10 cm, 10-30 cm, and >30 cm, dbh. Sugar maple abundance was evaluated in relation to absolute and relative abundance. The relationship between sugar maple abundance and calcium availability was evaluated in simple linear

regression. The analysis was performed in JMP Version 3.2.2. (SAS Institute Inc. 1997). The results are presented in Figure 2.7.

Analysis #2: The relationship between stand structure and calcium availability was re-examined in undisturbed stands on Brunisolic soils in order to standardize samples for soil parent material (calcareous till), soil order (Orthic Melanic Brunisol) and soil series (Otonabee loam). Soils with free calcium carbonate in the upper 15 cm of the soil profile were included in this analysis, in order to standardize samples with respect to *available* calcium. This analysis, as before, was restricted to quadrats with forest cover types 2 and 3, in order to standardize samples with respect to forest cover. This sample (N = 17 quadrats in 3 forest patches) provided the most uniform subset of quadrats in which to assess the response of sugar maple to differences in available calcium.

Stand structure was evaluated in relation to four size classes: 0-4 cm, 4-10 cm, 10-30 cm, and >30 cm, dbh. Sugar maple abundance was evaluated in relation to absolute and relative abundance. The relationship between sugar maple abundance and calcium availability was evaluated in simple linear regression. The analysis was performed in JMP Version 3.2.2. (SAS Institute Inc. 1997). The results are presented in Figure 2.8.

Analysis #3: The response of shade tolerant and intolerant herbs to increasing calcium availability and sugar maple abundance was examined in order to test the presumption of declining light levels on calcium rich soils. This analysis was extended to determine the calcium affinity of understory plants in relation to flowering phenology. Of interest was the degree to which life histories attributed to shade avoidance, early spring flowering and an ephemeral life history, were associated with calcium rich soils. Early spring flowering plants with persistent shoots were contrasted with mid to late season flowering plants to determine if calcium affinity were associated with the degree to which the life cycle was completed after canopy closure. Ephemeral and early spring flowering plants were classified in relation to Rogers 1982, and, to flowering data collected during the spring and summer vegetation survey. June 1st was taken to be the transition date between early and mid to late season flowering. Although leaf expansion may proceed as late as the summer solstice, the forest canopy was typically well developed by June 1%.

As in Analysis # 1, this analysis was restricted to undisturbed stands on Brunisolic and Luvisolic soils, overlying calcareous till, in order to standardize samples with respect to soil parent material, soil moisture, and recent site disturbance. Soils with free calcium carbonate in the upper 15 cm of the soil profile were excluded from the analysis in order to standardize samples with respect to exchangeable calcium. The analysis was restricted to quadrats with forest cover type 2 (sugar maple + red or white oak) and forest cover type 3 (sugar maple, no red or white oak, no wet-mesic, wet, trees) in order to standardize samples with respect to forest cover. N = 29 10m x 10m quadrats in 7 forest patches.

Shade tolerance was evaluated by simple linear regression: calcium affinity in relation to flowering affinity was evaluated by Wilcoxon rank sum test. The analyses was performed in JMP Version 3.2.2, (SAS Institute Inc. 1997). The results are presented in Figures 2.9 and 2.10.

2.2.4.5 Distribution of Plant Attributes on Environmental Gradients

The distribution of selected plant attributes on environmental gradients was examined in order to determine the degree to which the distribution of species may be explained by life form, life history, provenance, habitat affinity, shade tolerance and moisture affinity.

Analysis #1: The degree to which the selected attributes explained the dispersion of species scores in ordination space was investigated in CCA. The response variable was the proportion of taxa in a 10m x 10m quadrat with the attribute of interest. The significance of each contribution was determined by the Monte Carlo permutation test (n=1000 permutations). The analysis was performed in CANOCO Version 3.12 (ter Braak 1991). The results of the analysis are presented in Tables 2.16. 2.17 and 2.18.

Analysis #2: The distribution of provenance, habitat affinity and moisture affinity was summarized by microhabitat to provide contextual information for the interpretation of dispersal modes in Chapter 3. The response variable was the proportion of taxa (all life forms) in each habitat with the attribute of interest. The analysis was performed in JMP, Version 3.2.2. (SAS Institute Inc. 1997). The results are presented in Table 2.19.

2.3 Results

The tendency for species to be associated with particular environmental states and gradients will be reported in relation to the following headings: vegetation survey, relationships among environmental variables, species response to environmental gradients, species response to forest microhabitats, response of sugar maple and understory herbs to available calcium, local versus regional processes, and plant attributes.

2.3.1 Vegetation Survey

A summary of surveyed taxa by taxonomic rank. life form, life history and provenance is presented in Table 2.4. A summary of plant attributes by species is presented in Appendix2: a cross-referenced list of species codes is presented in Appendix 3. The distribution of species by soil parent material, soil moisture and canopy closure is presented in Appendix 4. A listing of species by microhabitat is presented in Appendix 5. The distribution of species by microhabitat is presented in Appendix 5. The distribution of species by microhabitat is presented in Appendix 5.

The sampled flora was composed of 413 species, 208 genera, 78 families and 45 orders. Twelve species occurred in \ge 50% of the quadrats surveyed: Acer saccharum(91.1%), Trillium grandiflorum (71.9%), Maianthemum canadense (67.8%), Epipactis helleborine (67.8%), Prunus virginiana (64.1%), Carex pensylvanica (62.5%), Tilia americana (55.7%), Erythronium americanum (55.7%), Dryopteris carthusiana (55.7%), Galium triflorum (54.7%), Taraxacum officinale (52.0%) and Arisaema triphyllum (50.5%). The three most species rich genera were Carex (43 taxa), Viola (10 taxa), and Aster (9 taxa). The three most genus rich families were Asteraceae (24 genera), Poaceae (21 genera) and Rosaceae (10 genera). The three most family rich orders were: Polypodiales (5 families), Ranunculales (4 families) and Solanales (4 tamilies).

Table 2.4. Summary of surveyed taxa by taxonomic rank. life form, life history and provenance. Life history: annual s.s = annual: biennial s.t = biennial, annual/biennial: perennial s.t = perennial, annual/perennial, biennial/perennial.

Plant Attribute	# Taxa	% (n=413)			
TAXONO	MIC RANK				
species	413	100.0			
genus	208	-			
family	78	-			
order	45	-			
LIFE	FORM				
tree	30	7.3			
shrub	55	13.3			
vine	9	2.2			
fem	23	5.6			
fern ally	8	1.9			
grass	36	8.7			
herb	252	61.0			
LIFE HISTORY					
annual s.s.	13	3.1			
biennial s.l.	22	5.3			
perennial s.l.	372	90.1			
unclassified	6	1.5			
PROVI	ENANCE				
native	349	84.5			
alien	58	14.0			
unknown	6	1.5			

Herbs were the most abundant life form (61.0% of surveyed taxa), followed by shrubs (13.3%), grasses (8.7%), trees (7.3%) and ferns (5.6%). The least abundant life forms were vines (2.2% and fern allies (1.9%). Most taxa were perennials (90.1%). Plants with biennial life histories (5.3%) were more abundant than plants with annual life histories (3.1%). The provenance of most taxa was native (84.5%).

The cover class of surveyed taxa in 10m x 10m quadrats is summarized by life form, life history, provenance, and habitat affinity in Table 2.5. The median cover for each element in the analysis was cover class 2 (0.5-1.0% cover). The maximum cover recorded was cover class 7 (50-75% cover). The life forms with the highest mean cover were the grasses (2.40) and fern allies (2.37). Plants with annual life histories had significantly greater mean cover (2.65) than plants with perennial (2.08) or biennial (1.88) life histories. The mean cover class of native taxa (2.12) was significantly greater than alien species (1.82). Plants with an affinity for "forest + open" habitats had a significantly higher mean cover (2.20) than plants with affinities for other habitats (1.79-2.11). Overall, plants with the lowest mean cover were plants with affinity for "open" habitats (1.79), alien species (1.82) and plants with biennial life histories (1.88).

The habitat affinity rating assigned to understory species is reported in Appendix 2. Species classified as occurring exclusively in forested habitats ("F") included: *Arisaema triphyllum*. *Maianthemum racemosa, Caulophyllum thalictroides, Solidago flexicatdis,* and, *Carex arctata,* Species classified as occurring primarily in forested habitats ("F=O") included: *Trillium grandiflorum, Erythronium americanum, Dryopteris carthusiana, Carex pedunculata,* and *Corylus cornuta.* Species classified as occurring primarily in open habitats ("O+F") included: *Carex pensylvanica, Taraxacum officinale, Impatiens capensis, Fragaria virginiana,* and *Poa pratensis.* Species classified as occurring exclusively in open habitats ("O") included: *Ranunculus acris.* Asclepias syriaca, Dancus carota, Lactuca serriola, and Vicia cracea.

A checklist of surveyed taxa is presented in alphabetical order by family in Table 2.6.

Table 2.5. Cover class of surveyed taxa in herb layer in10m x 10m quadrats by life form, life history, provenance and habitat affinity. Life history: annual *s.s* = annual; biennial *s.l*. = biennial, annual/biennial; perennial *s.l*. = perennial, annual/perennial, biennial/perennial. Cover class: 1=<0.5% cover. 2=0.5-1.0% cover. 3=1.0-3.0% cover. 4=3-15% cover, 5=15-25% cover, 6=25-50% cover. 7=50-75% cover. 8=75-100% cover. Highest value in bold when difference among attributes within category significant at p<0.05, after Bonferroni correction for n=16 tests. Wilcoxon rank sum tests, by column, independent samples.

	Cover Class								
Plant Attribute	Mean	Median	Range						
	LIFE	FORM							
tree	1.95	2	1-7						
shrub	2.05	2	1-7						
vine	1.80	2	1-5						
fem	2.21	2	1-6						
fern ally	2.37	2	1-5						
grass	2.40	2	1-6						
herb	2.11	2	1-7						
LIFE HISTORY									
annual s.s.	2.65	2	1-7						
biennial s.l.	1.88	2	1-4						
perennial s.l.	2.08	2	L-7						
PROVENANCE									
native	2.12	2	t- 7						
alien	1.82	2	1-6						
	LADITAT	AFEINITY							
Faraet	711		1.7						
forest + open	2.20		1-7						
open + forest	2.09	2	1-7						
open	1.79	2	1-6						

Table 2.6. Check list of surveyed taxa by family (alphabetical order).

ACERACEAE	Asarum canadense L.
Acer negundo L.	ASCLEPIADACEAE
Acer rubrion L.	Asclepias incarnata L. ssp. incarnata
Acer saccharinum L.	Asclepias syriaca L.
Acer saccharum Marshall ssp. saccharum	ASTERACEAE
Acer saccharum Marshall ssp. nigrum (Michaux	Achillia millefolium L. ssp. millefolium
t.) Desmarais	Ambrosia artemisiifolia L.
Acer spicatum Lam.	Antennaria neglecta E. Greene
ANACARDIACEAE	Arctium minus (Hill) Bernh. ssp. minus
Rhus radicans L. ssp. negundo (E. Greene) McNeill	Aster ciliolatus Lindley
Rinas typhina L.	Aster cordifolius L.
APIACEAE	Aster ericoides L.
Cicuta bulbifera L.	Aster lanceolatus L.
Cicuta maeulatum L.	Aster lateriflorus (L.) Britton
Cryptotaenia canadensis (L.) DC.	Aster macrophyllus L.
Daucus carota L.	Aster novae-angliae L.
Osmorhiza elaytonii (Michaux) C.B. Clarke	Aster puniceus L.
Sanicula marilandica L.	Aster umbellatus Miller
Sanicula trifoliata Bickn.	Bidens frondosa L.
Sanicula species	Carduus acanthoides L.
APOCYNACEAE	Cardutis nutans L.
Apocynum androsaemifolium L. ssp.	Chrysanthemum leucanthemum L.
androsaemifolium	Cirsium arvense (L.) Scop.
ARACEAE	Cirsium vulgare (Savi) Ten.
Arisaema triphyllum (L.) Schott ssp. triphyllum	Conyza canadensis (L.) Cronq.
ARALIACEAE	Erigeron annuus (L.) Pers.
Aralia nudicaulis L.	Erigeron philadelphicus L. ssp. philadelphicus
Panax quinquefolium L.	Erigeron strigosus Muhlenb. ex. Willd.
ARISTOLOCHIACEAE	Erigeran species

- .-

-

_ - - ._

Eupatorium maculatum L. Eupatorium perfoliatum L. Enpatorium rugosum Houtt, Euthamia graminifolia (L.) Nutt. Hieracium aurantiacum L. Hieracium caespitosum Dumort, ssp. caespitosum Lactuca canadensis L. Lactuca serriola L. Lactuca species Onopordon acunthium L. Prenanthes species Rudbeckia hirta L. Solidago altissima L. Solidago caesia L. Solidago canadensis L. Solidago flexicaulis L. Solidago gigantea Aiton Solidago juncea Aiton Solidago nemoralis Aiton Solidago rugosa Aiton ssp. rugosa Sonchus arvensis L. Sonchus oleraceus L. Taraxacum officinale G. Weber Tragopagon dubius Scop. Tussilago farfara L. BALSAMINACEAE Impatiens capensis Meerb. BERBERIDACEAE Caulophyllum thalictroides (L.) Michaux

Podophyllum peltatum L. BETULACEAE Almus incana (L.) Moench ssp. rugosa (Duroi) Clausen Betula alleghaniensis Britton Betula papyrifera Marshall Corvlus cornuta Marshall Ostrva virginiana (Miller) K. Koch BORAGINACEAE Hackelia virginiana (L.) I.M. Johnston BRASSICACEAE Cardamine diphylla (Michx.) A. Wood Cordamine pensylvanica Muhlenb. ex Willd. CAMPANULACEAE Lobelia inflata L. Lobelia species CAPRIFOLIACEAE Diervilla Ionicera Miller Lonicera canadensis Bartram Lonicera dioica L. Lonicera hirsuta Eaton Lunicera oblongifolia (Goldie) Hook. Sambucus canadensis L., Sambucus racemosa L. ssp. pubens (Michaux) House Triosteum aurantiacum E. Bickn. Vihurman acerfolium L. Vihurnum lentago L. Viburnum opulus L. Viburnum trilobum Marshall

CARYOPHYLLACEAE	Carex brevior (Dewey) Mackenzie ex Lunell
Cerastium fontanum Baumg, ssp. triviale (Link)	Carex cephaloidea Dewey
Jaias	Carex communis L. Bailey
Dianthus armería L.	Carex crimita Lam.
Silene vulgaris (Moench) Garcke	Carex cristatella Britton
Stellaria longifolia Muhlenb. ex Willd.	Curex dewevana Schwein.
CELASTRACEAE	Carex divitalis Willd.
Celastrus scundens L.	Carex gracillima Schwein.
CLUSIACEAE	Chear geomhais Muhlenh ex Willd
Hypericum perforatum L.	ti mer histifalis Maaltanzia
CONVOLVULACEAE	Caret Writebookieter Deuter
Calystegia sepium L.	Caree nuchcockiana Dewey
CORNACEAE	Carex intumescens Rudge
Cormis alternifolia L.S.	Carex lanuginosa Michaux
Cornus foemina Miller ssp. racemosa (Lam.) J.S.	Carex faxiflora Lam.
Wilson	Carex peckii Howe
Cornus rugosa Lam.	Carex pedunculata Muhlenb, ex Willd.
Cormes stolonifera Michaux	Carex pensylvanica Lam.
CUCURBITACEAE	Carex plantaginea Lam.
Echinocystis lobata (Michaux) Torrey & A. Gray	Carex platyphylla J.Carey
Sieyos angulatus L.	Carex prairea Dewey
CUPRESSACEAE	Curex projecta Mackenzie
Thuja accidentalis L.	Curex pseudo-cyperus L.
CYPERACEAE	Carex radiata (Wahlenb.) Small
Carex albursina E. Sheldon	Carex retrorsa Schwein.
Carex alopecoidea Tuckerman	Carex rosea Schk. ex Willd.
Carex arctata Boott	Carex sparganioides Muhlenb. ex Willd.
Carex backii F. Boott	Carex stipata Muhlenb. ex Willd.
Carex bebbii (L. Bailey) Olney ex Fem.	Carex tenera Dewey
Carex blandu Dewey	Carex tribuloides Wahlenb.

Carex vulpinoidea Michaux

Carex woodii Dewey

Carex specimen D719

Carex specimen D858

Carex specimen D868

Carex specimen D870

Carex specimen D879

Carex Section Ovales

Carex species

Scirpus atrovirens Willd.

DENNSTAEDTIACEAE

Pteridium aquilinum (L.) Kuhn

DRYOPTERIDACEAE

Athyrium filix-femina (L.) ssp. angustum (Willd.) Clausen

Athyrium thelypterioides (Michaux) Desv.

Cystopieris bulbifera (L.) Bernh.

Cystopteris fragilis (L.) Bernh.

Cystopteris tenuis (Michaux) Desv.

Dryopteris carthusiana (Villars) H.P. Fuchs

Dryopteris cristata (L.) Gray

Dryopteris intermedia (Muhlenb. ex Willd.) A. Gray

Dryopteris marginalis (L.) A. Gray

Gymnocarpium dryopteris (L.) Newman ssp. *dryopteris*

Matteuccia struthiopteris (L.) Tod.

Onoclea sensibilis L.

Polystichum acrostichoides (Michaux.) Schott

EQUISETACEAE

Equisetum arvense L. Equisetum hvemale L. ssp. affine (Engelm.)Stone Equisetum laevigatum A. Braun Equisetion scirpoides Michaux ERICACEAE Faccinium angustifolium Aiton FABACEAE Amphicarpaea bracteata (L.) Fern. Desmodium glutinosum (Muhlenb, ex Willd.) DC ex Loudon Medicago lupulina L. Melilotus alba Medikus Melilotus officinalis (L.) Pallas Rohma pseudoacacia L. Trifolium repens L. Vicia cracca L. FAGACEAE Fagus grandifolui Ehrh. Ouercus alba L. Quercus macrocarpa Michaux Quercus rubra L. Quercus species FUMARIACEAE Dicentra canadensis (Goldie) Walp. Dicentra cucullaria (L.) Bernh. GENTIANACEAE Gemiana andrewsii Griseb. GERIANIACEAE Geranium maculatum L.

Geranium robertianum L.

GROSSULARIACEAE

Ribes americanum Miller

Ribes cynosbati L.

Ribes glandulosum Grauer

Ribes lacustre (Pers.) Poiret

Ribes rubrum L.

Ribes triste Pall.

Ribes specimen D827

HYDROPHYLLACEAE

Hydrophyllum virginianum L.

IRIDACEAE

Iris versicolor L.

Iris species

JUGLANDACEAE

Carya cordiformis (Wangenh.) K. Koch

JUNCACEAE

Juncus tenuis Willd.

LAMIACEAE

Galeopsis tetrahit L. Leonurus cardiaca L. ssp. cardiaca Lycopus americanus Muhlenb. ex Bartram Lycopus uniflorus Michaux Mentha arvensis L. Prunella vulgaris L. Scutellaria lateriflora L. LILLACEAE Allium tricoccum Aiton

Clintonia borealis (Aiton) Raf.

Ervthronium americanum Ker Gawler ssp. umericanum Maionthemum canadense Desf. Malanthemum racemosum (L.) Link ssp. racemosum Maianthemum stellatum (L.) Link Polygonatum pubescens (Willd.) Pursh Streptopus roseus Michaux Trillium erectum L. Trillium grandiflorum (Michaux) Salisb. Uvularia grandiflora Smith LYCOPODIACEAE Lycopodium annotinum L. J.vcopodium dendroideum Michaux Excopodium obscurum E.var. obscurum Lycopodium tristachyum Pursh MENISPERMACEAE Memspermum canadense L.

MONOTROPACEAE Monotropa hypopithys L.

Monotropa uniflora L.

OLEACEAE

Fraxinus americana L.

Fraxinus nigra Matshall

Fraxinus pennsylvanica Marshall

ONAGRACEAE

Circea alpina L.

Circea lutetiana L. ssp. canadensis (L.) Aschers. & Magnus

Epilobium ciliatum Raf.

POACEAE Epilohium coloratum Biehler Agrostis gigantea Roth Epilobium leptophyllum Raf. Agrostis stolonifera L. Epilobium parviflorum Schreber **OPHIOGLOSSACEAE** Brachvelvtrum erectum (Schreber in Roth ex-Sprengel) P. Beauv. Botrychium matricariaefolium A. Braun ex Koch Bromus inermis Levsser Borrychium multifidum (S. Gmelin) Rupr. Cinna latifolia (Trevir.ex Goeppinger) Griseb. in Botrychium virginianum (L.) Sw. Ledeb. **ORBANCHACEAE** Dactylis glomerata L. Epifagus virginiana (L.) Barton Danthonia spicata (L.) P. Beauv. ex Roemer & Schultes ORCHIDACEAE Elymus repens (L.) Gould Cypripedium calceolus L. Elymus virginicus L. Liparis loeselii (L.) Rich. ex Lindley Festuca arundinacea Schreber OSMUNDACEAE Festuca pratensis Hudson Osminda claytoniana L. Festuca rubra L. Osmunda regalis L. Festuca subverticillata (Pers.) E. Alexeev. OXALIDACEAE Glyceria striata (Lam.) A. Hitch. Oxalis stricta L. Hystrix patula Moench Oxalis species Leersia orvzoides (L.) Sw. PAPAVERACEAE Leersia virginica Willd. Sanguinaria canadensis L. Milium effusum L. PINACEAE Midilenbergia frondosa (Poitet in Lam.) Fern. Abies balsameg (L.) Miller Muhlenbergia mexicana (L.) Trin. Picea glauca (Moench) Voss Orvzopsis asperifoliu Michaux Pinus strobus L. Panicum acuminatum Sw. Tsuga canadensis (L.) Carriere Panicum capillare L. PLANTAGENACEAE Phalaris arundinacea L. Plantago lanceolata L. Phleum pratense L. Plantago major L. Pou alsodes A. Gray Plantago rugelii Decne.

Poa compressa L.

Poa palustris L.

Poa pratensis L. ssp. pratensis

Poa saltuensis Fern, & Wieg.

Poa species

Schizachne purpurascens (Torrey) Swallen ssp. purpurascens

Sphenopholis intermedia (Rydb.) Rydb.

POLEMONIACEAE

Phlox species

POLYGALACEAE

Polygala paucifolia Willd.

POLYGONACEAE

Polygonum persicaria L.

Rumex orbiculatus A. Gray

PORTULACACEAE

Claytonia caroliniana Michaux

PRIMULACEAE

Lysimachia ciliata L. Lysimachia nummularia L. Lysimachia terrestris (L.) Britton. Stems & Pogg. Trientalis horealis Raf. ssp. borealis

PTERIDACEAE

Adiantum pedatum L. ssp. pedatum

PYROLACEAE

Chimaphila umbellata (L.) Barton *Pyrola elliptica* Nutt.

RANUNCULACEAE

Anemone canadensis L.

Anemone quingefolia L. Anemone virginiana L. Aquilegia canadensis L. Calthus palustris L. ssp. palustris Clematis virginiana L. Hepatica acutiloba DC. Ranunculus abortivus L. Rananculus acris L. Ranunculus hispidus Michaux Rommeulus recurvatus Poiret ex Lam. Thalictrum dioicum L. Thalictrum pubescens Pursh RHAMNACEAE Ceanothus americanus L. Rhammus alnifolia L'Her. Rhammas cathortica L. ROSACEAE Agrimonia gryposepala Wallr. Amelanchier arborea (Michaux f.) Fem. Amelanchier interior Nielson Amelanchier species Crataegus species #1 ('rataegus species #2' Crataegus species #3 Crataegus species Fragaria vesca L. ssp. americana (Porter) Staudt Fragaria virginiana Miller Geum allepicum Jacq. Geum laciniatum Murray

Geum rivale L. Geum urbanum L. Geum species Potentilla norvegica L. Potentilla recta L. Prunus serotina Ehrh. Prunus virginiana L. spp. virginiana Rosa blanda Aiton Rosa palustris Marshall **Rubus allegheniensis Porter** Rubus idaeus L. Rubus occidentalis L. Rubus odoratus L. Rubus pubescens Raf. Rubus specimen D840 RUBLACEAE Galium aparine L. Galium asprellum Michaux Galium circaezans Michaux Galium lanceolatum Torrey Galium obtusum Bigelow Galium palustre L. Galium triflorum Michaux Galium species Mitchella repens L. SALICACEAE Populus balsamifera L. Populus grandidentata Michaux Populus tremuloides Michaux

Sulix bebbiana Sarg. Solix discolor Muhlenb. Salix eriocephala Michaux Salix petiolaris Smith SAXIFRAGACEAE Mitella diphylla L. Tiarella cordifolia L. SCROPHULARIACEAE Verbascum thapsus L. Veronica officinalis L. Veronica serpvllifolia L. SMILACACEAE Smilas herbacea L. Smilax hispida Muhlenb. SOLANACEAE Solanum dulcamera L. TAXACEAE. Taxus canadensis Marshall THELYPTERIDACEAE Phegopteris connectilis (Michaux) Watt Thelypteris noveboracensis (L.) Nieuwl. Thelypteris palustris (Salisb.) Schott THYMELAEACEAE Dirca palustris L. TYPHACEAE Typha latifolia L. TILIACEAE Tilia americana L. ULMACEAE

Ulmus americana L.

URTICACEAE

Boehmeria cylindrica (L.) Sw. Laportea canadensis (L.) Wedd. Pilea pumila (L.) A. Gray Urtica dioica L. ssp. gracilis Urtica dioica L. ssp. dioica VERBENACEAE

Phryma leptostachya L. Verbena hastata L. Verbena urticifolia L. VIOLACEAE Viola affinis Le Conte Viola blanda Willd Viola canadensis L. Viola cucultata Aiton Viola labradorica Schrank Viola pubescens Aiton Viola rostrata Pursh Viola sororia Willd. VITACEAE Parthenocissus inserta (A. Kerner) Fritsch

Vita riparia Michaux

2.3.2 Relationships Among Environmental Variables

Relationships among the principal environmental variables are summarized in Tables 2.7-2.12. The edaphic and stand structure variables included in this summary were found to strongly influence the dispersion of species in CCA and DCA ordination space (see Sections 2.3.3 and 2.3.4). The patterns of association reported in this section provide general context for analyses reported in this, and later, chapters. A summary of environmental variables by quadrat is presented in Appendices 8-10.

The distribution of influential variables in relation to soil parent material is presented in Table 2.7. The frequency of edaphic variables varied by parent material. Seasonally dry depressions were more frequent than expected on glacio-fluvial materials, whereas, seasonally moist depressions were more frequent on lacustrine materials and less frequent on glacio-fluvial materials, respectively. Seasonally wet depressions were over-represented on calcareous outwash materials. Soil pH, available calcium, and percent soil organic matter, achieved their highest values on lacustrine parent materials.

In contrast, the frequency of forest cover types rarely differed on soil parent materials. The exceptions were oak - sugar maple stands which were over-represented on glacio-fluvial parent materials. Forest stands on calcareous outwash had the highest mean number of tree saplings 0-4 cm dbh. The latter pattern is more strongly correlated with moist and wet soils than with disturbance (see Table 4.3, Chapter 4).

The distribution of influential variables in relation to soil order is presented in Table 2.8. Seasonally dry depressions were more frequent than expected on Brunisolic soils and less frequent than expected on gleyed Brunisolic and gleyed Luvisolic soils. Seasonally wet depressions were over-represented on gleyed Luvisolic soils. Soil pH was highest on gleyed Brunisolic soils whereas available calcium and percent soil organic matter reached their highest mean values on Gleysolic soils.

Brunisolic soils were over-represented on glacio-fluvial parent materials, whereas Luvisolic soils were over-represented on calcareous outwash and under-represented on glacio-fluvial materials.

Table 2.7. Distribution of selected environmental variables by soil parent material. Cell values: number of quadrats with specified attribute (mean value of attribute for continuous variables). Chi-square tests of homogeneity, by row (categorical variables): Wilcoxon rank sum tests, independent samples, by row (continuous variables). Cell values in bold when differences among parent materials significant at p<0.05 after Bonferroni correction for number of cell or row tests in category. GF=glacio-fluvial, CT=calcateous till: L=lacustrine: CO=calcateous outwash, n = number of quadrats in category. Note: oak cover type excluded from analysis.

	Soil Parent Material							
Variable	,	GF		CT	L		CO	
	(n=40) obs. exp.		ubs.	exp.	(n=8) obs. exp.		obs.	exp.
SOIL ATTRIBUTES		_						
seasonally dry depressions	40	24.6	57	63.9	0	4,9	21	24.6
seasonally moist depressions	Q	8.8	32	22.8	8	1.8	2	8.8
seasonally wet depressions	0	6.7	15	17.3	0	1.3	17	6.7
soil pH	5,9			6.8	6.9		6.7	
available calcium (cmol/kg)	5.E			14.1	58.5		8.1	
% soil organic matter	5.5		14.5		69.2		LO.[
SOIL ORDER								
brunisol	40	14.6	26	37.9	0	2.9	4	14.6
gleyed brunisol	0	5.4	22	14.1	0	 	4	5.4
luvisot	0	[].9	32	30.9	0	2.4	25	11.8
gleyed luvisol	0	6.5	16	16.8	8	1.3	7	6.5
gleysol	0	1.7	8	4.3	0	0.3	0	2.9
FOREST COVER TYPE								
oak		0		0		0		1
oak + sugar maple	30	[1.3	20	29.3	0	2.3	0	2.9
sugar maple	10	15.4	44	40.1	1	3.1	0	2.9
sugar maple + wet mesic/wet	0	10.4	35	27.1	6	2.1	0	2.9
wet mesic /wet	0	2.7	j	7.0	I	0.5	0	2.9
STAND STRUCTURE								
% canopy closure		87.7		86.5	85.8			82.E
# live stems 0-4cm dbh		28.8		42.1	59.0		60.0	

Table 2.8. Distribution of selected environmental variables by soil order. Cell values: number of quadrats with specified attribute (mean value of attribute for continuous variables). Chi-square tests of homogeneity, by row (categorical variables): Wilcoxon rank sum tests, independent samples, by row (continuous variables). Cell values in bold when differences among soil orders significant at p<0.05 after Bonferroni correction for number of cell or row tests in category. B=brunisol: gB=gleyed brunisol: L=luvisol: gL=gleyed luvisol: G=gleysol. n = number of quadrats in category. Note: oak cover type excluded from analysis.

					A						
					Soil	Order					
Variable	₿		I	gB		Ĺ		gĹ		G	
	(n	=70)	(п	(n=26)		(n=57)		(n=31)		(n=8)	
SOIL ATTRIBUTES											
seasonally dry depressions	67	43.0	1	16.0	50	35.0	0	19.1	0	4.9	
seasonally moist depressions	0	15.3	19	5.7		12.5	18	6.8	4	1.8	
seasonally wet depressions	3	11.7	6	4.3	6	9.5	13	5.2	4	1.3	
soil pH	6.4			7.6		6.4		6.7		7.2	
available calcium (cmol/kg)	6.8			16.5		6.5		28.1		37.4	
% soil organic matter		6.3 14.0		4.0	8.9		33.6		41.0		
SOIL PARENT MATERIAL											
glacial-fluvial	-10	14.6	0	5.4	0	11.9	0	6.5	0	1.7	
calcareous till	26	37.9	22	14.[32	30.9	[6	16.8	8	4.3	
lacustrine	0	2.9	. 0	1,1	0	2,4	8	[.3	0	0.3	
calcareous outwash	4	14.6	- 4	5.4	25	11.9	7	6.5	0	1.7	
FOREST COVER TYPE											
oak		0	0		1		0			0	
oak ÷ sugar maple	45	19.9	0	7.4	9	15.8	0	8.8	0	2.3	
sugar maple	19	27.1	4	10.1	43	21.7	8	12.0	0	3.1	
sugar maple + wet mesic/wet	2	18.3	18	6.8	4	14.7	22	8.I	4	2,1	
wet mesic /wet	4	4.8	4	8.	0	3.8	Ι	2.1	4	0.5	
STAND STRUCTURE											
% canopy closure		85.4	:	83.6	87.6		84.0		90.1		
# live stems 0-4cm dbh		39.7		70.8	4	42.6		40.9		13.5	

Table 2.9. Distribution of selected environmental variables by soil moisture class. Cell values: number of quadrats with specified attribute (mean value of attribute for continuous variables). Chi-square tests of homogeneity, by row (categorical variables): Wilcoxon rank sum tests, independent samples, by row (continuous variables). Cell values in bold when differences among moisture classes significant at p<0.05 after Bonferroni correction for number of cell or row tests in category. Dry=quadrat with seasonally dry depressions; Moist=quadrat with seasonally moist depressions: Wet=quadrat with seasonally wet depressions. n = number of quadrats in category. Note: oak cover type excluded from analysis.

	1	S	ioil M	loisture Clas	S		
Variable	Dry		Moist		Wet		
	(n=11/) obs. exp. a		obs.	(n=42) exp.	obs.	(n=33) exp.	
SOIL ATTRIBUTES		-	I	-		· · ·	
soil pH		6.1		7.1		6.7	
available calcium (cmol/kg)		6.6	31.4			16.6	
% soil organic matter	7.3		27.7		20.8		
SOIL PARENT MATERIAL			• • • •				
glacial-fluvíal	40	24.6	0	8.8	0	6.7	
calcareous till	57	63.9	32	22.8	15	17.3	
lacustrine	0	4.9	8	1.8	0	£.[
caleareous outwash	21	24.6	2	8.8	17	6.7	
SOIL ORDER	*		,				
brunisol	67	43.0	0	15.3	3	11.7	
gleyed brunisol	1	16.0	19	5.7	6	4.3	
luvisol	50	35.0	1	12.5	6	9.ĵ	
gleyed luvisol	0	19.1	18	6.8	13	5.2	
gleysol	0	1.9	4	1.8	4	1.3	
FOREST COVER TYPE							
oak		1		0		0	
oak + sugar maple	54	33.1	0	16.3	0	9.0	
sugar maple	59	45.3	[0]	11.0	5	12.4	
sugar maple + wet mesic/wet	3	30.6	29	11.0	18	8.4	
wet mesic /wet	E	8.0	3	2.6	9	2.2	
CANOPY CLOSURE	-						
% canopy closure		87.7		86.0	80.7		
# live stems 0-4cm dbh		39.9		50.3	49.6		

Table 2.10. Distribution of selected environmental variables by forest cover type. Cell values: number of quadrats with specified attribute (mean value of attribute for continuous variables). Chi-square tests of homogeneity, by row (categorical variables); Wilcoxon rank sum tests. independent samples, by row (continuous variables). Cell values in bold when differences among cover types significant at p<0.05 after Bonferroni correction for number of cell or row tests in category. Cover Type: 1 = red or white oak, no sugar maple: 2 = red, white oak + sugar maple: 3 = sugar maple, no red, white oak, no wet mesic or wet tree species; 4 = sugar maple + black ash, silver maple or American elm; 5 = black ash, silver maple, American elm, no sugar maple, no red, white oak. Cover Type 1 excluded from analysis. n = number of quadrats in category

	Forest Cover Type								
Variable	1 (n=1) obs. exp.	2 (n=54) obs. exp.		3 (n=75) obs. exp.		4 (n=49) obs. exp.		5 (n=13) obs. exp.	
SOIL ATTRIBUTES									
seasonally dry depressions	1	54	33.1	59	45.3	3	8.4	1	1.2
seasonally moist depressions	0	0	11.9	10	16.3	29	11.0	3	2.9
seasonally wet depressions	0	0	9.0	5	12.4	18	8.4	9	2.2
soil pH	6.2	6.1		6.6		6.9		7.4	
available calcium (cmol/kg)	2.1	4.6		10.3		23.0		29.8	
% soil organic matter	3.l	6.1		11.1		22.9		29.9	
SOIL PARENT MATERIAL									
glacial-fluvial	0	30	í 1.3	10	15.5	Û	10.5	0	2.7
calcareous till	0	20	29.4	44	40.3	35	27.2	5	7.1
lacustrine	0	0	2.3	I	3.1	6	2.1	1	0.5
calcareous outwash	1	4	11.0	19	15.1	9	10.2	7	2,7
SOIL ORDER									
brunisol	0	-45	19.8	19	27.1	2	18.3	4	4.8
gleyed brunisol	0	0	7,4	4	10.1	18	6.8	4	1.8
luvisol	I	9	15.8	-43	21.7	4	14.6	0	3.8
gleyed luvisol	0	0	8.8	8	12.0	22	8.1	l	2.1
gleysol	0	0	2.3	0	3.1	4	2.1	4	0.6
Table 2.11. Attributes of forest stand structure by forest cover type. Cell values: mean percent of taxa in 10m x 10m quadrats with specified attributes (mean number where noted): Wilcoxon rank sum tests, independent samples, by row (Cover Type 1 excluded from analysis); non-parametric median tests, by row (Cover Type 1 excluded from analysis). Highest value in bold when differences among cover types significant at p<0.05, after Bonferroni correction for number of tests in attribute group. Cover Type: 1 = red or white oak, no sugar maple; 2 = red, white oak + sugar maple; 3 = sugar maple, no red, white oak, no wet mesic or wet tree species; 4 = sugar maple + black ash, silver maple or American elm; 5 = black ash, silver maple, American elm, no sugar maple, no red, white oak, n = number of quadrats in category.

		F	orest Cover Ty	pe	
Stand Attribute	1 (n=1)	2 (n=54)	3 (n=75)	4 (n=49)	5 (n=l3)
# LIVE TREE STEM	S > } m				
mean number	[6	53.9	49,7	68.0	59.9
median number	-	46.0	42.0	65.0	47.0
range	0	9-121	9-168	9-152	9-[8]
DBH SIZE CLASS					
% stems 0-4 cm	75.0	68.4	71.0	76.1	83.5
% stems 4-10 cm	25.0	16.2	17.3	13.4	11,4
% stems 10-30 cm	Û	13.1	9.3	8.5	2.9
% stems >30 cm	0	2.3	2.4	2,0	2.2
maximum DBH	8-10 cm Red Oak	88.0 cm White Oak	57.1 cm A. Beech	94.5 cm Silver Maple	46.4 Silver Maple
# LIVE TREE SPEC	IES > 1 m		_		
mean # species	8	6.7	4.5	7.8	5.6
median # species	-	6,5	4.0	8.0	5.0
range	0	2-12	1-10	2-12	[-12
% CANOPY CLOS	URE				
mean % closure	88.5	86.5	87.9	85.9	70.4
median % closure	-	88.5	90.5	85.9	76.8
range	0	48.3-98.2	53.3-99.6	57.4-99.0	39.2-97.2

Table 2.12. Distribution of selected environmental variables by patch size. Cell values: number of quadrats with specified attribute (mean value of attribute for continuous variables). Chi-square tests of homogeneity, by row (categorical variables): Wilcoxon rank sum tests, independent samples, by row (continuous variables). Cell values in bold when differences among size classes significant at p<0.05, after Bonferroni correction for number of cell or row tests in category. Large=>122 ha. Intermediate=43-121ha. Small= \le 42 ha. n = number of quadrats in category. Note: oak cover type excluded from analysis.

			Patch	Size Class		
Variable	I	Large	inte	ermediate		Small
	obs.	exp.	obs.	exp.]	obs.	(11-49) exp.
SOIL MOISTURE						· •···•
seasonally dry depressions	32	29.5	65	54.1	21	34.4
seasonally moist depressions	16	10.5	[5	19.3	l i	12.3
seasonally wet depressions	0	8.0	8	14.7	24	9.3
SOIL PARENT MATERIAL			_			
glacial-fluvial	16	10.0	24	18.3	Û	11.7
calcarcous till	16	26.0	56	47.7	32	30.3
lacustrine	8	2.0	0	3.7	0	2.3
calcareous outwash	8	10.0	8	18.3	24	(1.7
SOIL ORDER						
brunisol	24	17.5	40	32.1	6	20.4
gleyed brunisol	0	6.5	16	11.9	10	7.6
luvisol	8	14.3	24	26.1	25	[6.]
gleyed luvisol	16	7.8	0	14.2	15	9.0
gleysol	0	2.0	8	3.7	0	2.3
FOREST COVER TYPE		· · · · -	· _			
oak		<u> </u>		0	Ι	0
oak + sugar maple	24	13.3	30	24.9	0	15.8
sugar maple	13	18.2	38	34.1	23	21.7
sugar maple + wet mesic/wet	9	12.3	16	23.0	25	14,7
wet mesic /wet	1	3.2	4	5.0	8	3.8

Gleyed Luvisolic soils were over-represented on lacustrine deposits. Forest stands composed of oak and sugar maple were over-represented on Brunisolic soils, whereas sugar maple stands, *sensu stricto*, were over-represented on Luvisolic soils. Stands composed of sugar maple and wet mesic or wet trees were over-represented on gleyed Luvisolic soils, whereas stands composed of wet mesic and wet species were over-represented on Gleysolic soils. Forests on gleyed Brunisolic soils had the highest mean number of live tree stems 0-4 cm dbh. The latter pattern was more strongly correlated with soil moisture than with disturbance.

The distribution of influential variables in relation to soil moisture is summarized in Table 2.9. Soil pH, available calcium, and percent soil organic matter, achieved their highest mean values in quadrats with seasonally moist depressions. Stands composed of oak and sugar maple were over-represented in quadrats with dry depressions and were absent from quadrats with seasonally moist or seasonally wet depressions. Stands composed of sugar maple with wet mesic and wet species were over-represented in quadrats with seasonally moist, or seasonally wet, depressions. Stands composed of wet mesic and wet tree species were over-represented in quadrats with seasonally moist, or seasonally wet, depressions. Stands composed of wet mesic and wet tree species were over-represented in quadrats with seasonally moist.

The distribution of influential variables in relation to forest cover type is summarized in Tables 2.10 and 2.11. Soil pH, available calcium, and percent soil organic matter, achieved their highest mean values in stands dominated by wet mesic and wet trees. Patterns related to soil moisture, soil parent material and soil order were discussed previously. Stand structure variables also varied by forest cover type. Stands composed of sugar maple with oak had the highest mean percentage of stems in 10-30 cm dbh class, whereas stands composed of sugar maple. *sensu stricto*, had the highest mean, and median, percent canopy closure. Stands composed of sugar maple with wet mesic or wet species had the highest mean number of live stems, and, the highest mean, and median, number of tree species.

The distribution of influential variables in relation to patch size is summarized in Table 2.12. The greatest contrast was between large and small patches. Small patches were over-represented on

calcareous outwash materials and under-represented on glacio-fluvial materials. In keeping with this pattern, seasonally wet depressions were more frequent in small patches than in patches of intermediate or large size. Sugar maple stands with red or white oak were over-represented in large patches where soils were consistently drier than in patches of small or intermediate size.

2.3.3 Species Response to Environmental Gradients

The distribution of quadrats and patches in ordination space (DCA) is presented in Figure 2.1. In general, the species composition of quadrats tended to be more similar within, than among, forest patches. This tendency is revealed by the clustering of quadrats in patches 15, 18, and 22, lower right portion of the diagram, and by the clustering of quadrats in patches 10, 14 and 12, middle portion of the diagram. Canopy closure and moisture conditions within these patches are more uniform than in patches with quadrats that are broadly dispersed (patches 1, 3, 4, 6).

The species composition of quadrats with moist or wet depressions, seeps, and moist or wet forest floors, was typically more similar than the species composition of quadrats in upland settings. This pattern is revealed by tight quadrat clusters from dissimilar patches (e.g. quadrats from patches 4 and 20, 20 and 24, 19 and 24, 6 and 24, 6 and 19, 2 and 12, 4 and 12, 12 and 16, 2 and 6, 3 and 21, lower left portion of the diagram). A similar tendency was also observed in selected quadrats with closed forest floors or canopy gaps on dry soils (see tight clusters formed by quadrats from patches 1 and 7, 9 and 18, 15 and 18, 15 and 23, 15 and 22, and, 8 and 23, lower right portion of the diagram). Several of these clusters had similar parent materials and soil orders.

Taken together, these trends suggest that the primary influence on species composition has been similarity in site conditions. However, when edaphic conditions within patches are similar, other factors such as within-patch dispersal and disturbance history may intensify the similarity in species composition within patches. The similarity in species composition in moist and wet habitats suggests that species have had an opportunity to sample most habitats within the regional landscape during the post-glacial period. Pattern in the distribution of species in the present-day landscape, therefore, may primarily reflect historic opportunities for germination and persistence. The



DCA AXIS 1

Figure 2.1. Distribution of forest patches (N=24) in relation to DCA axes 1 and 2. Note the degree to which quadrats in each patch (N=8) cluster in ordination space.

availability of propagules appears not to have been a constraint, when viewed on long time scales.

The distribution of species in CCA ordination space (Figure 2.2) was strongly influenced by gradients in soil moisture, soil order, forest cover type, percent canopy closure, and soil parent material (see inter-set correlations, Table 2.13). The ordination explains 25.6% of the total inertia in species scores (F=3.748, p<0.001. Monte Carlo permutation test, n=9,999 permutations), after fitting patch membership as a co-variable: the first and second axis of the ordination collectively explain 42.8% of the variance explained by submitted environmental variables.

The first axis is a complex gradient dominated by soil moisture, cover type, and soil parent material (see inter-set correlations, Table 2.13). The second axis is a complex gradient dominated by soil order and percent canopy closure. Species on the upper left side of the diagram achieved their maximum fitted abundance in closed, dry, maple-oak stands, on Brunisolic soils overlying glacio-fluvial and calcareous till parent materials. In contrast, species on the upper right side of the diagram achieved their maximum fitted abundance in open, wet forest stands dominated by wet-mesic and wet tree species, on gleyed Brunisolic soils overlying calcareous outwash. Species on the lower right side of the diagram achieved their maximum fitted abundance in closed, and open, wet forest stands dominated by sugar maple and wet-mesic or wet tree species, on Gleysolic or gleyed Luvisolic soils overlying calcareous outwash parent materials. Species in the left-of-centre region of the ordination diagram achieved their maximum fitted abundance in closed, dry, sugar maple stands on Luvisolic and gleyed Brunisolic soils overlying calcareous outwash parent materials. Species in the right-of-centre region of the ordination diagram achieved moist and wet stands dominated by sugar maple with wet-mesic or wet species, on gleyed Luvisolic and gleyed Brunisolic soils overlying calcareous till and calcareous outwash parent materials. Species in the right-of-centre region of the ordination diagram achieved their maximum fitted abundance in closed, moist and wet stands dominated by sugar maple with wet-mesic or wet species, on gleyed Luvisolic and gleyed moist and wet stands dominated by sugar maple with wet-mesic or wet species, on gleyed Luvisolic and Luvisolic soils overlying calcareous till and lacustrine parent materials.

All but four of the sixteen environmental variables had a significant influence on the dispersion of fitted species scores, after fitting all remaining variables as co-variables, and, after Bonferroni correction for n=16 Monte Carlo permutation tests (n=1,000 permutations) (Table 2.13). The non-significant variables were: # trees 0-4 cm dbh, # trees 4-10cm dbh, # microhabitats, and disturbed

Legend Figure 2.2

<u>Annotation</u>	Environmental Variable	Arrow Points To:
MOIST	moisture class	seasonally dry depressions
PATCHarea	patch area	largest patches
10-30em	live tree stems 10-30 cm dbh	quadrats with highest # stems
CANCLOS	% canopy closure	high % closure (low light)
PI	patch isolation	high patch isolation
so	soil order	gleysols
#MH	# microhabitats	quadrats with highest # microhabitats
SOM	% soil organic matter	high % soil organic matter
SPM	soil parent material	calcareous outwash
омн	open microhabitats s.L	open microhabitats
СТ	cover type	wet mesic-wet trees (no sugar maple)
рH	soil pH	high soil pH
DIST	disturbance s.l.	quadrats with trails, regenerating
		fields, canopy gaps
#TS	# tree species	quadrats with highest # species
0-4cm	live tree stems 0-4cm dbh	quadrats with highest # stems

Ordination Details

- 1. Species in ordination space are located at the centroid of quadrats (not shown) that have the highest fitted mean abundance for the species (Scaling Mode 2).
- 2. The co-ordinates for the heads of arrows are the bi-plot scores for Axis 1 and Axis 2.
- 3. Patch membership was designated a co-variable: the effect of replicated samples within forest patches has therefore been removed from the ordination.
- 4. The submitted environmental variables explain 25.6% of the variance in species composition in the ordination (100 x Σ canonical eigenvalues - Σ unconstrained eigenvalues). The first and second axis collectively explain 42.8% of the variance explained by the submitted variables. The model and first axis are highly significant: F-ratio (overall test) = 3.748. p< 0.001: F-ratio (Axis I) = 12.074. p<0.001: Monte Carlo permutation tests (n=9.999).



Figure 2.2. Species ordination (CCA) constrained by environmental variables. See legend for description of annotated variables. See PANAQUIN, RANUABOR, KIBATYNO, SANGTANA, HIJAME, PHIJOAT, VRI PERR, 2. ABREBALS, ACERNEGIJ, AJ LITRU, CAREPLAN, CARLI HAL, CIR-Table 2.2 for environmental states at ends of arrows; see Appendix 3 for names of species. Species positions approximate. 1 ASTR-CLUDE, DRYOVART, DRYONDE, FRAGVESC, MILLEFEU, POLYACRO, RUBUODOR, VERBTHAP, J. AMBRARTE, ARCUMINU, CAREKADI, CARE 868, CORD, CARDDIPH, CAREARCT, CAREDEWE, CARELAXI, CAREROSE, CARICARO, CLAYCARO, CHINRORE, EQUISCIR, HYDRVIRG, LACT-CANA, EQUIARVE, TESTPRAT, FRAXPENN, JUNCTENU, JORE SP, FOA ALSO, POFENORV, RUMEORPI, SOLICANA, SOLIRUGO, TARAOFFI, VERBIJAST

Table 2.13. Relative importance of environmental variables in species ordination (CCA). % Total Variance = (Σ canonical eigenvalues to reneation strong all convariable Patch; % Unique =(Σ canonical eigenvalues for variable Σ unconstrained eigenvalues in the ordination) x 100, after fitting the co-variable Patch; % Unique =(Σ canonical eigenvalues to the ordination) x 100, after fitting the co-variable Patch; % Unique =(Σ canonical eigenvalues to the ordination) x 100, after fitting the co-variable Patch; % Unique =(Σ canonical eigenvalues to the ordination) x 100, after fitting all remaining variables as co-variables. Partial F-statistics and P-values for model (n=1000 permutations) after fitting termaining termaining termaining variables as co-variables as co-variables.

012.0	E111	9.0	1.4	961.0	140.0	ndb mo 01-4 soon
100'0	3.04	21	0.5	192.0	\$90'0	soloogs oon ovil 4
100'0	5,48	1'1	5,6	\$00.0-	080.0-	Elq lios
008'0	01.0	0'4	<u></u> 2'1	0.214	0,202	trees 0-4 cm dph
100'0	28.6	575	6'î	0/3/0-	0"51#	naitadosì itoliq
100'0	5.51	t'i	9 " [270.0.	FL2'0-	birtop suce
0£0.0	EE'1	8'0	81	612.0	0.284	statidationoim bodrauzib
06010	1:33	<i>L</i> '0	0.5	68£.0-	762.0	enenidadoroim #
100'0	12.2	£1	0'£	864.0-	976'0	voltara sinagro lios %
100.0	1'14	8.0	61	080'0-	622.0	ddb mo 08-01 soort
100'0	81.5	81	Sitz	872.0	78E.0	soil order
(00)0	991	0.1	5.6	90210	0"443	etabidadoroira neqo
100'0	591	610	3.4	+94.0-	LS#10-	amsola yqonna %
100.0	55.5	6'1	1.5	01110	015'0	loitenn material
100.0	5'30	£"1	4'2	842.0	604 0	cover (ype
100.0	310	8.1	£"†	0194	127.0-	ennisiom tios
aulav-9	P-statistic	supin!1 %	sometrey leto T %	CGV VXI2 5	T sixA ADD	
	bontakyal noiti	isoqmoD solooqB ni	оолытеУ %	noinslatio	later-set C	sldsingV listonnonivn3

microhabitats. The stand structure variables were strongly correlated with sugar maple abundance and soil moisture (see Table 4.3, Chapter 4), whereas, the composition of disturbed microhabitats was dominated by species with an affinity for closed forest conditions (see Figure 2.4 and related discussion). The non-significance of microhabitat number simply reflects the composite nature of this variable. The contribution of patch membership to the dispersion of species scores was significant (F=2.776, p<0.001), however, and accounted for 4.3% of the explained variance, and 1.2% of the total variance, in species scores. Patch membership was therefore designated a covariable, when testing for a treatment effect of environmental variables, to minimize the effects of replicated samples within patches.

The degree to which species were restricted to a particular edaphic condition, forest cover type, or disturbance state is summarized in Table 2.14. In general, the distribution of species was restricted more by differences in soil parent material and soil order than by differences in forest cover type and soil moisture. Comparatively few species were restricted by differences in canopy closure. Only 10.4% and 12.8% of species were present on every soil parent material, and soil order, respectively, whereas, 19.4% and 23.0% of species were present in every forest cover type, and soil moisture class, respectively. In contrast, 69.2% of species were recorded in both open and closed canopy conditions.

Inspection of the percentage of taxa restricted to a given environmental state reveals that proportionally fewer species were restricted to seasonally moist (5.4%) and seasonally wet microhabitats (5.0%) than to seasonally dry microhabitats (40.1%). This suggests that species of seasonally moist and wet habitats were more tolerant of dry conditions than species of dry habitats were of seasonally moist or wet conditions. A greater tolerance of excess moisture may account for the small percentage of species restricted to lacustrine parent materials (1.9%), and gleyed Brunisolic soils (4.7%), since the number of quadrats with seasonally moist depressions was significantly higher on these substrates (Table 2.9).

Species in this data set also showed a pronounced tolerance of gradients in canopy closure. In

Table 2.14. Number of species with a restricted and unrestricted spatial distribution in relation to soil parent material, soil order, soil moisture, cover type, canopy closure. Note: "oak" cover type (N=1 quadrat) included in "oak+sugar maple" for this analysis.

Environmental Variable	# Species	Specia to Envir ≠	es Restricted onmental State %	Speci in Ev #	es Presen very State %	
SOIL PARENT MATERIAL (total)	413	145	35.1	43	10	.4
giacio-fluvial	116	3	11.2		-	
calcareous till	334	70	21.0			
lacustrine	104	2	1.9		-	
calcareous outwash	307	60	19,5		-	
SOIL ORDER	413	110	26.6	53	12.	8
brunsisol	280	42	15.0		-	
gleyed brunisol	253	12	4.7		-	
luvisol	258	31	12.1		-	
gleyed luvisol	198	17	8.6		-	
gleysol	92	8	8.7		-	_
SOIL MOISTURE (total)	413	156	37.8	95	23.	0
seasonally dry microhabitats	337	135	40.1		-	
seasonally moist microhabitats	241	13	5.4		-	
seasonally wet microhabitats	159	8	5.0		-	
FOREST COVER TYPE (total)	413	107	25.9	80	19.	4
oak. and, oak + sugar maple	199	18	9.0		-	
sugar maple	305	30	9.8		-	
sugar maple + wet-mesic, wet	287	30	10.5		-	_
wet mesic + wet	333	29	8.7		•	
CANOPY CLOSURE (total)	413	127	30.8	286	69.	2
open microhabitats	341	55	16.1		-	_
closed microhabitats	358	72	20.1		-	

contrast to other variables, there were more, rather than fewer, species present in every state. This suggests that shade tolerant species are rarely displaced by taxa with an affinity for open habitats during disturbance events.

The relative contribution of patch variables, edaphic variables and matrix variables to the dispersion of species in ordination space is summarized in Figure 2.3. Overall, patch variables, *sensu stricto*, explained more of the variance in composition (36.1%) than did edaphic variables (28.1%) or matrix variables (10.2%). Edaphic variables explained 34.6% of the variance explained by patch variables, *sensu lato* $\{(19.2 \pm 1.4) \pm (36.1 \pm 19.2 \pm 1.4 \pm 2.8) = 34.6\}$, and, 21.7% of the variance explained by matrix variables. *sensu lato* $\{(1.4 \pm 2.2) \pm (10.2 \pm 2.8 \pm 1.4 \pm 2.2) = 21.7\}$. Matrix variables, on the other hand, explained only 7.1% of the variance explained by edaphic variables. *sensu lato*. The implications for the relative contribution of local versus regional processes are discussed in Section 2.4.

2.3.4 Species Response to Microhabitats

Species were responsive to the pragmatic classification of microhabitats used in this study (Figure 2.4). With few exceptions, the microhabitats were well separated in ordination space and were dispersed in relation to the moisture and disturbance gradients that defined them. The similarity in species composition (marked by the proximity of habitats in ordination space) varied among habitats but even the most widely separated habitats had at least some species in common since the length of the first axis was less than 4.0 standard deviation units long (ter Braak 1987). This result is consistent with the broad moisture tolerance of many of the taxa occurring in more than 25 microhabitats (Table 2.15). As expected, the composition of species in microhabitats was 2.737 versus 7.100 for the ordination of quadrats). Taken together, these patterns provide indirect evidence of the role of environmental heterogeneity in structuring the composition of forest patches.

The strongest overlap in species composition occurred in open stumps and closed moist depressions (overlapping annotation in Figure 2.4). This was due to the capacity of species of moist forest floors



Figure 2.3. Partial decomposition of variance in CCA species ordination: local versus regional processes Patch variables (cover type, % canopy closure, stem diameter class, # tree species, # tree stems, # microhabitats, open microhabitats, disturbed microhabitats, patch membership), *sensu stricto*, explain 59.5% of the variance explained by patch variables, edaphic variables (soil parent material, soil order, soil moisture, % soil organic matter), and matrix variables (patch area, patch isolation). Edaphic variables explain 34.6% of the variance in species composition explained by patch variables, *sensu lato*, and 21.7% of the variance explained by matrix variables, *sensu lato*, and 21.7% of the variance in species composition explained by matrix variables, *sensu lato*; matrix variables, in turn, explain 7.1% of the variance explained by patch variables, *sensu lato*. The variables in this model explain 100% of the explained variance, in the original species ordination.

Legend Figure 2.4

Annotation	Microhabitat
tioorD	seasonally dry forest floors-closed canopy (cc)
floorM	seasonally moist forest floors-cc
depM	seasonally moist forest depressions-cc
floorW	seasonally wet forest floors-cc
depW	seasonally wet forest depressions-cc
seep	seep-cc
gapD	seasonally dry gap-cc
gapM	seasonally moist gap-cc
mound	mound-ee
pitD	seasonally dry pit-ce
pitM	seasonally moist pit-ce
pitW	seasonally wet pit-co
log	log-ee
stump	stump-cc
rrm.	raised root mat-cc
stone	stone-cc
lane	lane/road-cc
ditch	ditch-cc
regfield	regenerating field-cc
FLOORd	seasonally dry forest floors-open canopy (oc)
FLOORm	seasonally moist floor-oc
DEPm	seasonally moist forest depressions-oc
DEPw	seasonally wet forest depressions-oc
SEEP	seep-oc
GAPJ	sasonally dry gap-oc
GAPm	seasonally moist gap-oc
GAPw	seasonally wet gap-oc
MOUND	mound-ne
PITd	seasonally dry pit-oc
PITw	seasonally wet pit-oc
LOG	log-oc
STUMP	stump-oc
RRM	raised root mat-oc
LANE	lane/road-oc
DITCH	ditch-oc
REGFIELD	regenerating field-oc
RIPMEADOW	riparian meadow-oc
RIPMARSH	riparian marsh-oc
RIPTHICKET	riparian thicket-oc



Figure 2.4. Distribution of microhabitats in relation to DCA axes 1 and 2. See legend for description of annotated variables. Note: FLOORd = dry open canopy gaps; FLOORm = moist open canopy gaps, Overlupping annotation STUMP, depM.

74

# Microhabitats	# Taxa	% of N=413 taxa
l	58	14.0
2-5	138	33.4
6-10	85	20.6
11-15	52	12.6
16-20	35	8.5
21-25	32	7.7
26-30	11	2.7
31-35	1	0.2
36-39	1	0.2

Table 2.15. Number of microhabitats occupied by surveyed taxa. See Notes for list of taxa present in more than 25 microhabitats.

Notes:

Taxa present in more than 25 microhabitats: Arisaema triphyllum, Aster lateriflorus, Carex gracillima, Circea lutetiana, Epipactis helleborine. Galium triflorum, Glyceria striata, Impatiens capensis. Parthenocissus inserta, Pilea pumila, Prunus virginiana, Solidago canadensis, Taraxacum officinale. to colonize open stumps. Approximately 80% of species on open stumps occurred in closed moist depressions whereas only 20% of species in closed moist depressions occurred on open stumps (Appendix 6, 7). The near overlap with species of open raised root mats is due primarily to the capacity of species of moist forest floors to colonize both stumps and raised root mats since open stumps rarely occurred in quadrats with open raised root mats.

The cluster of microhabitats on the extreme left of the ordination diagram reveals that the species composition of canopy gaps, tip-up mounds and dry tree pits in this data set was very similar to the species composition of closed, dry, forest floors. This suggests that species of the forest floor persist in canopy gaps and readily colonize dry mounds and pits created by wind-thrown trees. In keeping with this interpretation, 85% of the species recorded in the latter habitats (n=246 species) also occurred on closed, dry forest floors. The 36 species which did not occur on closed dry forest floors were found in open dry canopy gaps (30 species), closed mounds (4 species), open mounds (1 species), or closed canopy gaps (1 species). The majority of these taxa (80.6%) were species with affinity for "open" or "open – forest" habitats.

In general, the dispersion of microhabitats in ordination space appears to be influenced by differences in soil moisture, canopy closure and human disturbance. Moisture exerts the strongest influence over the composition of species, based on the tendency for habitats with similar moisture conditions to occur in the same sector of ordination space (seasonally dry habitats typically to the left, seasonally moist and seasonally wet habitats typically to the right). Open and closed habitats of the same type tend to occur together, and, in the sector of ordination space that reflects their moisture status. The tendency for open habitats to occur to the right of closed habitats reflects the shift in species composition arising from the germination requirements of light-demanding species.

The analysis of microhabitats was extended to clarify the influence of gap size on the composition canopy gaps on mesic soils. If colonization events during the gap phase were a significant influence on species composition, then one would expect to see an increasing departure from closed forest floors with increasing gap size, since species of open habitats should increasingly be favored under

conditions of increased light. Gaps in ordination space should therefore cluster in relation to gap size, and, larger gaps should be more remote from the ordination position of "closed dry forest floors" than small gaps. If other factors are more important determinants of the species composition of forest floors, then neither pattern should arise.

The results of this analysis are shown in Figure 2.5. The dispersion of gaps in DCA ordination space is only weakly patterned at best. The smallest gaps ("1", "2") are as likely to be close to the reference condition as distant from it, and, some of the largest gaps ("9", "10") are closer to the reference condition than many of the smaller gaps. If there is a general trend in the ordination, it is for gaps of larger size to be closer to the reference condition, and, for gaps of smaller size to be more distant. Gap size, *per se*, therefore, does not appear to be a significant contributor to the species composition of forest floors.

The factors responsible for this "pattern" are not readily apparent. Gaps in closest proximity to the reference condition tend to occur on calcareous till and outwash parent materials, whereas gaps remote from the reference condition tend to occur on glacial-fluvial parent materials and on calcareous till. Remote gaps on calcareous till occur exclusively on gleyed Luvisolic soils whereas proximate gaps on calcareous till occur on soils that are not gleyed. Differences in the species composition of gaps, therefore, may largely be due to differences in edaphic conditions.

The largest gaps in this analysis (labeled "10") differed sharply in the number of taxa with the capacity to persist in soil seed banks. The proportion of taxa with the capacity for prolonged dormancy in the gap nearest to the reference condition was <6% whereas the proportion of taxa in the gaps in the lower right portion of the diagram was >21%. The former gap was situated within an otherwise closed forest whereas the latter gaps were situated at the edge of a recently clear-cut forest stand. Differences in the species composition of these gaps , therefore, may be due in part to differences in seed bank size or in the degree of seed bank germination. If the latter, then large gaps (>>100m²) may be a significant contributor to the transient composition of species on forest floors.



Figure 2.5. DCA ordination of open, seasonally dry, canopy gaps. Legend: gap size class in 10m² increments (numbered 1 to 10); the microhabitat "closed, seasonally dry, forest floors" (floorD) has been included in the ordina-tion for reference purposes.

ž

The degree to which species were restricted to a given type of microhabitat is summarized in Figure 2.6. Four broad habitat categories were used for this analysis: closed, seasonally dry, forest floors *sensu stricto*; natural disturbance features (canopy gaps, tree-pits, tip-up mounds, stumps, logs); features created by human disturbance (regenerating fields, lanes, ditches); and, moist or wet habitats (forest floors, depressions, seeps, riparian meadow, riparian marsh, riparian thicket). Most species (79.2%) were recorded in more than one habitat category and approximately one-third of species (33.2%) were found in every category.

Species in this study were particularly tolerant of conditions created by the death or removal of a canopy tree (canopy gaps, tree pits, tip-up mounds, stumps, logs). Such features were colonized by 86.0% of the species on closed, dry, forest floors, 82.0% of the species in moist or wet habitats, and, 78.0% of the species in habitats created by human disturbance. However, the capacity of species to colonize or persist on features created by natural disturbance were not uniform. More species were recorded in canopy gaps (282) than on tip-up mounds (180), logs (120), pits (97), or stumps (64) (Appendix 5). This suggests that most species in these forests have access to alternative habitats where their competitors may do poorly or not survive, and, where populations of their own kind may expand.

Approximately 20% of the sampled flora (90 species) were restricted to features of one category. In keeping with the dispersion of microhabitats in Figure 2.4, restricted species were more constrained by moisture (42 species) than by human or natural disturbance (32 and 16 species, respectively).

2.3.5 Response of Sugar Maple and Understory Herbs to Available Calcium

The response of sugar maple to the availability of calcium eations in the upper 15 cm of the soil profile was examined to determine if there was evidence of increased survivorship of sugar maple saplings on calcium rich soils. The stand structure of undisturbed maple stands was evaluated in relation to increasing calcium availability on mesic soils overlying calcareous till. Preliminary analysis had revealed that soils of the Luvisolic order were typically more calcium rich than soils



Moist or Wet Habitats

Figure 2.6. Distribution of species among microhabitats. Annotation: number of species recorded in habitat category. Natural Disturbance = canopy gaps, tree pits. tip-up mounds, stumps, logs (in dry, moist or wet conditions): Human Disturbance = regenerating fields. lanes, ditches; Moist or Wet Habitats: floors, depressions, seeps, riparian meadow, riparian marsh, or riparian thicket. Unique Species: species restricted to features in habitat category.

of the Brunisolic order, and, that the latter soils often contained free calcium carbonate in the upper 15 cm of the soil profile. Sugar maple response was therefore evaluated in relation to non-reactive soils when the analysis contained more than one soil order.

The relative abundance of sugar maple on non-reactive soils is presented in Figure 2.7. The percentage of sugar maple stems increased with increasing calcium availability in all size classes. The response was significant in the 0-4 cm, 4-10 cm, and 10-30 cm size class. Calcium availability explained 13 % to 19 % of variance in sugar maple abundance in these size classes. The trends in absolute abundance (not shown) were similar in direction but weaker (only the response for the 10-30 size class was statistically significant).

The relative abundance of sugar maple on Brunisolic soils is presented in Figure 2.8. The samples in this analysis were from the same soil order and soil series, and, thus were the most uniform with respect to the degree of soil weathering and soil development. Forest stands were typically younger than in the preceding analysis and did not contain any stems in the >30 cm size class. The calcium gradient was approximately 50 % longer owing to the inclusion of reactive soils. As before, the percentage of sugar maple stems increased with increasing calcium availability in the 0-4 cm, 4-10 cm, and 10-30 cm size classes. The percent variance in sugar maple abundance explained by available calcium was much higher, however, and ranged from 54 % in the 0-4 cm size class to 35 % in the 10-30 cm size class. The trends in absolute abundance (not shown) were strongest in the 4-10 cm size class, where differences in calcium availability explained 70.9 % of the variance in the number of sugar maple stems in 10m x 10m quadrats. In contrast to previous results, available calcium differences in the number of sugar maple stems in the 0-4 cm size class.

Taken together, these results provide indirect evidence of increased survivorship of sugar maple stems in undisturbed second-growth stands on calcium rich, mesic, soils overlying calcareous till. Caution is required, however, since the sample size in each analysis was small. The contribution of quadrats without stems to the significance of the relationship is strong. In many analyses, this would be cause for concern. In this analysis, however, the absence of stems means that maturing sugar



Figure 2.7. Sugar maple abundance versus available calcium in upper 15 cm of soil profile on Brunisolic and Luvisolic soils overlying calcareous till. Response variable is the percentage of live tree stems (>1m) in specified size class that are *Acer saccharum*. N=29 10m x 10m quadrats in 7 forest patches. Soils with free calcium carbonate in upper 15 cm of soil profile excluded from analysis (see text). Quadrats with apparent human disturbance excluded from analysis. Forest cover = cover type 2 (sugar maple + red or white oak) and cover type 3 (sugar maple, no red or white oak, no wet-mesic, wet, species).



Figure 2.8. Sugar maple abundance versus available calcium in upper 15 cm of soil profile on Brunisolic soils overlying calcareous till. Response variable is the percentage of live tree stems (>1m) in specified size class that are *Acer saccharum*. N=17 quadrats in 3 forest patches. Soils with free calcium carbonate in upper 15 cm of soil profile included in analysis. Quadrats with apparent human disturbance excluded from analysis. Forest cover type = cover type 2 (sugar maple + red or white oak) and cover type 3 (sugar maple, no red or white oak, no wet-mesic or wet species).

maple stems did not survive. This interpretation is supported by the absence of "zero percent" quadrats in the 0-4 cm size class, and, the marked tendency for "zero percent" quadrats to increase with size class and to occur on low calcium soils. The justification for including them in this analysis, therefore, is that they are inherent to the hypothesis being tested.

The modest explanation of variance in the first analysis (Figure 2.7) is due in part to the inclusion of quadrats from stands on Luvisolic and Brunisolic soils. For a given concentration of calcium, the percentage of stems that were sugar maple was consistently greater on Luvisolic than Brunisolic soils. This suggests that factors other than calcium have contributed to this result. One apparent factor is antagonism in the uptake of potassium and magnesium (see Chapter 5). Other factors that may have contributed to observed differences in sugar maple abundance are examined in Chapter 4.

The response of shade tolerant and intolerant herbs to increasing sugar maple abundance and calcium availability is presented in Figure 2.9. The response was evaluated in undisturbed stands on Brunisolic and Luvisolic soils overlying calcareous till. The analysis was undertaken to test the presumption of declining light levels on calcium rich soils. In keeping with expectations, the percentage of shade tolerant herbs in 10m x 10m quadrats increased with increasing sugar maple abundance and calcium availability. In contrast, the number of shade intolerant herbs declined. The variance in response explained by increasing sugar maple abundance, and by increasing calcium availability, was similar.

The calcium affinity of plants that flower prior to, or after, canopy closure is presented in Figure 2.10. As in the preceding case, the analysis was restricted to undisturbed forest stands on Brunisolic and Luvisolic soils overlying calcareous till. Ephemeral spring herbs were typically found on more calcium-rich soils than plants with persistent shoots that flowered prior to, or after, canopy closure. Early spring flowering plants with persistent shoots occurred on more calcium rich soils than mid to late season flowering plants, but not significantly so. The apparent affinity of ephemeral spring herbs for calcium-rich soils is consistent with their distribution elsewhere in the Great Lakes region.



Figure 2.9. Herb response to increasing sugar maple abundance and available calcium on Brunisolic and Luvisolic soils overlying calcareous till. N=29 10m x 10m quadrats in 7 forest patches. Soils with free calcium carbonate in upper 15 cm of soil profile excluded from analysis (see text). Quadrats with apparent human disturbance excluded from analysis. Forest cover type = cover type 2 (sugar maple + red or white oak) and cover type 3 (sugar maple, no red or white oak, no wet mesic or wet tree species.



Figure 2.10. Mean calcium affinity and shade tolerance of understory plants in relation to flowering phenology on Brunisolic and Luvisolic soils overlying calcareous till. Legend: S-E=spring ephemeral herbs, S-P = early spring flowering plants with persistent shoots, MLS = mid to late season flowering plants with persistent shoots, H = high shade tolerance. M = moderate shade tolerance, L = low shade tolerance. N=29 10m x 10 quadrats in 7 forest patches. Soils with free calcium carbonate in upper 15 cm of soil profile excluded from analysis. Quadrats with apparent human disturbance excluded from analysis. Forest cover type = cover type 2 (sugar maple + red or white oak) and cover type 3 (sugar maple, no red or white oak, no wet-mesic ro wet tree species. Error bars: one standard error mean.

where they are typically confined to fertile, glaciated soils (Rogers 1982, Curtis 1959).

The calcium affinity of plants with persistent shoots was examined further to determine if the more shade tolerant plants of this functional group were restricted to the more calcium rich soils. In keeping with the initial analysis of shade tolerance (Figure 2.9), plants that possessed a high shade tolerance were found on soils that were more calcium rich than plants of intermediate or low shade tolerance. This pattern was present in taxa that flowered before, or after, canopy closure.

Taken together, these results suggest that plants in these forests have partitioned the calcium availability gradient in relation to the degree of shade stress to which they were exposed. An ephemeral habitat was more strongly associated with calcium-rich soils than early or mid to late season flowering.

2.3.6 Plant Attributes

The preceding analyses have shown that the distribution of species in sampled patches has been determined in part by differences in environmental conditions. The following analysis summarizes the degree to which the functional attributes of plants explain the distribution of species in the forest understory. The attributes included in the ordination analysis were life history, provenance, life form, habitat affinity, shade tolerance and moisture affinity.

The variance in species composition explained by these attributes is summarized in Table 2.16. Habitat affinity and life form, respectively, explained 10.5% and 9.7% of the dispersion of species scores in CCA, whereas, provenance and life history explained 5.4% and 2.4%, respectively. Habitat affinity and life form, collectively, explained 17.3% of the dispersion of species scores. By comparison, the comprehensive set of environmental variables examined in this study explained 25.6% of the dispersion in species scores in a related ordination (Figure 2.3). A large fraction of the variation in the composition of plants in the forest understory, therefore, can be explained by the habitat affinity and life form of species in the regional species pool.

tatistic p>l?	7.35 0.001	1.00.0	0.25 0.001	2.30 0.001	4.22 0.001
% Total Incruia F st Explained	10.5	9.7 3.	5.1 1(2.4	4
E Unconstrained	7.100	7.100	2,100	7.100	7.100
Σ Constrained Eigenvalues	0.745	0.689	0.363	0.169	1.225
Attribute	Habitat Affinity ¹	Life Form ²	Provenance ³	Life History ⁴	Habitat Affinity + Life Form ⁵

Table 2.16. Proportion of total inertia in the dispersion of species scores (CCA) explained by functional attributes of sampled plants. Overall test of ordination models by Monte Carlo permutation tests, n-1000 permutations.

Notes:

1. Forest affinity class excluded from analysis due to high collinearity with other variables.

2. Fern life form excluded from analysis due to high collinearity with other variables.

3. Native provenance class excluded from analysis due to high collinearity with other variables.

Biennial life history excluded from analysis due to high collinearity with other variables.

5. Forest affinity class and fern fife form excluded from analysis due to high collinearity with other variables.

The tolerance of life forms to shade, and to excess moisture, are summarized in Tables 2.17 and 2.18, respectively. The most shade tolerant life forms were the fern allies, ferns and trees. The vines, shrubs, and herbs were intermediate in shade tolerance, whereas, the grasses were the least shade tolerant. The shade tolerance of taxa within life forms was not uniform, however, and often ranged across the gradient. Nevertheless, the difference in shade tolerance among life forms was highly significant ($p > \chi^2 = 0.0001$, Likelihood ratio test).

The moisture tolerance of taxa also varied by life form. Shrubs, trees, herbs and grasses were more frequent on the drier soils, whereas, vines, ferns, and fern allies were more frequent on mesic and moist soils. The life forms most tolerant of wet soils were the grasses, herbs, shrubs and ferns. When trees are removed from the analysis, the difference in moisture tolerance among life forms is not significant ($p>\chi^2=0.0987$, Likelihood ratio test). Trees were highly over-represented in tolerance classs 2 (facultative upland), and absent from tolerance classes 1 (obligate upland) and 5 (obligate wetland).

The distribution of selected attributes within examined microhabitats is presented in Table 2.19. As expected, non-native species achieved their highest proportional abundance on open farm lanes and access roads, open and closed regenerating farm fields, and, in seasonally dry tree pits. With the exception of the latter habitat, these habitats were also rich in taxa with an affinity for open habitats. Species with an affinity for wet habitats were most abundant in open seasonally wet tree pits, closed and open seeps, and open riparian marshes. The widespread occurrence of species with a facultative or obligate affinity for wetland habitats was not expected. This pattern is in keeping with previous results (Table 2.14) and suggests that in these forests, at least, such taxa have a much broader moisture tolerance than their classification would indicate. The widespread but modest presence in closed habitats of taxa with an affinity for open habitats suggests that conditions for germination and establishment are more limiting than conditions for persistence.

Likelihood ratio tes	d). Shade tolera	ance classifica	tion denved Ir	on Nineriro i	und Isrand (199	(5) and Ellenby	erg (1988).	
		Shade Toleran	ce Class (% Cl	lassified Taxa,	É		Summary	
Life Form	_ .	2	3	4	5	Mean	% High ST	% Low ST
Tree	26.9	23,0	30.8	15.4	3.8	2,46	50.0	[9. <u>2</u>
Shrub	14.9	17.0	21.3	31,9	14.5	3.15	31.9	46,4
Vine	0	1.11	44.4	33.3	11.1	3.44	1.1.1	44,4
Геп	47,4	31.6	15.8	5.3	0	67.1	78.9	5.3
Fern Ally	40.0	40.0	0	0	20,0	2,20	80.0	20.0
Grass	0	0	17.6	l'2†	35.3	4.18	0	82.4
lterb	18.2	6.[1]	13.3	33.6	23.1	18.8	30.1	56.7
					Í			

Table 2.17. Shade tolerance of sampled flora by life form. Legend: Class 1=highest shade tolerance, Class 5=lowest shade tolerance. Ifigh ST = shade tolerance classes 1.2; Low ST = shade tolerance classes 4.5. Differences in shade class significant (χ^1 =66.44, p<0.0001, χ^2 = χ^2 , χ^2 ,

5=highest tolerance to excess moisture (obligate wetland). High MT = moisture tolerance classes 4,5; I, ow MT = moisture tolerance classes Table 2.18. Moisture tolerance of sumpled flora by life form. Legend: Class 1=lowest tolerance to excess moisture (obligate upland), Class 1,2. Differences in moisture class among life forms significant only when trees are included in the analysis ($\chi^2=57.11$, p<0.0002, Likelihood ratio test). Moisture tolerance classification derived from Oldham et al. (1995).

 Summary	Mean % Low MT % High MT	2.76 51.7 27.6	5 2.49 56.6 32.1	2.78 38.3 11.1	7 2.87 39,1 30,4	2.88 25.0 25.0	46.9 34.4	3 2.67 49,8 32.2
 SURCE DAUS	4 5	27.6 0	24.5 7.5	11.1 0	21.7 8.7	25.0 0	25.0 9.4	18.9 13.
 DCC C 1388 (76 C 13)	f	20.7	11.3	55.5	30.4	50.0	18.8	18.0
loisture Folera	ć	51.72	22.6	33.3	26.1	12.5	28.1	21.5
Z	1	0	34.0	0	13.0	12.5	18.8	28.3
	Life Form	Tree	Shrub	Vine	Fern	Fern Ally	Grass	Herb

primarily in open habitats but has capacity to invade closed habitats, O=taxa found only in open habitats; Moisture Affinity (N=387 species; all life forms): OFU=obligate or facultative upland taxa, it-lacultative taxa (found in both upland and wetland habitats), F=taxa found only in forested habitats, FO=taxa found primarily in forested habitats but also found in open habitats, OF=taxa found Legend: Provenance (N=407 species; all hite forms): N=native, A⁺⁻alien; Habitat Affinity (N=358 species; trees excluded from analysis): Table 2.19. Percentage of classified taxa (all life forms) in surveyed microhabitats by provenance, habitat affinity and moisture affinity. OFW=obligate or facultative wetland taxa. %SA = percent study area (19,200m2).

Microhabitat							Plant Att	ribute			
	à	7	Prover	amce		Habitat	Affinity		Μc	oisture A	Minity
Чапаріс	% SA	r Taxa	N 11=349	A 11=58	ן: מייז	1:0 11-02	01 [:] n=126	0 n=64	OPU n=190	94=ti S	0FW n≏121
seasonally dry forest floors/rises-ec	54,3	265	90.9	8.7	28.7	26.0	24.2	7.5	55.5	19.2	21.9
seasonally dry forest floors/rises-oc	6.8	205	86.8	13.2	26.3	24.4	24.9	11.2	57.6	21.4	18.0
seasonally moist forest floors/rises-co	5.4	261	88.8	10.2	17.3	25.9	35.0	7.6	38.6	6.91	40.6
seasonally moist forest floors/rises-or	2.2	153	84.3	13.7	18.3	26.1	34.0	8.5	35.9	19.6	42.5
seasonally moist forest depressions-ce	4.I	137	89.8	9.5	21.1	32.1	27.7	j.¢	36.5	17.5	43.1
seasonally moist forest depressions-oc	0.2	ų	85.7	14.3	21.4	35.7	21.4	9.5	26.2	611	6'19
seasonally wet forest floors/rises-ce	0.6	83	83.1	14.5	6.0	24.1	51.X	7.2	19,3	20.5	57.8
seasonally wet forest depressions-ee	4.0	\$1 I	1106	8.7	22.6	29,6	£-1£	4.1	34.5	13.9	49.6
seasonally wet forest depressions-oc	1.0	88	86.8	13.2	10.5	42,1	28.9	5.3	26.3	10.5	63.2
seep-ce	0.0	33	90.9	6.I	9.1	39,4	27.3	6.]	0	- : - : - :	78.8
o-daas	0.3	52	96.0	4.0	20.0	44,0	28.0	Ð	8.0	24.0	68.0

Microhabitat							Plaint Att	ribute			
			Prove	nance		Habitat	Affinity		Mc	visture A	funity
Variable	× YS	# Taxa	N 1349	A n=58	l≓ n−76	<u>1</u> Ω.1	0F) n=126	0 1=6-1	OFU 1=190	J [.] n≕76	0FW n=121
seasonally dry gap-cc	2.7	115	90.4	8.7	28.7	25.2	28.7	2.6	63.4	19.I	14.8
seasonally dry gap-oc	6,8	205	86.8	13.2	26.3	24.4	54.9	11.2	57.6	21.4	18.0
sensonally moist gap-cc	£.0	68	86.8	11.8	25.0	30.9	29.4	0	38.2	14.7	45.6
seasonally moist gap-oc	3.0	153	84.3	13.7	18.3	26.1	34.0	8.5	35.9	19,6	42.5
scasonally wet gap-oc	1,0	82	87.8	0.11	12.2	30.5	39.0	9, R	26.8	13.4	58.5
mound-cc	3.6	165	5.59	6.7	30.9	26.7	20.6	6.1	64.2	14.5	18.8
mound-oc	tı.4	76	84.2	15.8	31.6	28.9	19.7	9.2	68.4	18.4	11.8
seasonally dry pit -ec	0.4	81	95.1	4.9	34.6	28.4	18.5	0	71,6	17.2	1.1
sensonally dry pit-oc	0.1	26	80.8	19.2	26.0	30.8	26.9	3.8	61.5	15,4	23.1
seasonally moist pit-ce	0.3	. 5	100.0	+	40,0	40.0	20.0	0	-10.0	0	60.0
seasonally wet pit-cu	0.2	8	94.4	5.6	38.9	22.2	27.8	ŋ	38.9	22.2	38.9
seasonally wet pit-oc	0.1	_	100.0	0	¢	0	100.0	ŋ	0	a	100.0
pg-cc	0.5	102	92.2	7.8	22.5	31.4	26.5	2.9	39.2	17.6	41.2
log-ac	0.2	64	9.06	8.7	21.9	26.6	1,0 <u>5</u>	3.1	34.4	15.6	45.3

Table 4... Percentage of classified taxa (all life forms) in surveyed microhabitats by provenance, habitat affinity and moisture affinity.

ŝ

Microhabitat							Plant Ati	rrbute			
	3	-	Prove	nance		Habitat	Affinity		We	oisture A	v Dinity
Variable	s^ S	Taxa	N n-349	∧ 0−58	l; л–76	1:0	OF 11-126	0-64	OF() n-190	1: 1-76	OFW n=121
stump-cc	0.2	51	90,2	9.8	9.61	35.3	971	2.0	47.1	9.91	31.4
stump-oc	0.1	34	88.2	11.8	26.5	32.4	32.4	2.9	44.1	20.6	35.3
ruised root base-ee	1.3	138	91.3	7,2	24.6	27.5	25.4	4.3	42,8	20.3	eree
raised root base-oc	0.6	96	86.5	12.5	17.7	28.1	31.3	8.3	36.5	22.9	36.5
stone-cc	0.1	20	95.0	5.0	10.0	50.0	25.0	10.0	55.0	25.0	20.0
lane/road-cu	2.3	80	93.3	6.7	25.8	30.3	25.8].]	52.8	22.5	23.6
lane/road-oc	1.3	[8]	76.8	21.5	13.3	1.12	1.25	18.2	6'0) [,]	£.91	34,8
ditch-cc	0.3	68	88.2	8.11	C.01	22.1	42.6	11.8	6'LE	14.7	55.9
ditch-oc	0.1	33	90.I	6,0	3.0	30.0	39.4	18.2	£.7 <u>°</u>	9.1	63.6
regenerating field-cc	1.3	125	80.8	1 8.4	4.8	16.8	45.6	16.8	44.8	16.8	36.1)
regenerating field-oc	() ()	147	77.6	21.8	- I. I	15.6	44.2	23.1	44.2	16.3	38.1
riparian meadow-oc	0.7	75	85.3	13.3	13.3	20,3	38.7	13.3	22.7	14.7	60.0
riparian marsh-oc	0,4	٩ŀ	87.0	10,9	10.9	26.1	45.7	15.2	15.2	15.2	67.4
riparian thicket-oc	0.1	37	86.5	10.8	10.8	37.8	32.4	8.1	16.2	21.6	59.5

. Percentage of classified taxa (alt life forms) in surveyed microhabitats by provenance, habitat affinity and moisture affinity. Table 4,

÷

2.4. Discussion

i) Environmental Heterogeneity

On a regional spatial scale, environmental heterogeneity was an important contributor to the number of recorded taxa. Approximately two-thirds (64.2%) of the sampled flora was recorded on closed, seasonally dry, forest floors, *sensu stricto*. Differences in edaphic conditions, and the presence of disturbance features, therefore provided additional habitat for approximately one-third of the species recorded in this study.

In this study, the contribution of environmental heterogeneity to species richness was strongly scale dependent. Whereas 21 microhabitats contributed to a significant difference in species richness at the quadrat scale ($10m \ge 10m$), only 9 microhabitats did so at the patch scale. In keeping with this pattern, 38 of 39 microhabitats contained at least one unique species when evaluated at the quadrat scale, whereas, 33 and 13 microhabitats did so when evaluated at the patch scale and landscape scale, respectively (see Tables 4.4, 4.5 and 4.6, Chapter 4).

One apparent reason for this pattern is that heterogeneity provides alternative habitat for species of broad environmental tolerance. In these forests, most species (79.2%) were found to occur in more than one habitat category (*sensu* Figure 2.6), and several species (33.2%) were present in every category. Comparatively few species (21.8%) were confined to one habitat type. Of these, 32 species were confined to features created by human disturbance, 23 species were confined to moist or wet conditions. 19 species were confined to closed dry forest floors, and 16 species were confined to features created by natural disturbance. Of the species that did not occur on closed, dry, forest floors, 77 (52.0%) occurred in at least one other habitat category and 41 (27.7%) occurred in every other category. Taken together, these results suggest that most species were able to germinate or persist in a variety of settings and that only rarely were they confined to one type of habitat on the forest floor. Which habitats were occupied by which species depended on the scale of the analysis, the local configuration of habitats, and the environmental tolerance of the species involved. In general, the smaller the spatial scale, the greater the contribution of heterogeneity to species diversity.
In this study, species were particularly tolerant of conditions created by the death or removal of a canopy tree (canopy gaps, tree pits, tip-up mounds, stumps, logs). Such features were colonized by 86.0% of the species on closed, dry, forest floors, 82.0% of the species in moist or wet habitats, and, 78.0% of the species in habitats created by human disturbance. However, the capacity of species to colonize or persist on features created by natural disturbance was not uniform. More species were recorded in canopy gaps (282) than on tip-up mounds (180), logs (120), pits (97), or stumps (64). Only 20% of the species on these features were not found elsewhere on the forest floor.

Taken together, these results provide broad support for the hypothesis that environmental heterogeneity facilitates the coexistence of species through the spatial and temporal segregation of competing species (Hutchinson 1961, Levin 1974). By virtue of their capacity to colonize at least one other type of habitat, most species in these forests have access to alternative environments where competitors may do poorly or not survive, and, where populations of their own kind may expand (Pickett 1980, Comins and Noble 1985, Chesson 1986, Bazzaz 1991).

ii) Contribution of Canopy Disturbance to Species Composition

The species composition of canopy gaps was similar to the species composition of the adjacent forest floor. In general, the species composition of dry open canopy gaps was not responsive to differences in gap size $(1-100 \text{ m}^2)$. This suggests that the composition of plant assemblages is dominated by persistent taxa and that colonization or extinction events during the gap phase rarely alter the composition of the understory flora in a significant way. Nevertheless, the marked change in species composition of quadrats adjacent to a large, recent, clear-cut suggests that larger canopy openings may stimulate the germination of buried seeds and cause significant transient change in species composition (Metzger and Schultz 1984).

These findings are consistent with the results of studies of specific features in forests of the U.S. northeast. Studies of forest gaps that examined compositional differences have reported little difference between species in gaps and the adjacent forest floor (Ehrenfield 1980, Moore and Vankat 1986, Pickett 1987, 1988a, 1988b, Mladenoff 1990, and Goldblum 1997. Gaps in these studies were

created by tree fall, gypsy moth defoliation, standing dead trees, or experimental treatments; gap size ranged from 5-214 m² and from 1-30 years of age.

Similarities have also been reported in the species composition of forest floors, logs, and tree pits. Thompson (1980), for example, reported that logs and tree pits were readily colonized by herbs from the forest floor and that approximately 90% of the species on logs, and 85% of the species in pits, were recorded within 1 metre of these features. These findings are consistent with the results of this study. All species recorded in tree pits, and 94% of species recorded on logs, were recorded on closed forest floors.

Studies of tip-up mounds and tree pits, in contrast, have reported distinctive species assemblages associated with these features (Beatty 1984, Peterson and Pickett 1990, Peterson *et al.* 1990, Peterson and Campbell 1993). In keeping with the results of this study, mounds were typically more species rich than pits, although the reverse pattern has been reported for fresh pits and mounds in a large area of catastrophic windthrow (Peterson and Pickett 1990). Studies of environmental conditions reveal that the soil in tree pits is typically wetter, more alkaline, has a thicker litter layer, soil organic matter content, and, experiences less extreme temperature fluctuations, than the soil on adjacent mounds (Dwyer and Merriam 1981, Beatty and Stone 1986, Peterson and Campbell 1993). Leaf litter depth may be a limiting factor in these habitats since differences in species composition were non-significant when leaf litter was experimentally removed from pits in deciduous forests in central New York (Beatty and Sholes 1988). Excess moisture is also expected to be limiting in pits with seasonally saturated soils (Beatty 1984). In keeping with these expectations. Iow species richness was associated with both leaf litter and seasonally saturated soils in this study.

iii) Response of Sugar Maple to Available Calcium

Sugar maple was more abundant, in relative and absolute terms, in undisturbed second-growth stands on mesic soils that were rich in calcium cations. Stems that were subject to shade stress and self thinning were more responsive to differences in available calcium than were stems in the upper canopy. When differences among stands were standardized with respect to soil parent material, soil order, and soil series, available calcium explained 70.9 % of the variance in the number of sugar maple stems in the 4-10 cm size class, and, 54.0 % to 35.0% of the variance in the proportion of stems in the 0-4 cm, 4-10 cm, and 10-30 cm size class.

These results are consistent with the sharp reduction in mortality of juvenile sugar maple trees observed on calcium rich soils in oak transition-northern hardwood forests in northwestern Connecticut (Kobe *et al.* 1995, Kobe 1996). Juvenile trees in these studies were defined as any individual > 25 cm tall that did not have foliage reaching the canopy of the stand. The upper limit varied from site to site but did not exceed 10 cm dbh (diameter breast height). In deep shade (less than 5% full sun), the probability of mortality in sugar maple stems declined from 99.8 % on acid schist/gneiss uplands to 14.8 % on base rich soils overlying calcareous bedrock. Differences in sapling mortality and growth rates successfully predicted the composition of regional forests when incorporated in a model of forest dynamics (SORTIE).

The mechanisms by which trees benefit from calcium rich soils are presently unresolved. Recent studies of cold temperate trees have found an association between foliar calcium levels and dark respiration rates in red spruce (McLaughlin *et al.* 1991, McLaughlin and Kohut 1992). In these studies, dark respiration rates declined in the presence of increasing foliar calcium and contributed to net carbon assimilation in young saplings (1.2 - 2.0 m tall). Foliar calcium levels were positively correlated with calcium levels in the soil, a finding broadly supported by fertilization studies in both hardwood and conifer trees (Dr. V. Timmer, Department of Forestry, University of Toronto, *pers. com.*). In keeping with the widespread expectation that plants adapted to low light should have lower carbon losses via dark respiration. Lusk and Reich (2000) recently confirmed that juveniles (0.4 -1.5 m tall) of shade tolerant angiosperm trees typically have lower dark respiration rates than associated less-tolerant species.

An alternative mechanism by which calcium rich soils may contribute to sapling survivorship is increased nitrogen availability mediated by a calcium-based rise in soil pH. Dancet *et al.* (1973) have shown that nitrification rates are strongly and positively correlated with increasing soil pH over

the pH range 4.7 to 6.6. On calcium rich soils, soil pH is enhanced when calcium cations are taken up by tree roots and returned to the soil surface by litter feedback dynamics (Boerner 1984, Khanna and Ulrich 1991, Wilmot *et al.* 1995). In keeping with this mechanism, leaves of sugar maple seedlings on deeply shaded sites (<5% canopy openness) had higher nitrogen levels, and higher growth rates, on soils with higher nitrification rates in northern Wisconsin(Walters and Reich 1997). Soil moisture also varied on these soils, however, and the relative contribution of nitrogen and soil moisture could not be determined.

Studies have recently been initiated in the U.S. northeast to clarify which of these calcium-based explanations best explains tree growth and mortality relations in upland settings (Dr. A. Finzi, Department of Biology, Boston University, *pers. com.*: Dr. R. K. Kobe, Department of Forestry, Michigan State University, *pers. com.*). At present, the collective findings of these studies suggests that calcium nutrition has important consequences in the dynamics and distribution of north temperate trees. Local differences in soil chemistry may therefore lead to spatial patterning in the distribution of canopy trees that has heretofore been associated with regional differences in mineral substrate (Braun 1950, Curtis 1959, Pastor *et al.* 1984, Host and Pregitzer 1992, Reich *et al.* 1997, van Breemen *et al.* 1997).

iv) Response of Herbs to Understory Shade

The response of shade tolerant and intolerant herbs to increasing sugar maple abundance (Figure 2.9) is consistent with declining light at the forest floor. This was not unexpected. After beech and hemlock, sugar maple casts the deepest shade in forests in the Great Lakes region (Pacala *et al.* 1996), and, has the lowest percent transmission of photosynthetically active radiation (Canham *et al.* 1994).

Curtis (1959) has argued that the forest floor is a demanding environment that requires specialized traits for success and that it is the limited set of species that possess those traits that has led to the striking uniformity in species composition in the mesic hardwood forests of eastern North America. In keeping with this hypothesis, 92.5% of the species that were prevalent in the understory of forests

in the Maple-Basswood forest region in Wisconsin were present in maple dominated forests in this study (Appendix 14).

Early spring flowering is one of several plant traits that has been associated with deep shade in the forest understory. In herbs with low shade tolerance, early flowering is associated with an ephemeral (Sparling 1967), or winter annual (Rogers 1982). life history. Each facilitates net carbon gain by restricting the growth phase to periods when the canopy is leaf free. More commonly, however, the early flowering habit is associated with varying degrees of shade tolerance that enables shoots and leaves to persist until mid to late summer (Sparling 1967, Rogers 1982). The latter combination of characters was more common in the Peterborough area where only six of sixty-two early flowering species were spring ephemerals (*Allium tricnecum, Caulophyllum thalictroides, Claytonia caroliniana, Dicentra canadensis, Dicentra cucullaria, Erythronium americanum*); only one species (*Galium aparine*) was a known winter annual.

Related plant traits that may facilitate survival in deeply shaded habitats include winter-green leaves (Bierzychudek 1982) and the initiation of shoot growth (Taylor and Pearcy 1976) or flower initials (Bierzychudek 1982) in early autumn. The former trait greatly extends the period of carbon gain in species such as *Carex plantaginea*. *Hepatica acuailoba*. *Maianthemum canadense*, *Tiarella cordifolia*, *Trientalis borealis*, *Viola blanda*, and *Viola rostrata*, whereas the latter traits facilitate early spring growth and flowering in species such as *Allium tricoccum*, *Trillium grandiflorum*. *Arisaema triphyllum*, and *Geranium maculatum*.

The capacity of plants to tolerate deep shade has been attributed to a suite of traits that facilitate the capture and processing of light energy at the lowest net cost. Morphological characters associated with shade plants include: thin leaves with a large surface area (Grime 1965); a higher proportion of chlorophyll *b* relative to chlorophyll *a* (Boardman 1977); a chloroplast with large grana stacks oriented in more than one plane (Boardman 1977); a higher proportion of leaf nitrogen allocated to chlorophyll than to carboxylating enzymes and other proteins (Seeman *et al.* 1987, Niinemets 1997. Lusk and Reich 2000); a rapid stomatal response to changes in light intensity (Hicks and Cabot

1985); and leaves deployed in horizontal, non-overlapping layers (Grime 1965, Horn 1971). These traits facilitate the capture of energy in low light environments while minimizing the energetic cost to construct and maintain plant tissue. The latter is perceived to be especially important since it results in a lower leaf dark respiration rate and lowers the compensation point for net carbon gain (Grime 1965, Loach 1967, Lambers *et al.* 1983, Lusk and Reich 2000). These characters were not scored directly in this study owing to the lack of a suitable data set.

v) Response of Herbs to Available Calcium

Data from this study suggests that many plants of the forest understory may be responsive to differences in available calcium. Ephemeral spring herbs, for example, typically occurred on more calcium rich soils (mean concentration = 6.9 cmol/kg) than plants with persistent shoots that flowered prior to, or after, canopy closure. Early spring flowering plants with persistent shoots typically occurred on more calcium rich soils than plants which flowered mid to late season and completed most or all of their life cycle under a closed canopy (mean concentration = 5.7 and 5.4 cmol/kg, respectively), but not significantly so. The shade tolerant members of the latter functional groups, however, occurred on more calcium rich soils than species with moderate and low shade tolerance. This pattern suggests that plants in these forests have partitioned the calcium availability gradient in relation to the degree of shade stress to which they were exposed.

The mechanism(s) by which plants benefit from calcium rich soils have not been resolved. However, the greater availability of nitrogen arising from a calcium-mediated rise in soil pH may be particularly important for ephemeral spring herbs, since they typically complete their life cycle before the canopy closes (*Allium tricoccum*, the apparent exception, flowers mid to late summer). These species typically have a high light compensation point and a high saturation light intensity (Sparling 1967, Taylor and Pearcy 1976) and may therefore have a greater physiological requirement for nitrogen than more shade tolerant species. The reported affinity of ephemeral spring herbs for base rich, and particularly calcium rich, soils (Curtis 1959, Rogers 1982) may therefore be due in part to the greater availability of nitrogen on these soils. For shade tolerant herbs, however, the principal mechanism may be a calcium-mediated reduction in dark respiration rate. The morphology and physiology of these species typically emphasizes the conservation of reserves rather than photosynthetic performance (Went 1957, Grime 1965, Loach 1967) and thus may benefit more from a reduction in dark respiration rate than from a greater availability in nitrogen. Efficient use of high irradiance requires a high nitrogen investment in carboxylating enzymes and proteins responsible for photosynthetic electron transport (Niinemets 1997). Shade tolerant species, however, typically allocate proportionally more leaf nitrogen to chlorophyll than to carboxylation capacity (Seeman *et al.* 1987). This investment pattern is thought to be the primary reason why shade tolerant species have a lower respiration rate per unit of leaf N (Lusk and Reich 2000) and an intrinsically low photosynthetic plasticity (Niinemets 1997). In keeping with the latter finding, experimental transfers of plants between high and low light environments have shown that dark respiration rates can change much more rapidly than photosynthetic capacity (Azcon-Bieto and Osmond 1983, Sims and Pearcy 1991). Taken together, these findings suggest that shade tolerant plants on calcareous soils may benefit more from a reduction in dark respiration rate than from a greater availability in nitrogen.

vi) Contribution of Patch, Edaphic and Matrix Variables to Species Composition

Patch variables explained more variance in the distribution of species in the forest understory than did edaphic or matrix variables (36.1%, 28.1% and 10.2%, respectively). Patch variables in this analysis were measures of stand structure (forest cover type, stem diameter class, # tree species), stand disturbance (% canopy closure, regenerating fields, lanes, canopy gaps), and overall plant response to local site conditions and biotic interactions (patch membership). Edaphic variables were measures of inherent moisture retention capacity and soil fertility (soil parent material, soil order, soil moisture class, and percent soil organic matter). Matrix variables were measures of the degree of forest fragmentation in the local landscape (patch area and patch isolation).

For the purpose of this analysis, patch variables were viewed as indicators of processes operating on ecological time scales and at small spatial scales, whereas, edaphic and matrix variables were viewed as indicators of processes operating on long time scales and at larger spatial scales. Within this framework, the results of this analysis suggest that both local and regional processes have influenced the composition of sampled plant assemblages. In particular, the strong contribution of edaphic variables provides support for the view that the diversity of species in a given setting cannot be explained solely by processes operating on short time scales at the local spatial scale (Ricklefs 1987, Ricklefs and Schluter 1993).

The principal contribution of this analysis, however, is that it reveals the inherent difficulty in selecting appropriate indicators for such a test and in characterizing processes at the appropriate spatial and temporal scale.

In its present form, this analysis clearly overstates the regional contribution of edaphic processes by failing to acknowledge that ecological consequences of regional differences in soil moisture and soil fertility are expressed at the scale of the germinating seed and the competing plant assemblage. The contributions of recruitment limitation and success to species diversity are thus regional, in the sense that they are a consequence of glaciation and pedogenesis, and local, in the sense that operate at the scale of the forest patch and influence the strength of competitive interactions. The former reaffirms the importance of viewing local processes in broader context whereas the latter affirms the primacy of processes operating at the local scale.

In its present form, this analysis may also overstate the contribution of forest fragmentation, since large patches in this study were situated primarily on dry sandy soils, and, small patches were situated primarily on loam soils that were often seasonally moist or wet. Differences in species composition, therefore, may be due primarily to differences in edaphic conditions rather than to post-settlement colonization and extinction events. The degree to which the composition of present-day assemblages is the legacy of past migration events in a continuous forest environment is unknown. However, the modest percentage of alien taxa in vegetation samples (typically < 5%. Appendix 9), the similarity in composition of canopy openings and the adjacent forest floor (Figure 2.4), and the predominance of short-distance dispersal (see Chapter 3), suggests that the composition of the understory flora has been little modified by forest fragmentation.

The larger issue raised by Ricklefs (1987) remains, however. Periodic and chance events such as hurricanes, pathogens, fire, differential rates of migration, and past land use, when discerned, have all been shown to have a marked influence on the composition of present-day assemblages. While it is clear that a comprehensive explanation of community requires an integration of physical and biological factors, over space and time, the practical challenge of how to achieve that explanation has not been met.

vii) Percent Variance in Species Composition Explained by CCA Ordination

The percentage of the variance in species composition explained by environmental variables in CCA (\sum canonical eigenvalues – \sum unconstrained eigenvalues) was a modest 25.6%. A subset of these same variables, in contrast, explained 58.3 % of the variance in species richness in multiple linear regression (see Chapter 4.0). Reports of modest explanations of variance in CCA are widespread in the ecological literature (e.g. ter Braak 1987, Borcard *et al.* 1992, Okland and Eilertsen 1994, Aude and Lawesson 1998, Ohmann and Spies 1998, ter Braak 1999). In forest studies, the percent of total variance explained (%TVE) has ranged from 10% (Ohmann and Spies 1998) to 48% (Aude and Lawesson 1998) despite comprehensive environmental data.

In my study, four properties of the data may contribute to the modest percent of variance explained by environmental variables. First, the median species frequency was 7 quadrats. This property will generate a higher turnover in species composition for a given richness and thus reduce the percent of variance explained in CCA. This phenomenon has been characterized as a rare species effect by Ohmann and Spies (1998) but it applies more generally since most species are "rare" in most data sets.

Second, the median number of microhabitats per 10m x 10m quadrat was 3 (range 1-10). In principle, environmental heterogeneity should lead to differences in composition for a given value of species richness, since species vary in the type and range of habitats occupied. This in turn should lead to greater variability in species composition and reduce the percent of variance explained in CCA. In keeping with this expectation, habitat diversity explained more variance in species

richness (F=11.5, r²adj.=0.052, simple linear regression) than in the dispersion of species in CCA ordination space (F=3.83, Monte Carlo permutation test, %TVE=1.97%) (follow-up analyses, not shown).

Third, the spatial resolution of the environmental data is large relative to the conditions experienced by a given seed, seedling, or maturing plant. This should lead to a lower explained variance in species composition since recorded data will fail to account for variability in conditions that lead to differential germination, growth, and persistence. Soil conditions that contribute to differential germination success and to the availability of essential nutrients, for example, typically occur on the scale of centimetres (Harper *et al.* 1965, Harper 1977, Pacata and Silander 1990, Kosola *et al.* 1999) whereas soil properties in this study were measured on the scale of metres or taken from published regional soil surveys. The maximum correlation between vegetation and environment should be achieved when differences in plant-environment and plant-plant interactions reflect the average variation in the physical and biotic environment (Reed *et al.* 1993). The scale at which this occurs has been shown to be attribute dependent (Palmer 1990, Reed *et al.* 1993, Ohmann and Spies 1998).

In keeping with this pattern, attributes of the environment that varied at larger spatial scales (such as soil parent material, soil order, cover type) typically explained more variance in composition and richness than attributes that varied at smaller spatial scales (such as stand structure variables, disturbance features). Differences in soil parent material, for example, explained 4.5% of the variance in species scores in CCA, and 20.9% of the variance in species richness, whereas the presence/absence of disturbance features explained only 1.8% of the variance in species scores in CCA and 5.5% of the variance in species richness (Table 2.13 and follow-up analyses, not shown).

Fourth, the temporal resolution of the environmental data often fails to account for past events that are known contributors to differences in composition and richness, such as post-glacial plant migration (Davis 1981a), major storm events (Henry and Swan 1974), extreme values for temperature and moisture (Okland 1996, Ohmann and Spies 1998), and land use (Pettit *et al.* 1995, Motzin *et al.* 1996). While aspects of these events may be captured by analyses of pollen profiles

(Davis 1983), stand structure (Lorimer 1985), soil properties (Bormann *et al.* 1995), soil profiles (Beke and McKeague 1984), fire scars (Lorimer 1985), etc., the contribution of stand history is likely to remain largely undefined.

Additional sources of unexplained variation reside in certain technical properties of CCA. The goal of CCA is to select the linear combination of environmental variables that maximizes the dispersion of species scores in ordination space (ter Braak 1987). Unlike principal components analysis (PCA), CCA is not a maximal variance extraction technique and thus the fraction of inertia explained by CCA can be quite modest (ter Braak 1999). In strict terms, the "percentage variance" reported in CANOCO is not a true variance. Rather it is the portion of the total inertia that is explained by the extracted axes (ter Braak 1999). This proportion is an analogue to r² in linear regression but is not a true coefficient of determination. One consequence of this is that when the number of variables is less than the number of extracted axes, the fraction of inertia explained by the submitted variables will always be less than 100%. Moreover, as in CA, site scores on the second axis may form a quadratic relation with those on the first axis. This fault, known as the "arch effect", is typically less of a problem in CCA, but when present, will contribute to the inertia of the ordination and reduce (or enhance!) the amount of variance that can be explained (ter Braak 1999). As a consequence, ter Braak (1999) has cautioned users not to interpret a low (high) "variance explained" as a poor (good) fit of the ordination model.

The modest percentage variance in species composition explained by traditional data sets in CCA, therefore, may not be unreasonable. While the contribution of certain technical aspects of the method may not be readily quantified, there are several properties of traditional data sets that may account for the modest explanation of variance in species composition in CCA, and, for a modest explanation of variance in composition compared to species richness.

viii) Pattern of Functional Attributes in Relation to Environmental Gradients

The functional attributes of plants examined in this study individually explained more variance in the dispersion of species scores in CCA than did most attributes of the environment (Tables 2.16 and

2.13). Habitat affinity and life form individually explained more variance in the distribution of species than did provenance or life history, and, collectively explained 67.4% of the variance explained by environmental variables (Figure 2.2). The capacity of life form to explain variance in the distribution of species is the more interesting case, in this study, since habitat affinity was defined in relation to canopy closure.

The capacity of life form to explain variance in the distribution of sampled species is due in part to the difference among life forms in shade tolerance (Table 2.17). Grasses, herbs, shrubs and vines in these forests were relatively shade intolerant and were significantly more frequent in quadrats with low canopy closure (high light) (p<0.0003, Wilcoxon rank sum tests, not shown). In keeping with this outcome, the number of species of the most shade tolerant life forms (fern allies, ferns, trees) did not vary with canopy closure.

The moisture tolerance of species varied within, but typically not among, life forms (Table 2.18). With the exception of the fern allies, the species associated with each life form were more tolerant of upland conditions (moisture classes 1.2) than of lowland conditions (moisture classes 4.5). In apparent contrast to this pattern, the ferns, grasses and herbs were proportionally more species rich in quadrats with gleyed soil horizons and seasonally wet depressions. The contradiction is more apparent than real, however, since a majority of ferns, grasses and herbs were tolerant of moist to wet conditions when the taxa of intermediate tolerance (moisture class 3) are taken into account. The moisture regime of quadrats with seasonally wet depressions typically ranged from mesic to wet and thus provided micro-sites for taxa across the moisture continuum. In keeping with this perspective, the mean moisture tolerance rating of species in 10m x 10m quadrats did not vary among moisture classes ($p > \chi^2 = 0.98$, Wilcoxon rank sum test, not shown).

Life forms in these forests, therefore, appear to be more diversified with respect to moisture than to light. This pattern is not unexpected given that differences in moisture availability are less transient than differences in the availability of light. In keeping with this line of reasoning, the difference in shade tolerance among life forms was not significant when only "F" and "F+O" species were

considered. Species with an affinity for open habitats were also diversified with respect to moisture $(p>\chi^2=0.28)$. Likelihood ratio test, not shown) and were as likely to be found in natural canopy openings associated with large seeps as in upland habitats beneath canopy gaps created by human disturbance or by wind-throw. As in the previous case, the difference in shade tolerance among life forms is not significant when only "O" and "O+F" species are considered.

A sharp contrast in shade tolerance does arise, however, in life forms that are predominately shade tolerant, such as the ferns, fern allies, and trees, or, that are predominately shade intolerant, such as the vines and grasses ($p>\chi^2=0.0001$, Likelihood ratio test, not shown). These life forms may be found in both open and closed habitats, but not with equal frequency. In contrast, the herbs and shrubs were more diversified across the light gradient and species were equally as likely to be found in closed as in open habitats. Differences in shade tolerance among these life forms are not significant ($p>\chi^2=0.48$, Likelihood ratio test, not shown). These patterns persist when non-native species (which are predominately shade intoierant) are removed from the analysis.

Life form, therefore, is a significant contributor to differences in the composition and distribution of species in these forests. Based on the patterns of shade and moisture tolerance summarized above, the distribution of most life forms should be markedly non-random. Ferns and fern allies should occur primarily on moist soils under a closed canopy: vines should occur primarily on moist soils under a closed canopy: vines should occur primarily on moist soils under a thin or broken canopy; and, grasses should occur on a variety of soils but primarily under broken or open canopy conditions. Herbs and shrubs, on the other hand, should be found throughout the forest. And, not surprisingly, this is typically what was found.

2.5 Principal Findings

 Traditional environmental factors explained 25.6% of the variation in composition of plant assemblages in the understory of twenty-four, second-growth, northern hardwood, forests near Peterborough, Ontario. The distribution of species was most strongly influenced by gradients in soil moisture, soil order, forest cover type, percent canopy closure, and soil parent material. The patterns in plant response to differences in soil moisture, soil fertility, and available light are consistent with the results of previous studies of hardwood forests in the Great Lakes region and northeast United States.

2. The distribution of species within forest patches was strongly influenced by microhabitat features created by glacial history, landscape position, the death of canopy trees, and human disturbance: moist or wet forest floors, seeps, tip-up mounds, tree pits, stumps, logs, canopy gaps, raised root mats, lanes, ditches and regenerating fields. These features created habitat opportunities for 35.8% of the recorded flora that did not occur on closed, seasonally dry, forest floors, *sensu stricto*.

These results provide support for the hypothesis that environmental heterogeneity contributes to species diversity through the provision of novel resources.

3. Environmental heterogeneity at the patch scale provided alternative habitat for species of broad environmental tolerance. When the distribution of species was examined in relation to four generalized habitat categories (closed dry forest floors; moist or wet habitats; features created by human disturbance; features created by natural disturbance), most species (79.2%) were present in more than one habitat category, and several species (33.2%) were present in every category. Comparatively few species (21.8%) were confined to only one habitat type.

In these forests, species were particularly tolerant of conditions created by the death or removal of a canopy tree (canopy gaps, tree pits, tip-up mounds, stumps, logs). Such features were colonized by 86.0% of the species on closed dry forest floors, 82.0% of the species in moist or wet habitats, and 78.0% of the species in habitats created by human disturbance. However, the capacity of species to colonize or persist on these features was not uniform. More species were recorded in canopy gaps (282) than on tip-up mounds (180), logs (120), pits (97) or stumps (64).

These results provide support for the hypothesis that heterogeneity facilitates the coexistence of species through spatial and temporal segregation of competing species. By virtue of their capacity to colonize at least one other habitat in the local environment, most species in these forests have access to alternative environments where their competitors may do poorly or not survive and where populations of their own kind may expand.

- 4. The species composition of canopy gaps was similar to the species composition of the adjacent forest floor. The species composition of dry open canopy gaps was typically not responsive to differences in gap size (1-100 m²). This suggests that the composition of plant assemblages is dominated by persistent taxa, and, that colonization or extinction events during the gap phase rarely alter the composition of understory plants in a significant way. A marked change in the species composition of quadrats adjacent to a recent clear-cut stand suggests that larger canopy openings may stimulate germination of buried seeds and cause significant transient change in the composition of understory plants.
- 5. Marked differences in the survivorship of juvenile sugar maple stems was observed on a gradient of increasing calcium availability in undisturbed forest stands on mesic soils overlying calcareous till. Stems that were subject to shade stress and self-thinning were more responsive to differences in available calcium than were stems in the upper canopy. When differences among stands were standardized with respect to soil parent material, soil order and soil series, available calcium explained 70.9% of the variance in the number of stems in the 4-10 cm dbh size class, and, 54.0% to 35.0% of the variance in the proportion of stems in the 0-4 cm, 4-10 cm, and 10-30 cm dbh size class.

The data are consistent with the sharp reduction in mortality of juvenile sugar maple stems observed on calcium rich soils in oak transition-northern hardwood forests in northwestern Connecticut (Kobe *et al.* 1995, Kobe 1996) and suggest that on mesic soils the distribution and abundance of sugar maple may be secondarily constrained by the availability, and ratio. of base cations.

6. Data from this study suggests that plants of the forest understory may also be responsive to differences in available calcium. Spring ephemeral herbs, and shade tolerant plants generally, typically occurred on more calcium-rich soils than plants of moderate or low shade tolerance. Plants with persistent shoots that flowered prior to canopy closure typically occurred on more calcium rich soils than plants with persistent shoots that flowered mid to late season, but not significantly so. Traits that extend the period of carbon gain in deeply shaded habitats, such as a winter annual life history, winter-green leaves, and, the initiation of shoot growth or flower initials in the early autumn, were strongly associated with sugar maple stands on calcium-rich soils.

The mechanisms by which plants benefit from calcium rich soils have not been resolved. The greater availability of nitrogen arising from a calcium-mediated rise in soil pH may be particularly important for ephemeral spring berbs since they typically complete their life cycle prior to canopy closure. Shade tolerant herbs, however, may benefit more from a calcium-mediated reduction in dark respiration rate since the morphology and physiology of these species typically emphasizes the conservation of resources rather than photosynthetic performance.

7. Patch variables explained more variance in the distribution of species in the forest understory than did edaphic or matrix variables (36.1%, 28.1% and 10.2%, respectively). Patch variables were viewed as indicators of processes that operate at small temporal and spatial scales, such as competitive interactions, short-distance dispersal, and heterogeneity created by the death or removal of canopy trees. Edaphic variables were viewed as indicators of processes that operate at larger temporal and spatial scales an such as glaciation and pedogenesis. Matrix variables were viewed as indicators of dispersal and migration constraint within the present-day landscape.

Within this framework, the results of this analysis suggest that both local and regional processes have influenced the composition of sampled plant assemblages. The strong

contribution of edaphic variables provides support for the view that the diversity of species in a given setting cannot be explained solely by processes operating on short time scales at the local spatial scale.

8. Habitat affinity and life form explained 67.6% of the variance in species composition explained by environmental variables. The explanatory power of these variables was largely due to their capacity to account for differences in the availability of light. Ferns and fern affies occurred primarily on moist soils under a closed canopy; vines occurred primarily on moist soils under a thin or broken canopy; grasses occurred on a variety of soils but primarily under broken or open canopy conditions. Herbs and shrubs were found throughout the forest understory.

3.0 DISPERSAL PATTERNS

3.1 Introduction

Dispersal is the stage in a plant's life cycle in which the spore, seed, fruit, or vegetative propagule, detaches from the mother plant, travels through space, and comes to rest on a nearby or distant surface (Ridley1930, Berg1983, Little and Jones1980). Dispersal may be achieved in one or more episodes and involve one or more agents (Matlack 1989, Beattie and Lyons 1975, Venable and Levin 1985, Greene and Johnson 1997). The sequence of events may differ somewhat for vegetative propagules but must result in a detached, independently-rooted, ramet to be considered dispersal in this study. In spatially structured populations, dispersal is expected to serve three population-dynamic functions; i) reduction in the risk of low demographic success; ii) escape from the negative consequences of crowding; and, iii) escape from the negative consequences of interactions with siblings (Venable and Brown 1988, 1993).

Pattern in dispersal events is predicted to have profound consequences for populations and communities since dispersal governs variance in the size and composition of the seed rain (Clark and Yi 1995), affects the probability that a diaspore will land in a site suitable for germination (Harper1977, Sorensen 1978, Venable and Levin1985), determines the initial conditions that seeds and seedlings confront (Schupp and Fuentes 1995), affects the initial spatial array of individuals in a population (Thiede and Augspurger1996), determines who interacts with whom and with what intensity (Shmida and Ellner 1984, Pacala 1986, Silander and Pacala 1990, Rees 1996, Rees *et al.* 1996), influences local extinction rates by affecting the probability that declining or extirpated populations are rescued (Brown and Kodric-Brown 1977, Holt 1993), influences the rate at which plants colonize new habitat (Halpern *et al.* 1990, Matlack 1994, Kotanen 1997, Brunet and von Oheimb 1998) and the sequences in which they arrive (Drake 1991, Fastie 1995), and, influences the level of gene flow within and between populations and thus the degree to which neighboring plants are related (Williams and Guries 1994) and genetic variation is structured spatially (Levin 1981, Hamrick *et al.* 1993).

The diaspores of many species possess morphological features or chemical properties that facilitate dispersal by a particular agent or mechanism (Ridley 1930, Dansereau and Lems 1957, van der Pijl-1982, Thompson et al. 1997). Features considered important for dispersal in temperate forest habitats, and their associated dispersal syndrome, include the following: a fleshy pulp or aril that is caten by frugivores and the seeds are regurgitated or excreted unharmed (endozoochory); hooks, barbs, awns or scabrous hairs that cling to fur or feathers (epizoochory); an oil body (elaiosome) that is attractive to ants and the seed or fruit is moved before the body is eaten and the seed discarded unharmed (synzoochory s.l., myrmecochory s.s.); wings, plumes, hairs, flattened or inflated structures that increase air resistance and slow the rate of aerial descent of seeds and fruits (anemochory); a splash-cup or ballistic mechanism that forcefully expels the seed from the plant (ballochory); a hard seed coat or antifungal agent that facilitates the persistence of seeds in buried seed pools (chronochory); a detachable vegetative propagule or disintegrating rhizome, stolon or runner that permits the establishment of an independent ramet (various syndromes, autochory s.l.) (Ridley 1930, van der Pijl 1982, Sorensen 1986, Warr et al. 1992, Thompson et al. 1997. The terminology for dispersal syndromes follows van der Pijl (1982) and Little and Jones (1980); suffix "-chory" from chorein = to wander.

The spatial arrangement of established plants is governed initially by the distance diaspores travel in space or time, the nature of the dispersing agent, and the character of the receiving environment. The distance diaspores travel in space is influenced by a variety of factors including propagule morphology (Baker and O'Dowd 1982, Sacchi 1987, Willson 1993), the agent of dispersal (Hughes *et al.* 1994), plant height (Sheldon and Burrows 1973), inflorescence position (Trapp 1988), proximity and character of the surrounding vegetation (Thiede and Augspurger 1996), weather variables (Campbell 1983), seed processing method of animal vectors (Levey 1986), retention time in the digestive tract of animal vectors (Proctor 1968), retention time on the exterior of animal vectors (Bullock and Primack 1977), ferrying time by insects (Berg 1975, Jules 1996), and related factors. In general, diaspores with facilitating features travel farther than those without special devices (Willson 1993), smaller and lighter diaspores travel farther than larger and heavier diaspores of similar mode (Hoppes 1988, Matlack 1989), diaspores dispersed by animals travel farther than those dispersed on the wind or by ballistic mechanisms (Appendix 11), diaspores dispersed from taller plants and elevated positions travel farther than those dispersed from shorter plants and lower positions (Sheldon and Burrows 1973). For most plants, the proportion of diaspores dispersed declines sharply with distance creating seed dispersal curves with long tails (Portnoy and Willson 1993). The small fraction of diaspores that disperse beyond their immediate surroundings is considered critical for species of transient habitats (Harper *et al.* 1970, Meyer and Schmid 1999), and for metapopulations (Husband and Barrett 1996, Valverde and Silvertown 1997), and may be essential for plant fitness in general (Stebbins 1971, 1974).

The manner in which distance contributes to the population-dynamic functions of dispersal is presently being re-evaluated as studies reveal that many species lack features that may facilitate dispersal (Willson 1993, Hughes *et al.* 1994), that most diaspores travel only metres or tens of metres from the parent plant (Levin 1981, Willson 1993, Hughes *et al.* 1994, Cain *et al.* 1998, Appendix 11), that most dispersal events in spatially variable environments place diaspores in less favorable environments (Cohen and Levin 1991, Holt and McPeek 1996), and, that short distance dispersal may promote coexistence among species (Atkinson and Shorrocks 1981, Shmida and Ellner 1984, Pacala 1986, Silander and Pacala 1990, Hurtt and Pacala 1995, Rees *et al.* 1996). These and related studies suggest that for many plants the most important dispersal outcome is achieved at a distance of 1-2 canopy diameters from parent plant and that any further advantage to be gained at greater distances is not greatly affected by the dispersal mode adopted (Hughes *et al.* 1994, Portnoy and Willson 1993).

Nevertheless, that some propagules must achieve longer dispersal distances for overall plant fitness appears certain in view of the evolution of polychorous (multimodal) species and their investment in structures such as elaiosomes, and dimorphic seeds, which greatly extend the dispersal reach of diaspores (Berg 1969, Westoby 1981, Venable and Levins 1985); the continuous variation in morphology of dispersal structures of individual plants that extends the dispersal reach of unimodal species and monomorphic seeds (Sacchi 1987, Michaels *et al.* 1988); and, the skewed frequency distributions of dispersal distances achieved by all plants, no matter what the mode of dispersal

(Portnoy and Willson 1993). Distance matters, therefore, though not perhaps to the degree once thought (Ridley 1930).

The environment in which a diaspore lands is often a matter of chance alone. However, for some modes, and for propagules that travel only a short distance, the probability of landing in a particular type of environment is better than chance. The directed quality of such dispersal creates the basis for pattern in the spatial arrangement of plants. Direct analysis of the seed rain provides the strongest evidence for directed dispersal (*sensu* Howe and Smallwood 1982, Hanzawa *et al.* 1988) since the spatial arrangement of established plants can be influenced by differential germination success, competitive interactions, and other factors.

For plants dispersed by animal ingestion, the character of the seed rain, and where it lands, may be influenced by a variety of mechanisms, including the behavior and habitat preference(s) of the dispersing agent, the size and composition of the fruit consumed, the manner in which the fruit is eaten and seeds are processed, the degree to which seeds are voided alone or in the company of others, and the type of habitat in which post-foraging behavior occurs (Proctor 1968, Thompson and Willson 1978, McDonnell and Stiles 1983, Johnson et al. 1985, Levy 1986, Piper 1986, Stiles and White 1986, Hoppes 1988, Malmborg and Willson 1988, White and Stiles 1990. Stiles 1992. Schupp 1993. Kollmann 1995). Spatial pattern arising from these mechanisms has been detected in the seed-fall of bird dispersed plants in an Illinois woodland (Hoppes 1988, Malmborg and Willson 1988) where 53% of the seeds of fleshy fruits from artificial displays (seven species, representing four life forms) fell in tree-fall gaps. 33% fell within 9 metres of the gap edge, and 14% fell in undisturbed forest. Pattern in the receiving environment has also been reported for species of oak, beech, and pine arising from the seed caching behavior of blue jays (Darley-Hill and Johnson 1981, Johnson and Adkisson 1985) and Clark's nuteracker (Vander Wall and Balda 1977, Tomback and Linhart 1990). Pattern in the consumption of seeds by small mammals has rarely been detected owing to complex interactions among contributing factors (Willson and Whelan 1990. Whelan etal. 1991, Willson 1993b, Boman and Caspar 1995), although a tendency for scat to be deposited at the base of trees and on logs has often been observed.

For plants dispersed on the exterior of animals, the character of the seed rain and the receiving environment are expected to be less predictable, owing to differences in the "set" of individual barbs, the rate of uptake and retention of diaspores in different vegetation types, the rate of detection and removal of diaspores by animal vectors, and, the adhesive properties of diaspores and animal hides (Agnew and Flux 1970, Bullock and Primack 1977, Sorensen 1986, Williams and Guries 1994, Kiviniemi 1996). Unlike dispersal by ingestion, dispersal by adhesion is often a random process in which diaspores attach and detach as the animal moves. Depending on local circumstances and the vector involved, some diaspores may travel only a short distance, while others may travel farther than fruits dispersed by animal ingestion or the wind (Sorensen 1986, Matlack 1994, Brunet and von Oheimb 1998). Nevertheless, pattern has been detected in the spatial distribution of established plants. In an intercontinental survey of ten regional thoras, plants with adhesive fruits were more likely than other plants to occur in disturbed, meadow, lakeshore, and desert habitats, and, were significantly more common in woodlands than in other habitats examined (Sorensen 1986).

For plants dispersed by ants, the destination of the diaspore is the nest and associated disposal sites which are often located within metres to tens of metres of the parent plant (Berg 1975, Handel 1976, Culver and Beattie 1978). The character of the surrounding environment in which the seedling emerges, however, has rarely been assessed. In habitats where myrmecochory is common (dry sclerophyll vegetation in Australia: meadows and forest understories in North Temperate regions Rice and Westoby 1981,1993; Sernander 1906, as cited in Rice and Westoby 1986; Beattie and Culver 1981), ant-dispersed species are commonly found in a variety of microsites. The environment in which a seedling establishes, therefore, may have more to do with the degree of heterogeneity in the local environment than with the presence or absence of particular conditions. The relative importance of benefits arising from the escape from predators (O'Dowd and Hay 1980, Heithaus 1981), the distance diaspores move (Westoby and Rice 1981), or the placement of seeds in suitable microsites (Hanzawa *et al.* 1988), is still open to debate (Rice and Westoby 1986, Andersen 1988).

For plants dispersed by wind and by ballistic mechanisms, the character of the receiving environment is expected to be highly variable. For most diaspores, however, the receiving environment will be

the immediate surroundings of the parent plant since most diaspores travel only a short distance (Appendix 11). Pattern in the spatial array of these species, therefore, is expected to arise from differential germination success, and competitive interactions, rather than from the effects of dispersal *per se*.

For plants dispersed in persistent seed pools (*sensu* Thompson *et al.* 1997), the character of the receiving environment will be governed primarily by environmental heterogeneity since the seeds of these plants are typically dispersed by the wind, ballistic mechanism, or adhesive fruits. The environment in which germination takes place, however, will typically be disturbed habitats since dormancy is favored to evolve in small-seeded plants in variable environments (Venable and Brown 1988). The seeds of these plants typically lack the metabolic reserves to establish under a closed canopy (Saverimuttu and Westoby 1996, Westoby *et al.* 1996).

From information presented thus far, one may conclude that: i) dispersal has consequences for population and community structure; ii) dispersal may be a non-limiting process within habitat patches owing to the predominance of short-distance dispersal; iii) the environment in which a diaspore lands is governed by its mode of dispersal: iv) dispersal is primarily a deterministic process for diaspores that remain close to the parent plant: v) dispersal is primarily a stochastic process for those that travel more than a few metres from the parent plant; vi) both short and long-distance dispersal contribute to plant fitness; vii) pattern in the seed rain may arise from dispersal but may prove difficult to discern in established vegetation owing to the number of mechanisms involved.

The contribution of dispersal to pattern in plant communities may be affected by interactions with other plant traits that contribute to reproductive success. Interactions among traits that represent different ways of escaping unfavorable conditions have attracted particular attention since selection pressure that favors one trait may indirectly select against another (Levin *et al.* 1984. Venable and Brown 1988. Rees 1993). Reported interactions and apparent trade-offs of interest to this study include: seed size and habitat (Salisbury 1942, 1974, Baker 1972, Mazer 1989, 1990); seed size and seed number (Harper 1977); seed size and dispersal mode (Westoby *et al.* 1992, Hughes *et al.* 1994.

Rees 1996); seed size and dormancy (Rees 1996); dormancy and dispersal (Bulmer 1984, Rees 1996); seed size, dormancy and dispersal (Venable and Brown 1988); longevity and dormancy (Rees 1993, 1994); growth form and mode of dispersal (Harper *et al.* 1970, Hughes *et al.* 1994, Leishman *et al.* 1995); dispersal, dormancy, perenniality and iteroparity (Silvertown 1984); and, seedling success and dispersal (Morse and Schmitt 1985).

The relations between seed size and mode of dispersal have important implications for pattern since related trade-offs between seed size and germination success, and between seed size and dispersal distance, may restrict the range of babitats occupied by a particular dispersal mode. The finding by Westoby *et al.* (1992) that dispersal mode is rarely constrained by seed size is highly significant, therefore, since it implies that the habitats in which plants establish will rarely be constrained by the way in which they disperse. This greatly increases the range of solutions (to ecological problems) available to plants. Observed trends in the frequency of modes in relation to increasing seed mass (unassisted - wind - adhesion - ant - ballistic - vertebrate) (Hughes *et al.* 1994) suggest that some modes of dispersal may confer greater fitness benefits (Westoby *et al.* 1995), or be more constrained phylogenetically (Harvey and Pagel 1991, Harvey *et al.* 1995), than others for a given seed mass and habitat. These trends suggest that pattern may arise within and between habitats in the proportion of species dispersed by a given mode. Differences in dispersal spectra derived from published floras (Willson *et al.* 1990, Jurado *et al.* 1991, Westoby *et al.* 1992, Leishman *et al.* 1995), and from original studies (Dansereau and Lerns 1957, Frenkel 1970, Pojar 1974, Luftensteiner 1979, Ellner and Schmida 1981, Hoehne 1981, Westoby *et al.* 1990), are consistent with this expectation.

The relations between dispersal and germination biology have been studied rarely. That such relations occur are not in question since plants in which dispersal attributes fail to harmonize with germination requirements cannot replace themselves (consider, for example, the rapid loss of dispersal capacity among species of open inland habitats germinating on ocean beaches reported by Cody and Overton 1996). One may also conclude from first principles that the direction in which natural selection typically acts is from germination biology to dispersal attributes, since it is primarily the alleles (for dispersal) that are compatible with germination requirements that are passed

along in a colonization event. To date, the impact of germination biology on dispersal attributes has been examined in relation to three phenomena: seed dormancy, seed heteromorphism, and, polymorphism in germination requirements.

The impact of germination biology on dispersal processes is most clearly resolved in the case of spatial dispersal and seed dormancy (treated in this study as dispersal in time, after Willson er al. 1990, but arguably, and perhaps more appropriately, classified as a germination strategy, after Baskin and Baskin 1998). Modeling by several researchers reveals that the capacity for dormancy should rarely evolve in species with efficient spatial dispersal since dispersal reduces the likelihood that all seeds will be exposed to unfavorable conditions in any one year (e.g. Venable and Lawlor 1980, Cohen and Levin1987, Venable and Brown 1988). Similarly, dormancy is not expected to evolve in species with large seeds, since the attendant store of provisions significantly improves the chance that a seedling may establish under a closed canopy or other unfavorable conditions (Venable and Brown 1988, Rees 1996, Saverimutto and Westoby 1996). Dormancy may select for higher dispersal, however, when the probability is high that diaspores will find a suitable patch, and, when the amplitude of environmental variation is large (Cohen and Levin 1987, 1991). High dormancy and high dispersal, therefore, are most likely to arise in disturbed habitats in species that have small, wind-dispersed, seeds. The impact of germination biology on dispersal for species with perennial life histories has not been modeled, but is not expected to differ with respect to the trends reported here (Venable and Brown 1993).

The impact of germination biology on dispersal attributes has been examined more thoroughly in species with dimorphic (heteromorphic) seeds than in other functional groups (Harper 1965, 1977: Sorensen 1978; Flint and Palmblad 1978; Venable and Lawlor 1980; Baker and O'Dowd 1982; Olivieri and Berger 1985; Venable 1985; Venable and Levin 1985, Venable *et al.* 1995, and references therein). Seed heteromorphism is the production of seeds of different morphologies or behaviors by single individuals (Venable 1985). It is more common in annuals, plants of arid and semi-arid regions, and weeds, but it is not restricted to these plants (Venable *et al.* 1995). Heteromorphic seeds often display a polymorphism for both germination and dispersal (Harper 1965,

Silvertown 1984) thereby enabling individual plants to "hedge their bets" with respect to the timing and general location of germination events (Sorensen 1978, Seger and Brockmann 1985).

In general, the seed morph that has the more precise germination requirement, and the capacity for dormancy, remains in the immediate vicinity of the parent plant whereas the morph that can establish under a wider range of conditions, and lacks dormancy, disperses beyond the local patch (Venable and Lawlor 1980, Olivieri and Berger 1985). The former combination provides some measure of reproductive assurance (Pannell and Barrett 1998), as well as, a mechanism for reducing competition with siblings (Olivieri and Berger 1985) and non-related species. The latter combination, on the other hand, enables seeds to reach a wider range of habitats and to germinate rapidly in settings that provide suitable conditions for establishment. In the summer annual, <u>Heterosperma pinnatum</u> (Asteraceae), the relations between germination attributes and dispersal reach have been shown to be heritable and correlated with differences in habitat (Venable *et al.* 1995). In keeping with recent theory (Venable and Brown 1993), achenes that germinated rapidly and were capable of longer-distance dispersal were selected against at sites where early germination was hazardous. In those environments, flower heads contained a significantly lower proportion of rapidly germinating achenes than flowering heads in habitats where the risk of germination was less hazardous.

The degree to which monomorphic seeds possess germination polymorphisms has been examined rarely. According to Harper (1977) and Silvertown (1984), cryptic seed heteromorphism (variable seed behavior that is not accompanied by dramatic morphological differentiation) may be a widespread phenomenon affecting many plant species (Venable 1985). In a study of seedling performance and dispersal capacity in the common milkweed. <u>Asclepias syriaca</u>. Morse and Schmitt (1985) found that germination and seedling success were positively correlated with seed size and negatively correlated with dispersal reach. These results were interpreted as evidence of a conflict between seedling performance and dispersal capacity since the seeds that were most capable of reaching distant sites were the seeds least capable of establishment and growth. This conclusion may not reveal the true fitness of seeds dispersed over longer distances, however, since <u>Asclepias syriaca</u> has the capacity to form persistent seed banks (*sensu* Thompson *et al.* 1997, Burnside *et al.* 1996).

Given that approximately 50% of the seeds in this study failed to germinate, and that the seeds which failed to germinate were significantly lighter than those which did, it is possible that the smaller and more vagile diaspores of this species have the capacity for prolonged dormancy. If true, then the case for conflict between seedling performance and dispersal has not been demonstrated. However, if the smaller seeds in <u>Asclepias syriaca</u> were to possess more capacity for dormancy than larger seeds, then a germination polymorphism may be present in the monomorphic seeds of this weedy perennial.

The larger issue raised by the Morse and Schmitt study remains unresolved, however, and points to a fitness limitation that is inherent in the dispersal curve of all species and that may account for the small fraction of diaspores that leave the immediate vicinity of most parents. Although the variety of mechanisms by which germination polymorphism may arise (Silvertown 1984), and the various forms and degrees of dormancy that have evolved (Baskin and Baskin 1998), may each reduce the risk of regeneration failure in vagile diaspores, it would appear that these mechanisms have not been sufficient to modify the seed shadow of most species (Portnoy and Willson 1993). They may, however, permit a larger fraction of diaspores to disperse over longer distances and to colonize suitable habitat in patchy and disturbed environments.

From the preceding discussion, additional conclusions regarding the contribution of dispersal to pattern in established vegetation may be drawn; i) the habitat in which a species germinates should tarely be constrained by its mode of dispersal, although a given mode may be more frequent in some habitats than others; ii) reproductive success in plants requires some measure of harmonization between dispersal attributes and germination biology; iii) the fraction of seeds that disperses beyond the immediate vicinity of the parent plant may depend more on the germination biology of the species than on the mechanical constraints imposed by the mode of dispersal; iv) dispersal attributes are heritable and subject to natural selection.

Study Objectives and General Approach: The contribution of dispersal to spatial structure and species diversity in the forest understory has rarely been examined. The objectives of this chapter.

therefore, are to further this understanding by: i) identifying patterns of association between modes of dispersal and plant traits that may independently contribute to pattern in the distribution of species: and, ii) identifying patterns in the distribution of modes of dispersal within the forest understory and in relation to gradients in limiting resources.

The basic approach was to characterize the sampled flora in relation to the traits of interest and then to search for pattern in the established vegetation. The dispersal attributes of 413 vascular plants from 24 forest stands were classified in relation to the agent of dispersal and then examined for pattern in relation to plant traits (life form, life history, provenance, fruit type, taxonomic rank, environmental states (environmental gradients, microhabitats, habitat affinity, moisture affinity, shade tolerance), plant abundance (frequency class, richness class, cover class), and landscape properties (patch size and patch isolation). Particular attention was given to the herbaceous flora since it represented the full range of dispersal modes recorded in these forests. Pattern was assessed with reference to descriptive statistics, non-parametric statistics, detrended correspondence analysis (DCA), and canonical correspondence analysis (CCA).

This thesis represents the first time that the dispersal spectrum has been characterized in relation to microhabitats within the forest understory.

The contribution of dispersal to differences in species richness in sampled forest stands is analyzed in Chapter 4.

3.2 Methods

3.2.1 Classification of Dispersal Modes

3.2.1.1. Overview

Mode of dispersal was inferred from the morphology or known properties of the diaspore. Species were assigned to one or more of the following syndromes based on the known or presumed function of the facilitating feature: dispersal by animals (zoochory), dispersal by wind (anemochory),

dispersal by mechanical exputsion (ballochory), dispersal by prolonged dormancy in the soil (chronochory), dispersal by vegetative expansion (autochory), dispersal by unassisted means (atelechory), and, dispersal by more than one mode (polychory). Animal dispersal was further classified in relation to the proximate mechanism, in view of differences in habitat preferences and foraging patterns of the animal vectors; dispersal by ingestion (endozoochory), dispersal by adhesion (epizoochory), and, dispersal by active conveyance or handling (synzoochory). Dispersal by animal conveyance was further classified for similar reasons; dispersal by seed caching, and, dispersal by ants (myrmecochory).

Dormancy is considered to be a dispersal character, in this study, in recognition of the contribution of persistent seed pools to community structure and species richness (Venable and Brown 1988, 1993; Kalisz *et al.* 1997). Other plant traits that contribute to dispersal in time, such as iteroparity, perenniality, and persistent life history stages, are considered alternatives to dispersal since they do not affect the morphology or physical attributes of the diaspore. Serotiny, if present, would have been a dispersal character since it affects the release of the diaspore. Despite their short dispersal reach, rhizomes, stolons, and runners are treated as dispersal characters since they satisfy the definition of dispersal when they result in the establishment of a detached, independently rooted, ramet.

The purpose of the classification is descriptive: to identify the agents, structures and attributes that facilitate the dispersal of recorded taxa. It is concerned solely with the capacity of the agent to transport a spore or seed unharmed, and is silent with respect to the capacity of the agent to deliver the propagule to a "safe site" (dispersal efficiency), and, to the proportion of seedlings in a population that may be attributed to a particular vector (dispersal effectiveness) (Harper 1977, Reid 1989, Bustamente *et al.* 1992, Schupp 1993).

The classification is also silent with respect to the status of facilitating structures and properties as adaptations for dispersal. Depending on the definition used, these features may be adaptations for dispersal or not (Reeve and Sherman1993). If the definition requires that the feature be built by

natural selection for its current role (e.g. Gould and Vrba 1982), then most features would not be adaptations for dispersal since their original function may have been to protect the developing ovule (Stebbins 1974) or to serve some other purpose. If the definition only requires that the feature or property results in higher fitness than alternative variants in the same environment (Reeve and Sherman 1993), then many features in this study would be considered adaptive in some habitats but not, perhaps, in all. In keeping with the requirements of many definitions of adaptation, dispersal characters have been shown to contribute to plant fitness, to be heritable, and to be subject to natural selection (e.g. Cody and Overton 1996, Venable *et al.* 1995).

The morphology and dispersal properties of diaspores were evaluated in the field and with reference to the published literature and the Montgomery Seed Collection. Royal Ontario Museum, Toronto, Ontario. Fruit terminology follows Gleason and Cronquist (1991). The classification of recorded taxa is presented in Appendix 2; a listing of species codes and corresponding scientific names is presented in Appendix 3.

3.2.1.2 Classification Criteria and Related Considerations

Dispersal by Animals (Zoochory): Dispersal by animals may be achieved by ingestion (endozoochory), adhesion (epizoochory) or active conveyance (synzoochory) (Semander 1901, as cited in Fahn and Werker 1972). Facilitating properties and considerations are discussed below. The principal frugivores and myrmecochores in Eastern North America are presented in Appendices 12 and 13, respectively.

Animal Ingestion (Endozoochory): Dispersal by ingestion is achieved when the fleshy pulp of an aril or fruit is eaten by an animal and the seeds are regurgitated or excreted unharmed (Ridley 1930, van der Pijl 1982, Willson *et al.* 1990). Species were so classified if the seed possessed a fleshy aril: or, if the fruit was a berry, drupe, drupelet, pome: or, if an achene was embedded in a fleshy receptacle or enclosed by a fleshy hypanthium. Fleshy fruits that possessed an elaiosome were classified as ant dispersed. <u>Trillium erectrum</u> and <u>Streptopus roseus</u> were classified as ant dispersed but may be polychorous based on the analysis of <u>Trillium</u> by Berg (1958), as cited in Stebbins 1974,

and, on the classification of Streptopus by Willson 1986.

The criterion for dispersal by ingestion excludes seeds and fruits eaten by granivores since the hard seed coat, which is the facilitating property, cannot be discerned visually, and, since published tests of viability were not sufficient to classify the taxa in this study (e.g. McAtee 1947, Krefting and Roe 1949, Flaming and Proctor 1968). Non-fleshy diaspores consumed by animals were therefore classified as dispersed by "unassisted" means. While dispersal by granivores is generally discounted (Willson 1986), some fraction of consumed seeds invariably survives passage through the beak, gizzard, or stomach of the bird, mammal or insect, and contributes to the long-distance dispersal of species that lack fleshy diaspores (Krefting and Roe 1949, van der Pijl 1982, Morton and Hogg 1989, Stiles 1989, Schupp 1993). The surviving fraction varies with both the diaspore and the consuming agent. Collinge (1913) reports the germination of 281 species of herbaceous plants from 142 droppings of the house sparrow, bull finch and green finch, whereas Rossfer (1936), in contrast, reports that only 7 of 40,000 seeds of various herbaceous plants germinated in the scat of California linnets.

The criterion for dispersal by ingestion also excludes nuts swallowed and regurgitated by jays and other seed caching animals. While satisfying the operational requirement that seeds be regurgitated or excreted unharmed, the swallowing of nuts for the purpose of later retrieval and consumption (Darley-Hill and Johnson 1981, Johnson and Adkisson 1985) implies intentionality that is lacking in dispersal by frugivores. Nut fruits were therefore classified as being dispersed by active conveyance or handling.

Fleshy fruits in the forests of Eastern North America are consumed by birds, mammals, and at least one reptile, the Eastern Box Turtle (Martin *et al.* 1951). Most fleshy fruits in these forests are consumed by both birds and mammals based on data summarized in Martin *et al.* (1951). In that study, 46 of 48 genera of fleshy-fruited plants east of the Mississippi River were consumed by each animal class. Overlaps with birds and mammals have also been reported for genera consumed by the Eastern Box Turtle (Hamilton 1941, Wilson 1986, Rust and Roth 1981). Further differentiation among fleshy fruits arising from differences in foraging and processing within and among animal classes was not pursued, therefore, in view of the lack of capacity to discriminate among animal agents (based on the presence or absence of selected fruit characters).

Animal Adhesion (Epizoochory): Dispersal by animal adhesion is achieved when a diaspore becomes passively attached to the fur or feathers of an animal by means of barbed, hooked or scabrous awn, bristle, hair or spine, and the seed is subsequently removed or released unharmed (Sorensen 1986). The facilitating structure in this study may be present on the fruit (typically), hypanthium (Agrimonia), involucral bract (Arctium, Cirsium), lemma (Poaceae), stem or leaf (Galium, Leersia).

The criterion for dispersal by adhesion excludes fruits that become sticky when wet (viscid fruits) since this character is not readily observed and is rarely reported in technical manuals. Species that may be dispersed by mucilaginous seeds in this study are <u>Plantago major</u> and <u>Prunella vulvaris</u> (Ridley 1930, p.549). Kerner (1895, p.869) reports that the fruits of <u>Solanum</u> are sticky when overripe and adhere to the hair and bristles of animals.

The criterion for dispersal by adhesion also excludes diaspores that may be dispersed in mud adhering to the feet of birds or mammals (Ridley 1930) since facilitating structures are not required for this mode of dispersal. Species that are most likely to be dispersed by this mechanism are small-seeded plants of wet and wet-mesic habitats.

The adhesive status of diaspores with awns, bristles, hairs or spines that lack barbs, hooks or scabrous properties was selectively tested with fruits from the Montgomery Collection. Royal Ontario Museum, Toronto, Although diaspores lacking these structures were excluded by Sorensen (1986), their adhesive properties may be sufficient to facilitate short-distance dispersal, particularly when wet (Morton and Hogg 1989, personal observation). Diaspores that adhered to an inclined cotton shirt when dropped from an upraised hand were deemed to be "adhesive" and were included in the classification (cited as "Montgomery Collection Test" in Appendix 2).

Active Handling or Animal Conveyance (Synzoochory): Dispersal by animal conveyance is achieved when the diaspore is intentionally transported by an animal for some purpose and the seeds are regurgitated or discarded unharmed (Fahn and Werker 1972, van der Pijl 1982). When the diaspore is transported for the purpose of storing food, but is forgotten or not eaten, the dispersal syndrome is "seed caching" (Smith and Reichman 1984). "larder hoarding", or "scatter hoarding" (Stapanian and Smith 1978). When the diaspore bears an elaiosome (an outgrowth of a seed or fruit in which lipids are stored) that is eaten by ants, or by vespid wasps, and the seed is discarded unharmed, the syndrome is "myrmecochory" (Sernander 1906) and "vespichory" (Jules 1996), respectively. In this study, the elaiosome may be attached to the seed (typically), schizocarp (<u>Galium</u>), base of the style (<u>Carduus</u>, <u>Cirsium</u>), or, comprise the basal portion of the perigynium (<u>Carex</u>) (Sernander 1906). In this study, the acknowledged fruit type dispersed by seed caching is the nut, although other fruit types may be so dispersed.

The criterion for dispersal by animal conveyance leaves open the purpose for which the diaspore is conveyed. Diaspores passively conveyed on plant parts that are intentionally transported for nesting material (Ridley 1930, Morton and Hogg 1989) is not considered to be synzoochory in this study since the property of interest does not relate to the diaspore.

The traditional test for myrmecochory has been behavioral: transport by ants and disposal of the seed unharmed (Sernander 1906. Berg 1975. Hughes and Westoby 1992b). Chemical assays and morphological assessments are not sufficient owing to uncertainty regarding the nature of the attractant (Bresinsky 1963, Marshall *et al.* 1979, Howard *et al.* 1981, Gordon 1983, Skidmore and Heithaus 1988. Brew *et al.* 1989) and the status of arils, caruncles and strophioles as agents of myrmecochory (Fahn and Werker 1972, Roth 1977, Beattie 1985). Therefore, only species that have been tested and been shown to be dispersed by ants have been classified as myrmecochores. This requirement was waived for <u>Carex pensylvanica</u>, based on an analysis of lipid content conducted for this study by Dr. M. Kahn, Department of Botany, University of Toronto, and, for <u>Clavtonia caroliniana</u>, <u>Dicentra canadensis</u>, <u>Polvgala pauciflora</u>. <u>Viola labradorica</u>, <u>Viola</u> sp., based on the known status of conspecifies and the presence of prominent "elaiosomes". Additional genera not

classified as myrmecochores in this study, but reported to be ant-dispersed in Europe, are <u>Ranunculus</u> (Roth 1977). <u>Juncus</u> (Bresinsky 1963). <u>Allium, Convovulus, Festuca, Geranium, Impatiens, Irís, Polygala, Polygonum, Potentilla., Silene, Stellaria, Trientalis, and Waldsteinia</u> (Semander 1906).

Dispersal by Wind (Anemochory): Dispersal by wind is achieved when the rate of descent of the diaspore is slowed by an aerodynamic shape or by virtue of its small size and light weight (Ridley 1930, Fahn and Werker 1972, van der Pijl 1982, Burrows 1986). Structures that facilitate wind dispersal in this study are wings, accrescent tepals, an inflated perigynium, a detachable paniele, a coma of fine hairs, or, a pappus of plumose, capillary or barbellate bristles.

The criterion for wind dispersal makes provision for small, light diaspores that otherwise lack a facilitating morphology. This approach departs from Willson *et al.*(1990), who excluded such diaspores because of the problem in defining an appropriate threshold size. This argument has been accepted for seeds but not for spores. The former have been classified as "unassisted", whereas the latter have been classified as "wind dispersed". Orchid seeds, while dust-like in character (and thus classified as "unassisted" by Willson *et al.*(1990), have a flattened wing-like region surrounding the germ (Summerhayes 1951, Montgomery 1977, personal observation), or an inflated and air-filled seed coat (Rasmussen 1951, Arditti 1992), that contributes to their capacity to remain aloft for long periods, and thus are classified as wind dispersed in this study.

Wind-push contributes to the short-distance dispersal of many fruits through the catapult effect of swaying stems or direct pressure on a flattened profile. In keeping with the conservative approach adopted in this study, most of these species have been classified as dispersed by "unassisted" means since they lack a distinctive facilitating morphology. Selected species that have been classified as "wind ballists" (van der Pijl 1982) are described in the section entitled "Dispersal by Mechanical Expulsion (Ballochory)".

"Aerodynamic" structures, such as plumose hairs, wings, accrescent tepals, and inflated perigynia.

may enhance the buoyancy of diaspores on water as well. This dimension has not been classified owing to the lack of flowing water in surveyed stands. Classified species that may be water dispersed in other settings include: <u>Boehmeria cvlindrica</u>, <u>Carex intumescens</u>, <u>Carex retrorsa</u>, <u>Rumex</u> <u>orbiculatus</u>, <u>Salix</u> spp., and <u>Typha latifolia</u>.

Dispersal by Prolonged Dormancy (Chronochory): Dispersal by prolonged dormancy is achieved when a seed regenerates from a long-term persistent seed bank (Thompson *et al.* 1997). Species were classified as chronochores if their seeds have been reported to persist in the soil for at least five years, and, if the methodology for determining persistence satisfied the criteria of Thompson *et al.* (1997) for a Type 3 seed bank.

The threshold for defining long-term persistence was established pragmatically since five years represented the termination point for a significant proportion of burial experiments (Thompson *et al.* 1997). The ecological significance of this threshold is that it differentiates seeds with limited capacity to persist in the soil from those with greater capacity. Seeds that can persist at least five years in the soil often remain viable for much longer (Thompson *et al.* 1997) and thus represent an important source of propagules for the colonization of disturbed sites.

The term "chronochory" (khronos = time + chorein = to wander) originates with this study.

Dispersal by Mechanical Expulsion (Ballochory): Dispersal by mechanical expulsion is achieved when the seed is forcefully expelled from the plant by explosive dehiscence (Ridley 1930, Fahn and Werker 1972, Beer and Swain 1977, Stamp and Lucas 1983), a splash-cup or springboard mechanism activated by failing water droplets (Brodie 1951, 1955; Savile 1953, 1979; Savile and Hayhoe 1977), or, by a swaying stem activated by wind-push (Ridley 1930, van der Pijl 1982).

The criterion for mechanical expulsion by a swaying stem has been applied conservatively since the dispersal of most plants is enhanced by this mechanism. Only plants that are known for catapulting their seeds, or that possessed a specialized morphology (e.g. a censer mechanism, or an inflated

hypanthium) for controlling or enhancing the release of seeds from swaying capsule or follicle, were recognized as anemoballists in this study.

Dispersal by Vegetative Expansion (Autochory): Dispersal by vegetative expansion is achieved when a detachable propagule, or, a disintegrating rhizome, stolon or runner. gives rise to a detached, independently rooted, ramet (Bell 1991). Detachable propagules in this study are bulbils (<u>Cicuta bulbifera</u>, <u>Cystopteris bulbifera</u>), twigs and branchlets (<u>Salix spp.</u>).

Dispersal by vegetative expansion has been excluded as a dimension of polychory to facilitate comparisons between sexual and asexual modes of dispersal.

Dispersal by Unassisted Means (Atelechory): Dispersal by unassisted means is achieved by diaspores that lack an apparent feature or property to facilitate their movement through space or time. These species are presumably dispersed by animal ingestion since each has a sizable geographic range in Eastern North America (Gleason and Cronquist 1991).

Dispersal by Multiple Agents (Polychory): Dispersal by multiple agents is achieved when the diaspore is dispersed by two or more of the following modes: zoochory, anemochory, chronochory, or ballochory. Dispersal by vegetative expansion has been excluded as a dimension of polychory to facilitate comparison between sexual and asexual modes of dispersal.

The results of this classification are presented in Appendix 2.

3.2.2 Classification of Other Plant Traits

Additional plant traits evaluated in this study were life form (tree, shrub, vine, fern, fern ally, grass, herb), life history (annual, biennial, perennial), provenance (native, alien), and fruit type (achene, berry, caryopsis, capsule, drupe, follicle, legume, nut, nutlet, pome, samara, schizocarp, and silique). The spores of inventoried ferns and fern allies, and the arillate and winged seeds of inventoried gymnosperms, were classified as "fruit" for ease of use. The authority for provenance was Morton
and Venn (1990): the authority for life form, life history and fruit type was Gleason and Cronquist (1991). A summary of plant traits by species is presented in Appendix 2.

3.2.3 Identification of Pattern

In this study, a relationship between (among) variables was deemed to constitute a "pattern" when the relationship was significant statistically, or, when the variables of interest were aggregated in ordination space. Protocols and assumptions for testing patterns of association are described below.

3.2.3.1 Pattern in Relation to Plant Attributes

The relationship between modes of dispersal and other plant traits was examined with respect to life form, life history, provenance, modality, fruit type, and taxonomic rank. These traits are known, or have the potential, to vary with environmental conditions for independent reasons and thus may confound interpretations of the role of dispersal in structuring the composition and abundance of plant assemblages. Published data were not sufficient to characterize the germination traits of sampled species.

The relationship between modes of dispersal and examined traits was summarized by descriptive statistics and tested by chi-square tests of homogeneity by row or cell. In keeping with the subsequent focus on understory herbs, only the results for the herb life form are reported. The expected value for each test was the proportion of herbs in the sampled flora with the trait of interest. The defining equation was the following:

Expected value = (the number of herbs dispersed by mode in sampled patches+ the total number of herbs in the data set) X (the number of herbs with the trait of interest).

The total number of herbs in sampled patches was 252: however, the total number of herbs in a given test varied, since taxa were not always classifiable in relation to the trait of interest. Given the large number of tests in each analysis, the status of tests after Bonferroni correction for the number of tests was reported.

The analyses were performed in JMP, Version 3.2.2, SAS Institute. The results of this analysis are reported in Tables 3.1 to 3.8.

3.2.3.2 Pattern in Relation to Environmental Variables

The tendency for the dispersal modes of herbs to be associated with particular environmental states was examined in relation to habitat affinity, moisture affinity, shade tolerance, environmental gradients, and microhabitats (see Chapter 2 for definitions and descriptions of variables).

The relationship between modes of dispersal and the habitat affinities of herbs was tested by chisquare tests of homogeneity, by tow and cell. Chi-square tests of independence were not appropriate for this analysis since selected herbs were dispersed by more than one mode and thus were recorded in more than one dispersal category.

The relationship between modes of dispersal and selected environmental gradients was tested by Wilcoxon rank sum tests with independent samples. The response variable in each test was the proportion of taxa in 10 m x 10 m quadrats that were herbs dispersed by a given mode. Analysis of variance was not appropriate for this analysis since the response variable was based on count data and rarely satisfied the normality and equal variance assumptions of ANOVA. Only the major gradients (see Chapter 2) were included in this analysis.

Analyses were performed in JMP. Version 3.2.2., SAS Institute, Inc. The results of this analysis are reported in Tables 3.9 to 3.12,

Patterns of association were also examined in the ordination space of detrended correspondence analysis (DCA) in order that trends may be assessed in relation to compositional similarities among quadrats and to environmental affinities among species. DCA was preferred to CCA for this analysis, since the dispersion of quadrats in DCA is governed by species relations with the underlying environment rather than with the restricted set of variables chosen for study. Pattern analysis was conducted by visual inspection. In order to reveal the main trends in the data, only the quadrats in which the mode was prominent were labeled (i.e. quadrats in which the proportion of taxa dispersed by the labeled mode was a the 75% quantile of proportions for that mode in the study area).

The distribution of dispersal modes in relation to environmental variables was evaluated with reference to canonical correspondence analysis (CCA). Here the pattern of interest was the degree to which modes of dispersal were associated with the set of environmental variables examined in Chapter 2. For this analysis, the species codes of species were replaced by modes of dispersal which then functioned as "pseudo-species" in the ordination. The abundance values for each mode were the proportions of taxa dispersed by the mode in each 10m x 10m quadrat in which it occurred.

The DCA and CCA ordinations were performed in CANOCO. Version 3.12: the ordination diagrams were produced in S-Plus, Version 4.5, and re-formatted for presentation purposes in Microsoft Publisher 98. The results of this analysis are presented in Figures 3.1 to 3.6.

The relationship between modes of dispersal and environmental states was re-examined at the microhabitat scale since environmental conditions within quadrats were rarely uniform. Pattern was investigated in relation to three analytical contexts: i) difference in the number of herbs recorded in each microhabitat; ii) difference in the proportion of herbs in contrasting microhabitats, paired samples; iii) difference in the proportion of herbs in contrasting microhabitats amples.

Context (i) clarified the degree to which differences in the proportion of herbs in contrasting habitats were due to the number of herbs dispersed by the mode of interest: context (ii) clarified the degree to which differences in the proportion of herbs in contrasting habitats was due to factors other than dispersal: context (iii) revealed the degree to which the frequency of modes changed under contrasting habitat conditions but did not clarify the role of dispersal since the degree of dispersal limitation in sampled quadrats was not known.

The relationship between dispersal mode and microbabitat in context (i) was tested by chi-square tests of homogeneity by column, row, and cell. The relationship in context (ii) was tested by

Wilcoxon signed-ranks tests, using paired samples from each quadrat in which the contrasting habitats were present. The number of habitat contrasts that could be assessed by this method was constrained by the minimum sample size associated with this test (6 paired comparisons, all of like sign, are required for the test to be significant at the 5% level; Sokal and Rohlf 1995, p.444). The relationship in context (iii) was tested by Wilcoxon rank sum tests, using independent samples. Quadrats in which the mode was present in each habitat were excluded from the analysis.

Analyses were performed in JMP. Version 3.2.2., SAS Institute. Inc. The results of this analysis are reported in Tables 3.13 to 3.15.

3.2.3.3 Pattern in Relation to Abundance Variables

The relationship between mode of dispersal and plant abundance was examined in relation to species frequency, plant cover, and species richness.

The relationship between mode of dispersal and species frequency was tested by chi-square tests of homogeneity by row and cell. Three frequency classes were established for the analysis: "high" (\geq 25 quadrats), "intermediate" (3-24 quadrats), and "low" (\leq 2 quadrats). The thresholds for the "high" and "low" classes were taken, arbitrarily, to be the 75% and 25% quantiles of species frequency, respectively.

The relationship between mode of dispersal and plant cover was tested by chi-square tests of homogeneity by column (*df* 8), row (*df* 6), and cell (*df* 1). Eight cover classes were established for the analysis: trace (1-5 individuals or small clumps), <1%, 1-5%, 5-15%, 15-25%, 25-50%, 50-75%, and 75-100%. These thresholds conform to the Daubenmire cover scale when cover is greater than 25% and, with minor exceptions, to the Domin-Krajina cover scale when cover is below 25% (Mueller-Dombois and Ellenberg 1974).

The relationship between mode of dispersal and species richness was tested by Wilcoxon rank sum tests, by row. Three classes of species richness (number of species) were established for the

analysis: "high" (\geq 56 taxa per quadrat), "intermediate" (29-55 taxa per quadrat), and "low" (\leq 28 taxa per quadrat). The thresholds for the "high" and "low" classes were taken, arbitrarily, to be the 75% and 25% quantiles of species richness, respectively.

Analyses were performed in JMP, Version 3.2.2., SAS Institute, Inc. The results are reported in Tables 3.16 to 3.18.

3.2.3.4 Pattern in Relation to Spatial Scale

The relationship between mode of dispersal and spatial scale was examined in relation to patch size and patch isolation.

In each case, the relationship was tested by Wilcoxon rank sum tests, by row. Three patch size classes were established for the analysis: "high" (\geq 122 ha), "intermediate" (43-121 ha), and "small" (\leq 42 ha). Three patch isolation classes were established for the analysis: "high" (mean distance to the nearest 8 woodlots, in 45° sectors, \geq 477 metres), "intermediate" (mean distance 233-476 metres), "low" (mean distance \leq 232 metres). The thresholds for the "high" and "low" classes were taken, arbitrarily, to be the 75% and 25% quantiles of patch size, and patch isolation, respectively.

Analyses were performed in JMP. Version 3.2.2., SAS Institute, Inc. The results are reported in Tables 3.19 and 3.20.

3.3 Results

Analyses for pattern in the tendency for modes of dispersal to be associated with particular states will be reported in relation to the following functional groupings: plant attributes, environmental variables, abundance variables and spatial scale.

3.3.1 Pattern in Relation to Plant Attributes

The tendency for dispersal modes to be associated with particular plant attributes are summarized

in relation to life form, life history, provenance, fruit type, and taxonomic rank. Each attribute has the potential to confound interpretations of the role of dispersal in structuring the composition and abundance of herb assemblages on the forest floor.

3.3.1.1 Life Form

The mode of dispersal varied by life form (Table 3.1). Most trees, and all ferns and fern allies, were dispersed by the wind, whereas most shrubs and vines were dispersed by animal ingestion. Most grasses were dispersed by animal adhesion. Herbs were the only life form to be dispersed by all modes and were dispersed primarily by vegetative expansion, animal vectors or unassisted means.

All life forms were dispersed by the wind and all but the trees achieved vegetative expansion by rhizomes or stolons. With the exception of the ferns and fern allies, all life forms were dispersed by animals and by prolonged dormancy in the soil. Only the grasses and herbs were dispersed by adhesion to animals, and only the herbs and one tree (*Robinia pseudoacacia*) were dispersed by mechanical expulsion. Only the herbs were dispersed by ants. A minority of shrubs, and a modest fraction of grasses and herbs, were dispersed by unassisted means. All but the ferns and fern allies were dispersed by more than one mode.

The dispersal modes of life forms also changed with the stratum of the forest. Canopy trees were dispersed primarily by the wind, whereas shrubs and vines were dispersed primarily by animal ingestion. All modes were present in the herb layer on the forest floor.

Taxa in these forests were dispersed primarily by animals (38.5%), vegetative expansion (35.8%), or the wind (30.5%). Fewer taxa were dispersed by unassisted means (22.0%), prolonged dormancy in the soil (15.5%), and mechanical expulsion (4.6%). Approximately 12% of surveyed taxa were dispersed by more than mode.

These results are not supported by statistical analysis since the sparse data table did not permit

Table 3.1. Dispersal modes of surveyed taxa by life form. Legend: AI = animal ingestion, AA = animal adhesion, AC = animal conveyance, W = wind, PD = prolonged dormancy in the soil, ME = mechanical expulsion, U = unassisted, MM = multimodal, VE = vegetative expansion. Entries are the number of taxa recorded in 192, 10m x 10m quadrats.

Life I	orm						Dispersal	Mode			
			Ani	mal							
Form	# Taxa	AP	AA	AC	Total	W ¹	PD	ME	U	MM	VE
tree	30	1	0	6	7	22	1	1	0	l	0
shrub	55	44	0	1	45	6	3	0	3	3	17
vine	9	8	0	0	8	1	1	0	0	1	2
fern	23	0	0	0	0	23	0	0	0	0	18
fern ally	8	0	0	0	0	8	0	0	0	0	4
grass	36	0	19	0	19	2	9	0	10	7	13
herb	252	17	34	33	80	64	50	8	78	39	94
Total	413	70	53	40	159	126	64	19	91	51	148
%	100	16.9	12.8	9,7	38.5	30.5	15.5	4.6	22.0	12.3	35.8

Notes:

1. Column values for a given dispersal mode include species that are multimodal.

chi-square tests of independence or homogeneity.

The finding that mode of dispersal differs among life forms has important implications for analyses of pattern since correlations with dispersal mode may be confounded with life form. For this reason, the results of subsequent analyses are either summarized by life form or relate strictly to herbs.

3.2.1.2 Life History

Approximately 90 % of taxa were perennial (Table 3.2). Trees, shrubs, vines, ferns and fern allies were strictly perennial. Annual and biennial grasses and herbs were dispersed by prolonged dormancy in the soil (45.7 %), wind (37.1 %), adhesion to animals (28.6 %), mechanical expulsion (17.1 %), unassisted means (14.3 %), and ants (11.4 %). None was dispersed by animal ingestion, seed caching or vegetative expansion. Approximately one-third of the species dispersed by multiple modes were annuals or biennials.

The number of herbs dispersed by animal adhesion, wind, prolonged dormancy, mechanical expulsion, and multiple modes varied with life history (p<0.05, chi-square tests of homogeneity by row, *df* 2, not shown; tests for prolonged dormancy, mechanical expulsion and multiple modes significant after Bonferroni correction for 8 row tests). The number of taxa dispersed by prolonged dormancy, mechanical expulsion and multiple modes was greater than expected for annual herbs, whereas, the number of taxa dispersed by animal adhesion, wind, prolonged dormancy, and multiple modes was greater than expected for biennial herbs (p<0.01, chi-square tests of homogeneity by cell). Analyses of pattern involving these modes of dispersal were therefore assessed for possible interactions with life history.

3.3.1.3 Provenance

Approximately 85% of taxa were native (Table 3.3). Among life forms, only the ferns and fern allies were strictly native. The percentage of alien taxa among other life forms was highest among

Table 3.2. Life History of surveyed taxa by dispersal mode and life form. Annual = annual *s.s.*; Biennial = annual/biennial. biennial *s.s.*; Perennial= annual/perennial, biennial/perennial, perennial s.s., "-" = not applicable. Superscript¹: total includes taxa with unknown life history. Superscript²: includes taxa dispersed by more than one mode.

Dispersal Mode				Ţ	ife History		
Mode by Life Form	Taxa #	#	Annual %	#	Biennial %	Pe #	rennial %
ALL TAXA	413'	13-	3.1	22	² 5.3	372 ¹	90.1
Animal s.l.	159 ¹	3	1.9	9	5.6	146	91.8
Ingestion	70	0	0	0	0	70	100.0
Adhesion	53	3	5.7	7	13.2	42	79.2
Seed Caching	7	0	0	0	0	7	100.0
Ant	33	ı	3.0	3	9.1	29	87.9
Wind	126	2	1.6	11	8.7	113	89,7
Prolonged Dormancy	64	6	9.4	10	15.6	50	78.1
Mechanical Expulsion	[8]	4	22.2	2	11.1	12	66.7
Unassisted	95	2	2.1	3	3.2	85	94.5
Multiple Modes	49	5	10.2	n	22.4	33	67.3
Vegetative Expansion	148	0	0	0	0	148	100.0
TREES	30	-	-	-	-	30	100.0
Ingestion	Į	-	-	-	-	1	100.0
Seed Caching	6	-	-	-	-	6	t00.0
Wind	22	-	-	-	-	22	100.0
Prolonged Dormancy	1	-	-	-	-	1	100.0
SHRUBS	55	-	-	-	-	55²	100.0
Ingestion	44 ²	-	-	-	-	44	100.0
Seed Caching	1	-	-	-	-	t	100.0
Wind	6 ²	-	-	-	-	6	100.0
Prolonged Dormancy	31	-	-	-	-	3	100.0
Multiple Modes	3	-	-	-	-	3	100.0

Dispersal Mode				Life	e History		
Mode by Life Form	Taxa #	#	Annual %		iennial %	Po #	erennial %
VINES	9	0	0	0	0	92	100.0
Ingestion	8 ²	0	0	0	0	8	100.0
Wind	I	o	0	0	0	1	100.0
Prolonged Dormancy	12	0	0	0	0	τ	100.0
Multiple Modes	1	0	0	0	0	1	100.0
FERN / FERN ALLIES	31	-	-	-	-	31	100.0
Wind	31	-	-	-	-	31	100.0
GRASSES	36'	1	2.8	0	0	32 ²	88.9
Adhesion	192	0	0	0	0	19	100.0
Wind	22	1	50.0	0	0	1	50.0
Prolonged Dormancy	9 ²	0	0	0	0	9	100.0
Unassisted	101.2	0	0	0	0	9	90.0
Multiple Mode	7	0	0	0	0	7	100.0
HERBS	2521	122	4.8	22 ²	8.7	215 ²	85.3
Ingestion	172	0	0	0	0	17	100.0
Adhesion	34 ^{1,2}	3	8.8	7	20.6	23	67.6
Ants	33 ¹	Ŧ	3.0	3	9.1	29	87.9
Wind	64 ²	1	1.6	11	17.2	52	81.3
Prolonged Dormancy	50²	6	12.0	10.0	20.0	34	68.0
Mechanical Expulsion:	18 ²	4	22,2	2	11.1	12	66.7
explosive dehiscence	112	3	27.3	1	9.1	7	63.6
wind-push	5²	1	20.0	1	20.0	3	60.0
splash-cup	22	0	0	0	0	2	100.0
Unassisted	78 ¹	2	2.5	3	3.8	72	91.1
Multiple Modes	39	5	12.8	110	28.2	23	59.0

Table 3.2. Life history of surveyed taxa by dispersal mode and life history (cont'd).

Dispersal Mode				Pro	venance		
Mode by Life Form	Taxa	N	ative		Alien	. [Jnknown
	#	[#]	%	#	%	#	%
ALL TAXA	413	349	84.5	60	14.5	4	1.0
Animal s.l.	159	142	89.3	16	10.1	l	0.6
Ingestion	70	65	92.9	5	7.1	0	0
Adhesion	53	43	81. ľ	9	17.0	1	1.9
Seed Caching	7	7	100.0	0	0	0	0
Ant	33	30	90.9	3	9.1	0	0
Wind	126	107	84.9	19	15.1	0	0
Prolonged Dormancy	64	26	40.6	38	59.4	0	0
Mechanical Expulsion	18	13	17.2	5	27.7	0	0
Unassisted	91	83	91.2	8	8.8	3	0
Multiple Modes	51	28	54.9	23	45.1	0	0
Vegetative Expansion	148	132	89.2	16	10.8	0	0
TREES	30	29	96.7	t	0.1	0	0
Ingestion	1	1	100.0	0	0	0	0
Seed Caching	6	6	100.0	0	0	0	0
Wind	22	22	100.0	0	0	0	0
Prolonged Dormancy	l	0	0	1	100.0	0	0
SHRUBS	55	51	92.7	4	7.3	0	0
Ingestion	44	40	90,9	4	9.1	0	0
Seed Caching	1	1	100.0	0	0	0	0
Wind	6	6	100.0	0	0	0	0
Prolonged Dormancy	3	2	66.7	1	33.3	0	0
Multiple Modes	3	2	66.7	1	33.3	0	o

Table 3.3. Provenance of surveyed taxa by dispersal mode and life form. Elements do not sum to group totals when taxa dispersed by more than one mode.

Dispersal Mode				Li	fe History		
Mode by Life Form	Taxa #	#	Native %	#	Alien %	#	Unknown %
VINES	9	8	88.9	ι	11.1	0	0
Ingestion	8	7	87.5	l	12.5	0	0
Wind	l	1	100.0	0	0	0	0
Prolonged Dormancy	1	0	0	1	100.0	0	0
Multiple Modes	1	0	0	1	100.0	0	0
FERN / FERN ALLIES	31	31	100.0	0	0	0	0
Wind	31	31	100.0	0	0	0	0
GRASSES	36	27	75.0	7	19.4	2	5.6
Adhesion	19	17	89.5	2	10.5	0	0
Wind	2	2	100.0	0	0	0	0
Prolonged Dormancy	9	5	55.6	4	44,4	0	0
Unassisted	10	7	70.0	3	30.0	0	0
Multiple Mode	7	5	71.4	2	28.6	0	0
HERBS	252	203	80.6	47	18.7	2	0.8
Ingestion	17	17	100.0	0	0	0	0
Adhesion	34	26	76.5	7	20.6	1	2,9
Ants	33	30	9,9	3	9.1	0	0
Wind	64	45	70.3	19	29.7	0	0
Prolonged Dormancy	50	19	38.0	31	62.0	0	0
Mechanical Expulsion:	18	13	72.2	5	27.7	0	0
explosive dehiscence	11	9	81. 8	2	18.2	0	0
wind-push	5	2	40.0	3	60.0	0	0
splash-cup	2	2	100.0	0	0	0	0
Unassisted	78	70	89.7	5	6.4	3	3.8
Multiple Modes	39	20	51.3	19	48.7	0	0

Table 3.3. Provenance of surveyed taxa by dispersal mode and life history (cont'd).

the grasses (19.4%) and herbs (18.7%), followed by the vines (11.1%), shrubs (7.3%) and trees (0.1%).

The number of herbs dispersed by animal ingestion, wind, prolonged dormancy, multiple modes and unassisted means varied with provenance (p<0.05, chi-square tests of homogeneity by row, df), not shown; tests for prolonged dormancy and multiple modes significant after Bonferroni correction for 9 tests). Dispersal modes with a higher than expected number of alien taxa were wind (p<0.05), prolonged dormancy in the soil (p<0.001), and multiple agents (p<0.001) (chi-square tests of homogeneity by cell). Analyses of pattern involving these modes of dispersal were therefore assessed for possible interactions with species provenance.

3.3.1.4 Modality

Approximately 12% of taxa were dispersed by more than one mode (Table 3.4). Among life forms, only the ferns and fern allies were unimodal. The percentage of multimodal taxa among other life forms was highest among the grasses (19.4%) and herbs (15.5%), followed by the vines (11.1%), shrubs (5.5%) and trees (3.3%).

The number of herbs dispersed by animal adhesion, wind, prolonged dormancy, and mechanical expulsion varied with modality (p<0.001, chi-square test of homogeneity by row, *df*), not shown; all tests significant after Bonferroni correction for 7 tests). In each case, the number of herbs dispersed by multiple modes was significantly higher than expected (chi-square tests of homogeneity by cell: all tests significant after Bonferroni correction for 14 tests). Analyses of pattern involving these modes of dispersal were therefore assessed for possible interactions with species modality.

3.3.1.5 Fruit Type

Among the herbs, the most abundant fruit types were the achene (49.8 %) and capsule (21.1 %) (Table 3.5). Minor fruit types in descending order of abundance were the schizocarp, berry, nutlet.

Dispersal Mode		Modality					
Mode by Life Form	Taxa	່ ເ	inimodal		Multimodal		
	#	#	<u>%</u>	#	%		
ALL TAXA	413	362	87.6	51	12.4		
Animal s.l.	160	133	83.1	27	16.8		
Ingestion	70	66	94.3	4	5.7		
Adhesion	53	35	66.0	18	34.0		
Seed Caching	7	7	100.0	0	0		
Ant	33	25	75.7	8	24.2		
Wind	126	106	84.1	20	15.9		
Prolonged Dormancy	64	25	39 .t	39	60 .1		
Mechanical Expulsion	18	8	44,4	10	55.6		
Unassisted	91	91	100.0	0	0		
Multiple Modes	51	0	0	51	100.0		
Vegetative Expansion	148	132	89.2	16	10.8		
TREES	30	29	96.7	I	3.3		
Ingestion	I	3	100.0	0	0		
Seed Caching	6	6	100.0	0	0		
Wind	22	22	100.0	0	0		
Prolonged Dormancy	1	1	100.0	0	0		
SHRUBS	55	52	94.5	3	5,5		
Ingestion	44	42	95.5	2	4,5		
Seed Caching	l	1	100.0	0	0		
Wind	6	5	83.3	1	16.7		
Prolonged Dormancy	3	0	0	3	100.0		
Multiple Modes	3	0	0	3	100.0		

Table 3.4. Modality of surveyed taxa by dispersal mode and life form. Unimodal = taxon dispersed by one mode; multimodal = taxon dispersed by more than one mode.

.....

Dispersal Mode			Mod	ality	
Mode by Life Form	Taxa #	.#	Unimodal %	#	Multimodal %
VINES	9	8	88.9	ł	11.1
Ingestion	8	7	87.5	1	12.5
Wind]	1	100.0	0	0
Prolonged Dormancy	1	0	0	1	100.0
Multiple Modes	l	0	0	1	100.0
FERN / FERN ALLIES	31	31	100.0	Q	0
Wind	31	31	100.0	0	0
GRASSES	36	30	81.6	6	19.4
Adhesion	18	t 3	72.2	6	33.3
Wind	2	E	50.0	i	50.0
Prolonged Dormancy	9	4	44,4	5	55.6
Unassisted	12	12	100.0	0	0
Multiple Mode	6	0	0	6	0
HERBS	252	213	84.5	39	15.5
Ingestion	17	16	94.1	ī	5.9
Adhesion	34	22	64.7	12	35.3
Ants	33	25	75.7	8	24.2
Wind	64	46	71.9	18	2 8 .1
Prolonged Dormancy	50	21	42.0	29	58.0
Mechanical Expulsion:	18	8	44.4	10	55.6
explosive dehiscence	11	4	36.4	7	3.6
wind-push	5	2	40.0	3	60.0
splash-cup	2	2	100.0	0	0
Unassisted	78	78	100.0	0	0
Multiple Modes	39	0	0	39	100.0

Table 3.4. Modality of surveyed taxa by dispersal mode and life history (cont'd).

ں - رن	∥ ∑	ч ті	
on, A	ž	ц Ц	
dhesi	isiste	=drup	
mal a	unas	5 Y -	
 ឧពៀ		Drupt	
¥.	sion,	ats.	dal.
stion	udxa	tpenb	limo
inge	ical (10m	¢ mu]
nimal	schan	0m X	ોત્રી હાઉ
16 = 1	ž =	92, 11	axa th
hd: A	Ŵ	d in t	ude 18
l'egei	soil,	sorde	s incl
ode.	ц Бе	Na rec	value
al m	ii ya	l∦ta	י חוחנ
ispers	ormar	c tola	it cola
by d	т С	are th	es tha
251)	gnok	lrics	cnot
s (N=	nd =	n. En	jī.
herb	2	ansio	Dersei
reved	wind,	eexp	s. Sul
f sur	 	stativ	drup
v pe o	с. с.	= veg	bles-
ruit ty	/eyan		(liese)
.5. F	con	todal,	upc, t
5 olde	úma.	ullim	up-ay
<u> </u>	Ξ	Ē	

l

							lsiCl	bersal Mod	5			
	з	à		٧V	lemi							
Fruh Type	Taxa	78 n−251	٩l	١V٧	ACI	Total	W	اراط	ME	n	MM	VE
achene	125	49.8	5	17	30	26	53	23	0	46	31	43
berry	13	5.2	10	0	m	6	0	0	¢	0	0	7
capsule	53	21.1	0	2	15	17	4	10	12	17	7	28
capsular	Ξ	4,4	0	0	ς.	٣	-1	4	_	0	ł	9
drupe s.L	4	9.1	न्त	c	0	+	0	0	0	0	0	_
folliele	<u>ارم</u>	2.0	=	=	Ċ	o	en.	m.	-	c	сI	CI
legume	7	2.8	0		ĊI	цъ,	¢	- .	2	-	~	Ð
nutlet	12	4,8	9	-	a	1	0	ñ	0	œ	0	4
schizocarp	19	7.6	0	13	0	5	0	3	2	খা	ŝ	ci
silique	L:	0.8	0	()	0	0	0	0	0	2	0	1
Total	251	100	17	34	33	80	64	50	18	78	39	94

capsular fruit, legume, follicle, drupe *s.l.*, and silique. The number of modes by which a given fruit type was dispersed varied from one to six (excluding multiple modes and vegetative expansion). No fruit type was dispersed by all modes. Fruit types dispersed by a variety of modes were the achene (6), capsule (6), schizocarp (5) and capsular fruit (4). Two fruit types were dispersed by one mode: drupe *s.l.* and silique.

Each dispersal mode deployed the seeds of more than one fruit type but no mode deployed the seeds of all ten fruit types. The modes deploying the seeds of the greatest range of fruit types were animals s.L, (8), and prolonged dormancy in the soil (7). The modes deploying the seeds of the least number of fruit types were animal ingestion (3) and wind (4).

The predominant mode by which a given fruit type was dispersed varied. Achenes were dispersed primarily by the wind; berries by animal ingestion; capsules by unassisted means and ants; capsules by wind and prolonged dormancy; drupes s.l. by animal ingestion; follieles and legumes by prolonged dormancy in the soil; nutlets and siliques by unassisted means; and, schizocarps by animal adhesion.

The data table summarizing the relationship between fruit type and dispersal mode was too sparse to test the overall relationship with standard statistical tests. However, selective row tests (not shown) revealed that the number of berbs with achene and capsule fruits differed by dispersal mode (p<0.001, chi-square tests of homogeneity by row, *df* 8). In particular, the number of herbs dispersed by wind (p<0.001) was significantly greater than expected for achene fruits: the number of herbs dispersed by mechanical expulsion (p<0.001) and by ants (p<0.01) were significantly greater than expected for herbs with capsule fruits (chi-square tests of homogeneity by cell).

Collectively, these results suggest that fruit type may have influenced the frequency of dispersal modes in these forests. Further analysis is required, however, since this data table could not be tested comprehensively.

3.3.1.6 Taxonomic Rank

The distribution dispersal modes by taxonomic rank is presented in Tables 3.6 (genus), 3.7 (family), and 3.8 (order). The most important trend for this thesis is that the number of taxa dispersed by more than one mode increases with taxonomic rank: species (12.3 %), genus (25.8 %), family (33.3 %), and order (57.4 %). This pattern has consequences for the number of degrees of freedom associated with linear regression analyses reported in Chapter 4. The contribution of phylogeny to species richness in vegetation samples will be discussed there.

3.3.2 Pattern in Relation to Environmental Variables

The analysis for pattern in relation to environmental variables was restricted to the herbs since it was the one life form that was dispersed by all modes. The results are summarized in relation to habitat affinity, moisture affinity, shade tolerance, environmental gradients, and microhabitats.

3.3.2.1 Habitat Affinity

The number of herbs dispersed by animal ingestion, ants, wind, prolonged dormancy, wind-push mechanisms, unassisted means, and multiple modes varied with the habitat affinity of the taxon (Table 3.9). Herbs dispersed by animal ingestion, ants, and splash-cup mechanisms were over-represented in taxa with an affinity for forest habitats, whereas, herbs dispersed by wind, prolonged dormancy, wind-push, and multiple modes were over-represented in taxa with an affinity for open habitats. The number of herbs dispersed by animal adhesion, explosive mechanisms, and vegetative expansion did not vary with habitat affinity. The strongest evidence was found for herbs dispersed by prolonged dormancy and by multiple modes (chi-square tests of homogeneity by cell, p<0.001, Bonferroni correction for 52 cell tests).

The significance of zero values in the data is difficult to assess since absence may be due to sampling error associated with small samples or to habitat related factors constraining dispersal, germination, establishment and/or persistence. The absence of herbs dispersed by animal ingestion and splash-cup mechanisms in taxa with affinity for open habitats, and by wind-push mechanisms

Table 3.6. Distribution of dispersal modes by taxonomic rank I (genus). Legend: AI=animal ingestion; AA=animal adhesion; AC=animal conveyance (ants, seed caching); W=wind; PD=prolonged dormancy; ME=mechanical expulsion; U=unassisted; MM=dispersed by >2 modes. Cell entries are the number of genera dispersed by the given mode(s). Only the lower half of matrix is shown, "-" denotes combination not observed in vegetation samples. All life forms included in analysis. N = 208 genera.

GENUS	AI n=35	AA n=31	AC n=20	₩ n=6ວິ	PD n=59	ME n=14	U n=44	ММ n=6
AI	32							
AA	-	16						
AC	1	-	11					
w	-	2]	45				
PD	2	9	2	13	15			
ME	-	-	I	-	7	5		
U	-	-	-		6	1	33	
ММ	-	4	4	3	5	-	3	6'

Notes:

 Genera dispersed by more than two modes: Carduus: AC, AA, W, PD Carex: AC, W, U Cirsium: AA, W, PD Galium: AA, AC, PD Melilotus: AC, PD, U Poa: AA, PD, U.

Table 3.7. Distribution of dispersal modes by taxonomic rank II (family). Legend: AI=animal ingestion; AA=animal adhesion; AC=animal conveyance (ants, seed caching); W=wind: PD=prolonged dormancy; ME=mechanical expulsion: U=unassisted; MM=dispersed by >2 modes. Cell entries are the number of families dispersed by the given mode(s). Only the lower half of the matrix is shown. "-" denotes combination not observed in vegetation samples. All life forms included in analysis. N = 78 families.

FAMILY	Al n=19	AA n=11	AC n=14	W n=27	PD p=27	ME n=11	U n=25	MM n=15
AI	12							
AA	-	1						
AC	-	-	7					
w	-	-	-	18				
PD	l	-	-	1				
ME	-	_	1	-	3	2		
U	l	-	-	-	3	1	8	
ММ	5	10	6	7	1-1	4	13	15'

Notes:

1. Families dispersed by more than two modes: Apiaceae: AA, PD, U Asteraceae: AA, W, PD, U Betulaceae: AC, W, PD Caprifoliaceae: AI, PD, U Caryophyllaceae: PD, ME, U Cucurbitaceae: PD, ME, U Cyperaceae: AA, AC, W, U Fabaceae: AA, AC, PD, ME, U Liliaceae: AI, AC, U Onagraceae: AA, W, PD Poaceae: AA, W, PD, U Ranunculaceae: AJ, AC, W, PD, ME, U Rosaceae: AI, AA, PD, U Rubiaceae: AI, AA, AC, PD Urticaceae: AA, W, PD, U Verbenaceae: AA, PD, U.

Table 3.8. Distribution of dispersal modes by taxonomic rank III (order). Legend: AI=animal ingestion; AA=animal adhesion; AC=animal conveyance (ants, seed caching); W=wind: PD=prolonged dormancy; ME=mechanical expulsion: U=unassisted; MM=dispersed by >2 modes. Cell entries are the number of orders dispersed by the given mode(s). Only the lower half of the matrix shown. "-" denotes combination not observed in vegetation samples. All life forms included in analysis. Nomenclature follows Mabberly 1997. N = 47 orders.

ORDER	AI n=15	AA n=9	AC n=12	W n=19	PD n=25	ME n=9	U n=20	MM n=l9
Aſ	5							
AA	-	-						
AC	-	-	4					
W	I	-	-	7				
PD	-	-	-	2	2		_	
ME	-	-	-	-	I	-		
U	2	-	-	-	2	-		
ММ	7	9	8	9	18	9	۱6	19'

Notes:

1. Orders dispersed by more than two modes: Apiales: AI, AA, PD, U Asterales: AA, W, PD, ME, U Caryophyllales: AC, PD, ME, U Cyperales: AA, AC, W, PD, U Dipsacales: AI, PD, U Fabales: AA, AC, W, PD, ME, U Fagales: AC, W, PD Gentianales: W. PD, U Lamiales: AA, PD, U Liliales: AA, PD, U Malvales: W, PD, ME Myrtales: AA, W, PD Ranunculales: AI, AC, W, PD, ME, U Rosales: AI, AA, PD, ME, U Rubiales: AI, AA, AC, PD, U Scrophulariales: W. PD, ME, U Solanales: AI, PD, U Urticales: AA, W, PD, U Violales: AC, PD, ME, U

Table 3.9. Dispersal modes of surveyed herbs by habitat affinity. "Forest": taxa occur only in forested or shaded habitats; "Forest-Open": taxa occur primarily in forested habitats but move into open habitats (including thickets); "Open-Forest": taxa occur primarily in open habitats but move into forested habitats; "Open": taxa occur only in open habitats. N = 234 herbs. Expected number of taxa = (# taxa dispersed by mode \div 234) x (# taxa in habitat category). Habitat assignments based on descriptions in: Voss 1972, 1985, 1996; Cody and Britton 1989; Dore and McNeil11980, Gleason and Cronquist 1991. Dispersal mode and observed value marked by an asterisk when differences among habitat classes significant at p<0.05. chi-square tests of homogeneity by row, *df* 3, and cell. *df* 1: dispersal mode and observed value in bold when differences significant after Bonferroni correction for the total number of tests (13 tests by row; 52 tests by cell).

· ·					Habitat	Class			
Dispersal Mode	# Taxa (n=234)	For (n= Obs.	rest 55) Exp.	Fores (n= Obs.	t-Open •54) Exp.	Open- (n= Obs.	-Forest =82) Exp.	Op (n= Obs.	oen -43) Exp.
Animal	76	29*	17.9	27*	17.5	11*	26.6	9	14.0
ingestion*	17	9*	4.0	5	3.9	3	6.0	Ð	3.1
adhesion	31	8	7.3	10	7.2	5	10.9	8	5.7
ant	31	[3*	7.3	13*	7.2	3*	10.9	2	5.7
Wind	62	6*	14.6	6*	14.3	28	21.7	22*	[1.4
Prolonged Dormancy	50	0	11.8	4*	11.5	21	17.5	25	9.2
Mechanical Expulsion	17	3	4.0	3	3.9	6	6.0	5	7.9
explosive	10	L	2.4	3	2.3	5	3.5	I	1.8
splash-cup	2	2*	0.5	0	0.5	0	0.7	0	0.4
wind-push	5	0	1.2	0	1.2	1	1.8	4*	0.9
Unassisted	69	19	16.2	18	15.9	27	24.2	5*	12.7
Multimodai	39	3*	9.2	5	9.0	П	13.7	20	7.2
Vegetative Expansion	92	22	21.6	23	21.2	37	32.2	10	16.9

in taxa with affinity for closed habitats, cannot be assessed owing to the small number of taxa involved. The absence of herbs dispersed by dormancy among taxa with forest affinity may be due to habitat related factors since their absence is consistent with factors hypothesized to favor the evolution of seed dormancy (Venable and Brown 1988, 1993).

The dispersal mode of herbs seldom varied with the known moisture affinity of the taxon (Table 3.10). Herbs dispersed by animal vectors *s.l.* were under-represented in taxa with an affinity for wetland habitats whereas herbs dispersed by unassisted means were over-represented in taxa with an affinity for wetland habitats (chi-square tests of homogeneity by cell, p<0.05, uncorrected for the number of tests). The number of herbs dispersed by other modes did not vary with the moisture affinity of the taxon.

The dispersal mode of herbs did vary, however, with the known shade tolerance of the taxon (Table 3.11). Herbs dispersed by animal ingestion were over-represented in taxa with high shade tolerance, whereas, herbs dispersed by wind, prolonged dormancy, and multiple modes were strongly over-represented in taxa with low shade tolerance, and, strongly under-represented in taxa with high shade tolerance. The number of herbs dispersed by adhesion, ants, mechanical expulsion, and vegetative expansion did not vary with the shade tolerance of the taxon.

3.3.2.2 Environmental Gradients

Pattern in relation to environmental gradients was investigated by univariate and multivariate methods. The former revealed the degree to which dispersal modes were associated with particular gradients and portions of gradients, whereas, the latter revealed affinities among quadrats dominated by a particular mode of dispersal and among species with similar environmental affinities.

Univariate Analysis

In general, the proportion of herbs in a 10m x 10m quadrat dispersed by a given mode varied in relation to the principal environmental gradients examined in this study (Wilcoxon rank sum tests.

Table 3.10. Dispersal modes of herbs by moisture affinity. N = 233 herbs. Moisture affinity classification based on Oldham *et al.* 1995. Expected number of taxa = (# taxa dispersed by mode \div 233) x (# taxa in moisture affinity class). Dispersal mode marked by an asterisk when differences among habitat classes significant at p<0.05, chi-square tests of homogeneity by row. *df* 2; observed value marked by an asterisk when departure from expectation significant at p<0.05, chi-square tests of homogeneity by cell. *df* 1: no test significant after Bonferroni correction for the total number of tests (13 tests by row; 39 tests by cell).

				Moistu	re Affiniț	у	
Dispersal Mode	# Taxa (n=233)	Up (n= Obs.	land 146) Exp.	Interr (n: Obs.	nediate =42) Exp.	We (n [.] Obs.	tland =75) Exp.
Animal*	75	48	37.3	13	13.5	14*	24.1
ingestion	17	12	8.5	4	3.1	t	5.5
adhesion	31	19	15,4	3	5.6	9	10.0
ant	31	20	15.4	6	5.6	5	10.0
Wind	60	33	29.9	7	10.8	20	19.3
Prolonged Dormancy	49	27	24.4	12	8.8	10	15.8
Mechanical Expulsion	18	11	9.0	3	3.2	4	5.8
explosive	11	6	5.5	L	2.0	4	3.5
splash-cup	2	1	1.0	1	0.4	0	0.6
wind-push	5	4	2.5	L	0.9	0	1.6
Unassisted*	69	23	34,4	13	12.4	33*	22.2
Multimodal	38	25	18.9	6	6.8	7	12.2
Vegetative Expansion	91	45	45.3	16	16.4	30	29.3

Table 3.11. Dispersal modes of herbs by shade tolerance. N = 143 herbs. Shade tolerance classification based on Nimerfro and Brand (1993) and Ellenberg (1988). Expected number of taxa = (# taxa dispersed by mode \div 143) x (# taxa in shade tolerance class). "High" = shade tolerance class 1.2: "Intermediate = shade tolerance class 3; "Low" = shade tolerance class 4.5. Dispersal mode and observed value marked by an asterisk when differences among tolerance classes significant at p<0.05, chi-square tests of homogeneity by row, *df* 2, and cell, *df*1 : dispersal mode in bold when differences among classes significant after Bonferroni correction for the total number of tests (13 tests by row).

				- Shade '	Folerance		
Dispersal Mode	# Taxa (n=143)	Hi (n≂ Obs.	igh =43) Exp.	Intern (n= Obs.	nediate =19) Exp.	[(n Obs.	.ow =81) Exp.
Animal	54	28*	16.2	8	7.2	18"	30.6
ingestion	15	9"	4.5	2	2.0	4	8.5
adhesion	22	10	- 6.6	3	2.9	9	12.5
ant	17	9	5 .1	3	2.3	5	9.6
Wind	42	2*	12.6	5	5.6	35*	23.8
Prolonged Dormancy	43	3*	12.9	1	5.7	39*	24,4
Mechanical Expulsion	12	4	3.6	3	1.6	5	6.8
explosive	6	2	1.8	3	0.8	1	3.4
splash-cup	2	2	0.6	0	0.3	0	1.1
wind-push	4	0	1.2	0	0.5	4	2.3
Unassisted	25	8	7.5	3	3.3	14	14.2
Multimodal	29	2*	8.7	t	3.9	26*	16.4
Vegetative Expansion	65	21	19.5	8	8.6	36	36.8

independent samples) (Table 3.12). Most modes varied along each gradient (p<0.01, Bonferroni correction for 5 row tests; p<0.001, Bonferroni correction for 50 tests). The major exception was dispersal by vegetative expansion which did not vary in relation to any gradient.

All modes were present on each portion of the gradients examined. Patterns associated with particular gradients are summarized below.

Canopy Closure: Under low canopy closure (high light), the largest proportion of herbs for a given mode was dispersed by animal adhesion, wind, prolonged dormancy, unassisted means, and multiple modes. These results reflect an underlying interaction between dispersal mode and habitat affinity since herbs dispersed by these modes are over-represented by taxa with open habitat affinities. Animal ingestion was the dominant mode of dispersal under high canopy closure (low light) whereas dispersal by ants was the dominant mode under medium canopy closure. Dispersal by mechanical expulsion did not vary with canopy closure.

Moisture: In quadrats with seasonally dry depressions, the largest proportion of herbs for a given mode was dispersed by animal ingestion and ants. In quadrats with seasonally moist depressions, the dominant mode was dispersal by prolonged dormancy. In quadrats with seasonally wet depressions, the predominant modes were animal adhesion, wind, mechanical expulsion, unassisted and multiple modes. Moisture conditions in quadrats with seasonally moist and wet depressions were typically dry, however, since the latter habitats occupied less than 50% of the quadrat, on average. The percentages for dispersal modes, therefore, tend to be higher in quadrats with moist and wet depressions since they include taxa with an affinity for mesic soils.

Percent Soil Organic Matter (% SOM): In quadrats with low and medium soil organic matter, the largest proportion of herbs for a given mode was dispersed by animal ingestion and by ants, respectively. In quadrats with high soil organic matter, the predominant modes were dispersal by animal adhesion, wind, prolonged dormancy, mechanical expulsion, unassisted means and multiple modes. Higher percentages of soil organic matter in these forests tend to be associated with tree pits

Table 3.12. Distribution of herbs in 10m x 10m quadrats by environmental attribute and dispersal mode (mean %). Cell values: mean % o
taxa in 10m x 10m quadrats dispersed by mode. Wilcoxon Rank Sum Tests by attribute and mode. Highest value in bold when difference
among modes significant at p<0.01 (Bonferroni correction for 5 column tests); highest value underlined when differences among mode
significant after Bonferroni correction for 50 tests. Notes: 1. Canopy closure: low = <80% closure, medium = 80-94% closure, high = >949
closure (fish-eye photography); 2. Moisture class: dry = forest depressions dry spring and summer; moist = forest depressions moist spring
and/or summer; wet = forest depressions with standing water spring and/or summer; 3. % SOM: percent soil organic matter in upper 15 cn
of soil profile: low = $<6\%$ SOM; medium - $6-14\%$ SOM; high = $>14\%$ SOM; 4. Cover (ypc: 1 = red or white oak, no sugar maple; 2 = suga
maple + red, white oak; 3 = sugar maple, no red, white oak, no wet mesic or wet tree species; 4 = sugar maple + black ash, silver maple o
American clm; 5 = black ash, silver maple, American elm, no sugar maple, no red, white oak, 5. Disturbance: trail, regenerating field o
canopy gap present in 10m x 10m quadrat. N = 252 herbs. Values do not sum to 100% when the tolerance of species is broader than category
thresholds.

	urbance ⁵	No	27.9	<u>10.9</u>	5.1	12.0	8.6	4.0	3.0
	Dist	Yes	24.5	7.9	7.1	9.6	<u>13.7</u>	<u>8.1</u>	3.7
	I	ŝ	13.9	2.8	<u>8.7</u>	2.4	<u>22.9</u>	<u>13.8</u>	4.0
	Ъ°.	4	23.8	7.7	6.9	9.3	12.2	2.7	5.1
	ver Tyl	ŝ	<u>32.2</u>	10.7	6.6	<u>15,3</u>	8.7	4.9	3.1
lte	Co.	c 1	25.5	<u>13.0</u>	2.9	9.7	7.1	1,7	t. 1
Attribu		-	10.5	3.5	č.ť	5.3	21.1	12.3	<u>072</u>
mental	f	High	23.9	7.1	<u>8.6</u>	8.2	12.4	<u>7.9</u>	<u>6.4</u>
inviron	6 SOM	Med	29.3	10.Á	ō.ō	<u>13.5</u>	10.1	4.8	<u> </u>
щ	à.	No.1	24.3	<u>11.7</u>	3.3	+-6	7.9	3.5	2.0
	ass ²	Wet	23.6	5.7	<u>8.4</u>	5.6	<u>14.3</u>	7,8	<u>0'0</u>
	sture CI	Moist	22.7	X'L	7.1	7.8	12.2	<u>8.0</u>	5.0
	Mois	Ūry	<u> 29.2</u>	<u>12.0</u>	4.5	13.0	8.3	3.5	8'
	sure	High	27.5	<u>11.4</u>	4.3	12.0	8 ,4	3.3	2.9
	py Clos	Med	28,7	10.9	5.7	12.4	8,8	4.3	3.0
	Cano	Yow	22.4	6.9	7.2	8.4	<u>14,6</u>	<u>0'1</u>	3.8
	Dispersal Mode		ANIMAL.	ingestion	adhesion	ari	WIND	PROLONGED DORMANCY	MECHANICAL EXPULSION

1('d).
10a) (
% u
(mea
ode
al n
ispers
րր
lic a
tribu
al ut
nent
IUUI
cinvi
i hy
adrats
ոգլ
20 10
(0m 2
ž in
herb
1 of
ution
itribi
Dis
či,
ole 3.
Tat

Mode Ca TED I.2. IDAL 7.6	w N 4 IC	Closu Acd F	re 1.4 1.4 1.4 1.4 1.4 1.4 1.4 1.4 1.4 1.4	Mois Bry 3.5	Moist 12.5	ass ² Wet <u>15.9</u>	1.0w 7.9 2.8	inviron 6 SOM Med 10.6 4.6	High	Attribu	Le Co	ver Typ 3 10.8 5.2	oe ⁴ 4 7.0	5 18.1 11.4	Dia Yes 11.9	sturbance ⁵ No 10.0 4.1
25.	0 27	.1	26.9	27.5	25.4	26.0	27.5	27.1	25.4	2),1	2.7.2	27.6	26.0	21.8	26.3	27.0

and with moist to wet topographic depressions, which typically occupied less than 50% of a given quadrat, when present. The percentages for dispersal modes therefore tend to be higher in quadrats with medium and high soil organic matter since they typically include taxa with affinity for low to medium percent soil organic matter.

Cover Type: The predominant modes under cover type "1" (red or white oak, no sugar maple) were dispersal by mechanical expulsion and by multiple modes. The dominant mode under cover type "2" (sugar maple + red or white oak) was dispersal by animal ingestion, whereas the dominant mode under cover type "3" (sugar maple, no red or white oak, no wet mesic or wet tree species) was dispersal by ants. No modes were dominant under cover type "4" (sugar maple + black ash, silver maple, or American elm). The predominant modes under cover type "5" (black ash, silver maple, American elm, no sugar maple, no red or white oak) were dispersal by animal adhesion, wind, prolonged dormancy, and unassisted means.

Disturbance: In quadrats with a trail, regenerating field, or canopy gap, the predominant modes of dispersal were animal adhesion, wind, prolonged dormancy, and multiple modes. This pattern reflects differences in the proportion of herbs with affinity for "open+forest" and "open" habitats (p<0.001, Wilcoxon rank sum test, independent samples, data not shown), rather than invasion by alien taxa. The proportion of alien taxa did not vary with disturbance class (p>0.05. Wilcoxon rank sum test, independent samples, data not shown). The dominant mode in quadrats without disturbance features was animal ingestion. The proportion of herbs dispersed by ants, mechanical expulsion and unassisted means did not vary with disturbance class.

Multivariate Analysis

Further evidence of the tendency for dispersal modes to be associated with particular environmental states is provided by the distribution of dispersal modes in the ordination space of detrended correspondence analysis (DCA). DCA is an ordination technique which arranges vegetation samples along gradients (axes) that maximize their dispersion on the basis of differences in species

composition and abundance. Quadrats that are close together in stand ordination space (Figures 3.1-3.5) are more similar in species composition than quadrats that are far apart. Quadrats separated by more than 4 standard deviations in these figures have few if any species in common.

The distribution of dispersal modes in ordination space reveals that quadrats in which animal ingestion, ants, and vegetative expansion are the predominate mode of dispersal are largely confined to the right side of the stand ordination (Figures 3.1, 3.2, 3.5) whereas quadrats in which the predominate modes of dispersal are animal adhesion, wind, prolonged dormancy, mechanical expulsion, unassisted, and multiple modes are concentrated on the left side of the ordination (Figures 3.1-3.5).

Quadrats in which the predominate modes are animal adhesion, wind or prolonged dormancy often overlap (50% of occurrences) (not shown). Quadrats in which animal ingestion, ants and vegetative expansion are the predominate modes also tend to overlap (62% of occurrences) (not shown). In contrast, dispersal by wind, prolonged dormancy and mechanical expulsion are rarely major constituents in quadrats where animal ingestion is the predominate mode of dispersal (Figures 3.2, 3.3). This is true even in the central portion of ordination where all modes are present. This suggests that the environments in which animal ingestion predominates are well defined and different from other modes.

Inspection of the survey data reveals that quadrats in which animal ingestion predominates are characterized by moderate to high canopy closure, mesic soils, moderate to low percent soil organic matter, low disturbance, and forest cover types 2 and 3. In contrast, quadrats in which wind, prolonged dormancy, and mechanical expulsion predominate are characterized by moderate to low canopy closure, moist to wet soils, moderate to high percent organic matter, high disturbance, and forest cover types 4 and 5. Unlike the latter modes, dispersal by animal ingestion is often the predominate mode of dispersal on glacial fluvial parent materials and never achieves predominance on calcareous outwash or lacustrine parent materials.



DCA Axis I

Figure 3.1. Distribution of herbs dispersed by animal ingestion (I) and by animal adhesion (A) in relation to DCA axes 1 and 2. Only quadrats in which the proportion of herbs dispersed by designated mode is > 75% quartile are labeled. Note minor overlap (O) in quadrats where animal ingestion and animal adhesion predominate.



Figure 3.2. Distribution of herbs dispersed by ants (E) and by wind (W) in relation to DCA axes 1 and 2. Only quadrats in which the proportion of herbs dispersed by designated mode is $\gtrsim 75\%$ quartile are labeled. Note differences in the degree of overlap (O) in quadrats where designated modes and animal ingestion predominate.



Figure 3.3. Distribution of herbs dispersed by prolonged dormancy in the soil (D) and by mechanical expulsion (E) in relation to DCA axes 1 and 2. Only quadrats in which the proportion of herbs dispersed by designated mode is \Rightarrow 75% quartile are labeled. Note minor or no overlap (O) in quadrats where designated modes and animal ingestion predominate



Figure 3.4. Distribution of herbs dispersed by unassisted means (U) and by multiple modes (M) in relation to DCA axes 1 and 2. Only quadrats in which the proportion of herbs dispersed by designated mode is $\geq 75\%$ quartile are labeled. Note minor or no overlap (O) in quadrats where designated modes and animal ingestion overlap.



DCA Axis L

Figure 3.5. Distribution of herbs dispersed by vegetative expansion (V) and by selected animal agents (A) in relation to DCA axes 1 and 2. Only quadrats in which the proportion of herbs dispersed by designated modes is 575% quartile are labeled. Note substantial overlap (O) in quadrats where animal ingestion and abiotic agents predominate.

The environments in which dispersal by ants predominates are similar to animal ingestion (Figure 3.2). Unlike the latter, however, dispersal by ants was predominate in quadrats with wet depressions and in quadrats on calcareous outwash parent materials. The environments in which dispersal by mechanical expulsion predominates are similar to dispersal by wind and prolonged dormancy (Figures 3.2, 3.3), but trend towards the closed and wet ends of the environmental continuum. The quadrats in which dispersal by animal ingestion and ants predominate rarely overlap with quadrats in which dispersal by animal ingestion and ants predominate rarely overlap with quadrats in which dispersal by wind, prolonged dormancy or mechanical expulsion predominate (Figure 3.5).

An important trend in the data which has been masked by the decision to label only those quadrats in which the mode predominates is that some modes are more frequent than others. Herbs dispersed by vegetative expansion, animal ingestion and ants occur in virtually every quadrat whereas herbs dispersed by prolonged dormancy, mechanical expulsion and multiple modes are absent from more than 30 % of quadrats. Herbs dispersed by unassisted means, wind and animal adhesion are absent from 6 %, 10 % and 17 % of quadrats, respectively. Most absences occur in closed, dry, undisturbed habitats on calcareous till and glacio-fluvial parent materials.

The tendency for modes of dispersal to predominate in particular environmental states is readily apparent in the CCA ordination presented in Figure 3.6. In keeping with the pattern in the DCA ordination, the portion of ordination space occupied by herbs dispersed by animal ingestion and by ants is widely separated from the portion occupied by herbs dispersed by wind, prolonged dormancy, unassisted means, animal adhesion and multiple modes. Herbs dispersed by animal ingestion and by animal conveyance were most frequent in quadrats with a closed canopy, mesic soils, and large trees, whereas, herbs dispersed by wind and by prolonged dormancy were most frequent in disturbed habitats and quadrats with an open canopy. Herbs dispersed by unassisted means, adhesion to animals, and by multiple modes, were most frequent in quadrats with a high number of young stems, high soil organic matter, Gleysolic soils, and calcareous outwash or lacustrine parent materials.

Taken together, these results reveal that dispersal modes of herbs of the forest understory tend to be associated with particular environmental states at the quadrat scale. The mechanisms contributing


animal adhesion, C = animal conveyance (ants), I = animal ingestion, M = mechanical expulsion, S = prolonged dormancy in soil, U = unassisted means, W = wind; Environmental variables; CANCLO = % canopy closure, CT = forest cover type, DIST = disturbance patch area (ha), 0-4 cm = # tree stems 0-4 cm dbh, 4-10 cm = # tree stems 4-10 cm dbh, 10-30 cm = # tree stems 10-30 cm dbh, #TS(closed and open canopy gaps, lanes, regenerating fields), MOIST = soil moisture, OMH = open microhabitats, pH = soil pH, PA = Figure 3.6. Distribution of dispersal modes of herbs in relation to environmental variables (CCA). Legend: Dispersal modes: A = = # tree species. Environmental variables explain 51.6% of dispersion of dispersal modes in ordination space.

to this pattern require further analysis since the observed associations may be caused by differential dispersal, germination and establishment success. Nevertheless, the finding that all modes were present in each portion of the gradients examined suggests that the distribution of herbs in these forests has not been constrained by dispersal, or at least, has not been constrained absolutely.

3.3.2.3 Microhabitats

Environmental conditions within quadrats(10m x 10m) were rarely uniform. The relations between dispersal mode and environmental states were reassessed at the microhabitat scale in order that patterns may be examined in relation to more uniform states of forest cover, moisture, and disturbance. Definitions and descriptions of examined microhabitats were presented in Chapter 2.

Pattern was investigated in relation to three contexts: difference in the total number of herbs recorded each microhabitat (Table 3.13): difference in the proportion of herbs in contrasting microhabitats, paired samples (Table 3.14); and, difference in the proportion of herbs in contrasting microhabitats, independent samples (Table 3.15). The first context clarified the degree to which differences in the proportions of herbs in contrasting habitats were due to differences in the number of herbs present. The paired sample analysis clarified the degree to which dispersal modes were associated with particular environmental states when dispersal is known to be non-limiting. The independent sample analysis extends the latter analysis to cases where the contribution of dispersal is not known.

Number of Herbs within Microhabitats

Typically, the number of herbs dispersed by a given mode did not vary among microhabitats (Table 3.13). The exception was dispersal by animal ingestion which was over-represented in closed, dry, disturbed microhabitats (canopy gaps, tip-up mounds, tree pits, logs, stumps, farm lanes/access roads), closed seasonally dry forest floors, closed seasonally moist forest depressions, and, closed raised root mats (p<0.05, chi-square tests of homogeneity by column. *df* 38, Bonferroni correction for 9 modes). Among open microhabitats, the number of herbs dispersed by animal ingestion

AC = animal conveyance (ants), W = wind, PD = prolonged dormancy, ME = mechanical expulsion, \overline{U} = unassisted, MM = multiple modes, VE = vegetative expansion. Chi-square tests of homogeneity by column and row. Dispersal mode in bold when the number of herbs differs from expectation across microhabitats, chi-square tests of homogeneity by column, df 38, Bonferroni correction for 9 mode in specified microhabitat (upper); departure from expectation (lower). Modes: AI = animal ingestion, AA = animal adhesion, modes. Microhabitat marked by asterisk when number of herbs differs by dispersal mode, chi-square tests of homogeneity by row, *d*/ Table 3.13. Number of observed and expected herbs in surveyed microbubitats by dispersal mode. Cell values: # herbs dispersed by 8 (p<0.05*, p<0.01***, p<0.001***); microhabitat in bold when difference from expectation significant after Bonferroni correction for $p<0.001^{***}$. Expected cell value (not shown) = (proportion of herbs in data set dispersed by mode) x (# classified herbs in microhabitat). % SA = % of total area surveyed (19,200 m²). # Herbs – number classified herbs in microhabitat. Closed microhabitats (-cc); open 39 row tests. Cell value marked by asterisk when number of herbs for a given mode differs from expectation at p<0.05*, p<0.01** microhabitats (-oc), N=234 herbs.

Microhabitat						Di	spersal N	Aode			
Variable	% SA	# Herbs	AJ n=17	AA n=31	$AC_{n=31}$	W n≕62	ΡD n−50	ME n-17	U n=69	MM n=39	VI: n≖92
seasonally dry forest Roors/rises-ce*	5.1.3	£tri	16° 16,3	23 +4.7	28° +9.7	26 -9,3	22 -6.5	12 +2.3	35 -4.3	19 6.£-	60 † 7.6
seasonally dry forest Roors/rises-oc	5.2	121	12 +3.2	18 +3,0	19 14,0	32 +2.1	24 -0.1	9 +0.8	31 -1.2	22 +3.2	48 +3.6
seasonally moist forest floors/rises-ce	5.4	115	11 +3.2	15 (0.8	13 -1.2	31 +2.6	20 -2.9	7 -0.8	33 +1.4	14 -3.9	51 +8.8
seasonally moist forest floors/rises-oc	2.2	95	9 +2.6	10 -1.7	11 -0,7	2 8 +4,5	16 -2.9	7 + 1.6	28 (1.9	14 -0.8	46 +11.2
seasonally moist forest depressions-cc	4'I	6L	10° +4.6	10.2 +0.2	======================================	18 -1.5	13 -2.7	5 -0,4	23 +1.3	10 -2.3	36 17.0

Table 3.13. Number of observed and expected herbs (n=234) in surveyed microhabitats by dispersal mode (cont⁷d).

Microllabitat				:		ä	spersal N	Aode			
Variable	s S	# Herbs	AI n=17	AA n=31	AC h=31	W n=62	P1) n=50	MIS NH:	U n=69	MM n=39	VE 1=92
seasonally moist forest depressions-oc	0.2	20	-0.4	6' +3.5	- . .	5 +0.1	4 0.0	9°1+	4 -1.5	4 +0.9	5 -2.3
seasonally wet forest floors/rises-ec*	0.6	50	- 2.4		-5.2	-0.4	10 0.0	יי ב ב	24" +12.3	5 -2.8	۲۱ ٤.۱-
seasonally wet forest depressions-ce	4'I)	99	5 +0.5	7 - 1.2	7 2.1-	20 +3.7	8 -5.1	6 +1.5	22 +3.9	و ٤.۱-	30 +5.8
seasonally wel forest depressions-oc	1.0	74] -0.6	4 1. 0.	- 5	7	5 +0.2	Б 1.1-	10 • 3.4	7 +3.3	7 -1.8
oo-deas	0.9	19	0 <u>.</u>	ся Ц	- 7	÷0.7	2 	2 +0.7	9 +3.8	- 6.5	8 ().[+
ao-dans	0.3	6	-0-	3 +0.7	0 -2.3	5 +0.3	0 -3.8	ے +0.7	9 +3.2	- <u>-</u> 2.0	5 -2.0
seasonally dry gap-cc **		59	12***	6 - .3	6 6	13 6.1-	7 -4. 8	5 +1.0	13 2,6-	5 -4.2	30 18,4
seasonally dry gap-oc	6.9	125	12 +3.5	20 +4.6	21 +5.6	34 14.9	24 -0.9	9 •0.5	30 -4.4	23 13.6	50 F1.2
seasonally moist gap-cc	0.5	38	÷ بن 4	6 +1.7	5 +0.3	10 +0.6	4 -3.6	수 +1.4	10 -0.4	6 +0.1	16 +2.1

171

Table 3,13. Number of observed and expected herbs (n=234) in surveyed microhabitats by dispersal mode (cont³d).

Microhabitat						ī	spersal N	Aode			
Variable	% SA	# Herbs	AI n-17	۸۸ n÷31	AC n=31	W n=62	PD n=50	ME n≓17	09=10 U	MM 11-39	VI: ¤∺92
scasonally moist gap-oc	3.0	66	8 +1.7	-0.5	11 -0.5	28 +5.0	17 -1.5	7 +0.7	26 0.4	15 0.5	46 . +11.9
seasonally wet gap-ue	1:0	24	- 9.11-	+ 1.1.÷	-2.0	7	5 10.2	3 +1.4	10 +3.4	7 +3.3	7 -1.8
mound-cc	3.6	16	13"	9-2.2	20** +8.8	21 -1.5	10 -8.2	9 +3.8	2] -4,0	12 +2,1	44 +9.6
толид-см	0.4	6ŀ	10"" 16.3	7 10.9	9 12.9	6.0+	= <u>+</u>	- -2.3	9 +,t.	10 +2.4	24 +6.0
seasonally dry pit-ce	5	17	9** 16.5	5 +0.1	11" 16:1	s 1	2° -5.4	3 +0.5	5.2 -5.2		21° +7.4
seasonally dry pit-oc	0.1	15	-5.0 +2.0	±2+ 	و ۱.۱۰	4 +0.3	3 0.0	1 1	1 -3,1	3 +0.7	6 +0.5
seasonally moist pit-cc	0.3	5	0 -0.3	-1 -+ +		-0.2	0.1- 0	 +0.7	0 -1.4	0 -0.2	3 + I.2
seasonally wet pit-ce	0.2	=	2 +1.3	- 10.6	3 +1.6	1.7	 -1.2	2 +1,3	2 -1.0	2 +0,2	4 0.0
seasonally wet pit-ne	0.1	_	0 -0.1	0 ^{.1}	0 -0.1	0 -0.2	0.2	6'0+ .1	0.0-	0 -0.5	0 1'0

'Table 3.13. Number of observed and expected herbs (n=234) in surveyed microhabitats by dispersal mode (cont'd).

Microhabitut						Ž	spersal A				
Variable	% SA	ji Herbs	AI 1-17	AA AA I£=0	AC n=31	W n=62	PD n=50	ME: n=17	1) 11=69	MM n=39	VI: 11=92
log-cc	0.5	61	10" 15.9	8 10.5	10 12.5	12 -3.1	10	6.1+ 9	15 •1.8	10 +0.5	30 +7.6
ង្រុ-០០	0.2	38	<u>יד</u> ד ד. ד	7 +2.3	5 +0.3	12 +2.6	5 -2.6	· 4 + .∔	8 -2.4	8 +2.1	20 +6,1
slump-cc	0.2	28	6" †4.1	3 -0.5	4 +0.5	6 -0.9	3 -2.6	4 +2,1	7 -0,7	5 +0.6	16 +5.7
stump-oc	0.3	23	ي +.[+	2 -0.8	2 -0.8	9 £.£1	2 -2.6	2 10.4	5. 5	2 -1.6	12 +3,6
raised root mat-co	5	73	10. 15.1	071) (1	8 1,0,1-	15 -3.0	8 -6.5	5 • 0.1	23 1.61	9 -2.3	30 13.2
raised root intit-oe	9.0	57	\$ +2.1	0.1 1.0	7 0.0	9 -5.1	12 40.6	5 +1.1	12 -3,7	12 +3.1	25 +4.1
stune-cc	0.1	Ξ	1 10.3	- 1 -0.4	3. +1,6	: +1.3	0 -2.2	0 -0.7	1 -2.0	0 -1.7	5 +1.0
lane/rond-ce*	2.3	<u>50</u>	9 15.6	6 -0.2	9 +2.8	8 -4.4	4 -6.0	5 1.1	16 +2.3	4 -3.8	25 +6.7
lane/road-oc	L	123	6 -2.3	18 +2,8	15 -0,2	36 + 5.6	36° +11.5	8 -0.3	33 -0.8	26 +6.9	47 +1.9

173

Table 3.13. Number of observed and expected herbs (n=234) in surveyed microhabitats by dispersal mode (cont¹d).

Microhabitat						Ä	spersal N	lode			
Variable	% SA	# Herbs	AI 0≕17	AA n=31	AC n=31	W n=62	РI) 0-50	ME n=17	U n~69	MM n=39	VE n=92
ditch-ce	0.3	ए रु	3 0.0	c+ Å	6 10.6	14 13.1	6 -2.8	-2.0	6'E+	4 -7.8	22 +5.9
ditch-oc	0.1	20	- 0-	- I.5	1 -1.5	8 +3.1	3 -1.0	1 -0.4	6 +0.5	-2.1	11 +3.7
regenerating field-cc	5.1	72	3 6'1'	7 -1.9	6'1- L	27° +9.2	19 +4.7	4 - 0. 9	20 -0.2	14 +2.8	30 +4.6
regenerating field-oc*	1.3	86		9 9,1-	5 -5.6	33°°	24 +6,9	8 +2.2	24 +0.4	17 +3.6	32 +0.5
riparian meadow-oe	0.7	1 <u>5</u>	8'1-	5 -0.2	د 11-	14 13.6	9 10.6	3 10.1	13 (3.5	6 -0.5	-0.4 -0.4
riparian marsh-oc	0.4	26	0 -1.7	6 +2.8	- 2:2-	10 +3.6	ب 1-	2 +0,2	7 -0.1	4 .0.0	9 -0.5
ripariau thicket-oc	D.I	3	0 -1.6	7.4.4	- 1 .	8 +2.3	ی 1.6	 9.1-	6 -0.3	3 -1.6	4 4 4 4 2

departed from expectation only in dry tree pits.

Within microhabitats, the number of herbs rarely varied with mode of dispersal. The exceptions were seasonally dry or wet forest floors under closed canopies, disturbed microhabitats under closed canopies (canopy gaps with seasonally dry soils, tree pits with seasonally dry soils, stumps, lanes/access roads), and open regenerating fields (p<0.05, chi-square tests of homogeneity by row, *df* 8, rarely significant after Bonferroni correction for 39 row tests). The modes typically associated with this pattern were dispersal by animal ingestion and by ants.

The microbabitats in which modes of dispersal were significantly over (under) represented varied by mode. Dispersal by animal ingestion was over-represented in eleven microbabitats (see table), whereas dispersal by animal adhesion was over-represented only in open, seasonally moist, forest depressions. Dispersal by ants was over-represented on closed seasonally dry forest floors, and in closed or open seasonally moist depressions, whereas, dispersal by wind was over-represented in both closed and open regenerating fields. Dispersal by prolonged dormancy in the soil was overrepresented on open lanes/access roads and under-represented in dry closed tree pits. Dispersal by unassisted means was over-represented on closed seasonally wet floors, whereas, dispersal by vegetative expansion was over-represented in open seasonally moist gaps and in closed seasonally dry tree pits. Dispersal by mechanical expulsion and by multiple modes did not differ from expectation in any microhabitat.

All modes of dispersal were represented in most microhabitats. The exceptions were seven rare microhabitats with seasonally saturated or shallow soils (number of missing modes in brackets): closed seeps (1), open seeps (2), closed seasonally moist tree pits (4), open seasonally wet tree pits (8), boulders (3), open riparian matshes (1), and open riparian thickets (1). The modes most frequently absent from these habitats were dispersal by animal ingestion (5 habitats), prolonged dormancy (4 habitats) and multiple modes (3 habitats). The modes least absent from these habitats were dispersal by animal adhesion, wind, mechanical expulsion and vegetative expansion. These modes were absent only from open seasonally wet tree pits.

The trends in Table 3.13 are broadly consistent with the trends reported in Section 3.3.2.2.

Proportion of Herbs in Contrasting Microhabitats within the Same Quadrat

Within quadrats, the dispersal modes of herbs were often associated with particular microhabitats (Wilcoxon signed rank tests by cell, paired comparisons) (Table 3.14). This suggests that herb assemblages in these samples were affected more by factors governing germination, establishment, and persistence than by dispersal, *per se*, since herbs dispersed by each mode were already present in each quadrat.

The following patterns were observed:

i) Under closed canopy conditions, the proportion of herbs dispersed by animal ingestion, mechanical expulsion and vegetative expansion was significantly higher on disturbance features with organic substrates such as logs, stumps and raised root mats than on the adjacent dry forest floor. The proportion of herbs dispersed by these modes on mineral substrates, such as pit/mound complexes, and on features created by anthropogenic disturbance, such as lanes, access roads, and regenerating fields, did not differ from the proportion on the adjacent forest floor.

ii) Under closed canopy conditions, the proportion of herbs dispersed by animal adhesion, ants, and unassisted means was significantly lower on natural disturbance features with organic substrates than on the adjacent forest floor. As in the previous case, the proportion of herbs dispersed by these modes on mineral substrates, such as pit/mound complexes, and on features created by anthropogenic disturbance, did not differ (or rarely differed) from the proportion on the adjacent floor. The proportion of herbs dispersed by wind, prolonged dormancy and multiple modes did not differ from the adjacent forest floor under any conditions.

iii) The proportion of herbs recorded under open canopy conditions did not differ from the proportion recorded in the same habitat under closed canopy conditions.

Table 3.14. Difference in mean percent of herbs (n=234) dispersed by various modes in contrasting microhabitats within 10m x 10m
padrats. Wilcoxon signed-ranks tests, by cell, paired comparisons. AI = animal ingestion, AA = animal adhesion, A = ant, W = wind
3 D = prolonged dormancy, MF = mechanical expulsion, U = unassisted, MM = multiple modes, VE $-$ vegetative expansion. Difference
- lower case habitat - upper case habitat. Values marked with an asterisk when difference significant: 1 * p<0.05, ** p<0.01, ***
>(0.01; value in bold when difference significant after Bouterroni correction for n=9 tests per row. Quadrats with difference = "0"
excluded from analysis. N=192 10m x 10m quadrats; n = maximum # quadrats in analysis. Suffixes: "-cc" = closed curropy; "-oc"
= open canopy.

						ispersal Mc	ode			
Microhabitat ((=u)	14	٧V	V	×	CI4	Ш	D	MM	VE
DRY FOREST FLOOR/F	RISE-C	c.								
moist forest depressions-cc	<u>~</u>	-3.8	+5.7	-10'7	+4.8	-2.4	-2,8	+4.2	-3.5	-4.3
wet forest depressions-ec	11	-2.5	-1.5	-17.1	+14.5	+6.7	+0,I	-0.4	-1-9.2	-11.7
NATURAL DISTURBA	NCE-C	c.							_ 3	
natural disturbance-oc	8	4.9	ć.l+	+7,7	+3,9	-5.0	-5.5	+3.8	+5.2	-4.0
MOUND-CC										
pit-cc	36	+8.7	13.6	F1.1	-10.7	-11.0	+7.6	5.0- :	0.0-	+5,9
PIT/MOUND-CC										
log/stump/raised root mat-cc	16	1 8.6	-9.5	-11.3*	44.9	+10.3	+13.6	£~2-	+9.8	ŀ6.6
LOG/STUMP-CC						-				
raised root mat-cc	20	-1.1	+5.0	+2.6	+3.6	-13	-15.3*	+6.0	-1.4	-10.4*

Table 3.14. Difference in mean percent of herbs (n=234) dispersed by various modes in contrasting microhabitats (cont²d).

iv) The proportion of herbs dispersed by ants was significantly lower in moist and wet depressions under closed canopy conditions than on the adjacent dry forest floor.

v) The proportion of herbs dispersed by animal adhesion, ants, wind and prolonged dormancy on disturbance features varied with the feature and the nature of the substrate.

Proportion of Herbs in Contrasting Microhabitats, Independent Samples

The number of contrasting conditions that could be evaluated in paired samples was constrained by sample size and within-quadrat habitat combinations. By relaxing the requirement that contrasting habitats be present in the same quadrat, the analysis can be extended to the more general case; does the proportion of herbs dispersed by various modes change under contrasting habitat conditions. The greater generality achieved by this approach is offset by greater uncertainty regarding causal mechanism, since differences among sites cannot be controlled to the same degree, and, the degree to which dispersal is limiting is not known.

In general, the proportion of herbs dispersed by a given mode varied in response to contrasting habitat conditions (Wilcoxon rank sum tests, independent samples)(Table 3.15). Habitat contrasts that produced the most significant differences were the following (number of modes where the difference was significant in brackets): closed vs open dry forest floors (6); closed vs open forest floors/depressions (6), closed forest floors/depressions vs open human disturbance features (6), closed forest floors/depressions vs open natural disturbance features (5), dry vs moist closed forest floors (4), and, natural vs human disturbance features under an open forest canopy (4). Habitat contrasts in which the proportion of herbs did not vary for any mode were moist floors versus moist depressions, and, closed versus open pit/mound complexes. Modes of dispersal that were most sensitive to changes in habitat conditions were animal ingestion, prolonged dormancy in the soil, ants, and wind (difference in response was significant in >50% of habitat contrasts). Modes that were least sensitive to changes in habitat conditions were dispersal by unassisted means and by multiple modes (difference in response was significant in <20% of habitat contrasts).

Table 3.15. Mean percent of herbs (n=234) dispersed by various modes in contrasting microhabitats within 10m x 10m quadrats. Wilcoxon rank sum tests, independent samples, AI = animal ingestion, AA = animal adhesion, A= ant, W = wind, PD = prolonged dormancy, ME = mechanical expulsion, U = unassisted, MM = multiple modes, VE = vegetative expansion. Quadrats with mode in each habitat excluded from analysis. Mean % for microhabitat presented for summary purposes. Larger value marked by an asterisk when difference between habitats significant: |t| * p < 0.05, ** p < 0.01, *** p < 0.001; larger value in bold when difference significant after Bonferroni correction for n=9 tests per comparison. N = 192 quadrats; n = maximum number of quadrats in analysis. Suffixes: "-cc" = closed canopy; "-oc" = open canopy.

					D	ispersal M	lode			·
Microhabitat	(n=)	AI	AA	A	w	PD	ME	U	ММ	VE
dry forest floor/rise-cc	126	26.3***	12.4	27.9	17.9	11.2	9,2	19.7	10.8	60.4*
dry forest floor/rise-oc	12	13.4	16,9'	22,0	25.1	20,6"	9,5	18.4	20.5"	49,2
		_	-							
dry forest floor/rise-ee	138	26.4***	12,6	28,4****	18,2	11.0	9.3	19,7	10,7	60.4***
moist forest floor/rise-ce	18	15.7	15.2	11.2	25.4"	14.8	13.5	21.6	13.5*	47.6
moist forest floor/rise-cc	18	17.6	16.2	13.5	26.2	16.2	11.6	23.4	13.2	45.0
moist forest depressions-cc	19	15.6	19.0	16.4	33.7	17.0	16.4	25.4	15.9	47.4
			-		-					-
all forest floors/rises/ depressions-cc	143	27.1****	13.5	25.5	21.0	12.0	9.7	21.1	12.2	58.1"
all forest floors/rises/depressions-oc	18	12.1	17,4*	19,7	26,7*	18.4*	10.7	19.6	19.2**	47.6

Table 3.15. Mean per cent of herbs	disper	sed by var	ious mode	s in contr	asting mic	rohabitats ispersal Mi	within 10	m x 10m	quadrats («	:ont'd).
Microhabitat	(=u)	R	¥	<	×	۵ı	ME	=	MM	VE
all forest floors/rises/depressions-cc	147	25.0	13.0	26.4'	19.7	11.7	9.7	21.0	12.0	58.6
all natural disturbance features-oc	12	23.3	24.9	17.9	34.0*	19.7*	22.5	20.1	22.2'	61.3
all forest floors/rises/depressions-cc	157	25.1"	13.3	26.6"	20.1	11.7	9.9'	20.7	12.0	57.9
all human disturbance features-oc	2	8.3	16.3	15.0	42.3***	28.5	6.8	6.61	21.9**	50.5
all natural disturbance features-oc	6	21.6"	51 F	25.2	29.9	20.2	24.5'	21.6	22.7	55.8 -
all human disturbance features-oc	2	5.6	15.2	6.8	39.1	30.1	8.5	21.3	22.7	41.6
natural disturbance-cc	96	33.2	16.8	28.7	26.8	15.9	19.7	20.1	2.71	66.2
natural disturbance-oc	15	20.3	26.7	23.4	31.7	20.7	25.0	19.3	23.2	62.5
	Į									
pit/mound-ce	59	32.2	15.7	33.5	26.9	15.6	12.8	5.61	16.5	66.S
pit/mound-oc	5	32.0	18.4	29.1	25.6	22.6	14.3	10.1	20.8	70.1
pit/mound-ce	50	30.6	15.1	34.6	26.9	14.5	13.1	20.0	15.6	66.1
log/stump-cc	37	42.9	21.4	21.5	30.5	22.3"	35.3**	29.2'	25.5	66.8

Table 3.15. Mean per cent of herbs	indeline a	sed by var	rious mod	es in contr	asting mic	rohabitats	within 10.	m x 10m	quadrats (cont'd).
					D	ispersal M	ode 			
Microhabilat	(=u)	١٧	VV	۷	w	Δſ	ME	n	MM	VE
pit/mound-ce	53	32.4	15.2	34.5"	27.9	16.0	13.4	19.1	16.8	66.9"
raised root mat-cc	61	27.1	8'61	16,9	25.5	10.4	16.5	23.2	19.7	652.6
log/stump-cc	52	0.61	21.3	23.5	34.1	27.2	35.4	29.2	30.0	6.9
log/stump-oc	8	12.8	29.2	23.3	30,0	13.1	31.3	22.2	20.6	58.8
lane/road-cc	6	27.2"	13.7	30.0	16.8	8.5	9.5	22.2	11.0	59.2
lane/road-oc	6	6.8	191	17.8	35.5"	32.3'	7.0	22.3	24.0	-18.5
				-						
lane/road-cc	10	26.1	14.8	29.1"	: 16.I	9.0	7.4	23.4	12.6	58.6
regenerating field-co	5	8.4	9.7	8.5	42.3"	30.2'	7.7	24.0	20.7	47.4
lane/rond-oc	6	8.7	17.7	23.2	30.9	32.1	7.4	6'61	25.1	51.6
regenerating field-oc	5	6.J	6.11	8,0	45.6"	28.0	8.7	ب ۲ (م	19.3	5.44

Herbs dispersed by animal ingestion were most strongly associated with closed. dry forest floors: logs and stumps under closed canopies; and with natural rather human disturbance features (p<0.05. Bonferroni correction for 9 tests per comparison). In contrast, herbs dispersed by animal adhesion were most strongly associated with open natural disturbance features, whereas, herbs dispersed by ants were most strongly associated with closed forest floors, closed pit/mound complexes, lanes and access roads under closed canopies, and, with natural rather than human disturbance features. Herbs dispersed by the wind were most strongly associated with open human disturbance features such as lanes, access roads and regenerating fields, whereas, herbs dispersed by prolonged dormancy in the soil were most strongly associated with human disturbance features under an open canopy and with closed logs and stumps. Herbs dispersed by mechanical expulsion and by unassisted means were most strongly associated with closed logs and stumps. Herbs dispersed by mechanical expulsion and by multiple modes were most strongly associated with open forest floors, human disturbance features under open canopy and with closed logs and stumps. In contrast, herbs dispersed by vegetative expansion were most strongly associated with open forest floors, human disturbance features under open canopies, and closed logs and stumps. In contrast, herbs dispersed by vegetative expansion were most strongly associated with open forest floors, human disturbance features under open canopies, and closed logs and stumps. In contrast, herbs dispersed by vegetative expansion were most strongly associated with open forest floors, human disturbance features under open canopies, and closed logs and stumps. In contrast, herbs dispersed by vegetative expansion were most strongly associated with open forest floors, human disturbance features under open canopies.

The patterns of association revealed by independent and paired samples were similar in cases where direct or indirect comparison was possible. The sole exception was the affinity of selected modes for open habitats. In contrast to the pattern in paired samples, the proportion of herbs dispersed by animal adhesion, wind, prolonged dormancy, and multiple modes was often significantly higher under open than closed canopy conditions.

The latter result is intuitively more satisfying given that many of these herbs have affinities for open habitats. The difference in results is primarily in the relative strength of the outcome, however, since the proportion of herbs in open habitats was also greater in the paired samples analysis. If the positive, but non-significant, difference in richness in the paired samples simply reflects the low probability that a new taxon may land or germinate in a canopy opening before it closes, then the difference in results may be due to the larger sample size, and the different unit of measure, in the analysis of independent samples. The greater number of closed forests with low species richness in the independent samples analysis may also have strengthened the statistical relationship in that

analysis.

3.3.3 Pattern in Relation to Abundance Variables

The tendency for dispersal modes to be associated with particular states of plant abundance was examined in relation to frequency class, cover class and species richness class.

3.3.3.1 Frequency Class

The frequency class of herbs rarely varied with dispersal mode (Table 3.16). The sole exception were herbs dispersed by animal ingestion which were over-represented in the high frequency class (\geq 25 quadrats) (p<0.001, chi-square tests homogeneity by row and cell, Bonferroni correction for 10 row tests and 30 cell tests). The tendency for herbs dispersed by animal ingestion to occur with high frequency is consistent with the mobility and variety of animal taxa that consume fleshy fruits, and, with the capacity of these herbs to germinate and persist under closed canopy conditions.

Surprisingly, the frequency of wind dispersed herbs did not differ from expectation. Given the mobility of wind dispersed diaspores, one might have expected wind dispersed herbs to be over-represented in the high frequency class. The capacity for these taxa to colonize and persist in closed habitats is limited, however, and these factors may have offset the extended dispersal reach of these species.

3.3.3.2 Cover Class

The cover of a given herb within a 10m x 10m quadrat often varied with the mode of dispersal (chisquare tests of homogeneity by row and column) (Table 3.17). Herbs dispersed by the wind were over-represented in cover class 1 whereas herbs dispersed by animal ingestion, and by vegetative expansion, were over-represented in cover classes 4 and 5. Herbs dispersed by prolonged dormancy were over-represented in cover class 1 and under-represented in cover classes 2 and 4. Herbs dispersed by ants were over-represented in cover class 5 (chi-square tests of homogeneity by cell). Table 3.16. Dispersal modes of herbs (n=252) by frequency class. Cell values are the number of observed and expected herbs dispersed by a given mode at the designated frequency. Expected = (# herbs dispersed by mode $\div 252$) x (# herbs in the frequency class). Dispersal mode and observed value marked by asterisk(s) when departure from expectation significant: *p<0.05, **p<0.01, ***p<0.001, chi-square tests of homogeneity by row (*df2*) and cell (*df*1), respectively; dispersal mode and observed value in bold when departure from expectation significant after Bonferroni correction for 10 row tests and 30 cell tests, respectively. N=192 10m x10 quadrats.

Dispersal Mo	de			Frequen	icy Class		
Mode	# Herbs	Hi ≥25 qı n=49 Obs.	igh uadrats herbs Exp.	Intern 3-24 q n=127 Obs.	nediate uadrats herbs Exp.	L 1-2 q n=76 Obs.	ow uadrats i herbs Exp.
Animal	8[23	15.8	38	40.8	20	24.4
Ingestion	17	10***	3.3	4	8.6	3	5.1
Adhesion	34	5	6.6	20	[7.1	9	10.3
Conveyance (ant)	33	8	6.4	16	16.6	9	10.0
Wind	64	11	12.4	35	32.3	18	19.3
Prolonged Dormancy	50	5	9,7	31	25.2	14	15.1
Mechanical Expulsion	18	3	3.5	8	9.1	7	5.4
Unassisted	78	12	15.2	39	39.3	27	23.5
Multiple Modes	39	5	7.6	24	19.7	10	11.8
Vegetative Expansion	95	23	18.5	49	47.9	23	28.7

Table 3.17. Dispersal modes of herbs (n=252) by cover class. Legend: # = number of herbs dispersed by mode; Trace = 1-5 individuals or small clumps; Obs. = # species observed in 10m x 10m quadrats; Exp. = # species expected in 10m x 10m quadrats [expected number = (# herbs dispersed by mode ± 252) X (# herbs in cover class)]; chi-square tests of homogeneity by column (d/8), row (d/6) and hy cell (d/1): * p<0.05, ** p<0.01, *** p<0.001, no correction for number of tests; dispersal mode and cover class in bold when p<0.05 after Bonferroni correction for number of tests.

								Co	ver Cla	SS							
								+		ι γη		9		-		∞	
Dispersal Mode	*	Tr [=]	ace) (95	Σ.F	(%) 209	(1-1 n=1	5%) [33	(5-15	5%) 16	(15-2: n≓ṫ	5%) 5	(25-50 n≓9	(%	(50-7 <u>:</u> 11=]	3%)	(75-10 1-0	(%) (
		Ohs	Exp	Obs	Exp	Obs	Exp	Obs	Exp	Obs	Exp	Obs E	dx:	Obs 1	цхэ	Obs	Exp
Animal Ingestion**	17	51	13.2	١	14,1	13	9'D	8	3.1	3,	1,0	2	0.6	-	0.2	0	•
Animal Adhesion	34	¥5	26,3	Ę	28.2	Ы	17,9	۴	6.2		2.0		<u>ci</u>		0.4	0	e
Animal Conveyance (Ants)	33	F.	25.5	31	F'12	18	17.4	8	6.D	5	0.5	e	<u>_</u>	_	0.4	0	۰
Mind	64	60	31.0	<u>56</u>	53.1	37	33.8	13	11.7	_	3.8	_	2.3	0	0.8	0	0
Protonged Dormancy	50	22"	10.1	<u>34</u> "	41.5	17	26.4	÷1	9,1	_	3.0	0	1,8	0	0.6	0	0
Mechanical Expulsion	18	14	13.9	4	14.9	6	9.5	ę	3.3	3	1.1	6 1	0.6		0.2	0	•
Unassisted	78	51	40.2	64	64.7	43	41.2	10	14.2	-	4,ń	0	2.8	0	0.0	0	-
Multiple Modes	39	33	30.2	30	32.3	17	20.6	4	7.]	0	2.3	0	7.	0	0.5	0	•
Vegelative Expansion	95	75	73.5	5	78.8	58	50.1	30.	17.3	.tr	5.7	1	3.4	ر	I .1	0	0

The tendency for herbs dispersed by animal ingestion, ants and vegetative expansion to be overrepresented in the higher cover classes is consistent with their greater tolerance for deep shade and capacity for clonal growth. The tendency for herbs dispersed by wind and prolonged dormancy to be over-represented in cover class 1 is consistent with their mobility in space, or time, and their affinity for high light environments. Both recruitment and extinction processes may contribute to the tendency for the latter species to be found as isolated individuals or in small clumps.

3.3.3.3 Richness Class

The mean proportion of herbs in a 10m x 10m quadrat dispersed by a given mode varied with the number of taxa in the quadrat (Table 3.18). The proportion of herbs dispersed by animal ingestion and by ants was highest in quadrats with low species richness, whereas, the proportion dispersed by animal adhesion, wind, prolonged dormancy, mechanical expulsion, unassisted means, and multiple modes was highest in quadrats with high richness(Wilcoxon rank sum tests by row). The proportion of herbs dispersed by vegetative expansion did not vary with the species richness of the quadrat.

This pattern is broadly consistent with the tendency for taxa dispersed by these modes to be associated with closed or open habitats (Tables 3.9 and 3.12). Closed forest habitats tend to be comparatively species poor since taxa with affinity for open habitats cannot germinate or persist under a closed canopy. Open forest habitats, on the other hand, provide opportunities for colonization by taxa with affinities for high light conditions, and, retain taxa with affinity for moderate and low light conditions (since open canopy conditions are short-lived) and thus tend to be comparatively species rich.

3.3.4 Pattern in Relation to Spatial Scale

The tendency for the dispersal modes of herbs to vary in relation to the spatial scale of their surroundings was examined in relation to patch size and patch isolation.

Table 3.18. Dispersal modes of herbs (n=252) by species richness class. Cell value: mean percent of herbs in 10m x 10m quadrats dispersed by mode. Dispersal mode and class with the highest mean ranked sum marked by asterisk(s) when differences among richness classes significant: *p<0.05, **p<0.01, ***p<0.001, ***p<0.001, Wilcoxon rank sum tests by row, df 2); dispersal mode and cell value in bold when differences significant after Bonferroni correction for 10 tests, N=192 quadrats.

Dispersal Mode		Richness Class	
Mode	High ≥56 taxa/quadrat n=49quadrats	Intermediate 29-55 taxa/quadrat n=93quadrats	Low ⊴28 taxa/quadrat n=50quadrats
Anîmal	21.9	26.7	32.5****
Ingestion	6.6	9.4	14.5***
Adhesion	6.7****	6.6	3.1
Conveyance (ant)	8.6	10.7	14.9****
Wind	15.4****	9.9	5.6
Prolonged Dormancy	10.1****	4.4	2.0
Mechanical Expulsion	4.0****	3.6	1.7
Unassisted	14.9****	10.3	7.1
Multiple Modes	8.6****	4.8	2.1
Vegetative Expansion	25.7	25.8	29.6

3.3.4.1 Patch Size

The proportion of herbs dispersed by a given mode often varied in relation to patch size (Wilcoxon rank sum tests) (Table 3.19). The proportion of herbs dispersed by animals, ants, prolonged dormancy, and multiple modes was highest in patches of intermediate size, whereas, the proportion of herbs dispersed by mechanical expulsion, unassisted means, and vegetative expansion was highest in small patches. The proportion of herbs dispersed by animal ingestion, animal adhesion, and wind did not vary with patch size.

The causal mechanisms for these patterns are not immediately apparent. However, the pattern is consistent with the tendency for small patches to have wet depressions (Table 2.12), and, for patches of intermediate size to be rich in sugar maple (Wilcoxon rank sum test, not shown). The former may explain the greater frequency of herbs dispersed by unassisted means, mechanical expulsion and prolonged dormancy in small patches, given their apparent affinity for moist and wet soils, whereas, the latter may explain the greater frequency of herbs dispersed by ants patches of intermediate size, given their apparent affinity for sugar maple stands on mesic soils (Tables 2.12 and 3.12).

Only the most mobile modes did not vary with patch size: animal ingestion, animal adhesion, and wind. The degree to which this pattern is due to dispersal is unclear since the environmental states most strongly associated with these modes did not vary with patch size: canopy closure, mesic soils, high soil organic matter, cover type 5, and disturbance (Tables 2.12 and 3.12).

3.3.4.2 Patch Isolation

The dispersal modes of herbs rarely varied with patch isolation (Wilcoxon rank sum tests) (Table 3.20). The exceptions were dispersal by unassisted means and by vegetative expansion. The former were more abundant in quadrats in patches of intermediate isolation whereas the latter were more abundant in quadrats in patches of low isolation. These differences were not significant after Bonferroni correction for the number of tests. The mechanisms responsible for this result are not apparent.

Table 3.19. Dispersal modes of herbs (n=252) by patch size class. Cell value: mean percent of herbs in 10m x 10m quadrats dispersed by mode. Dispersal mode, and class with the highest mean ranked sum, marked by asterisk(s) when differences among size classes significant: p<0.05, p<0.01, p<0.001, p<0.0001. Wilcoxon rank sum tests by row: dispersal mode and cell value in bold when differences significant after Bonferroni correction for 10 tests. N=192 quadrats.

Dispersal Mode	Patch Size Class			
Mode	Large > 122 ha n=48 quadrats	Intermediate 43-121 ha n=88 quadrats	Small ≤ 42 ba n=56 quadrats	
Animal**	22.8	29.4"	26.9	
Ingestion	10.2	10.5	9.1	
Adhesion	5.7	5.7	5.9	
Conveyance (ant)	7.0	13.2***	11.9	
Wind	8.5	10.8	11.1	
Prolonged Dormancy*	3.4	5.9"	5.9	
Mechanical Expulsion	3.4	2.4	4.4***	
Unassisted	8.3	10.4	13.0****	
Multiple Modes*	3.4	5.8	5.5	
Vegetative Expansion*	24.6	27.3	27.7*	

Table 3.20. Dispersal modes of herbs (n=252) by isolation class. High: mean distance to nearest 8 woodlots in 45° sectors \ge 477m; Intermediate: mean distance 233-476 m: Low: mean distance \le 232 m. Cell value: mean percent of herbs in 10m x 10m quadrats dispersed by mode. Dispersal mode, and class with the highest mean ranked sum, marked by asterisk(s) when differences among richness classes significant: *p<0.05, **p<0.01, ***p<0.001, **** p<0.0001, Wilcoxon rank sum tests by row: differences among classes not significant after Bonferroni correction for 10 tests. N=192 quadrats.

Dispersal Mode	Isolation Class			
Mode	High ≥477 m n≕48quadrats	Intermediate 233-476 m n=96quadrats	Low ≤232 m n=48quadrats	
Animal	56.1	51.8	55.7	
Ingestion	20.7	21.2	22.3	
Adhesion	12.0	10.0	9,8	
Conveyance (ant)	23.4	20.6	23.7	
Wind	18.9	19.3	20.1	
Prolonged Dormancy	11.4	9.1	8.1	
Mechanical Expulsion	7.0	5.7	6.1	
Unassisted*	18.6	22.0*	18.2	
Multiple Modes	11.5	8.2	8.6	
Vegetative Expansion**	48.0	54.0	60.3**	

3.4 Discussion

Limitations of the Methodology

The composition and relative abundance of herbs in a given assemblage may be influenced by the composition, size and frequency of the seed rain: differential germination and establishment success: competitive interactions, and other factors. In principle, the distribution of plants will resemble the distribution of diaspores only when the probability of recruitment and persistence are independent of where a seed may land. In practice, this probability is not independent, and determination of the degree to which the composition of an assemblage has been influenced by dispersal limitation, site limitation, and persistence, is a complex undertaking. Inferences with respect to causal mechanism that may be drawn from pattern in the attributes and distribution of established plants, therefore, are limited.

In this study, the composition of an assemblage was deemed to have been influenced by dispersal if: i) the proportion of herbs dispersed by a given mode was significantly greater than its proportion in the sampled flora; and, ii) there was evidence of directed dispersal; or, iii) there was reason to conclude that the composition of the assemblage was not the result of differential germination. establishment, or persistence. This test of causality limits the accepted cases of influence to those in which seeds are dispersed beyond the immediate vicinity of the maternal plant, and, seedlings aggregate in sufficient numbers to be detected in a chi-square test of homogeneity or a Wilcoxon rank sum test. Seeds that are dispersed close to the maternal plant will be excluded from consideration since dispersal at this spatial scale is non-limiting and pattern in the established vegetation will be due to differential germination, establishment or persistence. Seeds that are dispersed at random beyond the immediate vicinity of the maternal plant will often be excluded since they will fail to aggregate in sufficient numbers to be detected in statistical tests.

Taken together, these conditions remove most seeds from consideration when evaluating the potential contribution of dispersal to the distribution and composition of herb assemblages in the forest understory (Portnoy and Willson 1993).

Has Dispersal Mode been a Limiting Factor in Sampled Herb Assemblages?

One inference that may be drawn, when modes of dispersal are consistently present in the sampled vegetation, is that the assemblage has not been constrained by the mode of dispersal, or at least, has not been constrained absolutely. This inference, while necessarily true, is not trivial when interpreted in the context of reported differences in the maximum reach of dispersal modes (Appendix 11) and concern that dispersal may be limiting in fragmented landscapes (Curtis 1956, Matlack 1994, Kalisz *et al.* 1997, Tilman 1997, Ehrlen and Eriksson 2000).

By this test, the composition of herb assemblages in these forests has not been constrained by the *mode* of dispersal, or at least, has not been constrained absolutely. Each mode of dispersal was represented on each portion of the environmental gradients examined (Table 3.12), and, with few exceptions (7 rare habitats with extreme conditions), in each type of microhabitat (Table 3.13). Although most modes were absent from at least one 10m x 10m quadrat, each mode of dispersal was present in all but one sampled patch (summaries of survey data, not shown). This pattern suggests that at least some seeds from each functional group were able to reach most patches and to germinate in most conditions recorded there.

Contrary to my expectations at the beginning of this study, the number of herbs recorded at low, intermediate and high frequencies was in proportion to their representation in sampled patches (Table 3.16). This suggests that the frequency of herbs in the sampled landscape was not constrained by mode of dispersal. The sole exception were herbs dispersed by animal ingestion, which were significantly over-represented in the high frequency class. Unlike other herbs, herbs dispersed by animal ingestion are both mobile and comparatively shade-tolerant and thus may experience greater colonization success in the forest understory. Taken together, these findings support the inference that mode of dispersal has not been a *limiting* factor in sampled patches.

The inference that mode of dispersal has been a non-limiting factor in the composition of sampled herb assemblages may depend critically on the level of generality in the analysis. Therefore, what may be true for functional groups may not apply to individual populations and species. Similarly,

what may be true for environmental states may not apply to individual localities. Or, at least, not at all temporal and spatial scales (Matlack 1994, Brunet and von Oheimb 1988, Cain *et al.* 1998).

Has the Composition of Herb Assemblages been Mediated by Dispersal?

The tendency for the dispersal modes of herbs to be associated with particular environmental states (Figures 3.1-3.6. Tables 3.12-3.15) is consistent with a dispersal mediated process that has facilitated the colonization of taxa in certain habitats but not others. According to this hypothesis, the observed tendency for herbs dispersed by animal ingestion to be over-represented in former canopy gaps, and on natural disturbance features associated with former canopy gaps (tip-up mounds, tree pits, logs and stumps) (Table 3.13), may be attributed to the habitat preferences and foraging behavior of their dispersal agents and to the methods by which seeds were processed. Similarly, the tendency for herbs dispersed by the wind to be found in open habitats may be attributed to higher wind speeds and the greater dispersal reach of diaspores in open vs closed habitats (Hughes *et al.* 1994). The evidence for such an hypothesis is examined below.

The pattern of colonization by herbs dispersed by fleshy fruits does appear to be consistent with a dispersal mediated process. First, birds that consume fleshy fruits are known to forage preferentially in canopy gaps (Thompson and Willson 1978, Willson *et al.* 1982, Malmborg and Willson 1988) and to regurgitate or excrete the majority of ingested seeds within and near gaps (Hoppes 1988). Moreover, forests and advanced second-growth habitats are known to support a greater diversity and abundance of frugivorous birds than early successional habitats and grasslands (Willson 1986, McDonnell and Stiles 1983). While the foraging patterns of mammals are apparently less defined (Willson and Whelan 1990), the scat of both mammals and birds has often been observed in tree pits (Thompson 1980) and on logs and stumps (Whelan *et al.* 1990, personal observation).

Second, all fleshy-fruited herbs observed in tree pits, or on tip-up mounds, logs and stumps, were also recorded on forest floors (summaries of survey data, not shown). And, all but two fleshy-fruited herbs recorded on forest floors were present in tree pits, or on tip-up mounds, logs and stumps. Thus there is little evidence that the distribution of fleshy-fruited herbs on these features has been

constrained by differential germination or persistence.

Third, inspection of the normative and compositional data (Tables 3.13 and 3.14, respectively) reveals that the over-representation of fleshy-fruited herbs on these features is due to an increase in the number of taxa rather than to an increased proportion of the taxa capable of colonizing these features. This suggests the principal mechanism contributing to this pattern has been dispersal enrichment. Although the direction of dispersal is unknown, the higher frequency and proportions of fleshy-fruited herbs on closed vs open features (Tables 3.13 and 3.15) suggests that the features were colonized over time and primarily from the forest floor. This pattern, and inference, are consistent with the patterns of herb colonization reported for tree pits and logs in three Illinois woodlands (Thompson 1980), and, with the seed rain reported for Trelease Woods, Illinois, (Willson *et al.* 1982, Hoppes 1988).

Interpretation of the colonization pattern of herbs dispersed by ants is less transparent. Although present in every patch, and in virtually every quadrat (189/192), herbs with elaiosomes were consistently over-represented on closed dry forest floors, tip-up mounds and dry tree pits (Table 3.13) and strongly under-represented (in selected habitat contrasts) on closed wet forest floors, logs and stumps, moist or wet depressions, and regenerating fields (Tables 3.13, 3.14, 3.15).

Aspects of this pattern are consistent with a dispersal mediated process. First, ants are known to carry elaiosome-bearing seeds back to the nest where the elaiosome is eaten and the seed is discarded unharmed (e.g. Handel 1978. Beattie and Culver 1981). In forests of the US northeast, nests of these species may be located in the soil (n=13 species), on decaying wood such as logs and stumps (n=3 species), or in both soil and decaying wood (n=2 species) (Appendix13). Most taxa excavate their nests in dry mineral soils (Creighton 1950). However, nests of *Myrmecina americana* and *Prenolepis imparis* are typically found in moist and damp soils, respectively (Creighton 1950), and, nests of *Formica neogagates, F. subsericea, Lasius alienus*, and *Tapinoma sessile* have been observed on organic soils in southern Quebec (Letendre *et al.* 1971). These tendencies are consistent with both the widespread occurrence and observed frequencies of elaiosome-bearing herbs. This

inference, however, depends on the unproved assumption that similar taxa and frequencies occur in sampled forest patches.

Second, ants that forage primarily in forest habitats have been shown to be less tolerant of elevated temperatures than taxa associated with old-field or other open habitats (Lynch 1981). Although the habitat affinities of known myrmecochores have not been widely assessed, most evaluated species show an affinity for forested habitats (Lynch 1981). However, at least one known myrmecochore. *Tapinoma sessile*, has an apparent affinity for recently abandoned fields, and two predominately forest litter species, including one of the most common ant species of the eastern deciduous forest. *Aphaenogaster rudis* (Lynch *et al.* 1980), have been recorded there (Lynch 1981). If these species were to nest in old-fields, then differences in the frequency and habitat affinity of dispersal vectors may contribute to the observed scarcity of elaiosome-bearing herbs in regenerating old-field habitats.

Third, the over-representation of elaiosome-bearing herbs on closed forest floors, tip-up mounds and dry tree pits was due to a greater number of herbs rather than to a higher proportion of taxa that were able to colonize these habitats (Table 3.13). Moreover, 21 of 26 species recorded on closed dry forest floors were also recorded on tip-up mounds and in tree pits; all taxa recorded in the latter habitats were recorded on closed dry forest floors (summaries of survey data, not shown). Taken together, these data provide little evidence of limitation by other factors and suggest that the principal mechanism contributing to this pattern has been dispersal enrichment.

The case for dispersal limitation as the principal mechanism in habitats where elaiosome-bearing herbs were under-represented is less strong. Although somewhat fewer species were recorded in regenerating fields, stumps, and open moist depressions (Table 3.13), the strength of the relationship was due primarily to differences in the proportion of elaiosome-bearing taxa in the contrasting habitat (Tables 3.14, 3.15). This was also the case for closed and open logs where more, rather fewer, species were present (Tables 3.13, 3.14). Taken together, these results suggest that dispersal was not the principal mechanism responsible for the under-representation of elaiosome-bearing herbs in these habitats. However, the under-representation of elaiosome-bearing herbs on closed wet

floors is consistent with a dispersal mediated process, since significantly fewer species were recorded in this habitat (Table 3.13) and comparatively few species of ants nest in such conditions.

The presumption that elaiosome-bearing herbs are dispersed exclusively by ants is not strictly true. Yellow jacket wasps in Oregon state have been observed transporting diaspores of *Trillium ovatum* to their nests (Jules 1996), and in Washington state. Peilmyr (1985) has observed yellow jacket wasps dispersing elaiosome-bearing seeds of *Vancouveria hexandra* (Berberidaceae). Elaiosomes represent a rich source of lipids in the forest understory and it is likely that deer and other animals consume and passively disperse the diaspores of many "ant-dispersed" herbs. Infrequent but chance (or consistent but unrecognized) dispersal by insect and animal vectors may therefore be an important mechanism for longer-distance dispersal of these herbs and may account for their tapid apparent rate of post-glacial migration in eastern North America (Cain *et al.* 1998).

The degree to which dispersal has contributed to patterns of association for other modes is less certain. Herbs dispersed by the wind, for example, were consistently associated with open microhabitats and with human disturbance (Tables 3.12 and 3.15). While propagules with wings and plumes may travel farther in these habitats than under a closed forest canopy (Hughes *et al.* 1994), the tendency for wind-dispersed herbs to be poorly represented in forest habitats is more likely to be have been caused by differential germination success arising from differences in seed weight (Salisbury 1942, 1974; Baker 1972, Luftensteiner 1979, Mazer 1990, Saverimuttu and Westoby 1996). Species of open habitats tend to have lower seed mass than congener species of closed habitats and thus lack sufficient reserves to sustain seedling growth in dense shade during the critical cotyledon stage (Saverimuttu and Westoby 1996). According to this view, species should differentially occupy habitats on the basis of seed size (Mazer 1990) rather than on the basis of dispersal reach.

Published data were not sufficient to test the seed size (seed reserve) hypothesis. However, in keeping with this hypothesis, approximately 10 % of the wind-dispersed herbs in this study had an affinity for closed habitats (Appendix 2): Aster macrophyllus, Boehmeria cylindrica, Galearis

spectabilis. *Prenanthes* sp., *Solidago caesia*, and *Solidago flexicaulis*. The reported mean seed mass for congeners of closed vs. open habitats (*sensu lato*) is consistent with the predictions of the seed size (seed reserve) hypothesis: 0.847 mg (n=1) vs 0.250 mg (n=5) for classified species of *Aster*: 0.346 mg. (n=1) vs. 0.090 mg. (n=7) for classified species of *Solidago* (Mazer 1990).

In addition, the mean frequency of wind-dispersed herbs in sampled patches was higher for herbs with affinity for "forest" habitats ($\bar{x} = 18.0$ quadrats, n=6 herbs) than for herbs with affinity for "open" habitats ($\bar{x} = 6.4$ quadrats, n=22 herbs) (p=0.03. Wilcoxon rank sum test, not shown). Herbs with affinities for intermediate conditions were present at higher frequencies ($\bar{x} = 43.2$ quadrats, n=6 herbs with affinity for "forest + open habitats"; $\bar{x}=16.8$ quadrats, n=28 herbs with affinity for "open + forest habitats"). These data suggest that lower wind speeds in forest interiors do not pose a colonization constraint for wind-dispersed herbs with affinities for closed habitats (*sensu lato*). The dispersal reach of asters and goldenrods may be enhanced by late season flowering and by higher wind speeds associated with a senescing canopy. The higher mean frequency of herbs with affinity for "forest+open" habitats may indicate that the dispersal reach of forest dependent taxa is constrained by their greater seed mass.

Wind-dispersed herbs were present in every sampled forest patch but were absent from 10 % of quadrats within patches. Closed canopy conditions were present in most of these quadrats. In the remaining cases, canopy gaps were either small (<2% of sampled area) or in a large forest stand where the sole wind dispersed herb was *Solidago flexicanlis*.

Taken together, these results suggest that closed canopy conditions pose less of a constraint for herbs with affinities for closed habitats ("forest", "forest+open") than for open habitats ("open", "open + forest"). While the colonization success of wind-dispersed herbs may be mediated to some extent by dispersal processes, the overall evidence from this study suggests that the principal mechanisms governing their distribution are differential germination, establishment or persistence.

Herbs dispersed by animal adhesion were over-represented in open seasonally moist depressions

(Table 3.13) and were consistently associated with open habitats in contrasting habitat comparisons (Table 3.15). While these habitats are associated with animal activity, the more likely explanation for this pattern is differential germination, establishment or persistence since dispersal by animal adhesion is primarily a stochastic process (Agnew and Flux 1970, Bullock and Primack 1977, Sorensen 1986, Williams and Guries 1994, Kiviniemi 1996). The habitat affinities of herbs dispersed by adhesion are broadly consistent with this inference: 74.2% of taxa have affinities for open habitats (*sensu lato*) and 30.4 % of these taxa have affinities for moist and wet habitats.

Herbs dispersed by mechanical expulsion were over-represented in open seasonally wet tree pits (Table 3.13) and were strongly associated with logs and stumps under closed canopies in habitat comparisons (Tables 3.14). At the local scale, dispersal by mechanical expulsion should be primarily a non-limiting process since most seeds land in the immediate vicinity of the parent plant (Appendix 11). The principal contributors to pattern in the local herb assemblage should therefore be differential germination, establishment and persistence. In keeping with this inference, only 50% of the ballochores recorded on closed forest floors were present on logs and stumps (summaries of survey data, not shown). The ballochores occupying open seasonally wet tree pits were habitat generalists that were recorded in more than 60% of clossified habitats.

Although the dispersal reach of herbs dispersed by mechanical expulsion is typically less than 5 metres (Appendix 11), ballochores were present in 23 of 24 sampled patches. Longer-distance dispersal may therefore be achieved by ingestion by birds, mammals or insects. Interestingly, herbs dispersed by mechanical expulsion in this study were never secondarily dispersed by animal adhesion.

Herbs dispersed by unassisted means were over-represented on closed seasonally wet forest floors (Table 3.13) and under-represented on closed logs and stumps in habitat contrasts (Table 3.14). Although the diaspores of these herbs lack apparent features that may facilitate their dispersal, herbs dispersed by unassisted means were present in every patch and in 181 of 192 quadrats. The contribution of dispersal to this pattern is not known but presumably is mediated by animal vectors.

Herbs dispersed by prolonged dormancy were initially dispersed by some other agent: unknown vectors (36%), wind (34%), vegetative expansion (26%), adhesion to animals (16%), mechanical expulsion (14%), ants (6%), and ingestion by animals (2%)(percentages do not sum to 100 since certain herbs were dispersed by more than one mode). The distribution of propagules arising from dispersal in space has presumably been masked by patterns of recruitment in time, owing to the inherent requirement for high light conditions for germination and establishment. In keeping with this inference, the habitats in which dispersal by prolonged dormancy was most prominent were open microhabitats associated with human disturbance and former canopy gaps (closed logs and stumps) (Table 3.15).

Herbs dispersed by vegetative expansion were often dispersed by some other agent: animal ingestion (9.9%), animal adhesion (4.4%), ants (22.0%), wind (35.2%), prolonged dormancy (14.3%), mechanical expulsion (8.8%), and multiple modes (13.2%). Herbs dispersed by vegetative expansion were present in every quadrat and rarely were associated with particular microhabitats. Their numbers were over-represented, however, in open, moist, canopy gaps and in closed, dry, tree pits (Table 3.13). The former association is unique among dispersal modes whereas the latter association was also shared by herbs dispersed by animal ingestion and by ants. The distribution of herbs dispersed by vegetative expansion may therefore be mediated to some extent by animals.

Dispersal in Time

Dormancy enables plants to germinate when conditions are favorable and to delay germination when they are not. Delayed germination is predicted to be favored in taxa of open habitats since it allows species to avoid exposure to unfavorable conditions and to specialize on conditions that maximize reproductive success (Brown and Venable 1986, Cohen and Levin 1987). Delayed germination is predicted to be less favored in forest habitats, since the enhanced seed reserves that facilitate establishment in dense shade also enable forest plants to specialize on shaded conditions and to avoid the reduction in mean annual fitness that arises when dormancy constrains the number of seeds that can be produced in favorable years (Venable and Brown 1988). Selection pressure for the evolution of dormancy, therefore, is expected to be greatest in permanently open habitats, where environmental quality varies more or less randomly in time, and to be intermediate in successional habitats, where the decline in environmental quality is progressive and persistent, and where non-seed-bank traits increasingly should be favored (Brown and Venable 1986). Plants with annual life histories are expected to benefit more from dormancy than plants with biennial or perennial life histories (Rees 1993, 1994).

Approximately 20% of the herbs recorded in this study are known to disperse in persistent soil seed banks (Table 3.1). In keeping with models of the evolution of dormancy, the majority of these herbs (92%) were species with affinities for "open" or "open+forest" habitats, and no herbs were species with an affinity for "forest" habitats (Table 3.9). Moreover, the number of taxa with capacity for prolonged dormancy increased with increasing affinity for open conditions (Table 3.9). These results, and current theory, suggest that in forested habitats, dispersal in time may be restricted to fugitive species and to sites of recent canopy disturbance. In keeping with this prediction, herbs with capacity for dormancy were significantly over-represented in the trace cover class (1-5 individuals) and under-represented in the higher cover classes (Table 3.17).

Unexpectedly, many seed-banking herbs (52%) in this study were also capable of long-distance dispersal (i.e., dispersal by wind, animal ingestion or animal adhesion). Typically, models of annual herbs in spatially and temporally variable environments predict an inverse relationship between dormancy and dispersal, since a reduction in the variance in reproductive success by dispersal should reduce selection pressure for the evolution of dormancy, and vice versa (e.g. Venable and Lawlor 1980. Cohen and Levin 1987). In plants with dimorphic seeds, reproductive success is maximized when low dispersal seeds have delayed germination and when high dispersal seeds have quick germination (Venable and Lawlor 1980). These combinations are also favored in plants with monomorphic seeds (Cohen and Levin 1987). A seed that is both dormant and mobile is therefore unexpected.

The most apparent explanation for taxa with the capacity for high dispersal and high dormancy is a seed polymorphism. In this study, such a species would be classified as having "high dispersal"

based on the morphology of the diaspore, and, as possessing "high dormancy" based on its demonstrated capacity to persist in the soil for ≥ 5 years. Based on the reasoning above, one would expect the dormant seeds to be dispersed close to the parent and the non-dormant seeds to be dispersed remote from the parent. The reverse pattern is less likely and has been observed only in selected species of Brassicaceae (Venable and Lawlor 1980). In keeping with this reasoning, one might also expect variability in the reported dormancy status of such species since their classification would depend on the morph that was sampled. All but one "high-dispersal, high-dormancy" taxa in this study (*Bidens frondosa*) were typically classified as "transient" or "short-term persistent" in Thompson *et al.* (1997). An assessment of seed polymorphism in these taxa, therefore, appears warranted.

Deterministic versus Stochastic Dispersal Processes

Most seeds of most North American herbs apparently land within a few metres of the parent plant (Portnoy and Willson 1993, Cain *et al.* 1998, Appendix 11). At this spatial scale, dispersal is primarily a deterministic, non-limiting, process and pattern in the structure of herb assemblages should be governed primarily by factors controlling germination, establishment and persistence. The maximum dispersal distance achieved by North American herbs is poorly understood but is apparently on the order of tens of metres for most herbs, and, rarely more than a thousand metres for seeds dispersed by the wind and animal adhesion (Appendix 11). Dispersal at these spatial scales is increasingly a stochastic process and may result in dispersal limitiation on short time scales. Dispersal by ants and by frugivorous birds may be important exceptions since these taxa have the potential for "directed" dispersal at these spatial scales.

When viewed from this perspective, dispersal is primarily a short-distance process that operates on the scale of metres to tens of metres. This is the scale where dispersal has the greatest control over where a seed may land, and, apparently is the scale at which reproductive success, on average, is maximized. Paradoxically, it is also the scale where differences in the dispersal reach of evolved morphologies are minimized. This is apparently true even for herbs of ephemeral habitats since most of their seeds, on average, also land in close proximity to the parent (Appendix 11). However, it is these seeds in which dormancy is most often expressed and thus travel greater distances in time (Flint and Palmblad 1978. Olivieri and Berger 1985. Venable and Lawlor 1980. Venable *et al.* 1995). Taken together, these tendencies imply that, on average, the fitness benefits to be derived from long-distance dispersal are outweighed by the benefits to be derived from "controlled" dispersal at the micro-scale. Paradoxically, it would appear that dispersal is most successful when the movement of seeds is most constrained.

3.5 Principal Findings

Most plants in the forest understory were dispersed by animals (38.5 %), vegetative expansion (35.8 %), and the wind (30.5 %). Others were dispersed by unassisted means (22.0 %), prolonged dormancy in the soil (15.5 %), multiple modes (12.3 %), or mechanical expulsion (4.6 %). Of those dispersed by animals, 16.9 % were dispersed by animal ingestion, 12.8% were dispersed by animal adhesion, and 9.7% were dispersed by ants or by seed eaching. These patterns are broadly consistent with reported dispersal profiles for temperate forest habitats in the Great Lakes region and northeast United States (Table 1.1).

The proportion of taxa dispersed by a given mode was found to be sensitive to environmental conditions at the micro spatial scale. Characterization of the dispersal profile at the patch or landscape spatial scale, therefore, may mask variation in dispersal frequencies that reflect differences in causal factors.

- 2. The mode by which plants were dispersed varied by life form. Most trees, and all ferns and fern allies, were dispersed by the wind, whereas most shrubs and vines were dispersed by animal ingestion. Most grasses were dispersed by animal adhesion. Herbs were the only life form to be dispersed by all modes and were dispersed primarily by vegetative expansion, animal vectors or unassisted means.
- The mode by which plants were dispersed also varied by life history, provenance, modality,
 fruit type, habitat affinity, and shade tolerance. Pattern in the frequency of dispersal modes
was also observed in relation to environmental gradients, microhabitat, patch size and plant cover class. Dispersal by animal ingestion and by ants was associated with low species richness in 10m x 10m quadrats, whereas dispersal by animal adhesion, wind, prolonged dormancy, mechanical expulsion, unassisted means, and multiple modes were associated with high species richness.

- 4. The composition of herbs in sampled patches has not been constrained by the modes of dispersal employed by herbs, or at least, has not been constrained absolutely. Each mode of dispersal was represented on each of the environmental gradients examined, and, with few exceptions, in each type of microhabitat. Although most modes were absent from at least one 10m x 10m quadrat, each mode of dispersal was present in 23 of 24 surveyed patches. These patterns suggest that at least some seeds from each functional group were able to reach most patches and to germinate there.
- 5. The species composition of selected habitats has been enriched by dispersal by frugivorous birds and mammals and by ants. The strongest evidence for directed dispersal by animal ingestion was found in former canopy gaps and on disturbance features associated with canopy openings (tip-up mounds, tree pits, logs, stumps, forest lanes and access roads). The strongest evidence for directed dispersal by ants was found on closed tip-up mounds and tree-pits.
- 6. With few exceptions, the tendency for modes of dispersal to be correlated with particular environmental states cannot be attributed to dispersal processes. The supporting evidence for this conclusion is not definitive, however, since the methodology has limited capacity to differentiate between effects caused by dispersal and effects caused by germination, establishment and persistence.
- Several herbs were classified as having the capacity for both long-distance dispersal and dormancy. This character combination is rare in the literature and counter to current theory

regarding the evolution of dormancy. The most apparent reason for this combination is seed heteromorphism, a property previously unreported for these species. Approximately 10 % of surveyed herbs, and 21% of herbs with the capacity for long-distance dispersal, possessed the traits of interest.

8. If most seeds of most herbs land within a few metres of the maternal plant, then dispersal is primarily a non-limiting, deterministic, process and pattern in established herb assemblages should be due primarily to differential germination, establishment, and persistence.

Evolved dispersal morphologies, however, enable at least some propagules to travel metres to hundreds of metres from the maternal plant, on short time scales. At this spatial and temporal scale, dispersal is increasingly a stochastic process and the structure of established herb assemblages may be mediated by the number and spatial distribution of dispersing seeds. Opportunities to detect change based on properties of the established vegetation are limited, however, since only rarely can the contribution of dispersal be differentiated from the effects of germination, establishment and persistence.

4.0 PATTERNS OF SPECIES RICHNESS

4.1 Introduction

The paradox posed by the apparent coexistence of many species in the same environment has attracted attention for more than a century. According to the competitive exclusion principle, two similar species cannot coexist indefinitely on a single limiting resource in a uniform environment (Grinnell 1904, Volterra 1926, Gause 1934). Under such conditions, one species should eventually exclude the other and the assemblage should be reduced to a single species (Hutchinson 1957, 1959, 1961; Hardin 1960). How is it, then, that so many natural habitats are species rich?

One productive approach to this question has been to examine the assumptions of the exclusion principle: what happens if species are not "similar". if interactions do not proceed to equilibrium. if there is more than one limiting resource, or, if the environment is not spatially and temporally uniform. This research effort has generated a vast literature and many alternative explanations (Connell 1978, Huston 1979, Pickett 1980, Sousa 1984, Petraitis *et al.* 1989, Hart and Horwitz 1991, Tilman and Pacala 1993, Ricklefs and Schluter 1993, Huston 1994, Palmer 1994, Heywood 1995, Grace 1999). A selected review of the principal themes follows.

Limiting Similarity: The Lotka-Volterra model of interspecific competition predicts the stable coexistence of species when competition between species is less significant than competition within species. One way this may be achieved is through niche differentiation. If competitors were to differ in their requirements for a limiting resource, or, were to consume that resource at a different time or in a different place, then the effects of competition would be concentrated more within than between species and coexistence would be favored (Begon *et al.* 1990).

The apparent necessity for coexisting species to possess distinct ecologies has been demonstrated in mathematical models in both uniform and variable environments (e.g. MacArthur and Levins 1967, May 1973, Chesson and Warner 1981, Pacala and Tilman 1994). An important prerequisite for coexistence is that populations be able to expand when rare. If individuals of different species in a competing assemblage were indistinguishable, then the competition experienced by individuals would depend solely on the total density of individuals in the assemblage. Under these circumstances, individuals of rare species experience no less competition than individuals of abundant species. And, since individuals possess the same capacity to reproduce and disperse, and propagules have the same response to differences in environmental quality, each species has the same chance of recruiting offspring into the population. Under these circumstances, it is not apparent how populations recover from fluctuations that take them to low density, and thus, how the criterion for coexistence may be satisfied (Chesson 1991).

There would appear to be some limit, therefore, to the similarity in the way in which species in a competitive assemblage may use, and respond to differences in, available resources. That limit is is now expected to be both environment and species assemblage specific (Abrams 1983).

The niche differentiation hypothesis may constitute a necessary, but not sufficient. explanation of species richness, however, owing to the degree of resource partitioning that is implied for a species rich assemblage (Connell 1978, Silvertown and Law 1987). In spite of the apparent utility of concepts such as the regeneration niche (Grubb 1977, 1986), and the resource ratio hypothesis (Tilman 1982, 1988), a major difficulty in applied studies has been to identify the important resource axes that may permit coexistence at the local scale (Silvertown and Doust 1993, Tilman 1993, Hooper and Vitousek 1997, Busing and White 1997).

Non-Equilibrium Conditions: An important limitation of hypotheses that seek to explain species coexistence under equilibrium conditions is that assemblages are embedded in environments that are subject to more or less continuous change. In a variable environment, species densities are expected to fluctuate over time, rather than stabilize, in response to patch dynamics (Watt 1947, Oliver 1981, Pickett and White 1985) and to periodic disturbance (Connell 1978, Pickett 1980, Sousa 1984, Petraitis *et al.* 1989). Under non-equilibrium conditions, community recovery from such processes is expected to be incomplete, owing to priority effects (Yodzis 1978, 1986), differences in the assembly sequence arising from chance colonization events (Drake 1991, Fastie 1995), and time lags in plant response (Davis 1981, 1986).

Hypotheses that account for spatial and temporal variability in the environment have proposed three general ways in which variability may facilitate species coexistence on ecological time scales: i) by slowing down the rate of competitive exclusion (e.g. Hutchinson 1941, Levin and Paine 1974, Connell 1978, Huston 1979, Shmida and Ellner 1984): ii) by reversing the order of competitive superiority among species (e.g. Hutchinson 1961, Chesson 1985, Pacala 1987); and, iii) by altering the availability of limiting resources (e.g. Tilman 1982, 1988). These processes are not mutually exclusive and may differentially apply in given circumstances (Warner and Chesson 1985). The first process emphasizes the conservation of existing species, whereas the latter processes emphasize how populations and diversity may increase (Chesson and Case 1986).

An important property of the slow dynamics hypothesis is that disturbance interrupts and sets back the process of competitive exclusion, by removing the competitive dominants and releasing resources for established species. This prevents competitive dominants from monopolizing available resources and promotes species coexistence by delaying the exclusion of inferior competitors. Factors expected to promote slow dynamics include: i) long life spans; ii) small differences in competitive strength; iii) highly aggregated or elumped distributions; iv) a surplus of suitable microsites; v) widely separated microsites; vi) intermittent competition; vii) slow growth rates; and, viii) intermediate rates of disturbance (Shmida and Ellner 1984, Huston 1979, Connell 1978). Related theories based primarily on small differences in competitive strength include the slow dynamics model of Hubbell and Foster (1986), and, the ecological and competitive combining ability model of Aarssen (1983).

Patterns and processes of diversity explained by this mechanism include: gap phase dynamics (Watt 1947, Oliver 1981, Pickett and White 1985): the initial floristic composition model of secondary succession (Egler 1954): species coexistence in the presence of keystone predators (Paine 1966, 1974); and, the displacement of characteristic species and assemblages in the absence of fire (Varga 1989, Kruger and Reich 1997), fluctuating water levels (Keddy and Reznicek 1982, 1986), herbivory (Whitney 1984, Grubb 1986), and land-use management (Gimingham 1972, Heil and Bruggink 1987).

An important property of the reversal in competitive rankings hypothesis is that a variable environment provides some place or time where competitors may perform poorly or not survive, and, where populations of low abundant species may expand. It is the latter process that distinguishes this explanation from the others and that provides the basis for the coexistence (Warner and Chesson 1985, Chesson 1986). Factors expected to promote reversals in competitive superiority in competing plant assemblages include: i) juvenile mortality rates that are sensitive to environmental conditions; ii) reproductive rates that are sensitive to environmental conditions; iii) adult survival rates that are insensitive to environmental conditions; iv) overlapping generations; v) iteroparity; and, vi) random dispersal (Chesson 1985, 1986; Pacala 1987).

In this analytical setting, species in strong competition may depress each other's recruitment rates. Nevertheless, each species may still be able to show positive average growth rates provided that it has periods when it is able to recruit well. In a variable environment, temporary reversals in the competitive advantage of species make this increasingly likely owing to the sensitivity of both juveniles and reproductive rates to changes in environmental conditions. This should lead to the strong recruitment of different species at different times, or in different places, and reverse the trend towards competitive exclusion for all species. Overlapping generations promote coexistence in this setting by ensuring that demographic gains made during favorable periods are "stored" in the surviving offspring and contribute to reproduction when favorable conditions return (Warner and Chesson 1985, Chesson 1985). Dispersal facilitates the reversal in competitive rankings through the randomization of neighbors in competing plant assemblages (Pacala 1987).

An important property of the resource availability hypothesis is that species utilize different portions of available habitats, or, differ in the range of habitats wherein they have a competitive advantage over other species, owing to differences in plant requirements and the differential availability of limiting resources (Whittaker 1965, Grime 1973, 1979; Tilman 1982, 1988; Comins and Noble 1985; Bazzaz 1991). In these models, coexistence is achieved primarily through the spatial segregation of competitors in a mosaic of suitable habitats. The C-S-R model of competition (Grime 1973, 1979) and the resource ratio hypothesis (Tilman 1982, 1988) emphasize competitive interactions among established plants, whereas, the lottery model of competition (Sale 1977,

Chesson and Warner 1981. Comins and Noble 1985) emphasizes the recruitment phase and the competition for suitable sites for establishment. Periodic disturbance facilitates coexistence and enriches local diversity by increasing spatial heterogeneity in the availability of limiting resources. In these models, the important consequence of disturbance is the change in the relative supply of limiting resources for which competition occurs. Competition is not so much interrupted as redirected.

Patterns explained by the differential availability of limiting resources include: the paradox of enrichment (Rosenzweig 1971, Al-Mufti *et al.* 1977, Tilman 1982, Wisheu and Keddy 1989, Abrams 1995): the facilitation model of plant succession (Connell and Slayter 1977, Finegan 1984): regeneration from persistent seed pools (Thompson *et al.* 1998), persistent seedlings, and serotinous cones: and, the differential composition and richness of plant assemblages along gradients of moisture (Spies and Barnes 1985, Caspersen *et al.* 1999), nutrients (Whitney 1991, Diekmann and Falkengren-Grerup 1998, Hutchinson *et al.* 1999), light (Thomas *et al.* 1999), and geomorpholgy (Pastor *et al.* 1984, Host and Pregitzer 1992, Reich *et al.* 1997).

Dispersal: The importance of environmental heterogeneity and periodic disturbance to species coexistence has awakened interest in the potential contribution of dispersal to the spatial and temporal segregation of competing species. Whereas the former segregates competitors by creating conditions locally suitable to different species, the latter may segregate competitors by chance dispersal to empty sites followed by population expansion of the favored species. Current models suggest that segregation by dispersal may arise at two spatial scales: the micro-scale of the competing plant assemblage (Atkinson and Shorrocks 1981, Shmida and Ellner 1984, Ives and May 1985, Pacala 1986, Geritz *et al.* 1988), and, the scale at which environmental heterogeneity enables each species to compete for establishment sites (Comins and Noble 1985, Pacala and Tilman 1994, Lavorel *et al.* 1994, Lavorel and Chesson 1995, Hurtt and Pacala 1995, Holmes and Wilson 1998, Bolker and Pacala 1999, Loreau and Mouquet 1999).

Rather than interact with a number of individuals in a plant assemblage, and thereby experience the average density of the population at large, plants interact primarily with individuals that lie within

a crown or root diameter (Harper 1977, Pacala and Silander 1985, Venable and Brown 1993). In the forest understory, this means that the distance over which most plants competitively interact is on the order of decimetres to metres (Pacala and Silander 1987). When the dispersal distance of plants is short, siblings tend to aggregate into monospecific clumps and competing species become segregated spatially. Under these circumstances, individuals tend to compete more with their own kind than with others and thus create the conditions for coexistence (Atkinson and Shorrocks 1981, Shmida and Ellner 1984, Ives and May 1985, Pacala 1986, 1987). An interesting property of these models is that the spatial heterogeneity required for coexistence is generated by the plants themselves.

When the dispersal distance extends beyond the immediate vicinity of the maternal plant, a declining number of conspecific seeds land in close proximity to one another and an increasing number of seeds land in environments that are less favorable than the home patch. These processes confer a competitive advantage to established plant assemblages by limiting the number of seeds from neighboring patches and by limiting the capacity of invading seeds to form aggregations of their own kind. These dynamics are expected to favor coexistence by enabling each species to be dominant in site establishment at some time or place (Shmida and Ellner 1984, Comins and Noble 1985, Loreau and Mouquet 1999).

The stochastic arrival of a diminishing number of seeds is expected to facilitate this process by randomizing the composition of neighbors in competing plant assemblages and by leaving a certain proportion of sites vacant for colonization (Pacala 1986, Geritz *et al.* 1988, Clark *et al.* 1998, Ehrlen and Eriksson 2000). The former process increases the likelihood that seeds germinate next to weaker competitors, or to stronger competitors weakened by a less favorable setting, whereas the latter process enables inferior competitors to win favorable sites in the absence of more dominant but recruitment limited competitors (Hurtt and Pacala 1995).

The degree to which long-distance dispersal is required for the persistence of competitively inferior species remains controversial. Recent modeling studies suggest that a long dispersal reach, *per se*, is not as important as regeneration niche, fecundity, disturbance, storage effects, and the spatial

pattern of suitable sites (Lavorel *et al.* 1994, Lavorel and Chesson 1995, Holmes and Wilson 1998, Bolker and Pacala 1999). In general, dispersal beyond the immediate vicinity of the maternal plant is favored whenever the distribution of the superior competitor is strongly clustered and the invading species is able to establish in the intervening gaps. The required dispersal distance to achieve this need not be great if both species are resident in the same habitat. Short distance dispersal is apparently favored in all other situations, owing to the competitive advantage that accrues when offspring are clustered and close to hand (Holmes and Wilson 1998, Bolker and Pacala 1999). With spatial segregation and environmental heterogeneity, these studies suggest that a mixed strategy which strongly favors short distance dispersal may offer the greatest return in most habitats.

Local versus Regional Processes: The assumption that the diversity of species in a given setting can be explained solely by properties of the local environment, and by interactions among resident species, has come under increasing scrutiny in recent years (e.g., MacArthur and Wilson 1967, Ricklefs 1987, Taylor *et al.* 1990, Cornell and Lawton 1992, Ricklefs and Schluter 1993, Caley and Schluter 1997. Cornell and Karlson 1997, Zobel 1997, Loreau and Mouquet 1999). Examination of processes that operate at larger spatial and temporal scales has revealed that the composition and richness of local assemblages may be influenced by historical processes such as time lags in plant response to climate change (Davis 1981a, 1986), alternative sequences of community development arising from chance colonization events (Drake 1991, Fastie 1995), establishment opportunities/ constraints arising from past disturbance (Motzkin *et al.* 1996, Abrams and Orwig 1996), extirpation by herbivory (Whitney 1984) or pathogens (Davis 1981b), and, by spatial processes such as dispersal (Holt 1993) and the scale of heterogeneity in the local environment (McLaughlin and Roughgarden 1993).

Recent studies investigating the relationship between local and regional species diversity have argued that the principal direction of control should be from regional to local (Cornell and Lawton1992. Eriksson 1993. Partel *et al.* 1996. Caley and Schluter 1997. Karlson and Cornell 1998). The principal reason for this is that in a variable environment biological interactions may not be sufficient to limit the number of species in the local assemblage. In the presence of dispersal, the richness of the local assemblage should reflect the number of species in the surrounding region that

are capable of establishing there. Empirical evidence from plant communities has been reported by Keddy 1981, Partel et al. 1996, Tilman 1997, Duncan et al. 1998, and Ehrlen and Eriksson 2000).

An alternative explanation for the positive correlation between local and regional richness observed in these studies is that the composition of plant assemblages is in fact constrained by local processes and that limits to regional richness are set by the number of distinct habitats, each supporting a saturated assemblage (Cornell and Lawton 1992). In the presence of migration, differences in the richness of local assemblages should reflect properties of the local environment rather than the number of species in the surrounding region.

Summary: Niche differentiation, competitive exclusion, slow dynamics, weak competitive interactions, differential resource availability, short-distance dispersal, recruitment limitation, and the consequences of periodic and chance events, are prominent elements in the vast number of individual hypotheses that have been advanced to explain observed differences in the composition and richness of plant assemblages. In a spatially and temporally variable environment, however, the synthesis for which we strive may well be pattern and scale dependent. If so, then a useful way forward is to clarify which of the alternative mechanisms apply to the pattern at hand and to what extent.

Study Objectives: The objectives of this chapter are the following: i) to identify the environmental variables that best explain observed differences in species richness in 10m x 10m quadrats: ii) to identify the plant attributes that best explain observed differences in species richness in 10m x 10m quadrats: and, iii) to compare the degree to which environmental and plant attributes explain observed differences in species richness.

4.2 Methods

4.2.1 Environmental Correlates of Species Richness

Environmental correlates of species richness were assessed in relation to edaphic, forest stand structure, human disturbance, environmental heterogeneity, and landscape, variables. Methods

related to the identification and evaluation of environmental correlates are presented in the following sections.

4.2.1.1 Generalized Linear Regression Models

Analysis #1: Environmental correlates of species richness in 10m x 10m quadrats were identified by simple linear regression. The normality of residuals was evaluated by the Shapiro-Wilk W test (SAS Institute Inc. 1997): environmental and response variables were transformed (square root, In, logit) where necessary to achieve, and/or maximize, normality. Pair-wise interactions among environmental variables were evaluated and all significant interactions (p<0.05) recorded. Analyses were performed in JMP, Version 3.2.2., SAS Institute.

Alternative combinations of variables that maximally explained the variance in observed species richness were identified through generalized linear regression (GLM) models that were fit manually by forward and backward selection. Separate models were fit for edaphic, forest stand structure, human disturbance, environmental heterogeneity, and, landscape, variables. Summary models were also developed to identify an overall set of environmental variables that maximally explained observed differences in species richness. Candidate models were evaluated in relation to Mallow's *Cp* statistic (Neter *et al.* 1996, Ryan 1997, Hocking 1996, Mathsoft Inc.1998) to ensure that only the most parsimonious models were retained for further analysis. Only significant, non-interacting, terms were retained in each model.

A "Best Model" was identified for each set of variables, based on the amount of explained variance $(r^2 adjusted statistic)$. Alternative models were retained for further analysis in cases where the next best model represented a better fit in relation to the Shapiro-Wilk W test, or, where an entirely different set of variables explained virtually the same amount of variance.

The results of this analysis are presented in Table 4.1. Scatter plots of important correlates of species richness are presented in Figures 4.1, 4.2, 4.3, 4.4. Mean species richness for categorical variables is presented in Table 4.2.

4.2.1.2 Contribution of Forest Stand Structure

Analysis #1: The contribution of stand structure to species richness (Table 4.1) was examined further to clarify the cause and effect relationship among selected variables. Of particular interest was the degree to which "number of tree species", and, "number of live stems 0-4 cm dbh", were causal mechanisms of species richness. Non-parametric correlations (Spearman's Rho) with selected stand structure variables, soil moisture, available calcium, and human disturbance, were computed with a view to identifying more proximate correlates of available moisture, nutrients and light on the forest floor.

Variables included in this analysis were: species richness; percent live tree stems Acer saccharum, percent live tree stems wet-mesic, or wet, tree species (Abies balsamea, Acer negundo, Acer saccharinum, Fraxinus nigra, Fraxinus pennsylvanica, Populus balsamifera, Thuja occidentalis, Ulmus occidentalis); percent live tree stems shade intolerant or very shade intolerant tree species (Acer negundo, Acer saccharinum, Betula papyrifera, Populus balsamifera, Populus grandidentata, Populus tremuloides, Robinia pseudoaccacia); percent canopy closure: soil moisture class (1 = seasonally dry depressions, 2 = seasonally moist or wet depressions; available calcium (cmols/kg); and, human disturbance (1 = no disturbance apparent in quadrat, 2 = trails, canopy gaps due to logging, or, regenerating fields, present in quadrat).

Trees associated with wet-mesic or wet soils were identified with reference to Oldham *et al.* (1996). Maycock (1963), and, Nimerfro and Brand (1993). Trees classified as obligate or facultative wetland species (wetland affinity rating = -1, -2, -3, -4, or -5)(Oldham *et al.* 1996) were chosen as representative indicators of wet-mesic or wet soils in this study. Trees that achieved their maximum average importance value in southern Ontario on wet-mesic soils (Maycock (1963) were typically rated as "facultative" or "facultative upland" species by Oldham *et al.* (1996), when abundant on other soils, and were not used as indicators in this study (*Betula papyrifera*, *Carpinus caroliniana*, *Carya cordiformis*, *Fraxinus americana*, *Tilia americana*, *Quercus macrocarpa*). Shade intolerant, and very shade intolerant, trees were identified with reference to Baker (1949). Barnes and Wagner (1981) and Oldham *et al.* (1996). The shade tolerance ratings of Barnes and Wagner (1981) were adopted for *Acer negundo*, *Acer saccharinum* (owing to uncertainties in the Baker's ratings) and *Prunus serotina* (owing to its relative shade tolerance when young); the shade tolerance ratings of Baker (1949) were used for all other species.

Analyses were performed in JMP, Version 3.2.2., SAS Institute. Results are presented in Table 4.3.

4.2.1.3 Contribution of Soil Fertility

Analysis #1: The interaction between available calcium and sugar maple abundance (Table 4.1) was investigated further to clarify the contribution of soil fertility to observed differences in species richness. This analysis was restricted to undisturbed forest stands on Brunisolic and Luvisolic soils, overlying calcareous till, in order to standardize samples with respect to soil parent material, soil moisture and recent site disturbance. Soils with free calcium carbonate in the upper 15 cm of the soil profile (positive reaction to 0.1N HCI) were excluded from this analysis in order to standardize samples with respect to exchangeable calcium (see contrasting treatment in Analysis 2). This analysis was further restricted to quadrats with cover type 2 (sugar maple + red or white oak) and cover type 3 (sugar maple, no red or white oak, no wet-mesic, wet, trees) in order to standardize samples with respect to forest cover. N = 29 10m x10m quadrats in 7 forest patches.

The contribution of available calcium to species richness, and to sugar maple abundance (% stems \geq 1 m *Acer saccharum*, # stems \geq 1 m *Acer saccharum*), was evaluated by simple linear regression. Analyses were performed in JMP, Version 3.2.2., SAS Institute. The results of this analysis are presented in Figure 4.5.

Analysis #2: The interaction between available calcium and sugar maple abundance was reexamined in undisturbed forest stands on Brunisolic soils in order to standardize samples for soil parent material (calcareous till), soil order (Orthic Melanic Brunisol) and soil series (Otonabee loam). The analysis was restricted to quadrats with cover type 2 (sugar maple + red or white oak) and cover type 3 (sugar maple, no red or white oak, no wet-mesic, wet, trees) in order to standardize samples with respect to forest cover. These stands (N = 17 quadrats in 3 forest patches) provided the most uniform subset of samples in which to assess the response of sugar maple to differences in available calcium.

The contribution of available calcium to species richness, and to sugar maple abundance, was evaluated by simple linear regression. Analyses were performed in JMP, Version 3.2.2., SAS Institute. The results of this analysis are presented in Figure 4.6.

Analysis #3: Stand history, edaphic variables, and available light, were examined further to determine whether one or more of these factors may have contributed to observed differences in species richness, or sugar maple abundance, in stands evaluated in Analysis #1. Indicators of potential differences in stand history were: stand structure (% stems in 0-4 cm dbh, 4-10 cm dbh, 10-30 cm dbh, and >30 cm dbh, size class); % live stems shade intolerant, very shade intolerant, trees; and, % canopy closure. Indicators of potential differences in soil properties were: available calcium, soil pH_{water} and % soil organic matter. Indicators of potential differences in available light were: % canopy closure; % taxa shade tolerant herbs (Nimerfro and Brand light rating 1, 2; Ellenberg light rating 1,2,3,4); and, % taxa shade intolerant herbs (Nimerfro and Brand light rating 4,5; Ellenberg light rating 6,7,8,9).

Differences among soil orders were evaluated by Wilcoxon rank sum tests. Analyses were performed in JMP. Version 3.2.2., SAS Institute. The results of this analysis are presented in Figures 4.7, 4.8, and 4.9.

4.2.1.4 Contribution of Patch Isolation and Patch Size

Analysis #1: The contribution of patch isolation and patch size to species richness was evaluated by simple and multiple linear regression. The independent contribution of patch isolation and patch size to species richness was evaluated by testing their interaction with edaphic, stand structure, and human disturbance variables. A significant interaction with one or more variables was taken as evidence that patch isolation, or patch size, *per se*, did not contribute to species richness. Analyses were performed in JMP. Version 3.2.2.. SAS Institute. The results of this analysis are presented in Table 4.1.

4.2.1.5 Contribution of Microhabitats

Analysis #1: The contribution of microhabitats to species richness was evaluated at the quadrat. forest patch, and landscape spatial scale. The objective was to determine the degree to which a given microhabitat contained species that did not occur elsewhere in the reference area. Observed differences in species richness ["total # species in quadrat or patch" versus "total # species in quadrat or patch - # species unique to microhabitat"] were evaluated by paired t-test, when the distribution of differences was normal, and by Wilcoxon signed rank sum test, otherwise. A minium number of six elements was required for the latter analysis (Sokal and Rohlf 1995, p.444). Quadrats or patches that contained only one microhabitat were excluded from the analysis.

Analyses were performed in JMP. Version 3.2.2., SAS Institute. The results of this analysis are presented in Tables 4.4, 4.5 and 4.6.

Analysis #2: The proportion of the study area occupied by habitats of contrasting moisture and canopy closure states was determined in order to evaluate the contribution of sampled area to the relative importance of moisture and light in models of species richness.

The results of this analysis are presented in Table 4.7.

4.2.2 Plant Trait Correlates of Species Richness

Plant trait correlates of species richness were assessed in relation to mode of dispersal, life history, provenance, life form, habitat affinity, shade tolerance, and moisture affinity. Methods related to the identification and evaluation of plant trait correlates are presented in the following analysis.

Analysis #1. Plant trait correlates of species richness in 10m x 10m quadrats were identified by simple linear regression. The normality of residuals was evaluated by the Shapiro-Wilk W test (SAS

Institute Inc. 1997); plant trait and response variables were transformed (square root. In. logit) where necessary to achieve, and/or maximize, normality. Pair-wise interactions among plant trait variables were evaluated and all significant interactions (p<0.05) noted. Analyses were performed in JMP. Version 3.2.2., SAS Institute.

Alternative combinations of variables that maximally explained the variance in observed species richness were identified through generalized linear regression (GLM) models that were fit manually by forward and backward selection. Separate models were fit for mode of dispersal, life history, provenance, life form, habitat affinity, shade tolerance, and moisture tolerance, variables. Summary models were also developed to identify an overall set of plant traits that maximally explained observed differences in species richness. Candidate models were evaluated in relation to Mallow's *Cp* statistic (Neter *et al.* 1996, Ryan 1997, Hocking 1996, Mathsoft Inc. 1998) to ensure that only the most parsimonious models were retained for further analysis. Only significant, non-interacting, terms were retained in each model.

A "Best Model" was identified for each set of variables, based on the amount of explained variance $(r^2 adjusted)$ and normality of residuals. An additional model was identified for modes of dispersal to permit a comparison among life forms.

The results of this analysis are presented in Table 4.8. Scatter plots of selected dispersal correlates of species richness are presented in Figures 4.10 and 4.11.

4.2.3 Comparison of Alternative Models of Species Richness

Analysis #1: Alternative models of species richness, composed of environmental variables, dispersal variables, and both environmental and dispersal variables, were compared within and among functional groupings, in relation to four statistical properties: F-statistic, r'adjusted value, mean square error, and, normality of residuals (Shapiro-Wilk W test). Interactions among explanatory variables prevented the use of a single evaluation criterion, such as Mallow's Cp statistic or the PRESS₀ statistic (Neter *et al.* 1996), to identify an overall best model. Models with a high $p \le W$

value and a low mean square error were considered superior to models with a low p<W value and a high mean square error. Superior models with a higher r²adjusted value were considered to be more informative than superior models with a lower r²adjusted value. Models that could not otherwise be differentiated were considered to be comparable models for the purpose of explaining observed differences in species richness.

One outlier, quadrat #160, was removed from each model (Cooke's D statistic = 0.226). Species richness in this quadrat was sharply reduced by seasonal standing water that covered more than 85% of the forest floor during the spring and summer survey, resulting in a strongly outlying data point in models with dispersal variables.

Analyses were performed in JMP, Version 3.2.2., SAS Institute. The results of the analysis are summarized in Tables 4.9 and 4.11.

Analysis #2: A graphical method (personal communication. Dr. Roger Hansell. Institute for Environmental Studies. University of Toronto) was subsequently employed to reveal the degree to which alternative models explained similar or dissimilar portions of sample space. Explanatory variables for each regression model were submitted to detrended correspondence analysis (DCA) (ter Braak 1987), and, the resulting "species" scores used to construct a polygon in ordination space. The ordination space enclosed by each polygon was interpreted to represent the portion of sample space maximally explained by each model. Regression models that overlapped in ordination space were considered less distinctive than models that did not.

The detrended correspondence analysis was performed in CANOCO. Version 3.12 (ter Braak 1991). The polygons were constructed in S-Plus, Version 4.5 (Mathsoft Inc. 1998). Representative results are presented in Figures 4.12 and 4.13. Summaries are presented in Tables 4.10 and 4.11.

4.2.4 Contribution of Phylogeny

Analysis #1: The contribution of phylogeny to explanations of variance in species richness (Table

4.8) was evaluated indirectly by refitting the regression models at more inclusive taxonomic ranks. Detection of a significant pair-wise interaction between terms was taken as indirect evidence of a phylogenetic contribution to observed differences in species richness.

The response variable in a given regression model was the number of species, general families, or orders, in a given $10m \ge 10m$ quadrat. The predictor variable for a plant trait was the number of species, general families, or orders with the trait of interest in a given $10m \ge 10m$ quadrat.

Analyses were performed in JMP. Version 3.2.2., SAS Institute. The results are summarized in Table 4.12.

Analysis #2: The contribution of phylogeny to observed interactions among selected variables at more inclusive taxonomic ranks was examined further by determining the loss of degrees of freedom associated with evaluating selected plant traits at more inclusive ranks. The traits addressed in this analysis were moisture affinity, shade tolerance, and mode of dispersal.

Analyses were performed in JMP, Version 3.2.2., SAS institute. The results are summarized in Table 4.13.

4.3 Results

4.3.1 Environmental Correlates of Species Richness

4.3.1.1 Generalized Linear Regression Models

Environmental correlates of species richness in 10m x 10m quadrats are reported in Table 4.1 in relation to the following groupings: edaphic variables, forest stand structure, human disturbance, environmental heterogeneity, landscape variables, and, overall model. Scatter plots of the most influential variables are presented in Figures 4.1, 4.2, 4.3, and 4.4; the mean species richness of categorical variables is presented in Table 4.2.

maple + wet-mesic or wet species, CT5=wet mesic or wet species, no sugar maple, no red or white eak; $\ln p/q = \ln [(proportion)^{2}(1-$ Table 4.1. Environmental correlates of species richness in 10m x 10m quadrats. Generalized linear regression (GLM) models, by row, See notes for interactions among predictor variables, and, for partial F statistics of selected models. See text for criteria for selection of "best" models. Legend: comcalcareous outwash; lacmlacustrine; ct=calcareous till; gf=glacio-fluvial; gb=gleyed brunsisol; gl=gleyed luvisol; b=brunisol; l=luvisol; g=gleysol; CT1= oak, no sugar maple, no red or white oak; CT2=sugar maple + oak; CT3=sugar maple; CT4=sugar proportion)].

Mudel	Response Variable	F	p-Value	r' adj.	PW'	relationship
1. EDAPHIC VARIABLES						
soil parent material ^{3,5,1,0,0,1,2,4} and 10,2,2	dds y ubs	18.99	0.0001	0.2203	0.29	cu>lac>cl>gf
soil order ^{1,4,10,13,10,13,10,1}	lu synt # spp	17.13	0.0001	0.2525	0.18	gb>g >b> s
soil moisture ^{t stat} te.te.st.et.e	sqri # spp	32.32	0.0001	0.2469	0.58	wet>dry
In % soil organic matter 439,14,00,00,00,00	dd≈ # u	11,52	0.0008	0.0522	0.26	positive
sold pH2 statements and statements and how	sdir # spp	[6:87	0.0001	£0F1/0	0.06	positive
available calcium ^{2,10,14,19,24,27}	synt <i>li</i> spp	5.52	0.0194	0.0260	0.15	positive
Best Model: soil moisture + soil parent material*	sqrt # ×pp	29.36	100010	0.3557	0.71	
2. FOREST STAND STRUCTURE						1
erenereren erendet son in son son son son son son son son son so	dds # 10s	19.25	1000'0	0.2765	0.09	CTS-CTI-CT4> CT3-CT2
% canopy closure (h/d ta) h/d (h/d ta)	sqrt # spp	34,78	1000'0	0.1503	0.05	negalive
tree species (> m) ^{\$,14,15} נגני 1,155	ln sqrt # spp	71.00	0.0001	0,2682	0.65	pasitive
In # wet-mesic or wet tree species ^{4,5} 8,10 to 17,25	dds # ubs	30.07	1000'0	0.2235	0.50	positive

Madel	Response Variable	4	p-Value	r² adj.	^{_W>d}	relationship
sqrt # live tree stems 0-4 cm dbls.in 1.5. m.2.2.2.2.25	dds # ubs	47.65	1000'0	0.1963	0.28	positive
sqrt # live tree stems 4-10 cm dbh ^{1,4,8,10,10,10,10,10,10}	sqrt # spp	1.49	0.2237	0.0026	0.07	positive
h # live tree stents 10-30 cm dbh23,0,0,0,0,0,0	dds # ubs vi	0.28	0.5990	0.0042	0.04	1
# live tree stems >30 cm dbh ^{t3}	sqrt # spp	7.37	0.0072	0.0323	0.07	negative
sqrt % stems sugar maple ^{s a to the stand set}	sqrt # spp	66.55	0,0001	0.2555	0.28	negative
sqrt % steins sugar maple 0-4cm dbhatter an at a steins sugar maple	dds # sqr	66.01	0.0001	0.2539	0.74	negative
sqrt % stems sugar maple 4-10 cm dbh ^{30,15,00,20,20,13}	dds # µbs	30.38	100010	166010	0.11	negative
% stems sugar maple 10-30 cm dbh ^{4 a tt to 33}	dds # ubs	19.14	0.0001	0.0959	0.07	negative
Best Model 1: ht # tree species (>1m) 1 canopy closure (In p/q) + forest cover type ¹²	अता # spp	39.67	1000.0	0.5485	0.68	
Best Model 2: sort # stems 0-4 cm dbh + campy closure (In p/q) + forest cover (ype ¹⁾	sqirt # spp	32.52	0.0001	0.4975	0.92	
3. HUMAN DISTURBANCE						
open Irail?	scht # spp	19.57	0.0001	0.0886	0.28	present > absent
closed trail+,11,24,36,2%	sqrt # spp	9.59	0.0022	0.0431	0.06	present > absent
open regenerating field ^{2,10,20,21,29}	sqrt # spp	17.41	0.0001	0.0791	0.03	present > absent
open canopy gap ^{4,12,14,18,19,24,26,31}	dds ∦ ⊔bs	9.21	0.0027	0.0412	0.04	present > absent

Table 4.1. Environmental correlates of species richness (cont'd).

223

Model	Response Variable	ŀ	p-Value	r² adj.	_w^	relationship
closed canopy gap ^{4, 19}	syct # spp	P0.04	0.8371	0.0050	0.02	present > absent
disturbed microhabitats ^{5,14,16, 20, 14,29, 20, 24,31,34}	sqri # spp	19,67	0,0001	0,0890	0.04	present > absent
Best Model: upen trail + open regenerating field30	dds # լսիs	15,03	1000'0	0.1280	0.20	
4. ENVIRONMENTAL HETHROGHNELEY						
# microliabitals ^{2,10,13,14,19,27,29}	ln sqrt # spp	11.84	2000'0	0.0537	0.05	positive
5. LANDSCAPE VARIABLES ¹²						
% forest cover (within 5km x 5km square centered on study site) ^{1,10,12,10,12,14}	sqit # spp	اط,ظ	0,0002	0.0658	0.14	negative
In mean distance to acarest forest patch (in 45° ares) ^{AddMagert}	dds # uhs u	101	691£.0	0.000.0	0.01	negative
sgrt patch area ^{2,0,4,00,11,11,11,11,18} [v,20,1,34	dds # 11ps	4.67	0.0319	0.0189	0.12	negative
Best model: % forest cover	sqrt # spp	14,44	0.0002	0.0658	0.14	negative
6. OVERALL MODEL: ENVIRONMENTAL CORRELA	TES					
Best Model 1; in # tree species (> tm) + canopy closure (in $p(q)$ + forest cover type + open microhabitats th	dds # ubs	39.04	0.0001	0.5823	0.66	
Best Model 2; canopy closure (In p/q) + sqrt # stems 0-4 cm dbh +moisture class + soil parent material ¹²	sqrt # spp	34.10	0.0001	0.5495	0.69	

Tuble 4.1. Environmental correlates of species richness (cont'd).

Notes: 1. Shapiro-Wilk W test for normality of residuals (residuals normal when p₂0.05).

28. Significant interaction with closed canopy gaps. 29. Significant interaction with disturbed microhabitats fent or sume wars, fanes, resementing fields - chosed or upen canopy).

Significant interaction with soil parent material.
Significant interaction with soil order.
Significant interaction with soil moisture.

Notes (cont'd).

335

Notes (cont'd).

- 30. Variables in model presented in descending order, based on partial F statistics; interactions among variables not significant; all elements in model significant, p<0.002. Partial F statistics: open trail: F=11.67; open regenerating field: F=9.60.
 - Significant interaction with # microhabitats.
 - N=24 patches; interactions based on mean # elements in a given patch. 计记载环境
 - Significant interaction with % forest cover.
- Significant interaction with In mean distance to nearest forest patch (in 45° arcs)
 - Significant interaction with surt patch area.
- forest cover type: F=16.93, open microliabitats (cut or souggap, lane, regenerating field, seep, riparian meadow, riparian thicket, riparian marsh elements in model significant, p<0.0001. Partial F statistics: In # tree species (> Im): F=75.88; percent emopy closure (In p/q): F=19.00; Variables in model presented in descending rank order, based on partial F statistics; interactions among variables not significant; all all features have clear view to sky); F=15.98. 36.
- elements in model significant, p<0.0001. Partial F statistics: In (p/q) canopy closure: F=43.31; sqrt # tree stems 0-4 cm dblr: F=31.68; moisture Variables in model presented in descending rank order, based on partial F statistics; interactions among variables not significant; all class: F=22.80; soil parent material; F=8.81. 37.



Figure 4.1. Scatter plots of selected correlates of species richness in 10m x 10m quadrats I. Legend: SPM = soil parent material: GF = glacio-fluvial. CT = calcareous till, L = lacustrine. CO = calcareous till: SO = soil order: G = gleysol, B = brunisol, GL = gleyed luvisol, GB = gleyed luvisol: MOIST = soil moisture: SD = seasonally dry depressions. SM = seasonally moist depressions. SW = seasonally wet depressions: pH = soil pH_{water} upper 15 cm soil profile.



Figure 4.2. Scatter plots of selected correlates of species richness in 10m x 10m quadrats II. Legend: CT = forest cover type: l = sugar maple + red, white oak; 2 = sugar maple, no oak, no wet-mesic or wet species: 3 = sugar maple + wet-mesic or wet species. 4 = oak, no sugar maple (1 quadrat), 5 = wet-mesic or wet species, no sugar maple, no oak; %CC = % canopy closure; #TSPECIES = # tree species (>1m); #Tstms0-4cm = # tree stems 0-4 cm dbh.



Figure 4.3. Scatter plots of selected correlates of species richness in 10m x 10m quadrats III. Legend: %stmsACERSACC = % live tree stms (> 1 m) in 10m x 10m quadrat Acer saccharum; %stms0-4cmASACC = % live tree stems 0-4 cm dbh in 10m x 10m quadrat Acer saccharum; %stms4-10cmASACC = % live tree stems 4-10 cm dbh in 10m x 10m quadrat Acer saccharum; %stms10-30cmASACC = % live tree stems 10-30 cm dbh in 10m x 10m quadrat Acer saccharum; %stms10-30cmASACC = % live tree stems 10-30 cm dbh in 10m x 10m quadrat Acer saccharum; %stms10-30cmASACC = % live tree stems 10-30 cm dbh in 10m x 10m quadrat Acer saccharum; %stms10-30cmASACC = % live tree stems 10-30 cm dbh in 10m x 10m quadrat Acer saccharum; %stms10-30cmASACC = % live tree stems 10-30 cm dbh in 10m x 10m quadrat Acer saccharum; %stms10-30cmASACC = % live tree stems 10-30 cm dbh in 10m x 10m quadrat Acer saccharum; %stms10-30cmASACC = % live tree stems 10-30 cm dbh in 10m x 10m quadrat Acer saccharum; %stms10-30cmASACC = % live tree stems 10-30 cm dbh in 10m x 10m quadrat Acer saccharum; %stms10-30cmASACC = % live tree stems 10-30 cm dbh in 10m x 10m quadrat Acer saccharum; %stms10-30cmASACC = % live tree stems 10-30 cm dbh in 10m x 10m quadrat Acer saccharum; %stms10-30cmASACC = % live tree stems 10-30 cm dbh in 10m x 10m quadrat Acer saccharum; %stms10-30cmASACC = % live tree stems 10-30 cm dbh in 10m x 10m quadrat Acer saccharum; %stms10-30 cm dbh in 10m x 10m quadrat Acer saccharum; %stms10-30 cm dbh in 10m x 10m quadrat Acer saccharum; %stms10 cm dbh in 10m x 10m quadrat Acer saccharum; %stms10 cm dbh in 10m x 10m quadrat Acer saccharum; %stms10 cm dbh in 10m x 10m quadrat Acer saccharum; %stms10 cm dbh in 10m x 10m quadrat Acer saccharum; %stms10 cm dbh in 10m x 10m quadrat Acer saccharum; %stm x 10m quadrat Acer



Figure 4.4. Scatter plots of selected correlates of species richness in 10m x 10m quadrats IV. Legend: DISTURB = disturbance absent/present (open or closed trails, canopy gaps, regenerating fields); OMHABS = open microhabitats absent/present (cut or snag canopy gaps, trails, regenerating fields, seeps, riparian meadow, riparian thicket, riparian marsh); FCOVER = % forest cover within a 5km x 5km square, centered on study site; PA = patch area (ha).

Table 4.2. Species richness (10m x 10m quadrats) of soil parent materials, soil orders, soil moisture classes, forest cover types, and disturbance classes. Wilcoxon rank sum tests, by attribute: ****** p<0.01, ******** p<0.0001. Highest mean value in attribute group in **bold**.

Attribute	# Quadrats	# Species Study Area	Mean # Species
1. SOIL PARENT MATERIAL""	<u> </u>		
glacio-fluvial	40	116	27.8
calcareous til)	104	334	42.6
lacustrine	8	104	48.5
calcareous outwash	40	307	55.2
2. SOIL ORDER""			
brunisol	70	280	36.6
gleyed brunisol	26	253	67.0
luvisol	57	258	36.6
gleyed luvisol	31	198	47.6
gleysol	8	92	34.0
3. SOIL MOISTURE CLASS""			
seasonally dry depressions	118	337	34.8
seasonally moist depressions	42	241	53.2
seasonally wet depressions	32	159	56.2
4. FOREST COVER TYPE""			
red oak. no sugar maple	ł	57	57
sugar maple – red oak	54	188	33.6
sugar maple	. 74	305	36.4
sugar maple + wet-mesic/wet	50	287	54.9
wet-mesic + wet, no sugar maple	13	333	63.9
5. DISTURBANCE CLASS"			
human disturbance present (closed or open canopy gaps, lanes, regenerating fields)	61	345	49.2
human disturbance absent	131	355	39.2

i) Edaphic Variables

Edaphic variables that maximally explained observed differences in species richness in 10m x 10m quadrats were soil order ($r^2adj.=0.253$), soil moisture ($r^2adj.=0.247$), and soil parent material ($r^2adj.=0.220$) (Table 4.1). Significant, but less informative variables, were soil pH_{water}($r^2adj.=0.140$), and, in % soil organic matter ($r^2adj.=0.052$). Available calcium (free calcium carbonate – exchangeable calcium) was a significant but weak predictor of overall species richness ($r^2adj.=0.026$).

Quadrats on calcareous outwash parent materials (mean richness = 55.2 species) were approximately twice as rich in species as quadrats on glacio-fluvial parent materials (mean richness = 27.8 species) (Table 4.2). Quadrats on lacustrine parent materials (mean richness = 48.5 species) were more species rich than quadrats on calcareous till (mean richness = 42.6 species). The latter contrast, however, was not significant (Tukey-Kramer HSD test).

Quadrats on gleyed Brunisolic soils (mean richness = 67.0 species) were more species rich than quadrats on gleyed Luvisolic soils (mean richness = 47.6 species), and far more species rich than quadrats on Brunisolic soils (mean richness = 36.6 species). Luvisolic soils (mean richness = 36.6 species). Luvisolic soils (mean richness = 36.6 species). Compared to the species of the spec

Quadrats with seasonally moist depressions (mean richness = 56.2 species) were marginally more species rich than quadrats with seasonally wet depressions (mean richness = 53.2 species), and significantly more species rich than quadrats with seasonally dry depressions (mean richness = 34.8 species)(Table 4.2)(Tukey-Kramer HSD test).

Species richness was positively correlated with percent soil organic matter, soil pH, and available calcium.

Soil moisture and soil parent material collectively explained 35.6% of the variance in species richness in 10m x 10m quadrats (Table 4.1).

Pair-wise interactions among edaphic variables (superscript 2-6), and with stand structure variables (superscript 10-21), were common (Table 4.1). Less common were interactions with human disturbance (superscript 24-29) and landscape variables (superscript 33-35).

ii) Forest Stand Structure Variables

Stand structure variables that maximally explained differences in species richness in 10m x10m quadrats were forest cover type ($r^2adj.=0.277$). # tree species ($r^2adj.=0.268$), % stems sugar maple ($r^2adj.=0.256$), % stems sugar maple 0-4 cm dbh ($r^2adj.=0.254$), # wet-mesic, wet, tree species ($r^2adj.=0.224$), and, # live tree stems 0-4 cm dbh ($r^2adj.=0.196$) (Table 4.1, variables variously transformed). Significant, but less informative, stand structure variables were % canopy closure ($r^2adj.=0.150$), % stems sugar maple 4-10 cm dbh ($r^2adj.=0.099$), % stems sugar maple 10-30 cm dbh ($r^2adj.=0.096$), and, # live tree stems >30 cm dbh ($r^2adj.=0.032$) (variables variously transformed). The number of live tree stems in the 4-10 cm dbh, and 10-30 cm dbh, size class were non-significant predictors of species richness.

Quadrats composed of wet-mesic or wet tree species (mean richness = 63.9 species), or composed of sugar maple and wet-mesic or wet tree species (mean richness = 54.9 species), were significantly more species rich than quadrats composed of sugar maple (no wet-mesic, wet, trees; no red or white oak) (mean richness = 36.4 species) or composed of sugar maple and red or white oak (mean richness = 33.6 species) (Table 4.2) (Tukey-Kramer HSD test).

Species richness was positively correlated with the number of tree species: the number of wet-mesic. wet, trees; the number of live stems 0-4 cm dbh; and, the number of live stems 4-10 cm dbh. Species richness was negatively correlated with the number of live stems >30 cm dbh; the percent stems sugar maple; the percent stems sugar maple 0-4 cm dbh; the percent stems sugar maple 4-10 cm dbh; and, the percent stems sugar maple 10-30 cm dbh.

The number of live tree species (>1m), degree of canopy closure, and type of forest cover, collectively explained 54.9 % of the variance in species richness in 10m x 10m quadrats (Table 4.1). The number of tree stems 0-4 cm dbh, degree of canopy closure, and type of forest cover type

collectively explained 49.8% of the variance in species richness. The latter model achieves a better statistical fit (p < W = 0.92 vs p < W = 0.68) and thus constitutes the "best" set of explanatory variables for this set of environmental variables.

Pair-wise interactions among stand structure variables (superscript 10-21), and with edaphic variables (superscript 2-6), were common (Table 4.1). Less common were interactions with human disturbance (superscript 24-29) and with landscape variables (superscript 33-35). Stand structure variables that interacted less frequently than other variables were # live tree stems >30 cm dbh. % canopy closure, # tree species (>1m), and. % stems 10-30 cm dbh.

iii) Human Disturbance

The disturbance variables that maximally explained differences in species richness in 10m x 10m quadrats were disturbed microhabitats (present/absent) (r²adj.=0.089), open trails (r²adj.=0.089), and open regenerating fields (r²adj.=0.079). Significant, but less informative, variables were closed trails (r²adj.=0.043) and open canopy gaps (r²adj.=0.041) (Table 4.1). Closed canopy gaps did not explain observed differences in species richness.

Quadrats with human disturbance (closed or open canopy gaps, trails, regenerating fields) were significantly more species rich than quadrats with no apparent disturbance (mean richness = 49.2 species versus 39.2 species) (Table 4.2).

Open trails and open regenerating fields collectively explained 12.8% of the variance in species richness in 10m x10m quadrats (Table 4.1).

Pair-wise interactions among disturbance variables (superscript 24-29), and with edaphic variables (superscript 2-6), and with stand structure variables (superscript 10-21), were common (Table 4.1). Open conditions interacted more frequently with other environmental variables than did closed conditions.

iv) Environmental Heterogeneity

The number of microhabitats recorded in a given quadrat explained 5.4% of the variance in species richness in 10m x 10m quadrats (Table 4.1).

Pair-wise interactions occurred with soil parent material (superscript 2), selected stand structure variables (superscript 10-24), and selected disturbance variables (superscript 24-29)(Table 4.1).

v) Landscape Variables

The variables that maximally explained differences in species richness in 10m x 10m quadrats were percent forest cover in the surrounding landscape (r²adj.=0.066), and, patch area (r²adj.=0.019) (Table 4.1). The mean distance to the nearest forest patch (in 45^e arcs) did not explain observed differences in species richness.

Species richness was positively correlated with the mean distance to the nearest forest patch (in 45° arcs), and, negatively correlated with the percent forest cover in the surrounding landscape, and, with patch area.

Pair-wise interactions occurred with soil parent material (superscript 2), soil order (superscript 3), soil moisture (superscript 4), selected stand structure variables (superscript 10-21), selected disturbance variables (superscript 24-29), and landscape variables (superscript 33-35).

vi) Overall Model

The combination of environmental variables that maximally explained differences in species richness in 10m x10m quadrats were the number of tree species (>1m), percent canopy closure, forest cover type, and open microhabitats (cut or snag gap, lane, regenerating field, seep, riparian meadow, riparian thicket, riparian marsh) (r^2 adj.=0.582) (Table 4.1). An alternative model composed of percent canopy closure, number stems 0-4 cm dbh, soil moisture class, and soil parent material, was equally successful in explaining observed differences in species richness (r^2 adj.=0.5495).

235

4.3.1.2 Contribution of Forest Structure

The contribution of stand structure to species richness (Table 4.1) was examined further to clarify the cause and effect relationship among selected variables. Of particular interest was the degree to which "number of tree species", and "number of live stems 0-4 cm dbh", were causal mechanisms of species richness. Non-parametric correlations (Spearman's Rho) with selected stand structure variables, soil moisture, available calcium, and human disturbance were computed with a view to revealing more proximate correlates of species richness (Table 4.3).

The number of tree species in a given 10m x 10m quadrat was negatively correlated with the percent stems that were sugar maple (p<0.0001), and, positively correlated with the percent stems that were wet-mesic, wet, trees (p<0.0001), or, shade intolerant, very shade intolerant, trees (p<0.0001)(Table 4.3). The number of tree species was not correlated with recent human disturbance (p=0.95) or with percent canopy closure (p=0.27).

The percent stems sugar maple in a given 10m x 10m quadrat was negatively correlated with seasonally moist or wet depressions on the forest floor (p<0.0001) and with the percentage of stems that were wet-mesic, wet trees (p<0.0001), or, shade intolerant, very shade intolerant, trees (p<0.0001) (Table 4.3). Sugar maple abundance was negatively correlated with available calcium (p=0.0049), owing to the strong positive correlation between available calcium and percent soil organic matter. Sugar maple abundance was positively correlated with percent canopy closure (p=0.0029) and weakly correlated with the absence of the human disturbance (p=0.0655).

Taken together, these results reveal that the number of tree species in given quadrat increased in the presence of seasonally moist or wet soils and in conditions that favored the establishment of shade intolerant and very shade intolerant trees, and, declined in the presence of sugar maple. This suggests that the number of tree species in a given quadrat is a complex variable that accounts for differences in soil moisture, stand history, and available light. Tree species number is thus a correlate of species richness and not its cause.

	Variables	Corre	lation
X	Å	Spearman Rho	Prob > Rho
# tree species	# species (all life forms)	0.5191	1000:0>
# tree species	% stems sugar maple	-0.5937	1000/0>
# tree species	% stems wet-mesic, wet, tree species	0,4430	<0.000.0>
# trev species	% stems shade intolerant, very shade intolerant, tree species	0.3220	I 000 0>
# tree species	moisture class (1-dry, 2-moist,wet)	0,1382	0.0562
# tree species	% canopy closure	-0.0804	0.2677
# tree species	human disturbance (1=nn, 2=yes)	-0.0044	0.9521
% stems sugar maple	# species (all life forms)	-0.4857	<0.0001
% stems sugar maple	# tree species	-0.6184	<0.0001
% steins sugar maple	% stems wet-music, wet tree species	-0.4521	<0000°>
🍬 sterns sugar maple	% stems shade intolerant, very shade intolerant, tree species	-0.3709	<0:0001>
% stons sugar maple	moisture class (1=dry, 2=moist,wet)	-0.3512	1000'0>
% stems sugar maple	% campy closure	0.2142	6200.0
% stems sugar maple	calcium (cmol/kg)	-0.2137	0.0049

Table 4.3. Selected correlations involving # tree species, % stems sugar maple, and, # tree stems 0-4 cm dbh.

237

	Variables	Corre	lation
×	λ	Spearman Rho	Prob > Rho
% stents sugar maple	human disturbance (1=no, 2=yes)	-0.1332	0,0655
# tree stems 0-4 cm dbh	# species (all life forms)	0.4646	<0.0001
# tree stems ()-4 cm dbh	# Iree species	0.5287	<0.0001
# tree stems 0-4 cm dbh	% stems sugar maple	-0.2503	0,0005
# Iree stems 0-4 en dbh	% stems wet mesic, wet, tree species	0.2659	0.0002
# tree stems 0-4 cm dbh	% stems shade intolerant, very shade intolerant	0.0256	0.7250
# tree stems 0-4 cm dbh	moisture class (1=dry, 2-moist,wet)	0.1602	0.026J
# tree stems 0-4 cm dbh	luman disturbance (1=no, 2=yes)	0.1389	0.0546
# tree stems 0-4 cm dbli	% cattopy closure	-0.0592	0.4150

Table 4.3. Selected correlations involving # tree species, % stems sugar maple, and, # tree stems 0-4 cm dbb (cont'd).

The number of live tree stems 0-4 cm dbh was positively correlated with the number of tree species (p<0.0001) and with the percentage of stems that were wet-mesic, wet, trees (p=0.0002) (Table 4.3). The number of live tree stems in this size class was weakly correlated with recent human disturbance (p=0.0546) and negatively correlated with the percentage of stems that were sugar maple (p=0.0005). Stem number was neither correlated with percent canopy closure (p=0.4150) nor with the percentage of stems that were shade tolerant or very shade tolerant (p=0.7250).

Taken together, these results reveal the number of live tree stems, 0-4 cm dbh, was more strongly influenced by the number of tree species, the percentage of stems that were wet-mesic, wet, trees, and, the percentage of stems that were sugar maple, than by recent and past stand disturbance. In general, the number of stems in this size class increased with increasing soil moisture and declined with increasing sugar maple abundance. This suggests that the number of live tree stems, 0-4 cm dbh, is a complex variable that primarily accounts for differences in soil moisture and available light. The number of tree stems, 0-4 cm dbh, is thus a correlate of species richness and not its cause.

The capacity of each variable to explain observed differences in species richness, therefore, is a function of their correlation with differences in soil moisture and available light. The variance in richness explained by the number of tree species also reflects the influence of past disturbance events and thus explains marginally more variance in species richness than the number of live tree stems 0-4 cm dbh (Table 4.1).

4.3.1.3 Contribution of Soil Fertility

The interaction between available calcium and sugar maple abundance (Table 4.1) was investigated further to clarify the contribution of soil fertility to observed differences in species richness. This analysis was conducted in two contrasting settings.

The first setting was restricted to forest stands on Brunisolic and Luvisolic soils, in view of reported differences in calcium availability in these soils (Hoffman and Acton1974, Gillespie and Acton 1981). Soils with free calcium carbonate in the upper 15 cm of the soil profile (positive reaction
to 0.1N HCl) were excluded from this analysis in order to standardize samples with respect to exchangeable calcium. Exchangeable calcium in these soils ranged from 0.7 to 13.1 cmol/kg.

The results of this analysis are summarized in Figure 4.5. Differences in sugar maple abundance explained 63.2 % of the variance in species richness in 10m x 10m quadrats, when expressed as the percentage of live stems (>1m) that were sugar maple, and 50.0 % of variance in species richness in 10m x10m quadrats, when expressed as the number of live stems (>1m) that were sugar maple.

Exchangeable calcium, in turn, explained 11.1 % of the variance in sugar maple abundance (p=0.054), when expressed as the percentage of live stems (>1m) sugar maple, and 2.9 % of the variance in sugar maple abundance (p=0.200), when expressed as the number of live stems sugar maple. Exchangeable calcium, *sensu stricto*, explained 6.6 % of the variance in species richness (p=0.109) on these soils. Exchangeable calcium is a significant predictor of sugar maple abundance for selected size classes, however, and explains 13% to 19% of the variance in the percentage of live stems (>1m) sugar maple in the 0-4 cm, 4-10 cm, and 10-30 cm, dbh size class (Figure 2.7).

The second setting was restricted to forest stands on Brunisolic soils in order to standardize samples for soil parent material (calcareous till), soil order (Orthic Melanic Brunisol), soil series (Otonabee loam), soil moisture (mesic), forest composition (Cover Type 2, 3), and recent site disturbance (no trails, canopy gaps, regenerating fields). Soils with free calcium carbonate in the upper 15 cm of the soil profile were included in the analysis in order to standardize samples for available calcium (free calcium carbonate + exchangeable calcium) in the upper 15 cm of the soil profile. The gradient in available calcium on these soils ranged from 0.3 cmol/kg to 20.2 cmol/kg. Soils with free calcium carbonate contained significantly higher levels of available calcium than non-reactive soils (16.6 cmol/kg versus 4.1 cmol/kg, respectively)(Wilcoxon rank sum test, not shown).

The results of this analysis are presented in Figure 4.6. Available calcium explained 62.8 % of the variance in sugar maple abundance, when expressed as the percentage of live stems (>1m) that were sugar maple, and, 45.9% of the variance in sugar maple abundance, when expressed as the number of live stems (>1 m) that were sugar maple. However, in contrast, to the first setting, differences in

R2adj.=0.50; p>F=<0.0001 60 60 50 50 # Species # Species 40 40 30 30 20 20 -10 10 Ō 20 40 60 60 100 Ŭ 10 z 50 60 30 40 % Stems >1m Acer saccharum # Stems >1m Acer saccharum R²adj.=0.11; p>F=0.054 R²adj.=0.03; p>E=0.20 60 100 % Stems Acer saccharum Stems Acer saccharum 50 **80** · 40 60 · 30 40 · 20 æ 10 Ħ: Q ٥ ٥ 8 12 4 16 0 4 8 12 16 Available Calcium cmol/kg Available Calcium cmol/kg

R²adj.=0.63; p>F=<0.0001

Figure 4.5. Species richness, sugar maple abundance, and available calcium, in 10m x 10m quadrats on Brunisolic and Luvisolic soils overlying calcareous till. Soils with free calcium carbonate in upper 15 cm of soil profile excluded from analysis (see text). Quadrats with apparent human disturbance excluded from analysis. Forest cover = cover type 2 (sugar maple + red or white oak) and cover type 3 (sugar maple, no red or white oak, no wet-mesic, wet, species). Regression statistics based on transformed data. Available calcium explains 6.6% of variance (p=0.11) in species richness (not shown). N-29 quadrats in 7 forest patches.

R2adj.=0.057; p>E=0.71

R²adj.=0.063; p>F=0.81



Figure 4.6. Species richness, sugar maple abundance, and available calcium, in 10m x 10m quadrats on Brunisolic soils overlying calcareous till. Soils with free calcium carbonate in upper 15 cm of soil profile included in analysis. Quadrats with apparent human disturbance excluded from analysis. Forest cover = cover type 2 (sugar maple + red or white oak) and cover type 3 (sugar maple, no oak, no wet-mesic, wet species). N=17 quadrats in 3 forest patches. Available calcium explains 6.0% of variance in species richness (p=0.76).

sugar maple abundance do not explain observed differences in species richness (p=0.71). As in the first setting, available calcium does not explain species richness on these soils (p=0.76).

The preceding results reveal that differences in exchangeable and available calcium do not explain observed differences in species richness. Or at least, not directly. The degree to which species richness is explained by differences in sugar maple abundance, and the degree to which sugar maple abundance is explained by differences in calcium availability in the upper 15 cm of the soil profile, depends on the setting in which the analysis is conducted and on the measure of sugar maple abundance that is used.

Available light, stand structure, and edaphic factors were examined further to reveal potential mechanisms that may explain the preceding results. The results of this analysis are summarized in Figures 4.7, 4.8, and 4.9.

The mean number of species, percent stems sugar maple, percent taxa shade tolerant herbs, and percent shade intolerant herbs, are compared in Figure 4.7. The mean number of species, and the mean percentage of taxa that were shade intolerant herbs, were significantly higher on Brunisolic soils whereas the mean percentage of live stems that were sugar maple, and the mean percentage of taxa that were shade tolerant herbs, were significantly higher on Luvisolic soils. This pattern suggests that more light is available in forest stands on Brunisolic soils than on Luvisolic soils, and, that the difference in species richness on these soils is due in part to the deep shade cast by maple saplings and trees. Observed differences in canopy closure were not significant (mean % canopy closure: 89.0% on Brunisolic soils versus 91.1% on Luvisolic soils, p=0.93. Wilcoxon rank sum test, not shown).

The abundance of sugar maple in selected size classes is presented in Figure 4.8. The data reveal that the number of sugar maple stems in a 10m x 10m quadrat is consistently higher on Luvisolic soils than on Brunisolic soils. The data also reveal that stands on Brunisolic soils have significantly fewer stems in the 10-30 cm dbh size class and no stems in the >30 cm size class. The most likely explanation for the latter patterns is that stands on Brunisolic soils were more intensively, or more



Figure 4.7. Comparison of mean plant response in 10m x 10m quadrats on soils of contrasting fertility I. Legend: B = Brunisolic soils. L = Luvisolic soils, overlying calcareous till. Soils with free calcium carbonate in upper 15 cm of soil profile excluded from analysis. Quadrats with apparent human disturbance excluded from analysis. N = 29 quadrats in 7 forest patches. Error bars: one standard error mean. Means: # SPECIES: B=39.6, L=23.9; % STEMS SUGAR MAPLE: B=15.1, L=76.7; % TAXA SHADE TOLERANT HERBS: B=42.1, L=56.1; % TAXA SHADE INTOLERANT HERBS: B=25.7, L=10.6. P-value: Wilcoxon rank sum test.



Figure 4.8. Comparison of mean plant response in 10m x 10m quadrats on soils of contrasting fertility II. Legend: B = Brunisolic soils. L = Luvisolic soils. overlying calcareous till. Soils with free calcium carbonate in upper 15 cm of soil profile excluded from analysis. Quadrats with apparent human disturbance excluded from analysis. N = 29 quadrats in 7 forest patches. Error bars: one standard error mean. Means: # SUGAR MAPLE STEMS 0-4 cm: B=9.6, L-17.1; # SUGAR MAPLE STEMS 4-10 cm: B=0.4, L=7.0; # SUGAR MAPLE STEMS 10-30 cm: B=0.3; L=1.9; # SUGAR MAPLE STEMS >30 cm: B=0.0, L=0.6. P-value: Wilcoxon rank sum test.



Figure 4.9. Comparison of mean plant response in 10m x 10m quadrats on soils of contrasting fertility III. Legend: B = Brunisolic. L = Luvisolic soils. overlying calcareous till. Soils with free calcium carbonate in upper 15 cm of soil profile excluded from analysis. Quadrats with apparent human disturbance excluded from analysis. N = 29 quadrats in 7 forest patches. Error bars: one standard error mean. Means: # SUGAR MAPLE STEMS: B=10.3, L=26.5: AVAIL-ABLE CALCIUM: B=4.3, L=6.9; pH_{water}: B=6.5, L=6.4; % SOIL ORGANIC MATTER: B=5.9, L=9.0. P-value: Wilcoxon rank sum test.

recently, logged than stands on Luvisolic soils. While such a difference in stand history may explain the relative scarcity of large diameter stems on Brunisolic soils, it does not readily explain the greater number of species, and the greater number of shade intolerant herbs, since the canopy in each stand in this analysis was intact. A more likely explanation, therefore, is that the number of sugar maple stems in stands on Brunisolic soils was not sufficient to cast a deep shade over the entire 10m x 10m quadrat. This explanation is in keeping with the relative and absolute difference in mean sugar maple abundance, and, with the difference in the mean percentage of taxa that were shade intolerant herbs (Figures 4.7 and 4.9). It also provides an explanation for the non-significant relationship between sugar maple abundance and species richness reported in Figure 4.6.

Selected soil properties and the mean number of sugar maple stems are presented in Figure 4.9. The mean number of sugar maple stems, the mean concentration of available calcium, and the mean percent soil organic matter, were significantly higher on Luvisolic soils than on Brunisolic soils. The difference in mean pH_{water} was not significant. The correspondence between available calcium and absolute sugar maple abundance on these soils is consistent with the calcium - mortality hypothesis proposed by Kobe *et al.* (1995) and Kobe (1996) (see Section 2.4, element iii). Other factors may contribute to the observed difference in sugar maple abundance on these soils, however, since the strength of the relationship between available calcium and sugar maple abundance depends on the parameters included in the analysis (non-significant when the analysis includes stands on both Luvisolic and Brunsiolic soils. Figure 4.5: significant when the analysis is restricted to Brunisolic soils. Figure 4.6).

Soil organic matter was positively and significantly correlated with the number of sugar maple stems in both analytical settings (p=0.042 when the analysis includes Brunisolic and Luvisolic soils: p=0.0007 when the analysis was restricted to Brunisolic soils) (supplementary correlation analysis not shown). The most apparent reason for this association is the high cation exchange capacity of soil organic matter (Brady 1990). In keeping with this property, the relationship between percent soil organic matter and available calcium on these soils was linear, positive, and strong (r²adj.=0.45 when the analysis includes Brunisolic and Luvisolic soils: r²adj.=0.68 when the analysis is restricted to Brunisolic soils). Soil organic matter may therefore contribute to differences in sugar maple abundance by making calcium differentially available to surface roots.

Soil organic matter also facilitates the retention of soil moisture (Brady 1990) and thus may contribute to differences in sugar maple abundance by facilitating the germination, establishment, and persistence of sugar maple seedlings. The moisture retention hypothesis is not supported by data, however, since the relationship between percent soil organic matter and the mean cover class of sugar maple seedlings (<im) was negative in both analytical settings (supplementary linear regression analyses not shown).

The relationship between percent soil organic matter and species richness was marginally significant (p=0.062) and negative when the analysis included both Brunisolic and Luvisolic soils, and, markedly non-significant (p=0.605) and positive when the analysis was restricted to Brunisolic soils (supplementary correlation analyses not shown). Differences in soil organic matter content, therefore, do not appear to have contributed to observed differences in species richness.

4.3.1.4 Contribution of Patch Isolation and Patch Size

The contribution of patch isolation and patch size to differences in species richness was evaluated by multiple linear regression (Table 4.1). Patch isolation, when expressed as the percent forest cover within a 5 km x 5km square centered on the study site, explained 6.6 % of the variance in species richness. In general, the species richness in a given 10m x 10m quadrat declined as the percent forest cover in the surrounding landscape increased (Figure 4.4). Patch isolation, when expressed as the mean distance to the nearest eight woodlots, measured in 45° ares, was markedly non-significant (p=0.32). Significant pair-wise interactions were detected between patch isolation and selected edaphic, stand structure, and disturbance variables. Interactions with important, and complex, explanatory variables such as soil order, and forest cover type, suggest that patch isolation, *per se*, does not make an independent contribution to species richness in this study.

Patch size, when expressed as the area of the forest patch in which the study site was situated, explained 1.9% of the variance in species richness. In general, the species richness in given quadrat

declined as the size of the forest patch increased. Significant pair-wise interactions were detected between selected edaphic, stand structure, and landscape variables. The interaction with soil parent material, soil order and forest cover type, in particular, suggest that patch size, *per se*, does make an independent contribution to species richness in this study.

4.3.1.5 Contribution of Microhabitats

The contribution of microhabitats to observed differences in species richness was evaluated at the quadrat (Table 4.4), forest patch (Table 4.5), and landscape (Table 4.6) spatial scale. The objective of this analysis was to determine the degree to which a given microhabitat contained species that did not occur elsewhere in the sampled area.

The microhabitats that contributed maximally to species richness at the quadrat scale were open lane/roads (mean # unique species = 24.3); open regenerating fields (mean # unique species = 19.33); closed, seasonally dry, forest floors (mean # unique species = 16.60); open riparian meadows (mean # unique species = 15.33); and, closed, seasonally moist forest floors (mean # unique species = 15.30 species) (Table 4.4). Microhabitats that occasionally contributed additional species were open and closed tree pits (mean # unique species = 0.50 and 0.54, respectively); open and closed stumps (mean # unique species = 0.43 and 0.90, respectively); open and closed logs (mean # unique species = 0.83 and 1.87, respectively). Only two of twenty-three evaluated microhabitats did not make a significant contribution to species richness at the quadrat spatial scale (open tree pits, all moisture classes; and, open stumps). Seventeen microhabitats were not evaluated (minimum sample size criterion for Wilcoxon signed rank sum test not satisfied).

The microhabitats that contributed maximally to species richness at the patch scale were closed, seasonally dry, forest floors (mean # unique species = 17.05), open regenerating fields (mean # unique species = 12.5), and open lanes/roads (mean # unique species = 9.00) (Table 4.5). Microhabitats that contributed a modest but significant number of species at the patch scale were closed, seasonally moist, forest floors (mean # unique species = 6.13); open, seasonally dry, canopy gaps (mean # unique species = 4.40); closed, seasonally wet, forest depressions (mean # unique

Table 4.4. Contribution of microhabitats to species richness in 10m x 10m quadrats. Legend: %SA = % of sampled area (19.200 m²). # Quads: number of quadrats in analysis; Mean difference = mean difference in species richness: (# species in quadrat) - (# species in quadrat - # species unique to microhabitat); paired t-tests when distribution of differences normal; Wilcoxon signed rank sum tests otherwise. NT = not tested (see text). Quadrats in which microhabitat is the only microhabitat excluded from analysis.

Microhabitat	% SA	# Quads	Mean Difference	p>₁t
seasonally dry forest floor-cc	54.3	102	16.60	0.0001
seasonally dry forest floor-oc	6.8	23	11.30	0.0001
seasonally moist forest floors-ee	3.0	21	15.30	0.0001
seasonally moist forest floors-oc	3.0	9	9.67	0.0028
seasonally moist forest depressions-co	4,1	20	5.35	0.0001
seasonally moist forest depressions-oc	0.2	2	4.00	NT
seasonally wet forest floors-cc	0.6	3	5.33	NT
seasonally wet forest floors-oc	1.0	-4	5.75	0.0431
seasonally wet forest depressions-cc	4.0	17	5.47	0.0001
seasonally wet forest depressions-oc	1.0	4	4.50	NT
seep-cc	0.9	3	4.67	NT
seep-oc	0.3	l	12.00	NT
seasonally dry gap-cc	2.2	3	9.00	NT
seasonally moist gap-cc	0.5	2	5.00	NT
pit/mound complexes-cc	5.6	64	3.66	0.0001
pit/mound complexes-oc	0.6	7	6.30	0.0378
mound-cc	3.6	60	3.12	0.0001
mound-oc	0.4	5	5.20	0.0466
pits-cc (all moisture classes)	1.8	46	0.54	0.0001
pits-oc (all moisture classes)	0.2	6	0.50	1.0000
seasonally dry pit-cc	1.5	41	0.38	0.0040
seasonally dry pit-oc	0.1	-4	0.75	NT
seasonally moist pit-ce	0.3	2	0.50	NT

Microhabitat	% SA	≓ Quads	Mean Difference	p> t
seasonally wet pit-ce	0.2	3	[.33	NT
seasonally wet pit-oc	0.1	I	0.00	NT
log-cc	0.5	45	1.87	0.0001
log-oc	0.2	12	0.83	0.0310
stump-ec	0.2	30	0.90	0.0001
stump-oc	0.1	7	0.43	0.0781
raised root mat-ce	1.3	28	3.79	1000.0
raised root mat-oc	0,6	12	4.25	0.0040
stone-cc	<0.1	l	4.00	NT
lane/road-ce	2.3	1	6.18	0.0003
lane/road-oc	L.3	8	24.13	0.0107
diteh-ce	0.3	4	5.00	NT
ditch-oc	<0.1	2	2.50	NT
regenerating field-co	1.3	5	[4.20	0.0351
regenerating field-oc	1.3	6	19.33	0.0014
riparian meadow-oc	0.7	3	15.33	NT
riparian marsh-oc	0.4	3	2.33	ΝΤ
riparian thicket-oc	0.1	l l	2.00	NT

Table 4.4. Contribution of microhabitats to species richness in 10m x 10m quadrats (cont'd).

Table 4.5. Contribution of microhabitats to species richness in surveyed forest patches. Legend: %SA = % of sampled area (19,200 m²). # Patches: number of forest patches in analysis; Mean difference = mean difference in species richness: (# species in patch) - (# species in patch - # species unique to microhabitat); paired t-tests when distribution of differences normal: Wilcoxon signed rank sum tests otherwise. NT = not tested (see text). Patches in which microhabitat is the only microhabitat excluded from analysis.

Microhabitat	% SA	# Patches	Mean Difference	p>it
seasonally dry forest floor-ce	54.3	21	17.05	0.0001
seasonally dry forest floor-oc	6.8]4	4,40	0,0010
seasonally moist forest floors-ce	3.0	8	6.13	0.0040
seasonally moist forest floors-oc	3.0	6	1.83	0.2500
seasonally moist forest depressions-cc	4.1	6	1.12	0.2500
seasonally moist forest depressions-oc	0.2	l	2.00	NT
seasonally wet forest floors-cc	0.6	2	2.50	NŤ
seasonally wet forest floors-oc	1.0	2	1.00	NT
seasonally wet forest depressions-cc	4.0	7	3.00	0.0167
seasonally wet forest depressions-oc	۱.0	2	2.00	NT
seep-cc	0.9	2	2.50	NT
seep-oc	0.3	1	7.00	NT
seasonally dry gap-cc	2.2	3	[.00	NT
seasonally moist gap-cc	0.5	1	2.00	NT
pi/mound complexes-cc	5.6	20	2.30	0.0001
pi∂mound complexes-oc	0.6	5	1.40	0.0400
mound-ce	3.6	20	1.35	0.0001
mound-oc	0.4	5	1.20	0.1250
pits-cc (all moisture classes)	1.8	15	0.33	0.1250
pits-oc (all moisture classes)	0.2	5	0.20	0.5000
seasonally dry pit-cc	1.5	15	0.27	0.5000
seasonally dry pit-oc	0.1	4	0.50	NT

Microhabitat	% SA	# Patches	Mean Difference	p> t
seasonally moist pit-ce	0.3	2	0.00	NT
seasonally wet pit-ee	0.2	2	0.00	NT
seasonally wet pit-oc	0.1]	0,00	NT
log-cc	0.5	L4	0.93	0.0310
log-oc	0.2	7	0.71	1.1250
stump-cc	0.2	10	0.400	0.2500
stump-oc	0.1	4	0.00	NT
raised root mat-co	1.3	9	1,68	0.0630
raised root mat-oc	0.6	6	2.33	0.1250
stone-cc	<0.1	:	0.00	NT
lane/road-cc	2.3	6	1.33	0.0606
lane/road-oc	i.3	5	9.00	0.0390
ditch-cc	0.3	2	0.50	NT
ditch-oc	<0.i	2	00.1	NT
regenerating field-cc	1.3	2	5.00	NT
regenerating field-oc	1.3	2	12.5	NT
riparian meadow-oc	0.7	: :	0.00	NT
riparian marsh-oc	0.4	I	3.00	NT
riparian thicket-oc	0.1	1	3.00	NT

Table 4.5. Contribution of microhabitats to species richness in surveyed patches (cont'd).

Table 4.6.	Contribution of	microhabitats	to species	richness at	t the l	andscape	scale.	Legend: %	6
SA = perc	ent total surveye	d area (19.200	m ²)						

Microhabitat	% SA	# Species Restricted to Habitat
seasonally dry forest floors-cc1	54.3	16
seasonally dry forest floors-oc ²	6.8	б
seasonally moist forest floors-cc ³	3.0	5
seasonally wet forest floors-cc4	0.6	1
seasonally wet forest depressions-cc5	4.0	3
seep-oc ⁶	0.3	
mound-ce ⁺	3.6	l
log-cc ⁴	0.5	l
raised root mat-oc"	1.3	1
lane/road-oc"	1.3	6
regenerating field-ce"	1.3	1
regenerating field-oc12	1.3	8
riparian marsh-oc ¹³	0.4	!

Notes:

 Botrychum multifidum, Carex communis, Carex platyphylla, Carex woodii, Chimaphila umbellata, Crataegus species #1, Crataegus species #2, Geum allepicum, Hieracium auramiacum, Lonicera hirsuta, Monotropa hypopithys, Panax quinquefolium, Rhamnus alnifolia, Sanicula trifuliata, Veronica officinalis, Viola rostrata.

 Calystegia sepium, Carduus acanthoides, Lohelia inflata, Nepeta cataria, Panicum capillare, Triosteum aurantiacum.

- 3. Amelanchier arhorea, Cypripedium calceolus, Phegopteris connectifis, Stellaria longifolia, Typha latifolia.
- 4. Aster umbellatus.
- 5. Carex pseudo-cyperus. Cicuta bulbifera. Osmunda regulis.
- 6. Carex crinita.
- 7. Monarda fistulosa.
- 8. Echinocystis lobata.
- 9. Sicyos angulatus.
- 10. Ambrosia artemisiifolia, Carduus mitans, Festuca pratensis, Potentilla norvegica. Verbena hastata, Verbena urticifolia.
- 11. Rudbeckia hirta.
- Antennaria neglecta, Aster ericoides, Carex prairea. Festuca arundinacea. Festuca rubra. Lycopodium tristachyum. Rhus typhina, Salix petiolaris.
- 13. Rumex orbiculatus.

species = 3.00); closed pit/mound complexes (mean # unique species = 2.30); open pit/mound complexes (mean # unique species = 1.40); closed tip-up mounds (mean # unique species = 1.35); and, closed logs (mean # unique species = 0.93 species). Only nine of twenty evaluated habitats made a significant contribution to species richness at the patch scale. Twenty-one microhabitats were not evaluated (minimum sample size criterion for Wilcoxon signed rank sum test not satisfied).

Thirteen microhabitats contributed to species richness at the landscape scale (i.e. contained species that were not recorded in any other habitat)(Table 4.6). The habitat that contributed the greatest number of species was closed seasonally dry forest floor (16 species). Habitats that contributed an intermediate number of species were open regenerating fields (8 species); open lanes/roads (6 species); open, seasonally dry, forest floors (open canopy gaps)(6 species); and, closed seasonally moist forest floors (5 species). Habitats that contributed a minor number of unique species at the landscape scale were closed, seasonally wet, forest depressions (3 species); closed, seasonally wet, forest floors (1 species); open seeps (1 species); closed mounds (1 species); closed logs (1 species); open raised root mats (1 species); closed regenerating fields (1 species); and, open riparian marsh (1 species).

Taken together, these results suggest that light is more limiting than moisture for many plants in these forests. Open disturbed habitats were particularly strong contributors to species richness at the quadrat scale, whereas, closed dry forest floors were the strongest contributors to species richness at the patch and landscape spatial scale. While closed seasonally moist forest floors and seeps were important contributors to species richness at the quadrat scale, moist and wet habitats were modest to weak contributors to species richness at larger spatial scales. Habitats created by natural disturbance (tip-up mounds, tree pits, logs, stumps, raised root mats) were typically weak contributors to species richness at all spatial scales.

These results may depend in part on differences in sampled area. Habitats in these forests were typically closed and seasonally dry (Table 4.7). Open microhabitats occupied 16.5% of the sampled area and were typically dry, whereas, seasonally moist or wet habitats occupied 21.3% of the

Table 4.7. Comparison of microhabitats by moisture and canopy closure class. Legend: % SA = % of total surveyed area (19,200 m²).

Microhabitat	# Quadrats	Mean Area (m ²)	Tota	l Area
	Present	per Quadrat When Present	m²	% SA
I. MOISTURE STATU	S			
Dry Habitats	188	80.3	15,105	78.7
closed canopy	170	76.8	13.051	68.0
open canopy	-43	47.8	2.054	10.7
Moist/Wet Habitats	71	57.7	4.095	21.3
closed canopy	58	51.5	2.986	15.6
open canopy	19	58.4	1,109	5.8
II. LIGHT STATUS		•		
Closed Habitats	174	92.2	(6,0 37	83.5
dry habitats	170	76.8	13.051	68.0
moist/wet habitats	58	51.5	2.986	15.6
Open Habitats	47	67.3	3,163	16.5
dry habitats	43	47.8	2,054	10.7
moist/wet habitats	19	58.4	1.109	5.8

sampled area and were typically closed. These patterns increase the probability of recording restricted species in both closed - dry and closed - moist habitats. Passive sampling effects, however, cannot account for the tendency for open lane/roads and open regenerating fields to contribute more to species richness than their closed counterparts, since the open phase of these habitats occupied an equal or smaller percentage of the total sampled area than the closed phase, at each spatial scale. Nor, for similar reasons, can passive sampling account for the tendency for open references than their closed root mats, to contribute more to species richness than their closed root mats, to contribute more to species richness than their closed seasonally moist floors to consistently contribute more to species richness than the open phase.

4.3.2 Plant Trait Correlates of Species Richness

4.3.2.1 Generalized Lincar Regression Models

Plant trait correlates of species richness in 10m x 10m quadrats are reported in Table 4.8. in relation to the following groupings: mode of dispersal. life history, provenance, life form, habitat affinity, shade tolerance, moisture affinity, and, overall model. Scatter plots of the most influential dispersal correlates are presented in Figures 4.10 and 4.11.

i) Mode of Dispersal

Modes of dispersal that maximally explained observed differences in species richness in 10m x 10m quadrats were dispersal by animal ingestion (r²adj.=0.397), dispersal by wind (r²adj.=0.334), dispersal by unassisted means (r²adj.=0.317), and, dispersal by multiple modes (r²adj.=0.246)(Table 4.8). Modes with intermediate explanatory power were dispersal by prolonged dormancy (r²adj.=0.199), dispersal by ants (r²adj.=0.152), and, dispersal by animal adhesion (r²adj.=0.114). Modes with minimal explanatory power were dispersal by mechanical expulsion (r²adj.=0.060) and dispersal by vegetative expansion (r²adj.=0.050).

Species richness was positively correlated with dispersal by animal adhesion, wind, prolonged dormancy, unassisted means, and multiple modes, and, negatively correlated with dispersal by

variables = 7a taxa in quantat with designated attri- description of dispersal modes. All models parsing	oue, oce noe nijous hased on	a lar uncrea Mallow's (p criterion.		alidnica, acc	
Model	Response Variable	Ч	p-Value	r² adj.	rW>q	relationship
1. MODE OF DISPERSAL ²						
animal ingestion ^{4,4,946,15,17,18,17,18,17,18,10}	sqrt # spp	126.67	0.0001	0.3969	0.81	onjagino
w. 7: at 2: a an a a a a a a a a a a a a a a a a a	dds și tibs	25.65	0.0001	0.1143	0.22	positive
tin(s ² /4, 0.15, 17, 22, 24, 24, 25, 36, 37, 75, 71, 72, 43	dds # 11bs	35.23	0.0001	0.1520	0.57	negative
WINI+10,15,10,124,34,36,46,46,46,46,46,46,46,46,46,46,46,46,46	surt # spp	96.76	0.0001	6666.0	0.79	positíve
prolonged domancy4.8.8.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.	sqrt # spp	38.56	0.0001	0.1992	08.0	positive
mechanical expulsion ⁴ 5.7.1.5 at 5.1.0.2 to 10.10.10 th	dds # ubs uj	15%	0,0026	0,0603	0.23	negalive
unassisted means ^{17,0,0,0,11}	dds # ubs	69'68	0.0001	0,3171	0.37	positive
multiple modes ^{3,4} w.1.1.8.16 (7,00,12,22,4,26,12,19,40,14,14,14,14)	dds ∦ ⊔bs	63.37	100010	0,2462	0.82	positive
vegetative expansion ^{4,5,0} ,0,15,17,30,38,29,99,41,42,41	dds # µbs	10.01	0.0011	0.0495	0.59	negative
MODEJ, 1 (HERJS) ¹² ; unassisted means + prolonged dormancy + ant + animal ingestion	dds #	61.35	10000	0.5583	0.95	
MODEL 2 (ALJ, LIFE FORMS) ¹³ ; unassisted means + animal conveyance (ants + seed eaching by birds and mammals) + prolonged dormancy	dds #	83.61	1000.0	0.5647	£9°0	

Table 4.8. Plant trait correlates of species richness in 10m x 10m quadrats. Generalized linear regression (GLM) models, by row. Predictor variables = % taxa in quadrat with designated attribute. See notes for interactions among predictor variables: see Chanter 3.2.1.1 for

258

Madel	Kesponse Variable	<u>ن</u>	p-Value	ને થતો.	p <w'< th=""><th>relationship</th></w'<>	relationship
2. LIFE HISTORY ¹⁴						
RF,54,11;46,08 56 56,92,82,52,42,52,51,91,81,71,61,81,71,71,71,71,11,11	dds # ribs	9.52	0,0023	0,0427	0.80	positive
biennia (************************************	sqrt # spp	35.63	0.0001	0.1535	0.03	positive
и, ст. њ. њ. ес. 7., 81, 12, 92, 92, 92, 12, 12, 91, 81, ел. 81, 11, 181, 87, 93, 64, 14, 16, 16, 16, 16, 16, 1	sqrt # spp	29,91	0.0001	0.1315	0.51	negative
MODEL: % taxa perenaial	sqrt # spp	29.91	0.0001	0.1315	0.51	ncgative
3. PROVENANCE ¹⁴						
10011VC 34.7.8.10.15.16.17.21.25.12.25.25.25.25.20.11.36	dds #	70.82	1000'0	0,2677	0.67	hegative
alienta (2000) attacts is a 25 period attact	dds #	t-č.:ôð	0.0001	0.2555	0.36	positive
MODEL: % taxa native	dds #	70.82	0.0001	0.2677	0.67	positive
4. LUFE FORM ¹⁴						
on, et , ite an see see see re, ite on , et ou see out	shtt# spp	65.69	0.0001	0.2632	0.76	negative
shrath* 15.16.17.10.19.24.20.30.36.30	sqrt # spp	2.03	0.0087	0.0306	0.14	negative
EN CL, EN CL, EN CL, AC, CL, AC, CL, AC, CL, AL, CL, AL	sqrt # spp	32.91	0.0001	0.1783	0.04	positive
fern ^{1,} 15,17,18,19,24,22,34,35,15	dds∦ubs	011	0.0009	0.0549	0.03	negative
fem ally ^{44,79,10,15,15,17,18,19,20,21,26,14,15,59,40,41,42,43}	ln syrt # spp	3.10	0,0800	0.0109	0.003	positive
BEASS 1.1 4.1 4.14.14.20 24.28.15	sqrt # spp	72.72	0.0001	0.2730	0.33	positive

Table 4.8. Plant trait correlates of species richness in 10m x 10m quadrats (cont'd).

250

1						
Madel	Response Variable	ند.	p-Value	r ³ adj.	p <w'< th=""><th>relationship</th></w'<>	relationship
herb ⁷ .4.1.24.26.46.47.16.36.42	sqrt # spp	31,37	0.0001	0,1372	0.73	positive
MODEL (ALL TAXA) ²⁷ : % tree+% vine	dds #	38.41	0.0001	0.3373	0.56	
5. HABUAT AFFINITY"						
forust ¹⁴ .5.4.10,01,15.17,02,22,25,25,25,20,50,60,61,51,54,54	dds #	68.63	0.0001	0.2615	0,45	negative
FEEDERACTIC/IC/IC/82/221/2/M//21/21/21/Prefitedor 15000	dds ∦ 11bs	16.04	1000.0	0.0730	0,07	negative
opens + forest (sort % "open + forest") ^{(0,10,10,10,10,10,10,10,10,10,10,10,10,10}	dds ⊭ µbs	90'2†	0.0001	6,1943	0.10	positive
th,۲.۲.۵ مغریف المرابط ا	dds #	121.85	0.0001	0.3875	0,04	positive
MODE3. (ALL TAXA) ³¹ : % open + % forest	#spp	68.56	0.0001	0.4143	0.03	
6. SHADE TOLERANCE ¹⁴						
shade tolerance class 1 (very shade toleram)\sits.m.o.20,20,20,20,20,20,00,00	dds # 11ps	53.02	0.0001	0.2140	0,14	negative
shade tolerance class 2 52.24,16, iu. 91, 51,16	dds ∦ 11µs	40.18	0.0001	0,1702	0.28	negative
shade tolerance class 345,15,75,15,77,16,17	sqrt # spp	5.20	0.0237	0.0215	0.19	negative
shade tolerance class 4 ^{1,4,5} ,6,16,12,12,02,26,26,17	sqrt # spp	20.19	0,0001	0,0913	10.0	positive
shade tolerance class 5 (very shade intolerant) ^{1,5,6,7,10} to 12,11,15,46	surt # spp	80.27	0.0001	6,2933	0.46	positive

Table 4.8. Plant trait correlates of species richness in 10m x 10m quadrats (cont*d).

260

					•	
Model	Response Variable	1	p-Value	r² adj.	P <w'< td=""><td>relationship</td></w'<>	relationship
MODEL (ALL TAXA) ³⁸ : % shade tolerance class 5 + % shade tolerance class 1	dds # ubs	46.92	1000.0	0.3247	0.47	
7. MOISTURE AFFINITY ¹⁴						
obligate upland ^{4,15,17,27,24,26,29,40,42,41}	dds # ubs	66.52	0.0001	0.2554	0.77	neyative
facultative upland ^{us, to,15} ,17,21,78,50,11,12,17	sqrt # spp	71.35	1000.0	0.2692	6.65	negative
facultative ^{ta} tantists.ss.st.st.as.as.	dds # ubs	90'0	08230	0,0051	0.02	•
facultative wetland ^{4.5,6,8} ,18,11,35,57,25,24,26,19,40,41,44	sqr # spp	85.58	1000'0	0.3014	0.83	positive
obligate wetland ^{14,3,6,7,8} in 11,25,07,22,25,24,19,40,41,32	dds # 11bs	52.92	0.0001	0.2137	0.34	positive
MODEL: % taxa facultative wetland	sılıt # spp	85.58	0.0001	0.3014	0.83	positive
8, OVERALL MODEL: PLANT ATTRIBUTES ¹⁴						
% taxa herbs dispersed by unassisted means + % taxa herbs dispersed by protonged dormancy + % taxa herbs dispersed by ants + % taxa herbs dispersed by animal ingestion ¹²	dds #	61.35	1000'0	0.5583	0.95	
Notes: L. Shapiro-Wilk W test for normality of residuals (residuals)	duals norma! wh	cu p≥0.05).			-	

Table 4.8, Plant trait correlates of species richness in 10m x 10m quadrats (cont'd).

Predictor variables are the percentage of taxa in a given 10m x 10m quadrat that are herbs dispersed by designated mode. Percentages for prolonged dormancy and mechanical expulsion were transformed (In p/q) to achieve normal residuals.

m.

Significant interaction with % taxa that are lierbs dispersed by animal injection. Significant interaction with % taxa that are herbs dispersed by animal adhesion. च

Notes (cont'd):

- 5. Significant interaction with % taxa that are herbs dispersed by ants.
- 6. Significant interaction with % taxa that are herbs dispersed by wind.
- Significant interaction with $\ln (p/q)$ taxe that are herbs dispersed by prolonged dormancy.
 - Significant interaction with In (p/q) taxa that are herbs dispersed by mechanical expulsion. зó
 - 9. Significant interaction with % taxa that are herbs dispersed by unassisted means
 - 10. Significant interaction with % taxa dispersed by multiple modes.
- Significant interaction with % taxa dispersed by vegetative expansion.
- elements in model significant, p<0.05. Partial F statistics: % unassisted means: F=32.06; % prolonged dormancy: F=25.17; % ant: F=14.44; 12. Variables in model presented in descending rank order, based on partial F statistics; interactions among variables not significant; all % unimal ingestion: F=6.16.
 - elements in model significant, p<0.05. Partial F statistics: % unassisted arcans: F-43.59; % animal conveyance (i.e., dispersal by ants or Variables in model presented in descending rank order, based on partial P statistics; interactions among variables not significant; all by seed caching): F=29.12; % protonged dormancy: F=24.25. ň
 - 14. Predictor variables are the percentage of taxa in a given 10m x 10m quadrat with the specifical attribute.
- Significant interaction with % taxa with an annual life history.
- 16. Significant interaction with % taxa with a bienuial life history.
- 17. Significant interaction with % taxa with a percunial life history.
- 18. Significant interaction with % tasa native.
 - 19. Significant interaction with % taxa alien.
- 20. Significant interaction with % taxa trees.
- 21. Significant interaction with % taxa shrubs.
 - 22. Significant interaction with % taxa vines.
- 23. Significant interaction with % taxa ferns.
- 24. Significant interaction with % taxa fern allies (hp/q).
 - 25. Significant interaction with % taxa grasses.
- 26. Significant interaction with % taxa herbs.
- 27. Variables in model presented in descending rank order, based on partial F statistics, interactions among variables not significant; all elements in model significant, p<0.05. Partial F statistics: % trees: F~37.94; In p/q vines: F=19.84.
 - Vielitieres de modes significant, provez, i antar i sonstrua. En teces i na crea un p 30. Otraticieres interestina mide of secondade an affician for Martin barter bablance
 - 28. Significant interaction with % taxa with an affinity for "forest" habitats.
- 29. Significant interaction with % taxa with an affinity for "forest + open" habitats. 30. Significant interaction with % taxa with up affinity for "open + forest" habitats.

31. Significant interaction with % taxa with an affinity for "open" habitats.
32. Variables in model presented in descending rank order, based on partial F statistics; interactions among variables not significant; all
elements in model significant, p<0.05. Partial F statistics: % affinity for "open" habitats: F=50.57, % affinity for "forest" habitats: F=9.69.
33. Significant interaction with % taxa with shade tolerance class 1.
34. Significant interaction with % taxa with shade tolerance class 2.
35. Significant interaction with % taxa with shade tolerance class 3.
36. Significant interaction with % taxa with shade tolerance class 4.
37. Significant interaction with % taxa with shade tolerance class 5.
38. Variables in model presented in descending rank order, based on partial F statistics; interactions among variables not significant; all
elements in model significant, $p < 0.05$. Partial F statistics: % shade tolerance class 5 (sqrt transformed): F=32, F4; % shade tolerance class 1:
P=9,84,

Notes (cont°d):

- 39. Significant interaction with % taxa with an obligate upland affinity.
 40. Significant interaction with % taxa with a facultative upland affinity.
 41. Significant interaction with % taxa with a facultative moisture affinity.
 42. Significant interaction with % taxa with a facultative wetland affinity.



Figure 4.10. Scatter plots of dispersal correlates of species richness in 10m x 10m quadrats I. Legend: %AIH = % taxa herbs dispersed by animal ingestion: %ANTH = % taxa herbs dispersed by ants: %WINDH = % taxa herbs dispersed by wind: %PDH = % taxa herbs dispersed by prolonged dormancy.



Figure 4.11. Scatter plots of selected correlates of species richness in 10m x 10m quadrats II. Legend: %UNASSH = % taxa herbs dispersed by unassisted means; %ACT = % taxa dispersed by animal conveyance (ants, or, seed caching by birds or mammals), all life forms: %PDT = % taxa dispersed by prolonged dormancy, all life forms: %UNASST = % taxa dispersed by uanssisted means, all life forms.

animal ingestion, ants, mechanical expulsion, and vegetative expansion.

The percentage of taxa that were herbs dispersed by unassisted means, prolonged dormancy, ants, and animal ingestion, explained 55.8 % of the variance in species richness in 10m x 10m quadrats. The percentage of taxa dispersed by unassisted means, animal conveyance (ants \div seed caching by birds and mammals), and prolonged dormancy (all life forms), explained 56.5 % of the variance in species richness in 10m x 10m quadrats. The former model represents a better statistical fit (p<W = 0.95 versus p<W = 0.67) and is considered to be the superior model for dispersal variables.

Pair-wise interactions were common among dispersal variables (superscript 3-11), and between mode of dispersal and life history (superscript 15-17), habitat affinity (superscript 28-31), and moisture affinity (superscript 39-43) (Table 4.8). Less common were interactions between mode of dispersal and provenance (superscript 18-19), life form (superscript 20-26), shade tolerance (superscript 33-37).

ii) Life History

Life history variables that maximally explained observed differences in species richness in 10m x 10m quadrats were the percentage of taxa that possessed a biennial ($r^2adj.=0.154$) or perennial ($r^2adj.=0.132$) life history (Table 4.8). The latter model is considered to be the superior model for life history variables in view of its superior statistical tit. The percentage of taxa that possessed an annual life history explained a significant but small fraction of the observed variance in species richness ($r^2adj.=0.043$).

Species richness was positively correlated with the percentage of taxa with an annual or biennial life history, and, negatively correlated with the percentage of taxa that possessed a perennial life history.

Pair-wise interactions were common among life history variables (superscript 15-17) and between life history variables and mode of dispersal (superscript 3-11), provenance (superscript 18-19), life form (superscript 20-26), habitat affinity (superscript 28-31), and shade tolerance (superscript 33-37). Plants with an annual or perennial life history interacted with moisture affinity (superscript 39-

43) whereas plants with a biennial life history did not.

iii) Provenance

The percentage of taxa that were native species explained marginally more variance in species richness than the percentage that were not ($r^2adj.=0.268$ versus 0.256, respectively) (Table 4.8). The former model is considered to be the superior model for provenance variables in view of its superior statistical fit and explanation of variance.

Species richness was positively correlated with the percentage of taxa that were alien species, and, negatively correlated with the percentage of taxa that were native.

Pair-wise interactions were common between provenance variables and mode of dispersal (superscript 3-11), life history (superscript 15-17), life form (superscript 19-26), habitat affinity (superscript 28-31), and shade tolerance (superscript 33-37). Provenance variables did not interact with moisture affinity (superscript 39-43).

iv) Life Form

Life form variables that maximally explained observed differences in species richness in 10m x 10m quadrats were the percentage of taxa that were grasses ($r^2adj = 0.273$) and trees ($r^2adj = 0.263$) (Table 4.8). Variables with intermediate explanatory power were the percentage of taxa that were vines ($r^2adj = 0.178$) and herbs ($r^2adj = 0.137$). The percentage of taxa that were ferns ($r^2adj = 0.055$) and shrubs ($r^2adj = 0.031$) explained a significant but minor fraction of variance in species richness. The percentage of taxa that were fern allies did not explain observed differences in species richness (p>F=0.08).

The percentage of taxa that were trees and vines collectively explained 32.5 % of the variance in species richness in 10m x 10m quadrats.

Species richness was positively correlated with the percentage of taxa that were vines, fern allies, grasses, and herbs, and, negatively correlated with the percentage of taxa that were trees, shrubs, and

ferns.

Pair-wise interactions between life form and life history (superscript 15-17) were more common than interactions among life forms and between life form and other plant traits. The percentage of taxa that were trees, shrubs, and grasses did not interact with moisture affinity (superscript 39-43).

v) Habitat Affinity

Habitat affinity variables that maximally explained observed differences in species richness in 10m x 10m quadrats were the percentage of taxa with an affinity for open ($r^2adj = 0.388$) and forested ($r^2adj = 0.214$) habitats (Table 4.8). The percentage of taxa with an affinity for "open + forest" habitats ($r^2adj = 0.194$) explained more than twice as much variance in species richness as taxa with an affinity for "forest + open" habitats ($r^2adj = 0.073$).

The percentage of taxa with an affinity for "open" and "forest" habitats collectively explained 41.4% of the variance in species richness in 10m x10m quadrats.

Species richness was positively associated with the percentage of taxa with an affinity for "open + forest" and "open" habitats, and, negatively correlated with the percentage of taxa with an affinity for "forest" and "forest + open" habitats.

Pair-wise interactions were common among habitat affinity variables (superscript 28-31), and, between habitat affinity variables and mode of dispersal (superscript 3-11), life history (superscript 15-17), provenance (superscript 18-19), life form (superscript 20-26), shade tolerance (superscript 33-37). Only the interactions between moisture affinity and the percentage of taxa with an affinity for "forest + open" and "open" habitats were significant.

vi) Shade Tolerance

Shade tolerance variables that maximally explained observed differences in species richness in 10m x 10m quadrats were the percentage of taxa classified as shade tolerance class 5 ($r^2adj.=0.293$), shade tolerance class 1 ($r^2adj.=0.214$), and, shade tolerance class 2 ($r^2adj.=0.170$) (Table 4.8). The

percentage of taxa classified as shade tolerance 4 ($r^2adj = 0.091$) explained more variance than taxa classified as shade tolerance class 3 ($r^2adj = 0.024$).

The percentage of taxa classified as shade tolerance class 5, and shade tolerance class 1, collectively explained 32.5 % of the variance in species richness in 10m x 10m quadrats.

Species richness was positively correlated with the percentage of taxa classified as shade tolerance class 4 and 5, and, negatively correlated with the percentage of taxa classified as shade tolerance class 1 and 2.

Pair-wise interactions were common among shade tolerance classes (superscript 33-37), and, between shade tolerance classes and mode of dispersal (superscript 3-11), life history (superscript 15-17), life form (superscript 20-27). Shade tolerance class interacted infrequently with provenance (superscript 18-19) and moisture affinity (superscript 39-43).

vii) Moisture Affinity

Moisture affinity variables that maximally explained observed differences in species richness in 10m x 10m quadrats were the percentage of taxa that were facultative wetland species (r²adj.= 0.301), facultative upland species (r²adj.= 0.269), obligate upland species (r²adj.= 0.255), and obligate wetland species (r²adj.= 0.214) (Table 4.8). The percentage of taxa with a facultative affinity for moisture did not explain observed differences in species richness (p>F=0.87).

The percentage of taxa that were facultative wetland species was considered to be the superior model for moisture affinity variables in view of its superior statistical fit and explanation of variance.

Species richness was positively correlated with the percentage of taxa classified as facultative wetland, and obligate wetland, species, and, negatively correlated with the percentage of taxa classified as facultative upland, and obligate upland, species.

Pair-wise interactions were common among moisture affinity variables (superscript 39-43), and,

between moisture affinity variables and mode of dispersal (superscript 3-11). life history (superscript 15-17), and, life form (superscript 20-26). Moisture affinity variables interacted infrequently with habitat affinity (superscript 28-31) and shade tolerance (superscript 33-37) and did not interact with provenance (superscript 18-19).

viii) Overall Model

The combination of plant attributes that maximally explained observed differences in species richness in 10m x10m quadrats was the percentage of taxa that were herbs dispersed by unassisted means, prolonged dormancy, ants, and animal ingestion. This model, which explains 55.8 % of the observed variance in species richness, was considered superior to all other models because of superior statistical fit and strong explanation of variance. An alternative model (composed of the percentage of taxa dispersed by unassisted means, animal conveyance and prolonged dormancy) explained marginally more variance in species richness (56.5 % versus 55.8%) but was a weaker statistical fit (p < W = 0.67 versus p < W = 0.95).

4.3.3 Comparison of Alternative Models of Species Richness

4.3.3.1 Generalized Linear Regression Models

The combinations of variables that maximally explain differences in species richness in 10m x 10m quadrats are presented in Table 4.9. Models E1 and E2 summarize the optimum combination of environmental variables: models D1 and D2 summarize the optimum combination of dispersal variables; models M1 through M5 summarize the optimum combination of environmental, dispersal, and life form variables. One outlier, quadrat #160, was removed from each model to improve the statistical fit (see Section 4.2.2.3).

Models D1, D2, and M4 were considered inferior models in view of the comparatively weak statistical fit (p < W = 0.33, 0.11, 0.22, respectively). Models M1, M3, and M5 were considered superior models in view of the excellent statistical fit (p < W = 0.88, 0.95, 0.89, respectively) and strong explanation of variance (r^2adj , = 0.659, 0.712, 0.642, respectively). Interactions among model elements prevented an overall statistical evaluation of superior and inferior models (see Section

Table 4.9. Comparison of selected GLM models of species richness. Model element F statistics, Response variable: sqrt # species; MSE = mean square error; $p \le W = Sha$ normal when $p \ge 0.05$); % taxa = % taxa (all life forms) in 10m x 10m quadrats. All n One outlier (quadrat #160) removed from each model.	s presented i piro-Wilk W todels parsin	n descending Lest for norm tonious based	rank order, ba ality of residu i on Mallow's	sed on partial als (residuals . Cp criterion.
		Model /	Attribute	
Model	<u>ب</u>	r ¹ udj	MSE	w>d
I. ENVIRONMENTAL VARIABLES				
MODEL E1: In # tree species (>1 m) + % canopy closure (In p/q) + forest cover type + open mfcrohabitats (cut or snag gap, lane regenerating field, seep, riparian meadow. riparian marsh) ¹	39.00	0.5833	0.8491	0.70
MODEL E2: % canopy closure (ln p/q) + sqrt # stems 0-4 cm dbh + soil moisture class + soil parent material ²	13,93	0.5482	0.9207	0.69
11. DISPERSAL VARIABLES				
MODEL D1 (ALL LIFE FORMS): % taxu dispersed by massisted means + % taxa dispersed by animal conveyance (ants + seed caching by birds and manumals) + % taxa dispersed by prolonged dormancy ³	19.68	0.6285	0.7571	0.33
MODEL D2 (HERBS ONLY): % taxa dispersed by unassisted means $\pm \%$ taxa dispersed by prolonged dormancy $\pm \%$ taxa dispersed hy ants $\pm \%$ taxa dispersed hy animal ingestion ⁴	74.10	0.6061	0.82026	11'0
III. MIXED VARIABLES				
MODEL, M1. % stems sugar maple + % taxa herbs dispersed by unassisted means + % taxa dispersed ty prolonged dormancy (all life forms)'	123.54	0.6593	0.6943	0.88

_	
Ś	
1	
5	
÷,	
ess	
Ē	
-2	
5	
਼ੁ	
ž	
5	
22	
-3	
- 2	
=	
2	
5	
÷	
- 2	
ž	
- S	
je	
ę	
- 23	
Ē	
ğ	
Ē	
ŭ	
Ъ,	
4	
ė	
4	
Ę.	

		Model /	Attribute	
Model	×	r'adj	MSE	p <w< td=""></w<>
III, MIXED VARIABLES (cont'd)				
MODEL M2: In # tree species (>1 m) + % canopy closure (In p/q) + forest cover type + % taxa dispersed by unassisted means (all life forms) ⁶	56.51	0.6716	0.6692	0.61
MODEL M3: sqrt # stems 0-4 cm dbh + % taxa grasses + % taxa harbs dispursed by prolonged dormancy + % taxa herbs dispersed by animal ingestion + % canopy closure ($\ln p/q$) + forest cover type ⁷	53,18	0.7119	0.5870	0.95
MODEL M4: sqrt # stems 0-4 cm dbh + % taxa herbs dispersed by prolonged dormancy \pm % taxa dispersed by animal conveyance (ants or seed caching by birds and mammals) \pm % canopy closure (ln p/q) \pm soil order \pm open regenerating fields [#]	44.89	0.6732	0.6660	0.22
MODEL M5: % taxa herbs dispersed by wind + sqrt # stems 0-4 cm dbh + soit moisture class 4 % canopy closure (ln $p(q)$) + soil parent material ⁴	43.63	0.6422	0,7291	0.89

Notes:

- 1. Model E.I. Partial F statistics: In # tree species: F=66.16; % canopy closure (In p/q): F=18.29; forest cover type: F=17.96; open microhabitats: F=15,09,
 - 2. Model E2. Partial F statistics: % canopy closure (la p/q): F=43.19; sqrt # live stems 0-4 cm dhb: 31.08; soil moisture class: F=22.85; soil parent material: F=8.10.
- 3. Model D1. Partial F statistics: % taxa dispersed by unassisted means (all life forms): F=59.36; % taxa dispersed by animal conveyance (all life forms): F=46.15; % taxa dispersed by prolonged dormancy (all life forms): F=25.92.
 - Model D2. Partial F statistics: % taxa herbs dispersed by unassisted means: F-32.63: % taxa herbs dispersed by prolonged dormancy: F=22.39; % taxa herbs dispersed by ants: F=19.99; % taxa herbs dispersed by animal ingestion: F=14.38. 4
- Model M1. Partial F statistics: % live stems sugar maple: F=92.35; % taxa herbs dispersed by unassisted means: F=58.19; % taxa dispersed by vi

Notes (cont'd):

- protonged domancy (all life forms): F=57.99.
- 6. Model M2. Partial F statistics: number tree species > Im: F=70.51; % canopy closure (in plu): F=20.33; forest cover type: F=4.29; % taxa dispersed by unassisted means (all life forms): F=69.64.
- 7. Model M3. Partial F statistics: surt # live stems 0-4 cm dbh: F=48.79, % tuxa grasses: F=34.58; % taxa herbs dispersed by prolonged dormancy: F=18.00; % taxa herbs dispersed by animal ingestion: F=15.64; % canopy closure (ht p/q): F=9.02; forest cover type: F=6.99
- Model M4. Partial F statistics: sqrt # live stems 0-4 cm dbh: F=36.40; % taxa herbs dispersed by prolonged dormancy: F=31.46; % taxa dispersed by animal conveyance: F=22.66; % canopy closure (In p/q): F=8.51; soil order: F=5.08; open regenerating fields: F=4.53. -
- 9. Model M5. Partial F statistics: % taxa herbs dispersed by wind: F=49,10; sqrt # five stems 0-4 cm dbh: F=29.75; soil moisture class: F=17.34; % canopy closure (In p/q): F-17.34; soil parent material: F=7.28.

4.2.2.3). In general, dispersal models explained more variance in species richness than did environmental models, and, mixed models explained more variance than either dispersal or environmental models, *sensu stricto*. Model M3, which included environment, life form and dispersal variables, explained 29.9% more variance in species richness than model E2, which included only environment variables.

4.3.3.2 Graphical Evaluation of Alternative Models of Species Richness

The degree to which elements in alternative models explained similar properties of the underlying environment was evaluated graphically by submitting the elements of each model to detrended correspondence analysis (DCA) and using the resulting "species" scores to construct a polygon in ordination space (see Section 4.2.2.3). The ordination space enclosed by each polygon was interpreted to represent the portion of sample space maximally explained by each model. Regression models that overlapped in ordination space were considered less distinctive than models that did not.

The results of this analysis are summarized in Table 4.10; representative graphical solutions are presented in Figure 4.12. The most distinctive model contrasts were between regression models E2 and M1 (no overlap in ordination space). E1 and D2 (minor overlap), and, M1 and M2 (minor overlap). All other models overlapped moderately or strongly in ordination space. Individual elements occupied distinct regions in ordination space when evaluated in contrasts with selected regression models: e.g. elements E14 (open microhabitats) and D12 (dispersal by animal conveyance) when present with elements in model D1 (Figure 4.12); element M34 (dispersal by animal ingestion) when present with elements in model M1 (Figure 4.13); element M53 (soil moisture) when present with elements in model M1 (Figure 4.13); and element M33 (dispersal by prolonged dormancy) when present with elements in model M5 (Figure 4.13). However, dispersal and environmental elements typically occupied proximate positions when present with elements in the superior models of species richness: M1, M3 and M5 (Figure 4.13). These results suggest that while each model addresses unique regions in ordination space, the models in this study primarily represent alternative ways of explaining underlying causal factors.

Table 4.10. Summary of graphical evaluation of leading models of species richness. Legend: E = environmental model, D = dispersal model, M = mixed model. Minor overlap: <10% area of smaller polygon; moderate overlap:10-50% area of smaller polygon: major overlap: >50% of area of smaller polygon.

Model	DCA Ordination Space	Degree of Overlap
E1 vs E2	overlap	moderate
D1 vs D2	overlap	major
M1 vs M2	overlap	minor
M1 vs M3	overlap	moderate
MT vs M4	overlap	moderate
M1 vs M5	overlap	moderate
M2 vs M3	overlap	major
M2 vs M4	overiap	moderate
M2 vs M5	overlap	major
M3 vs M4	overlap	major
M3 vs M5	overlap	major
M4 vs M5	overlap	major
Et vs D1	overlap	moderate
ET vs D2	overlap	major
El vs MI	overlap	major
ET vs M2	overlap	moderate
ET vs M3	overlap	major
E1 vs M4	overlap	moderate
El vs M5	overlap	moderate
E2 vs D1	overlap	minor
E2 vs D2	overtap	minor
E2 vs M1	separate	none
E2 vs M2	overlap	moderate
E2 vs M3	overlap	major
Model	DCA Ordination Space	Degree of Overlap
----------	----------------------	-------------------
E2 vs M4	overlap	moderate
Ë2 vs M5	overlap	moderate
D1 vs M1	overlap	major
DI vs M2	overlap	minor
D1 vs M3	overlap	moderate
D1 vs M4	overlap	moderate
Di vs MS	overlap	moderate
D2 vs ME	overlap	major
D2 vs M2	overlap	moderate
D2 vs M3	overlap	moderate
D2 vs M4	overlap	moderate
D2 vs M5	overlap	moderate

Table 4.10. Summary of graphical evaluation of leading models of species richness (cont'd).



Figure 4.12. Representative results from graphical evaluation of leading GLM models of species richness. Legend: Minor Overlap: <10% area of smaller polygon; Moderate Overlap: 10% - 50% area of smaller polygon; Major Overlap: >50% area of smaller polygon. See Table 4.14 for description of regression models.



Figure 4.13. Graphical evaluation of superior GLM models of species richness. Legend: M11 = mixed model, number 1, element 1. See Table 4.14 for description of elements and models. DCA = Detrended Correspondence Analysis (ter Braak 1987).

The results of the graphical evaluation of the "superior" models of species richness are presented in Figure 4.13. Model 1 was the most distinct, based on the degree of overlap in ordination space, whereas models M3 and M5 were the most similar. The portion of ordination space shared by all three models was very small.

Shared elements among models have presumably contributed to overlaps in ordination space when present (e.g. D2 vs M1, M3 vs M5). The contribution of shared elements to similarities in the fraction of variance explained by each model, however, is more difficult to discern, owing to the degree to which the elements in a given model co-vary. Inspection of the partial F statistics for each model (Table 4.11) reveals that the contribution of variables-in-common is largely context dependent.

4.3.4 Contribution of Phylogeny

The variables that explain species richness in a given linear regression model may depend in part on the taxonomic rank at which the relationships are examined. Variables that are significant, and non-interacting, at one taxonomic rank may not retain this status at more inclusive ranks owing to a reduction in the number of degrees of freedom. The regression models reported in Table 4.9 were re-evaluated at the ranks of genus, family and order to determine their dependence on taxonomic rank. The results of the analysis for the superior models are reported in Table 4.12.

As expected, the variance explained by a given model, and the quality of the statistical fit, declined with taxonomic rank. The number of interacting elements typically increased with taxonomic rank and, depending on the model, peaked at the rank of genus, family or order. The relative importance of variables often depended on the taxonomic rank of the analysis and elements in selected models did not retain their significance at the more inclusive ranks. The mean number of taxa in a given 10m x 10m quadrat was 42.5 species, 33.4 genera, 23.2 families, and 19.0 orders.

The most apparent cause of the interactions reported in Table 4.12 is the progressive inclusion of plant attributes at the higher taxonomic ranks (Table 4.13). This tendency is most strongly expressed at the rank of order, where more than 50 % of the orders included more than one mode of dispersal,

Table 4.11. Explanatory variables included in leading models of species richness (Table 4.9). Legend: E = environmental model, D = dispersal model. M = mixed model, ALF = all life forms, H = herb life form. See Table 4.9 for descriptions of models and summary statistics.

Variable	Model(s)	Partial F	p>F
# tree species	EI	66.16	<0.0001
	M2	75.51	<0.0001
% canopy closure	E1	18.29	<0.0001
	E2	43.19	<0.0001
	M2	20.33	<0.0001
	M3	9,02	0.0030
	M4	8.51	0.0040
	M5	12.64	<0.0001
forest cover type	E1	17.96	<0.0001
	M2	4.29	0.0024
	M3	6.99	<0.0001
open microhabitats	ËI	15.09	0.0001
# stems 0-4 cm dbh	E2	31.08	<0.0001
	M3	48.79	<0.0001
	M4	36.40	<0.0001
	M5	29.75	<0.0001
soil moisture class	E2	22.85	<0.0001
	M5	22.41	<0.0001
soil parent material	E2	8.10	<0.0001
	Mõ	7.28	0.0001
% unassisted means(ALF)	DI	59.36	<0.0001
	M2	69.64	<0.0001
% animal conveyance (ALF)	Di	46.15	<0.0001
	M4	22.66	<0.0001
% prolonged dormancy (ALF)	DI	25.29	<0.0001
	M1	57.99	<0.0001
% unassisted means (H)	D2	32.63	<0.000)
	MI	58.19	<0.0001
% prolonged dormancy (H)	D2	22.39	<0.0001
	M3	18.00	<0.0001
	M4	31.46	<0.0001
% ants (H)	D2	19.99	<0.0001

Variable	Model(s)	Partial F	p>F
% animal ingestion (Н)	D2 M3	14.38 15.65	0.0002 0.0001
% stems sugar maple	MI	92.36	<0.0001
% grasses	M3	34.58	<0.0001
soil order	M4	5.08	0.0007
open regenerating fields	M4	4.53	0.0347
% wind (H)	M5	49.10	<0.0001

Table 4.11. Explanatory variables included in leading models of species richness (cont'd).

Table 4.12. Contribution of phylogeny to superior models of species richness. Generalized linear regression (GLM) models, by row. Predictor variables are the percentage of species, genera, families, or, orders, in 10m x 10m quadrats, with the specified attribute. Sqrt = square root; $\ln p/q = \ln [(proportion) \div (1-proportion)]; n.s. - not significant (p>0.05); AI = % taxa herbs dispersed by animal ingestion; ASACC = % live stems >1m Acer saccharum; CC = % canopy closure; GRASS = % taxa grasses; MC = soil moisture class; PDH - % taxa herbs dispersed by prolonged dormancy; PDT = % taxa dispersed by prolonged dormancy (all life forms); SPM = soil parent material; WIND = % taxa herbs dispersed by wind; UNASS = % taxa herbs dispersed by unassisted means; 0-4 cm dbh = sqrt # stems 0-4 cm dbh. One outlier, quadrat # 160, removed from each model.$

Model	Response Variable	F Statistic	p- Value	r² adj.	p≤W ¹	Interaction
MODEL M1 - SPECIES: % stems sugar maple + % taxa herbs dispersed by unassisted means + % taxa dispersed by prolonged dormancy (all life forms) ¹	sgrt # spp	123.54	0.0001	0.6593	0.88	no
MODEL M1 - GENUS: % taxa dispersed by prolonged dormancy (all life forms) + % stems sugar maple + % taxa herbs dispersed by unassisted means ²	# genera	94.68	0.0001	0.5996	0.42	ΠØ
MODEL M1 - FAMILY: % taxa dispersed by prolonged dormancy (all life forms) + % stems sugar maple + % taxa herbs dispersed by unassisted means ³	# families	63.77	0.0001	0.4977	0,002	ASACC x PDT ASACC x UNASS
MODEL M1 - ORDER: % taxa dispersed by prolonged dormancy (all life forms) + % stems sugar maple + % taxa herbs dispersed by unassisted means ⁴	# orders	59,16	0,0001	0.4787	0.06	ASACC x PDT
MODEL M3 - SPECIES: sqrt # stems 0-4 cm dbh + % taxa grasses + % taxa herbs dispersed by prolonged domancy + % herbs dispersed by animal ingestion + % canopy closure + forest cover type ⁵	sqrt#spp	53.18	0.0001	0.7119	0.95	πο

Model	Response Variable	F Static Static	P- Value	r ³ «dj.	P≤W'	Interaction
MODEL M3 - GENUS; sqrt # stems 0-4 cm dbh +% taxa	# genera	47.07	0.0001	0.6857	0.77	PDH x GRASS
neros disperseu oy prototigeu domanuy 7 76 tava glasses 7 78 herðs dispersud by animal ingestion + % canopy closure + forest cover type ⁿ			_			
 MODEL M3 - FAMILY: sqrt # stems 0-4 cm dbh +% taxa therbs dispersed by prolonged dormancy + forest cover type + % canopy closure + % taxa herbs dispersed by animal ingestion + % taxa grasses (n.s.)? 	# families	25.65	1000'0	0.5387	0.26	PDH x 0-4 cm dbh CC x 0-4 cm dbh PDH x GRASS PD x AI
MODEL M3 - ORDER: sqr1 # stems 0.4 cm dbh 4% taxa licrbs dispersed by prolonged dormancy + % taxa herbs dispersed by animal ingestion $+$ forest cover type + % canopy closure + % taxa grasses (n.s.) ⁸	# orders	23.58	0,0001	0.5168	0.81	CC x 0-4 cm dbh PDH x GRASS AI x GRASS
MODEL, M5 - SPECHES: % taxa herbs dispersed by wind a sqrt # stems 0-4 cm dbh + soil moisture class + canopy closure (hr p/q) + soil parent material?	surt # spp	+3.63	1000.0	0.6422	0.89	цо
MODEJ, M5 - GENUS: % taxa herbs dispersed by wind +canopy closure (In p/q) + sqrt # stems 0-4 cm dbh + moisture class + soil parent material ²⁰	<i>ដ</i> ម្លូយាមពេ	36.00	1000.0	7292.0	0.12	MC x 0-4 cm dbh CC x 0-4 cm dbh
MODEJ, M5 - FAMILY: canopy closure (in p/q) + moisture class + sqrt # sterns 0-4 cm dbh + soil parent material + % taxa herbs dispersed by wind (n.s.) ¹¹	# families	30.09	1000.0	0.5505	0.40	CC x 0-4 cm dbh

Table 4.12. Contribution of phylogeny to superior models of species richness (cont'd).

283

Model	Response Vuriable	<u></u>	P- Value	r' adj.	p <w'< th=""><th>Interaction</th></w'<>	Interaction
MODE1. M5 - ORDER: soil parent material + sqrt # stems 0- 4 cm dbh + canopy closure (la p/q) + moisture class + % taxa herbs dispersed by wind (n.s.) ¹⁷	# orders	32.06	0.0001	0.5667	0.07	SPM x WIND SPM x CC CC x 0-4 cni dbli
Notes: I. MODEL M1: SPECIES: Partial F statistics; % live stems sug- discovered to contended dormancy (all life forms): F=57.99.	ar muple: F–97	2.35; % taxa	ı herbs dis	hersed by i	unassisted	means: F=58,19; % taxa
2. MODEL M1: GFNUS: Partial F statistics: % taxa dispersed 1. F48.18. % taxa herbs dispersed by upassisted means: F=11.	by prolonged d 42.	lormancy (a	ıl) life for	0s): F=l]	i.12; % liv	e stems sugar maple:
3, MODEL M1; FAMILY: Partial F statistics: % taxa dispersed	by prolonged	dormancy	(all life for	rms); F=6(.21; % liv	e stems sugar maple:
F=31.62; % taxa herbs dispersed by unassisted means. F =6.0 4. MODEL M1; ORDER: Partial F statistics: % taxa dispersed 7.12.51.02 means: F=0.6	by protonged 0	dormancy (all life for	ms): }=69	,72; % liv	e stents sugar maple:
5. MODEL, M3: SPECTES: Partial F statistics: sqrf # live stem	s 0-4 cm dbh: 1	Sar 18, 79, %	taxa grass	es: P=34.5 so the etc.	58; % (nxa ~ E0 07-1	herbs dispersed by pro- losest cover tone: F=6.09
Imged dormancy: F=13.00; % taxa herbs dispersed by annual 6. MODEL M3: GENUS: Partial F statistics: sqrt # stems 0-4 c	ringestion: F=38.0 em dhh: F=38.0	08; % Iaxa	herbs disp	ersed hy p	colonged d	lormaney: F=30.20;
% taxa grasses: F=24.91; % herbs dispersed by animal inges 	stion: F=9.41; •m.dhh: F=28.5	% canopy 4 57: % taxa b	stosure: Fi Jerbs dispe	=K,NO; for rsed by pr	est cover () alonged do	ype: t≃5.48. srmancy: F=25.16; _fores!
cover type: F=10.16; % canopy clasure: F=6.53; % taxt hort	bs dispersed by	/ animud ing 68- % taya	gestion: F≞ herths disn	-6.41; % I	axa grasse rolonged d	is: F=0.001 (n.s.). Jormaney: F=25.85: %
A. MULTINE TO ALL ALL ALL ALL ALL ALL ALL ALL ALL AL	wer type: F=6.	42; % can	apy closure	a F=6.14;	% laxa gr	asses: } =0.64 (n.s.).
9. MODEL M5: SPECIES: Partial F statistics: % taxa herbs disp	bersed by wind	F=40,10; s	qri # live s	tems 0-4 c	m dibh: F=	29,75; soil moisture class
F=17.54; % canopy closure (In p/q): F=17.54; %01 purch una 10, MODEL M5: GENUS; Partial F statistics: % taxa herbs disp	ersed by wind:	F=41.86;	% canopy «	closure (In	p/q): F=2	0,42; sqrf # stems 0-4 cm
Obh: F=18.86, moisture class: F=16.62; soil parent material: 11. MODEL M5: FAMILY: Partial F statistics: soil moisture cla	F=7.68. iss: F=22.15; c	anopy closu	ne (In p/q)); ['=21,37	; sqrt # ste	ans 0-4 cm dbh: P=21.08
soil parent material: F=15.65; % taxa herbs dispersed by win 12, MODEL M5: ORDER: Partial F statistics: soil parent mater	nd: F=0.95 (n.: ial: F=23.21; s	s.). qrt # stems	0-4 cm db	di: F-22,2	I; canopy	closure (In p/q): F=21.78
moisture class: $F=15.27$; % taxa herbs dispersed by wind: F	= .36 (n.s.).					

Table 4.12. Contribution of phylogeny to superior models of species richness (cont'd).

784

Table 4.13. Proportion of taxa with selected plant attributes at progressively more inclusive taxonomic ranks. Categories: Moisture Affinity: obligate upland, facultative upland, facultative, facultative wetland, obligate wetland; Light Affinity: Shade tolerance class 1 (very shade tolerant) - class 5 (very shade intolerant); Mode of Dispersal: animal ingestion, animal adhesion, animal conveyance, wind, prolonged dormancy, mechanical expulsion, unassisted means. Cell entries = % classified taxa with specified number of attribute categories.

Taxonomic	#			# Attribute	Categories					
Kank	taxa	1	2	3	4	5	>t			
MOISTURE /	AFFINITY				•					
Species	387	100	0	0	0	0	0			
Genus	204	74.5	17.2	4.4	2.5	1.5	25.5			
Family	76	46.1	22,4	17.1	3.9	10.5	53.9			
Order	45	28.9	17.8	26.7		15.6	71 .1			
LIGHT AFFINITY										
Species	266	100	0	0	0	0	0			
Genus	[7]	81.3	12.9	5.3	0.6	0	18.7			
Family	62	56.5	17.7	16.l	9.7	0	43.5			
Order	45	26.7	17.8	31.1	11.1	13.3	73.3			
MODE OF DISPERSAL										
Species	413	88.1	11.1	0.5	0.2	0	11.9			
Genus	208	86.5	12.5	1.0	0	0	13.5			
Family	77	66.2	14.3	11.7	2.6	5.2	33.8			
Order	45	42.2	20.0	22.2	4.4	11 .1	57.8			

and, where more than 70 % of the orders included more than one moisture or light affinity.

Taken together, these results reveal the models of species richness developed for this study are sensitive to the taxonomic rank at which they are assessed. This provides indirect evidence that phylogeny has contributed to the number of taxa recorded in 10m x 10 m quadrats in this study.

4.4 Discussion

i) Environmental Heterogeneity

Environmental beterogeneity, in this study, is spatial and temporal variation in conditions that govern the germination, establishment, growth and reproduction of plants. When measured in relation to traditional environmental variables, beterogeneity in the availability of moisture, nutrients and light explains up to 58.3% of the variance in species richness in 10m x 10m quadrats (Table 4.9). When measured in relation to microhabitats, beterogeneity in site conditions at the landscape scale has facilitated the germination and establishment of 38.3% of the sampled flora (158 species) that was not recorded on closed, seasonally dry, forest floors, *sensu stricto* (Appendix 5).

The contribution of microhabitats to species richness was strongly scale dependent. Whereas 21 of 23 evaluated microhabitats contributed to differences in species richness at the quadrat scale (Table 4.4), only 9 of 20 evaluated microhabitats contributed to differences in species richness at the patch scale (Table 4.5). In contrast, whereas 38 of 39 habitats contained at least one unique species at the quadrat scale (Table 4.4), only 13 of 39 microhabitats contained at least one unique species at the landscape scale, and only five of these contained five or more species not recorded elsewhere in the forest understory (Table 4.6).

One apparent reason for this pattern is that the distribution of species has been constrained by opportunities for germination, establishment and persistence. In a spatially variable environment, differences in plant traits are expected to lead to pattern in the distribution of species and in the composition of local plant assemblages. In this analytical setting, the probability of encountering a new species scales with the size of the sampled area and with the scale of heterogeneity in the local

environment. The observed decline in the number of species restricted to a given habitat is consistent with this scaling relationship, and, with the passive sampling effects associated with an increasing sample size. This suggests that an increasing proportion of the species capable of occupying a given habitat were recorded as the size of the sampled area increased from quadrat to patch to landscape. The sharp decline in the number of contributing habitats in the transition from quadrat to patch suggests that much of the regional variance in environmental heterogeneity was captured at the scale of the forest patch.

Additional factors that may have contributed to this pattern of scale dependence include: the degree of habitat specialization in the sampled flora: the thresholds used to differentiate continuous environmental states into discrete habitat types; the decision to evaluate the relationship at the habitat level, and not in relation to more generalized categories: the sampling decision to locate all eight 10m x 10m quadrats within a 35 metre radius of a fixed sampling point within each forest patch; and, plant dispersal.

The contribution of dispersal to observed differences in the distribution and composition of plant assemblages is less transparent, given the decision to sample plants rather than propagules in this study. Nevertheless, the repeated observation that most propagules land in close proximity of the maternal plant (Appendix 11) suggests that dispersal may tarely limit the availability of propagules within 10m x 10m quadrats, on ecological time scales. This suggests, in turn, that pattern in the distribution of species at this spatial scale is controlled primarily by factors that govern germination, establishment and persistence. Dispersal on the order of decimetres to metres may also be sufficient to sample all habitats with a given forest patch, on most ecological time scales, since all eight quadrats were located within a 35 metre radius of a fixed sampling point.

Dispersal to nearby forest patches currently requires a mean dispersal reach of 41m to1378m. depending on the degree of forest cover in the landscape. Dispersal over this distance should not pose an absolute constraint for taxa dispersed by birds and large mammals (38.5% of the sampled flora) and for taxa dispersed by spores or winged and plumed seeds of low mass (proportion of wind-dispersed taxa, 30.5% of the sampled flora, unknown). Such distances will be beyond the normal

reach of species dispersed by other means, however, and, except for rare long-distance dispersal events, the propagules of these species will remain within the current patch. Dispersal beyond these distances is presumably a low-probability event for even the most mobile of species and thus plant migration, rather than dispersal, is the more relevant consideration when evaluating the distribution of plants at the landscape scale.

Taken together, this reasoning suggests that the distribution and richness of plant assemblages in these forests has been not been constrained by dispersal, given the length of time available, post-glacially, for the dispersal and migration of less mobile taxa. The number of species restricted to a particular habitat, therefore, has been influenced primarily by the spatial and temporal scale of heterogeneity in the local environment and by the spatial scale of the analysis.

Heterogeneity, when measured as the number of microhabitats in a given 10m x10m quadrat, was a significant but weak contributor to species richness (p=0.0007, $r^2adj = 0.054$, Table 4.1). The mean number of habitats per 10m x 10m quadrat was 3.01 and ranged from one to ten habitats (supplementary analysis not shown). The weak explanation of variance suggests that diversity in the number of habitats is less important at this spatial scale than the attributes, and area, of the habitats that are present.

The habitats that contributed maximally to species richness, at every spatial scale, were closed seasonally dry forest floors, open regenerating fields, and, open lanes/roads (Tables 4.4,4.5.4.6). These habitats stand in contrast to one another, with respect to available light, habitat structure and site history; and, to related habitats in the forest understory, with respect to available moisture, substrate, transience, and sampled area. Seasonally dry forest floors were more species rich than open lanes/roads and open regenerating fields with respect to total species richness (255 species, 181 species, and 147 species, respectively, Appendix 11) but were far less species rich with respect to mean-richness-when-present (31.1 species, 50.7 species, and 58.0 species, respectively; supplementary analysis not shown). When differences in habitat transience and area are taken into account, open lanes/roads and open regenerating fields are markedly species rich, by any measure, since the latter habitats are transient in time and individually occupied only 1.3% of the total sampled area

(Table 4.4), whereas, closed seasonally dry forest floors persist for the life span of the canopy and occupied 54.3% of the total sampled area.

One apparent reason for the large number of restricted species in these habitats is the suite of adaptations required to germinate, establish, and persist in low and high light environments. One plant trait that has been differentially favored in these habitats is mode of dispersal. Herbs recorded on closed, seasonally dry forest floors were disproportionately dispersed by animal ingestion whereas herbs recorded on open lanes/roads were disproportionately dispersed by prolonged dormancy (Table 3.11). In contrast, herbs recorded in open regenerating fields were disproportionately dispersed by the wind. It may not be coincidental that herbs dispersed by these modes are the most mobile taxa in these forests (Appendix 11), since the seed rain from mobile taxa is more likely to include a greater proportion of species from the surrounding landscape with the capacity to establish in such habitats.

One habitat factor that presumably contributes to this pattern is available light at the soil surface. Establishment in dense shade, for example, has been shown to be facilitated by seed reserves that enable seedlings to survive to the cotyledon stage (Saverimuttu and Westoby 1996). This enables plants with large seeds to specialize on shaded conditions and to avoid the reduction in mean annual fitness that arises when dormancy constrains the number of seeds produced in favorable years (Venable and Brown 1988). Selection pressure for the evolution of dormancy is therefore expected to be weak in closed forest habitats. In contrast, dormancy is expected to be favored in open habitats where the capacity for delayed germination enables plants to avoid unfavorable conditions and to specialize on conditions that maximize reproductive success (Brown and Venable 1986, Cohen and Levin 1987). Selection pressure for the evolution of dormancy in these environments is expected to be greatest in small-seeded annual and biennial plants since the cost of reproductive failure is extirpation. In keeping with these expectations, the number of herbs in this study dispersed by prolonged dormancy increased with increasing light availability and no herbs with an apparent affinity for forested habitats were dispersed by prolonged dormancy (Table 3.7).

The tendency for herbs dispersed by prolonged dormancy to occur more frequently on open lanes

than in open regenerating fields may also be due to differences in available light since leaf litter was absent on the traveled portion of lanes but present in regenerating fields (personal observation). The periodic breakage of stems by passing vehicles and equipment, and the crushing of plants by tires, may also be important since on lanes and access roads the mean height of plants dispersed by prolonged dormancy was significantly lower than plants dispersed by wind or by both wind and dormancy (mean minimum height = 17.3 cm. 32.3cm and 32.3, respectively, $p>\chi^2 - 0.0097$; mean maximum height = 76.6 cm, 133.2cm, and 146.4 cm, respectively, $p>\chi^2 = 0.0001$; supplementary Wilcoxon rank sum tests, based on data on stem heights presented in Gleason and Cronquist 1991, not shown). The mean heights of plants dispersed by prolonged dormancy was also lower on open lanes than in open fields, but not significantly so. In general, however, herbs dispersed by prolonged dormancy (mean minimum height = 23.0 cm, 31.2 cm and 30.0 cm, respectively, $p>\chi^2 = 0.0333$; mean maximum height = 81.4 cm, 121.7 cm, and 139.5 cm, respectively, $p>\chi^2 = <0.0001$). Taken together, these results suggest that habitat conditions in lanes and fields have favored the persistence of plants with contrasting traits that are causally or passively correlated with mode of dispersal.

ii) Environmental vs Plant Correlates of Species Richness

Overall, plant attributes in this study explained more variance in species richness in 10m x 10m quadrats that did physical attributes of the environment (r^2 adj. = 0.6285. Model D1, versus r^2 adj. = 0.5833. Model E1. Table 4.9). Plant attributes that maximally explained differences in species richness were mode of dispersal (r^2 adj. = 0.5583), habitat affinity (r^2 adj. = 0.4143). life form (r^2 adj. = 0.3373), shade tolerance (r^2 adj. = 0.3247) and moisture affinity (r^2 adj. = 0.3014) (Table 4.8). In contrast, environmental attributes that maximally explained differences in species richness were stand structure (r^2 adj. = 0.5485) and soils (r^2 adj. = 0.3557) (Table 4.1). Significant but weak predictors of species richness were life history (r^2 adj. = 0.1315), human disturbance (r^2 adj. = 0.0537) (Tables 4.1 and 4.8).

The tendency for plant attributes to explain more variance than physical attributes of the environment was not unexpected. Plants in the forest understory sample spatial differences on the scale of centimetres and temporal differences over the course of a life span, whereas, apparent differences in moisture, nutrients and light were measured or characterized on the scale of decimetres to metres and over the course of one field season. The similarities in explained variance suggests that the scale of the environmental data was appropriate for the scale of heterogeneity in the environment. The overall fraction of explained variance, however, suggests that important factors may have been weakly characterized or overlooked.

Among plant attributes, mode of dispersal explained far more variance than life form, shade tolerance or moisture affinity. One apparent reason for this pattern is that the moisture affinity and shade tolerance of taxa both varied in relation to mode of dispersal ($p>\chi^2 = 0.02$, and <0.0001, respectively, supplementary log likelihood ratio tests, not shown). In contrast, only the shade tolerance of taxa varied with life form ($p>\chi^2 = <0.0001$); the moisture affinity and shade tolerance of classified species were not associated ($p>\chi^2 = 0.42$). Differences in moisture and light availability are thus more likely to be accounted for by modes of dispersal than by life form, or by moisture and light affinity alone.

Mode of dispersal also explained more variance than habitat affinity (r^2 adj. = 0.5583 versus 0.4143, respectively, Table 4.8). The most apparent explanation for this pattern is that mode of dispersal was also strongly correlated with calcium availability in the upper 15 cm of the soil profile whereas habitat affinity was not. Thus differences in moisture, nutrients and light are more likely to be correlated with mode of dispersal than with habitat affinity alone.

The degree to which dispersal has contributed to differences in species richness was not resolved. While there is indirect evidence of directed dispersal by birds in former canopy gaps, and by birds, mammals and ants on disturbance features associated with former canopy gaps (tip-up mounds, tree pits, logs, stumps, closed lanes), most propagules appear to be dispersed at random and in close proximity to the maternal plant (Chapter 3). This suggests that the distribution of plants in a given 10m x 10m quadrat has been determined primarily by factors that govern germination, establishment and persistence. However, the number of species with a particular attribute in a given quadrat may be influenced by dispersal since the seed rain from mobile taxa is more likely to reflect the diversity of species in the surrounding region that are capable of establishing there. In keeping with this line of reasoning, the mean percentage of plants dispersed by animal ingestion in 10m x 10m quadrats was significantly greater than the proportion in the flora as a whole (herbs:10.1 % vs 6.7 %, p>|z| = 0.0001: all life forms: 27.6 % vs 16.9 %, p>|z| = 0.0000). This pattern was also apparent in plants dispersed by animal conveyance (all life forms: 15.8% vs 9.7%, p>|z| = 0.0000). These taxa are typically shade tolerant and thus may be less constrained by understory conditions that species with an affinity for open habitats. Approximately 80% of the taxa dispersed by animal ingestion were recorded on closed, seasonally dry, forest floors.

The reverse pattern, however, was observed in plants dispersed by the wind (herbs: 10.1% vs 25.4%. p>|z| = 0.0000; all life forms: 29.69 % vs 30.5 %, p>|z| = 0.1054). This result was not expected and may reflect differences in the dispersal reach of seeds relative to spores, since the proportion of wind dispersed taxa approached their proportion in the sampled flora when ferns and fern allies were included in the analysis. This pattern may also reflect a greater diversity in germination and establishment requirements among wind than animal dispersed taxa, since only 67.6% of the species dispersed by wind were found in canopy openings and only 49.2% were recorded in apparently favorable habitats such as open roads and open regenerating fields (Appendix 11).

On balance, these results suggest that long-distance dispersal has enriched the proportion of species dispersed by animal ingestion and animal conveyance. Overall species richness has been governed by additional factors, however, since dispersal by animal ingestion was strongly associated with low species richness in 10m x 10m quadrats whereas dispersal by wind was strongly associated with high species richness (Table 3.16).

The degree to which dispersal and environmental factors accounted for a unique fraction of the variance in species richness was evaluated indirectly by detrended correspondence analysis (DCA). The results of this analysis suggest that while certain combinations of dispersal and environmental variables may occupy distinct regions in ordination space, they primarily represent alternative ways

of explaining the proximate mechanisms of causation. Modes of dispersal are thus strongly correlated with the habitat factors and plant traits that govern germination, establishment and persistence. Environmental variables, in turn, are correlated with modes of dispersal that determine the composition and size of the seed rain, the initial conditions that seeds and seedlings must confront, and, therefore, who interacts with whom and with what intensity.

In these models (Figures 4.12, 4.13), dispersal by animal ingestion was associated with soil moisture, whereas, dispersal by the wind was associated with the number of tree stems in the 0-4 cm dbh class, forest cover type, and soil parent material. In contrast, dispersal by prolonged dormancy was associated with sugar maple abundance and dispersal by the wind, whereas dispersal by unassisted means was associated with percent canopy closure.

Factors that may contribute to the correlation of environmental and dispersal variables include: i) the predominance of short-distance dispersal in plant communities (Portnoy and Willson 1993, Hughes *et al.* 1994): ii) a convergence of germination and dispersal biology, owing to the necessity for prior germination and reproductive success for the transmission of dispersal alleles from parent to offspring (Venable and Lawlor 1980, Olivieri and Berger 1985, Venable *et al.* 1995, Cody and Overton 1996); iii) the presence of habitat factors that favor dispersal by some modes but not others (Cohen and Levin 1987, Venable and Brown 1988, 1993); and, iv) heterogeneity in local conditions governing germination, establishment and persistence (Fowler and Antonovics 1981, Robertson *et al.* 1988, Lechowicz and Bell 1991, Palmer 1990).

These factors suggest that the contribution of dispersal to species richness is both fundamental and pervasive. The failure to find strong evidence of this contribution, therefore, may be due more to the choice of methodology than to the importance of dispersal, *per se*.

iii) Forest Stand Structure

In this study, the environmental variables that maximally explained species richness were attributes of forest stand structure: forest cover type (r^2 adj. = 0.2765), number of tree species (r^2 adj. = 0.2682), and, the relative abundance of sugar maple (r^2 adj. = 0.2555) (Table 4.1). Stand structure

variables explained more than 50% of the variance in species richness in 10m x 10m quadrats (Table 4.1) and were important elements in the superior models of species richness (Table 4.9).

The capacity of forest stand structure to account for species richness is due in part to the fact that measured attributes both respond to, and alter, the spatial and temporal availability of moisture, nutrients and light on the forest floor. As "consequence", stand structure reflects the past influence of physical and ecological processes that govern the distribution and abundance of plants in the forest understory. As "cause", stand structure controls the spatial and temporal availability of light within the forest canopy and on the forest floor (Minkler and Woerhiede 1965, Horn 1971, 1975, Messier and Bellefleur 1988, Canham *et al.* 1990, Canham *et al.* 1994, Brewer 1980); moderates the availability of nutrients and moisture in the rooting zone of forest soils (Aber *et al.* 1991, Pastor and Post 1986, Zinke 1962, Crozier and Boerner 1984,1986, Leininger and Winner 1988, Boerner and Koslowsky 1989); determines the quantity and quality of coarse woody debris on the forest floor (Harmon *et al.* 1986, Hale and Pastor 1998); influences the probability and size of tree pits created during wind-throw events (Putz *et al.* 1983); and, influences the timing and size of gaps in the forest canopy (Lorimer *et al.* 1988, Lorimer 1989, Frelich and Lorimer 1991).

As a causal factor, stand structure contributes to species richness by providing spatial and temporal heterogeneity in the form of logs, stumps, tree pits, tip-up mounds, raised root mats, and gaps in the forest canopy. These features provide novel resources for the germination, establishment and persistence of new species and for the co-existence of species already established on the forest floor. The results of this study suggest that contribution to coexistence may be far more important than the contribution to richness *per se*.

At small spatial scales, habitat features created by the death of canopy trees provide novel and alternative settings for the germination of propagules from nearby plants, and, the number of local species is substantially enriched (Table 4.4). At larger spatial and temporal scales, these features increasingly represent alternative rather than novel habitat, owing to the apparent tolerance of plants to conditions created by the death of a canopy tree (Figure 2.9), and, the number of new species recorded on these features declines (Tables 4.5 and 4.6). When viewed from the perspective of the

forest patch, or regional landscape, the contribution of habitats created by the death of canopy trees to species richness is strikingly modest relative to the contribution of the forest floor.

The predominance of short-distance dispersal in plant communities, however, suggests that the appropriate spatial scale for this analysis is on the order of metres to tens of metres. At this spatial scale, the availability of alternative habitat is expected to promote species coexistence by increasing the probability that there will be some place, or time, where competitors perform poorly or do not survive and where populations of low abundance species may expand (Hutchinson 1961, Levin 1974, Warner and Chesson 1985, Comins and Noble 1985, Hurt and Pacala 1995). In keeping with this expectation, more species in this study were recorded in canopy gaps (282) than on tip-up mounds (180), logs (120), pits (97) or stumps (64) (Appendix 11). This suggests that species in the understory of these forests differ in their capacity to colonize or persist on these features, and, that the principal contribution of habitats created by the death or removal of canopy trees may be the maintenance, rather than generation, of species richness.

The principal mechanism by which stand structure contributes to differences in species richness, however, is by altering the spatial and temporal availability of light within the forest canopy and on the forest floor. For plants in these settings, the composition of the canopy matters, owing to differences among species in the arrangement of leaves and stems (Horn 1971, Brown and Parker 1994), the transmission of light (Canham *et al.* 1994), the seasonal pattern of leaf expansion and senescence (Brewer 1980), the shade tolerance of seedlings and young saplings (Kobe *et al.* 1995, Lusk and Reich 2000), and, life span (Lorimer 1989). In the forests examined in this study, the species that exerted the greatest control was sugar maple (supplementary correlation analysis not shown: % stems *Acer saccharum*, % stems *Fagus grandifolia*, % stems *Tsuga canadensis*. % stems shade intolerant and very shade intolerant trees, % stems wet-mesic or wet trees, % taxa in shade tolerance class 1, class 2, class 4, class 5). Although capable of casting a deeper shade than sugar maple (Canham *et al.* 1994), beech and hemlock exerted little influence over the availability of light, owing to their low density in the forest canopy.

In keeping with these patterns, the number of species in 10m by 10m quadrats was strongly and

negatively correlated with the absolute and relative abundance of sugar maple. Sugar maple abundance explained 25.6% of the overall variance in species richness (Table 4.1) and up to 63.0% of the variance in species richness in undisturbed quadrats on Brunisolic and Luvisolic soils overlying calcareous till (Figure 4.5). The diversity of trees in the forest canopy, a strong correlate of overall species richness in these forests, and in forests elsewhere in glaciated eastern North America (Braun 1950, Curtis 1959, Whittaker 1965), declined sharply in the presence of sugar maple (Table 4.3), and was up to 50% lower in stands on mesic soils (Table 2.11).

Sugar maple achieved its highest abundance on mesic soils that were high in available calcium (Figures 4.5, 4.6, 4.7 and 4.9). This pattern is commonly observed in southern Ontario (Farrar 1995), where sugar maple abundance is strongly associated with lime rich soils: in the mesic forests of southern Wisconsin (Curtis 1959), where sugar maple dominance and low canopy diversity are associated with high levels of exchangeable calcium in the A₁ layer (36 to 64 cmol/kg); and, in mesic forests elsewhere in the Beech-Maple and Maple-Basswood forest regions (Braun 1950), where sugar maple dominance is strongly associated with glaciated soils that possess an argillic B horizon (Braun 1950, Brady 1990). The correlation between sugar maple abundance and calcium rich soils has also been observed in selected northern hardwood stands in northwestern Connecticut, where the mortality of sugar maple seedlings and young trees is sharply lower than on nearby acidic soils (Kobe *et al.* 1995, Kobe 1996).

This pattern contrasts sharply with abundance of sugar maple elsewhere in its range, where it is typically only one of several trees in the forest canopy. This is particularly true of the rich cove forests of the Mixed Mesophytic, and Oak-Chestnut, forest regions which typically contain up to 30 or more canopy trees (Braun 1950) and an exceptionally species rich understory (Whittaker 1965). The availability of base cations in these forests varies. Reported mean values for exchangeable calcium range from 3.3 cmol/kg in old growth stands in eastern Kentucky (Muller 1982) to 1.95 emol/kg in mesic eutrophic cove forests on the North Carolina Piedmont (Peet and Christensen 1980). These values are substantially lower than the mean concentration of available calcium in sugar maple dominated quadrats in this study (8.9 cmol/kg) and the mean concentrations reported by Curtis (1950) for exchangeable calcium in the A₁ horizon of sugar maple dominated forests in

southern Wisconsin (32 to 64 cmol/kg).

Exchangeable base cations have been identified as a potential contributing factor to species richness in forests on the North Carolina piedmont (Peet and Christensen 1980). In an analysis of 105 hardwood forest stands on soils of contrasting moisture and fertility, total exchangeable base cations (calcium + magnesium + potassium) explained 85% of the variance in species richness in 0.1 ha plots. In mesic eutrophic cove forests, the most species rich community, the mean concentration of total base cations in the upper 10 cm of the soil profile was 1043 ppm. Calcium ions accounted for 75.0% of this total. In keeping with cove forests elsewhere in the Oak-Chestnut forest region, the stand with the highest species richness contained 29 species of trees.

In contrast, total base cations in this study explained only 2.5% of the overall variance in species richness (p>F=0.02) and did not explain the variance in species richness in undisturbed maple stands on mesic soils overlying calcareous till. When two strong outliers were excluded from the analysis, soil order explained 71.8% of the variance in species richness in these stands and 86.2% of the variance in sugar maple abundance (measured as percent stems sugar maple). Sugar maple abundance, in turn, explained 69.9% of the variance in species richness. In keeping with cove forests, species richness was significantly higher in the quadrats with high tree diversity (mean # tree species per quadrat: 9.4 species on Brunisolic soils versus 4.2 species on Luvisolic soils) and with low sugar maple abundance (mean % stems sugar maple per quadrat: 15.1% on Brunisolic soils versus 76.7% on Luvisolic soils). In contrast to the cove forests, high concentrations of total base cations and exchangeable calcium in the upper 15 cm of the soil profile were associated with low rather than high species richness.

Taken together, these results suggest that species richness on mesic soils in the Peterborough area is governed primarily by soil order and by the percentage of sugar maple stems (>1 m) in the forest canopy. In contrast to the cove forests, calcium availability in Ontario stands is strongly correlated with sugar maple dominance which in turn sharply limits the availability of light within the canopy and on the forest floor. The mean concentration of base cations at which the transition to sugar maple dominance and low species richness may occur is not known, but in these forests, it appears to lie between 4.3 and 6.8 cmol/kg for available calcium, and between 1795 and 2780 ppm for total base cations. These values are substantially above those reported by Peet and Christensen (1980) for cove forests on the North Carolina piedmont.

The capacity of soil order to explain observed differences in species richness and sugar maple abundance requires further study, since the degree to which soil order stands for available calcium in the preceding analysis is not known. A narrow reading would suggest that properties other than available calcium may be involved since soil order in this analysis explained far more variance in species richness and sugar maple abundance than available calcium (71.8% vs 20.7%, species richness; 86.2% vs 14.2%, sugar maple abundance; supplementary regression analysis not shown). Caution and further analysis are required, however, since available calcium explained 63.0% of the variance in sugar maple abundance when the analysis was restricted to undisturbed second-growth stands on Brunisolic soils (Figure 4.6),

Taken together, however, these results suggest sugar maple abundance, mediated by differences in the availability of calcium, and potentially by unmeasured properties of differentially weathered soils, is the principal contributor to species richness in closed-canopied forests on mesic soils. The proximate mechanism by which this influence is exerted is primarily by control of the spatial and temporal availability of light within the forest canopy and on the forest floor.

iv) Forest Fragmentation

The observation that larger areas, in general, contain more species than smaller areas has recently, and critically, been explained in relation to three hypotheses:

- the "habitat diversity" hypothesis (Williams 1943.1964), which argues that species number is governed primarily by environmental factors and thus larger areas typically have more species because they have more habitats:
- the "passive sampling" hypothesis (Connor and McCoy 1979, Coleman 1981), which argues
 that species number is controlled primarily by passive sampling of the species pool and thus

larger areas typically contain more species because they represent larger samples:

iii) the "area per se" hypothesis (Preston 1960, 1962; MacArthur and Wilson 1963, 1967), which argues that when distance to the propagule pool is held constant, species number will be governed primarily by extinction rates; since these rates are inversely proportional to population size, the number of species in a given sample will be proportional to its area.

The fixed-area sampling design applied in this study provides a suitable context for an indirect test (Gilbert 1980) of the area *per se* hypothesis. The passive sampling and habitat diversity hypotheses each predict that no correlation should be found between patch area and species richness, whereas, the area *per se* hypothesis predicts that a positive correlation should be found (Kelly *et al.* 1989). The mechanisms underlying the passive sampling and habitat diversity hypothesis cannot be distinguished by this test, however, since diverse rather than uniform habitats were sampled in this study (Connor and McCoy 1979). Evidence of a negative relationship between extinction processes and area, however, would provide indirect support for the habitat diversity hypothesis since no relation should be found if passive sampling were the principal or only causal factor involved (Connor and McCoy 1979).

The results of this study were not consistent with the predictions of the area *per se* hypothesis. Species richness in 10m x 10m quadrats was negatively (rather than positively) correlated with patch area (Spearman Rho: -0.2208, p>1Rho; = 0.0021) and explained less than 2% of the variance in species richness in linear regression (Table 4.1). In keeping with the habitat diversity hypothesis, the number of microhabitats in 10m x 10m quadrats was positively correlated with species richness (Spearman Rho: 0.2460, p>1Rhoi = 0.0006) and environmental variables in multiple linear regression explained 58.2% of the variance in species richness in 10m x10m quadrats (Table 4.1). In keeping with the habitat diversity hypothesis, the number of taxa recorded primarily or exclusively in forested habitats ("Forest" and "Forest + Open", *sensu* section 2.2.3.4) was negatively correlated with patch area (Spearman Rho: ~ 0.2803, p>1Rho = 0.0001). These species are more likely to persist in the forest understory than species of open habitats and thus provide an indirect measure of post-settlement extinction rates in these forests. The negative correlation of apparent

extinction and patch area is contrary to the prediction of both the area *per se* and passive sampling hypotheses. In contrast to the predictions of the habitat diversity hypothesis, however, the number of habitats in 10m x 10m quadrats did not increase with patch area (Spearman Rho: - 0.1949, p>|Rho| = 0.0068).

One apparent explanation for the failure of the area per se hypothesis in this study is the ameliorating effect of dispersal on local extinction rates (Brown and Kodrik-Brown 1977, Holt 1992). Given the predominance of short-distance dispersal in plant communities (Portnoy and Willson 1993, Hughes et al. 1994. Appendix 11), dispersal is expected to be a non-limiting process at the scale of centimetres to metres. In the presence of spatial and temporal heterogeneity, and non-limiting dispersal, differences in plant traits should facilitate persistence by ensuring that there will be some place or time where competitors perform poorly or do not survive and where populations of low abundance may expand (Hutchinson 1961, Levin 1974, Warner and Chesson 1985, Comins and Noble 1985, Hurt and Pacala 1995). Moreover, given that the distance over which plants competeis on the order of centimetres to metres (Harper 1977, Pacala and Silander 1985, 1987, Venable and Brown 1993), the spatial aggregation of conspecifics resulting from short-distance dispersal should sharply diminish the potential for competitive displacement by ensuring that most interactions in a competing plant assemblage occur between individuals of the same species (Schmida and Ellner 1984, Pacala 1986, Lavorel et al. 1994, Rees et al. 1996). Under these conditions, extinction processes in the forest understory should proceed slowly. And, since both immigration and extinction operate primarily at the micro spatial scale, on ecological time scales, there is no apparent reason why extinction processes should proceed more rapidly in forest patches of smaller size.

The failure to detect a positive correlation between species richness and patch area is consistent with the few studies that have controlled for passive sampling (Harner and Harper 1976, Westman 1983. Kelly *et al.* 1989, Holt 1992, Fukamachi *et al.* 1996, and Scariot 1999). These studies encompass a wide variety of habitats: pinyon-juniper communities in Utah and New Mexico; xeric shrublands of the inner Channel Islands. California; beech forest and manuka scrub on islands in Lake Manpouri, New Zealand; a successional field in Kansas; old growth forest reserves in the cool temperate zone of Japan; and, palm communities in isolated forest fragments in central Amazonia.

respectively. The study by Kelly *et al.* (1989) is perhaps the most convincing demonstration of the contribution of passive sampling effects to species-area relations, since it was conducted on islands on which the entire flora had been systematically sampled by proportional sampling (Quinn *et al.* 1987). The percentage of variation in species richness explained by island area declined from 92% with proportional sampling to 17% (beech forest) and 10% (manuka scrub) with fixed-area samples. The latter results were statistically non-significant.

The negative correlation between habitat diversity and patch area in my study was unexpected and may be due to the fixed-area sampling design. For example, whereas the number of habitats at both the quadrat and patch scale declined with area, the increase in the mean number of habitats from quadrat to patch was highly significant (p>|t| = 0.0000; Wilcoxon signed rank sum test). This result conforms to the more traditional pattern when a proportional sampling design has been used.

On balance, the evidence from this study supports the view that the positive correlation that has traditionally been found between patch area and species richness is due primarily to passive sampling effects and to environmental heterogeneity.

4.5 Principal Findings

- Edaphic variables explained 35.6% of the variance in species richness in 10m x 10m quadrats. The variables that maximally explained differences in richness were soil order (r²adj. = 0.253), soil moisture (r²adj. = 0.247), and soil parent material (r²adj. = 0.220).
- 2. Stand structure variables explained 54.9% of the variance in species richness in 10m x10m quadrats. The variables that maximally explained differences in richness were forest cover type (r²adj. = 0.277), # tree species >1m (r²adj. = 0.268), % stems sugar maple (r²adj. = 0.256), % stems sugar maple 0-4 cm dbh (r²adj. = 0.254), # wet-mesic, wet tree species >1m (r²adj. = 0.224), and, # live tree stems 0-4 cm dbh (r²adj. = 0.196).

The variance in species richness explained by sugar maple abundance was substantially higher in undisturbed quadrats on Luvisolic or Brunisolic soils overlying calcareous till. In

these settings, the percentage of tree stems (>1m) that were sugar maple explained 63.0 % of the variance in species richness in 10m x 10m quadrats.

- 3. Human disturbance (lanes, roads, ditches, regenerating fields) explained 12.8% of the variance in species richness in 10m x 10m quadrats. The variables that maximally explained differences in richness were disturbed microhabitats (r'adj. = 0.089), open trails (r'adj. = 0.089), and open regenerating fields (r'adj. = 0.079).
- Environmental heterogeneity, when measured as the number of recorded microhabitats, explained 5.4% of the variance in species richness in 10m x 10m quadrats.
- 5. Environmental heterogeneity, when measured as the contribution to species richness by individual microhabitats, explained the presence of 38.3% of the sampled flora that did not occur on closed, seasonally dry, forest floors, *sensu stricto*.

Microhabitats that contributed maximally to species richness at the quadrat scale, when present, were open lanes/roads (mean # unique species = 24.3); open regenerating fields (mean # unique species = 19.3); closed, seasonally dry, forest floors (mean # unique species = 16.6), open riparian meadows (mean # unique species = 15.3), and, closed, seasonally moist forest floors (mean # unique species = 15.3)

Microhabitats that contributed maximally to species richness at the patch scale, when present, were closed, seasonally dry, forest floors (mean # unique species = 17.1), open regenerating fields (mean # unique species = 12.5), and open lanes/roads (mean # unique species = 9.0).

Thirteen microhabitats contributed to species richness at the landscape scale (i.e. contained species that were not recorded in any other habitat). Microhabitats that contributed the greatest number of species were closed, seasonally dry, floors (17 species), open regenerating fields (8 species), open lanes/roads (6 species), open canopy gaps on seasonally dry forest

floors (6 species), and, closed, seasonally moist, forest floors (5 species).

6. Patch size was negatively correlated with species richness and explained 1.9% of the variance in species richness in 10m x 10m quadrats. This pattern, and significant interactions with important explanatory variables such as soil parent material, soil order, and forest cover type, were interpreted as evidence that patch size, *per se*, did not make an independent contribution to species richness in this study.

The most apparent reasons for the failure of the area *per se* hypothesis to explain differences in species richness were the use of a fixed-area sampling design to control for the effects of passive sampling, and, the spatial scale at which extinction processes operate on ecological time scales.

- 7. Patch isolation, when measured as the mean distance to the nearest 8 forest patches, measured in 45° arcs, was negatively correlated with species richness and non-significant. When measured as the percentage of forest cover within a 5 km by 5 km square centred on the study site, path isolation was significantly and negatively correlated with species and explained 6.6% of the variance in species richness. The latter pattern was not regarded as supporting evidence for the MacArthur-Wilson hypothesis, owing to the strongly non-significant relationship for the more direct test of the immigration hypothesis, and, to significant pair-wise interactions between percent forest cover in the landscape and important explanatory variables such as soil order and forest cover type.
- 8. Dispersal explained 55.8% of the variance in species richness in 10m x 10m quadrats. The variables that maximally explained differences in richness were the percentage of taxa that were herbs dispersed by animal ingestion (r^2adj , = 0.397), ants (r^2adj , = 0.333), and the wind (r^2adj , = 0.317).
- 9. Life history explained 13.2% of the variance in species richness in 10m x 10m quadrats. The variable that maximally explained differences in richness was percentage of taxa that were

perennial herbs (r²adj. = 0.132).

- 10. Provenance explained 26.8% of the variance in species richness in 10m x 10m quadrats. The variable that maximally explained differences in richness was the percentage of taxa that were native herbs (r²adj. = 0.268).
- Life form explained 33.7% of the variance in species richness in 10m x 10m quadrats. The variables that maximally explained differences in richness were the percentage of taxa that were grasses (r²adj. = 0.273) and trees (r²adj. = 0.263).
- 12. Habitat affinity explained 41.4 % of the variance in species richness in 10m x 10m quadrats. The variables that maximally explained differences in species richness were the percentage of taxa classified as having an affinity for "open" and "forest" habitats (r²adj. = 0.388 and 0.262, respectively).
- 13. Shade tolerance explained 32.5% of the variance in species richness in 10m x 10m quadrats. The variables that maximally explained differences in richness were the percentage of taxa classified as being very intolerant, and very tolerant, of shade (r²adj. = 0.293 and 0.214, respectively).
- 14. Moisture affinity explained 30.1% of the variance in species richness in 10m x 10m quadrats. The variables that maximally explained differences in richness were the percentage of taxa classified as having a facultative affinity for wetland habitats (r²adj. = 0.301), and the percentages of taxa classified as having a facultative or obligate affinity for upland habitats (r²adj. = 0.269 and 255, respectively).
- 15. In general, linear regression models composed of dispersal and environmental variables explained more variance than models composed of dispersal or environmental variables alone. For example, mixed model M3 explained 71.2% of the variance in species richness in 10m x 10m quadrats, whereas dispersal model D1 and environmental model E1.

respectively, explained 62.9% and 58.3% of the variance in richness.

A graphical evaluation of alternative models of species richness revealed that while individual variables and models may occupy distinct regions in DCA ordination space. dispersal and environmental variables primarily represent alternative ways of explaining underlying causal factors. The most apparent reason for this convergence is that dispersal is a non-limiting process at the scale of centimetres to metres. A further contributing factor may the tendency for the dispersal and germination biology of plants to converge, owing to the necessity of prior germination success for the transmission of alleles for dispersal.

16. The linear regression models of species richness in this study are sensitive to the taxonomic rank at which they are assessed. The variance in richness explained by a given model, and the quality of the statistical fit, declined at progressively more inclusive ranks. The number of interacting elements typically increased with taxonomic rank, and, depending on the model, peaked at the rank of genus, family or order.

The most apparent cause of the interactions was the progressive inclusion of plant traits at the higher taxonomic ranks. This tendency was most strongly expressed at the rank of order, where more than 50% of the orders in the analysis included more than one mode of dispersal, and where more than 70% of the orders included more than one moisture or light affinity. Taken together, these patterns were interpreted as indirect evidence that phylogeny has contributed to the diversity of taxa recorded in these forests.

5.0 GENERAL CONCLUSIONS

The principal findings and conclusions of the thesis are summarized below. Specific findings related to the distribution of species, and to modes of dispersal, in relation to environmental variables are reported in Chapters 2.5 and 3.5 respectively. Specific findings related to the capacity of environmental variables and plant traits to explain observed differences in species richness are reported in Chapter 4.5.

- 1. Herbs, unlike other life forms, were dispersed by every mode of dispersal recorded in these forests. The reasons for this pattern are beyond the scope of this thesis but suggest in part that the outer integuments that evolved to protect the developing ovule in the angiosperms have been responsive to selection pressure when subsequent modifications facilitated dispersal by various means (Stebbins 1974). The capacity to disperse by one or by many modes, however, did not influence the number of habitats colonized by a given life form.
- 2. The mode by which a given herb dispersed was often associated with the type of fruit it possessed. Achenes, for example, were more likely to be dispersed by the wind than by other modes, whereas, berries and drupes were more likely to be dispersed by animal ingestion. Seeds in capsules were more likely to be dispersed by mechanical expulsion and by animal conveyance, whereas, schizocarps were more likely to be dispersed by adhesion to animals. Nutlets were more likely to be dispersed by unassisted means and by prolonged dormancy in the soil, whereas seeds in capsular fruits were more likely to be dispersed by prolonged dormancy in the soil and by animal conveyance.

The degree to which mode of dispersal was predetermined by the type of fruit a herb possessed was less clear. The berry, drupe, and silique were over-represented in herbs with an affinity for forest habitats whereas the legume was over-represented in herbs with an affinity for open habitats. This tendency is consistent with some measure of control by fruit type. However, the colonization pattern for fleshy fruits may also be explained, in part, by the tendency of seeds of fruits ingested by birds to be excreted in tree-fall gaps and the adjacent forest (Hoppes 1988, Malmborg and Willson 1988).

Evidence consistent with marginal or no control by fruit type among herbs was more pervasive. First, in contrast to the previous pattern, the majority of fruit types in this study (achene, capsular, capsule, follicle, nutlet, schizocarp) were *not* associated with the habitats in which they are typically found. Second, most fruits of herbs in this study were dispersed by several modes. The apparent exceptions were the drupe, which was dispersed by animal ingestion; the silique, which was dispersed by unassisted means; and, the berry, which was dispersed by animal ingestion and by animal conveyance. In contrast, the achene and the capsule were each dispersed by every mode save one (mechanical expulsion and animal ingestion, respectively). The latter fruit types were possessed by 70% of the herbs recorded in this study.

Third, the tendency for certain modes of dispersal to be more frequent in some habitats than others appears to be related more to germination constraints than to fruit type. For example, the tendency for wind-dispersed herbs to be over-represented in open habitats, and under-represented but present in forest habitats, may be explained in part by the differential germination success of large versus small seeds in deep shade (Salisbury 1942, 1974; Baker 1972, Luftensteiner 1979, Mazer 1990, Saverimutto and Westoby 1996). In keeping with this hypothesis, the reported seed mass for congeners of wind-dispersed herbs in this study was consistently heavier for herbs with an affinity for closed versus open habitats (*sensu lato*). The statistical significance of this trend could not established owing to the small sample size. In a similar fashion, the tendency for herbs dispersed by prolonged dormancy to be associated with an affinity for open habitats may also be explained by differential germination success in open versus closed habitats (Brown and Venable 1986, Cohen and Levin 1987, Venable and Brown 1988).

Taken together, the evidence from this study suggests that fruit type exerts little or no control over the mode of dispersal of herbs in the forest understory. Further study of the understory flora in the Great Lakes - St. Lawrence Forest Region (Rowe 1972) is necessary, however, in view of modest sample size (n=234 herbs) associated with this study. This pattern, if widespread, is keeping with the finding by Westoby *et al.* (1992) that mode of dispersal is

rarely constrained by seed size. Taken together, these findings suggest that the habitats in which plants establish will rarely be constrained by the way in which they disperse. This greatly increases the range of solutions (to ecological problems) available to plants.

- 3. The composition of herb assemblages in sampled patches has apparently not been constrained by the *mode* of dispersal, or at least, has not been constrained absolutely. Each mode of dispersal was represented on each portion of the environmental gradients examined, and, with few exceptions, in each type of microhabitat. The exceptions were 7 rare habitats with moist or wet soil conditions: closed seeps: open seeps: closed, seasonally moist tree pits: open, seasonally wet tree pits: open riparian marsh; open riparian thicket. Although most modes were absent from at least one 10m x 10m quadrat, each mode of dispersal was present in all but one of twenty-four sampled patches. These patterns reveal that at least some seeds from each functional group were able to reach most patches and to germinate in the conditions recorded there.
- 4. The habitats in which a given herb was found, however, were associated with the mode by which it dispersed. This suggests that certain modes of dispersal, or traits correlated with dispersal, may be favored in some habitats but not others. Herbs dispersed by animal ingestion, for example, were strongly over-represented (relative to their proportion in the sampled flora) in closed seasonally dry canopy gaps, on open tip-up mounds, and in closed seasonally dry tree pits. In contrast, herbs dispersed by the wind were over-represented in regenerating fields, and, herbs dispersed by prolonged dormancy were over-represented on open lanes and access roads. Herbs dispersed by animal adhesion, ants, mechanical expulsion, unassisted means, and vegetative expansion were over-represented in at least one of the habitats in which they were found.

The degree to which these patterns of association were caused by dispersal, or by plant traits correlated with dispersal, could not be answered with the sampling design applied in this study. However, the habitats in which certain herbs were typically over-represented were consistent with a distribution pattern that had been mediated by dispersal. The indirect

evidence for directed dispersal was strongest for herbs dispersed by birds and mammals: i) patterns of colonization that were consistent with the foraging and seed-processing patterns of frugivorous birds (Thompson and Willson 1978, Willson *et al.* 1982, Malmborg and Willson 1988. Hoppes 1988); ii) patterns of colonization that provided little evidence of differential germination or persistence: e.g., all fleshy-fruited herbs recorded in tree pits or on tip-up mounds, logs, and stumps were observed on forest floors, and, all but two fleshyfruited herbs recorded on forest floors were present in tree pits or on tip-up mounds, logs and stumps; and, iii) patterns of distribution which revealed that the over-representation of fleshy-fruited herbs on these features was due to the greater number of fleshy-fruited species rather than to a greater proportion of taxa that were able to colonize such habitats.

The degree to which dispersal has contributed to the distribution pattern of herbs dispersed by other means is less certain. Herbs dispersed by the wind, for example, were consistently associated with open microhabitats and with human disturbance. While propagules with wings and plumes may travel farther in these habitats than under a closed forest canopy (Hugbes *et al.* 1994), the tendency for wind-dispersed herbs to be poorly represented in forest habitats is equally or more likely to be caused by differential germination success arising from differences in seed weight (Salisbury 1942, 1974, Baker 1972, Luftensteiner 1979, Mazer 1990, Saverimuttu and Westoby 1996). In keeping with this line of reasoning, wind-dispersed herbs with an affinity for "forest" habitats were recorded in significantly more quadrats than were wind-dispersed herbs with an affinity for "open" habitats. This pattern suggests that the distribution of wind-dispersed herbs in the forest understory is constrained more by germination, establishment and persistence than by dispersal.

A similar case can be made for herbs dispersed by prolonged dormancy in the soil. In keeping with recent models of the evolution of dormancy (Venable and Lawlor 1980, Brown and Venable 1986, Cohen and Levin 1987, 1991, Venable and Brown 1988, 1993, Rees 1996), more than 90% of the seed-pooling herbs in this study were species with an affinity for "open" or "open + forest" habitats, and, none of the herbs with an apparent affinity for "forest" habitats was classified as a seed-banking species. Taken together, these patterns

suggest that in forested habitats dispersal in time may be restricted to fugitive species and to sites of recent canopy disturbance. In keeping with this line of reasoning, herbs with the capacity for dormancy were significantly over-represented in the trace cover class (1-5 individuals) and under-represented in the higher cover classes. Herbs with the capacity for prolonged dormancy were found in a wide variety of habitats, however, including closed forest floors. This suggests that these herbs may persist in the understory of second-growth forests for many years. For these herbs, the conditions required for germination and establishment appear to be more limiting than the conditions required for persistence (Grubb 1977, 1988; Grime 1997).

The colonization pattern of herbs dispersed by other modes was consistent with non-directed dispersal and factors governing germination, establishment and persistence. Longer distance dispersal of herbs dispersed by mechanical expulsion and by unassisted means is presumably facilitated by animal ingestion since each species has a sizeable North American range. However, pattern resulting from such dispersal events was not detectable by the sampling design applied in this study.

Taken together, the indirect evidence from this study suggests that distribution of herbs in the forest understory is governed more by differential germination, establishment and persistence than by dispersal. The apparent exceptions are plants dispersed by animal ingestion and by ants. The tendency in this study for modes of dispersal to be associated with particular habitats appears to be due more to plant traits that are correlated with dispersal than to the dispersal process *per se*.

5. The number and composition of plants in the forest understory was strongly influenced by the abundance of sugar maple. Mean species richness (all life forms) in undisturbed 10m x 10m quadrats declined from 46.0 species in quadrats with ≤25% stems (>1 m) sugar maple to 27.1 species in quadrats with ≥75% stems (>1 m) sugar maple. Associated with this 41.1% decline in mean richness was a marked rise in the proportion of prevalent species (≥20% frequency) that flowered before canopy closure (35.2% to 55.6%) and that were shade

tolerant (54.9% to 72.7%).

Curtis (1959) has argued that the forest floor is a demanding environment that requires specialized traits for success and that it is the limited set of species that possess those traits that has led to the striking uniformity in species composition in the mesic hardwood forests of eastern North America. In keeping with this hypothesis, 92.5% of the species that were prevalent in the understory of southern mesic forests in Wisconsin were present in maple dominated forests in this study. Traits that were prominent in each geographic region were early spring flowering and shade tolerance.

Early spring flowering is one of several plant traits that have been associated with deep shade in the forest understory. In herbs with low shade tolerance, early flowering is associated with an cphemeral (Sparling 1967) or winter annual (Rogers 1982) life history. Each facilitates net carbon gain by restricting the growth phase to periods when the canopy is leaf free. More commonly, however, the early flowering habit is associated with varying degrees of shade tolerance that enables shoots and leaves to persist until mid to late summer (Sparling 1967, Rogers 1982). The latter combination of characters was more common in the Peterborough area where only six of sixty-two early flowering species were spring ephemerals (*Allium tricoccum, Caulophyllum thalictroides, Claytonia caroliniana, Dicentra canadensis, Dicentra cucullaria, Erythronium americanum*); only one species (*Galium aparine*) was a known winter annual.

Related plant traits that may facilitate survival in deeply shaded habitats include winter-green leaves (Bierzychudek 1982) and the initiation of shoot growth (Taylor and Pearcy 1976) or flower initials (Bierzychudek 1982) in early autumn. The former trait greatly extends the period of carbon gain in species such as *Carex plantaginea, Hepatica acutiloba, Maianthemum canadense, Tiarella cordifolia, Trientalis borealis, Viola blanda, and Viola rostrata, whereas the latter traits facilitate early spring growth and flowering in species such as <i>Allium tricoccum, Trillium grandiflorum, Arisaema triphyllum, and Geranium maculatum.*
The capacity of plants to tolerate deep shade has been attributed to a suite of traits that facilitate the capture and processing of light energy at the lowest net cost. Morphological characters associated with shade plants include: thin leaves with a large surface area (Grime 1965); a higher proportion of chlorophyll *b* relative to chlorophyll *a* (Boardman 1977); a chloroplast with large grana stacks oriented in more than one plane (Boardman 1977); a higher proportion of leaf nitrogen allocated to chlorophyll than to carboxylating enzymes and other proteins (Seeman *et al.* 1987, Niinemets 1997, Lusk and Reich 2000); a rapid stomatal response to changes in light intensity (Hicks and Cabot 1985); and leaves deployed in horizontal, non-overlapping layers (Grime 1965, Horn 1971). These traits facilitate the capture of energy in low light environments while minimizing the energetic cost to construct and maintain plant tissue. The latter is perceived to be especially important since it results in a lower leaf dark respiration rate and lowers the compensation point for net carbon gain (Grime 1965, Loach 1967, Lambers *et al.* 1983, Lusk and Reich 2000). These characters were not scored directly in this study owing to the lack of a suitable data set.

The foregoing traits, and a large seed (Saverimuttu and Westoby 1996), are expected to be among those essential for germination, growth and reproduction on the forest floor. It is more likely, therefore, that it is traits correlated with dispersal, rather than the dispersal process *per se*, that fundamentally govern the distribution of plants on the forest floor. One important reason for this may be the necessity of prior germination and reproductive success for the transmission of dispersal alleles from parent to offspring.

6. Sugar maple abundance on mesic soils was significantly correlated with available calcium (exchangeable calcium + free calcium carbonate) in the upper 15 cm of the soil profile. When differences in soil parent material, soil order, soil series, and site disturbance were standardized, available calcium explained 42.1% of the variance in the number of sugar maple stems (>1m) in 10m x 10m quadrats and 62.8% of the variance in the percentage of stems that were sugar maple. In keeping with the calcium - mortality bypothesis (Kobe *et al.* 1995, Kobe 1996), calcium availability explained observed differences in the number of sugar maple stems subject to shade stress and self-thinning. In undisturbed stands on

Brunisolic soils overlying calcareous till, differences in available calcium explained 70.9% of the variance in stem number in the 4-10 cm dbh size class and 22.3% of the variance in stem number in the 10-30 cm size class. The decline in explained variance with increasing stem size is consistent with the decline in shade stress experienced by stems as they reach the middle and upper layers of the forest canopy.

Unexpectedly, however, available calcium did not explain the variance in the number of sugar maple stems (>1m) in the 0-4 cm class. These stems are typically the most deeply shaded stems within the forest canopy, and therefore among the stems that would most benefit from a calcium-mediated reduction in leaf dark respiration rates. The soils in this analysis were particularly rich in magnesium, however, and may be causing nutrient stress owing to an imbalance in base cations. To avoid antagonism in the uptake of potassium and magnesium in sugar maple in the Quebec Appalachians, Ouimet and Camire (1995) concluded that the potassium:magnesium ratio in the B soil horizon should be >1. To avoid calcium deficiencies associated with crown die-back, the calcium:magnesium ratio should be > 4. In contrast, the potassium:magnesium ratio was <1 in approximately 50% of the quadrats in this analysis, whereas, the calcium:magnesium ratio was < 4 in approximately 60% of the quadrats. In keeping with a cation imbalance hypothesis, the number of sugar maple stems in the 0-4 cm dbh size class on these soils increased with increasing values of the calcium:magnesium, and magnesium:potassium, ratio, and, declined in the presence of increasing magnesium.

Taken together, these results suggest that the abundance of sugar maple in the forest understory may be limited by the availability, and ratio, of base cations in the soil. This constraint was most apparent on mesic soils and suggests that on the optimal portion of the moisture gradient sugar maple may be secondarily constrained by the availability of base cations.

7. The data from this study suggests that many plants in the forest understory may be responsive to differences in available calcium. Ephemeral spring herbs, for example,

typically occurred on more calcium rich soils (mean concentration = 6.9 cmol/kg) than plants with persistent shoots that flowered prior to, or after, canopy closure. Early spring flowering plants with persistent shoots typically occurred on more calcium rich soils than plants which flowered mid to late season (mean concentration = 5.7 and 5.4 cmol/kg, respectively), but not significantly so. The shade tolerant members of these functional groups, however, occurred on soils that were significantly more calcium rich than species with moderate and low shade tolerance. This pattern suggests that plants in these forests have partitioned the calcium availability gradient in relation to the degree of shade stress to which they were exposed.

The mechanism(s) by which understory plants benefit from calcium rich soils are presently unresolved. Recent studies of cold temperate trees have found an association between foliar calcium levels and dark respiration rates (McLaughlin *et al.* 1991, McLaughlin and Kohut 1992) and sharp reductions in the mortality of sugar maple saplings on calcium rich soils (Kobe *et al.* 1995, Kobe 1996). The greater availability of nitrogen on calcium rich soils (Dancer *et al.* 1973) may be equally or more important for some species, however, and studies have recently been initiated in the U.S. northeast to clarify the relative contribution of calcium and nitrogen to tree growth and mortality relations (Dr. A. Finzi, Department of Biology, Boston University, *pers. com.*; Dr. R.K. Kobe, Department of Forestry, Michigan State University, *pers. com.*).

The greater availability of nitrogen may be particularly important for ephemeral spring herbs since they typically complete their life cycle before the canopy closes (*Allium tricoccum*, the apparent exception, flowers mid to late summer). These species typically have a high light compensation point and a high saturation light intensity (Sparling 1967, Taylor and Pearcy 1976) and may therefore have a greater physiological requirement for nitrogen than more shade tolerant species. The reported affinity of ephemeral spring herbs for base rich, and particularly calcium rich, soils (Curtis 1959, Rogers 1982) may therefore be due in part to the greater availability of nitrogen on these soils.

For shade tolerant herbs, however, the principal mechanism may be a reduction in dark respiration rate. The morphology and physiology of these species typically emphasizes the conservation of reserves rather than photosynthetic performance (Went 1957, Grime 1965, Loach 1967) and thus may benefit more from a reduction in dark respiration rate than from a greater availability in nitrogen. Efficient use of high irradiance requires a high nitrogen investment in carboxylating enzymes and proteins responsible for photosynthetic electron transport (Niinemets 1997). Shade tolerant species, however, typically allocate proportionally more leaf nitrogen to chlorophyll than to carboxylation capacity (Seeman etal. 1987). This investment pattern is thought to be the primary reason why shade tolerant species have a lower respiration rate per unit of leaf N (Lusk and Reich 2000) and an intrinsically low photosynthetic plasticity (Niinemets 1997). In keeping with the latter finding, experimental transfers of plants between high and low light environments have shown that dark respiration rates can change much more rapidly than photosynthetic capacity (Azcon-Bieto and Osmond 1983, Sims and Pearcy 1991). Taken together, these findings suggest that shade tolerant plants on calcareous soils may benefit more from a reduction in dark respiration rate than from a greater availability in nitrogen.

8. The contribution of environmental heterogeneity to species richness was strongly scale dependent. Whereas 21 microhabitats contributed to a significant difference in species richness at the quadrat scale, only 9 microhabitats did so at the scale of the forest patch. In keeping with this pattern, 38 of 39 microhabitats contained at least one unique species when evaluated at the quadrat scale whereas only 13 of 39 microhabitats did so when evaluated at the landscape scale.

In a spatially and temporally variable environment, differences in plant traits are expected to lead to pattern in the distribution of species owing to differential germination, establishment and persistence. In this context, the probability of encountering new species should scale with the size of the sampled area and the scale of heterogeneity in the sampled environment. The observed decline in unique species at increasingly larger spatial scales is consistent with this scaling relationship and suggests that an increasing proportion of the species capable of occupying a given habitat were recorded as the size of the sampled area increased from quadrat to patch to landscape. The sharp decline in the number of contributing habitats in the transition from quadrat to patch suggests that a large fraction of the regional variance in environmental heterogeneity was captured at the scale of the forest patch.

In this study, habitat features created by the death or removal of canopy trees were an important source of environmental heterogeneity at small spatial scales. Features such as logs, stumps, tree pits, tip-up mounds, raised root mats, and canopy openings provided novel resources for the germination and establishment of new species and alternative habitat for species already established on the forest floor. The latter process is expected to promote species coexistence by providing some place or time where competitors perform poorly or do not survive and where populations of low abundance may expand (Hutchinson 1961, Levin 1974, Warner and Chesson 1985, Comins and Noble 1985, Hurtt and Pacala 1995).

In keeping with this expectation, these habitats were colonized by 86% of species recorded on closed, seasonally dry, forest floors, 82% of species recorded in moist or wet habitats, and 78% of species recorded in habitats created by human disturbance. However, the capacity of species to colonize or persist on these features was not uniform: more species were recorded in canopy gaps (282) than on tip-up mounds (180), raised root mats (138), logs (120), tree pits (97), and stumps (64). In these forests, the provision of alternative habitat was far more prevalent than the provision of novel resources for new species since only 20% of the species recorded on these features were not found elsewhere on the forest floor.

Taken together, these results provide support for the hypothesis that heterogeneity facilitates the coexistence of species through the spatial and temporal segregation, and differential performance, of competing species (e.g. Levin 1974, Chesson and Warner 1981, Shrnida and Ellner 1984, Comins and Noble 1985, Warner and Chesson 1985).

9. Dispersal variables explained more variance in species richness in 10m x 10m quadrats than

did environmental variables (r²adj. = 0.6285. Model D1 versus r²adj. = 0.5833. Model E1). The statistical significance of this difference could not be determined, however, owing to interactions among the explanatory variables. The degree to which modes of dispersal and physical attributes of the environment accounted for the same variance was subsequently evaluated indirectly by detrended correspondence analysis (DCA). The results of this analysis revealed that while certain combinations of dispersal and environmental variables may occupy distinct regions in ordination space, they primarily represent alternative ways of explaining the proximate mechanisms of causation. Modes of dispersal are thus strongly correlated with habitat factors and plant traits that govern germination, establishment and persistence. Environmental variables, in turn, are correlated with modes of dispersal that determine the composition and size of the seed rain, the initial conditions that seeds and seedlings must confront, and, therefore, who will interact with whom and with what intensity in competing plant assemblages.

The most apparent explanations for this correlation are: i) the predominance of short-distance dispersal in plant communities (Portnoy and Willson 1993, Hughes *et al.* 1994, Appendix 11); ii) a convergence of germination and dispersal biology, at local spatial scales, owing to the necessity of prior germination success for the transmission of dispersal alleles from parent to offspring (Venable and Lawlor 1980, Olivieri and Berger 1985, Venable *et al.* 1995, Cody and Overton 1996); iii) the presence of habitat factors that favor dispersal by some modes but not others (Cohen and Levin 1987, Venable and Brown 1988, 1993); and, iv) heterogeneity in local conditions governing germination, establishment and persistence (Fowler and Antonovics 1981, Robertson *et al.* 1988, Lechowicz and Bell 1991, Palmer 1990).

Under these conditions, dispersal in the forest understory is primarily a non-limiting process and pattern in the composition and distribution of plant assemblages is due primarily to factors governing germination, establishment and persistence. 10. Dispersal is expected to have profound consequences for populations and communities since it determines the size and composition of the seed rain (Clark and Yi 1995), determines the initial conditions that seeds and seedlings confront (Schupp and Fuentes 1995), determines who interacts with whom, with what intensity, and over what time scale (Atkinson and Shorrocks 1981, Schmida and Ellner 1984, Pacala and Silander 1985, Pacala 1986, 1987, Lavorel *et al.* 1994. Rees *et al.* 1996), influences local extinction rates by affecting the probability that declining or extirpated populations are rescued (Brown and Kodric-Brown1977), influences the rate at which plants colonize new habitat (Halpern *et al.* 1990, Matlack 1994, Kotanen 1997, Brunet and von Oheimb 1998) and the sequences in which they arrive (Drake 1991, Fastie 1995), and, influences the level of gene flow between populations and thus the degree to which neighboring plants are related (Williams and Guries 1994) and genetic variation is structured spatially (Levin 1981, Hamrick *et al.* 1993, Hamrick and Godt 1997).

This suggests that the contribution of dispersal to plant dynamics is both fundamental and pervasive. The failure to find strong evidence of this influence in the composition and distribution of herbs in the understory of sampled forests is therefore surprising. On reflection, this result reflects both the nature of the dispersal process and the choice of methodology for this study.

If most seeds of most plants land within a few metres of the maternal plant, then the principal contribution of dispersal to spatial pattern is the spatial aggregation of conspecifics and the randomization of neighbors in competing plant assemblages. Theory predicts that the former pattern should facilitate the coexistence of species by increasing the frequency of competitive interactions among conspecifics, whereas, the latter pattern should enable populations of low abundance to expand by increasing the probability of reversals in relative competitive strength (Hutchinson 1961, Atkinson and Shorrocks 1981, Chesson and Warner 1981, Shmida and Ellner 1984, Pacala and Silander 1985, Warner and Chesson 1985, Pacala 1986, Lavorel *et al.* 1994, Rees *et al.* 1996). The contribution of the latter mechanism is expected to increase with increasing dispersal distance from the maternal plant and to be influenced

by the scale of spatial and temporal heterogeneity in the local environment (Levin et al. 1984, Comins and Nobel 1985, Chesson 1986, Pacala 1987, Lavorel and Chesson 1995).

This suggests that the principal contribution of dispersal to spatial pattern may occur at the scale of the competing plant assemblage. The degree to which these micro-scale processes have influenced the spatial distribution of plants in the forest understory could not be answered with the methodology applied in this study. The decisions at the outset of this study to map plants rather than propagules, and to evaluate functional groups in relation to microhabitats and environmental gradients rather than individual species in relation to spatial aggregation and competing plant assemblages, precluded subsequent examination of these matters when their potential significance became apparent during preliminary analysis of the data.

The inference that pattern in the composition and distribution of plant assemblages is due primarily to differential germination, establishment and persistence, therefore, understates the contribution of dispersal to plant dynamics and persistence at the micro spatial scale.

Concluding Remarks

The patterns of colonization examined in this study suggest that the principal contribution of dispersal to the spatial distribution, composition, and coexistence of species is made at the scale of the competing plant assemblage and at the scales at which environmental heterogeneity enables populations of low abundance species to expand. If most seeds of most plants land within one to two canopy diameters of the maternal plant, as the current literature suggests, then the principal contributions of dispersal to the spatial distribution, and coexistence, of plants may be the spatial aggregation of conspecifics near the base of the maternal plant, and the randomization of neighbors in adjacent competing plant assemblages. Paradoxically, but perhaps necessarily, the primary contribution to species coexistence, and to reproductive assurance, is made at a spatial scale where differences in the dispersal reach of evolved dispersal morphologies are minimized.

In the forest understory, habitat features created by the death or removal of canopy trees are an important source of environmental heterogeneity at small spatial scales. Canopy openings, tree pits, tip-up mounds, raised root mats, logs and stumps were widespread on the forest floor and apparently well within the dispersal reach of even the least mobile species. Evidence from this study suggests that these features provide novel resources for new species and alternative habitat for species already established on the forest floor. The former contributes to species richness in the forest understory, whereas, the latter may promote species coexistence by providing some place or time where competitors perform poorly, or do not survive, and where populations of low abundance may expand. In this study, the provision of alternative habitat appeared to be the more important process, since only twenty percent of the species recorded on these features were not found elsewhere on the forest floor.

Heterogeneity in site conditions provided novel habitat for 38.3 percent of the sampled flora (158 species) that was not recorded on closed, seasonally dry, forest floors, *sensu stricto*. Habitats that were the most significant contributors to species richness were closed seasonally dry forest floors, habitats created by human disturbance, and, habitats with moist and wet soils. Conditions in these habitats contrast sharply with respect to relative availability of light, moisture, and substrate, and provide the most apparent explanation for the distinctive flora associated with each setting.

The degree to which dispersal has contributed to differences in species richness was not resolved. While there is indirect evidence of directed dispersal by birds in former canopy gaps, and by birds, mammals, and ants, on disturbance features associated with former canopy gaps (tip-up mounds, tree pits, logs, stumps, closed lanes), most propagules appear to be dispersed at random and in close proximity to the parent plant. This suggests that the distribution and richness of plants in a given 10m x 10m quadrat has been determined primarily by factors that govern germination, establishment and persistence.

The capacity for longer distance dispersal, however, may influence the composition of species with a particular set of traits since the seed rain from mobile taxa is more likely to reflect the diversity of species in the surrounding region that are capable of establishing there. The strongest evidence for this source of influence was found in plants dispersed by animal ingestion and by ants. In each case, the mean proportion of plants in 10m x 10m quadrats dispersed by each mode was significantly higher than their proportion in the flora as a whole. These taxa are typically shade tolerant and thus inherently less constrained by understory conditions than species with an affinity for more open habitats. The low shade tolerance of species in open habitats may explain the failure to find a similar pattern in plants dispersed by the wind.

Overall species richness, however, has been governed by other factors, since dispersal by animal ingestion was strongly associated with low species richness whereas dispersal by wind and by prolonged dormancy was strongly associated with high species richness.

On balance, the principal contributors to differences in species richness have been the moisture and base cation status of forest soils. The relative availability of these limiting resources has strongly influenced the composition of the forest canopy and the consequent soil-vegetation complexes have mediated the composition and richness of species by controlling the spatial and temporal availability of light on the forest floor, and, the availability of alternative habitats where competitors may perform poorly or not survive and where populations of low abundance may expand.

On mesic soils, the most significant factor has been the absolute and relative abundance of sugar maple. The number of species in 10m x 10m quadrats was strongly and negatively correlated with each measure. Overall, sugar maple abundance explained 25.6 % of the variance in species richness and up to 63% of the variance in species richness on Brunisolic and Luvisolic soils overlying calcareous till. Mean species richness in undisturbed quadrats declined from 46.0 species in quadrats with $\leq 25\%$ stems (>1m) sugar maple to 27.1 species in quadrats with $\geq 75\%$ stems (>1m) sugar maple to 27.1 species richness was a marked rise in the proportion of species that flowered before canopy closure and that were shade tolerant.

6.0 LITERATURE CITED

- Aarssen, L.W. 1983. Ecological combining ability and competitive combining ability in plants: toward a general evolutionary theory of coexistence in systems of competition. <u>American</u> <u>Naturalist</u>. 122: 707-731.
- Aber, J.D. J.M. Mclillo, K.J. Nadelhoffer, J. Pastor, and R.D. Boone. 1991. Factors controlling nitrogen cycling and nitrogen saturation in northern temperate forest ecosystems. <u>Ecological</u> <u>Applications</u>, 1: 303-315.
- Abrams, M.D. and D.A. Orwig. 1996. A 300-year history of disturbance and canopy recruitment for co-occurring white pine and hemlock on the Allegheny Plateau, U.S.A. <u>Journal of Ecology</u>. 84: 353-363.
- Abrams. P.1983. The theory of limiting similarity. <u>Annual Review of Ecology and Systematics</u>, 14: 359-376.
- Abrams, P.A. 1997, Monotonic or unimodal diversity-productivity gradients: what does competition theory predict?, <u>Ecology</u>, 2019-2027.
- Agnew, A.D.Q. and J.E.C. Flux, 1970. Plant dispersal by hares (*Lepus capensis*) in Kenya. <u>Ecology</u>, 51: 735-737.
- Alex, J.F. 1992. <u>Ontario Weeds</u>. Publication 505. Ontario Ministry of Agriculture and Food. Toronto.
- Al-Mufti, M.M., C.L. Sydes, S.B. Furness, J.P. Grime and S.R. Brand, 1977. A quantitative analysis of shoot phenology and dominance in herbaceous vegetation. <u>Journal of Ecology</u>, 65: 759-791.
- Andersen, A.N. 1988. Dispersal distance as a benefit of myrmecochory. <u>Oecologia</u>, 75: 507-511.
- Andersen, M. 1991. Mechanistic models for the seed shadows of wind-dispersed plants. <u>American</u> <u>Naturalist</u>, 137: 476-497.
- Antonovics, J. and N.C. Ellstrand. 1985. The fitness of dispersed progeny: experimental studies with Anthoxanthum odoratum. In: P. Jacquard et al. (eds.). <u>Genetic Differentiation and Dispersal</u> <u>in Plants</u>. Springer-Verlag. Berlin. pp. 369-381.
- Antonovics, J., N.C. Ellstrand and R.N. Brandon. 1988. Genetic variation and environmental variation: expectations and experiments. In: L.D. Gotleib and S.K. Jain. <u>Plant Evolutionary</u> <u>Biology</u>. Chapman and Hall. London. pp. 275-303.

Arditti, J. 1992. Fundamentals of Orchid Bjology. John Wiley & Sons. New York.

- Arnold, R.M.1981. Population dynamics and seed dispersal of *Chaenorrhinum minus* on railroad cinder ballast. <u>American Midland Naturalist</u>. 106: 80-91.
- Argus, G. 1992. <u>Field Botanists of Ontario Workshop on Willow Identification</u>, 6-7 June 1992. unpublished manuscript. Canadian Museum of Nature. Ottawa.
- Atkinson, W.D. and B. Shorrocks. 1981. Competition on a divided and ephemeral resource: a simulation model. <u>Journal of Animal Ecology</u>, 50: 461-471.
- Atmospheric Environment Service, 1992, <u>Canadian Climate Normals 1961-1990, Ontario</u>, Environment Canada, Ottawa, Ontario.
- Aude, E. and J. Lawesson. 1998. Vegetation in Danish beech forests: the importance of soil, microclimate and management factors, evaluated by variance partitioning. <u>Plant Ecology</u>, 134: 53-65.
- Azcon-Bieto, J. and C.B. Osmond. 1983. Relationship between photosynthesis and respiration: the effect of carbohydrate status on the rate of CO₂ production by respiration in darkened and illuminated wheat leaves. <u>Plant Physiology</u>. 71: 574-581.
- Baker, F.S. 1949. A revised tolerance table. Journal of Forestry, 47: 179-181.
- Baker, G.A. and D.J. O'Dowd. 1982. Effects of parent plant density on the production of achene types in the annual *Hypochoeris glabra*. Journal of Ecology. 70: 201-215.
- Baker, H.G. 1972. Seed weight in relation to environmental conditions in California. <u>Ecology</u>, 53: 997-1010.
- Bakker, D. 1960. A comparative life-history study of Cirsium arvense (L.) Scop. and Tussilago farfara L., the most troublesome weeds in the newly reclaimed polders of the former Zuiderzee. In: J.L. Harper (ed.). <u>The Biology of Weeds</u>. Blackwell Scientific Publications. Oxford. pp. 205-222.
- Barnes, B.V. and W.H. Wagner, Jr. 1991. <u>Michigan Trees A Guide to the Trees of Michigan and the</u> <u>Great Lakes Region</u>. University of Michigan Press. Ann Arbor. MI.
- Baskin, C.C. and J.M. Baskin. 1998. <u>Seeds: Ecology. Biogeography, and Evolution of Dormancy</u> and Germination. Academic Press. San Diego.
- Bazzaz, F.A. 1991. Habitat selection in plants. American Naturalist, 137: S116-S130.

- Beattie, A.J. 1985. <u>The Evolutionary Ecology of Ant-Plant Mutualisms</u>. Cambridge University Press. Cambridge.
- Beattie, A.J. and D.C. Culver, 1981. The guild of myrmecochores in the herbaceous flora of West Virginia forests. <u>Ecology</u>, 62: 107-115.
- Beattie, A.J. and N. Lyons. 1975. Seed dispersal in *Viola* (Violaceae): Adaptations and strategies. <u>American Journal of Botany</u>. 62: 714-722.
- Beattie, A.J., D.C. Culver, and R.J. Pudlo, 1979. Interactions between ants and the diaspores of some common spring flowering herbs in West Virginia. <u>Castanea</u>, 44: 177-186.
- Beatty, S.W. 1984. Influence of microtopography and canopy species on spatial patterns of forest understory plants. <u>Ecology</u>, 65: 1406-1419.
- Beatty, S.W. and O.D.V. Sholes. 1988. Leaf litter effect on plant species composition of deciduous forest treefall pits. <u>Canadian Journal of Forest Research</u>. 18: 553-559.
- Beatty, S.W. and E.L. Stone. 1986. The variety of soil microsites created by tree falls. <u>Canadian</u> Journal of Forest Research. 16: 539-548.
- Beer, T. and M.D. Swaine. 1977. On the theory of explosively dispersed seeds. <u>New Phytologist</u>. 78: 681-694.
- Begon, M., J.L. Harper and C.R. Townsend.1990. <u>Ecology: Individuals, Populations and</u> <u>Communities</u>. Second Edition. Blackwell Scientific Publications. Boston.
- Beke, G.J. and J.A. McKeague. 1984. Influence of windthrow on the properties and classification of selected forested soils from Nova Scotia. <u>Canadian Journal of Soil Science</u>. 64: 195-207.
- Bell, A.D. 1991, <u>Plant Form, An Illustrated Guide to Flowering Plant Morphology</u>, Oxford University Press, Oxford.
- Bell. G. and M.L. Lechowicz. 1991. The ecology and genetics of fitness in forest plants. L. Environmental heterogeneity measured by explant trials. Journal of Ecology. 79: 663-685.
- Bell, G., M.J. Lechowicz and D.J. Schoen. 1991. The ecology and genetics of fitness in forest plants. III. Environmental variance in natural populations in *Impatiens pallida*. Journal of Ecology. 79: 697-713.
- Bengtsson, J., T. Fagerstrom and H. Rydin, 1994. Competition and coexistence in plant communities. <u>Trends in Ecology and Evolution</u> (TREE), 9: 246-250.

- Berg, R.Y. 1966. Seed dispersal of *Dendromecon*: its ecologic. evolutionary, and taxonomic significance. <u>American Journal of Botany</u>. 53: 61-73.
- Berg, R.Y. 1969. Adaptations and evolution in *Dicentra* (Fumariaceae), with special reference to seed, fruit and dispersal mechanism. <u>Nytt Magasin for Botanikk</u>, 16: 49-75.
- Berg, R.Y. 1975. Myrmecochorous plants in Australia and their dispersal by ants. 1975. <u>Australian</u> Journal of Botany, 23: 475-508.
- Berg, R.Y. 1983. Plant distribution as seen from plant dispersal: general principles and basic modes of plant dispersal. In K. Kubitzi (ed.). Dispersal and Distribution: An International Symposium. Sonderbande des Naturwissenschaftlichen Vereins in Hamburg, 7, pp. 13-36.
- Berg, R.Y. 1985. Myrmecochorous plants in Australia and their dispersal by ants. <u>Australian Journal</u> of Botany, 23: 475-508.
- Bierzychudek, P. 1982. Life histories and demography of shade-tolerant temperate forest herbs: a review. <u>New Phytologist</u>. 90: 757-776.
- Boardman, N.K. 1977. Comparative photosynthesis of sun and shade plants. <u>Annual Review of</u> <u>Plant Physiology</u>, 28: 355-377.
- Bloom, A.L. 1978. <u>Geomorphology A Systematic Analysis of Late Cenozoic Landforms</u>. Prentice-Hail. Englewood Cliffs, N.J.
- Boerner, R.E.J. 1984. Foliar nutrient dynamics and nutrient use effeciency of four deciduous tree species in relation to site fertility. Journal of Applied Ecology, 21: 1029-1040.
- Boerner, R.E.J. and S.D. Koslowsky. 1989. Microsite variations in soil chemistry and nitrogen mineralization in a beech-maple forest. <u>Soil Biology and Biogeochemistry</u>. 21: 795-801.
- Bolker, B.M. and S.W. Pacala, 1999. Spatial moment equations for plant competition: understanding spatial strategies and the advantages of short dispersal. <u>American Naturalist</u>, 153: 575-602.
- Boman, J.S. and B.B. Casper. 1995. Postdispersal seed predation in disturbed and intact temperate forest. <u>American Midland Naturalist</u>, 134: 107-116.
- Borcard, D., P. Legendre and P. Drapeau. 1992. Partialling out the spatial component of ecological variation. <u>Ecology</u>. 73: 1045-1055.
- Bormann, B.T., H. Spaltenstein, M.H. McClellan, F.C. Ugolini, K. Cromack Jr. and S.M. Nay. 1995. Rapid soil development after windthrow disturbance in pristine forests. <u>Journal of Ecology</u>. 83: 747-757.

- Boyd, R.S. 1996. Ant-mediated seed dispersal of the rare chaparral shrub Fremontodendron decumbens (Sterculiaceae). Madrono. 43: 299-315.
- Brady, N.C. 1990. <u>The Nature and Properties of Soils</u>. Tenth Edition. Macmillan Publishing Company, New York.
- Brandon, R.N. 1990. Adaptation and Environment. Princeton University Press. Princeton.
- Bratton, S.P. 1976. Resource division in an understory herb community: responses to temporal and microtopographic gradients. <u>American Naturalist</u>, 110: 679: 693.
- Braun, E. L. 1950. Deciduous Forests of Eastern North America. The Free Press, New York,
- Bresinsky, A. 1963. Bau, Entwicklungsgeschichte und Inhaltsstoffe der Elaiosomen. <u>Bibliotheea</u> <u>Botanica</u>, 126: 1-54.
- Brew, C.R., D.J. O'Dowd, and I.D. Rae. 1989. Seed dispersal by ants: behaviour-releasing compounds in elaiosomes. <u>Oecologia</u>, 80: 490-497.
- Brewer, R. 1980. A half-century of changes in the herb layer of a climax deciduous forest in Michigan. <u>Journal of Ecology</u>, 68: 823-832.
- Brodie, H.J. 1951. The splash-cup mechanism in plants. Canadian Journal of Botany, 29: 224-234,
- Brodie, H.J. 1955. Springboard plant dispersal mechanisms operated by rain. <u>Canadian Journal of</u> <u>Botany</u>, 33: 156-167.
- Brown, J.H. and A. Kodric-Brown, 1977. Turnover rates in insular biogeography: effect of immigration on extinction. <u>Ecology</u>, 58: 445-449.
- Brown, M.J. and G.G. Parker. 1994. Canopy light transmittance in a chronosequence of mixedspecies deciduous forests. <u>Canadian Journal of Forest Research</u>. 24: 1694-1703.
- Brown, J.S. and D.L. Venable. 1986. Evolutionary ecology of seed-bank annuals in temporally varying environments. <u>American Naturalist</u>, 127: 31-47.
- Brunet, J. and G. von Oheimb. 1998. Migration of vascular plants to secondary woodlands in southern Sweden. Journal of Ecology. 86: 429-438.
- Buell, M.F., A.N. Langford, D.W. Davidson and L.F. Ohmann, 1966. The upland forest continuum in northern New Jersey. Ecology, 47: 416-432.
- Bullock, S.H. 1989. Life history and seed dispersal of the short-lived chaparral shrub Dendromecon

rigida (Papavaraceae). American Journal of Botany, 76: 1506-1517.

- Bullock, S.H. and R.B. Primack. 1977. Comparative experimental study of seed dispersal on animals. <u>Ecology</u>. 58: 681-686.
- Bulmer, M.G. 1984. Delayed germination of seeds: Cohen's model revisited. <u>Theoretical Population</u> <u>Biology</u>. 26: 367-377.
- Bulow-Olsen, A. 1984. Diplochory in *Viola*: A possible relation between seed dispersal and soil seed bank. <u>American Midland Naturalist</u>, 112: 251-260.
- Burnside, O.C., R.G. Wilson, S. Weisberg, and K.G. Hubbard. 1996. Seed longevity of 41 weed species buried 17 years in eastern and western Nebraska. <u>Weed Science</u>, 44: 74-86.
- Burrows, F.M. 1986. The aerial motion of seeds. fruits, spores and poilen. In: D.R. Murray (ed.), Seed Dispersal. Academic Press. pp. 1-47.
- Burton, P.J. 1989. Constraints to tree invasion on a nutrient rich site during old-field succession. Ph.D. dissertation. University of Illinois. Urbana. (Cited in Greene, D.A. and E.A. Johnson 1996. op.cit.)
- Busing, R.T. and P.S. White. 1997. Species diversity and small-scale disturbance in an old growth temperate forest: a consideration of gap partitioning concepts. <u>Oikos</u>. 78: 562-568.
- Bustamente, R.O., J.A. Simonetti, and J.E. Mella. 1992. Are foxes legitimate and efficient seed dispersers? A field test. <u>Acta Oecologica</u>, 13: 203-208.
- Cain, M.L., H. Damman, and A. Muir. 1998. Seed dispersal and the holocene migration of woodland herbs. <u>Ecological Monographs</u>. 68: 325-347.
- Cain, M.L., S. Subler, J.P. Evans and M. Fortin. 1999. Sampling spatial and temporal variation in soil nitrogen availability. <u>Oecologia</u>. 118: 397–404.
- Cain, M.L., B.G. Milligan and A.E. Strand. 2000. Long-distance seed dispersal in plant populations. American Journal of Botany. 87: 1217-1227.
- Caley, M.J. and D. Schluter, 1997. The relationship between local and regional diversity. <u>Ecology</u>, 78: 70-80.
- Campbell, C.S. 1983. Wind disperal of some North American species of *Andropogon* (Gramineae). <u>Rhodora</u>. 85: 65-72.

Canada Soil Survey Committee. 1978. The Canadian System of Soil Classification. Publication

1646. Research Branch, Canada Department of Agriculture. Minister of Supply and Services Canada. Hull, Quebec.

- Canham, C.D., J.S. Denslow, W.J.Platt, J.R. Runkle, T.A. Spies and P.S. White. 1990. Light regimes beneath closed canopies and tree-fall gaps in temperate and tropical forests. <u>Canadian Journal of Forest Research</u>, 20: 620-631.
- Canham, C.D., A.C. Finzi, S.W. Pacala, and D.H. Burbank. 1994. Causes and consequences of resource heterogeneity in forests: interspecific variation in light transmission by canopy trees. <u>Canadian Journal of Forest Research</u>, 1994, 24: 337-349.
- Carey, P.D. and A.R. Watkinson. 1993. The dispersal and fates of seeds of the winter annual Vulpia ciliata. Journal of Ecology. 81: 759-767.
- Casper. B.B. 1987. Spatial patterns of seed dispersal and postdispersal seed predation of *Cryptantha flava* (Boraginaceae). <u>American Journal of Botany</u>. 74: 1646-1655.
- Caspersen, J.P., J.A. Silander, C.D. Canham and S.W. Pacala. 1999. Modeling the competitive dynamics and distribution of tree species along moisture gradients. In: Mladenoff, D.J. and W.L. Baker (eds.). 1999. <u>Spatial Modeling of Forest Landscape Change: Approaches and Applications</u>. Cambridge University Press. Cambridge. pp. 14-41.
- Chan, S.S., R.W. McCreight, J.D. Walstad, and T.A. Spies. 1986. Evaluating forest vegetative coverwith computerized analysis of fisheye photographs. <u>Forest Science</u>, 32: 1085-1091.
- Chesson, P.L. 1985. Coexistence of competitors in spatially and temporally varying environments: a look at the combined effects of different sorts of variability. <u>Theoretical Population</u> <u>Biology</u>. 28: 263-287.
- Chesson, P.L. 1986. Environmental variation and the coexistence of species. In: Diamond, J. and T.J. Case (eds.). 1986. <u>Community Ecology</u>. Harper and Row. New York, pp. 240-256.
- Chesson, P.L. 1991. A need for niches?. Trends in Ecology and Evolution. 6: 26-28.
- Chesson, P.L. and T.J. Case, 1986. Overview: non-equilibrium community theories: chance. variability, history and coexistence. In: Diamond, J. and T.J. Case (eds.), 1986. <u>Community</u> <u>Ecology</u>, Harper and Row, New York, pp. 229-239.
- Chesson, P.L. and R.R. Warner. 1981. Environmental variability promotes coexistence in lottery competitive systems. <u>American Naturalist</u>. 117: 923-943.

- Clark, J.S. and Y. Ji. 1995. Fecundity and dispersal in plant populations: implications for structure and diversity. <u>American Naturalist</u>, 146: 72-111.
- Clark, J.S., E. Macklin and L. Wood. 1998. Stages and spatial scale of recruitment limitation in southern Appalachiam forests. <u>Ecological Monographs</u>. 68: 213-235.
- Cody, M.L. and J.M. Overton. 1996. Short-term evolution of reduced dispersal in island plant populations. Journal of Ecology. 84: 53-61.
- Cody, W.J. and D.M. Britton. 1989. <u>Ferns and Fern Allies of Canada</u>. Publication 1829/E. Agriculture Canada. Ottawa.
- Cohen, D. and S.A. Levin. 1987. The interaction between dispersal and dormancy strategies in varying and heterogeneous environments. In: Teramoto, E. and Y. Yamaguti (eds). <u>Lecture</u> <u>Notes in Biomathematics</u>. Volume 71. Springer-Verlag. Berlin. pp.110-122.
- Cohen, D. and S.A. Levin, 1991, Dispersal in patchy environments: the effects of temporal and spatial structure. <u>Theoretical Population Biology</u>, 39: 63-99.
- Coleman, B.D. 1981. On random placement and species-area relations. <u>Mathematical Biosciences</u>, 54: 191-215.
- Collinge, W.E. 1913. The destruction and dispersal of weed seeds by wild birds. <u>Journal of the</u> <u>Board of Agriculture (Great Britain)</u>. 20: 15-26.
- Collins, B.S. and S.T.A. Pickett. 1987. Influence of canopy openings on the environment and herblayer in a northern hardwood forest. <u>Vegetatio</u>, 70: 3-10.
- Collins, B.S. and S.T.A. Pickett. 1988a. Response of herb layer cover to experimental canopy gaps. <u>American Midland Naturalist</u>, 119: 282-290.
- Collins, B.S. and S.T.A. Pickett. 1988b. Demographic responses of herb layer species to experimental canopy gaps in a northern hardwoods forest. Journal of Ecology. 76: 437-450.
- Comins, H.N. and I.R. Noble. 1985. Dispersal, variability, and transient niches: species coexistence in a uniformly variable environment. <u>American Naturalist</u>. 126: 706-723.
- Connell, J.H. 1978. Diversity in tropical rain forests and coral reefs. Science, 199: 1302-1310.
- Connell, J.H. and R.O. Slayter. 1977. Mechanisms of succession in natural communities and their role in community stability and organization. <u>American Naturalist</u>. 111: 1119-1144.

- Connor, E.F. and E.D. McCoy. 1979. The statistics and biology of the species-area relationship. <u>American Naturalist</u>. 113: 791-833.
- Cornell, H.V. and R.H. Karlson. 1997. Local and regional processes as controls of species richness. In: Tilman D. and P. Kareiva (eds.). 1997. <u>Spatial Ecology: The Role of Space in Population</u> <u>Dynamics and Interspecific Interactions</u>. Princeton University Press, Princeton, pp. 250-268.
- Cornell, H.V. and J.H. Lawton. 1992. Species interactions. local and regional processes. and limits to richness of ecological communities: a theoretical perspective. <u>Journal of Animal Ecology</u>. 61: 1-12.
- Corner, E.J.H. 1976. The Seeds of Dicotyledons. Cambridge University Press. Cambridge.
- Cote, B. and J.W. Fyles. 1994. Leaf litter disappearance of hardwood species of southern Quebec: interaction between litter quality and stand type. <u>Ecoscience</u>, 1: 322-328.
- Crawley, M.J. 1990. The population dynamics of plants. <u>Philosophical Transactions Royal Society</u> of London B. 330: 125-140.
- Creighton, W.S. 1950. <u>The Ants of North America</u>. Bulletin of the Museum of Comparative Zoology at Harvard College. Volume 104. Cambridge.
- Crocker, R.L. and J.Major, 1955. Soil development in relation to vegetation and surface age at Glacier Bay, Alaska, Journal of Ecology, 43: 427-448.
- Crockett, L.J. 1977. Wildly Successful Plants. A Handbook of North American Weeds. Collier Macmillan Publishers. London.
- Crozier, C.R. and R.E.J. Boerner. 1984. Correlations of understory herb distribution patterns with microhabitats under different tree species in a mixed mesophytic forest. <u>Oecologia</u>, 62: 337-343.
- Crozier, C.R. and R.E.J. Boerner. 1986. Stemflow induced soil nutrient heterogeneity in a mixed mesophytic forest. <u>Bartonia</u>. 52: 1-8.
- Culver, D.C. and A.J. Beattie. 1978. Myrmecochory in *Viola*: dynamics of ant-seed interactions in some West Virginia species. <u>Journal of Ecology</u>, 66: 53-72.
- Culver, D.C. and A.J. Beattie, 1980. The fate of *Viola* seeds dispersed by ants. <u>American Journal of</u> <u>Botany</u>, 67: 710-714.
- Curtis, J.T. 1956. The modification of mid-lattitude grasslands and forests by man. <u>International</u> <u>Symposium on Man's Role in Changing the Earth</u>, Princeton, pp. 721-736.

- Curtis, J.T. 1959. <u>The Vegetation of Wisconsin. An Ordination of Plant Communities</u>. Second Printing. University of Wisconsin Press. Madison.
- Dancer, W.S., L.A. Peterson and G. Chesters. 1973. Ammonification and nitrification of N as influenced by soil pH and previous N treatments. <u>Soil Science Society of America</u> <u>Proceedings</u>. 37: 67-69.
- Dansereau, P. and K. Lems, 1957. <u>The Grading of Dispersal Types in Plant Communities and their</u> <u>Ecological Significance</u>. Institute Botanique de L'Universite de Montreal. Montreal.
- Darley-Hill, S. and W.C. Johnson. 1981. Acom dispersal by the blue jay (Cvanocitta cristata). Oecología. 50: 231-232.
- David, M.B. 1988. Use of loss-on-ignition to assess soil organic matter carbon in forest soils. <u>Communications in Soil Science Plant Annals</u>, 19(14): 1593-1599.
- Davidson, D.W. and S.R. Morton. 1981. Myrmecochory in some plants (F. chenopodiaceae) of the Australian arid zone. <u>Oecologia</u>, 50: 357-366.
- Davies, B.E. 1974. Loss-on-ignition as an estimate of soil organic matter. <u>Soil Science Society of</u> <u>America Proceedings</u>, 38: 150-151.
- Davis, M.B. 1981a. Quaternary history and the stability of forest communities. In: D.C. West, H.H. Shugart and D.B. Botkin (eds.), 1981. <u>Forest Succession: Concepts and Application</u>, Springer-Verlag, New York.
- Davis, M.B. 1981b. Outbreaks of forest pathogens in Quaternary history. <u>International Palynology</u> <u>Conference, Lucknow</u>. 3: 216-228.
- Davis, M.B. 1983. Quaternary history of deciduous forests of eastern North America and Europe. <u>Annals of the Missouri Botanical Garden</u>, 70:550-563.
- Davis, M.B. 1986. Climatic instability, time lags and community disequilibrium. In: Diamond, J. and T.J. Case (eds.), 1986. <u>Community Ecology</u>, Harper and Row, New York, pp. 269-284.
- Davis, M.B., S. Sugita, R.R. Calcote, J.B. Ferrari and L.E. Frelich. 1994. Historical development of alternate communities in a hemlock-hardwood forest in northern Michigan. USA. In: R. May, N. Webb and P. Edwards (eds.). <u>Large-Scale Ecology and Conservation Biology</u>. Blackwell Science. Oxford. pp. 19-39.
- Diekmann, M. and U. Falkengren-Grerup. 1998. A new species index for forest vascular plants: development of functional indices based on mineralization rates of various forms of soil nitrogen. <u>Journal of Ecology</u>. 86: 269-283.

- Dore, W.G. and J. McNeill, 1980, <u>Grasses of Ontario</u>. Monograph 26. Research Branch, Agriculture Canada, Ottawa,
- Drake, J.A. 1991. Community-assembly mechanics and the structure of an experimental species ensemble. <u>American Naturalist</u>. 137: 1-26.
- Duncan, R.P. H.L. Hannah, S.C. Urlich, G.H. Stewart and J. Geritzlehner. 1998. Small-scale species richness in forest canopy gaps: the role of niche limitation versus the size of the species pool. Journal of Vegetation Science, 9: 455-460.
- Dwyer, L.M. and G. Merriam. 1981. Influence of togographic heterogeneity on deciduous litter decomposition. <u>Oikos</u>. 37: 228-237.
- Egler, F.E. 1954. Vegetation science, concepts I. Initial floristic composition, a factor in old-field vegetation development. <u>Vegetatio</u>, 4: 412-417.
- Ehrenfeld, J.G. 1980. Understory response to canopy gaps of varying size in a mature oak forest. Bulletin of the Torrev Botanical Club. 107: 29-41.
- Ehrlen, J. and O. Eriksson, 2000. Dispersal limitation and patch occupancy in forest herbs. <u>Ecology</u>. 81: 1667-1674.
- Ellenberg, H. 1988, <u>Vegetation Ecology of Central Europe</u>. Fourth Edition. Cambridge University Press, Cambridge.
- Elliott, L. 1978. <u>Social Behavior and Foraging Ecology of the Eastern Chipmunk (Tamias striatus)</u> in the Adirondack Mountains. Smithsonian Contributions to Zoology. Number 265. Smithsonian Institution Press. Washington.
- Ellner, S. and A. Schmida, 1981. Why are adaptations for long-range dispersal rare in desert plants. Oecologia, 51: 133-144.
- Energy, Mines and Resources Canada. 1990a. <u>Canada Climatic Regions Thornwaite Classification</u> <u>Moisture Regions</u>. National Atlas Information Service. Geographical Services Division. Canada Centre for Mapping. Ottawa.
- Energy, Mines and Resources Canada. 1990b. <u>Canada Climatic Regions Thornwaite Classification</u> <u>Thermal Efficiency Regions</u>. National Atlas Information Service. Geographical Services Division, Canada Centre for Mapping. Ottawa.
- Environment Canada. 1993. <u>Canadian Climate Normals 1961-1990</u>. Ontario. Ministry of Supply and Services Canada.

Eriksson, E. 1993. The species-pool hypothesis and plant community diversity. <u>Oikos</u>. 68: 371-374.

- Eriksson, O. and J. Ebrlen. 1992. Seed and microsite limitation of recruitment in plant populations. Oecologia. 91: 360-364.
- Evans, R.A., H.H. Biswell, and D.E. Palmquist. 1987. Seed dispersal in *Ceanothus cuneatus* and C. *leucodermis* in a Sierran oak-woodland savanna. <u>Madrono</u>, 34: 283-293.
- Fahey, T.J., J.J. Battles, and G.F. Wilson. 1998. Responses to early successional northern hardwood forests to changes in nutrient availability. <u>Ecological Monographs</u>. 68: 183-212.
- Fahn, A. and E. Werker. 1972. Anatomical mechanisms of seed dispersal. In: T.T. Kozlowski (ed.). Seed Biology. Volume 1. Academic Press. New York. pp.151-221.
- Farrar, J.L. 1995. <u>Trees in Canada</u>. Fitzhenry & Whiteside Limited and the Canadian Forest Service. Canada Communications Group-Publishing. Supply and Services Canada. Ottawa.
- Fastie, C.L. 1995. Causes and consequences of multiple pathways of primary succession at Glacier Bay, Alaska. <u>Ecology</u>. 76: 1899-1916.
- Fenner, M. 1985. Seed Ecology, Chapman and Hall, New York.
- Fenster, C.B. 1991. Gene flow in *Chamaecrista fasciculata* (Leguminosae) I. Gene dispersal. <u>Evolution</u>, 45: 398-409.
- Finegan, B. 1984, Forest succession, Nature, 312: 109-114.
- Fisher, T.R. 1988. <u>The Dicotyledoncae of Ohio. Part 3: Asteraceae</u>. Ohio State University Press. Columbus.
- Fitter, A.H. and R.K.M. Hay, 1987. <u>Environmental Physiology of Plants</u>. Second Edition. Academic Press. London.
- Flaming, V. and V.W. Proctor. 1968. Dispersal of aquatic organisms: viability of seeds recovered from the droppings of captive killdeer and mallard ducks. <u>American Journal of Botany</u>, 55: 20-26.
- Flint, S.D. and I.G. Palmblad, 1978. Germination dimorphism and developmental flexibility in the ruderal weed *Heterotheca grandiflora*. <u>Oecologia</u>, 36: 33-43.
- Fowler, N.L. and J. Antonovies. 1981. Small-scale variability in the demography of transplants of two herbaceous species. <u>Ecology</u>, 62: 1450-1457.

- Fox, J.F. 1977. Alternation and coexistence of tree species. American Naturalist, 111: 69-89.
- Frelich, L.E. and C.G. Lorimer. 1991. Natural disturbance regimes in hemlock-hardwood forests of the upper Great Lakes region. <u>Ecological Monographs</u>. 61: 145-164.
- Frelich, L.E., R.R. Calcote, M.B. Davis and J. Pastor. 1993. Patch formation and maintenance in an old-growth hemlock-hardwood forest. <u>Ecology</u>. 74: 513-527.
- Frenkel, R.E. 1970. <u>Ruderal Vegetation Along Some California Roadsides</u>. University of California Press. Berkely.
- Fukamachi, K. S. Iida, and T. Nakashizuka. 1996. Landscape patterns and plant species diversity of forest reserves in the Kanto region, Japan. <u>Vegetatio</u>, 124: 107-114.
- Gaddy, L.L. 1986. Twelve new ant-dispersed species from the southern Appalachians. <u>Bulletin of</u> <u>the Torrey Botanical Club</u>, 113: 247-251.
- Gashwiler, J.S. 1969. Seed fall of three conifers in west-central Oregon. Forest Science, 15: 290-295.
- Gates. B.N. 1941. Observations in 1940 on the dissemination by ants of the seeds of *Trillium* grandiflorum. <u>Rhodora</u>. 43: 206-207.
- Gates, B.N. 1942. The dissemination by ants of the seeds of bloodroot. *Sanquinaria canadensis*. <u>Rhodora</u>, 44: 13-15.
- Gause, G.F. 1934. The Struggle for Existence. Hafner, New York,
- Geiger, R. 1961. <u>The Climate Near the Ground</u>. Fourth Edition. Harvard University Press. Cambridge.
- Geritz, S.A.H., J.A.J. Metz, P.G.L. Klinkhamer and T.J. De Jong. 1988. Competition in safe sites. <u>Theoretical Population Biology</u>. 33: 161-180.
- Gilbert, F.S. 1980. The equilibrium theory of island biogeography: fact or fiction?. Journal of Biogeography. 7: 209-235.
- Gillespie, J.E. and C.J. Acton. <u>Soils of Peterborough County</u>. Report No. 45. Ontario Institute of Pedology. Agriculture Canada Research Branch. Ontario Ministry of Agriculture and Food. Department of Land Resource Science University of Guelph.
- Gillespie, J.H. and M. Turelli. 1989. Genotype-environment interactions and the maintenance of polygenic variation. <u>Genetics</u>. 121: 129-138.

- Gleason, H.A. 1952. <u>The New Britton and Brown Illustrated Flora of the Northeastern United States</u> <u>and Adjacent Canada</u>. 3 Volumes. Hafner Press, New York.
- Gleason, H.A. and A. Cronquist. 1991. <u>Manual of Vascular Plants of Northeastern United States and</u> <u>Adjacent Canada</u>. Second Edition. New York Botanical Garden. Bronx.
- Glenn-Lewin, D.C., R.K. Peet and T.T. Veblen (eds.), 1992. <u>Plant Succession: Theory and</u> <u>Prediction</u>. Chapman and Hall, London.

Gimingham, C.H. 1972. Ecology of Heathlands. Chapman and Hall. London.

- Goldberg, D.E. and A.M. Barton. 1992. Patterns and consequences of interspecific competition in natural communities: a review of field experiments with plants. <u>American Naturalist</u>. 139: 771-801.
- Goldberg, D.E. and T.E. Miller. 1990. Effects of different resource additions on species diversity in an annual plant community. <u>Ecology</u>. 71: 213-225.
- Goldblum, D. 1997. The effects of treefall gaps on understory vegetation in New York State. <u>Journal</u> of Vegetation Science. 8: 125-132.
- Gordon, D.M. 1983. Dependence of necrophoric response to oleic acid on social context in the ant *Pogonomyrmex badius*. Journal of Chemical Ecology, 9: 105-111.
- Gould, S.J. and E.S. Vrba. 1982. Exaptation a missing term in the science of form. <u>Paleobiology</u>. 8: 4-15.
- Graber, R.E. and D.F. Thompson. 1978. <u>Seeds in the Organic Layers and Soil of Four Beech-Birch-Maple Stands</u>. Forest Service Research Paper NE-401. Forest Service, U.S. Department of Agriculture. Northeastern Forest Experiment Station. Brumall. PA.
- Grace, J.B. 1999. The factors controlling species density in herbaceous plant communities: an assessment. <u>Perspectives in Plant Ecology, Evolution and Systematics</u>. 2: 1-28.
- Greene, D.A. and E.A. Johnson, 1989. A model of wind dispersal of winged or plumed seeds, <u>Ecology</u>, 70: 339-347.
- Greene, D.A. and E.A. Johnson. 1995. Long-distance wind dispersal of tree seeds. <u>Canadian Journal</u> of <u>Botany</u>, 73: 1036-1045.
- Greene. D.A. and E.A. Johnson. 1996. Wind dispersal of seeds from a forest into a clearing. Ecology. 77:595-609.

- Greene, D.A. and E.A. Johnson, 1997. Secondary dispersal of tree seeds on snow. <u>Journal of</u> <u>Ecology</u>, 85: 329-340.
- Grime, J.P. 1965. Shade tolerance in flowering plants. Nature, 208: 161-163.
- Grime, J.P. 1973. Competitive exclusion in herbaceous vegetation. Nature, 344-347.
- Grime, J.P. 1979. Plant Strategies and Vegetation Processes. John Wiley and Sons. Toronto.
- Grime, J.P., K. Thompson, R.Hunt, J.G. Hodgson, J.H.C. Cornelissen, I.H. Rorison, G.A.F. Hendry, T.W. Ashenden, A.P. Askew, S.R. Band, R.E. Booth, C.C. Bossard, B.D. Campbell, J.E.L. Cooper, A.W. Davison, P.L. Gupta, W. Hall, D.W. Hand, M.A. Hannah, S.H. Hillier, D.J. Hodkinson, A. Jalili, Z. Liu, J.M.L. Mackey, N. Matthews, M.A. Mowforth, A.M. Neal, R.J. Reader, K Reiling, W. Ross-Fraser, R.E. Spencer, F. Sutton, D.E. Tasker, P.C. Thorpe, and J. Whitehouse, 1997. Integrated screening validates primary axes of specialization in plants. <u>Oikos</u>, 79, 259-281.
- Grinnell, J. 1904. The origin and distribution of the chestnut-backed chickadee. Auk: 21: 364-382.
- Gross, K.L. 1986. personal communication, cited in Okubo and Levin (1989), op.cit.
- Gross, K.L. and P.A. Werner. 1982. Colonizing abilities of "biennial" plant species in relation to ground cover: implications for their distributions in a successional sere. <u>Ecology</u>, 63: 921-931.
- Grubb, P.J. 1977. The maintenance of specie-richness in plant communities: the importance of the regeneration niche. <u>Biological Reviews</u>, 52: 107-145.
- Grubb, P.J. 1986. Problems posed by sparse and patchily distributed species in species-rich plant communities. In: Diamond, J. and T.J. Case (eds.). 1986. <u>Community Ecology</u>. Harper and Row. New York. pp. 207-225.
- Gunther, R.W. and J. Lanza. 1989. Variation in attractiveness of *Trillium* diaspores to a seeddispersing ant. <u>American Midland Naturalist</u>. 122: 321-328.
- Gurevitch, J., L.L. Morrow, A. Wallace and J.S. Walsh. 1992. A meta-analysis of competition in field experiments. <u>American Naturalist</u>, 140: 539-572.
- Hale, C.M. and J. Pastor. 1998. Nitrogen content, decay rates and decompositional dynamics of hollow versus solid hardwood logs in hardwood forests of Minnesota. U.S.A. <u>Canadian</u> <u>Journal of Forest Research</u>. 28: 1276-1285.
- Hamilton, W.J. 1941. The food of small forest animals in eastern United States. Journal of

Mammology, 22: 250-263.

- Hamrick, J.L. and M.J.W. Godt. 1997. Effects of life history traits on genetic diversity in plant species. In: Silvertown, J., M. Franco. and J.L. Harper. <u>Plant Life Histories. Ecology.</u> <u>Phylogeny and Evolution</u>. Cambridge University Press. Cambridge. pp.102-118.
- Hamrick, J.L., D.A. Murawski, and J.D. Nason. 1993. The influence of seed dispersal mechanisms on the genetic structure of tropical tree populations. <u>Vegetatio</u>. 107/108: 281-297.
- Handel, S.N. 1976. Dispersal ecology of *Carex pedunculata* (Cyperaceae), a few North American myrnecochore. <u>American Journal of Botany</u>. 63: 1071-1079.
- Handel, S.N. 1978a. The competitive relationship of three woodland sedges and its bearing on the evolution of ant-dispersal of *Carex pedunculata*. Evolution. 32: 151-163.
- Handel, S.N. 1978b. New ant-dispersed species in the genera Carex. Luzula, and Claytonia. Canadian Journal of Botany, 56: 2925-2927.
- Handel, S.N., S.B. Fisch, and G.E. Schatz. 1981. Ants disperse a majority of herbs in a mesic forest community in New York State. <u>Bulletin of the Torrey Botanical Club</u>. 108: 430-437.
- Hanzawa, F.M., A.J. Beattie, and D.C. Culver. 1988. Directed dispersal: demographic analysis of an ant-seed mutualism. <u>American Naturalist</u>. 131: 1-13.
- Hardin, G. 1960. The competitive exclusion principle. Science, 131: 1292-1297.
- Harmon, M.E., J.F. Franklin, F.J. Swanson, P. Sollins, S.V. Gregory, J.D. Lattin, N.H. Anderson, S.P. Cline, N.G. Aumen, J.R. Sedell, G.W. Lienkaemper, K. Cromack Jr., and K.W. Cummins, 1986. Ecology of coarse woody debris in temperate ecosystems. <u>Advances in Ecological Research</u>, 15: 133-302.
- Harner, R.F. and K.T. Harper. 1976. The role of area, heterogeneity, and favorability in plant species diversity of pinyon-juniper ecosystems. <u>Ecology</u>. 57: 1254-1263.
- Harper, J.L. 1965. Establishment, aggression, and cohabitation in weedy species. In: H.G. Baker and G.L. Stebbins (eds.). <u>The Genetics of Colonizing Species</u>. Academic Press. New York. pp. 243-265.
- Harper, J.L. 1977. Population Biology of Plants. Academic Press. New York.
- Harper, J.L. P.H. Lovell and K.G. Moore. 1970. The shapes and sizes of seeds. <u>Annual Review of Ecology and Systematics</u>. 1: 327-356.

- Harper, J.L., J.T. Williams and G.R. Sagar. 1965. I. The heterogeneity of soil surfaces and its role in determining the establishment of plants from seed. <u>Journal of Ecology</u>, 53: 273-286.
- Hart, D.D. and R.J. Horwitz. 1991. Habitat diversity and the species-area relationship: alternative models and tests. In: McCoy. E. and G. Bell (eds.). 1991. <u>Habitat Structure</u>, London, Chapman and Hall, pp.47-68.
- Harvey, P.H. and M.D. Pagel. 1991. <u>The Comparative Method in Evolutionary Ecology</u>. Oxford University Press. Oxford.
- Harvey, P.H., AF, Read, and S. Nee. 1995. Why ecologists need to be phylogenetically challenged. Journal of Ecology. 83: 535-536.
- Hedman, C.W. and D. Binkley. 1988. Canopy profiles of some Piedmont hardwood forests. Canadian Journal of Forest Research. 18: 1090-1093.
- Heil, G.W. and M. Bruggink. 1987. Competition for nutrients between Calluna vulgaris (L.) Hull and Molinia caerulea (L.) Moench. <u>Oecologia</u>. 73: 105-107.
- Heithaus, E.R. 1981, Seed predation by rodents on three ant-dispersed plants. <u>Ecology</u>, 1981, 62: 136-145.
- Henry, J.D. and J.M.A. Swan. 1974. Reconstructing forest history from live and dead plant material - an approach to the study of forest succession in southwest New Hampshire. <u>Ecology</u>. 55: 772-783.
- Herben, T. 2000. Correlation between richness per unit area and the species pool cannot be used to demonstrate the species pool effect. *Journal of Vegetation Science*, 11: 123-126.
- Heywood, V.H.(ed.).1995. <u>Global Biodiversity Assessment</u>. Cambridge University Press. Cambridge.
- Hicks, D.J. and B.F. Chabot, 1985. Deciduous Forest. In: Chabot, B.F. and H.A. Mooney, 1985. <u>Physiological Ecology of North American Plant Communities</u>. Chapman Hall. New York. pp. 257-277.
- Hills, G.A. 1952. <u>The Classification and Evaluation of Site for Forestry</u>. Research Report No. 24. Ontario Department of Lands and Forests.
- Hinds, T.E. and F.G. Hawksworth, 1965. Seed dispersal velocity in four dwarfmistletoes. <u>Science</u>, 148: 517-519.

Hitchcock, A.S. 1971 (1950). Manual of the Grasses of the United States. Second Edition, 2

Volumes, Republication by Dover Publications, Inc. New York,

- Hoehne, L.M. 1981. The groundlayer vegetation of forest islands in an urban-suburban matrix. In: R.L. Burgess and D.M. Sharpe (eds.) <u>Forest Island Dynamics in Man-Dominated</u> <u>Landscapes</u>. Springer-Verlag. New York. pp. 41-54.
- Hoffman, D.W. and C.J. Acton. 1974. <u>The Soils of Northumberland County</u>. Report No. 42 of the Ontario Soil Survey, Research Branch, Agriculture Canada and the Ontario Agriculture College.
- Holmes, E.E. and H.B. Wilson, 1998. Running from trouble: long-distance dispersal and the competitive co-existence of inferior species. <u>American Naturalist</u>, 151: 578-586.
- Holsinger, K.E. and S.W. Holsinger, 1990. Multiple-niche polymorphisms in plant populations. <u>American Naturalist</u>, 135: 301-309.
- Holt, R.D. 1993. Ecology at the mesoscale: the influence of regional processes on local communities. In Ricklefs, R.E. and D. Schluter. 1993. <u>Species Diversity in Ecological</u> <u>Communities, Historical and Geographical Perspectives</u>. Unversity of Chicago Press. Chicago, pp. 77-88.
- Holt, R.D. and M.A. McPeek. 1996. Chaotic population dynamics favors the evolution of dispersal. <u>American Naturalist</u>, 148: 709-718.
- Hooper, D.U. and P.M. Vitousek. 1997. The effects of plant composition and diversity on ecosystem processes. <u>Science</u>. 277: 1302-1305.
- Hoppes, W.G. 1988. Seedfall pattern of several species of bird-dispersed plants in an Illinois woodlad. <u>Ecology</u>. 69: 320-329.
- Horn, H.S. 1971. <u>The Adaptive Geometry of Trees</u>. Princeton University Press. Princeton, New Jersey.
- Horn, H.S. 1975. Forest succession. Scientific American, 232: 90-98.
- Hosie, R.C. 1973. <u>Native Trees of Canada</u>. Seventh Edition, Canadian Forestry Service. Department of the Environment. Information Canada. Ottawa.
- Host, G.E. and K.S. Pregitzer. 1992. Geomorphic influences on ground-flora and overstory composition in upland forests of northwestern lower Michigan. <u>Canadian Journal of Forest</u> <u>Research</u>. 22: 1547-1555.
- Host, G.E., K.S. Pregitzer, C.W. Ramm, J.B. Hart, and D.T. Cleland. 1987. Landform-mediated

differences in successional pathways among upland forest ecosystems in northwestern lower Michigan. <u>Forest Science</u>. 33: 445-457.

- Host, G.E., K.S. Pregitzer, C.W. Ramm, D.P. Lusch, and D.T. Cleland. 1988. Variation in overstory biomass among glacial landforms and ecological land units in northwestern lower Michigan. <u>Canadian Journal of Forest Research</u>. 18: 659-668.
- Howard, D.F., M.S. Blum, T.H. Jones, and D.W. Phillips. 1982. Defensive adaptations of eggs and adults of *Gastrophysa cyanea* (Coleoptera: Chrysomelidae). <u>Journal of Chemical Ecology</u>. 8: 453-462.
- Howard, E.W. 1970. Dissemination and viability of seed from upland black spruce in central Newfoundland. <u>Bi-Monthly Research Notes</u>, 26: 29. Department of Fisheries and Forestry. Ottawa.
- Howard, W.E. 1961. Seeds stored by prairie deer mice. Journal of Mammalogy, 42: 260-263.
- Howe, H.F. and J. Smallwood, 1982. Ecology of Seed Dispersal. <u>Annual Review of Ecology and</u> <u>Systematics</u>, 13: 201-28.
- Howell, J.A. and J.L. Vankat, 1981. An ordination of the forest herb stratum of Abner's Hollow, south-central Ohio. <u>Ohio Journal of Science</u>, 81: 98-104.
- Hubbeil, S.P. and R.B. Foster. 1986. Biology, chance, and history and the structure of tropical rain forest communities. In Diamond, J.E. and T.J. Case (eds.). <u>Community Ecology</u>. Harper and Row. New York, pp. 314–329.
- Hughes, L. and M. Westoby. 1992a. Fate of seeds adapted for dispersal by ants in Australian sclerophyli vegetation. <u>Ecology</u>. 73:1285-1299.
- Hughes, L. and M. Westoby. 1992b. Effect of diaspore characteristics on removal of seeds adapted for dispersal by ants. <u>Ecology</u>. 1300-1312.
- Hughes, L., M. Dunlop, K. French, M.R. Leishman, B. Rice, L. Rodgerson, and M. Westoby. 1994. Predicting dispersal spectra: a minimal set of hypotheses based on plant attributes. <u>Journal</u> of Ecology. 82: 933-950.
- Hurtt, G.C. and S.W. Pacala. 1995. The consequences of recruitment limitation: reconciling chance, history and competitive differences between plants. <u>Journal of Theoretical Biology</u>. 176: 1– 12.
- Husband, B.C. and S.C.H. Barrett. 1996. A metapopulation perspective in plant population biology. Journal of Ecology. 84: 461-469.

Huston, M. 1979. A general hypothesis of species diversity. American Naturalist, 113: 81-101.

- Huston, M. 1994. <u>Biological Diversity The Coexistence of Species on Changing Landscapes</u>. Cambridge University Press. Cambridge.
- Hutchinson, G.E. 1941. Ecological aspects of succession in natural populations. <u>American</u> <u>Naturalist</u>. 75: 406-418.
- Hutchinson, G.E. 1957. Concluding remarks. <u>Cold Spring Harbor Symposium in Quantitative</u> <u>Biology</u>. 22: 415-427.
- Hutchinson, G.E. 1959. Homage to Santa Rosalia; or, why are there so many kinds of animals?. <u>American Naturalist</u>, 93: 145-159.
- Hutchinson, G.E. 1961. The paradox of the plankton. American Naturalist, 95: 137-145.
- Hutchinson, T.F. R.E.J. Boerner, L.R. Iverson, S. Sutherland and E.K. Sutherland, 1999. Landscape patterns of understory composition and richness across a moisture and nitrogen mineralization gradient in Ohio (U.S.A.) *Quercus* forests. <u>Plant Ecology</u>, 144: 177-189.
- Ives. A.R. and R.M. May. 1985. Competition within and between species in a patchy environment: relations between microscopic and macroscopic models. <u>Journal of Theoretical Biology</u>, 115: 65-92.
- Johnson, R.A., M.F. Willson, J.N. Thomson, and R.I. Bertin. 1985. Nutritional values of wild fruits and consumption by migrant frugivorous birds. <u>Ecology</u>. 66: 819-827.
- Johnson, W.C. and C.S. Adkisson, 1985, Dispersal of beech nuts by blue jays in fragmented landscapes, <u>American Midland Naturalist</u>, 113: 319-324.
- Johnson, W.C. and R.L. Patterson, unpublished data. Cited in Darley-Hill and Johnson (1981), op.cil.
- Jongman, R.H.G., C.J.F. ter Braak, and O.F.R. van Tongeren (eds.). 1987. <u>Data Analysis in</u> <u>Community and Landscape Ecology</u>. Pudoc Wageningen, the Netherlands.
- Jules, E.S. 1996. Yellow jackets (*Vespula vulgaris*) as a second seed disperser for the myrmecochorous plant, *Trillium ovatum*, <u>American Midland Naturalist</u>, 135: 367-369.
- Jurado, E., M. Westoby, and D. Nelson. 1991. Diaspore weight, dispersal, growth form, and perenniality of central Australian plants. <u>Journal of Ecology</u>, 79: 811-830.
- Kalisz, S.L., F.M. Hanzawa, S.J. Tonsor, D.A. Thiede, S. Voight, n.d. Ant-mediated seed dispersal

affects distance, density and spatial pattern of seed relatedness in *Trillium grandiflorum*. Unpublished manuscript.

- Kalisz, S., L. Horth, M.A. McPeek, 1997. Fragmentation and the role of seed banks in promoting persistence in isolated populations of *Collinsia verna*. In: M.W. Schwartz (ed.), <u>Conservation in Highly Fragmented Landscapes</u>. Chapman and Hall. pp. 286-312.
- Karam, A. 1993, Chemical properties of organic soils. In: M.R. Carter (ed.), <u>Soil Sampling and</u> <u>Methods of Analvsis</u>, Canadian Society of Soil Science, Lewis Publishers.
- Karlson, R.H and H.V. Cornell. 1998. Scale-dependent variation in local vs. regional effects on coral species richness. <u>Ecological Monographs</u>, 68: 259-274.
- Keddy, P.A. 1981. Experimental demography of the sand-dune annual, Cakile edentula, growing along an environmental gradient in Nova Scotia. <u>Ecology</u>, 69: 615-630.
- Keddy, P.A. and A.A. Reznicek. 1982. The role of seed banks in the persistence of Ontario's coastal plain flora. <u>American Journal of Botany</u>, 69: 13-22.
- Keddy, P.A. and A.A. Reznicek. 1986. Great Lakes vegetation dynamics: the role of fluctuating water levels and buried seeds. Journal of Great Lakes Research. 12: 25-36.
- Keily, B.J., J.B. Wilson and A.F. Mark. 1989. Causes of the species-area relation: a study of islands in Lake Manapouri, New Zealand. <u>Journal of Ecology</u>. 77: 021-1028.
- Kerner, A. 1895. <u>The Natural History of Plants</u>. Half-Volume IV. Henry Holt and Company. New York.
- Khanna, P.K. and B. Ulrich. 1991. Ecochemistry of temperate deciduous forests. In: Rohrig. E. and B. Ulrich (eds). 1991. <u>Ecosystems of the World 7: Temperate Deciduous Forests</u>. Elsevier. Amsterdam. pp 121-163.
- Kiviniemi, K. 1996. A study of adhesive seed dispersal of three species under natural conditions. <u>Acta Botanica Neerlandica</u>, 45: 73-83.
- Kjellsson, G. 1985. Seed fates in a population of Curex pilulifera L. Oecologia. 67: 416-423.
- Klinkhamer, P.G.L., T.J. de Jong, and E. van der Meijden. 1988. Production, dispersal and predation of seeds of the biennial *Cirsium vulgare*. Journal of Ecology, 76: 403-414.
- Knopps, J.M.H., W.D. Koenig, and T.H. Nash III. 1997. On the relationship between nutrient use efficiency and fertility in forest ecosystems. <u>Oecologia</u>. 110: 550-556.

- Kobe, R.S. 1996. Intraspecific variation in sapling mortality and growth predicts geographic variation in forest composition. <u>Ecological Monographs</u>, 66: 181-201.
- Kobe, R.S., S.W. Pacala, and J.S. Silander. 1995. Juvenile tree survivorship as a component of shade tolerance. <u>Ecological Applications</u>, 5: 517-532.
- Kohlermann, L. 1950. Cited in Geiger, R. (1961), op.cit.
- Kollman, J. 1995. Regeneration window for fleshy-fruited plants during scrub development on abandoned grassland. <u>Ecoscience</u>, 2: 213-222.
- Kosola, K.R. and K.L.Gross. 1999. Resource competition and suppression of plants colonizing early successional old fields. <u>Oecologia</u>, 118: 69-75.
- Kotanen, P.M. 1997, Effects of gap area and shape on recolonization by grassland plants with differing reproductive strategies. <u>Canadian Journal of Botany</u>, 75: 352-361.
- Krefting, L.W. and E.I. Roe. 1949. The role of some birds and mammals in seed germination. <u>Ecological Monographs</u>, 19: 270-286.
- Kruger, E.L. and P.B. Reich. 1997. Responses of hardwood regeneration to fire in mesic forest openings. I. Post-fire community dynamics. <u>Canadian Journal of Forest Research</u>, 27: 1822– 1831.
- Lambers, H. R. K. Szaniawski and R. deVisser. 1983. Respiration for growth. maintenance and ion uptake. An evaluation of concepts. methods. values and their significance. <u>Physiologia</u> <u>Plantarum</u>, 58: 556-563.
- Lavorel, S. and P. Chesson. 1995. How species with different regeneration niches coexist in patchy habitats with local disturbance. <u>Oikos</u>. 74: 103-114.
- Lavorel, S., R.V. O'Neill and R.H. Gardner. 1994. Spatio-temporal dispersal strategies and annual plant species coexistence in a structured landscape. <u>Oikos</u>. 71: 75-88.
- Leak, W.B. 1978. Relationship of species and site index to habitat in the White Mountains of New Hampshire. Forest Service Research Paper NE-397, U.S. Department of Agriculture.
- Leak, W.B. 1982. Habitat mapping and interpretation in New England. Forest Research Paper NE-496. U.S. Department of Agriculture.
- Lechowicz, M.J. and G. Bell. 1991. The ecology and genetics of fitness of forest plants. II. Microspatial heterogeneity in the edaphic environment. Journal of Ecology, 79: 687-696.

- Lee, T.D. 1984. Effects of seed number per fruit on seed dispersal in *Cassia fasciculata* (Caesalpiniaceae). <u>Botanical Gazette</u>, 145: 136-139.
- Leininger, T.D. and W.E. Winner. 1988. Throughfall chemistry beneath *Quercus rubra*: atmospheric. foliar, and soil chemistry considerations. <u>Canadian Journal of Forest Research</u>. 18: 478-482.
- Leishman, M.R., M. Westoby, and E. Jurado. 1995. Correlates of seed size variation: a comparison among five temperate floras. <u>Journal of Ecology</u>, 83: 517-530.
- Letendre, M., A. Francoeur, R. Beique, and J.-G. Pilon. 1971. Inventaire des fourmis de la station biologie de l'universite de Montreal. St. Hippolyte. Quebec (Hymenoptera: Formicidae). <u>Le</u> <u>Naturaliste Canadien</u>. 98: 591-606.
- Levey, D.J. 1986. Methods of seed processing by birds and seed deposition patterns. In: A. Estrada and T.H. Fleming (eds.). <u>Frugivores and Seed Dispersal</u>. Dr. W Junk Publishers, Dordrecht. pp. 147-158.
- Levin, D.A. 1981. Dispersal versus gene flow in plants. <u>Annals of the Missouri Botanical Garden</u>. 68: 233-253.
- Levin, D.A. and H. Kerster. 1969. Density-dependent gene dispersal in *Liatris*. <u>American Naturalist</u>, 103: 61-74.
- Levin, D.A. and H.W. Kerster. 1974. Gene flow in seed plants. In: T. Dobzhansky, M.K. Hecht and W.C. Steere (eds.). <u>Evolutionary Biology</u>. Volume 7. Plenum Press, New York, pp. 139-220.
- Levin, S.A. 1974. Dispersion and population interactions. American Naturalist. 108: 207-228.
- Levin, S.A. 1976. Population dynamic models in heterogeneous environments. <u>Annual Review of</u> <u>Ecology and Systematics</u>, 7: 287-310.
- Levin, S.A. and R.T. Paine. 1974. Disturbance. patch formation, and community structure. <u>Proceedings of the National Academy of Sciences, USA</u>. 71: 2744-2747.
- Levin, S.A., D. Cohen, and A. Hastings. 1984. Dispersal strategies in patchy environments. <u>Theoretical Population Biology</u>, 26: 165-191.
- Levins, R. 1968. <u>Evolution in Changing Environments Some Theoretical Explorations</u>. Princeton University Press. Princeton.
- Linn, D.M. and J.W. Doran. 1984. Effect of water-filled pore space on carbon dioxide and nitrous oxide production in tilled and nontilled soils. <u>Soil Science Society of America Journal</u>, 48:

1267-1272.

- Little, R.J. and C.E. Jones. 1980. <u>A Dictionary of Botany</u>. Van Nostrand Reinhold Company. New York.
- Livingston, R.B. and M.L. Allessio. 1968. Buried viable seed in successional field and forest stands. Harvard forest. Massachusetts. <u>Bulletin of the Torrey Botanical Club</u>, 95: 58-69.
- Loach, K. 1967. Shade tolerance in tree seedlings. New Phytologist. 66: 607-621.
- Loreau, M. and N. Mouquet. 1999. Immigration and the maintenance of local species diversity. <u>American Naturalist</u>, 154: 427-440.
- Lorimer, C.G. 1989. Relative effects of small and large disturbances on temperate hardwood forest structure. <u>Ecology</u>. 70: 565-567.
- Lorimer, C.G., L.E. Frelich, and E.V. Nordheim. 1988. Estimating gap origin probabilities for canopy trees. <u>Ecology</u>, 69: 778-785.
- Luftensteiner, H.W. 1979. The eco-sociological value of dispersal spectra of two plant communities. <u>Vegetatio</u>, 41: 61-67.
- Lusk, C.H. and P.B. Reich. 2000. Relationships of leaf dark respiration with light environment and tissue nitrogen content in juveniles of 11 cold-temperate tree species. <u>Oecologia</u>. 123: 318-329.
- Lynch, J.F. 1981. Seasonal, successional, and vertical segregation in a Maryland ant community. Oikos, 37: 183-198.
- Lynch, J.F., E.C. Balinsky, and S.G. Vail. 1980. Foraging patterns in three sympatric forest ant species. *Prenolepis imparis*, *Paratrechina melanderi*, and *Aphgenogaster rudis* (Hymenoptera: Formicidae). <u>Ecological Entomology</u>, 5: 353-371.
- Lynch, J.F., A.K. Johnson, and E.C. Balinsky, 1988. Spatial and temporal variation in the abundance and diversity of ants (Hymenoptera: Formicidae) in the soil and litter layers of a Maryland forest. <u>American Midland Naturalist</u>, 119: 31-44.
- Mabry, C., D. Ackerly and F. Gerhardt. 2000. Landscape and species-level distribution of morphological and life history traits in a temperate woodland flora. <u>Journal of Vegetation</u> <u>Science</u>, 11: 213-224.
- MacArthur, R.H. and R. Levins. 1967. The limiting similarity, convergence, and divergence of coexisting species. <u>American Naturalist</u>, 101: 377-385.

- MacArthur, R.H. and E.O. Wilson, 1963. An equilibrium theory of insular zoogeography. <u>Evolution</u>, 17: 373-387.
- Malmborg, P.K. and M.F. Willson. 1988. Foraging ecology of avian frugivores and some consequences for seed dispersal in an Illinois woodlot. <u>Condor</u>. 90: 173-186.
- Marshail, D.L., A.J. Beattie, and W.E. Bollenbacher, 1979, Evidence for diglycerides as attractants in an ant-seed interaction, <u>Journal of Chemical Ecology</u>, 5: 335-344.
- Martin, A.C., H.S. Zim, and A.L. Nelson, 1961, <u>American Wildlife and Plants, A Guide to Wildlife</u> <u>Food Habits</u>. Dover Publications, Inc. New York.
- Mathsoft, Inc. 1998. S-Plus 4.5 Professional Release 2, Copyright (c) 1988-1998, Mathsoft Inc.
- Matlack, G.R. 1987. Diaspore size, shape, and fall behavior in wind-dispersed plant species, <u>American Journal of Botany</u>, 74: 1150-160.
- Matlack, G.R. 1989. Secondary dispersal of seed across snow in *Betula lenta*, a gap-colonizing tree species. <u>Journal of Ecology</u>, 77: 853-869.
- Matlack, G.R. 1994. Plant species migration in a mixed-history forest landscape in eastern North America. <u>Ecology</u>, 75: 1491-1502.
- May, R.M. 1973. Stability in randomly fluctuating versus deterministic environments. <u>American</u> <u>Naturalist</u>, 107: 621-650.
- Maycock, P.F. 1962. The phytosociology of the deciduous forests of extreme southern Ontario. <u>Canadian Journal of Botany</u>, 41: 379-438.
- Mazer, S.J. 1989. Ecological. taxonomic, and life history correlates of seed mass among Indiana dune angiosperms. <u>Ecological Monographs</u>, 59: 153-175.
- Mazer, S.J. 1990. Seed mass of Indiana dune genera and families: taxonomic and ecological correlates. <u>Evolutionary Ecology</u>, 4: 325-357.
- McAtee, W.L. 1947. Distribution of seeds by birds. American Midland Naturalist, 38: 214223.
- McCaughey, W.C. Schmidt, and R.C. Shearer. 1986. Seed- dispersal characteristics of conifers in the inland mountain west. In: <u>Proceedings - Conifer Tree Seed in the Inland Mountain West</u> <u>Symposium</u>, United States Department of Agriculture, Forest Service, General Technical Report INT-203, pp. 50-62.

McClaugherty, C.A., J. Pastor, and J.D. Aber, 1985. Forest litter decomposition in relation to soil

nitrogen dynamics and litter quality. Ecology. 66: 266-275.

- McClintock, D. The transport of propagules by mammals. <u>Bulletin Mammal Society British Isles</u>, 24: 12-13.
- McDonald, A.W., J.P. Bakker, and K. Vegelin. 1996. Seed bank classification and its importance for the restoration of species-rich flood-meadows. *Journal of Vegetation Science*. 7: 157-164.
- McDonnell, M.J. and E.W. Stiles. 1983. The structural complexity of old field vegetation and the recruitment of bird-dispersed plant species. <u>Oecologia</u>. 56: 109-116.
- McEvoy, P.B. and C.S. Cox. 1987. Wind dispersal distances in dimorphic achenes of ragwort. *Senecio jacobea*. <u>Ecology</u>, 68: 2006-2015.
- McFadden, J.P., N.W. MacDonald, J.A. Witter and D.R. Zak. 1994. Fine-textured soil bands and oak forest productivity in northwestern lower Michigan, U.S.A. <u>Canadian Journal of Forest</u> <u>Research</u>. 24: 928-933.
- McKay, S.M. and P.M. Catling, 1979. <u>Trees, Shrubs and Flowers to Know in Ontario</u>, J.M. Dent & Sons (Canada) Ltd.
- McLaughlin, J.F. and J. Roughgarden. 1993. Species interactions in space. In Ricklefs. R.E. and D. Schluter. 1993. <u>Species Diversity in Ecological Communities</u>, <u>Historical and Geographical Perspectives</u>. Unversity of Chicago Press. Chicago, pp. 89-98.
- McLaughlin, S.B. and R.J. Kohut. 1992. The effects of atmospheric deposition and ozone on carbon allocation and associated physiological processes in red spruce. In: Eager. C. and M.B. Adams (eds.). 1990. Ecology and Decline of Red Spruce in the Eastern United States. Springer-Verlag. New York. pp. 338-382.
- McLaughlin, S.B., C.P. Andersen, P.J. Hanson, M.G. Tjoelker and W.K. Roy. 1991. Increased dark respiration and calcium deficiency of red spruce in relation to acidic deposition at highelevation southern Appalachian Mountain sites. <u>Canadian Journal of Forest Research</u>, 21: 1234-1244.
- McRill, M. and G.R. Sagar, 1973. Earthworms and seeds. Nature, 243: 482.
- Meagher, T.R. and E. Thompson. 1987. Analysis of parentage for naturally established seedlings of Chamaelirium luteum (Liliaceae). <u>Ecology</u>, 68: 803-812.
- Melillo, J.M., J.D. Aber, and J.F. Muratore. 1982. Nitrogen and lignin control of hardwood leaf litter decomposition dynamics. <u>Ecology</u>. 63: 621-626.
- Mergen, F.1954. Mechanical aspects of wind-breakage and wind-firmness. <u>Journal of Forestry</u>, 52: 119-125.
- Mesler, M.R. and K.L. Lu. 1983. Seed dispersal of *Trillium ovatum* (Liliaceae) in second-growth redwood forests. <u>American Journal of Botany</u>, 70: 1460-1467.
- Messier, C. and P. Bellefleur. 1988. Light quantity and quality on the forest floor of pioneer and climax stages in birch-beech-sugar maple stand. <u>Canadian Journal of Forest Research</u>. 18: 615-622.
- Metz, J.A.J., T.J. de Jong, and P.G.L. Klinkhamer. 1983. What are the advantages of dispersing: a paper by Kuno explained and extended. <u>Occologia</u>. 57: 166-169.
- Metzger, F. and J. Schultz. 1984. Understory response to 50 years of management of a northern hardwood forest in upper Michigan. <u>American Midland Naturalist</u>. 112: 209-223).
- Meyer, A.H. and B. Schmid, 1999. Seed dynamics and seedling establishment in the invading perennial Solidago altissima under different experimental treatments. <u>Journal of Ecology</u>, 87: 28-41.
- Michaels, H.J., B. Benner, A.P. Hartgerink, T.D. Lee, S. Rice, M.F. Willson, and R.I. Bertin. 1988. Seed size variation: magnitude, distribution, and ecological correlates. <u>Evolutionary Ecology</u>, 2: 157-166.
- Microsoft Corporation. 1998. Publisher 98. Copyright (c) 1991-1998. Microsoft Corporation.
- Minkler, L.S. and J.D. Woerheide. 1965. Reproduction of hardwoods 10 years after cutting as affected by site and opening size. *Journal of Forestry*. 63: 103-107.
- Mladenoff, D.J. 1990. The relationship of the soil seed bank and understory vegetation in old-growth northern hardwood-hemlock treefall gaps. <u>Canadian Journal of Botany</u>. 68: 2714-2721.
- Molinier, R. and P. Muller, 1938. Cited in Dansereau and Lems (1958), op.cit.,
- Montgomery, F.H. 1977. <u>Seeds and Fruits of Plants of Eastern Canada and Northeastern United</u> <u>States</u>. University of Toronto Press. Toronto.
- Moore, M.R. and J.L. Vankat. 1986. Response of the herb layer to the gap dynamics of a mature beech-maple forest. <u>American Midland Naturalist</u>. 115: 336-347.
- Morse, D.H. and J. Schmitt. 1985. Propagule size. dispersal ability. and seedling performance in *Asclepias syriaca*. <u>Oecologia</u>, 67: 372-379.

- Morton, J.K. and F.H. Hogg. 1989. Biogeography of island floras in the Great Lakes. II. Plant dispersal. <u>Canadian Journal of Botany</u>. 67: 1803-1820.
- Morton, J.K. and J.M. Venn. 1990. <u>A Checklist of the Flora of Ontario Vascular Plants</u>. University of Biology Series Number Thirty-Four. University of Waterloo. Waterloo.
- Motzin, G., D. Foster, A. Allen, J. Harrod, and R. Boone. 1996. Controlling site to evaluate history: vegetation patterns of a New England sand plain. <u>Ecological Monographs</u>. 66: 345-365.
- Mueller-Dombois, D. and H. Ellenberg, 1974. Aims and Methods of Vegetation Ecology, John Wiley and Sons, New York.
- Muller, R.N. 1982. Vegetation patterns in the mixed mesophytic forest of eastern Kentucky. <u>Ecology</u>. 63: 1901-1917.
- Nadelhoffer, K.J., J.D. Aber and J.M. Melillo. 1983. Leaf-litter production and soil organic matter dynamics along a nitrogen-availability gradient in Southern Wisconsin. <u>Canadian Journal of</u> <u>Forest Research</u>. 13: 12-21.
- Nelson, D.W. and L.E. Sommers. 1982. Total carbon, organic carbon, and organic matter. In: Page, A.L (ed.). <u>Methods of Soil Analysis, Part 2. Chemical and Microbiological Properties</u>. Second Edition. American Society of Agronomy and Soil Science of America. Madison, Wisconsin, p. 576.
- Neter, J., M.H. Kutner, C.J. Nachtsheim, and W. Wasserman. 1996. <u>Applied Linear Statistical</u> <u>Models</u>. Fourth Edition. WCB/McGraw-Hill. Boston. MA.
- Newman, E.I. 1973. Competition and diversity in herbaceous vegetation. Nature, 244: 310.
- Nicolai, V. 1988. Phenolic and mineral content of leaves influences the decomposition in European forest ecosystems. <u>Oecologia</u>, 75: 575-579,
- Niinements, U. 1997. Role of foliar nitrogen in light harvesting and shade tolerance of four temperate deciduous woody species. <u>Functional Ecology</u>, 11: 518-531.
- Nimerfro, K. and G. Brand. 1993. <u>General Technical Report NC-157</u>. The Microcomputer Scientific Software Series7: Data Recorder Program for Storing Plant Lists and Calculating Synecological Coordinates. North Central Forest Experiment Station. Forest Service. United States Department of Agriculture.
- Noble, I.R. and R.O. Slayter, 1980. The use of vital attributes to predict successional changes in plant communities subject to recurrent disturbance. <u>Vegetatio</u>, 43: 5-21.

- Nowacki, G.J., M.D. Abrams, and C.G. Lorimer. 1990. Composition. structure, and historical development of northern red oak stands along an edaphic gradient in north-central Wisconsin. Forest Science. 36: 276-292.
- O'Dowd, D.J. and M.E. Hay. 1980. Mutualism between harvester ants and a desert ephemeral: seed escape from rodents. <u>Ecology</u>, 61: 531-540.
- Ohara. M. and S. Higashi. 1987. Interference by ground beetles with the dispersal by ants of seeds of *Trillium* species (Liliaceae). <u>Journal of Ecology</u>. 75: 1091-1098.
- Ohmann, J.L. and T.A. Spies. 1998. Regional gradient analysis and spatial pattern of woody plant communities of Oregon forests. <u>Ecological Monographs</u>, 68: 151-182.
- Okansen, J. 1996. Is the humped relationship between species richness and biomass an artefact due to plot size. Journal of Ecology. 84: 293-295.
- Okland, R.H. 1996. Are ordination and constrained ordination alternatives or complementary strategies in general ecological studies. Journal of Vegetation Science, 7: 289-292.
- Okland, R.H. and O. Eilertsen, 1994. Canonical correspondence analysis with variance partitioning: some comments and an application. Journal of Vegetation Science, 5: 117-126.
- Okubo, A. and S.A. Levin. 1989. A theoretical framework for data analysis of wind dispersal of seeds and pollen. <u>Ecology</u>. 70: 329-338.
- Oldham, M.J., W.D. Bakowsky, and D.A. Sutherland. 1995. <u>Floristic Quality Assessment System</u> <u>for Southern Ontario</u>. Natural Heritage Information Centre, Ontario Ministry of Natural Resources. Peterborough.
- Oliver, C.D. 1981, Forest development in North America following major disturbances. <u>Forest</u> <u>Ecology and Management</u>, 3: 153-168.
- Oliver C.D. and E.P. Stephens. 1977. Reconstruction of a mixed-species forest in central New England. <u>Ecology</u>. 58: 562-572.
- Olivieri, I. and A. Berger, 1985. Seed dimorphism for dispersal: physiological, genetic and demographic aspects. In: P. Jacquard et al. (eds.). <u>Genetic Differentiation and Dispersal in</u> <u>Plants</u>. Springer-Verlag. Berlin. pp. 413-429.
- Olivieri, I and P. Gouyon, 1985. Seed dimorphism for dispersal: theory and implications. In: J. Haeck and J.W. Woldendorp (eds.). <u>Structure and Functioning of Plant Populations 2</u>, <u>Phenotypic and Genotypic Variation in Plant Populations</u>. North-Holland Publishing Company. Amsterdam. pp. 77-90.

- Ontario Institute of Pedology, 1985, <u>Field manual for describing soils</u>, 3rd edition, O.I.P. Publication No. 85-3. University of Guelph, Guelph, Ontario.
- Ontario Ministry of Natural Resources. 1978. Forest Stand Map 442784 (SE) (1:10.000 scale). Forest Resources Inventory.
- Ontario Ministry of Natural Resources. 1978. Forest Stand Map 442783 (NE, NW, SW) (1:10,000 scale). Forest Resources Inventory.
- Ontario Ministry of Natural Resources. 1978. Forest Stand Map 442782 (NE. SE) (1:10.000 scale). Forest Resources Inventory.
- Ontario Ministry of Natural Resources. 1978. Forest Stand Map 441784 (NE, SE) (1:10,000 scale). Forest Resources Inventory.
- Ontario Ministry of Natural Resources, 1978. Forest Stand Map 441783 (NW, SW) (1:10,000 scale). Forest Resources Inventory.
- Ontario Ministry of Natural Resources. 1979. Forest Stand Map 443781 (NE, SW, SE) (1:10.000 scale). Forest Resources Inventory.
- Ontario Ministry of Natural Resources, 1980. Forest Stand Map 442781 (NW, SW) (1:10.000 scale). Forest Resources Inventory.
- Ouimet, R. and C. Camire, 1995. Foliar deficiencies of sugar maple stands associated with soil cation imbalances in the Quebec Appalachians. <u>Canadian Journal of Soil Science</u>, 75: 169-175.
- Pacala, S.W. 1986. Neighborhood models of plant population dynamics. 2. Multi-species models of annuals. <u>Theoretical Population Biology</u>, 29: 262-292.
- Pacala, S.W. 1987. Neighborhood models of plant population dynamics. 3. Models with spatial heterogeneity in the physical environment. <u>Theoretical Population Biology</u>, 31: 359-392.
- Pacala, S.W. and J.A. Silander, Jr. 1985. Neighborhood models of plant population dynamic. I. Single-species models of annuals. <u>American Naturalist</u>, 125: 385-411.
- Pacala, S.W. and J.A. Silander, Jr. 1987. Neighborhood interference among velvet leaf, Abutilon theophrasti, and pigweed, Amaranthus retroflexus, Oikos, 48: 217-224.
- Pacala, S.W. and J.A. Silander, Jr. 1990. Field tests of neighborhood models of two annual weed species. <u>Ecological Monographs</u>. 60: 113-134.

- Pacala, S.W. and D. Tilman. 1994. Limiting similarity in mechanistic and spatial models of plant competition in heterogeneous environments. <u>American Naturalist</u>. 143: 22-257.
- Pacala, S.W., C.D. Canham, J. Saponara, J.A. Silander Jr., R.K. Kobe and E. Ribbens. 1996. Forest models defined by field measurements: estimation, error, analysis and dynamics. <u>Ecological</u> <u>Monographs</u>, 66: 1-43.
- Paine, R.T. 1966. Food web complexity and species diversity. American Naturalist, 100: 65-75.
- Paine, R.T. 1974. Intertidal community structure: experimental studies on the relationship between a dominant competitor and its principal predator. <u>Oecologia</u>, 15: 93-120.
- Palmer, M.W. 1990. Spatial scale and patterns of species-environment relationships in hardwood forest of the North Carolina piedmont. <u>Coenoses</u>, 5: 79-87.
- Palmer, M.W. 1994. Variation in species richness: towards a unification of hypotheses. Folia Geobotanica Phytotaxonomica, 29: 511-530.
- Pannell, J.R. and S.C.H. Barrett. 1998. Baker's law revisited: reproductive assurance in a metapopulation. Evolution. 52: 657-668.
- Parra, V., C.F. Vargas and L.E. Eguirarte. 1993. Reproduction biology. pollen and seed dispersal. and neighborhood size in the hummingbird-pollinated *Echeveria giffiflora* (Crassulaceae). <u>American Journal of Botany</u>. 80: 153-159.
- Partel, M., M. Zobel, K. Zobel and E. van der Maarel. 1996. The species pool and its relation to species richness: evidence from Estonian plant communities. <u>Oikos</u>. 75: 111-117.
- Pastor, J. 1986. Reciprocally linked carbon-nitrogen cycles in forests: biological feedbacks within geological constraints. In: Agren, G.I. (ed.). <u>Predicting consequences of intensive forest</u> <u>harvesting on long-term productivity</u>. Swedish University of Agricultural Science. Department of Ecology and Environmental Research. Report Nr. 26, pp. 131-140.
- Pastor, J. and W.M. Post. 1986. Influence of climate. soil moisture, and succession on forest carbon and nitrogen cycles. <u>Biogeochemistry</u>. 2: 3-27.
- Pastor, J., J.D. Aber and C. McClaugherty and J.M. Mclillo. 1984. Aboveground production and N and P cycling along a nitrogen mineralization gradient on Blackhawk Island. Wisconsin. <u>Ecology</u>. 65: 256-268.
- Patrick, W.H. Jr. and I.C. Mahapatra. 1968. Transformation and availability to rice of nitrogen and phosphorus in waterlogged soils. <u>Advances in Agronomy</u>, 20: 323-359.

- Patterson, G.T. 1978. Some criteria used in the identification of soil moisture and drainage in the Canadian system of soil classification. In: R.K. Jones (ed.). <u>Proceedings: Soil Moisture</u> <u>Regime Workshop, December 15, 1978, University of Guelph</u>. Ontario Institute of Pedology, Guelph, Ontario, pp.1-16.
- Paul, E.A. and F.E. Clark. 1996. <u>Soil Microbiology and Biochemistry</u>. Second Edition. Academic Press. San Diego.
- Perry, I. and P.D. Moore. 1987. Dutch elm disease as an analogue of Neolithic elm decline. <u>Nature</u>. 326: 72-73.
- Peet, R.K. and N.L. Christensen, 1980. Hardwood forest vegetation of the North Carolina piedmont. <u>Verroff, Geobot, Institute</u>, 69: 14-39.
- Pellmyr, O. 1985. Yellow jackets disperse Vancouveria seeds (Berberidaceae). Madrono, 32: 56.
- Pemberton, R. W. 1988. Myrmecochory in the introduced range weed, leafy spurge (*Euphorbia esula* L.). <u>American Midland Naturalist</u>, 119: 431-435.
- Peterson, C.J. and J.E. Campbell. 1993. Microsite differences and temporal change in plant communities of treefall pits and mounds in an old-growth forest. <u>Bulletin of the Torrev</u> <u>Botanical Club</u>, 120: 451-460.
- Peterson, C.J. and S.T.A. Pickett. 1990. Microsite and elevational influences on early forest regeneration after catastrophic windthrow. Journal of Vegetation Science, 1: 657-662.
- Peterson, C.J., W.P. Carson, B.C. McCarthy, and S.T.A. Pickett. 1990. Microsite variation and soil dynamics within newly created treefall pits and mounds. <u>Oikos</u>, 58: 39-46.
- Petraitis, P.S., R.E. Latham, and R.A. Niesenbaum. 1989. The maintenance of species diversity by disturbance. <u>The Ouarterly Review of Biology</u>, 64: 393-418.
- Pettit, N.E., R.H. Froend and P.G. Ladd. 1995. Grazing in remnant woodland vegetation: changes in species composition and life form groups. Journal of Vegetation Science, 6: 121-130.
- Pickett, S.T.A. 1980. Non-equilibrium coexistence of plants. <u>Bulletin of the Torrey Botanical Club</u>. 107: 238-248.
- Pickett, S.T.A. and P.S. White (eds.). 1985. <u>The Ecology of Natural Disturbance and Patch</u> <u>Dynamics</u>. Academic Press Inc. San Diego.
- Pierpoint, G. 1978. The recognition and classification of soil moisture regime in the Ontario physiographic site classification system. In: R.K. Jones (ed.). <u>Proceedings: Soil Moisture</u>

<u>Regime Workshop, December 15, 1978, University of Guelph</u>. Ontario Institute of Pedology, Guelph, Ontario, pp.39-59.

- Piper, J.K. 1986. Germination and growth of bird-dispersed plants: effects of seed size and light on seedling vigor and biomass allocation. <u>American Journal of Botany</u>, 73: 959-965.
- Platt, W.J. 1976. The natural history of a fugitive prairie plant (*Mirabilis hirsuta* (Pursh) MacM.). <u>Oecologia</u>. 22: 399-409.
- Platt, W.J. and I.M. Weis. 1977. Resource partitioning and competition within a guild of fugitive prairie plants. <u>American Naturalist</u>, 111: 479-513.
- Pojar, J. 1974. Reproductive dynamics of four plant communities of southwestern British Columbia. <u>Canadian Journal of Botany</u>. 52: 1819-1834.
- Ponnamperuma, F.N. 1972. The chemistry of submerged soils. Advances in Agronomy, 24: 29-96.
- Portnoy, S. and M.F. Willson, 1993. Seed dispersal curves: behavior of the tail of the distribution. Evolutionary Ecology, 7: 25-44.
- Pregitzer, K.S. and B.V. Barnes. 1982. The use of ground flora to indicate edaphic factors in upland ecosystems of the McCormick Experimental Forest. Upper Michigan. <u>Canadian Journal of</u> <u>Forest Research</u>. 12: 661-672.
- Pregitzer, K.S. and B.V. Barnes. 1984. Classification and comparison of upland hardwood and conifer ecosystems of the Cyrus H. McCormick Experimental Forest, upper Michigan, <u>Canadian Journal of Forest Research</u>. 14: 362-375.
- Preston, F.W. 1960. Time and space and the variation of species. Ecology, 41: 611-627.
- Preston, F.W. 1962. The canonical distribution of commonness and rarity. Part I. Part II. <u>Ecology</u>, 43: 185-215: 410-432.
- Primack, R.B. 1996. Lessons from ecological theory: dispersal, establishment, and population structure. In: D.A. Falk, C.I. Millar and M. Olwell (eds.). <u>Restoring Diversity: Strategies for</u> <u>the Reintroduction of Endangered Plants</u>. Island Press. Washington, D.C. pp. 209-233.
- Proctor, V.W. 1968. Long-distance dispersal of seeds by retention in digestive tract of birds. Science, 160, 321-322.
- Pudlo, R.J., A.J. Beattie, and D.C. Culver. 1989. Population consequences of changes in an ant-seed mutualism in *Sanquinaria canadensis*. <u>Oecologia</u>, 46: 32-37.

- Putz, F.E., P.D. Coley, K. Lu, A. Montalvo and A. Aiello. 1983. Uprooting and snapping of trees: structural determinants and ecological consequences. <u>Canadian Journal of Forest Research</u>, 13: 1011-1020.
- Quinghong, L. and S. Brakenhielm. 1995. A statistical approach to decompose ecological variation. <u>Water. Air and Soil Pollution</u>. 85: 1587-1592.
- Quinn, S.L., J.B. Wilson, and A.F. Mark. 1987. The island biogeography of Lake Manapouri. New Zealand. Journal of Biogeography. 14: 569-581.
- Rabinowitz, D. and J.K. Rapp. 1979. Dispersal abilities of seven sparse and common grasses from a Missouri prairie. <u>American Journal of Botany</u>. 68: 616-624.
- Rasmussen, H.N. 1995, <u>Terrestrial Orchids from Seed to Mycotrophic Plant</u>, Cambridge University Press.
- Raunkiaer, C. 1934. <u>The Life Forms of Plants and Statistical Plant Geography</u>. Reprint edition 1977. Arno Press. New York.
- Reed, R.A., R.K. Peet, M.W. Palmer and P.S. White, 1993. Scale dependence of vegetationenvironment correlations: a case study of a North Carolina piedmont woodland. <u>Journal of</u> <u>Vegetation Science</u>, 4: 329-340.
- Rees. M. 1993. Trade-offs among dispersal strategies in British plants. <u>Nature</u>, 366: 150-152.
- Rees. M. 1994. Delayed germination of seeds: a look at the effects of adult longevity, the timing of reproduction, and population age/stage structure. <u>American Naturalist</u>, 144: 43-54.
- Rees. M. 1996. Evolutionary ecology of seed dormancy and seed size. <u>Philosophical Transactions</u> <u>Royal Society London B</u>, 351: 1299-1308.
- Rees, M., P.J. Grubb, and D. Kelly. 1996. Quantifying the impact of competition and spatial heterogeneity on the structure and dynamics of a four-species guild of winter annuals. <u>American Naturalist</u>. 147: 1-32.
- Reeve, H.K. and P.W. Sherman. 1993. Adaptations and the goal of evolutionary research. <u>Quarterly</u> <u>Review of Biology</u>. 68: 1-32.
- Reich, P.B., M.D. Abrams, D.S. Ellsworth, E.L. Kruger and T.J. Tabone. 1990. Fire affects ecophysiology and community dynamics of central Wisconsin oak forest regeneration. <u>Ecology</u>, 71: 2179-2190.
- Reich, P.B., D.F. Grigal, J.D. Aber and S.T. Gower, 1997. Nitrogen mineralization and productivity

in 50 hardwood and conifer stands on diverse soils. Ecology, 78: 335-347.

- Reid, N. 1989. Dispersal of mistletoes by honeycaters and flowerpeckers: components of seed dispersal quality. <u>Ecology</u>, 70, 137-145.
- Rice, B. and M. Westoby, 1981. Myrmecochory in sclerophyll vegetation of the West Head, New South Wales. <u>Australian Journal of Ecology</u>, 6: 291-298.
- Rice, B. and M. Westoby, 1986. Evidence against the hypothesis that ant-dispersed seeds reach nutrient-enriched microsites. <u>Ecology</u>, 67: 1270-1274.
- Ricklefs, R.E. 1987. Community diversity: relative roles of local and regional processes. <u>Science</u>, 235: 167-171.
- Ricklefs, R.E. and D. Schluter. 1993. <u>Species Diversity in Ecological Communities, Historical and</u> <u>Geographical Perspectives</u>. Unversity of Chicago Press. Chicago.

Ridley, H.N. 1930. The Dispersal of Plants Throughout the World, L.Reeve and Co. Ashford.

- Roberts, T.L. and J.L. Vankat. 1991. Floristics of a chronosequence corresponding to old fielddeciduous forest succession in southwestern Ohio. II. Seed banks. <u>Bulletin of the Torrey</u> <u>Botanical Club</u>, 118: 377-384.
- Robertson, G.P., M.A. Huston, F.C. Evans and J.M. Tiedje, 1988. Spatial variability in a successional plant community: patterns of nitrogen availability. <u>Ecology</u>, 69: 1517-1524.
- Rogers, R.S. 1982. Early spring herb communities in mesophytic forests of the Great Lakes region. <u>Ecology</u>. 63: 1050-1063.
- Root, R.B. 1967. The niche exploration pattern of a blue grey gnateatcher. <u>Ecological Monographs</u>. 37: 317-350.
- Rosenzweig, M.L. 1971. Paradox of enrichment: destabilization and exploitation ecosystems in ecological time. <u>Science</u>, 171: 385-387.
- Roth, I. 1977. Fruits of Angiosperms. Gebruder Borntraeger, Berlin,
- Roughgarden, J.D. 1977. Patchiness in the spatial distribution of a population caused by stochastic fluctuations in resources. <u>Oikos</u>. 29: 52-59.
- Rowe, J.S.1972. <u>Forest Regions of Canada</u>. Department of Fisheries and the Environment. Canadian Forestry Service Publication No. 1300. Supply and Services Canada. Ottawa. Ontario.

- Ruhren, S. and M.R. Dudash. 1996. Consequences and timing of seed release of *Erythronium americanum* (Liliaceae), a deciduous forest myrmecochore. <u>American Journal of Botany</u>. 83: 633-640.
- Rust, R.W. and R.R. Roth. 1981. Seed production and seedling establishment in the mayapple *Podophyllum peltatum*. <u>American Midland Naturalist</u>. 105: 51-60.
- Ryan, T.P. 1997. Modern Regression Methods. John Wiley and Sons, Inc. New York.
- Sacchi, C.F. 1987. Variability in dispersal ability of common milkweed. Asclepius syriaca, seeds. Oikos, 49: 191-198.
- Sale, P.F. 1977. Maintenance of high diversity in coral reef fish communities. <u>American Naturalist</u>. 111: 337-359.
- Salisbury, E.J. 1942. The Reproductive Capacity of Plants. G. Bell and Sons. London.
- Salisbury, E.J. 1974. Seed size and mass in relation to environment. <u>Proceedings Royal Society</u> London B, 186: 83-88.
- SAS Institute Inc. 1997. JMP. Version 3.2.2. Copyright (c) 1989-1997. SAS Institute Inc.
- Saverimuttu, T. and M. Westoby, 1996. Seedling longevity under deep shade in relation to seed size. Journal of Ecology, 64: 681-689.
- Savile, D.B.O. 1953. Splash-cup dispersal mechanism in *Chrysosplenium* and *Mitella*. <u>Science</u>, 117: 250-251.
- Savile, D.B.O. 1979. Dispersal by falling water in Saxifragaceae. Davidsonia. 10: 65-69.
- Scariot, A. 1999. Forest fragmentation effects on palm diversity in central Amazonia. <u>Journal of</u> <u>Ecology</u>. 87: 66-76.
- Schall, B.A. 1980. Measurement of gene flow in Lupinus texensis. Nature, 284: 450-451.
- Schemske, D.W. 1978. Evolution of reproductive characteristics in *Impatiens* (Balsaminaceae): the significance of cleistogamy and chasmogamy. <u>Ecology</u>, 59: 596-613.
- Schmitt, J., D. Erhardt, and D. Swartz. 1985. Differential dispersal of self-fertilized and out-crossed progeny in jewelweed (*Impatiens capensis*). <u>American Naturalist</u>. 126: 570-575.
- Schulze, E.D., F.A. Bazzaz, K.J. Naddelhoffer, T. Koike and S. Takatsuki. 1996. Biodiversity and ecosystem function of temperate deciduous broad-leaved forests. In: Mooney, H.A., J.H.

Cushman, E. Medina, O.E. Sala, and E.D. Schulze (eds.), Functional Roles of Biodiversity: <u>A Global Perspective</u>, John Wiley and Sons, Ltd. pp. 71-98,

- Schupp, E.W. 1993. Quantity, quality and effectiveness of seed dispersal by animals. <u>Vegetatio</u>, 107/108: 15-29.
- Schupp, E.W. and M. Fuentes. 1995. Spatial patterns of seed dispersal and unification of plant population ecology. <u>Ecoscience</u>. 2: 267-275.
- Seeman, J.R., T.D. Sharkey, J.L. Wang and C. B. Osmond. 1987. Environmental effects on photosynthesis. nitrogen-use efficiency. and metabolite pools in leaves of sun and shade plants. <u>Plant Physiology</u>, 84: 796-802.
- Seger, J. and H.J. Brockmann. 1987. What is bet-hedging? In: P.H. Harvey and L. Partridge (eds). <u>Oxford Surveys in Evolutionary Biology</u>. Vol. 4. Oxford University Press. Oxford. pp. 182-211.
- Semple, J.C. and S.B. Heard. 1987. <u>The Asters of Ontario: Aster L. and Virgulus Raf. (Compositae:</u> <u>Asteraceae</u>). University of Waterloo Biology Series No. 30. Department of Biology, University of Waterloo. Waterloo.
- Semple, J.C. and G.S. Ringius. 1983. <u>The Goldenrods of Ontario: Solidago L. and Euthamia Nutt.</u> University of Waterloo Biology Series No.26. Department of Biology. University of Waterloo. Waterloo.
- Sernander, R. 1906. <u>Entwurf einer Monographie der Eurpaischen Myrmekochoren</u>, Kungl. Svenska Vetenskapsakademiens Handlingar, Band 41, No. 7. Almquist and Wiksells Boktryckeri-A-B. Uppsula.
- Sheldon, J.C. and F.M. Burrows. 1973. The dispersal effectiveness of the achene-pappus units of selected compositae in steady winds with convection. <u>New Phytologist</u>, 72: 665-675.
- Shmida, A. and S. Ellner, 1984. Coexistence of plant species with similar niches. <u>Vegetatio</u>, 58: 29-55.
- Siccama, T.G., F.H. Bormann, and G.E. Likens. 1970. The Hubbard Brook study: productivity, nutrients, and phytosociology of the herbaceous layer. <u>Ecological Monographs</u>, 40: 389-402.
- Silander, J.A. and S.W. Pacala. 1990. The application of plant population dynamic models to understanding plant competition. In. J.B. Grace and D. Tilman (eds.). <u>Perspectives on Plant</u> <u>Competition</u>. Academic Press. San Diego. pp. 67-91.
- Sims, D.A. and R.W. Pearcy, 1991. Photosynthesis and respiration in Alocasia macrorrhiza

following transfers to high and low light. Oecologia. 86: 447-453.

- Silvertown, J.W. 1984. Phenotypic variety in seed germination behavior: the ontogeny and evolution of somatic polymorphism in seeds. <u>American Naturalist</u>, 124: 1-16.
- Silvertown, J. and R. Law. 1987. Do plants need niches? Some recent developments in plant community ecology. <u>Trends in Ecology and Evolution</u>. 2: 24-26.
- Skidmore, B.A. and E.R. Heithaus, 1988. Lipid cues for seed-carrying by ants in *Hepatica* americana. Journal of Chemical Ecology, 14: 2185-2196.
- Smith, A.J. 1975. Invasion and ecesis of bird-disseminated woody plants in a temperate forest sere. Ecology. 56: 19-34.
- Smith. B.H., P.D. Forman and A.E. Boyd. 1989. Spatial patterns of seed dispersal and predation of two myrmecochorous forest herbs. <u>Ecology</u>, 70: 1649-1656.
- Smith. B.H., C.E. de Rivera, C.L. Bridgman, and J.J. Woida. 1989. Frequency-dependent seed dispersal by ants of two deciduous forest herbs. <u>Ecology</u>, 70: 1645-1648.
- Smith, J.M.B. 1989. An example of ant-assisted plant invasion. <u>Australian Journal of Ecolgy</u>, 14: 247-250.
- Smith, L.M. and L.T. Kok. 1984. Dispersal of musk thistle (*Carduus nutans*) seeds. <u>Weed Science</u>. 32: 120-125.
- Smith. T.M. H.H. Shugart and F.I. Woodward (eds.). 1997. <u>Plant Functional Types</u>. Cambridge University Press. New York.
- Snow, B. and D. Snow. 1988. <u>Birds and Berries: A Study of an Ecological Interaction</u>. T&AD Poyser Ltd. Calton. England.
- Sokal, R.R. and F.J. Rohlf. 1995. <u>Biometry. The Principles and Practice of Statistics in Biological</u> <u>Research</u>. Third Edition. W.H. Freeman and Company. New York.
- Soper, J.H. and M.L. Heimburger. 1982. Shrubs of Ontario. Royal Ontario Museum. Toronto.
- Sorensen, A.E. 1978. Somatic polymorphism and seed dispersal. Nature, 276: 174-176,
- Sorensen, A.E. 1986. Seed dispersal by adhesion. <u>Annual Review of Ecology and Systematics</u>, 17: 443-463.
- Sousa, W.P. 1984. The role of disturbance in natural communities. Annual Review of Ecology and

Systematics, 15: 353-91.

- Sparling, J.H. 1967. Assimilation rates of some woodland herbs in Ontario. <u>Botanical Gazette</u>. 128: 160-168.
- Spies, T.A. and B. Barnes. 1985a. A multifactor classification of the northern hardwood and conifer ecosystems of Sylvania Recreation Area. upper peninsula, Michigan. <u>Canadian Journal of Forest Research</u>. 15: 949-960.
- Spies, T.A. and B. Barnes. 1985b. Ecological species groups of upland northern hardwood bemlock ecosystems of the Sylvania Recreation Area, upper peninsula, Michigan. <u>Canadian</u> <u>Journal of Forest Research</u>. 15: 961-972.
- Stamp, N.E. 1989. Efficacy of explosive vs. hygroscopic seed dispersal by an annual grassland species. <u>American Journal of Botany</u>. 76: 555-561.
- Stamp, N.E. and J.R. Lucas. 1983. Ecological correlates of explosive seed dispersal. <u>Oecologia</u>, 59: 272-278.
- Stamp, N.E. and J.R. Lucas. 1990. Spatial patterns and dispersal distances of explosively dispersing plants in Florida sandhill vegetation. <u>Journal of Ecology</u>. 78: 589-600.
- Stapanian, M.A. and C.C. Smith. 1978. A model for seed scatterhoarding: coevolution of fox squirrels and black walnuts. <u>Ecology</u>, 59: 884-896.
- Starkman, J.M. 1994. Causes of soil nutrient heterogeneity at different scales. In: Caldwell, M.M. and R.W. Pearcy (eds.). <u>Exploitation of Environmental Heterogeneity by Plants.</u> <u>Ecophysiological Processes Above- and Belowground</u>. Academic Press. San Diego. pp. 255-284.
- Stebbins, G.L. 1971. Adaptive radiation of reproductive characteristics in angiosperms. II: seeds and seedlings. <u>Annual Review of Ecology and Systematics</u>, 2: 237- 260.
- Stebbins, G.L. 1974. Adaptations for seed development and dispersal and for seedling establishment. In: Stebbins, G.L. 1974. <u>Flowering Plants Evolution above the Species Level</u>. Belknap Press of Harvard University Press. Cambridge.
- Stergios, B.G. 1976. Achene production, dispersal, seed germination, and seedling establishment of *Hieracium aurantiacum* in an abandoned field community. <u>Canadian Journal of Botany</u>, 54: 1189-1197.
- Stiles, E.W. 1980. Patterns of fruit presentation and seed dispersal in bird-disseminated woody plants in the eastern deciduous forest. <u>American Naturalist</u>, 116: 670-688.

- Stiles, E.W. 1989. Fruits, seeds, and dispersal agents. In: W.G. Abrahamson (ed.). <u>Plant-Animal</u> <u>Interactions</u>. McGraw-Hill, New York.
- Stiles, E.W. 1992. Animals as seed dispersers. In: M. Fenner (ed.). <u>Seeds The Ecology of</u> <u>Regeneration in Plant Communities</u>. CAB International. Oxford. pp. 87-104.
- Stiles, E.W. and D.W. White. 1986. Seed deposition patterns: influence of season. nutrients. and vegetation structure. In: A. Estrada and T.H. Fleming (eds.). <u>Frugivores and Seed Dispersal</u>. Dr W. Junk Publishers. Dordrecht. pp. 45-54.
- Stratton, D.A. 1994. Genotype-by-environment interactions for fitness of *Erigeron annuus* show fine scale selective heterogeneity. <u>Evolution</u>. 48: 1607-1618.
- Summerhayes, V.S. 1951. Wild Orchids of Britain with a key to the species. Collins. London.
- Sydes, C. and J.P. Grime. 1981a. Effects of tree leaf litter on herbaceous vegetation in deciduous woodland. I. Field investigations. <u>Journal of Ecology</u>. 69: 237-248.
- Sydes. C. and J.P. Grime. 1981b. Effects of tree leaf litter on herbaceous vegetation in deciduous woodland. II. An experimental investigation. Journal of Ecology, 69: 249-262.
- Taylor, D.R., L.W. Aarssen and C. Loehle. 1990. On the relationship between r/K selection and environmental carrying capacity: a new templet for plant life history strategies. <u>Oikos</u>. 58: 239-250.
- Taylor, R.J. and R.W. Pearcy, 1976. Seasonal patterns of the CO₂ exchange characteristics of understory plants from a deciduous forest. <u>Canadian Journal of Botany</u>, 54: 1094-1103.
- ter Braak, C.J.F. 1987. Ordination. In: Jongman, R.H.G., C.F.J. ter Braak, and O.F.R. van Tongeren (eds). <u>Data Analysis in Community and Landscape Ecology</u>. Pudoc Wageningen, pp. 91-173.
- ter Braak, C.J.F. 1991, <u>CANOCO Version 3.12 April 1991</u>, Copyright (c) 1988-1991 Agricultural Mathematics Group DLO, Box 100, 6700 AC Wageningen, the Netherlands.
- ter Braak, C.J.F. 1994. Canonical community ordination. Part I: Basic theory and linear methods. <u>Ecoscience</u> 1: 127-140.
- ter Braak, C.J.F. 1999. Variance explained and variance partitioning. www.okstate.edu/artseibotany/ordinate/varpar.html.
- ter Steege, H. 1994. <u>HEMIPHOT. a Programme to Analyze Vegetation Indices. Light and Light</u> <u>Ouality from Hemispherical Photographs</u>. The Tropenbos Foundation, Wageningen, The Netherlands.

- Thiede, D.A. and C.K. Augspurger. 1996. Intraspecific variation in seed dispersion of *Lepedium* campestre (Brassicaceae). <u>American Journal of Botany</u>. 83: 856-866.
- Thomas, S.C., C.B. Halpern, D.A. Falk, D.A. Liguori and K.A. Austin. 1999. Plant diversity in managed forests: understory responses to thinning and fertilization. <u>Ecological Applications</u>. 9: 864-879.
- Thompson, J.N. 1980. Treefalls and colonization patterns of temperate forest herbs. <u>American</u> <u>Midland Naturalist</u>, 104: 176-164.
- Thompson, J.N. 1981. Elaiosomes and fleshy fruits: phenology and selection pressures for antdispersed seeds. <u>American Naturalisi</u>. 117: 104-108.
- Thompson, J.N. and M.F. Willson. 1978. Disturbance and the dispersal of fleshy fruits. <u>Science</u>. 200: 1161-1163.
- Thompson, J.N. and M.F. Willson. 1979. Evolution of temperate fruit/bird interactions: phenological strategies. <u>Evolution</u>. 33: 973-982.
- Thompson, K., J.P. Bakker, and R.M. Bekker. 1997. <u>The Soil Seed Banks of North West Europe:</u> <u>methodology, density and longevity</u>. Cambridge University Press. Cambridge.
- Thompson, K., J.P. Bakker, R.M. Bekker and J.G. Hodgson. 1998. Ecological correlates of seed persistence in soil in the north-west European flora. Journal of Ecology. 86: 163-169.
- Tilman, D. 1982. <u>Resource Competition and Community Structure</u>. Princeton University Press. Princeton.
- Tilman, D. 1987. Secondary succession and the pattern of plant dominance along experimental nitrogen gradients. <u>Ecological Monographs</u>, 57: 189-214.
- Tilman, D. 1988. <u>Plant Strategies and the Dynamics and Structure of Plant Communities</u>. Princeton University Press. Princeton.
- Tilman, D. 1993. Species richness of experimental productivity gradients: how important is colonization limitation. <u>Ecology</u>, 74: 2179-2191.
- Tilman, D. 1994. Competition and biodiversity in spatially structured habitats. Ecology, 75: 2-16.
- Tilman, D. 1997, Community invasibility, recruitment limitation, and grassland biodiversity, <u>Ecology</u>, 78: 81-92.
- Tilman, D. and J.A. Downing, 1994. Biodiversity and stability in grasslands. Nature, 367: 363-365.

- Tilman, D. and S. Pacala. 1993. The maintenance of species richness in plant communities. In: Ricklefs, R.E. and D. Schluter. 1993. <u>Species Diversity in Ecological Communities</u>. <u>Historical and Geographical Perspectives</u>. Unversity of Chicago Press, Chicago, pp. 13-25.
- Tilman, D., R.M. May, C.L. Lehman and M.A. Nowak. 1994. Habitat destruction and the extinction debt. <u>Nature</u>, 371: 65-66.
- Tomback, D.F. and Y.B. Linhart. The evolution of bird-dispersed pines. <u>Evolutionary Ecology</u>. 4: 185-219.
- Trapp, E.J. 1988. Dispersal of heteromorphic seeds in *Amphicarpu bracteata* (Fabaceae). <u>American</u> <u>Journal of Botany</u>, 75: 1535-1539.
- Valverde, T. and J. Silvertown. 1997. A metapopulation model for *Primula vulgaris*, a temperate forest understorey herb. <u>Journal of Ecology</u>. 85: 193-210.
- van Breeman, N., A.C. Finzi, and C.D. Canham, 1997. Canopy tree soil interactions within temperate forests: effects of soil elemental composition and texture on species distributions. <u>Canadian Journal of Forest Research</u>, 27: 1110-1116.
- van der Pijl, L. 1982. <u>Principles of Dispersal in Higher Plants</u>. Third Edition. Springer-Verlag, New York.
- van der Wall, S.B. and R.P. Balda. 1977. Coadaptations of the Clarke's nuteracker and the pinon pine for efficient seed harvest and dispersal. <u>Ecological Monographs</u>, 47: 89-111.
- Varga, S. 1989. <u>A Botanical Inventory and Evaluation of the High Park Oak Woodlands Area of Natural and Scientific Interest</u>. Ontario Ministry of Natural Resources. Parks and Recreational Areas Section, Central Region, Richmond Hill, ON.
- Venable, D.L. 1985. The evolutionary ecology of seed heteromorphism. <u>American Naturalist</u>. 126: 577-595.
- Venable, D.L. and J.S. Brown. 1988. The selective interactions of dispersal, dormancy, and seed size as adaptations for reducing risk in variable environments. <u>American Naturalist</u>. 131: 360-384.
- Venable, D.L. and J.S. Brown, 1993. The population dynamic functions of seed dispersal. <u>Vegetatio</u>. 107/108: 31-55.
- Venable, D.L. and L. Lawior. 1980. Delayed germination and dispersal in desert annuals: escape in space and time. <u>Oecologia</u>. 46: 272-282.

- Venable, D.L. and D.A. Levin. 1983. Morphological dispersal structures in relation to growth habit in the Compositae. <u>Plant Systematics and Evolution</u>. 143: 1-16.
- Venable, D.L. and D.A. Levin. 1985. Ecology of achene dimorphism in *Heterotheca latifolia*. 1. Achene structure, germination and dispersal. Journal of Ecology. 73: 133-145.
- Venable, D.L., E. Dyreson, and E. Morales. 1995. Population dynamic consequences and evolution of seed traits of *Heterosperma pinnatum* (Asteraceae). <u>American Journal of Botany</u>, 82: 410-420.
- Vetaas, O.R. 1992. Micro-site effects of trees and shrubs in dry savannas. Journal of Vegetation Science, 3: 337-344.
- Vickery, R.K., D.R. Phillips, and P.R. Wonsavage. 1986. Seed dispersal in *Mimulus* guttatus by wind and deer. <u>American Midland Naturalist</u>, 1986.
- Vitousek, P. 1982. Nutrient cycling and nutrient use efficiency. American Naturalist, 119: 553-572.
- Volterra. V. 1926. Fluctuations in the abundance of a species considered mathematically. <u>Nature</u>: 118: 558-560.
- Voss, E.G. 1972. <u>Michigan Flora. Part I. Gymnosperms and Monocots</u>. Cranbrooke Institute of Science and University of Michigan Herbarium. Ann Arbor.
- Voss, E.G. 1985. <u>Michigan Flora. Part II. Dicots (Saururaceae-Cornaceae)</u>. Cranbrooke Institute of Science and University of Michigan Herbarium. Ann Arbor.
- Voss, E.G. 1996. <u>Michigan Flora, Part III. Dicots (Pvrolaceae-Compositae)</u>. Cranbrooke Institute of Science and University of Michigan Herbarium. Ann Arbor.
- Waller, D.M. 1980. Environmental determinants of outcrossing in *Impatiens capensis* (Balsaminaceae). <u>Evolution</u>. 34: 747-761.
- Walters, M.B. and P.B. Reich. 1996. Are shade tolerance. survival and growth linked? Low light and nitrogen effects on hardwood seedlings. <u>Ecology</u>, 77: 841-853.
- Walters, M.B. and P.B. Reich. 1997. Growth of Acer saccharum seedlings in deeply shaded understories of northern Wisconsin: effects of nitrogen and water availability. <u>Canadian</u> <u>Journal of Forest Research</u>. 27: 237-247.
- Warner, R.R. and P.L. Chesson. 1985. Coexistence mediated by recruitment fluctuations: a field guide to the storage effect. <u>American Naturalist</u>. 125: 769-787.

- Warr, S.J., K. Thomson, and M. Kent. 1992. Antifungal activity in seed coat extracts of woodland plants. <u>Oecologia</u>, 92: 296-298.
- Watkinson, A.R. 1978. The demography of a sand dune annual: *Vulpia fasciculata*. III. The dispersal of seeds. Journal of Ecology.66: 483-498.
- Watt, A.S. 1947. Pattern and process in the plant community. Journal of Ecology, 35: 1-22.
- Webber, J.M. and P.W. Ball. 1984. The taxonomy of the *Carex rosea* group (Section Phaestoglochin) in Canada. <u>Canadian Journal of Botany</u>. 62: 2058-2073.
- Webber, L.R. and F.F. Morwick. 1946. Soil Survey of Durham County. Report No. 9 of the Ontario Soil Survey. Experimental Farms Service. Dominion Department of Agriculture and the Ontario Agricultural College.
- Weiblen, G.D. and J.D. Thomson, 1995. Seed dispersal in *Erythronium grandiflorum* (Liliaceae). <u>Oecologia</u>, 102: 211-219.
- Wein, G.R. and S.T.A. Pickett. 1989. Dispersal. establishment. and survivorship of a cohort of Erythronium americanum. <u>Bulletin of the Torrev Botanical Club</u>. 116: 240-246.
- Went, F.W. 1957. The Experimental Control of Plant Growth. Ronald Press Company. New York.
- Werner, P.A. 1975. A seed trap for determining patterns of seed deposition in terrestrial plants. <u>Canadian Journal of Botany</u>. 53: 810-813.
- Werner, P.A. and W.J. Platt. 1976. Ecological relationships of co-occurring goldenrods (Solidago: Asteraceae). <u>American Naturalist</u>. 110: 959-971.
- Westlaken, I.L. and M.A. Maun. 1985. Spatial pattern and seed dispersal of *Lithospermum* caroliniense on Lake Huron sand dunes. <u>Canadian Journal of Botany</u>, 63: 125-132.
- Westman, W.E. 1983. Island biogeography: studies on the exeric shrublands of the inner Channel Islands. California. Journal of Biogeography. 10: 97-118.
- Westoby, M. and B. Rice. 1981. A note on combining two methods of dispersal-for-distance. Australian Journal of Ecology. 6: 189-192.
- Westoby, M., B. Rice and J. Howell. 1990. Seed size and plant form as factors in dispersal spectra. <u>Ecology</u>. 71: 1307-1315.
- Westoby, M., E. Jurado, and M. Leishman. 1992. Comparative evolutionary ecology of seed size. <u>Trends in Ecology and Evolution</u>. 7: 368: 372.

- Westoby, M., M. Leishman, and J. Lord. 1996. Comparative ecology of seed size and dispersal. <u>Philosophical Transactions Royal Society London. B</u>. 351: 1309-1328.
- Wheeler, W.M. 1910. Ants, Their Structure. Development and Behavior. Columbia University Press. New York.
- Whelan, C.J., M.F. Willson, C.A. Tuma, and I. Souza-Pinto. 1991. Spatial and temporal patterns of post-dispersal seed predation. <u>Canadian Journal of Botany</u>, 69: 428-436.
- White, D.W. and E.W. Stiles. 1990. Co-occurrences of foods in stomachs and feces of fruit-eating birds. <u>Condor</u>. 92: 291-303.
- White, D.W. and E.W. Stiles. 1992. Bird dispersal of fruits of species introduced into eastern North America. <u>Canadian Journal of Botany</u>. 70: 1689-1696.
- Whitney, G.G. 1984. Fifty years of change in the arboreal vegetation of Heart's Content, an oldgrowth hemlock-white pine- northern hardwood stand. <u>Ecology</u>, 65: 403-408.
- Whitney, G.G. 1986. A demographic analysis of *Rubus ideaus* and *Rubus pubescens*. <u>Canadian</u> Journal of Botany. 64, 2916-2921.
- Whitney, G.G. 1991. Relation of plant species to substrate, landscape position, and aspect in north central Massachusetts. <u>Canadian Journal of Forest Research</u>, 21: 1245-1252.
- Whittaker, R.H. 1956. Vegetation of the Great Smoky Mountains. Ecological Monographs, 26: 1-80.
- Whittaker, R.H. 1965. Dominance and diversity in land plant communities. Science, 147: 250-260.
- Williams, C.B. 1943. Area and number of species. Nature, 152: 264-267.
- Williams, C.B. 1964, Patterns in the Balance of Nature. Academic, London,
- Williams, C.F. and R.P. Guries. 1994. Genetic consequences of seed dispersal in three sympatric forest herbs. I. Hierarchical population-genetic structure. <u>Evolution</u>. 48: 791-805.
- Willson, M.F. 1986. Avian frugivory and seed dispersal in eastern North America. In: R.F. Johnston (ed.). <u>Current Ornithology</u>. Volume 3. Plenum Press. New York. pp. 223-279.
- Willson, M.F. 1993a. Dispersal mode, seed shadows, and colonization patterns. <u>Vegetatio</u>. 107/108: 261-280.
- Willson, M.F. 1993b. Mammals as seed dispersal mutualists in North America. Oikos. 159-176.

- Willson, M.F. and C.J. Whelan. 1990. The evolution of fruit color in fleshy-fruited plants. <u>American</u> <u>Naturalist</u>. 136: 790-809.
- Willson, M.F., E.A. Porter, and R.S. Conit. 1982. Avian frugivore activity in relation to forest light gaps. <u>Caribbean Journal of Science</u>, 18: 1-6.
- Willson, M.F., B.L. Rice, and M. Westoby, 1990. Seed dispersal spectra: a comparison of temperate plant communities. <u>Journal of Vegetation Science</u>, 1: 547-562.
- Wilmot, T.R., D.S. Ellsworth and M.T. Tyree. 1995. Relationships among crown condition. growth. and stand nutrition in seven northern Vermont sugarbushes. <u>Canadian Journal of Forest</u> <u>Research</u>, 25: 386-397.
- Wisheu, I.C. and P.A. Keddy. 1989. Species richness standing crop relationships along four lakeshore gradients: constraints on the general model. <u>Canadian Journal of Botany</u>, 67: 1609-1617.
- Yodzis, P. 1978. <u>Competition for Space and the Structure of Ecological Communities</u>. Springer-Verlag. Berlin.
- Yodzis, P. 1986. Competition, Mortality and Community Structure. In: Diamond, J. and T.J. Case (eds.). 1986. <u>Community Ecology</u>, Harper and Row, New York, pp. 480-491.
- Zak, D.R., K.S. Pregitzer, and G.E. Host. 1986. Landscape variation in nitrogen mineralization and nitrification. <u>Canadian Journal of Forest Research</u>, 16: 1258-1263.
- Zak, D.R. and K.S. Pregitzer. 1990. Spatial and temporal variability of nitrogen cycling in northern lower Michigan. <u>Forest Science</u>, 36: 367-380.
- Zinke, P.J. 1962. The pattern of influence of individual forest trees on soil properties. <u>Ecology</u>, 43: 130-133.
- Zobel, M. 1997. The relative roles of species pools in determining plant species richness: an alternative explanation for species coexistence?. <u>Trends in Ecology and Evolution (TREE)</u>, 12: 266-269.

APPENDICES

APPENDIX 1

LOCATION OF STUDY SITES

Site	Quadrats	Region/County	Municipality/Township	UTM	NTS Sheet
		LANDSCA	VPE I		
31	1-8	Peterborough	South Monaghan	110E 966N	31D/1
W	9-16	Peterborough	Otonabce	264E 082N	31D/8
8X	17-24	Peterborough	Asphodel	362E 132N	31D/8
8	25-32	Peterborough	Asphodel	309E 141N	8/011
60	33-40	Peterboraugh	Asphodel	329E 118N	31D/8
 -	41-48	Peterboraugh	Otomahee	195E 005N	31D/I
EA EA	49-56	Peterborough	Otonabee	221E 037N	31D/I
PS	57-64	Peterlannagh	()Ionabce	296E 037N	31D/8
pi2	65-72	Peterharough	()tonabee	215E 961N	1/011
۲۱	73-80	Peterboraugh	Asphodel	357E 119N	31D/8
Y2	81-88	Peterbarough	South Monaghan	124E 950N	31D/1
YS	96-96	Peterborough	Asphodel	365E 134N	31D/8
		TANDSCA	DF: II		
BBI	to1-26	Durham Regional Municipality (R.M.)	Newcastle Town Municipality (T.M.)	833E 814N	31D/2
Idd	105-112	Victoria	Manvers	897E 975N	2/CH E

Location of Study Sites.
<u></u>
Appendix

Sheet												
STN 3		31D/2	31D/2	31D/2	3110/2	3112/2	31D/2	3113/2	311)/2	3113/2	3113/2	
UTM		803E 804N	835E 955N	836E 907N	801E 825N	784E 837N	796E 891N	789E 888N	848F; 834N	N1+8 311+8	923E 955N	
Municipality/Township	ll (cont'd)	Newcastle T.M.	Mauvers	Mauvers	Manvers	Scugog	Scugog	Sengog	Manvers	Manvers	Manvers	
Region/County	LANDSCAPE	Durham R.M.	Victoria	Victoria	Victoria	Durham R.M.	Durham R.M.	Durham R.M.	Victoria	Victoria	Victoria	
Quadrats		071-11	121-128	129-136	137-144	145-152	153-160	161-168	921-691	177-184	185-192	
Sile		GGI	602	MM2	MM8	NN2	ENN	NN	0012	Idd	Edd	

(cont'd).
sites (
study
ē
Location
<u>.</u>
Appendix

APPENDIX 2

SUMMARY OF PLANT ATTRIBUTES BY SPECIES

Legend: SPECIES: species code (see Appendix 3): # Q: number of quadrats (N=192): LF: life form: T= tree, S = shrub. V = vine, F= fern, FA =fern ally, G = grass, H = herb; LH: life history. A = annual, B = biennial, annual/biennial. P = perennial, annual/perennial, biennial/perennial; P: provenance. N = native, A = alien, U = unknown; HA: habitat affinity, F = forest, FO = forest + open. OF = open - forest, O = open; FRUIT TYPE: "arillate seed", "winged seed" (gymnosperms), "spore" (homosporous spores of ferns and fern allies) are regarded as fruit types in this classification; DISPERSAL MODES: AI = animal ingestion: FF = fleshy fruit; AA = animal adhesion: AD = adhesive fruit: AC = animal conveyance: EL = elaiosome bearing seed or fruit. NF = nut fruit: PD = prolonged dormancy: SP = long term persistent seed pool: W = wind: ME = mechanical expulsion: EM = explosive mechanism, WM = wind-push mechanism. SM = splash-cup mechanism: U = unassisted; MM = multiple modes; VE = vegetative expansion.

Sources: plant names and provenance: Morton and Venn 1990. Cody and Britton 1989: life form. life history and fruit type: Gleason and Cronquist 1991.

References: (1) Alex 1992, (2) Arditti 1992, (3) Beattie and Culver 1981, (4) Beattie and Lyons 1975, (5) Beattie et al. 1979, (6) Bell 1991, (7) Berg 1966, (8) Berg 1969, (9) Brodie 1955, (10) Bulow-Olsen 1984, (11) Burnside et al. 1996, (12) Cody and Britton 1989, (13) Corner 1976, (14) Crockett 1977. (15) Culver and Beattie 1978. (16) Darley-Hill and Johnson 1981, (17) Dore and McNeill 1980, (18) Elliott 1978, (19) Farrar 1995. (20) Fisher 1988, (21) Gaddy 1986, (22) Gates 1940. (23) Gates 1941, (24) Gleason 1952. (25) Gleason and Cronquist 1991, (26) Graber and Thompson 1978, (27) Gunther and Lanza 1989, (28) Handel 1976, (29) Handel 1978ab, (30) Handel et al. 1981. (31) Heithaus 1981. (32) Howard 1961. (33) Hitchcock 1971. (34) Livingston and Allessio 1968. (35) Martin et al. 1951, (36) McDonald et al. 1996, (37) McKay and Catling 1979. (38) Montgomery 1977, (39) Montgomery Collection Test 1997, (40) Pudlo et al. 1980, (40) Rasmussen 1995. (42) Ridley 1930. (43) Roberts and Vankat 1991, (44) Savile 1953, (45) Savile 1979. (46) Semple and Heard 1987. (47) Semple and Ringius 1983. (48) Semander 1906. (49) Smith and Reichman 1984. (50) Smith et al. 1989. (51) Snow and Snow 1988. (52) Soper and Heimburger 1982, (53) Sorensen 1986, (54) Stamp and Lucas 1983, (55) Stiles 1980, (56) Stiles 1989, (57) Thompson 1980, (58) Thompson 1981, (59) Thompson and Willson 1979, (60) Thompson et al. 1997. (61) Trapp 1988. (62) Webb and Willson 1985. (63) Wein and Pickett 1989. (64) White and Stiles 1992, (65) Whitney 1986, (66) Williams and Garies, (67) Willson 1986, ST - see text Section 3.2.1.

KEPEKENCIE	19	61	19	61	61	10	61	60; -12; 25	67	67	67	21	25, 38, 25	60: 25	60; 25	25	60; IV	24, 25; 60	t9
VE			•					NI:	•			VE	VE	VI:	VE	,			,
MM			-			-		MM		•							MM	MM	-
7	•	•	1	•	•	'	•	•	•	•			,			11	,	•	,
MI		,	-			•			: -	•	-		1				-	٠	
3	w	w	w	w	w	w	w	w				Ŵ	•	•			w		
61	•		-			•	-	SP	-			,	,	sР	ЧN	,	SP	ż	
٨¢	-		-		-		-		-			,	,			,			
٧V		•	•	: '				ı		-			dν		-			UV	
V	,	-	•	-	-			,	:l:f	EF.	i:)		,	,			-	,	al I
FRUIT TYPE	winged seed	schizocurp	schizoturp	schizocarp	schizocarp	schizocarp	schizeeun (the	adadat	herry	herry	berry	spare	achene	caryopsis	euryopsis	capsule	el 1986 els	uchene	atuend
VH	•	•		•				:lO	. ત	<u>د</u>	÷	:-	0;	,í0	0	<u>دد ،</u>	0	05	01:
^	z	z	z	z	z	z	z	۷	z	z	z	z	z	۷	z	z	z	z	z
Ŧ	1	4	а. 	a.,	*	-	-	-	<u> </u>	<u>-</u>	<u>~</u>	а.	<u> </u>	4	2	~	4	<	4
.:. 	L.	ľ	r.	.1.	-	÷	*	=	=	H	=	<u></u>	~	9	0	=	×	Ŧ	x
Ô#	61	t	86	175	22	~	34	Ē	20	77	ſŀt:	30	6	٤ı	3	25	-	Ē	-
FAMILY	Pinaceae	Aceraces	Асстароды	Accritectie	Acetuceue	Acertoche	Averaceae	Asteruccae	Ranuaculaceae	Kanunculaceae	Raturenlareae	Adiastoceae	Rosuceae	Ponceae	Ронселе	Lihacene	Behulaceae	Asternicenc	Rosaccae
		1	¥		z	_			E	ж	<u>.</u>	4	÷	Ņ	Ē		<	Ш	Q

RIFFERENCE	ł9	64	61	37; 25	3, 31; 25	25, 38	25, 38; 25	14, 25; 25	34; SU, 25	55, 64; 25	20, 25, 64	64	3,7,27,31,40, 57, 25	25	11; 25	-16; 25	46. 25	-16.23	-16: 25
MI:				V1:	VIC 1	•	VI):	VI:	NI:	NI:			ž			(i) (i)	ΥE	VE	VE
MM	-		-		-			•	MM		MIM	,	,	ı	MM			•	
1		•		,		·	•	•	-	•	•	'	۰.		•	•	-	•	_ '
ME			EM						WN								•		
*	•	,	-	ž	,	×	Ŵ	N.	,	,				w	W	W	W.	w	w
ીન						,			12		21				i.	L .			
AC	_				EI.			,		•			Ы.						
۲ ۲											(IV	-		.			,		
₹	÷	Ъ			<u> </u>		•			44]-F:	-				·		.
Fruit Type	pone	putite	եցսու	achene	achene	uchene	uchene	folliele	folkele	berry-like- drupe	achene	berry	copsular	fallicle	folliele	achene	achene	achene	achene
=	ō	аĿ	5	÷	E0	:10	a.	90	СĿ.		â	Ŀ,		01	÷	ЧC,	:ICI	0	5
-	z	z	z	z	z	z	z	z	z	z	~	z	z	z	z	z	z	z	z
Ξ	<u> </u>	<u>-</u>	<	<u>l</u> ≞	<u></u>	<u>^</u>	<u> </u>	<u></u>	<u>~</u>	-	=	-	-	-	-	=	<u>-</u>	-	-
11	~	×	=	=	=	=	=	=	=	=	=	=	=	=	=	=	π	=	=
0#	~	47	=	-	-	=	_	2	13	₽	*	16	38	24	2	۲.	13	-	~
FAMILY	Rossecae	Rosucene	Fubaceae	Ratianeulaceae	Ranneuluceae	Rananculaceae	Asterneeue	Аросущееве	Runneukaceue	Amliaceae	Asteracite	Алисан	Aristolochinceae	Aselepiaduceae	Aselepiaduceae	Asteraceate	Asteracene	Asteraceue	Asteraceae
SPECIES	AMELINTE	AMEL SP	AMPERAC	ANEMCANA	ANEMOUN	ANEMVIRG	ANTENEOL.	APOCANDR	AQUICANA	ARALNUTH	ARCTAINU	ARISTRIP	ASARCANA	ASCLINCA	ASCLEYRI	ASTECILI	ASTECORD	ASTERIC	ANTELANC

REFERENCE	46; 25	46; 25	46; 25	46; 25	46; 25	12	12	61	34; 1V	20, 25: 60	42, 25, 38	12	12	12	11, 19, 25	25	641	60: 25	25, 20	25
ΥE	VE	NI:	NI(λI	3iA	NI:	NE.			,			•	•)IN	V1(-	NI:	-	ЭΛ ;
MM							'	,		MM	MM	,	1			•	-	-	MM	1
=	<u> </u>		· ·	'	•	•			•	•	•		,	,	,	n		L		Ξ
ME		-	-					ı				: ,		•		. •				
÷	м	w	M	w	w	w	M	w	M	-	W.		W.	W	•		-		w	
CId		•	-						NI.	SP.	•	•	-	-		,	SP	ЗР		
X		,	·		•	•	-		-	-		•	-	-						
٧V		-	-	•	•	•			-	- OV	AD.		-	-	ЧŅ				AD.	
Ň	-	-	•			•							•	-		,	,		,	,
Froit Type	achene	achene	achene	achene	uchene	share	spare	DUNNING	UTERCIES	achene	achene	spute	spore	choic	sistlo.(un)	cery apsis	folliele	capsular	achene	silique
VII	Å		- HC	210	0],	[0	<u></u>	•		:10		0E	0	:I	<u>يد</u>	ö	30	•	•	-
<u> </u>	z	z	z	z	z	z	z	z	z	z	z	z	z	z	Z	<	z	z	<	z
:	ᆈ		-	-	<u></u>	<u>-</u>	<u>.</u>	<u>٦</u>	±.	<	-	·	-	~	<u> </u>	-	-	-	×	-
=	Ħ	Ξ	=	Ŧ	=	<u></u>		H		Ξ	=	÷-	-	<u>.</u>	σ	<u> </u>	=	Ξ	=	=
	66	ŗ	81	15	-	£4	+	32	29	74	28		-	÷	Ś	÷		_	1	¥
FAMILY	Asteraceae	Asteracone	Asteruceue	Asteriocito	Asternocae	Asplenincene	Aspleniaceae	isctulucene	Вениксене	Asteraceate	Urthenease	Ophioglossuceuc	Ophioglossoceae	Ophioglossucete	Poaceac	Posterate	Rammenlaceate	Convolvulaceae	Asterneene	Brassicarceac
SPECIES	ATTATE	ASTEMACR	ASTENUVE	ASTIPUNI	ASTEUMBE	THATCLY	ATHYTHEE	BETUALLE	BETUPAPY	HIDEFRON	BORICYLE	BOTRMATR	BOTRMULT	BOTRVIRG	BRACISREC	BROMINER	CALTPALU	CALYSEP	CARDACAN	CARDOINE

REFERENCE	25; 48, 60; 20	25	25	52	25	25	25	25	25	25	38. 205	25	25	25	25	25	25	25	25	35
NI:	-	-	-		-				-	-		-			-	•	νE	-	-	-
MM	MM	-	•		-						,			-	-			-		
===		11	n	ī,	1	1	;	n	1	Э		i i	n	1	n	n	L)	n	n	'
ME		•																		
W	W	•	•	,										•			-		,	w
Ê	SP SP						•									,				
W.	EL.	•		-					+	+	EI				,	,		,		
٧V	(IV										-					I				
र			'	,			, I		•		,							,		
Fruit Lype	achene	silique	uchene	tichicite	achedic	achene .		auartau	achene	achene	aehene	מכווינויונ	uchene	attatta	netene	achene	poliene	achene	achene	achene
	c		l:	tΟ	Ŀ	0ŀ;	:lO	FO	0	130	1:01	120	13	ł.	4	10	-10	61	Ŀ	0.1
-	<	z	z	z	z	z	z	z	z	z	z	z	z	z	z	z	z	z	N	z
=	≖	=	=	-	: =,	<u>.</u>	<u>.</u>	<u>a</u> ,	<u>م</u>	.⊶	-	-	<u>~</u>	٦	-	-	<u>~</u>	<u>~</u>	á	-
::: •••	=	=	=	=	=	=	=	=	=	=	Ξ	⊐	Н	н	Ħ	=	₽	Ħ	=	=
¢,	_	•	5	~	35	\$	÷	19	+	~	_	1	91	7 %	×.	63	9	÷	ſ	28
FAMILY	Asteracae	Brassicaceae	Cyperaceae	Cyperuceue	Cyperaceae	Cyperuceae	Cyperaceae	Cyperacene	Cyperaceae	Cyperaceae	Cy peraceae	C) periodic	Cyperaceae	Cyperneeue	Сурегаселе	Cyperaceae	Cyperucene	Суреписане	Сурсиссис	C'yperoceae
SPECIES	CARDNUTA	CARDPENS	CAREALBU	CAREALOP	CAREARCT	CAREBACK	CARENTER	CAREBLAN	CAREBREV	CARECEPH	CARECOMM	CARECREN	CARECRIS	CAREDEWE	CAREDIGI	CARECRAC	CAREORAN	CARIBELICT	CARBHITC	CARGINTU

REFERENCE	25	21	25	28, 204, 30; 25	ST; 25	52	3	25	25	25	25	Ŀ	25	25	25	25	25	25	25	25
3	₹			VE	VE		,		-	VE	· · ·	VĿ			-			-	VE	,
WW		-	-	-				-	-	-	-	-		-	-					
=	=		IJ		•	Ŀ,	•	=	=	11	=	 '	=	-	2	2	=	Ľ	-	LI I
ME	,	-	-		•	-	•		•					-			-			-
3	,		-	•	•						,		•				-	•		
- Cl-									,	•			•							-
ί.Υ		E1.	ŀ	EL.	EI.		ЕI.				•	•						,		
Ą		,					-					•	-							·
Ń	,		•			ı	1	,				•						,	,	
Fruit Type	achene	achene	uchunc	achene	achene	achene	peltere	achene	achene	atchicase	achene	achene	adation	achene	нскете	nchene	ទាលាផ្នែរជ	achene	achene	tichicitie
MI	40E	04	4	0/1	:10	:		ol:	0	с);	<u>ىد:</u>	ЕO	ł:	<u>~</u>	PO:	.+O	đ	Ë		,
<u> </u>	7.	7	z	z	z	z	z	z	z	z	z	z	Ň	z	z	z	Z.	z	z	z
	<u> </u>	<u>-</u>	-	-	-	d.	-	<u> </u>	ન	4	ť	-	<u>~</u>	<u>~</u>	<u>~</u>	<u></u>	<u>.</u>	<u>~</u>	•	4
3.1	=	=	=	=	Iŧ	Ξ	Ξ	=	=	=	=	=	Ħ	≖	=	=	=	=	=	=
0 #	6	≞	53	80	120	61	-	_	-7	-	<u>50</u>	٤t	11	*	16	13	÷	•	_	Ę
Y.IIMAN	Сурсинсам	(,Aperate	Cyperaceae	Cypenseae	Cyperaceae	Cyperneau	Cyperaceae	Cypernoene	Cyperaceae	Cypenicaie	C) peraceae	C) peraceae	Су регасине	Суреписан	Суратиска	Cyperaceae	4. yperneae	C'yperiociae	Cypergeeue	Cyperaeae
SPECIES	CARIE ANU	CARELAXI	CAREPECK	CARFEEDU	CAREPENS	CAREPLAN	CAREPLAT	CARFERAL	CAREPROJ	CAREPSEU	CARERADI	CARERISTR	CAREROSE	CARESPAR	CARESTIP	CARFFENE	CARFERIB	CAREVULP	CAREWOOD	CARE 719

ad Perence	25	25	25	25	25	61	32	59. 67	15	55	60; 35	15	60; 42; 25	25	25	33, 30	25, 38; 25	25, 3K; 25	25; 11; 60, 25
۲Ŀ.				•	L	•	-	,			VE	VE .	VE	VE	-	-	VI:	VE	VF.
MM		-			•	•	ı	•	,				MM				-	-	NIN I
-	n I	נו ו	11	5	LL	•		'	=	•		=		n	n		•	-	
ME			-		•		-				-								
×	•		-	•		Ŵ	-				-		W.		•			•	W.
Cl-1	-		I	1	-			-	-		sÞ		dХ	ı					12
УĊ.	•	•		-		1	NI:				-	-							,
٧V			ŀ		-			ı				-	ı			αv	dΝ	. dk	٩v مار
V	,							14		-i-i			,	,	,	,	-		-
Fruit Type	urbene	achene	achene	achene	schene	uutiet	nuf	resembles- drupe	capsule-like- drupe	ciprisule	capsule	othsqua	agender.	schizocarp	schizacarp	cary opsis	capsule	eapsale	achene
۴IA		,				: -		н.	зO	Ē	:IO	0:1	=	:(0	30		<u></u>	Ŀ	¢
<u> </u>	z	z	z	z	Z.	Z.	z	z	z	z	V	z	<	z	z	z	z	z	۷
Ξ	<u></u>	<u>~</u>	. E.	크	-	<u>-</u>	-	<u> </u>	4	<u> </u>	<u>~</u>	<u>a</u> .	5 .	.	<u> </u>	-	.	-	-
:[: 	=	=	=	=	E			Ξ	x	>	=	x	=	=	=	:	=	Ξ	₽
¢#	e1	_	-	5	ų	38	\$ 5	57	2	ŝ	=	5	Ţ	-	-	=	£	50	¥
EAMILY	Cypenicene	Cyperaceue	Cyperaceae	Cypemeete	Cyperatesise (Cyperatesise)	Batulacene	Jughandaceae	Berheridaceae	Краптиястае	Celastraceae	Carsy upbryl lackau	Pyroloceae	Asteriicene	Apiacete	Aplacate	Ронсске	<i>ភព2</i> ភពវងមែល()	()hagancee	Asteraceae
SPECIES	CARE 868	CARIE 870	CARE_879	CARE_OV	CARE_SP	CARUCARO	CARYCORD	CAULTHAL	CEANAMER	CULANCAN	CERAFONT	CITIMUMBE	CHRYLEIIC	CICUBULB	CICUMACU	CINNLATI	CIRCALPI	CIRCLUITE	CIRSARVE

species.
à
attributes
plant
đ
Summary
ei.
Appendix

REFERENCE	25; 60; 25	1.S 'SE	25, 38	67; 25	60, 20, 25	64	(*¥	55, th	55, tH; 25	25	(t-1	()-{	64	64	25	2, 4I	12	12	12	33, 39; 60
۸I				VE	•	ı			ΥE								VE	VE	VE	
MiM	MM				MM				I								I	I	-	MM
7		'					,	'		'	•	•		•	=		'			•
MELS	r		-										: '							,
×	M		Ŵ	,	з	1	,	,		,		•	,		,	з	a.	ы	W	
CI-I	dS		-		512		,	,	,	,	•						ı	-		ź.
АС		El.	-		-	-				ЯŔ										
٧V	٩Ŋ	-		-		-					,							,	-	ΟV
٧I			-	FF.		da.	:EI	:I:I	:Li			EE.	E.	:1:1		-	,			
Eruit Type	achene	cupsale	achene	herry	uchene	qrupe	գութ	drupe	drujec	aru i	- Juluic	poste	երուծ	երուն	schizocarp	сирзынг	spore	ands	i onds	caryopsis
¥I4	0	4	ы:	101	¢	0.1	OF	ΕŪ	:iO	£1,H	\$	0	5	3	Я	0;1	ىغذ	끒	:	÷
-	<	z	z	z	z	z	z	z	z	z	z	z	z	z	z	z	z	z	z	~
	æ	-	=	-	-	4	-	-	<u> </u>	<u>.</u>	-	<u>-</u>	-	-	-	<u> </u>	a	d	-	<u>-</u>
ж. -	=	=	>	=	Ξ	×	2	~	s	z.	z	x	x	~	Ħ	÷	<u>.</u>	<u></u>	4	::
٦ ۳	÷	51	면	1 1	Ś	06	16	20	5	5	~		_	56	ŕI	ĩ	35	7	7	9
FAMILY	Asterneew	Partulpeaceae	Ranunenlaceae	L.iliperae	Asteraceue	Сопасеае	Comacene	Contacente	Cornucena	Beruhacene	Rusenceae	Rosencese	Reseauciae	Rosencene	Apiaceae	Orabiduceae	Aspteniaceae	Aspteniaceae	Aspleniaceae	Рывсеве
SPECIES	CIRSVIN.G	CLAYCARD	CLEMVIRG	CLINBORE	CONYCANA	CORNALTE	CORNFOEM	CORNEUCIO	CORNSTOL.	CORPCORN	CRAT SPI	CRAT, SP2	CRAT_SP3	CRAT_SP	CRYPEANA	CRYPCALC	CYSTBULB	CYSTFRAG	CVSTTENU	DACTGLOM

-	tencre					55	1,58: 25														
	REFER	9E 'EE	38, 34	25, 38	25	8, 58; 2	8,30,57	ΡE	67	12	12	12	ะเ	I; H	6U; 25	33, 30	25	60; 25	25	25	52
	VE		-	-	-	VE	VE	•	ı	νE	NK.	NT:	NE		ΝE		,	•	-	VE	ЗA
	MM	•	ΜМ	-			•			-		•	-	MNI	-	•	•	MIN	•	-	•
	1				-	-	•	•	,		,					,	n	'	•		'
	ME	-	-		WM			•			.			Ш							
	я									A.	×.	#	w.		,	,	,	w	W.	W.	W.
	г.	,	ЧS	I	•	-	-	SP.					ſ	SP	ЧS	-		dN.	•		
	л.	L	•	-		EI.	EL.	•			•	-				-		ı			
	٨A	٩Ŋ	QV	d٨		•	-					-	•			ΩV					
	М		•		•				4:1		,			,		,		ı	ı		
	Fruit Eype	cury opsis	schizucarp	legume	enpsuke	capsular	capsular	capanic	thrupe	spore	slbate	sport	spore	capsule	caryopsis	ejsda.Gab	cupsule	capsule	eapsule.	capsule	capsule
	ШΑ	-10	0	FO.	0	·	ا <i>:</i> ا	0.1	А	60	0k.	<u>Q</u>	÷	: <u>-</u>	ö	СЧ С		:UE	01	0F	ol:
	Ч	z	v	z	v	z	N	z	z	z	z	z	ż	z	N.	z	z	z	z	z	<
	E.H	۵.	æ	-	æ	-	1	÷	<u> </u>	~	~	4	-	<	<u>~</u>	-	-	-	4	a.	.
	1.1	5	=	=	÷	-	Ξ	×	×.	<u></u>	-	<u></u>	<u></u>	=	C	c	=	Ξ	Ξ	Ξ	=
	¢ ¢	+	y	7	÷	18	+	12	~	107	23	11	33	_	~	'n	~	5	(iĘ	+	7
	FAMILY	Poncae	Арінсенс	l'abgeac	Caryophyllaceae	Fumuriaceae	Fumuriaceae	Caprifoliaceae	Thymehoene	Asplettineeue	Aspleninceae	Asplegiaceae	Asplentacea	Cueuchitaceae	Peacene	Postoae	Orahuschuccue	Unagraceae	Опарписни	Omgracene	Ongraceae
	SPECIES	DANTSPIC	DAUCARO	DESMOLUT	DIANARME	DICTICANA	DICHERTORY	DIEKLON	DIRCPALU	DRYOUART	DRYOCRES	DRYOINTE	DRYOMARG	ECHLOBA	ELYMREPI:	JELYMVIRG	BHFVIRG	THEFTAR	BPLC01.0	EPILL, JEPT	TEPLE PARA

KEEPERGNCTE	1+"č	. 1.	12	12	12	60; 25	25	35	25	30, 40, 63, 25	24, 25, 25	20, 25, 25	34, 43, 20, 25	47; 25	tg, IN	25	25	60; 25	25	64; 60; 25
VE		ΛE	NF:	VE:	VIE	-	r			VE		•		жv	,			VE	·	ΝE
MM	ı	•	•		•	WW		,	ı	ŀ	-	•	MM	,	,			•		MM
=			•	•	•	÷	•	1	1	,		•	•	'		⇒	1	•	n	. '
ME			•												1			•		
3	W'	W	w	W	W	W	w	з	3		W.	W.	w	3				•		
(tđ		•	-			SP		. ,				-	sp	,	,	,		-1S		Я
 W	,				,	,		-	-	1:1			ı	ı	ż	,				
Ý		1			ı		,	, ,	, ,	,	-			,	,	,	,			
۲		1			,					,	,	,			-	,	,	•	,	14
Finit Eype	capsula	spure	spore	spore	spore	achere	achene	achene	uchene	capsule	מלוובשה.	achene	исично	nchene	riul.	cury opsis	citi upsis	caryopsis	car) opsis	acheac
ΥH	F()	10	0	≎	10	р	0	:to	,	ЪC	:10	je.	0,1	. 1 6)		c	÷	•	<u></u>	50
<u> </u>	<	z	z	z	z	z	z	z	z	z	z	z	z	z	z	۷	Y	ż	z	ž
FI	a.	=	-	-	-	=	<u></u>	æ	æ	-	-	-	-	<u></u>	<u></u>	-	=	-	a.	4
Г. Г.	=	2	EA.	EA.	Υġ	≖	=	=	=	=	=	=	=	=	÷	э	9	Ģ	÷	=
0	ŝ	ŧ	1	~	÷	10	Ŧ	×	~1	Ξ	s:	-	12	5	67	_	2	ī	30	27
FAMILY	Orchidaecae	Equisclucese	Equisciace	Equiserances	Equisetaceae	Азвенисае	Asterpread	Asteraceac	Asteracette	L. Husterne	Astenieete	Asteraucae	Asteraceae	Asteracac	Fagarette	Puatera	ן _א מנוהכסה	athatlo ₄ [puttere	Resources
SPECIES	EPIPIELL.	EQUIARVE	ROTHYEM	EQUILAEY	EQUISCIR	ERIGANNU	RRIGPHIL.	ERIGSTRU	ERIG_SP	HRYTAMICK	TEDAMACET	BUPAPERF	BUPARDGO	RUDICKAM	FACILICIRAN	FESTARUN	FESTPRAT	FESTRUBR	FESTSUBV	FRAGVESC

REFERENCE	64; 25	61	61	61	2. 41; 25	(1)	24, 25; 48	24, 25	24, 25, 21	31' TZ	24, 23	24, 25; 60	24, 25	24, 25	25	54; 25	60; 41	24, 25, 38	24, 25, 38	24, 25, 38
AE.	VI:	•		r	VE	'	•		1	•			-			VE	-			
MM		•					MM		MM			MM		•			MM			
Ξ	·	•	•	'	,	•			,	ı	•	·			э	'	-	· .	•	,
ME													•			ISM	ВМ			
w	,	×	Ŵ	N.	З		ı			,			,					•	-	
G	•		,			SP	ŀ			-		۶ľ		,			*	,	-	,
AC ¹		,			-	•	ы.	1	EI.		-		-	,	-		,			
VV			-		-	-	(IV	αv	QV QV	(N	ΩW	AD.	0V	AD.	-			(IV		(IV
V	:Li	,	'_	٦			,							,	<u> </u>		,			,
Froit Type	attette	Sanhara	samara	sumara	cupsular	nuter	schizocurp	schizocarp	schizocarp	schizocarp	schizocarp	schizocanji	schizocarp	schrocusp	capsule	schizwearp	schræcarp	achene	uchene	achene
L I I	ē				<u>ند</u>	:10	£	10	Ч	·	94	÷	5	,	ā	ŧ	:0	ē	5	Ę.
-	z	z	z	z	z	<	z	z	z	z	z	z	z	z	z	z	<	z	z	z
Ξ	<u> -</u>	<u> </u>	<u> </u>	<u></u>	-	<	<	-	<u>~</u>	<u>a</u>	<u> </u>	<u>^</u>	-	-	<u></u>	a.	=	<u>.</u>	-	
1	=			-	=	=	=	=	=	=	=	=	=	=	≍	=	=	=	=	=
O,#	ĹĒ	50	55	15	2	~	[]	~	-	I	13	-	501	_	~	드	3	_	11	5
FAMILY	Rasuecie	Oleaceae	Ојевистеме	Oleaceae	Orchiduceae	E.matiaecae	Rubinccue	Rubinecue	Rubioceae	Rubiwene	Rubineue	Кирінсеас	Rubincene	Rubiaccae	Gentiamaceae	(ieturhisktrad	Gerahiateate	Rusancad	Rosaccie	Rusticette
SPECIES	FRACIVIRG	FRAXAMER	FRAXNICIR	FRAXPENN	GALESPEC	GALETER	GALIAPAR	GALIASPR	GALICIRC	GALILANC	GALIOUSTU	GALIPALA	GALITRIF	OALL SP	GIENTANDR	ORAMACU	GERAROME	GELMALEP	GIGIMCANA	GEUMLAU
REFERENCE	24, 25, 38	24, 25, 38, 53	24, 25	57 .	12; 25	25, 38	3, 5, 30, 50	20, 25	20, 25	25	61)	66.(0	Đ-l	20, 25	25	25	Ŕ()	20, 25	60; 20, 25	20, 25
------------	------------	----------------	----------	-----------	--------------	-------------	---------------	------------	------------	-----------------	-------------	------------	--------------	------------	-----------	-----------	--------------	------------	------------	------------
ξ	,				VE			۳.	ЯE	VE					VE	VI:		-		
MM	-	-	-		ı	-			-		-	•	ı	-	-	-	,		MM	
=		•	-	IJ	'	_ '	,	•	-	11	-	•	L	-	11	ιr	-	-	•	•
Ж		,		•							-		EM		-	-	•	L		
≩		•	-		W			w	w	-	-	-	•	W	-	-	•	w	w.	3
04		•	•	•	-	-		•	-		sP			-	-	-	с.	-	4N	
Ъ,	· .						Ē						•	-	-	-				
AA	٩N	(IV	ΩV			(IN	ı	,			-	(IV	•	-	-	I		,	,	
N			•	•	ı		ı		-	-		-	•	ŀ	-			•	,	
Fruit Type	uchane	attaitat	uchene	caryopsis	spore	ntilet	исћење	uchene	achene	capsule	capsule	car) opsis	capsule	achene	cupsule	capsule	capsule	achene	achene	achene
нΑ	50	OF:	,	- 04	0.†	10.ł	÷.	-10	оĿ.	();]	. 0	61	OF:	0	01:	-	0E	Юŀ,	÷	
Â.	z	<	a	z	z	z	z	<	<	7.	Y	z	z	×	z	L)	2	Ż	ĸ	Ŧ
≡	<u> </u>	₽ .	-	-	-	≠	~	<u>~</u>	۵.	<u>а</u> .	γ	-	۲	<u>.</u>	4	4	4	B	Ξ	8
ň	۳	=	Ħ	9	-	=	=	=	=	=	I I I	Li	H	н	Π	=	=	Π	=	11
¢.	2	-1	23	62	24 24	Ŧ	32	_	60	۲·I	H	ſ	60	2	1	5	3	\$	5	r.
FAMILY	Rosaceae	Rosuceac	Rospeche	Ризсере	Aspteniuccie	Bongingceue	Kanunculaceac	Asteracene	Asteraceue	11ydrophyllacae	C'hushurene	heateste	Balsanimucae	Asteracore	Iridacene	Iridacene	Juncaceae	Asteraciae	Asterneac	Asteraceae
SPECIES	GEOMBIVA	GEUMURBA	GEUM_SP	GLYCSTRI	GYMNDKYO	HACKVIRG	HEPAACUT	HIBRAUKA	UIBRCAES	IIYDRVIRG	IIYPEPERE	IIYSFPATU	HAY. YAMI	INDUTIEL	IRISVERS	IRIS_SP	INNEL. JNITE	LACTOANA	LACTSERK	LACT_SP

REFERENCE	25	33, 39, Mr 25	31, 39; 25	25	2, 41	75, ST	25	55, 64	55, 6 1	55, 64	25	12	12	12	12	25	25	(1)	25	64; 25
VE	ЗA .	ΥE	VE	•							ЯĿ	-		,	'	VE	ΥE	,	V1:	NIX .
MM		MM	-	-							-		-		-		,			1
11	- I-	•		0		•	n			. '	-	•	•	•	'	=	=	,	11	,
ME	1	-	-			WM	-					•				•				
*			,		ж			,	,	-	•	W.	w.	ĸ	W.	,	,	,	-	,
([:]		sP	-				-				,				-		,	as		,
AC.	-			4	•															
A A		UΝ	đV												,					
Ŧ	ı			-	,			: :::	:4:1	. 4 :I	١.			-						:1:
Fruit Type	uchene	caryopsis	caryopsis	talbur	capsular	capsule	capsule	herry	њету	herry	mulei	spure	ърние	spore	spore	untici	enpsule	capsule	capsule	hum,
VII	101	0	0.1	зЮ	oĿ	0	,	ı:	63	();	10	{) .[0,1	· ,	64	đ	ö	:tO	0F	63
<u></u>	z	z	z	<	z	z	z	z	z	z	z	z	z	z	z	z	z	v	Z	z
E	4	4	el .	ri,	.	<	a	~	-	4	-	-	4	<u>~</u>	=.	-	-	4	ų	
31	Ξ	σ	9	=	=	=	Ħ	×.	*	s	Ξ	РA	\mathbf{V}_{i}^{\dagger}	Y:	ĿУ	=	=	1	H	=
ð#	18	÷	_	ŀ	Ŧ	-		32	1	01	ş	6	Ę	→	_	ŝ	₫	0	_	130
FAMILY	Unticaceae	Рояссие	Puaceue	Lumiaceate	Orchiducene), surproduction	Cumpandiateae	Caprifollaceae	Cuprifoliaceae	Caprifoliacore	- apprique conce	Lycopodiaceae	Lycopodiaceae	Lycopodiaceae	Lycopodiaceae	Lamiacenc	Primulaceae	Primulaceae	Primulaceae	Liliaceae
SELES	LAPOCANA	LEERORY7	LERVIRG	LEONCARD	LIFALOES	LOBENEL.	LUBP_SP	LONICANA	IORINO I	LONITIES	LYCOAMER	LYCOANNO	UNECODEND	LYCOOROB	LYCOTRIS	LYCODNIE	FIXECILI	LYSINUMM	1.YSUTERR	MARACANA

areasonche	. 25	: 25		; 60		. 60	. 59			: 25	45. 25	-			, 39; 25	, 39; 25			: 25	.39
<u>/ :</u>	71: 64	11: 04	/1: 25	81·	25	H.	55	٤I E	55	1: 64	/E 9,	1: 25	25	25	76 33	1: 33	25	72 12	60	8
	-	-	_	- W		- N		•	•	-	-	_	•	•	-	-	'	_	м -	<u> </u>
N I	'	· ·		W	•	М	,	,	·			. <u>.</u>	· ·	· -	•	•	-		MI	
\vdash	'	'	'		-	•	1						-	-	•	'	-	'	•	•
E		•		,		,			,		KS.						,	,		
×			×	,				'	•									¥	w	
Ē			• •	÷		ż		dy.											÷.	
ΥU	,			H.		E														
V							,	,				,		,	٩٧	٩V	,	,	-	4D
ž	4.4	ЫR		:,	ı		FI:		,	÷:	•						,			
Fruit Type	herry	њену	spore	legume	едиле	legume	drupe	anul ici	cary opsis	heny	capsule	nuka	eapsule	capsule	ciaryopsis	sistleAna	nulet	spore	achene	cary opsis
É	<u></u>	() ()		dF.	0	÷	5	ē	<u>~</u>	-	· <u>·</u>	30		 	¢	ä	\$	9	c	<u></u>
÷.,	z	z	z	 	۷	V	z	z	z	z	z	z	z	z	z	z	<	z	V	z
Ξ	-	. <u>~</u>	-	≏	÷	÷	÷-	<u>-</u>	<u></u>	<u>-</u>	<u></u>	<u> </u>	~	-	-	<u> </u>	<u> </u>	a.,	£	s.,
<u>يند</u> 	=	=	÷-	=	=	=	>	=	3	>	Ξ	Ξ	Ħ	Ħ	3	5	=	<u>.</u>	=	Ģ
) *	5	Ξ	ŝ	=	÷	5	F 4	<u>.</u>	-	8	Fi	-	ы	45	_	-	_	Ş	3	S.
FAMILY	Liliaceae	Liljacene	Onseleptene	Fabattae	Гаћисас	beautiged.	Menispernueeue	Lemissread	Poaceac	Rubisceae	Navifragnecae	Lathnactare	Monotrapaceae	Monotropaceate	Punceau	Panceue	adaatiitititi	anananant 1	Asteraceae	Papeare
SPECIES	MAIARACE	MAIASTEL.	MATTSTRU	M&DILUPU	MELIALIBA	MELIOPPI	MENICANA	MENTARVE	MILIEPO	MERRER	MUTEDIM	MONAFIST	MONORYPO	MONOHNIL	MUER PRON	MUELIMEXT	NEPECATA	ONOUNENS	NV,)VdONO	ORYZASPE

GENCE	66				25				25			AH, 25		60					
RIEP HI	25, 53, (ก	15	19	60; 42;	75	67	25	31, 43; 3	25	6 .	33, 49; 6	12	33, 39, (25	38	61	25	₹
ž			•	,	VE.				•			VE	VE			-			i '
MM	,	-	-	ı	MM	-	ŀ		ММ		-	MM		MM	I	+	-	I	
-		,	,	'		'		11	•	1			,	,	1	,	'	11	,
MIE	•	•	-	ı	ЕM	EM		-			-	-					-	-	
¥	•	w	w	w	-	-			w				ж	,			w		¥
ŝ				-	st	•			SP			SI		л.	-	4		-	
¥			-				ļ ,			-	•		•	F			-	•	
×	٩٧			-		•		-		-		αv		άN		٩٧			_
¥	Ŀ				-		HF.	-	-	-	·ŀ	•						,	,
l'ruit Type	schizacarp	sporv	spare	nutci	capsule	capsule	berry-like- drupe	sisto.(inc	cury opsis	cary opsis	herry	canyopsis	spore	caryopsis	capsule	achene	winged seed	achene	winged seed
۷II	· 12.	();I	U.f		:#: I		iin.	:lO	÷	-	0F.	•	¢	5		<u>نہ</u>	ı	FO.	
<u></u>	z	z	N	z	z	z	z	z	z	z	z	z	z	<	N	z	z	z	z
Ξ	-	<u>-</u>	4	*	-	<u> </u>	-	-	<	=	÷.	÷	~	-	-	<u> </u>	4	<	<u>+</u>
1.F	=	-	4		Ξ	≍	=	9	÷	:=	~	0	*	9	=	=		=	÷
) ₹	2	5	1	81	5	_	-	~	_	~	(JC)	ſ	_	¥7.	Ē	61	Ś	ม	_ <u>_</u>
EAMILY	Аріасене	Osstupducene	Osmunducene	Betulaceae	Oxalidacene	Oxulidareae	Areliaceuc	Potentie	Poaceae	Posecse	Vitaceae	Poaceae	Thetypteridaceae	Pouceae	Polemanineare	Verbenaceae	Pippeene	Unicaceae	Pinuceae
SPECIES	OSMOCLAY	OSMUCLAY	OSMURACIA	OSTRVIRG	OXALSTRU	OXAL_SP	PANAQITIN	PANIACUM	PANCAPI	PANISP	JISNU JV	PEIALARUN	PHEGCONN	PHLEPRAT	PHLO SP	PERVLET	PICEOLAU	IWIN-The	PINUSTRO

GETERLENCE	0	0	5	3, 39	7, 33, 39, 25	3, 39	3,30,60,25	2	5	7	2	K, S.I., 25	ę	7; 25	0	6	6	0	\$	\$
Vh. I R		ء -	•	-	VE I	ر ا .	VI: 3	.] 2		φ -	VIC 1	VI: 3	-	VE 6	-		-	÷	-	۳۱ ۱
MM	-			-			MM		•	•										ı
=		•	1	-		-	•	Ξ		•	'	'	•		1		'	ı	a	
NIE N			1	-	T				-	•	-	-		-	-	-		-		
÷	-	-	-							•	w	-	•	-	w	Ň	я			≥
ŝ	ŝ	sР	-	•		-	sl			•	-	-	SP			-		4s		
.уĘ	ı	-	- '			-	-		-			LJ.	•	-	·		•	-	•	
٧V		-	1	٨D	AD.	AD	٩Ŋ	•	•	-			-		,		,			
AI A			,			•		•	•	лu	-	-	-	1-h	-					
Henit Type	alusqu	eupsule	capsale	earyopsis	caryopsis	sisdovica	car) opsis	caryopsis	cary upsis	herry	spore	capsula	archene	hurry	capsule	capsule	capsule	achene	achene	achene
VI	0	ö	зłО	64	сŀ	¢	0]:	0:1	,	сця :	Q.	ΡO	0		,			0ŀ,	лO	<u> </u>
<u> </u>	<	<	z	z	z	z	z	z	z	z	z	z	۷	z	z	z.	z	۷	Y	z
=	<u>-</u>	<u>-</u>	-	<u>a</u> .	<u>.</u>	54	=-	a.	-	-	-	<u>م</u>	A	-	<u>~</u>	<u>a</u> .	<u>г</u> .	đ	Ч	<u>a.</u>
Ë	=		⊐	5	9	<u>е</u>	σ	σ	<u>с</u>	=		=	=	=			<u> </u>	=	Ξ	=
) *	~	\$	9	£		53	26	ri -	÷.	ч ^с ,	а 1	ş		92	6	30	R	-	2	R
FAMD.Y	Plantaginucae	Pentugimucae	Pluntuginuceae	Poaceac	Ponceac	Poncate),tuticate	putication	Postcene	Rerberidaceae	Aspleniacenc	Pulyguluceae	Polygomacor	Laliquene	Salicareae	Sulivarene	Sulicacene	Rosucene	Rosaccae	Asteniceate
SPECIES	PLANLANC'	PLANMAJO	PLANKUGE:	POA_ALSO	POA_COMP	POA_PALU	FOA, PRAT	POA SALT	FOA_SI	PODOPELT	POLYACRO	POLYPAUC	POLYPERS	POLYPRINE	POPUBALS	POPUGRAN	POPUTREM	POTENORY	POTERECT	PREN_SP

REFERENCE	6 1	64; 25	60	12	25	35, 49	35, IV	32, 49	35, 49	25	61)	25	25	53, 67	51, 67	64, 25	55, 64; <u>1</u> 5	55	55	25
VF		ЧE		ΥE	VE		-		ŀ			NE.		-		VE.	VE			
MM		-		,	-			-						-	-	-		-		
Ξ	•	I	ı	,	ı.		,	•	•	11		-	1		'		•	,		
ME			•							•	-	•		-	-	-		•		
*	•		,	W		'		-					,			,	•			
เร		,	SI'					-	1								-		, ,	
VC.	,					nı;	ΪN	ль:	ä				,	,			,	-		,
VV.						-		-		-			,	,	,		-	-	•	,
र	alat	ĿI;				-				,			,	÷	÷	÷	лd	ĿЕ:	лı	49
Eruit Type	drupe	drupt	. captraut	spure	capsule	100	W	nit	Lint	uchene	achene	achene	achene	drupe	drupe	drupaceous	snoaaduup	berty	burry.	herry
۲	ŀ	-10	:10	.dO	2		,			60	=	ţ:	<u>~</u>	ō	ă.	.40	d0	Ŀ0.	FO.	2
<u> </u>	z	z	z	z	z	z	z	z	z	z	۲.	z	z	z.	v	z	z	z	z	z
ET)	-	<u> </u>	<u>.</u>	<u></u> л	\$4	۵.	-	-	-	-	<u>.</u>	-	*	-	<u> </u>	: ~	-	4	a	-
1,F	I.	x	=	-	×	÷			-	=	=	=	=	×	×	~	x	×	×	x
С#	98	123	:	42	21	\$	2	1	_	50	16	_	5	-	7ħ	45	_	ĿI	1 6	÷
FAMILY	Rosaceae	Kospene	Lumiucauc	Dennstaedtiuceue	Pyrolaecae	Fagacone	Fagaceae	Fagueeue	Fagneeue	Ranoneulacene	Rammendaceae	Rauncoluceae	Ranunculaceae	Khempaccac	Rhummeene	Anseardincese	Anucardiaceae	f i rossidiar juicepe	Gressuluriaceae	Orossolurineate
PECTES	RUNSERO	RUNVIRG	MUNVULO	TERAQUI	PYROBILI	QUERALBA	QUERMACK	QUERRUBR	QUER_SP	RANUABOR	KANUACRI	RANIJIJICA	RANDRECH	KIYMVI'NI	BUAMCATH	RIDSKADI	RILISTYPH	RIBEAMER	RIBECYNO	RIBECTAN

REFERENCE	25	25	25	25	60	 Στ	25	64	6-1, 60, 65; 25	6-11 <u>25</u>	52	64; 25	25	52	75	14: 25	10, 25	19: 25	19: 25	6H; 25
ΥI:			ı	•	ı		'	ı	ΥE	NI:	NI:	AE.	ΥE	ı		NI:	VI:	NI:	NE.	AI:
MM				,	MM			I	MM	-							I	,		
Ξ	·	•	'	•	ı	,	1	,	•	'	,		•	-		ı	'	,	'	,
NIK				•	EM			ı		-		ı						•	-	
з				•			,			·	,				ы	a.	3	×	3	,
(ii	,			•	SP	ı	-		is .			,		ı	-		,			,
YC.			,	,			,	-				,								
ž		•	ı		ı	1		,	,		,	,							,	
₹	14	ьk	цĿ	ЪР		ż	14	÷	i:	÷	:1:1	ЧH	끈	. '	,					11
Fruit Type	berry	liurry:	herry:	herry:	legume	achene .	neltene	drupelet	drupelet	drupelet	drapolet	drupelet	drupolet	achene	achene	copsule	capsulo	cupsule	eapsule	herry-like
ΝĮ	63	-10		-	ŀ	лo	:IO	-0	ö	ы	c	þ	,	σ£	0	:-	ö	÷	÷	ЗD
<u></u>	z	۲	z.	z	ĸ	z	z	z	<	z	z.	z	z	z	z	z	z	z	z	z
Ē	-	<u>-</u>	-	<u> </u>	a.	•	-	<u> </u>	<u>-</u>	-	<u>-</u>	đ	*	곡	-	-	-	-	1	-
÷	<i>9</i> .	~	×	×		×	×	~	×	*	x	x	x	=	=	s.	×.	×	*	×
0 *	_	•	Æ	-	2	÷.	÷	×	53	٩ ا	=	÷	-1	_		-7	ŝ	10	I	×
			ر	2											- Inc					ceate
LAMIL	Grossulariacea	Orossulariaceu	Grossulariacen	Grossulurince	Fabaceae	Rosaccae	Rosaccae	Rasacene	Rosucae	Russuene	Resaccate	Krisaceae	Rusacene	Asternotic	Polygonace	Solicueae	Salicatene	Sulicaceae	Salicacene	Capitolia.

REFERENCE	55; 26, 60	3,7,23,30,01,40,57,25	25, 3 K , 53	25, 38, 53	25, 38, 53	4E 'EE	25	25	25	60); 25, 42	35	25	(i-1)	59	55, 64; 43, 60; 25	47, 25	47; 25	60; 47; 25	47; 25	17, 25
ΝE	,	NR.	-	•	•	•	•	ΥE	,			,	,		лĿ	YE	VI:	ΥE	VI:	ЧE
NM	MM	-	-	•	•		•			MM	-		•		MIM	-	- -	ΜМ	-	,
Ξ	•	-	,	•	•		•	=	=	-	-	11	•	1	•		•	•		
ME			-			-				INM		-	•	-	-	-	•	•	-	
×	•	-	-	-	-	-	ı						•	-		W.	W	W.	W	W
เน	SP	-	-			-		•	ı	dS		-	•		ЧN		,	÷.		
۶¢.	•	EL,	-		-	-		ı	ı								•	,		
VV	-	•	ŧΨ	dΝ	ίN	άk	()V	,	,		,		-							
Z	E]:			-			-	,	r				FР	Η	1.1			,	-	
Fruit Type	berry-like	capsule	schizucarp	schizucurp	schizwerp	unyopeis	uchene	nutlet	capsele	capsule	capsule	schrocarp	herry	het i S	be ну	већене	achene	achene	achene	achene
۲I	БŪ	-	FO	Ŀ	-	÷	0	0/	0:1	0	=	сŀ:	OF.	dO	60	зЮ	:	0F	Ľ	дO
<u> </u>	z	z	z	z	z	z	z	z	z	Y	z	z	z	z	٧	z	z	z	z	z
	<u> </u>	<u>-</u>	-	8	=	<u>-</u>	-	-	<	5	-	<u>.</u>	4	<u>-</u>	4	-	4	4		<u>.</u>
н. Т	×	=	Ξ	Ħ	Ξ	9	=	=	=	=	=	Ξ	>	>	~	=	Π	=	=	=
Э́т	Ę	=	ţ	1	-	24	×	=			٠	~	51}	ĩ	ΰŞ	28	4	ş	13	£
FAMILY	Caprifuliacae	Раричегасеве	Apiecee	Apiucoue	Apinene	Postsic	Cyperaceae	Lamincene	Courbitaceae	Caryophy Bacene	Indaceae	Apiateate	Smilacaceae	Smilaencene	Solomocae	Asteracene	Asteriteeae	Asteraceae	Asteratoate	Asterneete
SPECIES	SAMIBRAPU	SANGCANA	SANIMARI	SANTRUE	SANL_SP	ANDAILESS	SCHRATIRO	SCUTATE	SICYANGU	DURVEUS	SISYMONE	NVDSINDIS	SMILHERB	SMILIEP	SOLADULC	RUALIT	SOLICAES	SOLICANA	XEDUDS	SOLIGICA

16F

KIEFERENCTE	47; 25	47; 25	47; 25	60; 25	hU; 25	33, 39; 17	25	SI'	25	60; 25	19	25	25	12, 25	12, 25		15, 25	61	25	25
VE	VE	•	VI:	VI:		'	'	NI:				•		VE.	VI:		ΥI	•		ЯN
MM		•	•	MM	MM	MNE		-	-	MM	-				I	1	1	,	•	
=	•	•	•	,	•	•	()	I			,	=	۲ı	•	'	-		-	•	=
MK				1				, ,	ŗ	-			-	•	-	-	WS	-	-	
≥	з	N.	N.	M	w	W	•	,	,	w	•		-	W	, w	w		w	w	• ,
Cld				ЧS	ЗP				,	ż			-	-	-	•				
¥.				,	,	•		EI.		-				· · · ·	-				-	
V			1	ı		Π								-				-		
₹				,		•			÷	,	ĿĿ.			-			-			,
Fruit Eype	nchene	achene	achene	achene	achene	caryopsis	capsule	berry	վախտ	achene	arillate seed	achene	achene	shote	ajods	winged seed	eupsule	nut-like	achene	capsule
٧Ħ	;to	ö	ö		0	Ē	:0	ж	50	:4C)		0:1	-i0	0,I	-10		:		0	0.1
-	z	z	z	<	<	z	Z.	N	z	V	Z.	z	z	z	z	Ÿ	'n	z	٨	z
II	<u>.</u>	<u> </u>	<u>a.</u>	<u> </u>	<	<u>-</u>	6 .,	^	<u>-</u>	<u>-</u>	<u>-</u>	: =	-1	-	-	<u>-</u>	п .	đ	Ħ	4
÷.	=	=	=	=	=	:=	=	=	×	Ξ	×	Ξ	=	<u></u>	<u></u>	-	=		=	=
0,4	~	=	.	~	2	15	-	*	ş	66	s.][5	₹	۶	Ŧ	47	50	511	5	Ŧ
FAMILY	Asteraceae	Asteraceae	Asteraceae	Asteraceae	Asteraceae	Poscese	Curyophythecene	Lilliaceae	Cuprifoliaceue	AMeraceae	Дихичене	Ranusculaceae	Rommentaceae	Aspleniaceae	Aspleniaceae	Cupressaceae	Savifinguene	Tibaceae	Asteraceae	Primulaceac
SPECIES	SOLIFUNC	SOLINEMO	SOLIRUGO	SONCARVE	SONCOLER	SPEILINTE	STELLONG	STREROSE	OB.IA9MYS	TARAOPPI	TAXIITANA	TRALING	THALPUBE	THELNOVE	(TTVA' CHJ	LYDOMITI	TIARCORD	NEIWVITEL	BUILD VAL	TRIJEBORU

NCE			30; 25										\$	58, 62					
REFERE		3, 27; 25	3, 22, 27,	19	61	25	60; 25	19				25	36, 60, 25	7, 15, 57,	()	25	60; 42	د ر	16 441
VIC.		VE	٧£	-	VI	-	VE	-	-	•	-	VI:	NI:	NI:	NI:			•	
MM		•					MM	L	-					•	•	ŗ	ЫМ	-	
1:		-		ı	-	'					•	5	'	•	·	L	-	LI I	
ME		-		-	-	-			-	-			-	•	•		WN	-	
×		-			w	M	w	w				•				ı	-	-	
62	N.	-		•	Ľ		sР						SP		-		ЯP		12
AC		EI.	EL,		•		•			-				EI.					
Ŷ			•										-			,			
2		,	•	:H:							,		-	•	:Lit	,			
Fruit Type	legume	herry	lkerty	drupe	winged seed	achene	achene	SURVICE	caryopsis	caryopsis		achene	achene	cupsule	Quad	nutlet	capsular	gutiet	to the second
VII	0	*	2	2	,	0	¢	ı			-	0:1	0.1	0:1	- 30	3	с	F()-	121
<u>-</u>	<	z	z	z	z	<	z	z	-	1	=	z	٧	z	ž	z	Y	N	Y
=	<u> </u>	۵.	-	-	-	-	<u> </u>	<u> </u>	=	=	-	-	-	*		*	=	-	-
Ξ	=	=	=	=	-	=	≍	-	э	e	=	=	=	=	×	=	=	=	-
≎ *	r.	2	138	_	3	¢.	-	99	_	-	-	÷.	_	=	5	_	ſ	. –	-
FAMILY	Fabuceuc	Liliaceae	Liliaceae	Caprifoliucene	Pinaceae	Asteraceae	Турћисење	1)[ntercence	Ронсае	Paperan	•	Unicaceae	Uthieucene	Lillistocae	liticheette	Verbensteelte	Nerophulariaceae	Verbenaceae	· · · · · · · · · · · · · · · · · · ·
SPECIES	ATANAINT.	TRILEREC	TRILCIRAN	TRIOAURA	TSUGCANA	TUSSFARF	TYPILATI	ULMUAMER	UNKN_751	LINKN_701	INNN R22	REFURER	UKTIDIOI	LIVULGRAN	VACCANGU	VERBIAST	VERBTHAP	VERBURT	The second second

REFERENCE	šl, 35 , 64	51, 55, 64	\$1	51	60; 42	7, 25	3, 5, 10, 30; 4; 25	30; 25	15; 4; 25	ST; 25	-18, 15	3, 5, 15, 31; 4; 25	15; 25	ST; 25	ST; 25	64	<u>15</u>
VE :	•					VE	VE	VĽ.	. VE	VE	Vŀ.	VE	٧E	٧R	VI(VI:
MM	-	-			MM		MM	-	MM		-	MIM	1	-			
=	,		•		•	•	•			'				,	•		•
MIS		-			EM		ΕM	-	MH	-		IN3			•	-	•
ž		-	•			•		-	•	-		-	•	•	•		•
(년 -	-		-	-	SP	•		-	•	-					•		•
ΑŬ						EI.	EL.	ĿF.	Ю.,	- 1:1		÷		Ē	EI.		
٧V			•		,	-	-			-				•			
А	ΕĿ	ŀF	н.	РF:	-						,			,		Яd	÷
Prini Type	ղորե	րորե	վուրբ	drupe	legune	cupsule	cupsule	capsule	cupstile	cupsule	capsule	capsule .	capsule	capsule	caps ule	berry:	uchene
11A		:40	0,1	ĿО	0	64	FO.	-	5	0E	<u>ــــــــــــــــــــــــــــــــــــ</u>	·	0. L		,	Ċ.	ĿО
<u> </u>	z	z	۷	z	А	z	z	z	z	z	z	z	z	z	z,	z	z
H-1	-	-	'n	<u> </u>	<u>-</u>	÷	-	÷	<u> </u>	<u> </u>	-	<u>a</u>	<u>-</u> .	<u></u>	2	۹.	: -
1,17	z	x	×	~	Ξ	ī.			н	Ξ	=	=	=	=	=	>	=
0#	15	28	ŕ1	ł	2	4	6 t	*	13	15	89	-	-1	_	5	63	23
FAMILY	Caprifoliaceae	Cuprifolituceae	Сиргіїніясяю	Copri folisecae	Pabaceae	Vjoliacese	Violacoae	Violaceae	Violacue	Vinhisene	Violaceae	Vintuceue	Viulaceae	Violuceue	Violacenc	Vituceae	Rasaceae
SPECIES	VIBUACER	VIBULENI	VIBUODRII.	VIBUTRU.	VICICRAC	VIOLAPPE	NV BEIOTA	VIOLCANA	VIDLAU	VIOLABIC	VIOLIPIRE	VIOLROST	VJOI,SORO	V101788	ASIOTV	VITRIPA	WALDERAG

APPENDIX 3 SPECIES CODES

Species Code	<u>Scientific Name</u>
ABIEBALS	Abies balsamea (L.) Miller
ACERNEGU	Acer negundo L.
ACERRUBR	Acer ruhrum L.
ACERSACC	Acer saccharum Marshall ssp. saccharum
ACERSACN	Acer saccharinum L.
ACERSANI	Acer saccharum Marshall ssp. nigrum (Michaux f.) Desmarais
ACERSPIC	Acer spicatum Lam.
ACHIMILL	Achillia millefolium L. ssp. millefolium
ADIAPEDA	Adiantum pedatum L. ssp. pedatum
AGRIGR YP	Agrimonia gryposepala Wallr.
AGROGIGA	Agrastis gigantea Roth
AGROSTOL	Agrostis stolonifera L.
ALLITRIC	Allium tricoccum Aiton
ALNUINCA	Alnus incana (L.) Moench ssp. rugosa (Duroi) Clausen
AMBRARTE	Ambrosia artemisiifolia L.
AMELARBO	Amelanchier arborea (Michaux f.) Fern.
AMELINTE	Amelanchier interior Nielsen
AMEL_SP	Amelanchier species
AMPHBRAC	Amphicarpaea bracteata (L.) Fem.
ANEMCANA	Anemone canadensis L.
ANEMQUIN	Anemone quinqefolia L.
ANEMVIRG	Anemone virginiana L.
ANTENEGL	Antennaria neglecta E. Greene
APOCANDR	Apocynum androsaemifolium L. ssp. androsaemifolium
AQUICANA	Aquilegia canadensis L.
ARALNUDI	Aralia nudicaulis L_i
ARCTMINU	Arctium minus (Hill) Bernh. ssp. minus
ARISTRIP	Arisaema triphyllum (L.) Schott ssp. triphyllum
ASARCANA	Asarum canadense L.
ASCLINCA	Asclepias incarnata L. ssp. incurnata
ASCLSYRI	Asclepias syriaca L.
ASTECILI	Aster ciliolatus Lindley
ASTECORD	Aster cordifolius L.
ASTEERIC	Aster ericoides L.
ASTELANC	Aster lanceolatus L.
ASTELATE	Aster lateriflorus (L.) Britton
ASTEMACR	Aster macrophyllus L.
ASTENOVA	Aster novae-angliae L.
ASTEPUNI	Aster puniceus L.

Species Code	Scientific Name
ASTEUMBE	Aster umbellatus Miller
ATHYFILI	Athyrium filix-femina (L.) ssp. angustum (Willd.) Clausen
ATHYTHEL	Athyrium thelypterioides (Michaux) Desv.
BETUALLE	Betula alleghaniensis Britton
BETUPAPY	Betula papyrifera Marshall
BIDEFRON	Bidens frondosa L.
BOEHCYLI	Boehmeria cylindrica (L.) Sw.
BOTRMATR	Botrychium matricariaefolium A. Braun ex Koch
BOTRMULT	Botrychium multifidum (S. Gmelin) Rupt.
BOTRVIRG	Botrychium virginianum (L.) Sw.
BRACEREC	Brachyelytrum erectum (Schreber in Roth ex Sprengel) P. Beauv.
BROMINER	Bromus inermis Leysser
CALTPALU	Calthus palustris L. ssp. palustris
CALYSEPI	Calystegia sepium L.
CARDACAN	Carduus acanthoides L.
CARDNUTA	Carduus nutans L.
CARDDIPH	Cardamine diphylla (Michx.) A. Wood
CARDPENS	Cardamine pensylvanica Muhlenb. ex Willd.
CAREALBU	Carex albursina E. Sheldon
CAREALOP	Carex alopecoidea Tuckerman
CAREARCT	Carex arctata Boott
CAREBACK	Carex backii F. Boott
CAREBEBB	Carex hebbii (L. Bailey) Olney ex Fern.
CAREBLAN	Carex blanda Dewey
CAREBREV	Carex hrevior (Dewey) Mackenzie ex Lunell
CARECEPH	Carex cephaloidea Dewey
CARECOMM	Carex communis L. Bailey
CARECRIN	Carex crinita Lam.
CARECRIS	Carex cristatella Britton
CAREDEWE	Carex deweyana Schwein.
CAREDIGI	Carex digitalis Willd.
CAREGRAC	Carex gracillima Schwein.
CAREGRAN	Carex granularis Muhlenb. ex Willd.
CAREHIRT	Carex hirtifolia Mackenzie
CAREHITC	Carex hitchcockiana Dewey
CAREINTU	Carex intumescens Rudge
CARELANU	Carex lanuginosa Michaux
CARELAXI	Carex laxiflora Lam.
CAREPECK	Carex peckii Howe

Species Code	Scientific Name
CAREPEDU	Carex pedunculata Muhlenb. ex Willd.
CAREPENS	Carex pensylvanica Lam.
CAREPLAT	Carex plantaginea Lam.
CAREPLAT	Carex platyphylla J.Carey
CAREPRAI	Carex prairea Dewey
CAREPROJ	Carex projecta Mackenzie
CAREPSEU	Carex pseudo-cyperus L.
CARERADI	Carex radiata (Wahlenb.) Small
CARERETR	Carex retrorsa Schwein.
CAREROSE	Carex rosea Schk. ex Willd.
CARESPAR	Carex sparganioides Muhlenb. ex Willd.
CARESTIP	Carex stipata Muhlenb. ex Willd,
CARETENE	Carex tenera Dewey
CARETRIB	Carex tribuloides Wahlenb.
CAREVULP	Carex vulpinoidea Michaux
CAREWOOD	Carex woodii Dewey
CARE_719	Carex specimen D719
CARE_858	Carex specimen D858
CARE_868	Carex specimen D868
CARE_870	Carex specimen D870
CARE_879	Carex specimen D879
CARE_OV	Carex sect. Ovales
CARE_SP	Carex species
CARYCORD	Carya cordiformis (Wangenh.) K. Koch
CAULTHAL	Caulophyllum thalictroides (L.) Michaux
CEANAMER	Ceanothus americanus L.
CELASCAN	Celastrus scandens L.
CERAFONT	Cerastium fontanum Baumg, ssp. triviale (Link) Jalas
CHIMUMBE	Chimaphila umbellata (L.) Barton
CHRYLEU	Chrysanthemum leucanthemum L.
CICUBULB	Cicuta bulbifera L.
CICUMACU	Cicuta maculatum L.
CINNLATI	Cinna latifolia (Trevir. ex Goeppinger) Griseb. in Ledeb.
CIRCALPI	Circea alpina L.
CIRCLUTE	Circea lutetiana L. ssp. canadensis (L.) Aschers.& Magnus
CIRSARVE	Cirsium arvense (L.) Scop.
CIRSVULG	Cirsium vulgare (Savi) Ten.
CLAYCARO	Claytonia caroliniana Michaux
CLEMVIRG	Clematis virginiana L.

Species Code	<u>Scientific Name</u>
CLINBORE	Clintonia borealis (Aiton) Raf.
CONYCANA	Conyza canadensis (L.) Crong.
CORNALTE	Cornus alternifolia L.f.
CORNFOEM	Cornus foemina Miller ssp. racemosa (Lam.) J.S. Wilson
CORNRUGO	Cornus rugosa Lam.
CORNSTOL	Cornus stolonifera Michaux
CORYCORN	Corylus cornuta Marshall
CRAT_SP1	Crataegus species #1
CRAT_SP2	Crataegus species #2
CRAT_SP3	Crataegus species #3
CRAT_SP	Crataegus species
CRYPCANA	Cryptotaenia canadensis (L.) DC.
CYPRCALC	Cypripedium calceolus L.
CYSTBULB	Cystopteris bulbifera (L.) Bernh.
CYSTFRAG	Cystopteris fragilis (L.) Bernh.
CYSTTENU	Cystopteris tenuis (Michaux) Desv.
DACTGLOM	Dactylis glomerata L.
DANTSPIC	Danthonia spicata (L.) P. Beauv. ex Roemer & Schultes
DAUCCARO	Daucus carota L.
DESMGLUT	Desmodium glutinosum (Muhlenb. ex Willd.) DC ex Loudon
DIANARME	Dianthus armeria L.
DICECANA	Dicentra canadensis (Goldie) Walp.
DICECUCU	Dicentra cucultaria (L.) Bernh.
DIERLONI	Diervilla lonicera Miller
DIRCPALU	Dirca palustris L.
DRYOCART	Dryopteris carthusiana (Villars) H.P. Fuchs
DRYOCRIS	Dryopteris cristata (L.) Gray
DRYOINTE	Dryopteris intermedia (Muhlenb. ex Willd.) A.Gray
DRYOMARG	Dryopteris marginalis (L.) A. Gray
ECHILOBA	Echinocystis lobata (Michaux) Torrey & A.Gray
ELYMREPE	Elymus repens (L.) Gould
ELYMVIRG	Elymus virginicus L.
EPIFVIRG	Epifagus virginiana (L.) Barton
EPILCILI	Epilobium ciliatum Raf.
EPILCOLO	Epilobium coloratum Biehler
EPILLEPT	Epilobium leptophyllum Raf.
EPILPARV	Epilobium parviflorum Schreber
EPIPHELL	Epipactis helleborine (L.) Crantz
EQUIARVE	Equisetum arvense L.

Species Code	Scientific Name
EQUIHYEM	Equisetum hyemale L. ssp. affine (Engelm.) Stone
EQUILAEV	Equisetum laevigatum A. Braan
EQUISCIR	Equisetum scirpoides Michaux
ERIGANNU	Erigeron annuus (L.) Pers.
ERIGPHIL	Erigeron philadelphicus L. ssp. philadelphicus
ERIGSTRI	Erigeron strigosus Muhlenb. ex. Willd,
ERIG_SP	Erigeron species
ERYTAMER	Erythronium americanum Ker Gawler ssp. americanum
EUPAMACU	Eupatorium maculatum L.
EUPAPERF	Eupatorium perfoliatum L.
EUPARUGO	Eupatorium rugosum Houtt.
EUTHGRAM	Euthamia graminifolia (L.) Nutt.
FAGUGRAN	Fagus grandifolia Ehrh.
FESTARUN	Festuca arundinacea Schreber
FESTPRAT	Festuca pratensis Hudson
FESTRUBR	Festuca rubra L.
FESTSUBV	Festuca subverticillata (Pers.) E. Alexeev.
FRAGVESC	Fragaria vesca L. ssp. americana (Porter) Staudt
FRAGVIRG	Fragaria virginiana Miller
FRAXAMER	Fraxinus americana L.
FRAXNIGR	Fraxinus nigra Marshall
FRAXPENN	Fraxinus pennsylvanica Marshall
GALETETR	Galeopsis tetrahit L.
GALIAPAR	Galium aparine L.
GALIASPR	Galium asprellum Michaux
GALICIRC	Galium circaezans Michaux
GALILANC	Galium lanceolatum Torrey
GALIOBTU	Galium obtusum Bigelow
GALIPALU	Galium palustre L.
GALITRIF	Galium triflorum Michaux
GALI_SP	Galium species
GENTANDR	Gentiana andrewsii Griseb.
GERAMACU	Geranium maculatum L.
GERAROBE	Geranium robertianum L.
GEUMALEP	Geum allepicum Jacq.
GEUMLACI	Geum laciniatum Murray
GEUMRIVA	Geum rivale L.
GEUMURBA	Geum urbanum L.
GEUM_SP	Geum species

Species Code	Scientific Name
GLYCSTRI	Glyceria striata (Lam.) A. Hitch.
GYMNDRYO	Gymnocarpium dryopteris (L.) Newman ssp. dryopteris
HACKVIRG	Hackelia virginiana (L.) I.M. Johnston
HEPAACUT	Hepatica acutiloba DC.
HIERAURA	Hieracium aurantiacum L.
HIERCAES	Hieracium caespitosum Dumort, ssp. caespitosum
HYDRVIRG	Hydrophyllum virginianum L.
HYPEPERF	Hypericum perforatum L.
HYSTPATU	Hystrix patula Moench
IMPACAPE	Impatiens capensis Meerb.
IRISVERS	Iris versicolor L.
IRISSP	Iris species
JUNCTENU	Juncus tenuis Willd.
LACTCANA	Lactuca canadensis L.
LACTSERR	Lactuca serriola L.
LACT_SP	Lactuca species
LAPOCANA	Laportea canadensis (L.) Wedd.
LEERORYZ	Leersia oryzoides (L.) Sw.
LEERVIRG	Leersia virginica Willd.
LEONCARD	Leonurus cardiaca L. ssp. cardiaca
LIPALOES	Liparis loeselii (L.) Rich, ex Lindley
LOBEINFL	Lobelia inflata L.
LOBE_SP	Lobelia species
LONICANA	Lonicera canadensis Bartram
LONIDIOI	Lonicera dioica L.
LONIHIRS	Lonicera hirsuta Eaton
LONIOBLO	Lonicera oblongifolia (Goldie) Hook.
LSYINUMM	Lysimuchia nummularia L.
LYCOAMER	Lycopus americanus Muhlenb. ex Bartram
LYCOANNO	Lycopodium annotinum L.
LYCODEND	Lycopodium dendroideum Michaux
LYCOOBOB	Lycopodium obscurum L.var. obscurum
LYCOTRÍS	Lycopodium tristachyum Pursh
LYCOUNIF	Lycopus uniflorus Michaux
LYSICILI	Lysimachia ciliata L.
LYSITERR	Lysimachia terrestris (L.) Britton. Stems& Pogg.
MAIACANA	Maianthemum canadense Desť.
MAIARACE	Maianthemum racemosum (L.) Link ssp. racemosum
MAIASTEL	Maianthemum stellatum (L.) Link

<u>Species Code</u>	Scientific Name
MATTSTRU	Matteuccia struthiopteris (L.) Tod.
MEDILUPU	Medicago lupulina L.
MELIALBA	Melilotus alba Medikus
MELIOFFI	Melilotus officinalis (L.) Pallas
MENICANA	Menispermum canadense L.
MENTARVE	Menthu arvensis L.
MILIEFFU	Milium effusum L.
MITCREPE	Mitchella repens L.
MITEDIPH	Mitella diphylla L.
MONOHYPT	Monotropa hypopithys L.
MONOUNIF	Monotropa uniflora L.
MUHLFRON	Muhlenbergia frondosa (Poiret in Lam.) Fern.
MUHLMEXI	Muhlenbergia mexicana (L.) Trin.
ONOCSENS	Onoclea sensihilis L.
ONOPACAN	Onopordon acanthium L.
ORYZASPE	Oryzopsis asperifolia Michaux
OSMOCLAY	Osmorhiza claytonii (Michaux) C.B. Clarke
OSMUCLAY	Osmunda claytoniana L.
OSMUREGA	Osmunda regalis L.
OSTRYVIRG	Ostrya virginiana (Miller) K. Koch
OXALSTRI	Oxalis stricta L.
OXAL_SP	Oxalis species
PANAQUIN	Panax quinquefolium L.
PANIACUM	Panicum acuminatum Sw.
PANICAPI	Panicum capillare L.
PARTINSE	Parthenocissus inserta (A. Kerner) Fritsch
PHALARUN	Phalaris arundinacea L.
PHEGCONN	Phegopteris connectilis (Michaux) Watt
PHLEPRAT	Phleum pratense L.
PHLO_SP	Phlox species
PHRYLEPT	Phryma leptostachya L.
PICEGLAU	Picea glauca (Moench) Voss
PILEPUMI	Pilea pumila (L.) A. Gray
PINUSTRO	Pinus strobus L.
PLANLANC	Plantago lanceolata L.
PLANMAJO	Plantago major L.
PLANRUGE	Plantago rugelii Decne.
POA_ALSO	Poa ulsodes A. Gray
POA_COMP	Paa compressa L.

<u>Species Code</u>	<u>Scientific Name</u>
POA PALU	Poa palustris L.
POAPRAT	Poa pratensis L. ssp. pratensis
POA SALU	Poa saltuensis Fern. & Wieg.
POA_SP	Poa species
PODOPELT	Podophyllum peltatum L.
POLYACRO	Polystichum acrostichoides (Michaux.) Schott
POLYPAUC	Polygala paucifolia Willd.
POLYPERS	Polygonum persicaria L.
POLYPUBE	Polygonatum pubescens (Willd.) Pursh
POPUBALS	Populus halsamifera L.
POPUGRAN	Populus grandidentata Michaux
POPUTREM	Populus tremuloides Michaux
POTENORV	Potentilla norvegica L.
POTERECT	Potentilla recta L.
PREN_SP	Prenanthes species
PRUNSERO	Prunus serotina Ehrh.
PRUNVIRG	Prunus virginiana L. spp. virginiana
PRUNVULG	Prunella vulgaris L.
PTERAQUI	Pteridium aquilinum (L.) Kuhn
PYROELLI	Pyrola elliptica Nutt.
QUERALBA	Quercus alba L.
QUERMACR	Quercus macrocarpa Michaux
QUERRUBR	Quercus rubra L.
QUER_SP	Quercus species
RANUABOR	Ranunculus abortivus L.
RANUACRI	Ranunculus acris L.
RANUHISP	Ranunculus hispidus Michaux
RANURECU	Ranunculus recurvatus Poiret ex Lam.
RHAMALNI	<i>Rhamnus alnifolia</i> L'Her.
RHAMCATH	Rhamnus cathartica L.
RHUSRADI	Rhus radicans L. ssp. negundo (E. Greene) McNeill
RHUSTYPH	Rhus typhina L.
RIBEAMER	Ribes americanum Miller
RIBECYNO	Ribes cynoshati L.
RIBEGLAN	Ribes glandulosum Grauer
RIBELACU	Ribes lacustre (Pers.) Poiret
RIBERUBR	Ribes rubrum L.
RIBETRIS	Ribes triste Pall.
RIBE_827	Ribes specimen D827

Species Code	Scientific Name
ROBIPSEU	Robinia pseudoacacia L.
ROSABLAN	Rosa blanda Aiton
ROSAPALU	Rosa palustris Marshall
RUBUALLE	Rubus allegheniensis Porter
RUBUIDAE	Rubus idaeus L.
RUBUOCCI	Rubus occidentalis L.
RUBUODOR	Rubus odoratus L.
RUBUPUBE	Rubus pubescens Raf.
RUBU_\$40	Rubus specimen D840
RUDBHIRT	Rudbeckia hirta L.
RUMEORBI	Rumex orbiculatus A.Gray
SALIBEBB	Salix bebbiana Sarg.
SALIDISC	Salix discolor Muhlenb.
SALIERIO	Salix eriocephala Michaux
SALIPETI	Salix petiolaris Smith
SAMBCANA	Sambucus canadensis L.
SAMBRACE	Sambucus racemosa L. ssp. pubens (Michaux) House
SANGCANA	Sanquinaria canadensis L.
SANIMARI	Sanicula marilandica L.
SANITRIF	Sanicula trifoliata Bickn.
SANI_SP	Sanicula species
SCHIZPURP	Schizachne purpurascens (Torrey) Swallen ssp. purpurascens
SCIRATRO	Scirpus atrovirens Willd
SCUTLATE	Scutellaria lateriflora L.
SICYANGU	Sieyos angulatus L.
SILEVULG	Silene vulguris (Moench) Garcke
SMILHERB	Smilax herbacea L.
SMILHISP	Smilax hispida Muhlenb.
SOLADULC	Solanum dulcumera L.
SOLIALT	Solidago altissima L.
SOLICAES	Solidago caesia L.
SOLICANA	Solidago cunadensis L.
SOLIFLEX	Solidago flexicaulis L.
SOLIGIGA	Salidago gigantea Aiton
SOLIJUNC	Solidago juncea Aiton
SOLINEMO	Solidago nemoralis Aiton
SOLIRUGO	Solidago rugosa Aiton ssp. rugosa
SONCARVE	Sonchus arvensis L.
SONCOLER	Sanchus oleraceus L.

<u>Species Cade</u>	<u>Scientific Name</u>
SPHENINTE	Spenopholis intermedia (Rydb.) Rydb.
STELLONG	Stellaria longifolia Muhlenb. ex Willd.
STREROSE	Streptopus roseus Michaux
TARAOFFI	Taraxacum officinale G. Weber
TAXUCANA	Taxus canadensis Marshall
THALDIOI	Thalictrum dioicum L.
THALPUBE	Thalictrum pubescens Pursh
THELNOVE	Thelypteris noveboracensis (L.) Nieuwl.
THELPALU	Thelypteris palustris (Salisb.) Schott
THUJOCCI	Thuja occidentalis L
TIARCORD	Tiarella cordifolia L.
TILIAMER	Tilia americana L.
TRAGDUBI	Tragopagon dubius Scop.
TRIEBORE	Trientalis borealis Raf. ssp. borealis
TRIFREPE	Trifolium repens L.
TRILEREC	Trillium erectum 💭
TRILGRAN	Trillium grandiflorum (Michaux) Salisb.
TRIOAURA	Triosteum aurantiacum E. Bickn.
TSUGCANA	Tsuga canadensis (L.) Carriere
TUSSFARF	Tussilugo farfara L.
TYPHLATI	Typhu latifolia L
ULMUAMER	Ulmus americana L.
URTIDIGR	Urtica dioica L. ssp. graeilis
URTIDIOI	Urtica dioica L. ssp. dioica
UVULGRAN	Uvularia grandiflora Smith
VACCANGU	Vaccinium angustifolium Aiton
VERBHAST	Verbena hastata L.
VËRBTHAP	Verbascum thapsus L.
VERBURTI	Verbena urticifolia L.
VEROOFFI	Veronica officinalis L.
VEROSERP	Veronica serpyllifolia L.
VIBUACER	Viburnum acerfolium L.
VIBULENT	Viburnum lentago L.
VIBUOPUL	Viburnum opulus L.
VIBUTRI	Viburnum trilobum Marshall
VICICRAC	Vicia cracca L.
VIOLAFFI	Viola affinis Le Conte
VIOLBLAN	Viola hlanda Willd.
VIOLCANA	Viola canadensis L.

<u>Species Code</u>	Scientific Name
VIOLCUCU	Viola cucullata Aiton
VIOLLABR	<i>Viola labradorica</i> Schrank
VIOLPUBE	Viola puhescens Aiton
VIOLROST	Viola rostrata Pursh
VIOLSORO	Viola sororia Willd.
VITARIPA	Vita riparia Michaux
WALDFRAG	Waldsteinia fragarioides (Michaux) Tratt.

APPENDIX 4

DISTRIBUTION OF SPECIES BY SOIL PARENT MATERIAL, SOIL ORDER, SOIL MOISTURE AND CANOPY CLOSURE

<u>Legend</u>

Column Heading	Description	Unit
# Q	# quadrats	# quadrats
GF	Glacio-fluvial parent material	# quadrats
CT	Calcareous till parent material	# quadrats
LAC	Lacustrine parent material	# quadrats
CO	Calcareous outwash parent material	# quadrats
В	Brunisol soil order	# quadrats
GB	Gleyed Brunisol soil order	# quadrats
L	Luvisol soil order	# quadrats
GL.	Gleyed Luvisol soil order	# quadrats
G	Gleysolic soil order	# quadrats
SD	Seasonally dry depressions	# quadrats'
SM	Seasonally moist depressions	# quadrats!
SW	Seasonally wet depressions	# quadrats'
OC	Open canopy	# quadrats ²
CC	Closed canopy	# quadrats ²

Notes:

- 1. Frequency based on presence in dry, moist and wet microhabitats, respectively: frequency may not sum to number of quadrats recorded in column 2 (#Q) since occurrences on logs, stumps, raised root mats, lanes, ditches and regenerating fields were not included.
- 2. Frequency based on presence in open and closed microhabitats, respectively; frequency may not sum to number of quadrats recorded in column 2 (#Q) since species sometimes present in both open and closed microhabtats in the same quadrat.

	ſ	Soil Pa	irent Mai	erial		Sul Or	der				Soil Mc	oisture		Canopy	
SPECIES	Ş	: 10	CI	TAC	8	Ð	GB		GI.	э	CS S	NS	MN	ŏ	ដ
ABIEBALS	61	•	38	5	8	01	6	21	51	e.	47	÷	_	7	57
ACERNEOU	4	÷	J	0	0	_	0	0	•	_	m	_	0	c	Ŧ
ACERUBR	86	81	30	30	20	46	ŝ	2	₫	¢	65	r~	cı	20	33
ACERSACC	175	ę	96	9	2	68	2	56	28		152	2	ŝ	62	157
ACERSACN	5	0	2	ç	0	+	ŝ	-	\$	Ŷ	.	¥)	ý.	0	13
ACERSANI	7	0	6	0	_	-	v.	0	_	0	÷	_	0	•	r
ACERSPIC	Ψ	0	26	~	c	-1	m .	Ŧ	07 07	¥.	2	ŝ	~ 1	2 0	2
ACHEMILL	Ŷ	0	~ :	-	*	٦	Ģ	0	0	0	0	-1	_	-	►
ACTAPACH	29	ŝ	5	0	=	ç	0	5	ς.	0	ŝ	•	Ģ	7	5
ACTARUBR	55	ò	8	_	-	ŝ	\$	Ģ	•	6 1	17	m	0	-1	ā
ACTA_SP	Ç.	Ŷ	r.	_	12	2	Ś	É	9	-1	8	_	с ч	4	35
ADIAPEDA	50	0	0	•	ŝ	÷	×	~	÷	a	17	۳٦	e	€n	1
AGRIGRYP	đ	5	47	0	7	-1	•	~	_	ġ	ŝ	e.	0	6 1	F
AGROGIGA	1	¢	~ 1	0	×	Ŧ	ŝ	0	C	0	0	Ŧ	cı	ġ	~
AGROSTOL.	ę,	¢	6 1	0	-	_	6 0	÷	c	9	c	G		3	_
ALLUTRIC'	5	÷	÷	0	٤L	5	~ .	\$	ю,		원	D	Ŧ	7	
ALNUNUA.	-	Ð	_	0	•	Ð	0	=	-	¢	=	a	-	0	
AMBRARTE	τų	÷	ר ו	•	=	¢	רי ו	-	¢	Ð	0	ç	0	- 1	•
AMELARBO	_	÷	¢		=	¢	₽	÷	_	0	0	_	0	0	_
AMELINTE	ŝ	r,	_	0	CI	Ċ,	0	C1	0	•	-	•	₽	c i	-+
AMEL_SP	47	38	<u>م.</u>	0	Ŧ	닦	0	<i>\$</i>	0	0	5 17	•	⇒	r 1	÷
AMPHBRAC	=	0	m	e	×	.	.	÷	ŕι	0	r ~	-	-	~1	÷
ANEMCANA	J	c	0	0	- 7	ŝ	_	¢	•	•	¢	_	D	r h	cı
ANEMQUIN	_	ç	-	0	•	0		0	-	•	_	Ō	ç	0	_
ANEMVIKG	Ξ	Ð		_	ç	w.	-1	r,	_	c	ج ا	_	ç	80	Ŷ
ANTENEQ1.		0	e	0	_	0	0	_	0	0	¢	Ō	¢	_	•
APOCANDR	91	7	ç	G	ç	6	_	Ģ	0	0	<u> </u>	_	•	_	9
AQUICANA	ŝ	6	•5	c	~	đ	c	4	0	Ċ	=	0	•	~	잌
ARALNUD	٩C	<u>6</u>	61	ŝ	0	9	-+	•+	2	c	38	20	0	ę	5
													י !		:

d canopy closure.
soil moisture an
soil order,
rent material,
soil pa
species by
Distribution of
Appendix 4.

		Soil P	arent Ma	terial	:	Soll Q	der				Soil M	oisture		Cunop	 >
SPECIES	Ŷ	0F	CL	TVC	3	æ	GB	-	CI,	0	dS	SM	ΜS	S	5
ARCIMINU	6	•	200	-	0	. –	. +	m	1	0	v		0	-+	v.
ARISTRIP	97	Ģ	69		8	r,	ñ	37	F	ş	72	8	2	ព	83
ASARCANA	38	0	1	Ð	15	a	6	20	6	0	35	m	ы	Ŀ	Æ
ASCLINCA		•	•	٥	-1	0	_	_	D	•	0	•	Ċ	-	-
ASCLSYRI	16	0	8	Ģ	~	-	+	5	¢	0	×	•71	0	=	2
ASTROILJ	5	ŝ	ŝ	¢	01	2	r)	÷	¢	c	16 1	0	_	2	2
ASTECORD	n	~ ;	Ś	Ģ	9	4	-	¢	rı	•	Ξ	=	_	۲I	=
ASTEERIC	-	•	-	c	_	Ģ	0	_	¢	0	Ð	=	8	-	Ċ
ASTELANC	\$	0	۴ı	•	۴٩	_	с1 С1	-	_	0	¢	0	0	Ś	c
ASTELATE	99	•	14	Ŷ	16	9	52	×	2	-7	5 1 2	2	Q,	ŝ	58
ASTEMACR	2	ŝ	~	c	0	~	0	c	e	Ċ	7	0	•	_	9
ASTENOVE	8	0	0	_	20	÷	ŝ	_	1 7.	÷	9	6 1	~1	œ	ŕ
ASTEPUNI	2	-	9	_	×	Ţ	æ	a	-	_	0	ŝ	ć	2	2
ASTEUMBE	-	0	Q	0	_	0	-	•	=	G	0	0	_	0	-
ATHYFILL	ž	•	얶	6	91	~	17	5	20	e con	7	2	80	ī	57
A THY THEL.	9	0	c	÷	9	÷	÷	.	••	-	ve.	=	r.	÷	¢
BETUALLE	ŝ	•	됩	5	2	÷	ų	v .	17		¢	÷	m	-7	ñ
BETUPAPY	50	'n	2	-1	9	13	-7	×	- 1	~	5	_	÷	÷	74
BIDEFRON	Ŧ	e	≠	e	0	•	Ģ	_	ŕ	÷	ŝ	-1	m.	×	÷
BOERCYLI	20	0	10	0	ប	0	=	9	6	×	2	vî.	2	9	20
BOTRMATR	~1	_	c	G	_	_	Ċ	_	0	Ð	_	8	•	_	-
BOTRMULT	_	c	c	0	<u> </u>	0	0	_	0	•	_	8	=	¢	_
BOTRVIRG	÷	¢	<u>1</u>		2	8	c	위	.	¢	04		•	Ģ	4
BRACERFC	¢.,	-	-	G	'n	_	0	÷	0	•	ŝ	0	÷	۳ı	Ċ
BROMINER	m	¢	ы	¢	-	_	~1	0	c	=	Ð	ç	_		-
CALTPALU	_	٥		Ģ	0	ð	¢	0		•	ð	_	_	c	-
CALYSEPI	-	0	-	Đ	0	Ð	¢	_	c	e	_	÷	•	_	0
CARDACAN	_	¢	=	0		ð	Ð	_	¢	•	<u> </u>	ç	c	_	•
CARDDIPH	47	-	38	1	13	-	2	16	3 .	÷	37	<u>1</u>	m	Ξ	Ŧ
										:					•

, soft moisture and canopy closure.	
material, soil order	
Distribution of species by soil parent r	
Appendíx 4.]	

		с 			
	ene-moofmn-oofailea a a	20000-X-00000-44		00-07-07-00000 100500-00000 100500-000000 100500-0000000 100500-0000000 100500-0000000 100500-0000000 100500-000000 100500-00000 100500 100500-0000 1005000 1005000 1005000 100500000000	
	ne-maximmaxanan a x	00NF0000NA-%/	-4-4-6-6		
орососососососососососососососососососо	с-маю ^с ми-со ^д айиа.			- e ² r gão - 200 a 20	• • • • • • • • • • • • • • • • • • •
00000000000000000000000000000000000000		NF0	чдосоорински чдо-у-срдоуч	адияца-оросон-соди идопосо-сороса идопососо-соро идопососо-соро идопососо-сороса идопоса идопососо-сороса идопоса и идопоса и идопоса и идопоса и и и и и и и и и и и и и и и и и и и	е 4 мессососсееса m = o m & o = o o o o o o o o o o o o o o o o o
00000000000000000000000000000000000000	an Rostao-19m2aan	- u			4400000000000 2008000000000 2008000000000 200800000000
00000000000000000000000000000000000000	олкод со-ил 200 2-2-2008-хао		onys-o-ryshy	~~≈≈~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	Nococco Nococc
00000000000000000000000000000000000000			wys-s-rzoht		2000000000000000000000000000000000000
			8		Колинана тожоосоон ЯмХоонной Сосоосоо
0000Ro ⁰ 0mooo 000Ro ⁰ 0mooo 00Ro ⁰ 0mooo				е-честен 1998 еесее	Коба о о о о о о т о о о о о о о о Я о Ю о о о о о о о о о о о о о о о
000Ro ² amnoo 00Ro ² amnoo 00Ro ² amnoo			도 - 신 +		
00000000000000000000000000000000000000			ㅋㅋㅋㅋㅋㅋㅋ	Койина тожоос Поййоон	
o-11+ ² amNoo 11+ ² amNoo 11+ ² amNoo	00 ² 02400	•••••		Ko E 4 - F 8 8 0 0	Коби тожоо За 2000 сожоо
- Fit - Common - Fit - Common - Fit - Common - Fit - Common - Fit - Fit - Fit - Common - Fit	822840 92280		► 국 수 위 구	R 0 E 4 7 0 8 0 2 0 8 0	с 3 3 3 3 3 5 5 5 5 5 5 5 5 5 5 5 5 5
- 0 + 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	20840 8-800		포슈위 ·	R o E 7 0 8 2 2 2	8 22 8 2 2 0 8 2 1 0 0 7 7 0 17
0 7 0 0 0 0 0 0 0 - 0 0 0 0 0 7 - 0 0 0 0 0 7 - 0 0 0 0 0 1			6 년 4	ि स्ट	0 5 0 0
9 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	200		치 -+	27 P CS	N 7 12 0
	~ o		-7		
	0			++ 0 5	-+
0 6 0 0 0 6 0 0			9	0 0 6	0 0 9
600 600	¢		۲ı	1 0 2	0 I 0 2
0 °	0 5		2	16 0 12	0 16 0 12
< .	~ 7		7	0 0 7	0 0 0 7
	н У	•	'n	8 0 5	1 8 0 5
0 2 0	=	-	- -	T 0 0	0 0 0
8 19 24 4	5 IX	-	l F	J 14 66 16 1	10 JR 6 16 1
7 36 9]	7 17		26 5	58 3 26 5	33 58 3 26 5
6 3 0	-	¢	ۍ د	4 0 5 0	0 4 0 5 0
0 0 0	Ð	_	0	0 0 1	0 0 1 0
0 0 0	•				0 0 0
• -	~:		÷	2 0 3	0 2 0 3

	l parent material, soil order, soil muisture and canopy closure.
	oy sc
•	f species b
	Distribution of
	Appendix 4.

		Soil P.	arent Ma	terial		Sul O	rder				Soil N	loisture		Canop	 _
SPECIES	Q₽	GF	5	LAC	9	Ð	GB		CI.	σ	SD	WS	МS	S	ខ
CAREPSEU	-	0	. =	Ģ	-	Ċ	_	Ð	ð	0	0	0	 	=	_
CARERADI	64	0	9C	ب	91	œ	38	=	3 0	m	25	<u>7</u>	÷	=	4]
CARERETR	딘	=	÷	0	۴ŋ	¢	4 7)	0	~	Ŧ	¢	r)	۲.	é	9
CAREROSE	69	~ 7	48	7	ï	5	25	2	=	0	2	입	¢1	9	63
CARESPAR	ŝ	0	ম	_	c	4	•	Ģ	_	0	Ŧ	_	•	-	4
CARESTIP	17	¢	٩	_	r.	²	30	-	ŕ	÷	c	۶.	9	÷	6
CARETENE	<u>+</u>	¢	ΓI	c	<u>~</u>	+	+	•7	÷	0	C1	m	-1	r-	00
CARETRUS	¢	0	¢1	¢	4	0	بر ،	_	0	•	Ð	m	_	¢1	Ŧ
CAREVITE	6	•	•71	¢	7	_	3 0	0	0	0	Ċ	e.	·~,	á	ন
CAREWOOD	_	0	_	¢	•	0	_	=	8	=	_	•	0	0	-
CARE 719	-	0	0	¢	_	_	÷	0	ġ	e	a	•	0	0	-
CARE 868	2	0	cı	÷	•	a	r١	c	¢	•	0	0	0	ณ	¢
CARE 870		0	0	•	_	-	9	a	•	0	2	0	¢	_	Ċ
CARE 879	_	•	0	•	_		÷	0	0	0	¢	0	ç	_	•
CARE OV	ŝ	-	~	-	ci	¢	-	۳ı	ŕ1	0	¢1	-	_	_	÷
CARE SP	÷	=	v.	÷	_	_	÷	~	_	-	m	_	c	۳.	÷
CARPCARO	9F	-	5		91	9 0	_	n	ŝ	5	â	r.	¢1	n	Z,
CARYCORD	70	25	90	ŝ	¢	÷	_	9	¢	c	61	e.	c	ð	67
CAULTHAL.	57	-	40	0	91	-7	2	Я	9	¢	54	c 1	c 1	r~	ç,
CEANAMER	C 1	¢	2	0	¢	^1	0	•	0	¢	e 1	•	0	-	-
CELASCAN	æ	Ξ	6	_	민	ŝ	-	=	r 1	0	50	0	0	۲-	38
CERAFONT	2	0	4	c	9	-	Ŷ	0	0	•	~	-	_	~	J
CHIMUMBE	<u>-</u>		0	c	•	C1	•	0	0	•	പ	÷	0	9	2
CHRYLEUC	•	0	¢;	G	Ś	÷	r٩	0	¢	0	0	÷	÷	-	÷
CICUBULB	_	0	÷	Ð	_	a	-	e	0	0	0	∍	_	¢	_
CICUMACU	Ŧ	c	μ	â	rı	_	۴ı	÷	_	•	÷	÷	'n	~1	rı
CINNLA'II	Ξ	•	*	I	¢1	0	•	0	~	m	_	m	-7	¢1	Ġ
CIRCALPI	36	0	13	r-	<u>2</u>	0	~	×	fi	'n	2	F	m	20	ę.
CIRCLUTE	33	-	69	×	17	n	5	ē	3	se.	71	28	5	27	18

canopy closure.
nd o
moisture a
soil
order,
soil
material,
parent
soil
ş
species
0
oution
listril
Ξ.
Appendix 4

		Suit P	arent Mut	terial		Soil O	rder				Soit M	loisture		Canop	
SPECIE8	ş	0F	CL	LAC	8	8	GB	 ;	СL	ŋ	SD	SM	ΜS	ğ	g
CIRSARVE	9	0	Ŷ		÷	÷		: :	-	=	F	י רו	:0	6	÷
CIRSVULG	9	0	-7	0	r 1	~	0	~	0	_	r,	-	÷	Q	o
CLAYCARO	13	6 1	S	0	ó	ri	0	2	-	•	압	•	0	ςή	0
CLEMVIRG	엽	Ċ	•0	-	c 3	0	9	_	-+	_	-	ų	rī.	Ś	99
CI.INBORE	. .1	1	e	_	o	-	-	Ċ	_	0	-	_	0	0	~1
CONYCANA	5	ç	Ś	ð	¢	0	0	-7	÷	_	÷	÷	÷	۰°1	ð
CORNALTIE	8	n t	65	×	-	19	17	26	<u>75</u>	~	92 2	5	с 1	0	68
CORNFOEM	91	e	5	÷		<u>-</u>	m	ņ	-	0	Ŧ	~	÷	۲I	<u>+</u>
CORNRUGO	દ	6	16 1	0	4	6	5	_	_		38	^ 1	-+	¢	22
CORNSTOI.	22	∍	<u>-</u>	_	••	ŝ	21	o	Ŧ	_	r.	~		2	?
CORYCORN	4 2	£	ŝ	ç	7	Ē	¢	=	~	¢	4		•	7	÷
CRAT_SPI	m.	-	•	ç	0	~	¢	0	a	¢	ŝ	ç	•	0	m
CRAT_SP2	_	_	-	ç	=	-	¢	•	a	¢	-	¢	•	-	-
CRAT_SP3	_	_	•	¢	=	_	¢	0	¢	÷	-	¢	•	0	-
CRAT SP	29	5	٢	ç	£	2	ন	÷	r,	÷	8	<u> </u>		¢	ŝ
CRYPCANA	۴ı	Ð	ث ا	•	Ð	0	¢	_	_	¢	r I	÷	÷	rı	_
CYPRCALC	-1	÷	-	•	_	Ģ	21	•	¢	÷	÷	ŕ,	÷	đ	r)
CYSTBULB B	56	¢	ŝ	•	Ê		16	•	Ξ	ŕ١	12	<u>~</u>	r-	5	15
CYSTFRAG	т	c		0	-	Ģ	0	^ 1	C 1	0	rn	0	•	¢	⊸
CYSTTENU	~ i	•	<u>~</u> 1	9	0	→	-	0	0	0	~	0	÷	-	_
DACTGLOM	Ŷ	0	ŝ	0	_	¢1	2	-1	0	•	m	-	¢	7	÷
DANTSPIC	न	Ċ	r.	=	_	¢1	_	-	0	•	m	0	ð	~1	~
DAUCCARO	s	Ċ	r.	0		ר ו	en.	_	0	÷	0	÷	a	ŝ	-
TO, IDESMOLUT	~	0	_	0	\$	_	0	9	0	÷	ç	¢	a	~ 1	r~
DIANARME	_	0	0	0	_	_	0	¢	•	¢	ç	÷	ç	-	0
DICECANA	8	Ð	0	0	÷	•	0	<u>*</u>	0	D	8	÷	Ş	9	13
DICECUCU	Ŧ	•	m	¢	-	⇒	e	-+	c	¢	-+	₽	¢	0	-+
DIERLONI	2	ç	Ŷ	c	e	6	rı	-	G	c	5	÷	¢	_	=
DIRCPALU	9	÷	-	¢	÷¢	_	ġ	÷C	0	0	5	Ð	•	Ŧ	Ŷ
															:

413

. .

		Soil P	arent Ma	terial		Soil O	ruler			:	Soil M	loisture		Cartop	ž
SPECIES	Ċ,	GF	СT	LAC	9	Ħ	GB	<u>_</u>	GL	0	SD	MX	MS	ŏ	ਇ
DRYOCART	107		60	9	X	13	3	37	ន	: 		11		14	92
DRYOCRIS	8	0	91	-7	m	¢	9	сI	엽	•	9	6	C I	v.	Ś
DRYONTE	5	=	37	20	91	17	_	72	67	m	51	입	-	1	6l
DRYOMARG	5	-	되	20	r-i	r.	n	r I	ដ	÷	Ē	입	_	oc	32
ECHILOBA	_	0	_	0	0	0	0	ð	0	_	•	÷	¢	¢	-
ELYMREPE	~	0	_	0		•	_	_	0	e	¢	_	¢	م	-
ELYMVIRG	2	0	9	0	m.	0	9	Ċ	_	-1	•	s.	- 1	-+	~
EPIFVIRG	Ŧ	•	Ŧ	0	-	-	=	Ŧ	-	•	4	•	¢	-	m
EPILCH	ŝ	0	•	Ð	ŝ	2	* "1	Ċ	0	•	•	c 1	-	-	4 71
13P1.COLO	0 M	0	20	⇒	ô	έı	đ		12	J	w.	=	Ś	16	1
TY LEAD AND THE PARTY AND THE	Ŧ	Ċ	=	=	ъ	ς.	el.	Ċ	=	=	0	_	¢	-	J
EPILPARV	1- -	¢	v.	~1	¢	=	_	c	•	_	•		_	w.	L 1
EPIPIHSLI.	130	2	۲,	×	5	53	t,	29	17	Ŧ	100	22	-1	2	117
EQUIARVE	ž	¢	5	÷	01	%	17	÷	Ŧ	-	17	=	9	51	Ē
ROUTIVEM	5	r,	÷	¢	-	ç	¢	-	Ð	0	¢	8	•	-	¢
EQUILARY	۲J	÷	r 1	÷	•	÷	- 1	ŧ	÷	3		_	•	_	~ 1
EQUISCIR	-7	=	┯	Ð	•	_	_	ŕ١	¢	0	•Դ	¢	•	э	7
ERIGANNI	2	=	6	8	_	Ċ	1		0	0	· ~ 5	~	ζ.	ç	7
ERIGPHIL	÷	•	5	¢	6	7	8	÷	~1	0	ñ	¢	٢Ì	2	50 20
ERIGSTRI		•	2	0	ó	~	~	~1	0	9	0	-	0	ç	÷
ERIG_SP	cı	•	2	0	0	0	L 1	0	0	0	0	-	0	-	-
ERYTAMER	107	1	69	0	77	곱	07	Ŧ		0	86	61	ŝ	<u>•</u>	\$
FUPAMACU	2	0	÷	_	Ŷ	5	*	-	rt)	0		~	÷.	Ξ	ţ.
EUPAPERF	~	e		0	'n	ς Γ	÷	-	0	0	-	-1	0	ŝ	÷,
EUPARUGO	5	¢	=	0	2	•	겁	2	7	0	=	~	Ś	÷	2
EUTHGRAM	ç	÷		0	Ś	Ŧ	-	_	•	•	_	0	¢	Ŷ	_
FAGUGRAN	67	2	37	~1	ñ	2	1	R	۲	•	57	Ŧ	•	÷	59
FESTARUN	-	Ð	0	0	-	-	0	0	0	0	ç	c	¢	-	a
FESTPRAT	c1	0	1	•	¢	•	сı	÷	•	÷	•	•	¢	¢ι	c

Appendix 4. Distribution of species by soil parent material, soil order, soil moisture and canopy closure.

+ + +

:

:

:

moisture and canopy closure.
soil
order,
soil
material,
parent
soil
s by
specie
ē,
ution
trib
Dis
Ť
Appendix

SPECIES RQ GF CT LAC CO B GI I GI G SN SN SN CC C FESTRUBR 2 0 0 2 1 1 7 5 3 0 0 2 0 0 2 1 3 2 0 </th <th>SPECIES #Q GF CT LAC CO B GB L GB FESTRUBK 2 0 0 0 2 2 0</th> <th>5.anoxo72acaca->%cca7.</th> <th>Q の 2 2 + 2 2 2 0 0 = 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0</th> <th>×</th> <th>8 08888644-mmm4=-8-va</th>	SPECIES #Q GF CT LAC CO B GB L GB FESTRUBK 2 0 0 0 2 2 0	5.anoxo72acaca->%cca7.	Q の 2 2 + 2 2 2 0 0 = 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0 - 0	×	8 08888644-mmm4=-8-va
FESTRUBR 2 0<	FESTRUBR 2 0 0 2 2 0 0 2 2 0<		ong12+285500=>∞2		0%8888899995555599
FESTSUBV 20 2 19 0 9 14 7 6 3 0 28 1 5 3 FMAXNER 32 23 49 5 1 4 7 6 3 4 7 11 3 21 4 23 4 7 11 3 21 4 23 1 9 23 1 9 23 1 3 24 4 1 1 1 3 21 4 23 3 1 9 3 1 9 3 1 9 3 1 9 3 1 9 3 1 9 3 1 <	FESTSUBV 30 2 19 0 9 14 7 6 0 7 10 2 6 0 FRAGVESC 28 2 25 0 1 4 12 6 0 FRAXMER 25 23 49 3 20 7 10 2 8 5 FRAXMER 25 0 34 7 11 3 21 4 7 6 3 FRAXMER 25 0 34 7 11 3 20 47 5 12 5 5 6 3 FRAXMER 52 0 34 7 11 3 20 47 5 12 5 5 13 6 5 5 13 6 5 5 17 5 12 5 17 5 17 5 17 5 17 5 17 5 17 5 17 5 16 6 17 10 6 17 10<	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	813+995500mmのm3-r		88888440-mmm41-8-va
FMAGVESC 28 2 25 0 1 4 12 5 6 1 22 3 0 9 25 FRAXMER 53 2 10 6 10 7 10 2 8 1 14 8 8 8 1 14 8 8 8 8 1 14 8 8 8 1 14 8 8 8 1 14 8 8 8 1 14 8 8 1	FKAGVESC 28 2 25 0 1 4 12 5 FKAGVIKG 28 2 10 6 10 7 10 2 6 FRAXMER 95 23 49 3 20 47 9 33 6 FRAXMER 95 28 2 10 6 10 7 10 2 8 FRAXMER 95 16 5 24 7 11 3 21 4 21 4 22 8 6 17 10 2 11 3 21 4 10 21 4 10 1 10 1 21 4 21 21 4 21 21 4 21 21 4 21 21 4 21 21 4 21 21 4 21 21 4 21 21 4 21 21 4 21 21 4 21 21 4 21 21 4 21 0	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	2 + 2 2 5 0 0 = 0 m 2 - 1 m = 2 2 5 5 0 - 0 - 0 - 0 0 m 0		2894444-844444444
FFAAAMER 28 1	FKAGVIKG 28 2 10 6 10 7 10 2 8 FRAXAMER 95 23 49 3 20 47 9 3 8 FRAXAMER 95 23 49 3 20 47 9 3 8 FRAXNIGR 52 0 34 7 11 3 21 4 3 6 FRAXNIGR 52 0 34 7 11 3 21 4 2 4 4 1 6 1 4 3 2 4 3 5 1 4 2 1 4 2 1 4 5 1 4 5 1 4 3 6 6 1 1 2 4 2 1 2 1 4 2 1 1 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2	**************************************			2994422-89251525-23
FRAXMER 95 23 49 3 20 47 9 33 6 1 86 5 1 18 8 FRAXER 52 0 34 7 11 3 21 4 21 3 19 20 7 18 4 FRAXPENN 51 0 30 5 16 5 24 7 0 2 7 20 6 7 18 4 GALEFFER 3 0 2 0 1 0 1 2 1 3 19 20 7 18 4 GALEFFER 3 0 2 0 1 0 1 2 0 1 2 1 3 19 20 7 18 4 GALEFFER 3 0 2 0 1 0 1 0 1 2 0 1 1 0 0 2 1 3 0 GALEFFER 3 0 2 0 1 0 1 0 1 1 2 0 0 2 1 0 2 1 3 0 GALEFFER 3 0 2 0 1 0 1 0 1 1 0 1 3 0 0 1 3 0 GALEFFER 3 0 2 0 1 0 0 1 1 0 1 3 0 0 1 1 3 0 0 1 3 0 GALEFFER 3 0 2 0 1 0 0 1 1 0 1 1 0 1 3 0 0 1 3 0 0 1 3 0 GALEFFER 3 0 2 0 1 0 0 1 1 0 1 1 0 1 1 0 1 0 1 3 0 0 1 1 3 0 GALEFFER 1 1 0 1 0 0 1 1 0 1 1 0 1 1 0 1 1 0 1 1 0 1 0 0 1 0 0 0 1 0 0 GALEFFER 3 0 0 2 0 0 1 1 0 0 1 1 0 1 1 0 1 1 0 1 0	FRAXAMER 95 23 49 3 20 47 9 33 6 FRAXNER 52 0 34 7 11 3 21 4 3 6 FRAXNER 52 0 34 7 11 3 21 4 3 6 FRAXPENN 51 0 30 5 16 5 24 5 17 9 33 6 GALESPEC 2 0 1 0 1 0 1 23 24 5 17 9 33 6 7 1 3 21 4 21 4 21 4 21 4 21 4 21 4 21 4 21 4 21 21 4 21 21 4 21 4 21 4 21 21 4 21 21 4 21 21 4 21 21 4 21 21 4 21 21 21 21 21	∞n====================================	80000000000000000000000000000000000000		<u>8440-8004</u> =78-03
$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	FRAXNIGR 52 0 34 7 11 3 21 4 2 FRAXPENN 51 0 30 5 16 5 24 5 24 5 24 5 24 5 24 5 24 5 24 5 24 5 24 5 24 5 24 5 24 5 24 5 24 5 24 5 24 5 24 5 26 0	n	응원a-e-eeee	►×00000040	<u>844-mww</u> 42-8-w2
FIAXZPENN 51 0 30 5 16 3 24 5 17 0 27 23 5 16 4 GALERER 3 0 2 0 1 0 1 0 27 23 5 16 4 GALERER 3 0 2 0 1 0 1 0 27 23 5 16 4 GALLASPR 3 0 2 0 1 0 1 0 1 1 0 1 1 0 1 0 1 3 0 0 1 3 0 0 1 3 0 0 1 0 0 1 0 0 1 3 3 0 0 1 3 3 0 0 1 1 1 1 1 1 1 1 1 1 1 1 1	FRAXPENN 51 0 30 5 16 5 24 5 17 9 GALESPEC 2 0 1 0 1 0 1 2 9 1 24 5 17 9 GALESPEC 2 0 1 0 1 0 1 2 0 9 24 5 17 9 GALESPEC 2 0 1 0 1 0 1 2 0 9 6 2 0 1 24 2 0 <t< td=""><td></td><td>200<u>-0-00</u>00</td><td>×00000040</td><td><u>40-80041-8-0</u>3</td></t<>		200 <u>-0-00</u> 00	×00000040	<u>40-80041-8-0</u> 3
GALIGSPEC 2 0 1 0 1 0 1 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 2 0 0 1 0 0 1 0 0 1 0 0 1 0 0 1 0 0 1 0 0 1 0 0 1 1 0 0 1 1 0 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	000000-0%c007	0-0-00m0	000000#0	0-mmnt <u>t</u> s
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	GALETETR 3 0 2 0 1 0 1 2 0 GALIAPAR 11 0 3 0 2 0 1 2 0 0 GALIAPAR 11 0 3 0 2 0 1 2 0 0 GALIASPR 3 0 2 0 4 0 1 2 0 8 GALIANC 15 11 4 0 0 4 0 4 0 0 4 0 <	eeeee-e%eee% eereem-eeene		0000040	- œ m m = = = = = = = = = = = = = = = = =
GALIAPAR II 0 3 0 8 0 1 0 4 8 GALIARR 3 0 2 0 1 0 1 0 1 0 1 3 0 3 0 0 1 3 3 4 1 1 3 3 4 1 3 3 4 1 1 3 3 4 1 1 3 3 4 1 3 3 4 1	GALIAPAR 11 0 3 0 8 3 0 8 GALIASPR 3 0 2 0 1 0 8 3 0 GALIASPR 3 0 2 0 1 0 1 0 8 GALIASPR 3 0 2 0 1 0 1 0 8 GALIANC 15 11 4 0 0 4 0 0 GALLANC 15 11 4 0 0 4 0 0 GALLIANC 15 0 3 0 1 0 0 0 GALLINE 10 3 0 1 0 0 0 0 GALL_SP 1 0 0 1 0 0 1 0 GALL_SP 1 0 0 3 3 4 0 0 GALL_SP 1 0 0 3 3 4 0 0	0000-070007 00000-0000			®mn±=-8-n≘
GALIASPR 3 0 2 0 1 0 1 0 2 0 1 0 0 3 0 1 0 0 1 3 0 1 0 0 1 3 0 1 1 1 3 0 1 1 1 3 0 1 3 0 0 1 3 0 1 3 4 10 0 1 3 4 10 0 1 3 4 10 0 1 3 4 1 0 0 1 3 4 10 1 3 4 10 1 3 4 10 1 3 1 1 3 1	GALIASPR 3 0 2 0 1 0 1 0 GALIANC 15 11 4 0 0 4 0 0 4 0 GALILANC 15 11 4 0 0 4 0 0 4 0 GALILANC 15 11 4 0 0 15 0 0 0 GALILANC 15 11 4 0 0 15 0 0 0 GALLIANC 15 0 3 0 1 1 0 0 0 GALLIANC 15 0 0 0 0 0 0 0 0 GALLINE 10 1 59 5 30 36 19 24 25 GALL_SP 1 0 0 3 1 1 0 0 GALL_SP 1 0 0 3 3 4 0 0	000->%cco7;	- 0 0 n 0	000*0	m n i I - 8 - 10 I
DALICIRC 4 0 4 0 4 0 4 0 3 0 1 3 0 1 3 0 1 3 0 1 3 0 1 3 0 1 3 0 1 3 0 0 1 3 0 0 1<	GALICIKC 4 0 0 4 0 0 4 0 GALILANC 15 11 4 0 0 15 0 4 0 GALILANC 15 11 4 0 0 15 0 0 4 0 GALILANC 15 11 4 0 0 15 0 0 0 GALLIPALU 3 0 3 1 50 36 19 24 25 GALLINE 10 0 0 1 0 0 1 1 0 GALLINE 10 17 0 0 7 3 4 0 0 GALLINE 10 17 0 0 7 3 4 0 0	00-0700070 000-07000		² ni	~==~=~=
OALILANC 15 1 4 0 15 0 0 15 0 0 1 <td< td=""><td>GALLANC 15 11 4 0 0 15 0 0 0 GALLANC 15 11 4 0 0 15 0 0 0 0 GALLANC 15 0 7 0 8 4 10 0 1 GALLPALU 3 0 3 0 0 0 0 0 1 GALLSP 1 1 59 5 30 36 19 24 25 GALLSP 1 0 0 1 0 0 1 0 0 GENTANDR 7 0 0 0 7 3 4 0 0</td><td>o-oficoofi oom-coofic</td><td>۳ م</td><td>0 * 0</td><td>2 - 2 - v 2</td></td<>	GALLANC 15 11 4 0 0 15 0 0 0 GALLANC 15 11 4 0 0 15 0 0 0 0 GALLANC 15 0 7 0 8 4 10 0 1 GALLPALU 3 0 3 0 0 0 0 0 1 GALLSP 1 1 59 5 30 36 19 24 25 GALLSP 1 0 0 1 0 0 1 0 0 GENTANDR 7 0 0 0 7 3 4 0 0	o-oficoofi oom-coofic	۳ م	0 * 0	2 - 2 - v 2
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	GALIOBTU 15 0 7 0 8 4 10 0 1 1 0 0 1 1 0 0 0 0 0 0 0 0 0 0			* 0	I - 8 - M 9
UALIPALIU J U J U J U J U J U U U U U U U U U U	GALIPALU 3 0 3 0 0 0 0 0 0 0 3 GALIRIE 105 11 59 5 30 36 19 24 25 GALI_SP 1 0 0 0 1 0 1 0 0 1 0 0 0 GENTANDR 7 0 0 0 7 3 4 0 0 0 0		۔ •	~1 i	- 8 - v 9
JALLINIF 105 11 59 5 30 36 10 24 25 1 76 13 5 39 94 GALL_SP 1 0 0 1 1 0 0 1 5 9 9 GENTANDR 7 0 0 1 1 0 0 1 0 0 1 5 9 9 GENAMACU 12 0 1 1 0 0 1 1 0 0 1 0 0 1 1 0 0 1 1 0 0 1 0 0 1 0 0 1 0 0 1 0 0 1 0 0 1 0 0 1 0 0 1 0 0 1 1 0 0 0 0 1 1 0 0 1 1 0 0 0 0 0 0 0 0 0 0 0 0	CALITRIF 105 11 59 5 30 36 19 24 25 1 CALL_SP 1 0 0 0 1 0 0 1 0 0 1 0 0 GENTANDR 7 0 0 0 7 3 4 0 0 0 0 CEDEMAACTI 17 0 17 0 0 0 11 1 0 0 0	Neeeg.			8 - 2 9
GALL_SP 1 0 0 1 0 0 1 0 0 1 0 0 1 0 0 1 0 0 0 1 0 </td <td>GALL_SP 0 0 0 1 0 0 1 0 0 1 0 0 1 0 0 0 0 0 0</td> <td>0007.</td> <td>76 13</td> <td>62 9</td> <td>- ~ 9</td>	GALL_SP 0 0 0 1 0 0 1 0 0 1 0 0 1 0 0 0 0 0 0	0007.	76 13	62 9	- ~ 9
GENTANDR 7 0 0 0 7 3 4 0 0 0 2 1 4 5 5 5 7 1 4 5 5 5 15 4 5 5 1 5 5 1 1 2 4 5 5 1 5 5 1 1 2 4 5 5 1 5 5 1 1 2 4 5 5 1 5 5 1 1 2 4 5 5 1 5 5 1 1 2 4 5 5 1 5 5 1 1 2 4 5 2 8 1 5 5 1 5 5 1 5 1 1 2 4 5 2 8 1 5 5 1 5 1 1 2 4 5 2 8 1 5 5 1 5 1 1 2 4 5 2 8 1 5 5 1 5 1 1 1 1 5 0 0 0 0 0 0 1 1 2 4 5 2 8 1 5 5 1 5 1 1 2 4 5 1 1 1 2 4 5 1 1 2 4 5 1 1 2 4 5 1 1 1 2 4 5 1 1 2 4 5 1 1 1 2 4 5 1 1 1 2 4 5 1 1 1 2 4 5 1 1 1 2 4 5 1 1 1 2 4 5 1 1 1 2 4 5 1 1 1 2 4 5 1 1 1 2 4 5 1 1 1 2 4 5 1 1 1 2 4 5 1 1 1 2 4 5 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	GENTANDR 7 0 0 0 7 3 4 0 0 0 Generation 17 0 17 0 0 11 1 0 0 1		•	•	~ [_]
GRAMACU 12 0 12 0 12 0 12 0 12 0 12 0 12 0 12 0 12 0 12 0 11 1 0 0 1 0 0 0 0 1 0 0 1 0 0 1 0 0 1 0 1 5 0		9 71 9 9 10 9	-1		9
JIRAKOHE 53 0 35 8 10 1 24 5 28 15 41 GRIMALEP 1 0 1 0 0 1 0 0 1 5 28 15 5 15 GRIMALEP 1 0 1 0 0 1 0 0 1 5 5 15 41 GRIMALEP 1 0 1 1 5 0 7 0 1 5 1 7 5 1 7 5 1 7 5 1 7 5 1 7 5 1 7 5 1 7 1 7 1 5 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 <		50 v	0	0	Ż
GRUMALEP 1 0 1 0 0 1 0<	GERAROBE \$3 0 35 8 10 0 13 11 24 3		28 15	5 15	7
OEUMCANA 13 0 10 2 1 1 5 0 7 0 4 5 9 OEUMLACI 21 0 13 2 6 0 5 4 7 5 6 8 1 7 9 OEUMINACI 21 0 13 2 6 0 5 4 7 5 6 8 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 6 1 7 1 6 1 7 1 7 1 7 1 7 1 7 1 6 1 1 7 1 6 1	GRUMALEP 1 0 1 0 0 1 0 0 0 (0	0	0 0	-
GEUMLACI 21 0 13 2 6 0 5 4 7 5 6 8 1 7 GEUMRIVA 2 0 2 0 0 0 0 0 1 2 0 GEUMIRBA 2 0 2 0 0 0 0 0 1 2 0 2 GEUMIRBA 2 0 14 6 3 4 8 1 6 4 2 0 1 2 0 2 GEUM SP 23 0 14 6 3 4 8 1 6 4 6 9 5 7 1 GEUM SP 23 0 14 7 15 3 23 5 23 8 27 26 11 20 5 GI/YCSTR1 62 0 4 0 1 3 3 3 3 7 1 7 1 GI/YCSTR1 62 0 4 0 1 3 3 3 3 3 7 1 2 3 7 1 1 2 3 7	GEUMCANA 13 0 10 2 1 1 5 0 7 0	7 0	~	••, _	6
OBUMRIVA 2 0 2 0 0 1 2 0 2 OBUMRBA 2 0 2 0 0 0 0 1 2 0 2 ORUMIRBA 2 0 2 0 0 0 0 0 1 2 0 2 ORUMIRBA 2 0 14 6 3 4 8 1 6 4 6 9 5 7 1 ORIVESTRI 62 0 40 7 15 3 23 5 23 8 27 26 11 20 5 OYNDRYO 24 0 9 11 0 15 3 23 5 23 8 27 26 11 20 5 5 5 54 1 2 3 5	GEUMLACI 21 0 13 2 6 0 5 4 7 5	7 5 6	80	- 1	16
GEUMURBA 2 0 2 0 0 0 0 0 2 0 2 0 2 0 1 1 GEUM SP 23 0 14 6 3 4 8 1 6 4 6 9 3 7 1 GLYCSTRI 62 0 40 7 15 3 23 5 23 8 27 26 11 20 5 GYMNDRYO 24 0 9 5 10 1 3 9 11 0 15 4 1 2 2 2 HACKVIRG 4 0 4 0 0 0 0 4 0 0 4 0 0 4 0 0 4 0	CJEUMRIVA 2 0 2 0 0 0 0 0 2 (0	_	0 0	ς.
GEUM SP 23 0 14 6 3 4 8 1 6 4 6 9 3 7 17 GLYCSTR1 62 0 40 7 15 3 23 5 23 8 27 26 11 20 50 QYMNDRYO 24 0 9 5 10 1 3 9 11 0 15 4 1 2 2 27 HACKVIRG 4 0 4 0 0 0 0 4 0 0 7 4 0 0 4 0	GEUMURBA 2 0 2 0 0 0 0 0 2 (•	0	1 0	-
GLYCSTR1 62 0 40 7 15 3 23 5 23 8 27 26 11 20 50 GYMNDRYO 24 0 9 5 10 1 3 9 11 0 15 4 1 2 2 HACKVIRG 4 0 4 0 0 0 0 4 0 0 4 0 0 4 0	GEUM SP 23 0 14 6 3 4 8 1 6 3	ч 9	÷	5 7	<u>۲</u>
GYMNDRYO 24 0 9 5 10 1 3 9 11 0 15 4 1 2 27 HACKVIRG 4 0 4 0 0 0 0 4 0 0 4 0 8 4 0	GLYCSTR1 62 0 40 7 15 3 23 5 23 8	23 &	21 26	11 20	50
IACKVIRG 4 0 4 0 0 0 0 4 0 0 4 0 0 4 0	0YMNDRYO 24 0 9 5 10 1 3 9 11 (0 11	12 T	-	23
	HACKVIRG 4 0 4 9 0 0 4 0 0 4	• •	-	† 0	

ntion of species by soil parent material, soil order, soil moisture and canopy closure.
Distríbutio
Appendix 4.

		Soil P:	areni Mai	leriaț		Soil Or	der		-	:	Soil Mo	oisture		Canopy	
SPECIES	ð,	5	5	LAC	8	£	68	. :	GL.	9	SD	NS	MS	8	8
IIEPAACUT	े 	۴ŋ	12	0	9	6	m	: c	c	. 0	2	_	0	4	ล
HIERAURA		0	-	0	0	0	_	¢	•	•	-	•	•	0	_
HIERCAES	60	0	ą	7	=	91	ង	6	<u>1</u>	c	93	0	ri	5	Ŧ
HYDRVIRG	∓ ∵	•1	Ŷ	0	m	ŝ	-	v.	m	0	=	0	_	_	m
аяядэдли	=	e	-1	¢	÷	¢	-7	-1	=	¢	-		0		ò
HYSTPATU	ŝ	e	C 1	¢	_	ò	- 1	_	¢	a	_	-1	_	6 1	ei
IMPACAPE	60	0	38	-	25	~1	=	5	26	9	27	Ξ	15	5	2
INUGHELI	cl	0		¢	_	ð	-1	0	÷	0	0	_	_	-1	•
IRISVERS	1	0	•	0		0	-	0	•	0	0	•	•	_	0
IRIS SP	ŝ	0		0	J	_	4	0	0	0	•	0	רי ז	rı	ŝ
JUNCTENU	m	0	ŝ	¢	•	÷	r)	_	0	0	_	•	=	~	0
LACTUANA	٦	•	۳I	¢	2	÷	0		0	0	_	•	0	Ŧ	_
LACTSERR	¥1	0	m	¢	0	•	- i	r:	÷	_	r.	÷	0	Ś	•
LACT SP	CI	¢	rı	0	¢	¢	_	e	-	ç	3	¢	0	-	_
LAPOCANA	2	0	2	0	"	_	5	¢	ŕ	0	6	~	9 C	r ~	2
LEERORYZ	~	8	ves.	÷	r:	ŕ1	÷	_	٦	0	÷	-1	^ 1	ç	_
LEERVIRG	_	ò	÷	-	0	÷	÷	÷	_	¢	3	0	5	-	•
LEONCARD	-	a	4	=	Ģ	e	Ð	4	¢	•	. ,	•	¢	rt.	_
LIPALOES	Ţ	ç	0	•	Ŧ	_	~	0	o	0	ç	_	-		-
LOBEINFL	_	0	_	=	ç	=	0	_	ç	0	-	Ð	•	_	ċ
LOBE SP	_	0	_	0	¢	•	_	•	0	0	0	•	•	_	c
LONICANA	ŝ	ŝ	=	cı	0	36	0	÷	~1	•	Z	0	÷	e n	8
IORINO 1	ī	9	99	0	0	n	0	_	÷	•	7	0	•	_	<u></u>
1,0NIHIRS	01	Ê	¢	0	•	2	9	0	•	•	=	0	=	c	2
1.YCOAMER	Ģ	÷	c 1	c	-+	-	~	÷	Ċ1	•	=		e	cı.	7
LYCOANNO	ŝ	۴ı	r,	0		÷	•	~1	e	÷	ŝ	0	-	0	er.
LYCODEND	2	-	-	¢	=	Ċ1	•	0	0	0	_	÷	=	Ċ	Ċ1
LYCOOBOB	7	Ŧ	0	¢	0	7	=	•	=	Ġ	-	c	=	¢	J
LYCOTRIS 11	-	¢	0	¢	-	0	=	_	c	¢	÷	0	-		•
•	•	•													

closure.
l canopy
sture and
oil mui:
order, s
, soil (
naterial
parent
y soil
pecies b
on of s
Distributi
<u>п</u>
Appendix 4

SPECIES		Soil P	arent Ma	terial		Soil O	Inder				Soil N	doisture		Canop	ኦ
	ç	Ð	Ē	JVT	2	n	69	<u></u>	СI,	5	CIS	SM	ΝS	9	8
LYCOUNIF	8	¢	12	•	\$	rt	: . ×	÷	•	\$	Ċ	-	5	5	=
I, YSICILI	Ŧ	0	Ŷ	•	×	ਜ	×	0	-1	0	0	w.	~	7	1
L YSINUMM	0	÷	J	÷	~	e	ŝ	0	ज	0	-	r∽	ŝ	-	×
LYSTIERR	_	¢	=	c		-	a	•	0	0	0	c	0		-
MALACANA	001	36	99 9	\$		60	8	Ĩ	2	_	107	01	¢	5	5
MALARACE	16	Ĩ	45	r 1	13	52	=	22	ŝ	_	22	F	_	ġ	86
VALASTEL.	10	N	Ţ	0	÷	v٩.	~1	m	Ċ	=	×	-	0	-	6
MATTSTRU	18	0	*	_	ç	÷,	6 2	m	Ś	0	Ξ	ŝ	÷	Ś	91
MEDILUPU	=	0	Ģ	¢	'n	2	-	⇒	_	•	~	0	0	20	9
MELIALIHA	4	0	-	0	ŝ	m	•	_	•	0	_	0	•	-7	c 1
MELIOFFI	-1	÷	÷	¢	r.	e.	•	0	¢	0	9	0	•	r.	_
MENICANA	r 1	•	a	0	-	0	0	e.	0	0	~1	0	9	_	2
MENTARVE	12	ci	ŝ	0	ŝ	~	×	c	÷	-	_	~	^ 1	s	~
мпленчи	7	•	m	0	Ŧ	c	•	Ś	r)	c	►	c	e	-	•
MITCREPH	7	17	à	0	÷	Ľ-	•	Ċ	÷	÷	17	a	9	_	9
MICELIA	1	2	¢	=	~	≘	÷	۴ı	Ξ	Π	3	~	÷	7	2
MONAFIST		-	ç	•	0	_	Ċ	a	=	a	-	Ċ	0	3	_
MONOHYPO	-		•	•	a	را م	Ċ	Þ	•	0	ţ,	¢	•	¢	2
MONOUNIF	15	¢	9	0	o	₫	ç	-	÷	-	15	¢	0	¢	15
MUHLFRON	_	ç	0	0	_	_	•	•	0	Ģ	9	•	¢	-	-
MURLMEXT	-	¢	r)	c	w.	~ .	Ŧ	-	÷	0	0	_	_	ŝ	m
VEPECATA	_	0	_	₫	•	÷	•	-	÷	Ð	-	0	0	-	e
ONOCSENS	ŧ	0	т гі	÷.	:	4	2	•••	린	6	<u>1</u>	=	2	2	30
DNOPACAN	m	0	~	0	0	Ċ	=	÷	0	0	-	0	¢	en.	0
JKYZASPE	51	33	2	0	÷	ŧ	÷	ŝ	a	0	51	0	0	÷	\$†
DSMOCLAY	36	Ţ	2	0	2	÷	÷	;]	ŝ	Ð	¥	0	_	0	38
DSMUCLAY	ر ،	m	-1	0	0	m	•		_	c	•†	-	0	•	Ś
DSMUREGA		0	_	Ģ	0	÷	÷	÷	_	0	¢	•	-	0	-
DSTRVIRG	82	<u>ا</u> ب	쉮	_	ž	Ľ	0	R	Ŧ	0	77	r:	•	¢	92

41 J

r closure.
: and canop)
moistur
soil
order,
soil
material,
parent
lios
s by
specie
, o
uìor.
tribi
Dis
Ť
Appendix

		Soil P	arent Ma	terial		Soil Or	der				Soil M	oisture		Canop)	
SPECIES	Ŷ	<u>G</u>	5	LAC	9	Ð	6B	:	ы С	С	GS	SM	ΜS	ğ	ដ
OXALSTRI	÷	c	•	0	m	0	¢	~	'a	. a	-	•	a	 	 רי
OXAL_SP	_	0	_	0	0	0	¢	-	G	ċ		c	0	0	_
PANAQUIN	_	¢	0	0	_	Ģ	0	-	¢	ð	÷	0	c	•	_
PANIACUM	÷	c	G	=	m		•	2	¢	Ċ	c	c	¢	m	^c ı
PANICAPI	_	÷		=	÷	Ŧ	Ð	_	¢	Ċ	_	0	ç	-	0
PANI SP	r I	þ	6 1	0	0		0	_	¢	ð	_	0	¢	_	_
PARTINSE	60	Ċ	99	m	27	80	5	13	Ξ	r.	E	15	91	5	\$
PHALARUN	-	٩	0	0	m	m	•	0	¢	•	¢	0	Ģ	~ 1	c 1
PHEGCONN	-	•	0	•	_	-	Ð	0	_	0	-	0	0	÷	_
PHLEPRAT	v n	٩	C 1	0	m	~1	m	0	0	0	0	0	¢	ŝ	rı
PHLO_SP	cı	0	0	0	~1	¢	e	~1	0	đ	¢.	0	0	_	-
PILRYLEPT	6 1	c	s	0	2	r n .	ŕι	Ξ	¢1	Ċ	21	_	0	20	엄
PICEGLAU	10	G	Ś	0	0	51	ŝ	ġ	¢	0	-	0	0	_	÷Ŧ
PILEPUME	25	0	2	0	÷	~	Ŷ	m.	•		Ŷ	ι η	ç	16	<u>+</u>
ONUSUNIA	13	-	9	•	2	Ś	en.	w.	¢	Ċ	면	_	0	_	
PLANLANC'	rı	0	⇒	0	C 1	14	÷	0	9	1	0	÷	3	-	_
PLANMAJO	=	0	ç	•	5	ا م	20	_	÷	8	_	-•		w.,	ų
PLANRUGE	<u>0</u>	0	r,	0	5	~1	9	2	•	a	_	-:	-	7	ŝ
POA_ALSO	16	6 0	••	c	-1	r.	<i>1</i> 7.	ų	ŝ	a	Ē	rı.	4	e.	Ŧ
POA_COMP	5	٩	Ξ	-	÷	Ē	w.	-;	ŕ)	8	믿	-1	e,	2	2
POA PALU	23	ċ	⊒	c	5	m	16	-	- 1	-	e	Ξ	e	Ξ	5
POA PRAT	27	ç	⊐	Ċ	=	Ը	÷	÷	0	¢	0	-1	r.	8	2
POA SALT	~	Ŀ1	_	÷	Ð	~ n	⇒	Þ	•	₽	~		0	_	ن ب
POA SP	•	ç	ς.	0	0	0	~1	0	•	¢		0	0	0	r-t
PODOPELT	Ś	¢	4	c	-	m	0	_	_	Ģ		•	0		÷.
POLYACRO	2	¢	~	0	~	_	6	C 1	_	Ð	6	_	c	•	≘
POLYPAUC	Ģ	ç	ç	e	•	÷n	Ċ	m	0	Ð	'n	÷	•	2	ज
POLYPERS	.,	•	Ţ	a	0	0	÷	c	-	÷	•	0	ا ت	4	0
POLYPUBE	8	0 1	÷3	a	0	60	ŝ	٦.	"	0	83	-1	-	2	39 39

		Soil P	burent Ma	terial		Soil C	rder				Soil N	doisture		Canop	2
SPECIES	ą	5	сI	LAC	9	~	CB	<u></u> ;	GI,	o	SD	SM	SW	ğ	8
POPUBALS	13	. -	20	•	5	4	2	: · _	॑ ᠇	m	-+	•	0	5	=
POPUGRAN	ŝ	÷	16	•	a	9	Ċ	~	0	0	20	0	0	0	ន
POPUTREM	30	9	=	e	6	16	\$	0	0	•	5	-1	_	ŕ	8
POTENORV		e	_	•	c	0	-	Ð	c	÷	0	Ð	0	-	0
POTERECT	٢	c	-	¢	9	ਜ	_	cI	c	0	o	•	¢	Ŷ	m
PREN SP	00	m	lá	a	-	91	~	0	_	0	<u>2</u>	-	0	Ŧ	<u>~</u>
PRUNSERO	80	24	%	~	15	7	۲-	5	Ś	¢	<u>66</u>	Ś	0	2	5
PRUNVIRG	133	ž	Ŧ	m	저	57	9	32	2	ŝ	101	2	4	입	011
PRUNVULG	1	÷	ų	Ċ	æ	4	10	e	0	0	ы	\$	-	×	÷
PTERAQUI	뎧	ŝ	5	•	ŝ	l f	0	Ξ	¢	¢	39	8	•	6	Ŧ
PYROELLI	20	ų	Ŷ	-	ŝ	2	-	ŝ	ы	ç	16	¢	-	_	8
QUERALBA	÷	Ģ	¢	0	0	ŝ	0	-+	0	0	80	•	e	_	¢
QUERMACR	<u>e</u>	Ð	r-	÷	s	Ð	ſ~	wî.	0	0	\$	ċ١	0	_	=
QUERRUBR	11	2	5 8	0	<u>e</u>	2	0	2	•	•	67	e	<u>-</u>	۲-	89
QUER SP	-	÷	-	÷	9	-	a	•	÷	•	_	•	0	=	-
RANUABOR	Ċ,	*	Ē	_	*	16	Ξ	51	Ξ	÷	ŧ	Ē	→	13	÷
KANUACRI	17	0	ន	_	2	£	9	-	_	-	2	÷	÷	₫	Я
RANUFIICA	_	0	-	¢	-	c	=	Ċ	_	Ċ	÷	_	_	÷	
RANURECU	61	0	÷	9	î	-	20	Ξ	17		60	81	~	ŝ	- The second se
RHAMALNI	_	a	_	•	0	_	⇒	D	D	Ð	-	Ð	÷	=	-
REAMCATE	26	, ,	58	-	13	ន	=	a	ŝ	r-	50	0	~I	2	61
RHUSRADI	17 T	<u>45</u>	5	-	.	5	₽	c 1	rı	÷	8	÷	6	۲-	유
RHUSTYPH	_	Þ	•	0	-	•	0	_	0	÷	=	Ð	ŧ		0
RUBEAMER	寸	¢	Ð,	•	ŝ	-	6	÷	-1	m	_	¥.	v,	-+	2
RIBECYNO	5	17	(3	.	인	2	2	5	₫	5 0	20	8	~1	Ż	28
RIBEGLAN	Ŧ	-	J	ð	¢	Ð	ç	•	Ŧ	∍	-	~	~	-	~ ~
RIBELACU	-	-	-	à	c	÷	•	0	-	0	a	-	ð	¢	-
RIBERUBR	ŝ	=	ŕ	a	-1	÷	m.	-	-	÷	e,		-	ri	m
RIBUTRIS	Ê	-T	€	ŝ	ר ו	9	Ξ	0	Ξ	~1	2	=	+	<u>2</u>	2
closure.															

d canopy															
oisture an															
soil me															
order,															
soil															
material,															
parent															
soil															
s by															
specie															
, G															
lio															
Distrib															
त्तं															
Appendix															

		d lig	arvel Ma	Terial		Soil O					N I N	oietures		Canon	.
19-19-5-19-5-1	¢.							-	5				1111		ξ
SPECIES	Ž.	5	-	TVC.	9	•	â	-:	.ici	;	ĺ.	MC	× :	5	
RIBE 827	_	ð	_	¢	0	Ģ	_	ċ	÷	8	0	0	_	¢	_
ROBIPSEU	61	ð	C1	Ð	81	~	rr.	Ξ	m	8	Ŧ	Ð	rı	7	9
ROSABLAN	÷	C1	0	0	÷	C 1	_	đ	÷	•	e.	_	ð	Ģ	~ 1
ROSAPALU	÷	0	¢	•	r,	_	- 1	8	0	¢	•	-	_	_	c1
RUBUALLE	8	ŝ	9	_	9	90	•	*	r-1	÷	÷	-	_	v 7:	9
RUBUIDAE	53	Ģ	æ	7	ñ	9	ŝ	16	07	Ŷ	8	ç	_	5	5
RUBUOCCI	16	þ	=	0	4 5	r)	m.	9	_	=	13	_	ר ו	6	0
RUBUODOR	Ξ	9	7	÷	•		÷	•		9	9	ŕ١	ç	r.	9
RUBUPUBE	47	¢	ŝ	7	Ē	Ś	<u>~</u>	د م	R	5	16	16	ŝ	16	₩E
RUBU 840	cı	0	CI	0	•	0	Ģ	0	c 1	Ð	Đ	c	¢	_	<u> </u>
RUDBILIRT	_	0	0	0	_	_	•	•	¢	•	¢	c	0	÷	-
RUMEORBI	_	0	_	0	=	a	-	0	¢	0	0	0	-	_	0
SALIBEBD	4	c	0	a	Ŧ	. .1	ci	0	¢	0	0	_	-	гі	7
SALIDISC	s	e	0	c	ç		C 1	•	0	•	Ð	_	•	v.,	m.
SALIERIO	a	•		¢	a	4	ŝ	-	=	Ċ	Ð	÷.		ų	5
SALIPETI	rı	=	Ð	¢	<u>^</u> 1	~	=	¢	÷	Ō	0	÷	=	- 1	÷
SAMBCANA	Ľ	=	~	0	3	¢	•	¢	9	_	~1	_	r i	᠇	ŝ
SAMBRAPU	40 F	÷	20	ς.	81	e.	_	R	80	0	Ħ	_	0	0	R
SANGCANA	=	ç	¢	0	v٦	9	0	r i	c 1	9	Ξ	•	=	-1	9
SANIMARI	9	0	~1	0	4	c)	c	4	÷	¢	Ŷ	0	0	2	÷
SANITRIF	-	0	•	0	_	÷	¢	_	0	¢	-	¢	c	c	-
SANI SP	_	ð	-	0	ç	_	þ	¢	0	e	_	¢	e	0	-
SCHIPURP	fi.	\$	50	=	Ŧ	5	1	7	0	=	8	-:+	c	ŝ	92
SCIRATRO	ŝ	0	¢Π	=	'n	-	4	0	÷	Ð	•		cı	m	~1
SCUTLATE SCUTLATE	Ξ	÷	9	C 1	'n	сı	ç	Þ	-1	_	e	m	۴ı	~	7
SICYANGU	_	÷	_	÷	0	÷	¢	•	0	_	¢	=	Ċ	_	•
SILEVULG	-1	0	-	⇒	-	r 1	¢	•	Ċ	=		Ð	Ċ	-	_
SISYMONT	~	0	9	÷	r .	۰ı	•	-	0	÷	÷	0	ð	ci	-1
SIUMSUAV	~I	0	۰t	÷	¢	0	-	=	e	~	0	0	_	⇒	ci

.

closure.
d canopy
noisture an
, soil n
order,
soil
naterial,
parent i
by soil
species
ion of
stribut
õ
× 4
Appendi

SWILHERE #0 GF LAC CO B GR L GL G SM SW OC CC SWILHERE 30 7 10 6 7 10 6 7 7 0 11	SPECIES #0 GF LAC CO B GR L GL G SM SW OC CC SMILHERB 20 7 10 0 1<	SPECIES SMILHERB SMILHERB		Soil P ₂	arent Mat	terial		Soil O	rder				Soil M	loisture		Canot	ž
SMILHER 50 72 10 0 8 40 0 10 0 46 0 7 1 1 1 28 20.1ALT SSMILHER 7 2 7 0 1 1 7 0 1 1 1 0 0 2 0 1 1 1 1 1 28 20.1ALT SSMILHER 7 2 7 0 1 1 1 7 0 1 1 1 0 0 1 1 1 1 1 1 1 1	SMLHERS 50 77 10 6 8 40 0 10 0 7 46 0 6 7 5 5 2 2 0 11 7 7 8 11 17 8 5 2 2 0 0 1 4 17 17 1 1 1 7 8 1 1 1 1 1 1 1 1 1 1 1 1	SMILHERB	Q#	GF	СТ	JYC	8	-	63	<u> </u>	сi.	5	SD	WS	SW	ğ	g
SMUHISP 2 0 0 0 0 2 0 1 </td <td>SMUHISP 2 0 0 0 2 0 0 1<!--</td--><td>SMILHISP</td><td>20</td><td>E</td><td>2</td><td> </td><td>×</td><td>- 07</td><td></td><td>2</td><td>. =</td><td>c</td><td>46</td><td>0</td><td>. 0</td><td>ų.</td><td>19</td></td>	SMUHISP 2 0 0 0 2 0 0 1 </td <td>SMILHISP</td> <td>20</td> <td>E</td> <td>2</td> <td> </td> <td>×</td> <td>- 07</td> <td></td> <td>2</td> <td>. =</td> <td>c</td> <td>46</td> <td>0</td> <td>. 0</td> <td>ų.</td> <td>19</td>	SMILHISP	20	E	2	 	×	- 07		2	. =	c	46	0	. 0	ų.	1 9
Sol.ADILC #9 0 33 7 9 0 13 1 17 8 16 11 4 16 35 Sol.ACMA 67 2 40 7 18 10 14 17 17 3 41 9 7 7 3 90 Sol.CAMA 67 2 40 7 18 10 14 17 17 3 41 9 7 7 3 90 Sol.CAMA 67 2 40 7 18 10 14 17 17 3 41 9 7 7 3 90 Sol.CAMA 67 2 0 7 1 8 1 18 8 5 0 18 4 7 7 7 9 Sol.CAMA 67 2 0 7 1 8 1 18 8 5 0 13 4 4 2 17 7 9 Sol.CAMA 67 2 0 7 1 8 1 18 8 5 0 13 4 4 2 17 7 9 Sol.CAMA 67 2 0 7 1 8 1 18 8 5 0 13 4 4 2 17 7 9 Sol.CAMA 10 0 5 7 1 8 1 18 1 1 2 0 0 1 2 0 1 4 2 17 7 9 Sol.CAMA 11 1 1 2 1 1 2 1 2 1 2 1 2 1 2 1 2 1 2	Sol.Abluc # 1 33 7 9 0 13 11 17 8 16 11 14 16 33 Sol.Actrs 9 0 8 0 1 8 17 17 3 14 16 17 73 Sol.Actrs 9 0 8 0 1 8 17 17 3 14 1 9 3 7 3 9 Sol.Actrs 1 2 40 7 18 16 14 17 17 3 14 1 9 3 7 3 9 Sol.Actrs 1 2 40 7 18 16 14 17 17 3 14 1 9 3 7 3 9 Sol.Actrs 2 0 7 1 8 11 18 8 7 0 1 4 2 17 3 Sol.Actrs 2 0 7 0 7 1 8 1 1 18 8 7 0 1 4 2 17 3 Sol.Actrs 2 0 7 0 1 2 1 2 1 2 1 7 3 Sol.Actrs 1 0 1 2 1 2 1 2 1 2 1 7 3 Sol.Actrs 1 0 1 2 1 2 1 2 1 2 1 7 3 Sol.Actrs 1 0 1 2 1 2 1 2 1 2 1 2 1 7 3 Sol.Actrs 1 0 1 2 1 2 1 2 1 2 1 2 1 7 3 Sol.Actrs 1 0 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 7 3 Sol.Actrs 1 0 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 1 4 1 4		6 -1	r:	-	=	0	r)	0	0	0	¢	ы	Ð	0	_	_
SOLIALTI 28 11 14 3 11 7 9 5 7 0 11 4 2 16 17 SOLIALTI 28 11 14 3 11 7 9 5 7 0 11 4 2 16 17 SOLIALIX 12 5 29 0 8 11 18 8 5 0 38 4 9 7 7 9 SOLICIAS 16 0 7 18 16 14 17 17 7 3 41 9 7 7 39 SOLICIAS 16 0 7 1 8 1 1 18 8 5 0 38 4 9 7 7 39 SOLICIAS 16 0 7 1 8 1 1 18 8 7 0 1 1 4 2 1 7 7 SOLINEMO 14 0 5 0 7 1 2 0 0 1 1 0 1 1 1 7 7 39 SOLIALIX 2 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	SOLIALTI 28 11 14 3 11 7 9 5 7 0 11 4 2 16 17 SOLIALTI 28 11 14 3 11 7 9 5 7 0 11 4 2 16 17 SOLIALISX 7 5 20 16 14 17 17 3 14 9 7 7 3 SOLIALISX 7 5 20 16 14 17 17 3 14 9 7 7 3 SOLIALISX 7 5 20 2 0 2 1 4 17 17 3 14 9 7 7 3 SOLIALISX 7 2 6 7 1 8 16 14 17 17 3 14 9 7 7 3 SOLIALISX 7 2 6 7 1 8 16 14 17 17 3 14 9 7 7 3 SOLIALISX 6 0 7 1 8 5 1 18 8 5 0 12 1 4 2 1 7 7 SOLIALISX 6 0 7 1 8 5 1 1 8 8 5 0 12 1 4 1 7 7 7 SOLIALISX 6 0 7 1 8 5 1 1 8 8 5 0 12 1 2 1 7 3 SOLIALISX 6 0 7 1 8 5 1 1 2 1 2 1 2 1 7 3 SOLIALISX 6 1 0 1 2 1 2 1 2 1 2 1 2 1 7 3 SOLIALISX 8 3 1 0 1 2 1 2 1 2 1 2 1 2 1 7 3 SOLIALISX 8 3 1 0 1 2 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	SOLADULC	6F	•	33	~	ņ	Ð	2	Ξ	17	~	8	=	Ţ	91	35
Sourcars 9 0 8 0 1 0 9 0 0 1 9 0 1 9 0 0 1 9 0 0 1 9 0 0 1 9 0 0 0 0 0 0 0 0 0 0 0 0 0 0 1<	Solucins 9 0 8 0 1 8 0 1 0 0 9 0 0 1 9 2000 M 6 2 2 40 7 1 8 5 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	SOLIALTI	38	=	4 1	m	=	7	¢	ŝ	Ľ	0	=	Ŧ	~ 1	91	Ľ
SOLICANA 67 2 40 7 18 16 14 17 17 3 41 9 3 25 80 SOLICANA 67 2 40 7 18 16 14 17 17 3 41 9 3 25 80 SOLIUNC 2 6 0 7 1 8 5 1 18 8 5 0 38 4 3 7 7 9 SOLIUNC 2 0 7 0 7 1 8 5 1 18 8 5 0 38 4 3 7 7 9 SOLIUNC 2 0 7 1 8 1 1 1 2 1 2 0 0 1 4 2 1 7 7 9 SOLIUNC 5 0 7 1 1 2 1 2 0 0 1 4 1 7 1 7 7 9 SOLIUNC 5 0 7 1 1 2 1 2 0 0 1 4 1 7 1 7 7 9 SOLIUNC 5 0 7 1 1 2 1 2 0 0 1 4 1 7 1 7 7 9 SOLIUNC 5 0 7 1 1 2 1 2 0 0 1 4 1 7 1 7 7 9 SOLIUNC 5 0 7 1 1 2 1 2 0 0 1 4 1 7 1 7 7 9 SOLIUNC 5 0 7 1 1 2 1 2 0 1 2 0 1 7 1 2 1 7 7 9 SOLIUNC 5 0 7 1 1 2 1 2 1 2 0 1 7 1 2 1 2 1 2 0 1 7 1 1 7 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Soluciona 67 2 40 7 18 16 14 17 17 3 41 9 3 73 80 Soluciona 67 2 40 7 18 16 14 17 17 3 41 9 3 73 80 Soluciona 2 0 2 0 1 4 2 1 7 3 90 Soluciona 2 0 2 0 1 4 2 1 7 3 90 Soluciona 2 0 2 0 1 4 2 1 7 3 90 Soluciona 2 0 2 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	SOUCAES	6	0	\$	0	_	a¢	¢		0	¢	6	0	0	_	¢
SOURLEX 42 5 29 0 8 11 18 8 5 0 38 4 3 7 39 SOURLEX 12 5 29 0 8 11 18 8 5 0 38 4 3 7 39 SOURIGO 16 0 7 1 8 5 9 0 2 0 1 4 1 1 1 SOURIGO 4 0 1 2 1 2 1 2 0 0 1 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	SOURLEX 12 5 29 0 8 11 18 8 5 0 38 4 3 7 90 201 FLEX 12 5 29 0 7 1 8 5 1 18 8 5 0 38 4 3 7 90 201 FLEX 16 0 7 1 18 7 5 9 0 2 0 1 1 4 2 1 7 7 30 50. INEMO 18 0 5 7 1 8 5 9 0 2 0 2 1 4 2 1 7 7 50 50. INEMO 18 0 5 0 1 0 2 0 1 1 4 1 2 1 2 1 2 50 50. COLREX 2 0 2 1 0 1 2 0 1 1 0 1 1 0 1 1 1 1 1 1	SOLICANA	67	ы	40	r-	8	91	Ī	13	13	-m	÷	с,	m	3	20
Souchistand 16 7 1 8 5 9 0 2 0 1 4 2 12 7 Souchistand 10 0 2 0 2 0 0 2 1 4 2 12 7 Souchistand 5 0 2 1 2 1 2 1 4 2 12 7 Souchistand 5 0 2 0 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1<	Solucidida 16 0 7 1 8 5 9 0 2 0 1 4 2 12 7 Solucidida 16 0 7 1 8 5 9 0 2 0 1 4 2 12 7 Solutivida 1 0 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2	SOLIFLEX	ç	ŝ	ŝ	9	30	=	8	æ	ŝ	•	38	÷	m	r ~	9
Solutions 2 0 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Solutino: 2 0 2 0 0 0 2 0 0 0 2 0 0 1 1 1 2 2 1 2 2 0 2 0	SOLIGIGA	16	•	ţ	_	×	ŝ	¢	0	сı	0	_	÷	2	2	r ~
SOLINEMO 10 0 5 0 5 7 0 3 0 6 0 0 5 7 20 3 0 0 1 0 5 7 3 0 5 0 1 2 0 2 0 0 1 1 0 1 0 1 0 0 5 3 3 SOLINEMO 1 0 1 2 1 2 1 0 2 0 0 1 1 0 1 0 1 0 0 1 0 0 1 0 0 1 0 0 1 0 0 1 0 0 1 0 0 0 1 0 0 0 1 0	SOLINEMO 10 0 5 7 7 0 3 0 5 7 0 0 0 5 0 0 0 5 7 8 20.NUEMO 10 0 5 7 7 0 3 0 7 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	SOLJUNC	-1	0	r:	Ģ	c	CI	¢	0	0	0	L 1	0	=	-	-
Soulititio 4 0 1 2 1 0 0 2 2 0 2 1 0 2 2 1 0 2 2 1 0 2 2 1 2 2 0 2 0	Solurado 4 0 1 2 1 0 2 2 2 0 2 1 0 2 2 1 0 2 2 2 0 2 1 0 2 2 2 2	SOLINEMO	01	0	ŝ	¢	ųr.		÷	~	0	0	Ŷ	c	9	ιcη.	►-
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	SONCARVE 5 0 4 0 1 2 1 2 0 0 4 0 1 4 0 1 2 1 2 0 0 4 1 4 0 1 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	SOLIRUGO	.,	0	-	~	_	Ċ	¢	۲ı	¢.	9	¢۱	-	Ð	CI	2
SONCOLER 2 0 0 0 2 0 1 0 2 0 1 0 2 0 1 0 2 0 1 0 2 0 2 0 2 0 2 0 2 0 1<	SONCOLER 2 0 2 0 0 0 0 1 0 1 0 1 0 1 0 1 0 2 0 1 1 0 1 1 0 1 1 0 1 0	SONCARVE	ŝ	c	Ŧ	0	_	. .1	-	C1	c	•	÷	ç	=	4	_
SPHENTE 15 0 10 4 1 4 5 1 4 1 4 2 9 7 STELENKI 1 0 1 0 0 1 0 0 1 4 1 2 9 7 STELENKI 1 0 1 0 0 1 0 0 1 0 1 4 2 9 7 SYMMARIE 0 1 0 0 1 0 0 1 0 1 1 0 1 1 0 1 1 0 1 1 0 1 1 0 1 1 0 1 1 0 1 1 0 1 1 0 1	SPHEINTE 15 0 10 4 1 4 5 1 4 1 4 2 9 7 STELLONG 1 0 1 0 1 0 0 1 0 1 1 4 1 4 1 4 1 4 1 4 1 4 1 4 1 4 1 4 1 4 1 4 1 4 1 4 1 4 1 4 1 4 1 4 1 1 0 1 0 1 0 1 1 0 1	SONCOLER	~1	Ċ	e.	0	0	ç	•	د م	0	•	-	ç	=	ب م	0
STELLONG 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 0 1 0 1 0 0 1 0 1 0 0 1 0	STELLONG 1 D 1 0 1 0 0 0 1 0 1 0 0 1 0 0 1 0 0 1 0 0 1 0 0 1 5 5 10 0 1 6 6 1 13 5 2 3 0 5 1 0 1 0 1 6 6 1 13 5 2 3 0 1 0 1 6 1 13 5 2 3 0 1 1 0 1 1 0 1 1 1 0 1 1 1 0 1 1 1 1	SPHEINTE	ŝ	Ċ	9	7	_	÷	۰¢	-	Ŧ	-	-	+	τı	G	-
STRERONE 8 3 3 0 5 1 0 1 6 6 0 0 0 1 8 SYMIALIBIL 6 6 0 0 1 6 6 0 0 1 5 1 5 5 23 5 23 20 6 6 1 5 3 8 27 21 50 6 6 1 1 1 1 1 5 23 20 6 6 1	STREROSE 8 3 3 0 2 3 0 5 10 0 10 0 0 0 0 0 0 0 0 5 11 1 1 1	STELLONG	_	ċ	_	0	ġ	¢	÷	0	-	÷	÷	<u> </u>	0	Ģ	_
SYMIALBU 6 0 0 0 0 0 0 0 1 5 TARAOFFI 100 1 66 6 27 21 26 27 20 6 64 13 5 28 82 TARAOFFI 100 1 66 6 27 21 26 27 20 6 64 13 5 28 82 THALDIOI 16 3 12 0 1 14 1 1 6 6 1 4 THALDIOI 16 3 12 0 1 14 1 1 4 6 1 4 4 4 4 6 6 1 1 1 1 4 6 <td>SYMIVALBUIL 6 0 0 0 0 0 0 1 5 TAXACMENT 100 1 66 6 27 21 26 27 20 6 64 13 5 28 82 TAXUCANA 5 0 2 3 1 1 1 1 4 0 1 2 1 4 4 1 4 1 4 4 4 1 1 1 4 1 4 1 4 4 1<</td> <td>STRERONE</td> <td>×</td> <td>~</td> <td>~</td> <td>•</td> <td>-1</td> <td>·~</td> <td>=</td> <td>w.</td> <td>=</td> <td>÷</td> <td>×</td> <td>ç</td> <td>÷</td> <td>Ş</td> <td>*</td>	SYMIVALBUIL 6 0 0 0 0 0 0 1 5 TAXACMENT 100 1 66 6 27 21 26 27 20 6 64 13 5 28 82 TAXUCANA 5 0 2 3 1 1 1 1 4 0 1 2 1 4 4 1 4 1 4 4 4 1 1 1 4 1 4 1 4 4 1<	STRERONE	×	~	~	•	- 1	·~	=	w.	=	÷	×	ç	÷	Ş	*
TARAOFH 100 1 66 6 27 21 26 27 20 6 64 13 5 28 R2 TAXAOFH 100 1 66 6 27 21 26 27 20 6 64 13 5 28 R2 TAXUCANA 5 0 2 1 14 1 1 1 1 0 1 14 1 THALPUBE 2 0 1 14 1 1 1 1 0 1 14 1 THALPUBE 2 0 1 1 1 1 0 0 1 1 1 1 0 1 1 1 1 1 THALPUBE 2 0 1 1 1 1 1 0 1 1 1 0 1 1 1 0 1 1 1 1 1 THALPUBE 2 0 1 1 1 1 0 0 0 1 1 1 1 0 1 1 1 1 0 1 1 1 THALPUBE 2 0 1 1 1 1 0 0 0 1 1 1 1 0 1 1 1 1 0 1 1 1 1 1 1 1 1 1 1	TARACIFI 100 1 66 6 27 21 26 27 20 6 64 13 5 28 82 TAXUCANA 5 0 2 3 12 0 1 14 1 2 13 5 28 82 TAXUCANA 5 0 2 3 12 0 1 14 1 1 20 6 13 5 28 82 THALPUBE 2 0 1 1 1 1 1 0 1 1 4 1 1 4 1 1 4 1 4 1 4 4 1 4 6 4 4 6 4 4 6 4 4 6 4 6 4 6 4 </td <td>SYMPALBU</td> <td>ų,</td> <td>¢</td> <td>2</td> <td>÷</td> <td>•</td> <td>¢</td> <td>=</td> <td>¢</td> <td>=</td> <td>•</td> <td>ţ,</td> <td>ç</td> <td>•</td> <td>_</td> <td>١Ċ.</td>	SYMPALBU	ų,	¢	2	÷	•	¢	=	¢	=	•	ţ,	ç	•	_	١Ċ.
TAXUCANA 5 0 2 3 0 0 1 4 0 1 2 0 1 4 0 1 2 0 1 1 4 1 1 1 0 0 1 5 1 0 1 1 4 1 1 1 1 0 0 1 5 1 0 1 1 4 1 1 1 1 0 0 1 5 1 0 1 1 4 1 1 1 1 0 0 1 1 5 1 0 0 2 0 1 1 1 1 1 1 1 0 0 1 1 5 1 1 1 1 1 1 1	TAXUCANA 5 (1 2 3 (1 14) 1 1 1 2 0 1 2 1 4 0 1 2 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	TARAOFFI	<u>100</u>	-	6 6	ç	71	5	26	ŗ,	50	¢	64	5	vs,	ñ	82
THALDIOI 16 3 12 0 1 14 1 1 0 0 15 1 0 5 THALDIOI 16 3 12 0 1 14 1 1 0 0 5 THALPUBE 2 0 1 1 1 0 0 0 2 0 0 2 0 0 2 0 0 2 THELNOVE 2 0 2 0 1 1 1 0 1 1 0 0 1 THELNOVE 2 0 2 0 1 1 2 0 0 0 2 THELNOVE 2 0 2 0 1 1 2 0 1 2 0 0 0 2 THELNOVE 2 0 2 0 1 1 2 0 1 2 0 1 2 0 1 2 0 1 2 0 1 1 2 1 0 0 0 2 THELNOVE 2 0 1 1 1 0 1 1 1 0 1 1 2 1 1 0 0 0 1 2 0 1 4 2 0 0 0 1 THELNOVE 115 19 68 8 20 43 21 2 9 18 4 88 19 2 2 22 101 TRAGRUH 2 0 3 3 5 2 4 2 7 8 0 3 2 3 0 0 0 1 2 0 1 2 1 TRAGRUH 2 0 3 2 2 3 0 0 0 3 2 2 3 0 0 0 7 2 2 2 101 TRAGRUH 2 0 3 2 2 3 0 0 0 5 7 3 3 7 14 4 1 9 3 3 1 TRAGRUH 2 0 1 0 1 0 1 2 0 1 2 0 1 2 0 1 2 0 1 2 0 1 1 2 1 1 TRAGRUH 2 0 1 0 1 0 1 2 0 1 2 0 1 2 0 1 1 2 1 1 TRAGRUH 2 0 1 0 1 0 1 1 0 0 0 1 1 2 1 1 TRAGRUH 2 0 3 2 2 3 0 0 0 5 7 3 3 1 TRUEBORE 7 0 3 2 2 3 0 0 0 5 7 3 3 1 TRUEBORE 7 0 3 0 3 2 2 2 3 0 0 0 5 7 3 3 1 TRUEBORE 7 0 3 0 3 2 2 2 3 0 0 0 5 5 2 1 0 1 2 0 1 1 2 1 1 TRUEBORE 7 0 3 0 0 0 5 5 3 1 14 14 19 3 3 1 TRUEBORE 7 0 3 0 0 0 1 1 0 0 1 1 2 1 10 10 1 1 1 10 0 0 1 1 2 1 10 10 1 1 10 1 10 1 10 1 1 10 1 10 1 1 10 1 10 1 1 10 1 10 1 1 10 1 10 1 1 10 1 1 10 1 1 10 1 1 10 1 1 10 1 1 10 1 1 10 1 1 10 1 1 1 10 1 1 10 1 1 1 10 1 1 1 10 1 1 1 1 10 1	THALDIOI 16 3 12 0 1 14 1 1 0 5 1 0 5 1 THALDIOI 16 3 12 0 1 14 1 1 0 5 1 0 5 1 0 5 1 0 5 1 0 5 1 0 5 1 0 5 1 0 5 1 0 5 1 0 2 0 0 0 1 1 1 0 1 1 1 0 0 1 1 1 0 1 1 1 0 0 0 0 0 0 1 <td>TAXUCANA</td> <td>6</td> <td>ç</td> <td>¢.</td> <td>77</td> <td>ç</td> <td>•</td> <td>•</td> <td></td> <td>-+</td> <td>0</td> <td>_</td> <td>-1</td> <td>c</td> <td>-</td> <td>,</td>	TAXUCANA	6	ç	¢.	7 7	ç	•	•		-+	0	_	-1	c	-	,
THALPUBE 2 0 1 1 1 0 0 2 0 8 2 0 1 1 1 0 0 1 1 2 0 0 1 1 1 0 1 2 0 0 1 1 1 1	THALPUBE 2 0 1 1 0 0 2 0 2 0 2 THELNOVE 2 0 1 1 0 0 1 1 0 2 0 2 THELNOVE 2 0 2 0 0 0 1 1 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 2 0 1 1 0 1 1 0 1 1 0 1 1 0 1 1 0 1 1 0 1 1 0 1 1 0 1	LUIGTVHJ.	16	m	<u>-</u> 1	0	_	7	-	_	¢	0	15		Ċ	5	2
THELNOVE 2 0 2 0 0 1 1 1 0 1 2 0 2 0 2 1 1 0 1 2 1 0 1 1 1 1	THELNOVE 2 1 2 0 0 0 1 1 1 0 1 2 0 0 1 1 1 0 1 2 1 0 2 0 1 2 1 2	THALPUBE	~1	=	_	_	₽	÷	•	•	ς.	0	÷	-1	Ċ	•	c,
THELPALU 4 0 4 0 0 0 2 0 1 2 0 4 0 4 1 1 10 18 11 8 0 29 14 2 9 40 11 11AMEK 115 19 68 8 20 43 11 8 0 29 14 2 9 40 11 11AMEK 115 19 68 8 20 43 11 8 0 29 14 2 9 40 11 11AMEK 115 19 68 8 20 43 21 29 18 4 1 29 18 19 2 22 10 11 11AMEK 115 19 $68 - 24 - 2 - 7 - 8 = 0 - 3 - 2 - 2 - 2 - 2 - 2 - 2 - 2 - 2 - 2$	THELPALU 4 0 4 0 0 0 2 0 1 2 0 2 0 1 2 1 1 2 1 0 4 1 1 1 1 1 0 1 8 1 1 8 0 29 1 4 2 9 40 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	THELNOVE	~1	÷	e.	Ċ	0	0	0	-	_	0		_	a	0	CI
THIJOCCI 47 0 30 6 11 10 18 11 8 0 29 14 2 9 40 TJAKCORD 65 0 39 7 19 0 13 17 30 5 37 14 2 9 40 TJAKCORD 65 0 39 7 19 0 13 17 30 5 37 14 1 19 53 TILIAMER 115 19 6 13 17 30 5 37 14 1 19 53 TRAGNUH 2 0 1 1 1 1 1 1 14 1 19 53 TRAGNUH 2 0 1 1 1 1 1 14 1 19 53 TRAGNUH 2 0 1 1 1 0 1 17 10 TRAGNUH 2 0 3 5 2 7 3 3 3 TRAGNUH 2 0 3 5 3 2 3 3 TREBORE 41 24 8 <	THILLOCCI 47 0 30 6 11 10 18 11 8 0 29 14 2 9 30 TLAKCORD 65 0 39 7 19 0 13 17 30 5 37 14 2 9 30 TLAKCORD 65 0 39 7 19 0 13 17 30 5 37 14 4 19 53 TLLAMEK 115 19 68 8 20 43 21 29 18 4 88 19 25 101 TRAGNUH 2 0 1 1 0 1 1 0 3 2 22 101 TRAGNUH 2 0 1 1 0 1 1 0 1 1 2 22 101 TRAGNUH 2 0 3 5 2 7 8 1 1 1 2 1 TRAGNUH 2 0 3 5 2 3 3 1 1 2 1 1 1 2 2 2 1 1	T}IELPALU	4	⇒	Ŧ	-	=	÷	r -1	9	ŕ١	Ċ			_	-	Ŧ
TIAKCORD 65 0 39 7 19 0 13 17 30 5 37 14 4 19 53 TILIAMEK 115 19 68 8 20 43 21 29 18 4 83 19 2 22 101 TRAGDUH 2 0 1 0 1 0 1 0 0 1 0 1 2 TIREBORE 41 24 8 3 6 24 2 7 8 0 32 3 2 7 37 TREBORE 41 24 8 3 6 24 2 7 8 0 32 3 2 7 37 TREBORE 41 24 8 3 6 24 2 7 8 0 32 3 2 7 37 TREBORE 41 24 8 3 6 24 2 7 8 0 32 3 2 7 37 TREBORE 41 24 8 3 6 24 2 7 8 0 32 3 2 7 37 TREBORE 41 24 8 3 6 24 2 7 8 0 32 3 2 7 37 TREBORE 41 24 8 3 6 24 2 7 8 0 32 3 2 7 37 TREBORE 41 24 8 3 6 24 2 7 8 0 32 3 2 7 37 TREBORE 41 24 8 3 6 24 2 7 8 0 32 3 2 7 37 TREBORE 41 24 8 3 6 24 2 7 8 0 32 3 2 7 8 7 37 TREBORE 7 0 3 0 3 2 2 7 8 0 32 3 2 7 8 7 10 1 2 1 10 10 1 2 1 10 10 1 10 10 10 1 10 10 10 10 10 10	TIAKCORD 65 0 39 7 19 0 13 17 30 5 37 14 4 19 53 TILIAMEK 115 19 68 8 20 43 21 29 18 4 83 19 2 22 101 TRAGDUH 2 0 1 0 1 1 0 1 1 0 1 1 2 7 8 0 32 3 2 20 101 11KIEBORE 41 24 8 3 6 24 2 7 8 0 32 3 2 2 7 37 7 14 14 1 2 4 1 2 4 8 3 6 2 4 2 7 7 8 0 32 3 2 2 7 37 7 11KIEBORE 7 0 3 0 3 2 1 2 7 8 0 32 3 2 2 7 37 11KIEBORE 7 0 3 1 2 2 3 0 0 0 3 2 2 3 2 101 11KIEBORE 7 0 3 1 2 2 3 0 0 0 3 2 2 3 2 101 11KIEBORE 7 0 3 1 2 2 3 0 0 0 3 2 2 3 2 101 11KIEBORE 7 0 3 1 2 2 3 0 0 0 3 2 2 3 2 101 11KIEBORE 7 0 3 2 2 3 0 0 0 3 2 2 2 2 101 11KIEBORE 7 0 3 1 2 2 3 0 0 0 3 2 2 2 2 101 11KIEBORE 7 0 3 2 2 3 0 0 0 3 2 2 2 2 101 11KIEBORE 7 0 3 1 2 2 3 0 0 0 3 2 2 2 2 101 11KIEBORE 7 0 3 1 2 2 3 0 0 0 3 2 2 3 2 101 11KIEBORE 7 0 3 1 2 2 3 10 0 3 1 2 2 2 3 101 11KIEBORE 7 0 3 1 2 2 3 10 0 0 3 2 2 2 2 2 101 11KIEBORE 7 0 1 1 2 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	THIJOCCI	47	•	Ċ£	9		0	20	=	20	•	2	Ŧ	~ 1 ·	-	2
TILIAMEK 115 19 6X 8 20 43 21 29 18 4 83 19 2 22 101 TRAGDUH 2 0 1 0 1 0 1 0 1 2 22 3 3 TRAGDUH 2 0 1 0 1 0 1 2 3 3 3 TREBORE 41 24 8 3 6 24 2 7 8 0 32 3 2 7 37 TREBORE 41 24 8 3 6 24 2 7 8 0 32 3 2 7 37 TREBORE 41 2 0 3 0 3 2 2 7 8 0 32 3 2 7 37 TREBORE 41 2 4 8 3 6 2 4 2 7 8 0 32 3 2 7 37 TREBORE 41 2 4 8 3 6 2 4 2 7 8 0 32 3 2 7 37 TREBORE 41 2 4 8 3 6 2 4 2 7 8 0 32 3 2 7 37 TREBORE 41 2 4 8 3 6 2 4 2 7 8 0 32 3 2 7 37 TREBORE 41 2 4 8 3 6 2 4 2 7 8 0 32 3 2 7 3 7 TREBORE 41 2 4 8 3 6 2 4 2 7 8 7 8 0 32 3 2 7 7 8 7 37 TREBORE 7 0 3 0 3 2 1 2 7 8 7 7 7 7 7 7 7 8 0 3 2 1 2 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	TILIAMER 115 19 6X 8 20 43 21 29 18 4 83 19 2 22 101 TRAGDUH 2 0 1 0 1 0 1 0 0 1 1 2 TRAGDUH 2 0 1 0 1 0 1 2 2 3 TREBORE 41 24 8 3 6 24 2 7 8 0 32 3 2 7 37 TREBORE 7 0 3 0 3 2 3 0 0 3 3 2 7 37 TREFEPE 7 0 3 0 3 2 2 3 0 0 5 5 2 7 37 TREFEPE 7 0 3 1 2 1 1 1 1 2 1 2 1 2 1 1 1 1 1 1 1 1	TJARCORD	3	•	ŝ	r- -	6	ċ	<u>_</u>	17	œ	5	57	Ī	-+	Ś	2
TRAGIULH 2 0 1 0 1 0 1 1 2 1 2 1 2 1 2 1 2 1 2 1	TRAGIUM 2 0 1 0 1 0 1 1 0 1 0 0 1 1 2 1 2 1 2 1	TILIAMEK	115	6	88	æ	2	÷	친	2	2	-7	ŝ	5	τı	2	0
TRIEBORE 41 24 8 3 6 24 2 7 8 0 32 3 2 7 37 TRIEBORE 7 0 3 0 3 2 2 2 3 0 0 3 0 0 5 2 TRIERE 7 0 3 1 0 5 2 1 37	TRIEBORE 41 24 8 3 6 24 2 7 8 0 32 3 2 7 37 TRIEBORE 7 0 4 0 3 2 2 3 0 0 3 0 0 5 2 7 37 	TRAGDUBI	¢1	e	-	•	_	_	9	_	•	¢	_		•	_	-
TXIFREPE 7 0 4 0 3 2 2 3 0 0 3 0 0 5 2	TXUFREPE 7 0 4 0 3 2 2 3 0 0 5 2 1	TRUEBORE	ŧ	54	×	r,	Ŷ	7.	2	F		¢	ň	÷۳.	헌	~	37
······································		TRUFREPE	7	•	÷	•	÷	~1	e)	r r ,	c	¢	m	Ð	0	ŝ	∩ i
121	421											•					 -
121	12H																
									÷	_							

					·	Soil Or	rdor			ĺ	Soil Mo	oisture		Canopy	-
		Solution	urcin Ma	I CLI 31							12	CDA	372	с О	Ç
SPECIES	0#	3 5	5	LAC	8	-	GB	L	5	ן כ	1			ŝį	ļ
		 	5	: -	15	•	9	32	1	0	цц	9	m	÷	국
DANATINI	2	., :			: :	1.4	- 1	Ň	Ű	_	128	~	_	티	128
TRULGRAN	22	Ş	60	N :	- -	5 -	2			=	_	c	c		0
TRIOAURA	_	_	0	¢	⇒		Þ			• •	. :	: 4		~	Ч. Т
TSUGCANA	96	_	36	\$	~	Ę,	2	= .	÷,	= <	5	• =	,	ۍ د	
THISSFARF	<u>~</u>	0	c	c	~1	6 1	¢	•	ð	-	5	.		•	• -
TVPHLA'	ı —	0	0	•		ð	-	•	c	-	0	_ :		• 5	- ç
FILMINAMER	. 99	c	Ş	-	20	9	7	2	ŝ	ç	~	<u>8</u>	= ,	2;	1 -
UNKN 751	3	. 0	Þ	c	-	÷		•	0	•	=		.		
TINK N 701			Ċ	c	÷	⇒	0	÷	_	Ð	=	•	_		- <
UNKN 877		. 0		0	¢	•	0	1	0	0	•	a (- ·	_ •	∍ -
INTER CONTRACT	• ••	0	ŝ	0	7	0	0	Ģ	ei	_	ሰሳ	c .	- 1	т -	
	•	9	G	0	_	0	_	¢	•	÷	•		-	<u> </u>	_ :
	. =	• =	- 0			-0	0	¢1	_	e	=	¢	0	_	-
	: .	5 -		, c	. 0	v.	c	a	0	0	м.	0	•	_	el :
VALLANUU VIED LET	•, -			-	. 0	•	_	0	•	=	•	÷	0	_	ç
VEKBHAAI	_ ,	: :					=	~	_	=	٢ı	÷	=	-1	-
VERBINAP	-, -		·ı -	: =		; ;		. –	0	=	5	=	0		=
VERBURT	- 1	= <			. :		• •		¢	0	r:	0	0	c	c,
VEROOFF	~1		-1 r		•	2 6	، ،	• =	. 0		r-1	•	9	-	۲I
VEROSERP	~ ,	-	-n :	-	5 •	5 7	n 2	, "	. =		80	0	0	~	8
VIBUACER	R :	ភ្.	<u>s</u> :		• -	0, 1	5 6			. 0	1	-7	2	Ś	23
VIBULENT	26	•	9	5	2.	<u>-</u> ;	- 1				ie	G	-	_	-
VIBUOPUI.	-1	c	∍	0	~	.		•	•	2 6		; -		_	r.
VIBUTRIL.	m	c	_	9	<u>-</u> 17	-1	- :	5 0	3 4	• •				2	-
VICICIRAC	-	÷	ò	Ċ	~1	~1	-	÷.		2 4	-		. ~	; ₽	-
VIOLAFFI	2	÷	Ŀ	0	8	-	2	_	.	• •	- -	, ,	n -		4
VIOUBLAN	2	à	¢	¢	2	•	-1	5	9	-1	Ξ.	•t -	- :		
VIOLCANA	r i	c		•	9	-	-	9	ç		2 1	- :	5 -	- 4	
VIOLOTICI)		•	2	=	-	•	ń	0	~~ .	9		. ¢	•1 8	5 1	, I
VIOL 1 ARK	2	_	=	•	ر م	_	17	÷	ç	-	=	Ţ	•	-1	<u>t</u>
Y 11714-11714											:		:	:	:

겁

:

:

		Suil F	Parent Ma	uerial		Sail O	Irder				Soil A	Abisture		Canop	~
SPECIES	ç	ö	СГ	I.AC	8	≏	GB	-:	CI.	Ü	SD	WS	SW	3	2
VIOLPUBE	68	%	- 78	0	22	27	0	29	<u>0</u>	0	64	-	 rt.	ย	- 39
VIDLROST	ŕι	¢I	c	o	0	6 1	0	•	0	c	~1	c	•	Ū	-
VIOLSORO	c)	0	c	¢	¢1	0	0	-1	•	¢	_	a	•	-	L1
VIOL 788	-	0	0	¢	_	0	0	-	e	¢	_	a	0	c	-
VIOL SP	ec	•	9	0	ri		m	-	r-i	¢	÷	¢	_	-	æ
VITIRIPA	64 64	-	옃	-	20	2	33	Ξ	~:	0	45	Ξ	•	5	53
WALDFRAG	23	20	_	0	- 1	5	∍	¢	r.	0		-	0	ŝ	5

:

1

i

-

Appendix 4. Distribution of species by soil parent material, soil order, soil moisture and canopy closure.

APPENDIX 5

LISTING OF SPECIES BY MICROHABITAT

Appendix 5. Listing of species by microhabitat. Species codes preser	ted in Appendix 3.
--	--------------------

CLOSED	CARECOMM	DICECANA	ιονιρίοι	OUERALBA
CLOSED	CARECONIN	DICECUCU	LONIHIRS	OUERMACR
CANOPY	CARECKIS	DIFRIONI	LYCOANNO	OUERRUBR
1. Dry Floors		DIRCPALL	LYCODEND	RANUABOR
(n=265 species)	CAREGRAC	DRVINGART	LYCOOBOB	RANUACRI
ABIEBALS	CARGORAL	DRIOCRIS	LYSINUMM	RANURECU
ACERNEGU	CAREGRAM	DRYOINTE	MAIACANA	RHAMALNI
ACERRUBR	CARELITE	DRYOMARG	MAIARACE	RHAMCATH
ACERSACC	CAREINTH	FPIFVIRG	MAIASTEL	RHUSRADI
ACERSACN	CARELAXI	EPIPHEI I	MATTSTRU	RIBEAMER
ACERSANI	CAREBACK	FOLIARVE	MEDILUPU	RIBECYNO
ACERSPIC	CAREPEDU	FOUTHYEM	MENICANA	RIBEGLAN
ACTAPACH	CAREFEER	FOULAEV	MENTARVE	RIBERUBR
ACTARUBR	CAREPLAN	FOUISCIR	MUJEFFU	RIBETRIS
ACTA SP	CAREPLAT	ERIGANNU	MITCREPE	ROBIPSEU
ADIAPEDA	CARERADI	FRIGPHIL	MITEDIPH	ROSABLAN
AGRIGRYP	CAREROSE	FRVTAMER	MONOHYPO	RUBUALLE
ALLITRIC	CARESDAR	ELIPARUGO	MONOUNIE	RUBUIDAE
AMELINTE	CARETENE	FAGUGRAN	ONOCSENS	RUBUOCCI
AMEL_SP	CAREWOOD	FESTSUBY	ORYZASPE	RUBUODOR
AMPHBRAC	CARE OV	FRAGVESC	OSMOCLAY	RUBUPUBE
ANEMVIRG	CARE SP	FRAGVING	OSMUCLAY	SAMBCANA
APOCANDR	CARPCARO	FRAXAMER	OSTRVIRG	SAMBRAPU
AQUICANA	CARYCORD	FRAXNIGR	OXALSTRI	SANGCANA
ARALNUDI	CAULTHAT	FRAXPENN	PANAOUIN	SANIMARI
ARCTMINU	CELASCAN	GALESPEC	PARTINSE	SANITRIF
ARISTRIP	CERAFONT	GALIAPAR	PHEGCONN	SAN1 SP
ASARCANA	CHIMUMBE	GALICIRC	PHLO SP	SCHIPURP
ASCLSYRI	CINNLATI	GALILANC	PHRYLEPT	SMILHERB
ASTECILI	CIRCALPI	GALIOBTU	PICEGLAU	SMILHISP
ASTECORD	CIRCLUTE	GALITRIF	PILEPUMI	SOLADULC
ASTELATE	CLAYCARO	GALL SP	PINUSTRO	SOLIALTI
ASTEMACR	CLINBORE	GERAMACU	POA ALSO	SOLICAES
ASTENOVE	CORNALTE	GERAROBE	POA COMP	SOLICANA
ATHYFILI	CORNFOEM	GEUMALEP	POA PALU	SOLIFLEX
ATHYTHEL	CORNRUGO	GEUMCANA	POAPRAT	SOLINEMO
BETUALLE	CORNSTOL	GEUMLACI	PODOPELT	SOLIRUGO
BETUPAPY	CORYCORN	GEUMURBA	POLYACRO	SPHEINTE
BIDEFRON	CRAT SPI	GEUM SP	POLYPAUC	STREROSE
BOEHCYLI	CRAT SP2	GLYCSTRI	POLYPUBE	SYMPALBU
BOTRMATR	CRAT SP3	GYMNDRYO	POPUBALS	TARAOFFI
BOTRMULT	CRAT SP	HEPAACUT	POPUGRAN	TAXUCANA
BOTRVIRG	CRYPCANA	HIERAURA	POPUTREM	THALDIO
BRACEREC	CYSTBULB	HIERCAES	PREN SP	THELNOVE
CARDDIPH	CYSTERAG	HYDRVIRG	PRUNSERO	THELPALU
CAREALBU	CYSTTENU	HYPEPERF	PRUNVIRG	THUJOCCI
CAREARCT	DACTGLOM	IMPACAPE	PRUNVULG	TIARCORD
CAREBACK	DANTSPIC	LAPOCANA	PTERAQUI	TILIAMER
CAREBLAN	DESMGLUT	LONICANA	PYROELLI	TRAGDUBI

CLOSED	APOCANDR	CORNSTOL	MAIARACE	SOLICANA
CLUSED		CORYCORN	MAIASTEL	SOLIFLEX
CANOPY	ARCTMINH	CRAT SP	MATTSTRU	SOLIGIGA
1. (cont'd)	ARISTRIP	CYPRCALC	MENTARVE	SOLIBUGO
TRIEBORE	ASARCANA	CYSTRUI R	MITEDIPH	SPHEINTE
TRIFREPE	ASCI SVRI	DRYOCART	MUHIMEXI	STELLONG
TRILEREC	ASTELATE	DRYOCRIS	ONOCSENS	TARAOFFI
TRILGRAN	ASTENOVE		OSTRVIRG	ΤΔΥΠΓΔΝΔ
TSUGCANA	ASTEDUNI		PARTINSE	THALDIOL
ULMUAMER		FLYMVIRG	PHRVIEPT	THALPURE
URTIDIGR	RETUALLE	EPH CH I	PILEPLIMI	THELPALL
UVULGRAN	DETUALLE		POVIJSTRO	THUÔCCI
VACCANGU	BETUPAPT	EPILCOLO EDIT L'EDT		TINECORD
VERBTHAP	BIDEFKON	EPILLOF I		TULAMED
VEROOFFI	BOERCYLI	EPILFARY	PLANKUUC DOA ALCO	
VEROSERP	BOIKVIKG	EPIPPELL	POA_ALSO	
VIBUACER	CARDDIPH	EQUIARVE	POA_PALU	TRILEREC
VIBULENT	CARDPENS	ERIOPHIL,	PUA_PKA I	TRILGKAN
VIBUTRII.	CAREALOP	ERIGSTR	POLYACRO	TSUGCANA
VIOLAFEL	CAREARCT	ERYTAMER	POPUBALS	TYPHLATI
VIOL BLAN	CAREBEBB	EUPAMACU	POPUTREM	ULMUAMER
VIOLOUKU	CAREBLAN	EUPAPERF	PRUNSERO	UNKN_751
VIOLIABR	CAREBREV	EUPARUGO	PRUNVIRG	URTIDIOI
	CARECEPH	FAGUGRAN	PRUNVULG	VIBULENT
	CARECRIS	FESTSUBV	QUERMACR	VIBUTRIL
VIOL 789	CAREDEWE	FRAGVIRG	RANUABOR	VIOLAFFI
	CAREGRAC	FRAXAMER	RANUACRI	VIOLBLAN
VIUL_SP	CAREINTU	FRAXNIGR	RANURECU	VIOLCANA
	CARELANU	FRAXPENN	RHAMCATH	VIOLCUCU
WALDEKAG	CAREPEDU	GALETETR	RHUSRADI	VIOLLABR
	CAREPENS	GALIASPR	RIBEAMER	VIOLPUBE
2. Moist Floors	CARERADI	GALIOBTU	RIBECYNO	VIOL_SP
(n=197 species)	CAREROSE	GALITRIF	RIBEGLAN	VITIRIPA
ABIEBALS	ÇARESPAR	GENTANDR	RIBERUBR	WALDFRAG
ACERNEGU	CARESTIP	GERAROBE	RIBETRIS	
ACERRUBR	CARETENE	GEUMCANA	ROSABLAN	3. Wet Floors
ACERSACC	CARETRIB	GEUMLACI	ROSAPALU	(n=152 species)
ACERSACN	CAREVULP	GEUM SP	RUBUALLE	ACERRIAR
ACERSANI	CARPCARO	GLYCSTRI	RUBUIDAE	ACERSACN
ACERSPIC	CARYCORD	GYMNDRYO	RUBUODOR	ACENSACI
ACHIMILL	CALILTHAL	HIFRCAES	RUBUPUBE	ACHIMILL
ACTARUBR	CERAFONT	HYPEPERF	SALIBEBB	AUKUUIUA
ACTA SP	CIRCALPI	IMPACAPE	SALIDISC	AMPHERAU
ADIAPEDA	CIRCILITE	[APOCANA	SALIERIO	ASCLINCA
AGRIGRYP	CIRCADVE		SAMBRADI	ASTECILI
AGROGICA	CLEMVIDO		SCHIDIND	ASTELATE
	CLEMIVINU	L VCOUNTE	SCIPATEO	ASTEUMBE
AMPHRPAC				BIDEFRON
ANEMCANA	CORNALIE			BOEHCYLI
ANEXAUA	CORNEUEM	LT SUNCIVINI	FOLIALT	CAREALOP
ANEMAIKO	CORNRUGO	MAIACANA	SOLIALTI	

CLOSED	POA_COMP	CARÉBLAN	GALITRIF	SOLIALTI
CANOPY	POA_PALU	CARECRIS	GERAROBE	SOLICANA
3 (cont'd)	POA_PRAT	CAREDEWE	GEUMCANA	SOLIFLEX
CADEDEDD	POPUTREM	CAREGRAC	GEUMLAÇI	TARAOFFI
CAREDEDD CADEDI AN	PRUNVIRG	CAREGRAN	GEUMRIVA	THELNOVE
CAREDLAN	PRUNVULG	CAREINTU	GEUM_SP	THELPALU
CARECRIS	RANUABOR	CAREPEDU	GLYCSTRI	THUJOCCI
CARECRIS	RANUACRI	CAREPENS	GYMNDRYO	TIARCORD
CAREGRAC	RHAMCATH	CAREPLAN	HEPAACUT	TILIAMER
CAREURAN	RHUSRAÐI	CAREPROJ	HIERCAES	TRIEBORE
CAREPANI	RIBETRIS	CARERADI	HYSTPATU	TRILEREC
CARERADI	ROBIPSEU	CARERETR	IMPACAPE	TRILGRAN
CAREAUSE	ROSAPALU	CAREROSE	LAPOCANA	TSUGCANA
CARESTIC	RUBUPUBE	CARESTIP	LYSICILI	ULMUAMER
CARETERE	SALIBEBB	CARETENE	LYSINUMM	VIBULENT
	SALIERIO	CARE_OV	MAIACANA	VIOLAFF]
CERAFOUR	SCUTLATE	CARE_SP	MAIARACE	VIOLBLAN
CICLIMACU	SOLADULC	CAULTHAL	MAIASTEL	VIOLLABR
CICOMACO	SOLICANA	CINNLATI	MATTSTRU	VIOLPUBE
CORNSTOL	SOLIGIGA	CIRCALPI	MITEDIPH	VITIRIPA
CKAI_SP	TARAOFFI	CIRCLUTE	ONOCSENS	
COLUMNE	ULMUAMER	CIRSARVE	OSMUCLAY	5. Wet
EQUIARYE	VIBULENT	CLEMVIRG	OSTRVIRG	Donrossiane
ERIGPHIL	VIBUOPUL	CORNALTE	PARTINSE	(n=115 emerging)
FRAGVIKG	VIOLAFFI	CORNFOEM	PILEPUMI	(u~115 species)
FRAANKIK	VITIRIPA	CORNSTOL	PLANMAJO	ABIEBALS
CALIODIU		CYSTBULB	POA_COMP	ACERSACC
GALIOBIU	4. Moist	DACTGLOM	POA_PALU	ACERSACN
CENTANDR	Depressions	DRYOCART	POLYPUBE	ACEKSPIC
GEUM CB	(n=136 energies)	DRYOCRIS	POPUBALS	ACTA_5P
GLUM_ar	ADIEDA(S	DRYOINTE	PRUNVIRG	ADIAPEDA
	ACERPIIAR	DRYOMARG	PRUNVULG	AGROGIGA
DADACADE	ACERSACC	ELYMREPE	RANUABOR	ALLITRIC
	ACERSACN	ELYMVIRG	RANUACRI	ALNUINCA
	ACERSPIC	EPILCOLO	RANUHICA	AMPHBKAC
LIFALUES EVCOAMED	ACTABLIER	EPIPHELL	RANURECU	ARISTRIP
LICOAMER	ΑΓΙΑΘΕΊΛΑ	EQUIARVE	RHAMCATH	ASARCANA
		EQUILAEV	RHUSRADI	ASCLINCA
L TSICILI I Veinitikkka	ARACHODI	ERIGANNU	RIBEAMER	ASTECORD
LISINOMINI MATTETOI?	ASARCANA	ERIGPHIL	RIBECYNO	ASTELATE
MALISING		ERIG_SP	RIBEGLAN	ASTENUVE
	ACCENTE	ERYTAMER	RIBELACU	ASTERUNI
NUTUNEAL	ASTERIO	EUPARUGO	RIBETRIS	ATHYTHE
ONUCSENS	ATUVELI	FAGUGRAN	RUBUIDAE	
PAKTINGE		FRAGVIRG	RUBUPUBE	DETUALLE
PILEPUMI		FRAXAMER	SAMBCANA	BOEUCYLI
FLANMAJU BEANDUCE		FRAXNIGR	SCHIPURP	
		FRAXPENN	SCUTLATE	CARDDIPH
				CADEDIAN

CLOSED	LAPOCANA	(p=33 species)	ARISTRIP	HIERCAES
CANOPY	LIPALOES	ASTENOVE	ASCLSYRI	IMPACAPE
E (constid)	LYCOUNIF	BOEHCYLI	ASTECILI	LONICANA
	LYSICILI	CALTPALU	ASTECORD	LONIDIO
CARECEPH	LYSINUMM	CAREINTU	ASTELATE	MAIACANA
CARECRIS	MAJARACE	CARESTIP	ATHYFILI	MAIARACE
CAREGRAU	MATTSTRU	CARE OV	BETUPAPY	MAIASTEL
CAREHIKI	MENTARVE	CICUMACU	BOTRVIRG	MENICANA
CAREINTU	ONOCSENS	CINNLATI	CARDDIPH	MONOUNIF
CAREPEDU	OSMUREGA	EPILCOLO	CAREBLAN	ONOCSENS
CAREPLAN	PARTINSE	EOUIARVE	CAREDEWE	ORYZASPE
CAREPSEU	PILEPUMI	FRAXNIGR	CAREDIGI	OSMOCLAY
CARERADI	POA ALSO	GEUMRIVA	CAREGRAC	OSTRVIRG
CARERETR	POA PALU	GLYCSTRI	CAREPEDU	PANI SP
CARETENE	POLŸPUBE	GYMNDRYO	CAREPENS	PARTINSE
CARETRIB	PRUNVIRG	IMPACAPE	CAREPLAN	PLANMAJO
CARPCARO	RANUABOR	LAPOCANA	CARERADI	POA COMP
CAULTHAL	RANURECU	L YCOAMER	CAREROSE	POA PRAT
CICUBULB	RHAMCATH	LYCOUNIE	CARESPAR	PODOPELT
CINNLATI	RHUSRADI	EYSICILI	CARPCARO	POLYPUBE
CIRCALPI	RIBEAMER	LYSINUMM	CARYCORD	PRUNSERO
CIRCI.UTE	RIBECYNO	MATTSTRU	CEANAMER	PRUNVIRG
CORNALTE	RIBETRIS	ONOCSENS	CELASCAN	PTERAOUI
CORNRUGO	RIBE 827	PILEPUMI	CIRCLUTE	PYROELLI
CORNSTOL	RUBUIDAE	POA PALLI	CORNALTE	ÓUÉRRUBR
CYSTBULB	RUBUOCCE	RANUHICA	CORNFOEM	RANUACRI
DRYOCART	RUBUPUBE	RIBEAMER	CORNELIGO	RANURECU
DRYOCRIS	SALIERIO	RIBEGLAN	CORYCORN	RHAMCATH
DRYOINTE	SCIRATRO	SAMBCANA	CRAT SP	RHUSRADI
DRYOMARG	SIUMSUAV	SOLADULC	CYSTERAG	RIBECYNO
EPILCILI	SOLADULC	TIARCORD	DESMGLUT	RIBETRIS
EPILCOLO	SOLIALTI	LII MUAMER	DICECANA	ROBIPSEL
EPILPARV	SOLICANA	INKN 701	DIRCPALIE	RUBUALLE
EPIPHELL	SOLIFLEX		DRYOCART	RUBUIDAE
EQUIARVE	TARAOFFI	+10 <u>0</u> _51	DRYOINTE	SAMBRAPU
EUPAMACU	THELPALU	7 Dev Can	DRYOMARG	SMILHERB
EUPARUGO	THUJOCCI	fi Diy Gap	EPIPHELL	SOLADULC
FESTSUBV	TIARCORD	(D=115 species)	ERYTAMER	SOLIALTI
FRAXAMER	TILIAMER	ABIEBALS	FESTSUBV	SOLICANA
FRAXNIGR	TRILEREC	ACERRUBR	FRAGVESC	SOLIELEX
FRAXPENN	TRILGRAN	ACERSACC	FRAGVIRG	SOLUTING
GALIOBTU	HILMHAMER	ACERSPIC	FRAXAMER	TARAGEFL
GALITRIF	VIOI BI AN	ACTAPACH	GALESPEC	
GERAROBE	VIOLOUCII	ACTARUBR	GATITRIE	THUDOCCI
GLYCSTRI	VIOLPURE	ACTA_SP	GERAMACU	TIARCORD
HIERCAES	VITIRIPA	ADIAPEDA	GERAROBE	TILIAMER
HYDRVIRG	TTING A	ALLITRIC	GEUMLACI	TRUGRAN
IMPACAPE	6 Seen	AMEL_SP	GEUM SP	VIBUACER
IRIS_SP	a neeh	APOCANDR	GLYCSTRI	VIBULENT

_

CLOSED	CERARORE			PRINSERO
CLOSED	GEUMCANA	ARISTRIP	FOLIADVE	
CANOPY	GEUMLACI	ASAPCANA	EQUIANT	PTERAOUT
7. (cont'd)	GEVOSTRI	ASCESVRE	FRIGPHII	PVROFILL
VIOLBLAN	IMPACAPE	ASTECILI	ERVTAMER	OUERRUBR
VIOLLABR	LACT SP	ASTECORD	FAGUGRAN	OUER SP
VIOLPUBE		ASTELATE	FESTSURV	RANL'ABOR
VIOL_SP	MAIACANA	ASTEMACR	FRAGVESC	RANUACRI
VITIRIPA	ONOCSENS	ASTENOVE	FRAXAMER	RANURECU
WALDFRAG	PARTINSE	ATHVEIL	FRAXMICR	RHAMCATH
	POPUBALS	ATHYTHE	FRAXPENN	RHUSRADI
8. Moist Gap	PRUNVIRG	RETIPARY	GALIAPAR	RIBECVNO
(n=68 species)	RANUAROR	BOTRVIEG	GALIATAK	RIBETRIS
ABIEBALS	RANURECH	CARDDIPH	GERAROBE	ROBIOSELL
ACERRUBR	RHAMCATH	CAREARCT	GEUMI ACI	ROSABLAN
ACERSACC	RIBECYNO	CAREBACK	GLUMERCI GLVCSTRI	RUBUALLE
ACERSACN	RIBERUBR	CAREBLAN	GYMNDRYO	RUBUIDAE
ACERSPIC	RIBETRIS	CAREDEWE	HEPAACUT	RUBUOCCI
ARALNUDI	RUBUIDAE	CAREDIGI	HIERCAES	RUBUODOR
ARISTRIP	RUBUPUBE	CAREGRAC	HYDRVIRG	RUBUEUBE
ATHYFILI	SAMBCANA	CAREHITC	IMPACAPE	SAMBRAPU
BOEHCYL1	SOLADULC	CARFINTU	LEONCARD	SANGCANA
CAREBEBB	SOLIALTI	CARFLAXI	LONICANA	SANIMARI
CAREBLAN	SOLICANA	CAREPEDII	1 YCOOBOB	SCHIPLIRP
CAREDEWE	SOLIGIGA	CAREPENS	MAIACANA	SILEVULG
CAREGRAC	TARAOFFI	CAREPLAN	MAIARACE	SMILLERB
CAREPEDU	THALPUBE	CARERADI	MAIASTEL	SOLIALTI
CARERADI	TIARCORD	CAREROSE	MATTSTRU	SOLICAES
CAREROSE	TRILEREC	CARYCORD	MITCREPE	SOLICANA
CAULTHAL	ULMUAMER	CAULTHAL	MITEDIPH	SOLIFLEX
CINNLATI	VIOLCUCU	CELASCAN	MONAFIST	SOLINEMO
CIRCALPI	VIOLPUBE	CIRCALPI	ONOCSENS	SONCARVE
CIRCLUTE		CIRCLUTE	ORYZASPE	STREROSE
CORNALTE	9. MOUND	CLAYCARO	OSMOCLAY	SYMPALBU
CORNSTOL	(n=165 species)	CORNALTE	OSTRVIRG	TARAOFFI
CYSTBULB	ADIEDATS	CORNRUGÓ	OXAL SP	THALDIOI
DRYOCART	ACEDDIED	CORYCORN	PARTINSE	THUJOCCI
DRYOCRIŞ	ACEDEACC	CRAT SP3	PHRYLEPT	TIARCORD
DRYOINTE	ACEDEANI	CRAT SP	PILEPUMI	TILIAMER
DRYOMARG	ACENSANI	DANTSPIC	PINUSTRO	TRIEBORE
EPILCOLO	ACENSFIC	DESMGLUT	POA ALSO	TRILEREC
EPILPARV	ACTADUDD	DICECANA	POACOMP	TRILGRAN
EPIPHELL	ACTA SD	DICECUCU	POA SALT	TSUGCANA
ERYTAMER		DIERLONI	POA SP	ULMUAMER
EUPAMACU	ALLITRIC	DRYOCART	PODOPELT	UVULGRAN
FRAGVESC		DRYOINTE	POLYPUBE	VIBUACER
FRAXNIGR		DRYOMARG	POPUGRAN	VIBULENT
FRAXPENN	ΔΠΠΓΑΝΑ	EPIFVIRG	POPUTREM	VIOLAFFI
GALITRIF	AVOICHUA	EPILCOLO	PREN_SP	VIOLBLAN

•

	CVANIDBYO	CIRCALDI	EDIT 1 EDT	TIARCORD
CLOSED			CFILLUF I	THIAMER
CANOPY	HEPAACUT	CIACLUIE	COULADVE	ΤΡΙΘΟΛΡΗ
9. (cont'd)	MT DRVIRU DADACADE		EDVTAMED	
VIOLCUCU	IMPACARE	INIFACAFE	CNT I MALEN	TRUCCANA
VIOLPUBE	LUNICANA		EDAGUESC	LUNGIAMED
VIOL SP	MAIAUANA	14. Log	FRAGYESC	VIOLDI AN
VITIRIPA		(n=102 species)	PRACOVINO PRACOVINO	
WALDFRAG	MAIASIEL	ABIEBALS	FRAXAMER FRAXAMER	
	MITCREPE	ACERRUBR	PRAANIOK	VIOLPUBE
10 Dry Pit	MITEDIPH	ACERSACN	FRAXPENN	VIUL_3P
(91	ORYZASPE	ACERSPIC	GALIASPK	VITIKIPA
(n=ai species)	OSMOCLAY	ACTASP	GALIPALU	WAEDFRAG
ABIEBALS	OSTRVIRG	ANEMQUIN	GALITIKIF	
ACEKRUBK	PINUSTRO	ARALNUDI	GEKAROBE	13. Stump
ACERSACC	POA_ALSO	ARISTRIP	GEUMLACI	(n=51 species)
ACTA_SP	POLYACRO	ASTELATE	GLYCSTRI	ABIEBALS
AMEL_SP	POLYPUBE	ATHYFILI	GYMNDRYO	ACERRUBR
ARALNUDI	POPUGRAN	BETUALLE	HEPAACUT	ACERSACC
ARISTRIP	PRUNSERÓ	BETUPAPY	HIERCAES	ACERSPIC
ASARCANA	PRUNVIRG	BIDEFRON	IMPACAPE	ARALNUDI
ASTEMACR	PTERAQUI	BOEHCYL1	LAPOCANA	ARISTRIP
ATHYFILI	PYROELLI	CARDDIPH	LYCOUNIF	ASTELATE
BOEHCYLI	QUERRUBR	CAREBLAN	LYSINUMM	BETUALLE
BOTRVIRG	RANUABOR	CARECRIS	MAIACANA	BOEHCYUI
CAREDEWE	RHUSRADI	CAREDEWE	MAIARACE	CARECRIS
CAREPEDU	RIBECYNÓ	CAREGRAC	MENTARVE	CAREDEWE
CAREPENS	ROBIPSEU	CAREINTU	MILIEFFU	CAREGRAC
CARPCARO	ROSABLAN	CAREPEDU	MITEDIPH	CAREPEDU
CARYCORD	RUBUALLE	CAREPENS	ONOCSENS	CAREPENS
CAULTHAL	SAMBRAPU	CARERADI	PARTINSE	CARE OV
CELASCAN	SANGCANA	CAREROSE	PILEPUMI	CIRCALPI
CIRCLUTE	SMILHERB	CARESTIP	POA_PALU	CIRCLUTE
CLAYCARO	SOLIFLEX	CARETENE	POLYPUBE	DRYOCART
CORNALTE	SYMPALBU	CARPCARO	RANUABOR	DRYOINTE
CORNRUGO	TARAOFFI	CARYCORD	RANURECU	EPIPHELL
CORYCORN	THUJOCCI	CAULTHAL	RHAMCATH	FAGUGRAN
DESMGLUT	TILIAMER	CINNLATI	RIBECYNO	FRAGVIRG
DICECANA	TRIEBORE	CIRCALPI	RIBEGLAN	GERAROBE
DRYOCART	TRILEREC	CIRCLUTE	RUBUALLE	GYMNDRYO
DRYOINTE	TRILGRAN	CLEMVIRG	RUBUIDAE	IMPACAPE
DRYOMARG	VIBUACER	CORNALTE	RUBUODOR	MAIACANA
EPIEVIRG	VIBULENT	CORNEDEM	RUBUPUBE	MAIARACE
EPIPHELL	VIOLPUBE	CVSTRUI R	RUBU_840	PARTINSE
ERYTAMER	WALDFRAG	DOVOCADT	SOLADULC	PICEGLAU
FAGUGRAN		DRIGGARI	SOLIALTI	PILEPLIMI
FESTSUBV	11. Maist Pit	DRIGENIJ	SOLICANA	POI VPAUC
FRAXAMER	(n=5 enosion)	ECUILOBA	SOLIFLEX	POLYPIIRE
GALITRIF	(II-5 species)	EURILOBA EDR COLO	TARAOFFI	OUERALRA
GERAROBE	CAREINTU	EPILCOLO	THUJOCCI	QUERREDA

CI OSED	CARECRIS	GYMNDRYO	TARAOFEI	ARISTRIP
CLOSED	CAREDEWE	HIERCAES	TAYLICANA	ASTELATE
CANOPY	CAREGRAC	HYDRVIRG	TVAL DURE	ATHVENI
13. (cont'd)	CARENTU	IMPACAPE		BOSHCVII
QUERRUBR	CAREPECK	LACT SP		BOTRVIEG
RANUABOR	CAREPEDU		TIAPCORD	CARDDIRH
RIBEAMER	CAREPENS		THIAMED	CAREARCT
RUBUIDAE	CAREPLAN	1 VCOLNIE	TRIEBORE	CAPERIAN
RUBUODOR	CARERADI	ΜΔΙΔΓΔΝΔ	TDILEDEC	CAREDEWE
RUBUPUBE	CAREROSE	MAIACANA	TRILEREC TRILCEAN	CARECEAC
SAMBRAPU	CARESTIP	MEDILUPI	TSUGCANA	CAPERITH
SOLADULC	CARETENE	MIT IFFET	LI MUAMER	CAREPEDII
SOLICANA	CARE OV	MITENIPH	UDTIDICD	CAREPENS
SOLIFLEX		OSMOCLAV		CAREFENS
SPHEINTE		OSTRVIEG		CAREFROJ
TARAOFFI	CRUCINAL	DADTINCE		CARERADI
THUJÓCCI		PINIETRA	VICL_SE	CARENOSE
TIARCORD	CINALATI		VITINIPA	CARETENE
TRIEBORE		POA_ALSO	15 64	CAREINID
TSUGCANA			15. Stone	CARE_SP
ULMUAMER			(n=20 species)	CARTCORD
VIOLCUCU	CODNALTE		ANEMQUIN	CROLINAL
	CORNALTE	DDEN CD	ASTELATE	
14. Raised Root	CORNETOL	PREN_OF	CAREBLAN	CIRCLUIE
Mat		PRUNOERU	CAREPEDL	DRYOCART
(n=1.38 emonion)	CVETERAC		CAREPENS	DRYOCARI
(II-100 species)	DEVOCART	OUEDNIACO	DRYOCRIS	DRYOCKIS
ADICOALO	DRIUCARI	QUERMACK	EPIPHELL	DRYUINTE
ACENAUOR	DRIOCKIS	DANUADUK	EQUIHYEM	EMIPHELL
ACERSACE	DRIVINE	RANUACRI	ERIGPHIL	EQUISCIN
ACEROAUA		RANURECU	FRAXAMER	EKYTAMEK
ACCASHIC		RHAMCATH	GALITRIF	FAGUGKAN
ACTARUDE		RECONIC	LONIDIO	FRAGVESC
ACTA_SP	EQUIARYE	NIDECTINU DIDECTIANI	MAIACANA	PRAXAMER
ALLEIKK	EQUISCIN	RIBEGLAN	PARTINSE	CALILANC
ARALNUDI	ENIGEN		PRUNVIRG	CLYCETEL
	FAGUGRAN	RUBUALLE	RHUSRADI	
ARISTRIP		KUBUIDAE	RIBECYNO	HIEKCAES
ASAKLANA	PRAAAMER ED & YNVER	RUBUODOK	SCHIPURP	HTDKVIKG
ASTECORD	PRAANIOR ED & YDENDY	KUBUPUBE	SOLIFLEX	IMPACAPE
ASTELATE	CALLEDO	SAMBLANA	TRILGRAN	LYCODEND
ASTENUVE	GALIASPK	SAMBRAPU		MAIACANA
ATRYFILI	GALIOBIU	SCHIPURP	16. Lane	MAJAKACE
BETUALLE	GALIPALU	SCUTLATE	(n=89 species)	MENTARVE
BETUPAPY	GALITRIF	SIUMSUAV	ABIEBALS	MITCREPE
BIDEFRON	GERAROBE	SOLADULC	ACERRUBR	UNOCSENS
BOEHCYLI	GEUMCANA	SOLICANA	ACERSACC	ORYZASPE
CARDDIPH	GEUMLACI	SOLIFLEX	ACTA SP	OSMOCLAY
CAREARCT	GEUM_SP	SOLIRUGO	ARALNUDI	OSMUCLAY
CAREBLAN	GLYCSTRI	SPHEINTE		OSTRVIRG

CLOSED	CARECRIS	SALIERIO	CARYCORD	PILEPUMI
CANOPY	CAREGRAN	SANGCANA	CELASCAN	PLANLANC
16. (cont'd)	CARELANU	SCUTLATE	CERAFONT	PLANRUGE
POA PRAT	CARERADI	SOLIALTI	CHRYLEUC	POA_COMP
	CAREROSE	SOLIGIGA	CIRSARVE	POA_PALU
	CARESTIP	TARAOFFI	CORNSTOL	POA_PRAT
POPLICRAN	CARETENE	TILIAMER	CORYCORN	POPUBALS
POPUTREM	CAREVULP	TUSSFARF	CRAT_SP	POPUTREM
	CARYCORD	ULMUAMER	DAUCCARO	POTERECT
PRUNYING	CLAYCARO	VIOLPUBE	DESMGLUT	PRUNSERO
PTERACU	CORNRUGO	VIOL_SP	ELYMVIRG	PRUNVIRG
	CORNSTOL	VITIRIPA	EPILCILI	PRUNVULG
PANUABOR	EPILCILI		EPILLEPT	PTERAQUI
RANGAGON	EPILLEPT	18. Regenerat-	EPIPHELL	QUERRUBR
RHUSPADI	EPIPHELL	ing Field	EQUIARVE	RANUACRI
RIBECYNO	EQUIARVE	(n=125 species)	ERIGPHIL	RHAMCATH
	ERIGPHIL	ARIERAIS	ERIGSTRI	RIBEAMER
	ERYTAMER	ACEDRIAD	EUPAMACU	RIBECYNO
RUBUIOCCI	EUPAMACU	ACERSACE	EUPAPERF	RUBUALLE
SALIEBIO	EUPAPERF	ACERSACN	EUTHGRAM	RUBUIDAE
SALIERIO	FRAGVIRG	ACHIMILI	FRAGVIRG	RUBUPUBE
SANCCANA	FRAXNIGR	ACTARIAR	FRAXAMER	RUDBHIRT
	FRAXPENN	ACRIGRAD	FRAXNIGR	SALIBEBB
	GALIOBTU	AGROCICA	FRAXPENN	SALIDISC
SOLICIANA	GALITRIF	AGROGIOA	GALICIRC	SALIERIO
TADAGET	GLYCSTRI	AMELINITE	GALIÔBTU	SISYMONT
	HYDRVIRG	AMELINIC	GALITRIF	SMILHERB
TIARCORD	IMPACAPE	ADOCANDD	GENTANDR	SOLIALTI
THIAKED	LEERORYZ	APOCANOR	HIERCAES	SOLICANA
TRIEDARE	LYCOUNIF	AQUICANA ASCI SVD1	HYPEPERF	SOLIGIGA
TRIEDUKE	LYSICILI	ASTECUL	IMPACAPE	SOLINEMO
TORCHAN	LYSITERR	ASTELLI ASTELLIE	IRI5_SP	TARAOFFI
TREORAN TRATIAMED	MAIACANA	ASTELATE	LACTCANA	THUJOCCI
	MEDILUPU	ASTEDUNI	LIPALOES	TILIAMER
	MENTARVE	RETEIDADV	LYCOAMER	TRAGDUBI
	MUHLMEXI	BOTRVIRG	LYCOUNIF	TRIFREPE
VIOLEUDE	ONOCSENS	BOMNER	LYSICILI	TRILGRAN
VITIDIDA	PARTINSE	CARERI AN	MAJACANA	ULMUAMER
YTTKIPA	PHALARUN	CAREDUAN	MATTSTRU	VIBUACER
17 Diteb	PILEPUMI	CARECRIS	MEDILUPU	VIBULENT
17. DRCM	PLANRUGE	CAREGRAN	MELIALBA	VICICRAC
(¤≖68 species)	POA_PALU	CAPELANI :	MELIOFFI	VIQLAFFI
ACERSACC	POA_PRAT	CARECANO	MUHLFRON	VIOL_SP
ACERSACN	POLYPUBE		ONOCSENS	VITIRIPA
ACHIMILL	PRUNVIRG		OXALSTRI	
AGROGIGA	RANUACRI	CARESTIC	PANIACUM	OPEN
ANEMCANA	RANURECU		PARTINSE	CANOPY
ASTELATE	ROBIPSEU		PHALARUN	
ASTEPUNI	SALIDISC	CARECARU	PHLEPRAT	17. Dry Floors

___ .

<u></u> ,				
OPEN	CAREROSE	GALIAPA R	POA_SALT	TRIOAURA
CANOPY	CARESPAR	GALILANC	POLYPAUC	TSUGCANA
10 D Vissen	CARE_OV	GALITRIF	POLYPUBE	ULMUAMER
19. DEV F100ES	CARE_SP	GERAMACU	PREN_SP	URTIDIGR
(n=205 species)	CARPCARO	GERAROBE	PRUNSERO	UVULGRAN
ABIEBALS	CARYCORD	GEUMCANA	PRUNVIRG	VACCANGU
ACERRUBR	CAULTHAL	GEUMLACI	PTERAQUÍ	VERBTHAP
ACERSACC	CEANAMER	GEUMURBA	QUERALBA	VIBUACER
ACERSACN	CELASCAN	GLYCSTRI	QUERMACR	VIBULENT
ACERSPIC	CERAFONT	GYMNDRYO	QUERRUBR	VIOLBLAN
ACTAPACH	CIRCALFI	HACKVIRG	RANUABOR	VIOLPUBE
ACTA_SP	CIRCLUTE	HEPAACUT	RANUACRI	VIOLSORO
ADIAPEDA	CIRSARVE	HIERCAES	RANURECU	VITIRIPA
AGRIGRYP	CIRSVULG	HYDRVIRG	RHAMCATH	WALDFRAG
ALLITRIC	CLAYCARO	HYSTPATU	RHUSRADI	
AMELINTE	CLEMVIRG	IMPACAPE	RIBECYNO	20. Moist
AMEL_SP	CONYCANA	JUNCTENU	RIBETRIS	Floors
AQUICANA	CORNALTE	LACTCANA	ROBIPSEU	(n=153 species)
ARALNUDI	CORNFOEM	LACTSERR	RUBUALLE	ARIERALS
ARCTMINU	CORNRUGO	LAPOCANA	RUBUIDAE	ACERRUBR
ARISTRIP	CORYCORN	LEONCARD	RUBUOCCI	ACERSACC
ASARCANA	CRAT SP	LOBEINFL	RUBUPUBE	ACERSACN
ASCLSYRI	CRYPCANA	LONICANA	SAMBRAPU	ACERSPIC
ASTECILI	CYSTBULB	LONIDIOI	SANGCANA	ACHIMUT
ASTELATE	DACTGLOM	MAIACANA	SANI_SP	ACTARUBR
ASTEMACR	DANTSPIC	MAIARACE	SCHIPURP	ACTA SP
ASTENOVE	DICECANA	MAIASTEL	SMILHERB	AGROGIGA
ATHYFILI	DICECUCU	MEDILUPU	SMILHISP	AMPHBRAC
BETUPAPY	DIERLONI	MELIALBA	SOLADULC	ANEMVIRG
BOEHCYLI	DIRCPALU	MENICANA	SOLIALTI	ARALNUDI
BOTRVIRG	DRYOCART	MILIEFFU	SOLICAES	ARISTRIP
BRACEREC	DRYOCRIS	MITCREPE	SOLICANA	ASARCANA
CALYSEPI	DRYOINTE	MITEDIPH	SOLIFLEX	ASCLSYRI
CARDACAN	DRYOMARG	NEPECATA	SOLIGIGA	ASTECILI
CARDDIPH	EPIFVIRG	ONOCSENS	SOLINEMO	ASTELANC
CAREALBU	EPILCOLO	ONOPACAN	SQLIRUGO	ASTELATE
CAREARCT	EPIPHELL	ORYZASPE	SONCARVE	ASTENOVE
CAREBACK	ERIGANNU	OSMOCLAY	SONCOLER	ASTEPUNI
CAREBLAN	ERIGPHIL	OSTRVIRG	SPHEINTE	ATHYFILI
CARECRIS	ERYTAMER	PANICAPI	SYMPALBU	BETUALLE
CAREDEWE	EUPAPERF	PARTINSE	TARAOFFI	BIDEFRON
CAREGRAC	EUPARUGO	PHLO_SP	THALDIOI	BOEHCYLL
CAREHITC	FAGUGRAN	PHRYLEPT	THUJOCCI	CARDDIPH
CAREINTU	FESTSUBV	PILEPUMI	TIARCORD	CAREARCT
CARELAXI	FRAGVESC	PINUSTRO	TILIAMER	CAREBEBB
CAREPEDU	FRAXAMER	PLANRUGE	TRIEBORE	CAREBLAN
CAREPENS	FRAXNIGR	POA_COMP	TRIFREPE	CAREBREV
CAREPLAN	FRAXPENN	POA_PALU	TRILEREC	CARECRIS
CARERADI	GALETETR	POA_PRAT	TRILGRAN	

.—

OBEN	CENTANDA	BA THE INC		
OPEN	GENTANDK	KUBUPUBE	CAREINTU	RIBEAMER
CANOPY	GERAKOBE	SALIDISC	CAREPROJ	RIBECYNO
20. (cont'd)	GEUMUANA	SALIERIO	CARERADI	RIBETRIS
CAREDEWE	GEUMLACI	SAMBÇANA	CARERETR	RUBUOCCI
CAREGRAC		SAMBKAPU	CAREROSE	SALIERIO
CAREGRAN	ULTUSIKI	SCUILATE	CARESTIP	SAMBCANA
CAREINTU	HIERCAES	SOLADULC	CARETRIB	SCUTLATE
CARELANU	IMPACAPE	SOLIALTI	CAREVULP	SOLADULC
CAREPEDU	INULHELI	SOLICANA	CIRCLUTE	SOLICANA
CAREPENS	IRISVERS	SOLIFLEX	CLEMVIRG	SOLIGIGA
CARERADI	IRIS_SP	SOLIGIGA	CORNRUGO	SPHEINTE
CAREROSE	LACT_SP	SOLIRUGO	CORNSTOL	TARAOFFI
CARESTIP	LAPOCANA	SONCARVE	CYSTBULB	THUJOCCI
CARETENE	LEERVIRG	SPHEINTE	DRYOCART	TILIAMER
CAREVILLE	LIPALOES	TARAOFF)	ELYMVIRG	ULMUAMER
CALETHAL	LYCOUNIF	THUJOCCI	EPILCOLO	VIOLCUCU
CELASCAN	LYSICILI	TIARCORD	EPILPARV	VITIRIPA
CERACONT	MAIACANA	TILIAMER	EPIPHELL	
	MATTSTRU	TRIEBORE	EQUIARVE	22. Moist
	MEDILUPU	TRILEREC	ERIGANNU	Depressions
CINNLATI	MENTARVE	TRIEGRAN	EUPAMACU	Depressions
CIRCALPI	MITEDIPH	TSUGCANA	EUPAPERF	(n=42 species)
CIRCLUIE	ONOCSENS	ULMUAMER	EUPARUGO	ABIEBALS
CORNAL TE	OSMOCLAY	VIBUOPUL	FRAXNIGR	ACERSACN
CORNSTOL	PARTINSE	VIOLAFFI	GALIOBTU	ACERSPIC
CRAT_SP	PHLEPRAT	VIOLBLAN	GALIPALU	ARISTRIP
CRYPCANA	PILEPUM	VIOLCUCU	GERAROBE	ASTELATE
CYSTBULB	PLANMAIO	VIOLPUBE	GEUMI ACI	BOEHCYUI
DRYOCART	PLANRUGE	VITIRIPA	GEUM SP	CARECRIS
DRYOCRIS	POA ALSO		GLYCSTRI	CAREGRAC
DRYOINTE	POA PALU	21 Wet Floore	HYSTPATH	CAREINTU
DRYOMARG	POA PRAT		IMPACAPE	CARERADI
ELYMYIRG	POLYPUBE	(D=82 species)		CIRCALPI
EPILCOLO	POPLIBALS	ACERSACN	LABOCANIA	CIRCLUTE
EPILPARV	POPLITREM	ACERSPIC	LEEDODV7	CORNALTE
EPIPHELL	PRINVIRG	ADIAPEDA	LECRON 12	CORNSTOL
EQUIARVE	DP LINVERC	AGROGIGA		CYSTBULB
ERIGANNU	PANUADOD	ARISTRIP	MENTARYE	DRYOCART
ERYTAMER	PANILACDI	ASTELANC	ONOCSENS	DRYOINTE
EUPAMACU	BANURCH	ASTELATE	DADTENS	EQUIARVE
EUTHGRAM	RANURELU	ASTEPUNI	PAKTINSE	ERIGPHIL
FAGUGRAN	RHAMCATH	ATHYFILI	FILEPUMI	ERYTAMER
FRAGVESC	RIUSKADI	BIDEFRON	POA_COMP	FRAXNIGR
FRAGVIRG	RIBEAMER	BOEHCYLI	PUA_PALU	GALIPALU
FRAXAMER	RIBEGLAN	CAREBEBB	POA_PRAT	GALITRIF
FRAXNIGR	RIBERUBR	CAREBLAN	PRUNVULG	GERAROBE
FRAXPENN	RIBETRIS	CARECRIS	RANUACRI	GEUMLACI
GALIOBTU	ROBIPSEU	CAREDEWE	RANURECU	GLYCSTRI
GALITRIE	RUBUIDAE	CAREGRAC	RHAMCATH	IMPACAPE
	RUBUODOR		RHUSRADI	

				· ·
OPEN	LYCOUNIF	ADIAPEDA	PRUNVIRG	SAMBRAPU
CANOPY	ONOCSENS	ARALNUDI	QUERRUBR	SOLADULC
11 (4-1)	PILEPUMI	ARISTRIP	RANUABOR	TARAOFFI
22. (cont'd)	RANURECU	ASARCANA	RHAMCATH	
ONOCSENS	RHAMCATH	ASTELATE	RUBUIDAE	27. Wet Pit
PARTINSE	SAMBCANA	CARDDIPH	RUBUOCCI	(n= 1 species)
POA_PALC	SOLADULC	CAREARCT	SAMBRAPU	(n= t species)
PRUNVIRG	SOLICANA	CAREBLAN	SANGCANA	IMFACAFE
RANUACRI	TARAOFFI	CAREDIGI	SMILHERB	
RANURECU	TILIAMER	CAREPEDU	SOLADULC	28. Log
RHAMCATH	ULMUAMER	CAREPENS	SOLICANA	(n= 64 species)
RIBETRIS	VIOLCUCU	CARERADI	SOLUUNC	ACERSACC
RUBUIDAE		CAREROSE	SONCARVE	ACERSPIC
RUBUPUBE	74 Seen	CARYCORD	SONCOLER	AGROSTOL
SAMBCANA	24. Seep	CALL THAL	SYMPALBU	ARALNUDI
SOLADULC	(n=25 species)	CIRCULITE	TARAOFFI	ARISTRIP
TARAOFFI	ARISTRIP	CIRSARVE		ASTELATE
TIARCORD	ATHYFILI	CIRSARTE CIRSVIII G	TILIAMER	ASTEPUNI
ULMUAMER	BOEHCYLI	CONVEANA	TUICDEDE	ATHYFILI
	CAREÇRIN	CONICANA	TRIFNEFE TRILCOAN	BETUALLE
23. Wet	CARECRIS	CORNECEM		BETUPAPY
Depressions	CAREGRAC	CVETTENU		BOERCYLI
/n=10 spectors	CAREINTU	DICECANA	WALDERAG	CAREARCT
(u-bo species)	CAREPROJ		37 Dec. 014	CAREDEWE
ACERSALN	CARERETR	DIEKLONI	26. Dry Pit	CAREINTE
ACEKSPIC	CARETENE	DRYOUARI	(n=26 species)	CAREPEDI
ARISTRIP	CARPCARO	DRYDINTE	ACERSACC	CARERADI
ASTELANC	CIRCALPI	EPIFVIKG	ARISTRIP	CICUMACE
ATHYFIL	EPILCOLO	EPILCOLO	ASARCANA	CINNI ATI
BIDEFRON	EUPAMACU	EPIPHELL	CIRCLUTE	CIRCALPI
BOEHCYLI	FRAXPENN	ERYTAMER	CIRSVULG	CIRCUSTE
CAREBEBB	GALITRIF	EUTHGRAM	DICECANA	CLEMVIDC
CAREBLAN	GLYCSTRI	FRAGVESC	DRYOCART	
CARECRIS	IMPACAPE	FRAXAMER	DRYOINTE	DEVOCART
CAREDEWE	LYCOAMER	GALITRIF	EPILCOLO	DRICEARI
CAREGRAC	ONOCSENS	GERAMACU	ERYTAMER	CLYNDURG
CAREINTU	RANUABOR	GERAROBE	EUPAMACU	
CARERETR	RANURECU	GYMNDRYO	FAGUGRAN	
CARESTIP	SQLADULC	HACKVIRG	FESTSUBV	EPILPARV
CORNRUGO	TIARCORD	MAIACANA	FRAXAMER	FRAGVESC
ELYMVIRG	URTIDIGR	MAIARACE	GALITRIF	FRAGVING
EPILPARV		MITCREPE	GERAROBE	FRAXNIGR
EQUIARVE	25. Mound	ONOPAÇAN	HACKVIRG	FRAXPENN
FRAXNIGR	(n=76 anasies)	ORYZASPE	MAJARACE	GALIAPAR
GALIPALU	(n=70 species)	OSMOCLAY	POLYPURE	GALIOBTU
GERAROBE	ABIEBALS	OSTRVIRG	PRINVIRG	GALIPALU
GEUMLACI	AUEKKUBK	PHRYLEPT	RANUABOR	GALITRIF
GLYCSTRI	ACERSAUC	POA PRAT	RUBUIDAE	GERAROBE
IMPACAPE	ACTAPACH	POLYPAUC	RUBLOCCI	GLYCSTRI
LAPOCANA	ACTARUBR	POLYPUBE		HIERCAES

OPEN	ERIGPHIL	CORNSTOL	SAMBRAPH	CARDDIPH
CANOPY	FESTSUBV	CYSTBULB	SCUTLATE	CARDNUTA
	FRAGVIRG	DAUCCARO	SICYANGU	CAREARCT
28. (cont'd)	GALITRIF	DRYOCART	SOLADULC	CAREBACK
	GLYCSTR1	DRYOCRIS	SOLIALTI	CAREBEBB
LACT_SP	HIERCAES	DRYOINTE	SOLICANA	CAREBLAN
LYCOUNIF	IMPACAPE	DRYOMARG	SOLIFLEX	CAREGRAC
MAIACANA	MAIACANA	EPILCOLO	SPHEINTE	CAREGRAN
OSTRVIRG	PARTINSE	EPILPARV	TARAOFFI	
PARTINSE	RANURECU	EPIPHELI	THUROCCI	CARFLANU
PILEPUMI	RUBUIDAE	EOUIARVE	TIARCORD	CARFLAXI
PRUNVIRG	RUBUPUBE	ERIGANNU	THIAMER	CAREPENS
RANURECU	SOLICANA	ERIGPHIL.	TRIFBORE	CAREPROL
RIBETRIS	SOLIELEX	ERIG SP		CAREBETR
RUBUIDAE	SOLIRUGO	FUPAMACU	TRILGRAN	CAREROSE
RUBUPUBE	SPHEINTE	FRAGVESC	TSUGCANA	CARESTIP
RUBU_840	TARAOFEI	FRAGVIRG	TILMUAMER	CARETENE
SOLIALTI	TIARCORD	FRAXNIGR	VIOI BLAN	CARETRIB
SOLICANA	TIME OND	FRAXPENN	VIOLEAR	CAREVINE
SOLIGIGA	10 D. 1 I.D. 4	GALITRIE	VIOLENBR	CARE 868
SPHEINTE	30. Raised Root	GERAMACIU	VITIRIPA	CARE 870
TARAOFFI	Mat	GERARORE	YTHNEA	
TAXUCANA	(n=96 species)	CELINCANA		CARVCORD
TIARCORD	ACERRUBR	GEUMCARA	31. Lane	CALLERAL
TRIEBORE	ACERSACC	GLUM _SF	(n= 181 species)	CERADONT
TSUGCANA	ACERSPIC		ACERSACC	CURVIEUZ
ULMUAMER	ACTA SP	INDACADE	ACERSACN	CIDCALDI
VIOLAFFI			ACHIMILL	CIRCALPI
VIOLCUCU	ARALNUDI	LACISERK	ACTA_SP	CIRCLUTE
VIOL_SP	ARISTRIP	MALACANIA	AGROGIGA	CIRSARYE
	ASTELATE	MAIACANA	AGROSTOL	CIRSVULG
29. Stump	ASTEPUNI	MATISIKU	AMBRARTE	CLEMMIKO
(n= 34 species)	BETUALLE		ANEMCANA	CONTCANA
ACERSACC	BIDEFRON	DADTRICE	ANEMVIRG	CORNECCO
ARISTRIP	BOEHCYLI	PARTINGE	ARCTMINU	COKINSTOL
ASARCANA	CARDPENS	FICEGLA()	ARISTRIP	
ASTELATE	CAREARCT		ASÇESYRI	DAUTOLOM
ASTEPUNI	CAREBLAN	PUA_PRAI	ASTECILI	DAUCCARO
BETUPAPY	CAREDEWE	PREN_SP	ASTECORD	DIANAKME
CAREDEWE	CAREGRAC	PRUNVIKG	ASTELANC	DICECANA
CAREGRAC	CAREPEDU	NANUADUK	ASTELATE	DKYOÇAKI
CAREPEDU	CAREPENS	RANUACRI	ASTENOVE	EL TAIREFE
CAREROSE	CARE SP	KHAMCATH	ASTEPUNI	EPILLOLU
CAREVULP	CIRCALPI	KHUSKADI	ATHYFILI	EPIPHELL
CIRCALPI	CIRCLUTE	RIBECTNO	BETUPAPY	EQUIARVE
DRYOCART	CIRSVULG	RUBUIDAE	BIDEFRON	EQUILAEV
DRYONTE	CLEMVIRG	RUBUOCCI	BOEHCYLI	ERIGANNU
FPIPHELI	CONVCANA	RUBUODOR	BOTRVIRG	ERIGPHIL
FOULABLE	CORNALTE	RUBUPUBE	BROMINER	ERIGSTRI
LQUIARVE	CONTRACTOR D	SAMBCANA		ERYTAMER

OPEN	PARTINSE	URTIDIGR	SOLIGIGA	CERAFONT
CANOPY	PHALARUN	URTIDIQI	TUSSFARF	CHRYLEUC
31 (sont'd)	PHLEPRAT	VERBHAST		CIRSARVE
	PHRYLEPT	VERBURTI	33. Regenerat-	CIRSVULG
CUPAMACO	PILEPUMI	VEROSERP	ing Field	CORNRUGO
EUFAFEAF	PLANLANC	VIBUACER		CORNSTOL
CESTERAT	PLANMAJO	VICICRAC	ACEDDIED	CRAT_SP
FDACVERC	PLANRUGE	VIOLAFFI	ACERROOK	DANTSPIC
ED ACUIDO	POA_ALSO	VIOLBLAN		DAUCCARO
ED A VAMED	POA_COMP	VIOLCANA	ACHIMIT	DIANARME
ED A VNICO	POA_PALU	VIOLCUCU	ACRICEVE	ELYMREPE
ED A VDENNI	POA_PRAT	VIOLPUBE	ACROCICA	ELYMVIRG
CALIORITI	POLYPERS	VIOLSORO	AMBUDDAC	EPILCILI
	POLYPUBE	VITIRIPA	AMENDAAC	EPILLEPT
CERADORE	POPUTREM		ANEMICANA	EPIPHELL
CEUMI ACL	POTENORV	32. Ditch	ANTENICO	EQUIARVE
CEUM SD	POTERECT	(n=33 species)	ABOCANOP	EQUIHYEM
	PRUNSERO	ACERSACC	ADUCANON	ERIGPHIL
	PRUNVIRG	AGROGIGA	AQUICANA	ERIGSTRI
	PRUNVULG	ASTECUL		EUPAMACU
	RANUABOR	ASTELANC	ASCENTA	EUPAPERF
LIEDCAES	RANUACRI	ASTELATE	ASTERNIC	EUTHGRAM
LIVEREDE	RANURECU	ASTEPUNI	ASTELANC	FAGUGRAN
UVSTBATTI	RHAMCATH	CARETENE	ASTELANC	FESTARUN
INDACADE	RHUSRADI	CICUMACU	ASTENOVE	FESTRUBR
	RIBECYNO	CORNALTE	ASTEROVE	FESTSUBV
INS_SF	RUBUALLE	CORNSTOL	ASIEFUNI DETTIDADV	FRAGVIRG
LACTCANA	RUBUIDAE	EPILCILI	BOTDMATD	FRAXAMER
	RUBUOCCI	EPILCOLO	BOTRVIBC	FRAXNIGR
LACTOCAN LEEDODV7	RUBUPUBE	EOUIARVE	DOMINED	FRAXPENN
	SALIERIO	ERYTAMER	CAREALOR	GALIOBTU
LEONCARD	SAMBRAPU	FRAXAMER	CAREBERR	GALITRIF
	SCIRATRO	GALIOBTU	CAREBLAN	GENTANDR
1 VCOUNTE	SCUTLATE	GLYCSTRI	CAREBOEN	GEUMSP
	SOLADULC	IMPACAPE	CARECRIS	GLYCSTRI
MALACANIA	SOLIALTI	LEERORYZ	CAREDEWE	HIERCAES
MATTSTOL	SOLICANA	LYCOUNIF	CAREGRAC	HYPEPERF
MEDILIPII	SOLIFLEX	LYSITERR	CAREGRAN	LACTCANA
MEDILOTO	SOLIGIGA	MENTARVE	CARELANT	LYCOTRIS
MENICANA	SONCOLER	PARTINSE	CAREEANS	LYCOUNIF
MENTADVE	SPHEINTE	PILEPUMI	CAREPENS	LYSICILI
MINIMENI	TARAOFFI	POA PALU	CAREPROL	MAIACANA
ONOCSENS	TIARCORD	POLYPUBE	CARERADI	MATTSTRU
ONODACAN	TRIFREPE	RANUACRI	CARESTIR	MEDILUPU
ONUTACAN OSMOCIAV	TRILEREC	RIBEAMER	CAREFENE	MELIALBA
OSTRVIDC	TRILGRAN	RUBUPUBE	CAREVILLE	MELIOFFI
OYAL STPI	TSUGCANA	SALIERIO	CARE \$70	MENTARVE
DANI CD	ULMUAMER	SCUTLATE	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	MUHLFRON
cant_or	UNKN_822		CARTCINU	MUHLMEXI

ODEN			E UD + EUG =	
OPEN	TRACOUS	FULTPERS	EUPARUGO	ACERRUBR
CANOPY	THEEPE	SULIGIGA	FRAXNICR	ASTELATE
33. (Cont'd)	LUSSPARF	AGKUGIGA	GALIOBIU	ASTEPONI
ONOCSENS	ULMUANER	BUCHUYLI	GEUM_SP	AIHYFILI
OXALSTRI	VERBINAP	COKNSTOL	GEYCSTRI	BEICALLE
PANIACUM	VIBULENI	EQUIARVE	HYPEPERF	BIDEFRON
PARTINSE	VIBUTRIL	EUPAMACU	HYSTPATU	BOEHCYLI
PHALARUN	VICICRAC	GALIOBIU	IMPACAPE	CAREBEBB
PHLEPRAT	VIOLAFEI	GLYCSTRE	INULHELI	CAREGRAC
PHRYLEPT	VIOLBEAN	IMPACAPE	LAPOCANA	CARESTIP
PILEPUMI	VITIRIPA	LEERORYZ	LEERORYZ	CIRCLUTE
PLANLANC	•· •· ·	PARTINSE	LYCOUNIF	CLEMVIRG
PLANMAJO	34. Riparian	POA_PALU	LYSICILI	CORNSTOL
PLANRUGE	Marsh		LYSINUMM	EPILCOLO
POA COMP	(n=46 species)	35. Riparian	MATTSTRU	EPIPHELL
POA PALU	AGROSTOL	Meadow	MENTARVE	EQUIARVE
POA PRAT	ASTELATE	(p=75 species)	ONOCSENS	ERIGPHIL
POPUBALS	ATHYFILI	ADIAPEDA	PARTINSE	EUPAMACU
POPUTREM	BIDEFRON	AGROGIGA	PILEPUMI	FRAXPENN
POTERECT	BROMINER	ARISTRIP	POA_COMP	GALIOBTU
PRUNSERO	CARDPENS	ASTELANC	POA_PALU	GALITRIF
PRUNVIRG	CICUMACU	ASTELATE	POA_PRAT	GEUMCANA
PRUNVULG	CIRCLUTE	ASTEPUNI	PRUNVULG	GEUMSP
PTERAOUI	CLEMVIRG	ATHYFILI	RANUACRI	GLYCSTRI
OUERRUBR	CORNRUGO	BIDEFRON	RHUSRADI	IMPAÇAPE
RANUACRI	ERIGANNU	BOEHCYLI	RIBEAMER	PARTINSE
RHAMCATH	EUPAPERF	CAREBEBB	RIBECYNO	PILEPUMI
RHUSTYPH	GEUMLACI	CAREGRAC	RIBETRIS	POA_PALU
RIBECYNO	IRIS SP	CAREPROL	RUBUOCCI	POA_PRAT
ROBIPSEU	LYCOUNIF	CARERADI	RUBUPUBE	RANUABOR
ROSAPALU	MATTSTRU	CARERETR	SALIERIO	RANUACRI
RUBUIDAE	MENTARVE	CAREROSE	SCUTLATE	RANURECU
RUBUPUBE	PILEPUMI	CARESTIP	SOLADULC	RIBETRIS
SALIBEBB	POA COMP	CARETRIB	SOLICANA	RUBUPUBE
SALIDISC	RIBERUBR	CAREVULP	SOLIGIGA	SOLADULC
SALIERIO	RUMEORBI	CIRCLUTE	SPHEINTE	TARAOFFI
SALIPETI	SCIRATRO	CLEMVIRG	THUJOCCI	VIOLAFFI
SCIRATRO	SOLADULC	CORNELIGO	TIARCORD	
SILEVULG	SOLIALTI	CORNETOL	TILIAMER	
SISYMONT	SPHEINTE	CYSTRUL B	ULMUAMER	
SMIL HERB	VIOLCUCI	DRYOCART	VIOLAFFI	
SOLIALTI	VITIRIPA	DRYOCRIS	VIOLCUCU	
SOLICANA	ASTEPIINI	EPH COLO	VIOE_SP	
SOLIGIGA	CARESTIP	EDIDWET I	VITIRIPA	
SOI INFMO	CVSTRIBR	SOLIARVE		
TARAOFFI	FPILCOLO	FRICANNU	36. Rinarian	
THUDCCI		FURAMACU	Thislast	
THIAMEP	ONOCSENS			
TERMEN	ONOCJENS	EUFAFERF	(n=57 species)	

APPENDIX 6

DISTRIBUTION OF SPECIES BY MICROHABITAT: CLOSED CANOPY

Legend

Column Heading	Description	Unit
dFLR	seasonally dry forest floor	# quadrats
mFLR	seasonally moist forest fluor	# quadrats
wFLR	seasonally wet forest floor	# quadrats
mDEP	seasonally moist depression	# quadrats
wDEP	seasonally wet depression	# quadrats
SEEP	seep	# quadrats
dGAP	seasonally dry gap	# quadrats
mGAP	seasonally moist gap	# quadrats
MOUN	tip-up mound	# quadrats
dPIT	seasonally dry tree pit	# quadrats
mPIT	seasonally moist tree pit	# quadrats
LOG	log on dry, moist or wet substrate	# quadrats
STMP	stump on dry, moist or wet substrate	# quadrats
RMAT	raised root mat on dry, moist or wet substrate	# quadrats
STON	stone	# quadrats
LANE	lane or access road	# quadrats
DITCH	ditch associated with lane/access road	# quadrats
F	regenerating farm field	# quadrats

SPECIES	dP1.R	mFLR	wFLR	mDEP	WDEP	SEEP	dGAP	mGAP	MOUN	TIAL		1.00	STMP	RMAT	ST'ON	LANE	DITCH F
		•			 -		ſ		r	_		.		t	.	_	
ABGEBALS	•	c	·	•	-		•	-		-		3					•
ACERNEGU		_	ı			,								,	,		• •
ACERRUBR	د.	5	_	-		ı	2	_	ç	-			5	ç		_	-
ACERSACC	_	æ		ŝ	en.		ŝ	L 1	5	r^.			ณ	_	ı	_	-
ACERSACN	ŝ	~	~	4	9			ר ו				_		2		ı	-
ACERSANI	• ∩	_		,	ı				_		ı		4				•
ACERSPIC	. –	- 20		6	r n		_	F -1	~		ı	20	~	_			
ACHIMILL	,	cı	_	ı	ı					,			·	ı			۰ ۲
ACTAPACH	_		ı				-1		4	ı	•						•
ACTARUBR	_	2	Ł	ŝ			-		-		ı			<u> </u>			-
ACTA SP	2	_	ı		_		-1	ı	m,	_	ı	_		Ŧ		_	'
ADIAPEDA	-	_			_		_		2		ı				1		
AGRIGRYP	4	2	,	,	,	1					ı	,	ı			,	-
AGROGIGA		r.			_						,	ı		ı		ı	-
AGROSTOL.		,	,					,		ŗ	ı	ı		,		ı	-
ALI.ITRJC	~1						_	,	¢1					_		ı	,
ALNUINCA	,				_								,	,			•
AMBRARTE						ı			•				,			,	•
AMELARBO	,	-	,			ı							ı				• •
AMEL,INTE	e							ı									-
AMEL SP	4		,				-1	ı	7	2			ŀ				, ,
AMPHBRAC		-	-		c 1												•
ANEMCANA		1	ı						,		1		ŀ	4			
ANEMQUIN			ı									_					•••
ANEMVIKO	1	_	ı						r	ı	,					,	۔ 4
ANTENEGL		,	ı						,						L		• •
APOCANDR	∞	_	ı		,		_		\$		ı		,				-
ΑΟΠΟΑΝΑ	œ								•0		ı						-
ARALNUDI	61	Ś	ı	-			,	_	m	m .	ı	ŝ	Ţ	_	1	-	•

441

ł

:

i

.

.....

.

:

:

.

:

SPECIES	dFLR	inFLR	wFLR	mDEP	WDEP	SEEP	dGAP	mGAP	MOUN	dPlT	mPIT	LOG	STMP	RMAT	STON	LANE	DITCH	F
ARCTMINU	3	1				-	_	-	-	-	-	-		2		•	-	-
ARISTRIP	6	1	-	2	9	-	2	3	1	7	-	6	2	1	-	5	-	-
ASARCANA	2	2	-	I I	2	-	-	•	6	6	-	-	-	2	-	-	-	-
ASCLINCA	-	-	1	-	I I	-	-	-	-	-	-	-	-	-	-	-	-	•
ASCLSYRI	2	2	•	1	-	-	1	-	I I	-	-	-	-	-	-	-	-	3
ASTECILI	9	-	1	-	-	-	2	-	3	-	-	-	-	-	-	-	-	4
ASTECORD	8	-	-	-	I I	-	1	-	I I	-	-	-	-	1	-	-	-	•
ASTEERIC	-	-	-	-	-	-	-	-		-	-	-	-	-	-	-	•	•
ASTELANC	-	-	-			-	-	-	-	-	-	-	-	•		-	•	•
ASTELATE	2	I I	2	I	3	-	I I	-	7	-	-	7	2	7	I I	1	2	4
ASTEMACR	6	-	•	-	-	-	-		2	1	-	-	-	•	-	-	•	•
ASTENOVE	3	2		-	1	1	-	-	3	-	-	-	•	1	-	•		2
ASTEPUNI		4	-	i	1	-	-	-	-	-	-	-	-	-	-	-	I I	3
ASTEUMBE	-	-	1		-	-	-	-	-	-	•	-	-	-	-	-	-	-
ATIIYFILI	3	1	-	4	3	-	+	1	6	2	-	I	-	7	-	2	-	•
ATHYTHEL	5	-	-	-	2		-	-	1	-	•	-	-	-	-	-	•	•
BETUALLE	6	2		-	2	-	-	-	-	-	-	I I	3	6	•	•	-	-
ΒΕΤΟΡΑΡΥ	9	1	-	-	-	-	2	-	-1	-	-	4	•	I .	-	-	-	\$
BIDEFRON	3	1	1	-	1	-	-	-	-	-	-	5		2	-	•	-	-
BOEHCYLI	6	4	3	2	7	1	•	1	-	1	-	3	I I	3	-	2	•	-
BOTRMATR	1	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	•
BOTRMULT	1	-	-	-	-	-	-		-	-	-	-	-	-	-	-	-	•
BOTRVIRG	3	1	-	-	-	-	2	-	4	2	-	-		-	-	1		Ι
BRACEREC	3	-	-	-	-	-	-			-	-	-	-	-	-	-	-	•
BROMINER	-	-	-	-	-	-	-		-	-		-	-	-	-	-	•	1
CALTPALU	-	-	-	1	-	1	-			-		-	-	-	-	-		•
CALYSEPI	-	-	-	-	-	-	-		-			-	-	-	-	-		-
CARDACAN	•	-	-		-	-	-	-	-	-	-	-	-	-	-		-	-
CARDDIPH	э	4	-	8	3	-	I	-	5	-		1	-	2	-	I	-	-
													-			··· ··		

.

Appendix 6. Distribution of species by microhabitat: closed canopy.

..____

SPECIES	dFLR	mFLR	wPLR	mDEP	d:ICI ∾	SIEP	dOAP	mGAP	MOUN	dPIT	T29m	1.06	STMP	RMAT	STON	LANE	DITCH F
CARDNUTA		: 1 '	,	,		•		•	ŀ								
CARDPENS		_	,	,	ı			,				ı					•
CAREALBU	en.				1			,				ı					
CAREALOP		_	_							,							•
CAREARCT	6 .1	~				,			-	,		ı		Ċ1		c :	•
CAREBACK	ŝ								¢.	,				,			•
CAREBEBB		-	_	-1				_				,		1		1	•
CAREBLAN	r.	Ś	1	-7	_		- 1	_	s			C1	,	m		<i>с</i> і	-
CAREBREV		_	Ч	,								ŀ		1		1	•
CARECEPH		Ч			_					,		ı		1			•
CARECOMM	<u>م</u>									Ŀ		ı		1			•
CARECRIN			,						,	,		,					•
CARECRIS	_	r.	~ 1	¢1	_	·			ı			CI	_	-			-
CAREDEWE	-7	¢		7			_	_	_	_		~	_	_		-1	'
CAREDIG	r.		,				_		_		ŀ					,	
CAREGRAC	-	-	_	Ó		,	C1		_	,		7	_	7		~1	-
CAREGRAN	_		_	_													-
CAREHIRT	ę	4			~					,				ı			•
CAREHITC	ŕt	,			,				_	r						,	•
CAREINTU	••	_		~1	ч	_			C 1		_			_		_	•
CARELANU	ı	_	_	r	,	,											-
CARELAXI	•	ı			,				_								•
CAREPECK	~1				,									_			•
CAREPEDU	4	_		S	_		~1	r)	_	2		¢	5	_	_		•
CAREPENS	×	v.		. .	,		-1	ı	~ 1	_		÷۱	~~;	~	-	r-	
CAREPLAN	7	ı			_		-		_					_			•
CAREPLAT	_																•
CAREPRAL		,		,	,	,				,					1	•	•
CAREPROJ		ı	,	-1	ı				•							_	'
:	· · ·	;								:							

SPECIES	dFLR	nFLR	wFLR	mDEP	wDEP	SEEP	dGAP	mGAP	MOUN	TIdb	mPIT	1'0C	dWIS	RMAT	NOIS	LANE	HOLICH	ί τ
CAREPSEU	,	.			: . 1			.						1			.	۰ ا
CARERADI	_	7	61	×	_		~	1	_		ı	CI		_	4	_	-	-
CARERETR	,	ı	•	-	Ś		·											ı.
CAREROSE	ъ	7	_	r-i		,	÷	-		,	,	r.		v .		61	_	ı.
CARESPAR	61	_			ı		_								E	ŀ		
CARESTIP		cı	¢;	_	,	-	,	,			,	1		_	·		c 1	
CARETENE	(°)	5	-	-1	~1							3		5		_	_	÷,
CARETRIB		-	-		-						ı			ı		_		
CAREVULP		ĉ	¢1	٢							,		L	ı		,	_	•
CAREWOOD	-	ı																
CARE 719	·	ı														F		-
CARE 868	ŀ	ı	ı									,						
CARE 870	,	ı	ı		,											,		
CARE 879	,	ı		,								,			r	,		
CARE OV	_	ı				_		,					-	_				•
CARE SP	~1	•				ı										_		ı.
CARPCARO	_	rı		ı	_		~ 1			F 1		_						Ċ,
CARYCORD	<u>ب</u>	ĊI		ı			-		_	Ś		-		Ŧ		-1	_	Ċŀ.
CAUL/THAL	4	-	ı	e:	¢1			_	_	\$		_		r.		_	,	•
CEANAMER	,			,			_											ı.
CELASCAN	~ 1		ı	ı			~		~ 1	1				_	ŀ			ы
CERAFONT	_	-	-	ı							,	1	ı				,	_
CHIMUMBE	~1	ı		•		ŗ			1			1						ı.
CHRYLEUC	,	ı	ı			,	,	,	1			ı						÷
CICUBULB		ı	ı		_													٢
CICUMACU		F				_			•	,								
CINNLATI	_	ı	ı	τ·ι	÷	_		_				m		•				
CIRCALPI	_	9	ı	~	c i		,	0	r.	,	_	6	-1	e,		_		
CIRCLUTE	ŝ	-	ı	_	~1	•		•73	_	-7	_	9	_	Ŀ.		Ś		
:				:											;	:		1

:

SPECIES	dFLR	mFLR	wFLR	mDEP	wDEP	SEEP	dGAP	mGAP	MOUN	dPtT	mPIT	LOG	STMP	RMAT	STON	LANE	DITCH	F
CIRSARVE	-	1		1	-	-	-	-	-	-	-	-	-	•	•	-	-	I
CIRSVULG	-		-	-	-	-	-	-	-	•		-	-	-	•	-	-	-
CLAYCARO	9				-	-	-	-	2	1		-	-	-	-	-	I	-
CLEMVIRG	-	5		3	-	-	-	-	-	-		Ł	-	2	-	-	-	-
CLINBORE	I I	1			-	-	-	-	-	-	-	-	-	Ł	•	-	-	-
CONYCANA	-	-			-	-	-	-	-	-		-	-	-	-	-	-	-
CORNALTE	6	1		1	2	-	3	2	I	1	-	3	-	8	-	1	-	-
CORNFOEM	1	t	-	2	-	-	1	-	•	-	•	1	-	-	-	-	-	•
CORNRUGO	1	i i	-	-	2	-	2	-	6	3	-	-	-	1	-	-	I I	•
CORNSTOL	3	2	2	3	3	-	-	1	-	-	-	-	-	1	-	-	2	3
CORYCORN	3	L		-	-	-	3	-	7	4	-	-	-	-	-	-	-	ł
CRAT_SPI	3	-	-	•	-	-	-	-	-	•	•	-	-	-	-	-	-	-
CRAT SP2	1	-	-	•	-	-	-	-	-	-	•	-	-	-	-	-	-	-
CRAT SP3	1	-	-	-	-	-	-	-	1	•		-	-	-	-	-	-	-
CRATSP	1	1	1	-	-	-	I I	-	2	•	•	-	-	•	-	-	-	2
CRYPCANA	1	-	-	-	-	-	-	-	-	-	•	-	-	-	-	-	-	-
CYPRCALC	-	2	-	-	-	-	-	-	-	-	•		•	-	-	-	-	-
CYSTBULB	2	4	-	9	+	-	-	1	-		-	3	•	3	-	-	-	-
CYSTFRAG	2	-	-	-	-		1		-	-	-	-	-	1	-	-	-	-
CYSTTENU	1	-	-	-	-			-	-	-	-	-	-	-	-	-	-	•
DACTGLOM	2	-	•	1	-		-	•	-	-	-	-	-	-	-	-	•	•
DANTSPIC	2	-	-		-		•	-	1	-	-	-	-	-	-	•	•	•
DAUCCARO	-	-	-	•	-	-	-	-	-	-	-	-	-	-	-	•		L
DESMGLUT	4				-	-	2	-	1	1	-	-	•		-	-	-	L
DIANARME	-	-		•	-	-	-	-	-	-	-		•		-	-	-	-
DICECANA	1	•			-	-	I I	-	2	2	-	-	-	•	•	•	-	-
DICECUCU	3	-	-	-	-	-	-	•	1	-	•	-	-	-	-	-	-	•
DIERLONI	1	-	-	•	-	-	-	-	I .	-	-	-	-	-	-	-	-	-
DIRCPALU	5	-			-	-	1	-	-	-	-	-	-	•		•	-	-

Appendix 6. Distribution of species by microhabitat: closed canopy.

PECIES	A FLR	mFLR	wFLR	mDEP	wDEP	SEEP	dCAP	mGAP	MOUN	TIPL	mPFF	D0.1	- IMIS	RMAT	STON	LANE	DITCH F
DRYOCART	 •∩	 _	•	+	י רו		- n	Гі	-			_	6	_		Ψ.	•
ORYOCRIS	J	<u>.</u>		e	6 1			_	'			~1		~	_	_	•
RYONTE	~	Ľ~		6	_	,	r	e,	-				Ŧ	-		мд.	•
RYOMARG	-	3 9			_	ı	_	_	~	_		ı		•3			•
CHILOBA			ı		·								ŀ				•
JLYMREPE				_													• •
ELYMVIRG	1	¢١	r.i	Π,		ı	ı	ŗ		_	,		,				-
SPIFVIRG	m	,		ı			,		_		,		1				
EPILCILI	·	2		E	_	,	,		'				,				
EPILCOLO	ı	Ŧ		Ţ.	_	_	,		_			7		. .			
EPILI,EPT	,	_	,	1			,				,	_		,			
EPILPARV	,	_	,	,			,	_	'		1					ı	
EPIPILELL					2		m	2	-	~		، ۳۱		_	_	r~-	_
COULARVE	-	r-	2	F	6 71	~1	,				,	_	ı	_			сı
зо́инуем	ŝ		,	ı		,	,					,		L	_		•
EQUILAEV	-						,									, .	' '
SQUISCIR	'n		,			,							·	c 1		_	•
GRIGANNU	~1	,		-	ı			,	_				ı				
ERIGPHIL		20	-	e de la constante de la consta	,		ı					ı	ı	_	_		- ·
FRIGSTRI		-1	ı		ı							1	ŀ			,	
ERIG SP	•		,	_			,		•							, 1	•
ERYTAMER	70	4		_	ı		_	~1	-	s.	_	_	,	,		-	· · ·
EULAMACU	ı	Ŧ			-		ı	_									_ ·
EUPAPERF	ı	_		,	,		ı	,	•								_
SUPARUGO	÷	rı		ŝ	Ś		ı					L.1		,			•
EUTHGRAM					ı				•		,		·		,		-
FAGUGRAN	4	m		_			,		Ś	_			_	~1	,	-1	
PESTARUN	ı	ı				,	,		•						,		'
FESTPRAT	ı			,		۲							ı		ı		
		ļ						•		:	:		!		:		:

:

i

. !

SPECIES	d£LR	mFl,R	wFLR	mDEP	wDEP	SEEP	dGAP	mGA₽	MOUN	dPIT	mPlT	1.0G	STMP	RMAT	STON	J.ANE	DILCH 1	2
FESTRUBR	-	-	_	•	•	-	-	-	-	-	_	-	-	-	-	-		
FESTSUBV	2	1	-	-	1	-	1	-	3	I I	-	-	-	-	-	-		
FRAGVESC	I I	-	-	-	-	-	I I	I I	2	-	-	2	-	-		I I		
FRAGVIRG	3	1	1	4	-	-	1	-	-	-	-	1	i i	4	-	-	1 4	Ļ
FRAXAMER	7	3	-	1	1	-	4	-	1	5	-	1	-	1	1	5	- 2	2
FRAXNIGR	1	1	2	1	3	1	-	2	2	-		1		2	-	-	<u>2</u> l	L
FRAXPENN	2	9	2	I I	3	-	-	I I	4	-		1	-	3	-	-	2 4	Ļ
GALESPEC	1	-	-	-	-	-	1	•	-	-	-	-	-	-	-	-		,
GALETETR	-	1	-	-	-	-	-	-		-	-	-	-	-	-	-		
GALIAPAR	8	-	-	-	-	-	-	-	1	-	-	-	-	-	-	-		
GALIASPR	-	1	•	-	-			-		-		1	•	1	-	•		
GALICIRC	2		-	-	-		-	•	-	-		-	-	-	•	-	- 1	l
GALILANC]	-	-	-	-		-	•	-	-	-	-	-	•	-	1		
GALIOBTU	1	3	2	l	3	-	-		-	-	-	-	-	1	-		L 3	ł
GALIPALU	-	-	-	-	-				-	-	-	1	-	1	-			
GALITRIF	5	9	I I	6	2		4	1	1	2		6	-	7	1	3	3	;
GALI SP	1	-	•	-	-	-	-		-	-	-	-	-	-	-	-		
GENTANDR	-	2	1	-	-	-	-	-	-	-	-	-	-	-	-		- 2	ł
GERAMACU	8	-	-	-	-	-	2	-	-	-	-	-	-	-	-	-		
GERAROBE	1	1		1	ч	-	1	3	4	1	-	8	2	5	-			
GEUMALEP	I I	-	-	-	-		-	-	-	-	-	-	-	-	-			
GEUMCANA	3	7	-	2		-	-	1	-	-		-		I	-	-		
GEUMLACI	4	7	-	2		-	1	4	1	-		1	•	1	-	-		
GEUMRIVA			-	1		2		-	-			-	•					
GEUMURBA	1	-	-		-		-	-	-	-	-	-	-					
GEUM SP	5	7	1	2	-		1		-		-		-	1	-	•		
GLYCSTRI	2	I	J	I I	7	I I	1	2	4	-	-	5	-	3	-	2	1 -	
GYMNDRYO	1	3	-	1		1	-	-	5	2		I I	1	в	-	-		
HACKVIRG	-	-			-	•	-		-	•		-	-	-	-	-		

Appendix 6. Distribution of species by microhabitat: closed canopy.

SPECIES	dFLR	mFLR	wFLR	mDEP	WDEP	SEEP	dGA₽	mGAP	MOUN	dPIT	mPIT	LOG	STMP	RMAT	STON	LANE	DITCH	F
REPAACUT	2	-	-	1	-	-	-	-	4	2	_	1		-	-	-		
HIERAURA	I I		-	-	-		-	-	-	-	-	-	-	-	-	-	-	-
HIERCAES	2	1	1	3	1	-	1	-	8	-	-	8	-	ā	-	I I	-	2
HYDRVIRG	9		-	-	1	•	-		3	I	-	•	-	I I	-	2	l	-
HYPEPERF	1	1	-	-	-	-	-	-	-	-	-	-	-	-	-		-	4
HYSTPATU	-	-	-	ł	-	-	-			-	-	-	-	-	-	-	-	-
IMPACAPE	2	1	3	8	5	3	2	2	5	3	1	1	6	I I	-	2	i i	1
INULHELI	-	-	-	-	-	-	-			-	-	-	-		-	-	-	-
IR/SVERS	-	-	-	-	-	-	-	-	-	-			-	-	-	-		-
IRIS SP	-	-	2		1		~		•	-	-	-	-	-	-	-	-	Т
JUNCTENU	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
LACTCANA	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	١.
LACTSERR	-		-	-	-	-	-	-	-	-	-	-	-	-	•	-		-
LACT SP	-		-	-	-	-	-	1	-	-	-	-	-	1	•	-		-
LAPOCANA	8	2	-	ł	4	1	-	1	-	-	-	I I	-	-				-
LEERORYZ	-	-	-	-	-	-				-		-				•	l	-
LEERVIRG	-	-	-	-	-	-	-	-	-	-	-	•					•	-
LEONCARD	-	-	-	-			-		1	-		-				•	-	-
LIPALOES	-	1	2	-	I I	-	-	-	-		-	-	•				-	Т
LOBEINFL	-	-	-		-	-	-				-	-		-	-	-	-	-
LOBE SP	-		-		-					•	-	-	-	-	-	-	-	-
LONICANA	2		-		•	-	1	-	4	3	-	-	-	2	-	-	-	-
LONIDIO	1		-	•	-		1	•		-	-	-	-	-	1	-	-	-
LONHIRS	I I	-	-		-	-						-	•	-	-	-	-	-
LYCOAMER	-	I.	1	•	-	l			-	-	-	-	-	-	-	-	-	Т
LYCOANNO	5	-	-		-	•					-		•	1	-	-	-	-
LYCODEND	1	-	-	-	-	-	-	-	-	-	-	-				l	-	-
LYCOOBOB	3	-	-	-	-	-	-	-	1							-	-	-
LYCOTRIS	-	-	-	-	-	-	-	-	-	-	-	-	-		-		-	-

Appendix 6. Distribution of species by microhabitat: closed canopy.

_____ · . .

								'		 							
SPECIES	AFT.K	mFl.R	wFL.R	mDEP	WDEP	SHEP	dGAP	mGAP	MOUN d	IPJT 	mPTT	1.06	STMP	RMAT	ST'ON	LANE	DITCH F
TYCOUNIF	1	-	- ~	. 1	_	÷1			•		1	Ŧ					-
TYSICILI	ı	য	2	C1		_										,	-
WW/INISAT	_	_	cı	_	~1	~						_			r	ı	•
1.YSITERR	,	,							'							ı	•
MAIACANA	6		ŀ	~~	ı		ъ.	_	-			_	_	_	_	Ś	1 2
MAIARACE	۲.	Ŷ	ı	-	1		2			_		ŝ	_	-		ŝ	•
MAIASTEL	÷	ci L		_			_		-				1		,		•
MATTSTRU	¢	7	_	ĉ	_	J.							1				
MEDILUPU	7								•				ı	_	ı		ہے –
MELIALBA		,							•						1		
MELIOFFI				ľ					·	-							-
MENICANA	_			ı			_			-			ı				•
MENTARVE	_	•••	_		_							_	,				•
MILLEFU	~			•		ŗ			•			_		_			
MUTCREPE								.,								_	
MITEDIPH	_	~1		_		,			-	-	,	-7		-			
MONAFIST				ı					· _	-							,
ONTYHONOM	-1		E	ı						-					,	,	•
MONOUNIF	I		,			,		·	'	-					1		
MUHLFRON		ı								-							 ,
MUHLMEXI			_			1			•	-							
NEPECATA		•		r				•	•	-							
ONOCSENS	6	2 0		¢	9	61	_	_	·	-		_				_	~
ONOPACAN	•					,	•		'	-							•
ORYZASPE	4		ŗ					-	_				ſ			m 3	•
OSMUCLAY	۴I						_	-		·					ŀ	_	•
OSMUCLAY	4	,							'							_	•
OSMUREGA	•		,	ı	_										1		
OSTRVIRG	9	-		_	1	•	m		*	-				_		r 1	•

440

i

:

DITCH F **C**1 . с1 L ı ı . I STMP RMAT STON LANE 1 ļ : ٦ . 00,1 . mPU' ı **JGAP INGAP MOUN dPTT** . ı Appendix 6. Distribution of species by microhabitat: closed canopy, . . . ı SEEP mDEP wDEP e t mFLR wFLR I ÷ dFLR i ÷ OXAL_SP PANAQUIN POLYACRO PHALARUN PHEGCONN PLANMAJO PLANRUGE POA_COMP PANIACUM PLANLANC POA_ALSO PODOPELT P111.EPRAT POA_PALU PHRYLEPT PICEGLAU PINUSTRO POA_PRAT POA SALT PARTINSE PILLO_SP POA___SP **OXALSTRI** PANICAPI PANL_SP PIL.EPUMI SPECIES

65 .

.

I

.

i

PN :

C1

CI.

rt I

177

POL YPAUC

POLYPERS

POLYPUBE

450

SPECIES	dFl.R	mFLR	wFLR	лıDEP	WDEP	SLEP	dGAP	mGAP	MOUN	dPlT	inP17	LOG	STMP	RMAT	STON	LANE	DITCH	F
POPUBALS	4	3	•	2	-	-	-	1	•	•	•	-	-	1	-	-	-	
POPUGRAN	1	-	•	-	-	-	-	-	3	2	•	-	•	-	-	L	-	-
POPUTREM	2	2	I I	-	-	-	-	-	1	-	•	-	-	-	-	1	-	3
POTENORV	-	-	-	-	-	-	-	-	-	•	•	-	-	-	-	-	-	•
POTERECT	-	-	-	•	-	-	-	-	-	-	-	-		-	-	-	-	3
PREN SP	1	-	-		-		-	-	4	-	-	-		1	-	-	-	-
PRUNSERO	5	5	-		-		3		4	2	-	-	-	ī	-	2	•	2
PRUNVIRG	9	6	2	3	2		3	2	ł	6	-	-	-	4	1	4	1	3
PRUNVULG	2	3	2	I I	-	-	-	-	-	-	-			-	-	-	-	3
PTERAQUI	3	-			-	•	2	-	X	\$	-	•		-	-	I	-	2
PYROELLI	1	-	•	-	-	-	1	-	4	1		-	•	2	-	-	-	•
QUERALBA	8	-	•		-	-		-	-	-	-	-	1	-	-	•	~	-
QUERMACR	8	2	-	-	-			-	-	•	-	-	-	Ì	•	-	-	•
QUERRUBR	6	-				-	2	-	7	3	-	-]	-	-	4	-	1
QUER SP	-	-	-	-	-	-	-	-	1	-	•	-	-	-	-	-	-	-
RANUABOR	2	4	I	3	1	-	-	1	1	1		2	1	1	-	5	-	-
RANUACRI	9	7	I	3	-	-	I I	•	1	-	-	-	-	1	-		I	4
RANURICA	-	-	-	I	-	I		-	-	-	-	-	-	-	-	-		-
RANURECU	3	6	-	I	4	-	3	1	4	-	-	I I	-	2	-	-	I	-
RHAMALNI	1	-	-	•		-	•	-	-	-	-	-	-	-		-	-	-
RHAMCATH	3	6	2	4	2	-	3	2	Ó	-	-	4	-	6	•	2	-	4
RHUSRADI	3	2	1	3	1	-	2	-	3	I	-		-	3	1	1	-	-
RHUSTYPH	-	-	-	-	-	-	-	-	•	-	-	-	-	-	-	-	-	-
RIBEAMER	1	3	-	1	2	2	-	-	-	-	-	-	I I	-	-	-	-	Ι
RIBECYNO	5	9	-	7	1	•	4	2	5	2	-	1	-	ó	I I	4	-	Ι
RIBEGLAN	1	1	-	2	-	2	~	-	-	-	-	2	-	1	-	-	•	-
RIBELACU	-	-	-	1	-		-	÷	-	-	-	-	•	l	-	-	-	-
RIBERUBR	2	1	-	-	-	-	-	1	•	•	-		-	-	-	-	-	·
RUBETRIS	6	8	1	6	1		1	1	I		-	-	-	-	-	•	-	•

Appendix 6. Distribution of species by microhabitat: closed canopy.

. ... _____

SPECIES	ALIAb	mFLR	wFLR.	mDEP	q:ICIw	SEEP	dCiAP	InGAP	NUOM	TIdb	TIAID	100	STMP	RMAT	STON	LANE	DITCH F
UBE 827			. '		-		ı		,				ı		ŀ		'
tOBIPSEU	_	Ľ	2			ı			-						ı	_	_
NA BLAN	_	_								_							•
toSAPALU		_	_														•
NUBUALLE	_	_							~ 1	_		_	·	_			_ ,
NUBUIDAE	-	7		\$	-		-1	-1	~1			90	r-1	m			-1
UBUOCCI	9				_				_								•
NUBUODOR	9	r -1					,	,	_			m	_	_			• •
NBUPUBE	_	_	2	20	~		,	_	_			¢.	r h	×			
RUBU 840	r	ı					,	ı				_					•
UDBHRT	ı	,				,	ı										-
RUMEORBI	,	,			,		ı										
SALIBEBB	•	_	_			,	,					,	,	,			
BALIDISC		_		,							·	,					
SALIERIO	•	CI	_	·	-									,			<u>-</u> -
BALIPETI		,		·	,	,						,	,	E			•
SAMBCANA	÷1	,		^ 1		-		_			·	,	,	_			•
SAMBRAPU	~ 1	-					.	,	~	_	·		~1	س	,	-1	
ANGCANA	÷	ı					ı		4	÷							_
SANIMARI	~		ı					,	_			,					
SANJTRIF	-	,	ı	,						ı							
SANI SP	_	1	,		,					ı							•
SCIRPURP	ŝ	r.I		6 1	·			,	-7					c۱	_		' '
SCIRATRO		-	,		_			,				,					•
SCUTLATE	,	۲٦	_	_			,							_			'
SICYANGU	,			•													
SILEVULG	ı			•		ı		,	_	ı	,						•
TNOMYSIS	ı					,	,				,		,				
VAUSUAV	,	,		ŀ				,	,			,					•
				; ;	:					•	:						I

SPECIES	dF1,R	տԲԼԶ	wFLR	mDEP	wDEP	SEEP	dGAP	mUAP	NIOM	dufi'	1.J.đut	DOL	STMP	RMAT	STON	LANE	DITCH	<u>[1</u>
SMILHERB	- m	,	.		-		-									5	,	-
SMILHISP	_							,			ı							
SOLADULC	4	_	rı	ç	_	_	-1	¢1			,	Ŧ	-	ŝ				
SOUALTI	Ś	6 1	,	~1	_		<u>.</u>	_	_			Ē	,	,		_	_	ы
SOLICAES	6							ı	v :	•			,	,		,		
SOLICANA	, 1	5	_	e,	_		- 1	_	ŗ			6	-	7				r n
SOLIFLEX	m	_	,	3	•		-		5	-		¢,	_	. .1				ī
SOLIGIGA	F	сı	_	,	,			_									ŕt.	•••
SOL, IJUNC			,	ı			_											
SOLJNEMO	r,	ı	ı	ı	ı		,		¢1					ľ				Ċ,
SOLJRUGO	1	_	ı							,				_	ı			
SONCARVE		ı	,						_	r		ŀ						
SONCOLER	ı	•					,	,						,			,	E
SPHEINTE	(1 1	сı	,										_	C 1		ı		
STELLONG		_																
STREROSE	7								_				,					
SYMPALBU	νn								_	_			,					
TARAOFFI	┯	_		•	6 1		۲ı	_	_	-		6	Ċ)	_	ı	r.	_	m
TAXUCANA	_	£.							·					_	ı			
THALDIOI	_	_					_		_	,			,		4	_		
THALPUBE	,	_	ı					_				ı		_	ı	ı		
THEI.NOVE	_	ŗ	ı	_											ı			
THELPALD	-	Т	ı	_	_							,		_	,			
TILUJOCCI	2	6	,	va	_		F 1		7 1	_		'n		_	F	ı		-
TIARCORD	<u>-</u> 1	_		6	r-1	_	~1	~1	*0			cı	Ŷ	_	F	_		
NEMARK	r-		,		2		-+		r-	^ 1		cı.		•	,	C 4	_	m
TRAGDUBI	_	,	ı										ŀ	ı				
TRIEBORE	r #.	_		_					ŷ	en en		-	-+	-ī-				
TRIFREPE										ı							_	_

453

.

i

-

į

i

PECIES	dFLR	mFLK	wFL,R	mDEP	wDEP	SEEP	dAP	mGAP	MOUN	TIAD	ntPIT	10C	JMIS	RMAT	STON	TANE	DITCH F
"RILEREC	m	: ; ; ~			~	: : : ·	! :		6	_	I	-	,	_			
RILGRAN	-	C 1		m	-1		w.		сл)	_		,	ı	3	_	8	-
RIOAURA					ı				ŀ				ı				•
SUGCANA	ę			_	ı				Ŧ			ó	-	S.			•
USSPARF										,						,	' -
TVPHILATI		_			ı							,			ı		•
JLMUAMER	C 1	7	_	_	20	_		ጉና	_		·	Ŧ	-	ŝ		_	
151 NXN		_	,						ı								'
NKN 791		ı	,			_			ı				1			ı	
NKN 822			,		ŀ												'
RTIDIGR	_	ı												•		_	•
RTIDIO)		_											1		ı		
VULGRAN	0		ŗ		ı		ı		•						ı	•	
ACCANGU	7		r		ı		,		•		,		,	ı		1	
/ERBHAST	,											ı					
/ERBTHAP		•	,													,	•
/ERBURTI		ı	,					,									•
/EROOFFI	r)											,					
/EROSERP	-1		,		•			ı								,	•
/IBUACER	'n	,	1		,		_		6	-7						,	_ ` ,
TRUUENT		c 1	6 1	3	ı		~ 1		~	~ 1				¢1		,	' 4
/IBUOPUL	·	,			ı		,										•
VIBUTRIL.	<u> </u>				ı	ı										1	•
/ICICRAC			ı			ı			,	1						,	·
/IOLAFFI	4	J	2	el.		ı			_	,						,	•
VIOLBLAN	_	-	ı						2		ı	~~.		-1			•
VIOLCANA		-	ŀ	ı					E	1	ı					ŀ	•
norcocu	6	_	ı	ı	_			_	c 1		ı			ı		•	'
/IOLLABR	_	_	F	~1	ı	,	_				ı	-		I	ı	_	

454

I

:

.....

:

÷

PICIES	dPL.R	m.Fl.R	wFL.R	nDEP	wdep	SEEP	dADb	mGAP	MOUN	TIAD	TIAm	LOG LOG	STMP	RMAT	STON	LANE	DITCH
/IOLPUBE	s		 1	 ~	 ~	.		-		 ~		~1	- - - - -			-	_
/IOLROST	2	,		,	•		,		·			,			ı		
/JOJ.SORO	4		,							,	·	,	,	۴	,		
/101. 788						,				ŗ	,	,	,	ı	,	ı	
/IOL_SP	r)	_				-	_		-		,	_	,	-	,	,	_
/JTIRIPA	~~	9	_	Ţ	~ 1		-		20			-		_		_	_
VALDFRAG	_	_					_		ŝ	è.	,	_	,	ı			
APPENDIX 7

DISTRIBUTION OF SPECIES BY MICROHABITAT: OPEN CANOPY

Legend

Column Heading	Description	Unit
dFLR	seasonally dry forest floor	# quadrats
mFLR	seasonally moist forest floor	# quadrats
wFLR	seasonally wet forest floor	# quadrats
mDEP	seasonally moist depression	# quadrats
wDEP	seasonally wet depression	# quadrats
SEEP	seep	# quadrats
MOUN	tip-up mound	# quadrats
dP(T	seasonally dry tree pit	# quadrats
wPIT	seasonally wet tree pit	# quadrats
LOG	log on dry, moist or wet substrate	# quadrats
STMP	stump on dry, moist or wet substrate	# quadrats
RMAT	raised root mat on dry, moist or wet substrate	# quadrats
LANE	lane or access road	# quadrats
DITCH	ditch associated with lane/access road	# quadrats
RFLD	regenerating farm field	# quadrats
RMA	riparian marsh	# quadrats
RME	riparian meadow	# quadrats
RT	riparian thicket	# quadrats

SPECIES	dFLR	mFLR	wFLR	mDEP	wDEP	SEEP	MUUN	dPIT	wpur		STMP	RMA1	LANE	DITCH	KFLD	RMA	RME	RT .
ABIEBALS	 m	1					-	ι	: ,			· ·						.
ACERNEGU		,					,	ŗ		,						,	ı	
ACERRUBR	~	40.		,	ı		ς.								ন			_
ACERSACC		m		,			s	1 75		_	_	ন	Ψ	_	_		ı	
ACERSACN	_	4	ŝ	_			,	E	ı				_		-+		ı	,
ACERSANI		,															ı	
ACERSPIC	÷	9	_	CI	_		ı			Ŧ		•					,	
ACHIMILL		-	,				,					•	-	,	7	ı		
ACTAPACH	Ś	ı	,	,			_		ı			•				ı	ı	
ACTARUBR	r			,			_					•		,		ı	ı	
ACTA SP	_	_		,	ı							_	_				,	
ADIAPEDA	_	,		,	ı		_			ı		_					_	
AGRIGRYP		,		,				,		1					_	,		
AGROGIGA		_	-	,	ı		•					•	<u>م</u>	_		ŝ	~ n	
AGROSTOL							,			_			¢.			_		
ALLITRIC	4			,	,		ŀ			ı								
ALNUINCA		·		,			ı											
AMBRARTE				,			ı						C 1					
AMELARBO			,	,								•						
AMELINTE	_								ı								4	
AMEL_SP	_		,	,								•		•				
AMPHBKAC		- +		ı		ı			ı						_			
ANEMCANA		,		,		ı						-	_		m .		,	
ANEMQUIN	ı	•									,					ı		
ANEMVIRG	ı	~										-	_		6	ı	,	,
ANTENEGL	ı	ı	,	,								•		I	_	ı		
APOCANDR	,	,	,											1	_	ı		
AQUICANA	-	,												•	_	ı		
ARALNUDI	4										,	· -				I	I	

:

ŀ

:

:

i

1

i

•

ł

SPECIES	dFLR	mFL.R	wFLR	mDEP	WDEP	SEEP	MOUN	apr	wPfT	1.06	STMP	RMAT	LANE	DITCH	RFLD	RMA	RMÊ	RT
ARCTMINU	3	-	•			-	-	 •	-	-			!		-		-	-
ARISTRIP		9	4	3	3	I I	1	1	-	I I	1	4	1	-	-	-	2	-
ASARCANA	5	1	-	-	-	•	ļ	1	-	-	I.	-	•		-	-	•	-
ASCLINCA	-	-	-	-	-	-		-	-	-		•	-	-	1	-	-	-
ASCLSYRI	5	1	-	-	-	-	-	-		-		-	2	-	4		-	-
ASTECILI	3	1	-		-	-			-	-	-	-	1	I I	5	-	-	-
ASTECORD	-	-	-	-	-	-	-	-	-	-	-	-	1	-	-	-	-	-
ASTEERIC	-	-	-		-	-	-	-	-	-	-	-	-	-	1	-		-
ASTELANC	-	1	1	-	1	-	-	-	-	•	-	-	2	1	1	-	l	-
ASTELATE	3	5	1	1	-	-	1	-	-	3	1	3	3	1	4	1	2	1
ASTEMACR	2	-		-	-	-	•	-	-	•	-	-		-		-		-
ASTENOVE	1	1	-	-	-	-	-	-	-	-	-	-	3	-	4	-	-	-
ASTEPUNI	-	2	I I	-	-	-	-	-	-	1	I.	L	3	1	3	2	3	I.
ASTEUMBE	-		-		-	-	-	-			-	-			-	-	~	-
ATHYFILI	3	5	2	-	1	1			-	1		-	2	-	-	I I	2	1
ATHYTHEL	-	-	-	-	-	-	-	-	-	-	-	-	-			-		-
BETUALLE	-	1	-	-	-		-	-	-	2	-	1	-	-	-	•		Ι
BETUPAPY	2	-	-	-	-	-	-	-	-	1	ł	-	1	-	4	-	-	-
BIDEFRON	-	1	2	-	1	-	-	-	-	-	-	1	3		-	1	2	T
BOEHCYLI	I I	A	3	1	<u>ר</u>	1	-	-	-	I I	-	2	2	-	-	3	3	T
BOTRMATE	-	-	-	-	-	•	-	-	-	-	-	-		•	L		-	-
BOTRMULT	-	-	-	-	•	•		-				•	-	-	-	-	-	-
BOTRVIRG	4	-	-	-	-	-	-	-	-	-			1	-	ł	-	-	-
BRACEREC	2	-	-	-	-	-		-		-		-	-	-	-	-	-	-
BROMINER	-	-	-	-	-	-		-	-	-	-	-	2	-	ŧ	I	-	-
CALTPALU	-	-	-	-	-	-	÷	-	-	-	-	-	-	-		-	•	-
CALYSEPI	1	-	-	-	-		-	-	-	-	-	-	-	-				-
CARDACAN	I I	-	-				-	-	-	-	-			-		-		-
CARDDIPH	6	3	-	-	-	•	I	-	-	-	-	-	1	-	•	-		-

ANDNUTA ANDNUTA ANDFINS ANDFINS AREALBU 2 AREALEN 4 AREBENS 1 AREBUN 4 ARECUN 4 AREDUN 4 AREDUN 4 AREDUN 4 ARELANU 4 AREPLAN 4 AREPLAN	· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·	• • • • • • • • • • • • • • •			· · · · · · · · · · · · ·	
ARDFENS AREALBU 2 AREALBU 4	· · · · · · · · · · · · · · · · · · ·		. - -	· · · · · · · · · · · · · · · · · · ·			
AREALBU 2 AREALBU 2 AREALOP - - - - AREARCT 6 1 - - - AREBACK 2 - - - - - AREBACK 6 1 - - - - - AREBUSIS - 1 1 - - - - - - ARECHI - - 1 -	· · · · · · · · · · · · · · · · · · ·			•••••••••••••••••••••••••••••••••••••			
AREALOP - </td <td>••••••••••••••••••••••••••••••••••••••</td> <td></td> <td>· - · · - · · · · · · -</td> <td>. ~ ~ ~ ~</td> <td> – – –</td> <td></td> <td></td>	••••••••••••••••••••••••••••••••••••••		· - · · - · · · · · · -	. ~ ~ ~ ~	– – –		
AREARCT 6 1 7 AREBEJB 1 1 1 1 AREBEJB 1 1 1 1 1 AREBACK 2 1 1 1 1 1 AREBUAN 1 4 1 1 1 1 1 AREBURY 1 4 1 1 1 1 1 1 ARECUNI 1 1 1 1 1 1 1 1 1 ARECUN 1	· · · · · · · · · · · · · · · · · · ·			or→or≄ · · · · · · ·			
AREBACK 2 AREBACK 2 A AREBEUB 1 1 1 1 1 AREBUAN 1 1 1 1 1 1 AREBUAN 1 1 1 1 1 1 1 AREDUN - 1 1 1 1 1 1 1 ARECUNI - - 1 1 1 1 1 1 1 ARECUNI - - 1	· · · · · · · · · · · ·		· · - · · · · · · -				
AREBEJB 1 </td <td>· · · · · · · · · · ·</td> <td></td> <td>· - · · · · · · -</td> <td></td> <td></td> <td></td> <td></td>	· · · · · · · · · · ·		· - · · · · · · -				
AREBLAN 1 </td <td> </td> <td></td> <td></td> <td>• • • • • • • • •</td> <td></td> <td></td> <td></td>	 			• • • • • • • • •			
AREBREV I </td <td>· · · · · · · ·</td> <td></td> <td>· · · · · -</td> <td>• • • • • • •</td> <td></td> <td>4 1 1 1 4</td> <td></td>	· · · · · · · ·		· · · · · -	• • • • • • •		4 1 1 1 4	
ARECEPTI -<	· · · · · ·			••••			
ARECIMIM -<	· · · · ·	61	· · · -	• • • •			
ARECRIN - </td <td>· · · ·</td> <td>, , , ,</td> <td>· · -</td> <td>•••</td> <td></td> <td></td> <td></td>	· · · ·	, , , ,	· · -	•••			
CARECRIS 1<	· · ·	, ei	· -	• •	۴ı		
CAREDEWE AREDEWE A CAREDEWE - - CAREDEME - - CAREDEME - - CAREDEME - - CAREDEME - - CAREGRAN - + CAREGRAN - + CAREDENC - + CAREDENC - + CAREDENC - + CARELINT - - CARELAXI - - CARELAXI - - CAREDECK - - CAREPECK - - CAREPENS - - CAREPEN	• • •	۲٦ •	-				
AREDIGI - </td <td>-</td> <td></td> <td></td> <td></td> <td>•</td> <td>•</td> <td></td>	-				•	•	
AREGRAC + 2 2 - </td <td></td> <td>,</td> <td>•</td> <td>•</td> <td></td> <td>,</td> <td></td>		,	•	•		,	
CAREGRAN -<	•	•	۲ -	' +	21	ı	F 1
CAREILIKT ·		•		-	د،		
CARENTIC 1 3 1 3 1 3 1 3 CARENNU 1 1 3 1 3 1 3 1 4 4 4 4 3 1 3 1 4 4 5	•	•	•	•		,	
CAREINTU 1 3 1<	•	•					
CARELANU - 1 ·<	•		•	•	,		
ZARELAXI 2	•	•	•	-	J	•	
ZAREPECK	•	•	•	•	,		
ZAREPEIJU 5 6 · · · · · · · · · · · · · · · · · ·	•	•	•	•	,		
CAREPENS I I 2	•	• '	-	•	•		
CAREPLAN I	- 2 -	•	-	' r1	61		
CAREPLAT • • • • • • • •	•	•	•	1	ı		
		•	•	•	ı		
CAREPRAI	r	•	•	•	-		
CAREPROJ - I - I	•	•	•	' _	_		_

SPECIES	dFLR	mFt.R	wFI.R	mDEP	wDEP	SEEP	MOUN	dPIT	wPIT	LOG	STMP	RMAT	LANE	DITCH	RFLD	RMA	RME	RT
CAREPSEU	-		-	-	-		-		-	-		-	-	•	•	_		
CARERADI	Э	3	I I	1	-	-	2	-	-	2		-		•	2		2	-
CARERETR	-		3		2	I I		-	-	-	-		I.	-	-	-	2	-
CAREROSE	4	1	I I	-	-		L	-	-	-	I I	-	1	-	-	-	1	•
CARESPAR	2	-	-	-	-	-	-	-	-	-	-	-	-	-	•	-	-	-
CARESTIP	-	2	2	-	1	-	-	-	-		-	-	1	-	2	2	3	1
CARETENE	-	2	-	-	-	1	-	-	-		-	-	2	ł	4	-	-	-
CARETRIB	-	-	l			-	-	-		-		-	1	-	-	-	1	-
CAREVULP	•	1	۱.			-	•	-		-	1	-	2	-	1	-	1	-
CAREWOOD		-				-	-	-	-	-		-	-		-	-	-	-
CARE 719	-	-				-		-	-	-		-	-	-	-	-	-	-
CARE 868	-	-			-	-	-	-	-	-		-	2	÷	-	-	-	•
CARE 870	-	-	-		-	-	-	-	-	-			ł.	-		-	-	•
CARE 879	-	-	-				-	-		-			-		I I	-	-	•
CARE OV	1		-	-		•		-	-	-	•	-	-	-	-	-	-	-
CARE SP	Ł	-	-	-	-	-	-	-	-	-	-	1	1	-	-	-	-	-
CARPCARO	ł		-	-	-	1		-	-	-	-	-	-	-	-	-	-	-
CARYCORD	7	-	-	-	-	-	1	-	-	-			I.	-	1			-
CAULTHAL	6	I I	-	-	-	-	1	•		-	-	-	4	-	-	-	-	•
CEANAMER]	-	-	-	-	•		-	-	-	-	-	-	-	-	-	-	
CELASCAN	4	I I	-	-	-	•	-	-	-	-	-	-	-	-	-	-	-	-
CERAFONT	1	1	-	-	-	•		-	-	-	-	-	2	-	3	-	-	-
CHIMUMBE	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
CHRYLEUC	-	1	-	-	-	-	-	-	-	-	-	-	3	-	4	•	-	•
CICUBULB	-	-	-			-		-		-	-		-	•	-			-
CICUMACU	-	-	-	-		-		-	-	1			-	I	-	L C	-	-
CINNLATI	-	2	-	-	-	-	-			1		-	-		-		-	-
CIRCALPI	3	5	-	1	-	I I	-	-	-	3	2	4	ł		-	-		-
CIRCLUTE		8	1	3	-	-	2	ł	-	I	-	6	5	•	•	I	2	Т

arneina	dFLR	mFLR	wFLR	mDEP	wDEP	SEEP	MOUN	46LL	wPIT	LOG	STMP	RMAT	LANE	DITCH	RFLD	RMA	RME	RT
CIRSARVE	2	-	•	-			1	• ·	-	•	•	-	3	-	3			- <u>-</u> -
CIRSVULG	3	-	-	•	-	-	1	1	-	-	-	1	1	-	2	-		-
CLAYCARO	3	-	-	-	-	-	-	-	-	-	•	-			-	-		-
CLEMVIRG	1	•	I I	-	-	-	-	-	-	I I	-	2	1	-	-	1	2	I.
CLINBORE	-	-	-	-			-	-		-	-	-	-	-	-	-	-	-
CONYCANA	3	-	-	-	-		I I	-	-	•	•	1	1	-	-	-	-	-
CORNALTE	4	5		1	-		-	-	-	-	-	2	-	I I	-	-		-
CORNFOEM	1	-		-	-	-	I		-	-	-	-	-	-	-	-	-	-
CORNRUGO	4	-	2	-	1	-	1		-	-	-	-	2	-	I I	1	2	-
CORNSTOL	-	3	1	1	-	-	-	•	-	-	•	1	2	I I	4	3	3	I.
CORYCORN	4	-	-	-	-	-	-	-	-	-	-	-					-	-
CRAT SP1	-	-	-	-	-	-	-	-	-	-					-	-		-
CRAT SP2	-	-	-	-			-	-	-	-	-		-			-	-	-
CRAT SP3	-	-	-	-			-	-	-	-	-	-			-	-	-	-
CRAT [®] SP	3	2	-	-	-	-	-	-	-	-	-		-		4	-		-
CRYPCANA	1	I		-			-	-	-	-	-	-	-	-	•	-	-	-
CYPRCALC	-	-	-	-	-	-	-	-	-	-		•	-			-	-	
CYSTBULB	2	2	1	2	-	-			-	1		1	1		-	3	2	-
CYSTFRAG	-	-	-	-		•			-	-	•	-	-	-		-		-
CYSTTENU	-	-	-	-	-	-	1	-	-	-	•	-	-	-		-	-	-
DACTGLOM	1	-	-	-	-	-	-	-	-		•	-	3	-		-	-	
DANTSPIC	1	-		-	-	-	-	-	-	-	•	-	-	-	1		-	
DAUCCARO	-			-		-	-		-	-	-	1	2	-	2	-	-	
DESMGLUT	-	•	-	•	-	-	•	•	-	-		-	-	-	-	-	-	-
DIANARME	-		-	-		-	•		-	-		-	I I	-	I I	-	-	-
DICECANA	6	-	-	-	-	-	1	1					1	-	-	-	-	-
DICECUCU	1	-	-	-	•	-	-	-		-		-		-	-	-	-	-
DIERLONI	1	-	-	-	•	-	I I	-	•	-		-	-	-	-	-	-	-
DIRCPAI.U	2	•	-	-	-	-	-	-	-		-	•	•		-	•	•	-

SPECIES	dFLR		wFLR	mDEP	wDEP	SIEP	MOUN		wPIT	001	STMP	RMAT	LANE	DITCH	RFLD	RMA	RME	-
DRYOCART	•	9	_	-			_	. –	•	1	-	. 4	न				. กา	
DRYOCRIS	_	0			,				,	,	,	_					_	
DRYOINTE	9	9		_			63			m .	÷	8						
DRYOMARG	4	-			F	,						-				,		
ECHILOBA	,	ı			۰	,		,							,			
ELYMREPE	ı	ı	,		,					,	,		_		-			
ELYMVIRG	,		_		_	ı				_	,	,		ı	_			
EPLEVIRG	-		,	,	·	•	_	,				ı				ı		
EPILCILJ			,	ı	,	•						ı		_	_			
EPILCOLO	-1	ų ę ,	_				-	-		_		<u>م</u>	~	_		ŕ٦	~n	_
EPILLEPT			•			ı							ı		_	ŀ		
SPILPARV		2	-		_	,				_		_				,	,	
EPIPIELL		¥	-				_				_	5	-1		2	,	~	_
EOULARVE		-	54	ci i	_	ı		,				Č 1	m.	_	ó	~ 1	<u>.</u>	_
единуем			ı	,											_		,	
RQUILARV						ı			,				-	L			,	
EQUISCIR	,	4														ŀ		
ERIGANNU	сı	-	-	•		ı						_	_			_	_	ŀ
ERIOPHIE.	9		ı			,			,	ı	_	_	ŝ	ı	L 1	,	,	_
ERIGSTRI	ı			ı									m .		n i			
ERIG SP	ı			,		4						_	ı					
ERYTAMER	-	٩	,	_		•	_	_					·C.	_		,		
EUPAMACU	,	~	_			_		_		,		_	C 1	ı	2	m		_
EUPAPERF	_	,	_							,			_		_	_	ר יו	
EUPARUGO		,								,	,	,	2			ı	-1	,
EUTRIGRAM			۲	,		,	_			,					÷			
FAGUGRAN	t~		•	,		,		_							_		,	
FESTARUN				,	,			,				,		ı	_	,		E
FESTPRAT	•			ı	ı							,	~		,	ı		
			:							•			: ! !	:			ļ	÷

. *
canopy
open
microhabitat:
Å
species
र्च
Distribution
ς.
Appendix

									!						:	:		i
SPECIES	dFL.R	anF.LR	wFLR	mDEP	4:HCFW	SIEP	NUOM	dPIT'	wPIT	10C	STMP	RMA'I	LANE	DITCH RF	a cu	۲W	RME	RJ'
FECTRI IRP		. •					 .				: .			· • ·	• !	ĺ		,
FESTSUBV	7	,	,					_		,	_			-	•			
FRAGVESC	4	ţ.				ı	_			_		_	~	•	t			
FRAGVIRG		4				ŀ				_	_	2	~	, •	r			
FRAXAMER		2		,	•		5	-					_	1	'			
FRAXNIGR	Þ	5	m	2	~1					_		_	-1	~ ,	I		_	
FRAXPENN	Ţ	J				_				-		न	-1	-	I			
GALESPEC			,	ı						,				•	'			
GALETETR	e 1				,					ı				•	'			
GALIAPAR	4		,		,		ı			_				•	'			
GALIASPR	ı	,					,			,				•	•			,
GALICIRC	ı					,	,							•	,			
GALILANC	~	1							,					•	•			,
GALIOBTU	ı	_	_			,				_			5	-	ς.υ		e	_
GALIPALU		,	¢1		6 1					_				•	'			
GALITRIF		•*i		~1		_	~1	-		\$	_	~		•	•			_
UALL SP		,	,	ı	,	,						,			'			
GENTANDR		_	ŗ												'			
GERAMACU	_		•	•			_			,		-			'			
GERAROBE	80	ŝ	~1	m	сı		_	_		_		×.	-	•	'			
GEUMALEP									,		,	1		•	'			
GEUMCANA	_	m										-1		•	'			-
GEUMLACI	_	~	_	_	_	,				,			~	•	-	-		
GEUMRIVA			ı			,								•	'			
GEUMURBA			ı			,									•			
GEUM SP		61	_			,							~	-				
GLYCSTRI	-	*	Ŧ	÷	r n	_				c 1	_	- -	~	-	нт,		19	_
UYMNDRYU	2			,	,		_						_	•	'			
HACKVIRG	Ċ						_	_		1			_	'				
													•		1	:		

164

SPECIES	dFLR	mFLR	wFLR	mDEP	WDEP	SEEP	MOUN	dPTT	wPiT	LOG	STMP	RMAT	LANE	DITCH	RFLD	RMA	RME	RT
HEPAACUT	3	_	-	<u> </u>	-	•	•	•	-	-	-	-	I	-	-	-	-	-
HIERAURA	-	-	-	-	-	•	•		•	-	-	-	-	-	-	-	-	-
HIERCAES	6	2	-	-	-	-	-	-	•	2	2	3	4	-	4	-	-	-
HYDRVIRG	I I		-	-	-	-			-	-	•		•	-	-	-	-	-
HYPEPERF	-	-	-	-	-	-	-	-	-	•	•		3	-	5	-	I I	-
HYSTPATU	I		I I	-	-	-	-	-	-	-	-		1		•	-	1	-
IMPACAPE	6	I I	4	2	3	l	-	-	I I	8	3	8	3	1	-	3	3	1
INULHELI	-	I I	1	-	-	-	-	-		-	-	-	-	-	-	-	I I	-
IRISVERS		1	-	-	-	-	-	-		-	•	-	-	-	-	-	-	-
IRIS_SP		I I	-	-	-			•	-	-	•	-	1	-	-	!	-	-
JUNCTENU	I I	-	-	-	÷	-	-	-	-	-	-	-	2	-	-		-	-
LACTCANA	I	-	-	-	-	-	-	-	-	-	-	-	1	-	2	-	-	-
LACTSERR	2		-	-	-	-	-	-	-	-	-	I I	2	-	-	•	-	•
LACT SP		I I	-	-	-	-	-	-	-	I		-	-	-	-	-	-	-
LAPOCANA	I.	2	2	-	1	-	-	-	-	-	-	-		•	-	2	3	-
LEERORYZ		-	1	-	-	-	-	-	•	-	-	-	2	1	-	3	3	-
LEERVIRG		1	-	-	-	-	-	-	-	-	-	-		-			-	-
LEONCARD	3		-	-	-	-	-	-	-	-	•	-	1	-	-	-	-	-
LIPALOES		1	-	-	-	-	-	-	-	-	-	-	•	-	-	-	-	-
LOBEINFL	I I	•	-	-	-	-	-	-	-	-	-	-		-	-	-	-	-
LOBE_SP	-	-	-	-	-	-	-		-	-	-	-	1	-		-	-	-
LONICANA	3	•	-	-	-	-	-	-	-	•		-	-	-	-	-	-	-
LONIDIO	I I	-	-	-	-	-	-	-	-	-	•	-	•	-	-	-	-	-
LONIHIRS	-	-	-	-	-	-	-	-	-			-	•	-	-	-	-	•
LYCOAMER	-	-	-	-	-	1				-	-	-	1	-	-	-	-	•
LYCOANNO	-	-	-	-	-	•	•	•		-	-	-	-	-	-	-	•	•
LYCODEND	-	-	-	•	-	-	-	-	-	-	-	-	-	-		-	-	-
LYCOOBOB	-	-	-		-	-	-	-	-	-	-	-	-			-	•	-
LYCOTRIS	•	-	-	-	-	-	-	-	-	-	-	-	•	-	I	-	-	-
<u> </u>			· · · · ·		· . -			-								····· —		

YEGUNIF 2 </th <th>SPECIES</th> <th>dFLR</th> <th>mfi.R</th> <th>wPl.R</th> <th>щDEP</th> <th>wDEP</th> <th>SEEP</th> <th>MUOM</th> <th>IPIT wPY</th> <th>r LOG</th> <th>STMP</th> <th>KMAT</th> <th>LANE</th> <th>A HOTICI</th> <th>FLD R</th> <th>MA</th> <th>RME</th> <th>8</th>	SPECIES	dFLR	mfi.R	wPl.R	щDEP	wDEP	SEEP	MUOM	IPIT wPY	r LOG	STMP	KMAT	LANE	A HOTICI	FLD R	MA	RME	8
YSICIL 1 2 3 1 L'SINUMM 2 2 3 1 <	LYCOUNIF	 		2		-	-	•	•	~1	i .	•	1	-	-		r 1	
XSINUMM	LYSICILI	,		_			ı	•	•	,	,		¢1	•	I			
JYSTTER . </td <td>LYSINUMM</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>•</td> <td>ı</td> <td>•</td> <td></td> <td>,</td> <td></td> <td>•</td> <td>•</td> <td></td> <td>_</td> <td>ī</td>	LYSINUMM							•	ı	•		,		•	•		_	ī
MAIACANA 2 4 A MACANA 2 4 <th4< th=""> 4 4</th4<>	LYSITERR	,					ı	•	•	ı					•			
MALARACE 8 .<	MAIACANA	-	C i					~.	•	_	_	~1	1	-	'			
MAASTEL I </td <td>MAIARACE</td> <td></td> <td>4</td> <td></td> <td></td> <td></td> <td></td> <td>_</td> <td>•</td> <td>ı</td> <td></td> <td></td> <td></td> <td></td> <td>•</td> <td></td> <td></td> <td>ı</td>	MAIARACE		4					_	•	ı					•			ı
MATTSTRIL I	MAIASTEL	_		,				•	ı	,			,	, ,	•			,
MEJILUPU I ·<	MATTSTRU	,	_						,			-	_		-		_	
MEI.IALA I<	UPU.IKI3M	_	_	,							,	,	ñ	।	'			
MELLOFFI .<	MELIALBA	_	,	,							,	,	_	- -	'			
MENICANA 1 -<	MELIOFFI	ı	ı				ı			٢	,			, ,	'		ı	
MENTARVE 1<	MENICANA	-	ı						ı	•			_	•	'			
MILEFFU - </td <td>MENTARVE</td> <td>I</td> <td>-</td> <td>_</td> <td></td> <td>,</td> <td></td> <td>·</td> <td>•</td> <td>,</td> <td></td> <td>→</td> <td>_</td> <td>_</td> <td>-</td> <td></td> <td>51</td> <td></td>	MENTARVE	I	-	_		,		·	•	,		→	_	_	-		51	
MITTEIDIE I	MILIEFFU	_			ı	,								•	I			
MITEDIPH 3 1 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	MITCREPE	_						' _	ı						•			
MONAFIST	HAIGHLIM	~ 1	_	,					ı				ı	'	•		·	ī
MONOHYPO ·<	MONAFIST			,											'			
MONOUNIF ·<	MONOHYPO	,		,								ı			'			
MUHLFRON	MONOUNIF			,					ı				,		•			
MUHLMEXI	MUHLFRON							•	ı		ŗ			_	•			
NEPECATA I · · · · · · · · · · · · · · · · · ·	MUHLMEXI			,				•	ı				÷ ∄	_ _	•			
ONOCSENS 1 6 1 1 1 2 3 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 3 2 1 2<	NEPECATA	_						'	ı	·				'	•			
ONOPACAN 3 .<	ONOCSENS	_	ĥ	_	_	_	_	•	۰	,			~ `	, ,	2		_	
ORYZASPE 7	ONUPACAN	÷							,				_		•			
OSMOCLAY 6 1	ORYZASPE	<i>۲۰</i>					,	۔ ~	ı			,			ŗ		,	,
OSMUCLAY	OSMOCI.AY	÷	_				,	_	I	•	,	,	_	•	'		,	
OSMUREGA - · · · · · · · · · · · · · · · · · ·	OSMUCLAY						,		ı	•				•	'		,	,
OSTRVIRG 7 2	OSMUREGA	,							ı					'	,			
	OSTRVIRG	1					,	' ~1	ı	-	,			•	,		,	

01/01/20	a sak		wFLR.	mDEP	wDEP	SECP	MOUN JPIT	wPU	1,00	STMP	RMAT	LANE	DITCH	RFLD	KMA	RME	RT
XALSTRI						,	•					_		_			
XAL SP	,	,															
ANAOUIN		r					•										•
ANIACIUM		,	ı			,			,					ri)		1	
ANICAPI	-	,	ı			,	•								,		•
ANI SP	,		1				•			,							, .
ARTINSE	ŝ	ŝ	_	C 1		ı	, ,		~	_	r 1	ŝ	_	न ।	 .	•	_
IBALARUN	,	,					•					_		~1			ı
HECCONN	,											,					•
HEERAT		_					ı ,			,		6 1		r.i			
HLO SP	_	,	ı	1			•		ı								•
HRYLEPT	ম্য		,			,	_		,			m		_		,	
PICEGLAU	,	,					•		,								
ILEPUMI	4	2	¢.		-	,	•	•	-	,	-1	1 17	_	-1	_		-
ONUSTRO	_	•								,			,				ı
PLANLANC			,			·	•							_			
UANMAJO		_	,		,	,	•				·	.					
LANRUGE	_	_	,			,						÷.		ħ١			
ON ALSO		-				ı			,	•		-		, .	, -		
POA_COMP	÷	۰	_			ŀ	•		,	,		~ ·		•	_ ,	_ ,	• -
POA_PALU	6 1	-	_	_		,	•						_	~ ~	-	. .	
PRAT	4		_			,	_				-1			•		ň	_
POA_SALT	_	,		,		ı	•	,					,	·			
POA SP	,			,			•						,				
*ODOPELT			ı				•						ı	ı			
POLYACRO							•						ı	1			,
POLYPAUC	_					ı	_						•				
POLYPERS	ŗ			,		ı	•					C 1	, .		-1		
POLYPUBE	9	_					3					_	-			,	
						:				:	:	:		!			

±67

SPECIES	dFLR		wFLR	тDEP	wDEP	SEEP	MOUN	dPIT	wPIT	LOG	STMP	RMAT	LANE	DITCH	RFLD	RMA	RME	RT
POPUBALS	-			-	-	-			-	. —	•	-	•		2	-	-	-
POPUGRAN	-	-	-	-	-	-	-	-	-	-		-	-	-	-	-	-	-
POPUTREM	-)	•	-	-	-	-	-		-		-	1	-	5		-	-
POTENORV	-	-	-	-	-	-		-		-	-	-	1	-	-	-	•	-
POTERECT	-	-	-	-	-	-		-		-	-	-	2	-	5	-	•	-
PREN SP	4	-			-	-	-			-	-	1	-	-	-	-	•	-
PRUNSERO	7	-	-		-	-				-	-	-	2	•	4	•	•	-
PRUNVIRG	8	4	-	1	-	-	1	1		1		1	3	-	5	-	-	-
PRUNVULG	-	1	1		-	-	-	-	-	-	•	-	3	-	4	-	1	-
PTERAQUI	5	-	-		-	-			-	-	-	-		-	2	-	-	-
PYROELLI	-	-	-		-	-	-	-	-	-	-		•		•	-	-	-
QUERALBA	1	-	-	-	-	-		•	-	•	-		-	-		-	-	-
QUERMACR	1	-	-	-	-	-	•	-	-		•	-	-	-	-	-	-	-
OUERRUBR	5	-	-	-	-	-	ļ			-	-	-	-	-	ł	-	-	-
QUER SP	-	-	-	-				-		•	-	-	-	-	-	-	-	
RANUABOR	5	3	-	-	-	I	1	1				3	2	-	•	-	-	1
RANUACRI	3	2	i	I I	-		-	-	-	-		2	3	I .	5		2	1
RANUHICA		-	-	•	-	-	-	-	-	-		-		•		-	-	-
RANURECU	2	2	1	I I	l	I I		-	-	۱.	I I	-	2	-	-	•	-	1
RHAMALNI	-	-	-	-	-	-	-	-			-	-	-	-	-	-	-	-
RHAMCATH	9	5	2	1	2	-	2	-	-		-	1	2	-	5	-	-	-
RHUSRADI	4	1	L.	-	-	-			-		-	1	2	-	•	-	1	-
RHUSTYPH	-	-		-	-	-	-	-	-	-	-	-	-	-	1	-	-	-
RIBEAMER	-	1	1	-	-	-	-	-	-		-	-	-	1	-	-	Ĵ	-
RIBECYNO	4	-	1						-	-		2	l	-	L	•	2	-
RIBEGLAN	-	1	-						-	-	-	-	-	-	-	-		-
RIBELACU		-	-		-	-			-	-	-	-	-		-	•	-	-
RIBERUBR		1	-			-			-	-		-	-		-	1	-	-
RIBETRIS	2	6	1	1	-	-	-	-		2	-	-	-	-	-	-	3	ι

	 JCI D					SELD.				1.00	STMD	DMAA'T	LANC	DITCH	DELD	DA A	DAND	<u>рт</u>
SPECIES			WFLK		WDEP	SEEP	MOON	ur11	wr+1			0.04751	1.71.76	ынсп				X1
RIBE_827	-	-	-	-	-	•	•	-	-	-	-	-	-	-	-	-	-	•
ROBIPSEU	2	I I	-	-	•	-	•	•	-	-	-	-	-	-	1	-	-	-
ROSABLAN	-	-	-		-	-		-	-	-	-	-	-	-	•	•	•	•
ROSAPALU	-	-	-	-	-	-	-	-	-	-	-	-	-	-	1	-		•
RUBUALLE	3	-	-	-	-	-	-	-	•	-	-	-	I	-	-	-	-	•
RUBUIDAE	9	7	-	2	-	-	i	I	•	3	-1	6	5	-	4		-	-
RUBUOCCI	7	-	1	-	-		I I	1	•	•	-	1	2	-	-	-	L	•
RUBUODOR	-	1	-	-		-	-	-	-		-	2	-	-	•		•	•
RUBUPUBE	I I	6	-	L	-	-	-	•	•	5	2	8	I	1	3		1	I.
RUBU_840		-	-	-	-	-		-	-	I	-	-	-	-	-	-	-	•
RUDBHIRT	-	-	-	-	-	-	-	-	-	-	-	•	-	-	-	-	-	•
RUMEORBI	-	-	-	-	-	-			-	-	•	-	-	-	-	1	-	•
SALIBEBB	-	-	-	-	-	-	-	-	-	-	-	-	-	-	2	-	-	-
SALIDISC		1	-	-	-	-	-	-	-	-	-	-	-	-	4	-	-	-
SALJERIO	-	Ł	1	•	-	-	-	-	-	-	-	-	1	I	4		1	-
SALIPETI	•	-	-	-	-	-	-	-	-	-		•	•	•	2	-	-	-
SAMBCANA	•	3	1	I.	I.	-	-	-	-		•	1	•	-	-	-	-	-
SAMBRAPU	7]	-	-	-	-	1	1	-		•	2	2	-	-	-	-	-
SANGCANA	1	-	-	-	-	-	<u>1</u>	-		•	-	-	-	-	-	-	-	-
SANIMARI	-	-	-		-	-	-		-	-	-	-	-	-	•	-	•	-
SANITRIF	-	•	-	•	-		-	-	-	-	-	-	•	•	•	-	•	-
SANI_SP	1	-	-	-	-	-	-	-		-	-	-	-	-	-	-	-	-
SCHIPURP	6	-	-	-	-	-	-	-	-	•	•	-	-	•	-	-	-	-
SCIRATRO		-	-	-	-	-	-	-	-	-	•	-	l	-	L	1	-	-
SCUTLATE		1	I I	-	-	-	-	-	-	-	-	1	1	1	-	-	3	-
SICYANGU	-	•		-	-	-	-	-	-	-	-	1	•		•	-	-	-
SILEVULG	-	-	-	-	-				•		-	-	-	-	l	•	•	-
SISYMONT	-	-	-	-	-	-	-	-	-	-	-	-	-	-	2	•	-	-
SIUMSUAV	•	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
·																		

SPECIES	dFLR	nıFLR	wFLR	mDEP	wDEP	SEEP	MOUN	dPIT	wPIT	LOG	STMP	RMAT	LANE	DITCH	RFLD	RMA	RME	RT
SMILHERB	3	-	·			•	I	-	-			<u>.</u>		-	1	-		-
SMILHISP	1	-	-	-			-		-	-	-	-	•	-	-	-	•	-
SOLADULC	7	4	3	2	3	1	1	1	-	-	-	2	2	-		I I	I I	1
SOLIALTI	4	3	•	-		•	•		-	1	-	2	4	-	4	1	-	-
SOLICAES	2		-	-	-	•	•	-		-	•	-	-	-	-	-	-	-
SOLICANA		4	3	-	2	-	2	-	-	3	I I	3	5	-	3	-	1	•
SOLIFLEX	3	I	-	-				-		-	l	1	1	-	-	-	-	-
SOLIGIGA	I I	3	1	-	-	-	-	-		1		-	3	I	3	2	3	-
SOLIJUNC	-	-	-	-	-		1				-	-	-	-	-	•	-	-
SOLINEMO	I I	-	-	-	-	-	-	-	•		-	-	-	-	3	-	-	-
SOLIRUGO	I I	1	-	-	-	-		-	-	-	Ι	-	-	-	-	•	-	-
SONCARVE	2	1	-	-	-	-	1	-	-	-	-	-	-	-		-	-	-
SONCOLER	I	-	-	-	-	•	1	-	-	-	-	-	l	-	-	-	-	-
SPICEINTE	2	1	1	-	-	-	-	-		2	2	1	3	-	-]	2	-
STELLONG	-	•	-	-	-	-	-	-	•				-	-	-	-	-	-
STREROSE	-	-	-	-	-		-					-	-	-	-	-	-	-
SYMPALBO	I I	-	-	-	-	-	1	-	•				-	-	-	-	-	-
TARAOFFI		3	1	۱.	1	-	l	1		2	I	2	6	-	5	•		1
TAXUCANA	-	-	-	-	-	•			-	I I	-	-	-	-	-	-	-	-
THALDIOI	6		-	-	-	-	1	-	-	-	-	-	-	-	-	-	-	-
THALPUBE	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
THEI.NOVE	-		-	-	-	-	-	-	-	-	-		-	-	•	-	-	-
THELPALU	-	-	-		-	-	-	•	-	-	-	-	-	•		-	•	-
THUJOCCI	4	2	1	-	-	-	-	-	-	-	-	1	-	-	t	-	I	-
TIARCORD	5	9	-]	-]	-	-	-	7	3	7	3	-	-	-	1	-
TILIAMER		5	t	-	1	-	3	-	-	-	-	2	-	-	I	-	1	-
TRAGDUBI	-	-	•	-	-	-	-	-	-	•		-	-	-	I	-	-	-
TRIEBORE	4	2	-	-	-	-	-	-	-	2	-	2	-	-	-	-	-	-
TRIFREPE	2	-	-		-	•	1		-	-	-	-	3	-	1	-	-	-

... _____

....

SPECIES	dFLR	mFLR	wFLR	mDEP	WDEP	SEEP	MOUN	dPIT	wPiT	LOG	STMP	RMAT	LANE	DITCH	RFLD	RMA	RME	RT
TRILEREC	7	1	•		•		-	-	-	-		3	1	-	•	•	•	-
TRILGRAN		I I	-	-		-	4	-	-	•	-	1	4	-	-	-	-	-
TRIOAURA	1	-	-	-		-		-	-	-	-	-	-	-	-	-	-	-
TSUGCANA	5	1	-	-	-	-	-	-	•	1		1	1	-	•	-	-	-
TUSSFARF	-	-	-	-	-	-	-	-	-	-	-		-	L	2	-	-	-
TYPHLATI	-	-	-	-	-	-	-	-	-	-	-	-		•	-	-	-	-
ULMUAMER	4	7	3	1	2	-	-	-	-	1		1	2	-	2	-	2	-
UNKN 751	-	-	-	-	-	-	-	-		-		-	-	÷	-	-	-	-
UNKN 791	-	-		-	-	-	-	-	-	-	-	-	-		-	-	-	-
UNKN_822	-	-	-	-	-	•	-	-	-	-	-	-	l		-	-	-	-
URTIDIGR	2	-	-	-	-	;	-	-	-	-		-	1	-	-	-	-	-
URTIDIO	-	-	-	-	-	-	-	-	-				1	-	-	-	-	-
UVULGRAN	1	-	-	-	-		-	-	-	-			-	-	-	-	-	-
VACCANGU	I I	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
VERBHAST	-		-	-	-	-	-	-	-	-	-	-	I I	-	•	-	•	-
VERBTHAP	1		-	-	-	-	-	-	-	-	-			-	1	-	-	-
VERBURT	-	-	-	-	-	-	-	-		•		•	1	•	-	-	-	-
VEROOFFI	-	-	-	-	-	-	-	-	-				-	-	-	-	-	-
VEROSERP	-	-	-	-	-	-	-	-	-				l	•	-	-	-	-
VIBUACER	4	-	-	-	-	-	-	-	-	-			1	-	-	-	-	-
VIBULENT	2	-	-	-	-	-	-	-	-		-		-	-	3	-	-	-
VIBUOPUL	-	1	-	-	-	-	•			•	-	•	-	-	-	-	-	-
VIBUTRIL	-	-	-	-	-	-		•		-		-	-	-	1	-	-	-
VICICRAC	-	-	-					-		-	-	-	1	-	2	-	-	-
VIOLAFFI	-	1	-		-	-	-	-		t	-	-	3	-	4	-	2	1
VIOLBI.AN	1	2	-			-		-	-	-	-	1	1	-	1	-	-	-
VIOLCANA	-	-	-		-		-	-	-	-	-	-	1	-	-	-	-	•
VIOLCUCU	-	2	I I		L	-	-	-	-	I	-	-	ł	-	-	1	ł	-
VIOLUAR	-	-		-	-	•	-		-	-	-	I	-	-	-		•	•
							-											

SPECIES	dFLR	mFLR	wPLR	nDEP	WDEP	SEEP	MOUN	dPTT d	TIGw	DOJ	STMP	RMAT	LANE	DITCH	RIA.D	RMA	RME	R
VIOLPUBE		 ^	 •	 ,	 _	: •	1						i _			.		,
VIOLROST	•		1	,			I							•				
VIOLSORO	_	,	,			ı		,	ŀ			_	-					
VIOL_788				,	ı													
VIOL SP		,			,			,		_							-	
VITIRIPA	r-	2	_	,			-					-		.,	5	-		r
WALDFRAG	4			ı	ı	4	C 1		,				•	•				
					i				:	!	:						:	

APPENDIX 8

ENVIRONMENTAL DATA SUMMARY BY QUADRAT (I)

<u>Legend</u>

Column Heading	Description	Unit/Label
PA#	patch number (label)	numeric label
Q#	quadrat number (label)	numeric label
# SPP	# species recorded in quadrat	number
# MH	# microhabitats recorded in quadrat	number
SPM	soil parent material class:	
	glacio-fluvial parent material	l
	calcareous till parent material	2
	lacustrine parent material	3
	calcareous outwash parent material	4
SO	soil order class:	
	brunisol	1
	gleyed brunisol	2
	luvisol	3
	gleyed luvisol	4
	gleysol	5
MOIS	soil moisture class	
	seasonally wet depressions present	Ι
	seasonally moist depressions present	2
	all depression seasonally dry	3
SOM	% soil organic matter	percentage
pН	soil pH	value
CALC	available calcium	emol/kg
Ca:Mg	calcium:magnesium ratio	value
K:Mg	potassium:magnesium ratio	value
OMH	open microhabitat class	Y (yes)
		N (no)
SEEP	seep	Y (yes)
		N (no)
DI ST	disturbance class:	
	regenerating field	F
	canopy gap	G
	trail/lane/access road	Т
	no disturbance features present	N
PA	patch area	hectares
FC	% forest cover in 5km x 5km square	percentage
	centered on quadrat	
90	patch isolation: mean distance to nearest	metres
	8 woodlots, measured in 45 degree arcs	

PA#	Q#	#SPP	#MH	SPM	SO	MOIS	SOM	рH	CALC	Ca:Mį	g K:Mg	OMI	SEEP	DIST	РА	FC	PI
]	I	57	2	4	3	3	3.1	6,2	2.1	0.8	0.5	Y	N	F	183	49	228
1	2	53	2	4	3	3	3.9	6.7	2.4	1.2	0.9	N	N	F	183	49	228
1	3	44	4	4	3	3	7.8	7,1	7.3	14.3	5.9	Y	N	6	183	49	228
1	4	55	2	4	3	3	6.4	7.0	5.3	1.5	0.5	Y	N	G	183	49	228
I I	5	32	1	4	3	3	6.6	7.2	5.9	5.7	1.9	N	N	N	183	49	228
I I	6	38	3	4	3	3	6 .1	7.3	7.0	2.7	1.0	N	N	N	183	49	228
I I	7	41	L	4	3	3	7.4	7.2	6.7	1.9	0,8	N	Ν	N	183	49	228
I I	8	44	3	4	3	3	10,3	7.5	14,9	2,4	0,4	N	N	N	183	49	228
2	9	63	3	2	2	2	14.5	7.5	21.8	2.5	0.3	N	N	N	53	41	50
2	10	58	4	2	2	1	19.6	7.4	12.4	2.4	0.5	N	N	Ν	53	41	50
2	11	56	4	2	2	2	11.0	7.7	19.9	3.0	0,4	N	N	N	53	41	50
2	12	59	4	2	2	3	14.0	7.8	18,0	2.7	0,4	N	N	N	53	41	50
2	13	58	4	2	2	2	15.8	7.8	17.0	3.0	0.4	N	N	N	53	41	50
2	14	62	4	2	2	2	12,8	7.5	19.1	2,1	0.3	N	N	N	53	41	50
2	15	37	2	2	2	3	12.0	7.4	11.3	3.1	0.8	N	N	N	53	41	50
2	16	57	4	2	2	I I	16.5	7.5	19,0	1.1	0,2	Ν	N	N	53	-11	50
3	17	42	4	4	3	3	8.4	5.6	2.0	2.0	1.1	N	N	Т	42	43	415
3	18	51	7	4	3	3	10.5	5.5	1.4	0.4	0.3	Y	N	TG	42	43	415
3	19	49	3	4	4	1	9.4	5.9	2.2	4.3	2.7	Y	Y	G	42	43	415
3	20	36	4	4	3	1	10.8	5.7	1.1	0.3	0.2	N	N	N	42	43	415
3	21	47	3	4	3	I.	8,6	5.7	3.1	6.0	2.1	Ν	N	N	42	43	415
3	22	56	4	4	3	3	8.1	5.9	2.4	0.4	0.2	Y	N	TG	42	43	415
3	23	35	3	4	4	1	19,8	6.7	13.7	0.8	U, I	Ν	Y	N	42	43	415
3	24	30	3	4	4	Ł	19.1	6.9	0.2	0.0	0.1	N	N	N	42	43	415
4	25	107	3	2	2	2	15.4	-	-	-	-	Y	N	Т	78	30	450
4	26	113	4	2	2	2	15.5	-	-	-	-	Y	N	Т	78	30	450
4	27	49	2	2	2	2	13.0	-	•	-		Ν	N	N	78	30	450

Appendix 8. Environmental data summary by quadrat (1).

								!						;			
PA#	\$	ddS#	HW#	SPM	so	MOIS	SOM	pH	CALC	Ca:Mg	j K∶Mg	SIMO	SEEP	DIST	Ł	ų	≂
.		5	-		,	-	991	 ,			· ۱	z	z	z	78	30	450
4		6	·1 •	4 6	• •	• -		,				z	z	z	78	30	450
म भ	62	5	T •	4	Чг	4 -					,	; >	z	G	78	<u>8</u>	450
ч [,]	S ;	2 i	a -	40	ч с		2 Y C					· >	z	0	78	ŝ	450
4	i i	e (. 1	ч г	-1 r		8 00			,		~	z	z	78	30	450
4	2	21		2	1 -	4 -	0.02	, r	16.4	5	,	· >	z	0	98	41	356
vî (÷.		4 •	-1-		• •			18.6		0.4	z	z	z	98	4	356
vn v	5	40 1 5	4 4	יו ר				XL	19.8	3.2	0.9	z	z	z	8 6	4 	356
~ `	93	CF CF	.	46		. ~	0	81	15.3	5.0	1.5	z	z	z	86	41	356
<u>~</u>	55	r s	ų -	4 F		· ~	1.1	7.6	58	- ri	0,6	z	z	z	8 6	4	356
<u> </u>		Č,		46		n .+•	9 61	2.5	15.4	÷	0.9	z	z	z	86	4	356
^ '	5	 		4		n ••		01	20.2	2.6	0.6	z	z	z	86	4	356
~ •	5	5 0	- ~	.ı r	+	n 	101	1.8	6.6	5.0	0.7	z	z	z	86	4 T	356
<u>^</u>	9 7	9 2	. r	4 r		. ~	5.0	!,		,		z	z	Ŀ.	2	2	487
<u>م</u>	- + +	9	4 0	t r			1 1			,		z	z	z	<u>2</u>	30	487
0.	1	23	ή ν	יי	1 -		013			,	,	≻	z	5	인	30	487
c \	<u> </u>	2	ጉ ጦ	1 6		1 6	0 Pl				,	z	z	z	2	90	487
۰ م	1 -	9	n 7	4 r	10	יו	0		,	,	,	z	z	z	입	9	487
۰ م	÷.	23	n r	h r	- I -	1 ~			,	,	,	z	z	z	<u>61</u>	30	487
۵.	ç ;	1	n -	4 0		، د	0 21		,	,	,	z	z	z	<u>-</u> :	œ	487
ρ,	- a 	53	t r	4 C	، د	• •	1.5			,	•	z	z	Ż	12	90	487
		ē ę	1 רי	10	ı –	I	1	12	7.1	39.00	1.1	z	z	z	68	55	300
- 1	4	q 4	, ,	۱ ۲		, <i>~</i>		7.0	1,2	2	0.7	≻	z	Ģ	68	55	300
~ r	2	6	4 -	16		, ~	23	13	90 10	8 Ci	1.1	۲	z	0	68	55	300
~ r	1	2 3	r -	10		. د.	64		5.7	Ľ.	0.5	z	z	z	68	55	ĝ
- 1	25	14		، د	. –		5.6	7.7	8.0	3.9	. .1	z	z	z	68	55	300
~ ~		67		101		س ا	6.0	7.0	6,0	6.1	0.8	z	z	9	8	55	300
-															ļ		

476

PA#	Q#	₿SPP	#MH	SPM	SO	MOIS	SOM	pН	CALC	Ca:Mg	; K:Mg	OMH	SEEP	DIST	PA	FC	Pi
7	55	50	2	2	1	3	6.1	6,8	4.4	1,2	0.6	N	N	N	68	55	300
7	56	48	5	2	1	3	7.4	7.1	6.1	2.0	0.7	Y	N	G	68	55	300
8	57	30	5	2	3	3	6.1	-	-	-	-	Y	N	G	101	31	431
8	58	16	3	2	3	3	6.8	-	-	-	-	N	N	N	101	31	431
8	59	14	3	2	3	3	8.2	-	-	-	-	N	N	N	101	31	431
8	60	21	2	2	3	3	8.9	-	-	•	-	N	N	N	101	31	431
8	61	17	2	2	3	3	7.3	5.9	4.5	33.9	12.0	N	N	N	101	31	431
8	62	21	I	2	3	3	6.1	5.8	3.0	37.0	23.7	N	Ν	N	101	31	431
8	63	21	3	2	3	3	6.6	5,9	2.6	64.0	42.6	N	N	N	101	31	431
8	64	25	2	2	3	3	9,1	5.9	4.0	43.0	27.1	N	N	N	101	31	431
9	65	33	1	2	1	3	5.9	6.6	4.6	221.3	62,8	N	N	N	267	45	337
9	66	34	2	2	I.	3	5.6	6.1	2.4	116.7	88.9	N	N	N	267	45	337
9	67	35]	2	1	3	5,9	5.6	1.3	128.8	180.9	N	N	Ν	267	45	337
9	68	37	1	2	1	3	6.2	6.2	4.1	78.8	29.2	N	N	Ν	267	45	337
9	69	46]	2	1	3	5,0	6.2	0.7	0.5	0.3	N	N	N	267	45	337
9	70	34	1	2	l	3	5.1	6.5	5.1	70.6	44.8	N	N	Ν	267	45	337
9	71	37	1	2	1	3	8.1	5.6	1.1	0.6	0.8	N	Ν	N	267	45	337
9	72	42	2	2	1	3	6.4	5.9	0.3	0.2	0.5	N	N	G	267	45	337
10	73	53	1	4	3	3	8.3	6.5	6.1	117.9	16.4	Y	N	G	82	39	294
10	74	48	4	4	3	3	6.1	6.7	8.5	91.6	26.5	Y	Ν	G	82	39	294
10	75	28	3	4	3	3	8.2	6.4	4.4	216.0	33.9	N	Ν	G	82	39	294
10	76	35	4	4	3	3	10.2	6.8	9.5	77.3	11.8	N	N	N	82	39	294
10	77	34	3	4	3	3	8.0	6.6	7.6	73.8	12.5	N	N	N	82	39	294
10	78	60	I I	4	3	3	8.4	6.3	4.1	1.4	0.2	N	N	N	82	39	294
10	79	47	4	4	3	3	8.9	7.4	12.8	88.7	18.3	Y	N	N	82	39	294
10	80	41	1	4	3	3	10.5	7.2	13.5	65.5	12.7	Ν	N	N	82	39	294
1	81	96	4	4	I.	1	5.1	7.7	11.7	113.2	13.1	Y	Ν	TF	34	41	244

Appendix 8. Environmental data summary by quadrat (1).

PA#	Q#	#SPP	#MH	SPM	so	MOIS	SOM	pН	CALC	Ca:Mg	K:Mg	ОМН	SEEP	DIST	₽А	FC	PI
н	82	85	3	4]	I	3.9	7.9	9.4	455.5	43.8	Y	N	TF	34	41	244
11	83	76	2	4	1	1	4.0	8.0	10.5	1021.6	77.1	Y	N	F	34	41	244
11	84	84	2	4	1	3	7.1	7,7	14.3	154. l	18.0	Y	N	F	34	41	244
11	85	79	1	4	2	2	10.1	7.8	18.6	62,4	6,5	Y	N	G	34	41	244
11	86	9i	2	4	2	1	12.1	7,6	19.8	48.1	4.9	N	N	Ν	34	41	244
11	87	89	1	4	2	2	8.8	7.6	12.8	113.3	15.1	N	N	N	34	41	244
11	88	88	3	4	2	ł	8.6	7,7	8.2	2.4	0.1	N	N	N	34	41	244
12	89	57	5	4	3	ł	18.5	5.9	5.9	2.2	0.2	N	N	N	42	42	394
12	90	58	5	4	4]	18.3	6.5	6,4	2.4	0.3	N	N	N	42	42	394
12	91	62	4	4	3	1	12.6	5.6	6.4	104.3	26.4	N	Ν	N	42	42	394
12	92	58	4	4	3]	16.7	6.0	9.2	74.4	8.2	N	N	N	42	42	394
12	93	58	3	4	3	3	17.2	5.6	4.0	1.9	0.2	N	N	N	42	42	394
12	94	54	2	4	4	1	14.6	6.2	10,9	96,2	9.6	N	N	N	42	42	394
12	95	60	3	4	4	L	17.6	6.5	14,6	59.1	6.1	N	N	N	42	42	394
12	96	55	4	4	Ļ	I I	15.1	6.4	12.5	121.4	9,5	N	N	N	42	42	394
13	97	21	3	1	t	3	5.2	7.0	10.4	252.8	27.0	N	N	1'	64	67	106
13	98	21	3	ł	1	3	5.7	7.1	8.2	2.7	0.1	N	Ν	ΤĽ	6-1	67	106
13	99	20	3	I I	1	3	7.3	6,4	10.1	109.1	16.3	N	N	N	64	67	106
13	100	17	4	I I	1	3	7.6	6,3	8.2	2,2	0.2	N	N	Ν	64	67	106
13	101	23	3	1	1	3	7.8	6.4	8.8	2.6	0.1	Ν	N	G	64	67	106
13	102	17	2	1	1	3	6.6	6.9	11.3	1100.4	33.6	N	N	N	64	67	106
13	103	18	3	I I	1	3	8.2	6,0	2.5	1.4	0.2	N	Ν	N	64	67	06
13	104	21	5	1	1	3	4.7	6.8	38.6	70.7	0.5	Y	N	TG	64	67	106
14	105	16	2	2	3	3	8.7	6.9	10.5	203.1	26.1	N	N	Т	10	38	647
14	106	28	4	2	3	3	7.7	6.8	8.0	194.4	35.6	N	Ν	Т	10	38	6 47
14	107	24	3	2	3	3	7.7	6.6	5,6	1.4	0.3	N	N	N	10	38	647
14	108	13	I I	2	3	3	8.5	6.5	9.5	116.0	20.0	N	N	N	10	38	6 47

Appendix 8. Environmental data summary by quadrat (1).

																2	
₽A#	#ð	ddS#	HM#	MdS	So	MOIS	SOM	Ηď	CALC	Ca:Mg	K:Mg	HWO	SEL		≤	<u>د</u>	±
	19	4	-	۰ ۱	~	-	10.8	6.6	8.6	838.2	(60.3	z	z	z	01	38	647
<u>य</u> :	60	1 1		16		1	911	7.0	13.2	127.8	16.d	z	z	z	0	38	647
4	2	<u>t</u> 2	n -	4 6	•	۲	86	9.9	6	286.5	60.]	z	z	z	01	38	647
1 :	= =	<u>, 1</u>	- 6	4 6	- ~	- -	11.7	6.6	10.4	201.9	36.2	z	z	z	10	80	647
<u>1</u>			. .	4 -	• -	n .~	99	5.5	29	178.9	158.5	z	z	z	325	19	332
<u>_</u>	1					n r *	2 4 4		0.6	0.2	0.5	z	z	z	325	ĥ	332
<u>-</u> :	* •	3,2	n -				,	ŝ	0.7	10	0.5	z	z	z	325	61	332
2 3	212 1	1 V 1 V				י ר	4		6.0	50	4.0	z	z	z	325	61	332
<u>,</u>		98	-			n ~	0		0.8	9.1	0.8	z	z	z	325	61	332
<u>-</u> :	2	93	n -			n e r	40	5	0.7	0.3	0.6	z	z	z	325	61	332
<u>-</u>		9 8	- r	- -			0	2	10	0.2	0.6	z	z	z	325	[9]	332
<u>^</u> :			יור	<u> </u>		۰ «	4	5	0.1	0.1	0.2	z	Z	z	325	61	332
2 2		55	4 5		1	n ¢	45.9	6.8	45.9	120.7	4.6	۲	z	G	547	46	422
2 1	15	58	দেব	•		ורו	23.1	6.8	26.8	123.9	5.1	z	z	z	547	46	423
2 2	12		- 73		-1	I Ci	25.0	7.0	27.2	330.8	E.01	>	z	J	547	1 6	5
2 1	22	07			1	ור	20.7	5.6	13.7	102.1	6.3	z	۷	z	547	46	57
	5 6	00	r vr		3	1 51	33.7	7.0	33.2	1.9.1	يد ري	۲	Z	5	547	46	422
2 1		2	יר	15	4	i ri	36.8	6.4	33.0	94.4	4.9	z	z	z	547	9	422
2 1			। च	10	4	- r-1	18.1	Ú,	25.1	152.3	4.7	z	z	9	547	1	422
2 4	20	2	ر . م		ন	1 64	13.6	6.6	15.7	58,8	3.1	z	z	z	547	46	422
	9 C	1			~	m	7.3	6.0	5.3	3.5	0.0	z	z	Z.	101	30	1378
		ĉ		1 r	••		6.7	6.0	3.5	<u>-</u> !	0.4	z	z	z	101	30	1378
	2 2	64 Y	, r	1 6			6.8	7.0	- 30	<u>ع.</u> 2	0,2	z	z	z	101	ŌĒ	1378
			•••		-		13	6.5	6.2	100.8	21.4	z	z	z	101	30	1378
- :	1 6	2	16	16			5.7	6.4	0.7	0.5	0.4	~	z	<u>9</u>	101	05	1378
25		9 7 7	ı -	• •	۰ <i>~</i>		4	6.1	9 7	149,6	40.8	7	z	Ŋ,	101	30	1378
2 5		3 2	- ~			. ~	5	6.0	3.7	60,4	16.9	۲	z	c	101	30	1378
-	<u>ה</u> -	1	4	•	,	1											

470

i

₽A#	₽	43₽₽	₩NH	SPM	S0	MOIS	NOS	pfl	CALC	Ca:Mg	K:Mg	HMO	SFEP	DIST	ΡA	FC	=
17	136	57	÷	2	m		6.4	6.8	3.2	1.5	0.2	~	z	0	101	30	1378
81	137	32	_	_		m	6,1	5,6	<u>ק</u>	65.9	60.9	z	z	z	94	22	16 <u>2</u>
18	138	21	4	-	4	e	5.7	5.6	0.1	0.3	0.3	z	z	Ţ	94	52	291
18	139	22	2	-	1	÷	5.3	5.3	0.8	0.3	+'0	z	z	Ð	8	22	291
18	140	22	-	-	. →	ŝ	4.9	5.7	3.4 2	83.4	33.2	z	z	z	94	52	291
18	141	29	_	-		m	6.2	5.3	0.6	0.1	0.4	z	z	z	94	52	291
<u>80</u>	142	8	e	_		m	6.5	6.0	6,0	65.0	16.7	z	z	z	94	52	<u> 5</u> 91
*	143	30	C1	_	_	-	5.3	5.6	ŗ.	60,6	65.3	z	z	_	6 4	52	201
81	144	27	_	_	_	Ē	5.2	5.4	0,9	0.3	0.3	z	z	z	94	5	162
61	145	57	_	ţ.	ŕ	Ē	12.1	6.6	ŝ	5	0.3	z	z	z	37	28	663
61	146	41	e	رم ا	÷	~ 1	4 .8	6.0	Q,6	64.0	13,4	z	z	z	37	28	663
61	147	59	L.J	~1	~	~	0.11	6.4	6.9	47,6	5.6	z	z	z	37	38	663
61	148	40	ъ	6 1	ۍ.	e	8,8	6.1	2	0.8	0.3	≻	z	σ	37	28	663
6	149	76	••	ר ו	না	_	14.0	6,8	8 .	130.6	4 1	z	×	z	37	38	663
6	150	58	ŝ	L 1	.,	- 1	17.5	6.9	5	74.4	8.6	~	Y	z	37	38	663
6	151	58	.	~1	-+	-	40.9	6.7	27.5	86.3	7.0	7	2	z	17	8 5	663
<u>6</u>	152	99	~	cı	4	. -1	19.0	6.7	5.11	101.7	10.4	۲	z	с С	37	28	663
20	153	41	÷	2	ŝ		42.6	7.1	52.6	21,8	8.1	Y	z	0	60	26	644
20	154	28	~1	ι Γ	Ś	2	13.2	7.7	19.7	42,6		z	z	z	60	26	644
20	155	55	01	-1	ŝ		20.9	7.3	29.2	34.6	3.2	~	z	5	6 0	26	644
20	156	0†	4	2	·^		69.2	6.7	50,7	11.4	1.7	z	z	z	09	36	644
30	157	R	m.	5	Ś	с 1	13.5	1.7	21.5	-i Ci	0.1	z	z	0	09	36	644
20	158	25	~	ר ו	vî,		16.8	7.6	24.8	0.8	0.1	z	z	z	60	26	644
20	159	37	Ŧ	2	Ś	-	74.0	6.7	50.7	С.П.	2	z	z	z	9	26	644
20	160	ন	ᠳ	с1	Ś	_	68.7	6.7	50.1	14.7	1.7	z	z	z	60	36	644
21	161	27	न	cı,	PP 7.	رب ا	11.4	6.5	6.2	3.0	0.8	z	z	z	신	31	653
ទ	162	27	v	cı	m	m	13.6	7,0	13.2	2.0	0.3	Y	z	5	13	1E	653

480

i

PA#	Ő#	ddS#	#MII	SPM	8	MOIS	SOM	pH	CALC	Ca:Mg	K:Mg	OMII	SEEP	DIST	۲٩	FC .	ε
21	163	8	म	2	-	•~		6.7	~	3.2	0.6	z	z	z	12	31	653
21	164	50	Ŷ	cı	4	_	12.8	6.9	12.7	1.5	0.3	۲	z	9	싙	31	633
21	165	50	9	¢1	-, +	_	19,9	6.9	25.0	4.1	0.1	~	z	υ	<u>1</u>	ΕĒ	653
21	166	32	c.	7	•*1	_	12.8	6.8	13.2	- ci	0. 1	z	z	z	겁	31	653
21	167	34	4	r-i	4	_	9.6	5.5	13.7	Ľ.	0.4	z	z	z	2	Ξ	653
21	168	51	Ş	¢I	Ŧ	_	15.2	6.9	12.0	2.9	0.6	z	z	z	2	31	653
22	169	42	4	_	_	~	5,9	6.5	0.11	구 다	0.4	z	z	z	125	65	4
52	170	36	_	_	_	rē.	4,8	5.5	0.1	0.8	0.9	z	z	z	125	<u>5</u> 9	4
22	171	40	ŝ	_	_	ب ،	5.6	6.5	7.4	8 .0	30. 	z	z	z	125	<u>č</u> 9	4
22	172	36	4	_	_	ŝ	6,0	6,7	7.2	4.7	ε.I	z	z	z	125	65	41
22	173	37	_	_	_	r۴	5.0	6.4	8.6	ণ্ ব	0.0	z	z	z	125	65	41
52	174	44	0	_	_	۰	4.5 2	5.9	8	म्	4 -	¥	z	: ;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;;	125	65	4
72	175	35	ŝ	_	_	m	5.1	5.5	<u>י</u> י	1.0	0.1	z	z	z	125	65	4]
22	176	45	ĊÌ	_	-	Ē	ы С	5.5	0.7	2	90 ⊂1	∻	z	c	125	65	┯
53	177	35	ŝ	-	-	en ((i)†	7.0	9.6	1.7	0.0	z	z	÷	=	68	Ĩ
2	178	ភ	4	_	_	m	5.1	6.8	L.,7	E. El	<u>6.</u>	ż	z	. . .	Ē	89	Ξ
23	179	25	~	_	_	ŝ	5.9	5.8	3.6	90 C 1	0.6	z	z	z	1	68	Ē
23	180	34 4	÷	-	_	ŝ	5.0	6.0	-	3.0	0.8	۲	z	: :	13	68	EII
23	181	ŝ	m	-	_	~	6.2	6.3	4.0	3.9	1.0	z	z	z	113	68	Ē
ង	182	8	~		_	~ 1	6.0	6.1	9.6	8.6	0.4	z	z	z	113	68	113
23	183	26	÷	_	-	ŝ	6.0	6.4	6.9	13.5	61	z	z	z	113	68	Ξ
23	184	08	_	_		ŝ	5.6	5.8	99. 	6.1	0.5	z	z	z	113	89]]3
24	185	24	c 1	e	- it	C 1	75.5	6.2	5:1:4	0.4	0.1	z	z	z	154	34	131
24	186	58	ব	Ē	-1	rı	76.7	7.0	55.5	0.6	0'0	z	z	z	154	34	Ξ
24	187	61	Ŧ	m,	-7	ר ו	62.7	7.2	62.1	0.6	0.1	z	z	z	154	3	Ξ
24	188	44	4	Ē	4	c1	£'69	7.0	60.6	0.7	0.0	7	z	U U	154	34	131
24	189	59	Ŧ	÷	4	6 -1	62.9	2.2	56.2	0.6	0.0	z	z	z	154	31	131

18

|

:

Apper	adix 8,	Enviror	omental	data su	ynnary	. pà dnad	rat (]).										
₽A#	8	#SPP	HM₩	MdS	8	MOIS	MOS	튭	CALC	Ca:M	s K:Mg	OMH	SEEP	DIST	Vd	ξ	4
24	190	44	-	- ~	-7	~	74.9	8.0	69.8	0,4	0.0	Y	z	0	154	코	131
24	161	5 I	9	r n	4	r-1	64.1	7.0	59.2	9.0	0.0	۲	z	0	154	ц4	[3]
24	192	47	4	m	4	c)	67,0	7,0	50.2	0,7	0.1	z	z	z	154	34	Ξ

<u>.</u>
_
-
guadrat
à
summary
data
Environmental
ź
endix
슶

APPENDIX 9

ENVIRONMENTAL DATA SUMMARY BY QUADRAT (II)

<u>Legend</u>

Column Heading	Description	Unit/Label
PA#	patch number (Iabel)	numeric label
Q#	quadrat number (label)	numeric label
#SPP	# species recorded in quadrat	number
#GEN	# genera recorded in quadrat	number
#FAM	# families recorded in quadrat	number
#ANN	# annual species recorded in quadrat	number
#BIE	# biennial species recorded in quadrat	number
#PER	# perennial species recorded in quadrat	number
#NAT	# native species recorded in quadrat	number
#ALI	# alien species recorded in quadrat	number
#F	# species with affinity for forest habitats	number
#FO	# species with affinity for forest + open habitats	number
#OF	# species with affinity for open - forest habitats	number
#O	# species with affinity for open habittats	number
СТ	cover type class:	
	red/white oak (no sugar maple)	1
	red/white oak + sugar maple	2
	sugar maple (no red/white oak: no wet-mesic, wet species)	3
	sugar maple + wet-mesic or wet species	4
	wet-mesic or wet species (no sugar maple)	5
#LTS	# live tree stems (all diameter size classes)	number
#0-4	# live tree stems 0-4 cm dbh	number
#4-10	# live tree stems 4-10 cm dbh	number
#10-30	# live tree stems 10-30 cm dbh	number
#>30	# live tree stems >30 cm dbh	number

NOTE: species attributes may not sum to species total since unclassified species not included in summary.

PA#	Q#	#SPP	#GEN	#FAM	#ANN	#BIE	#PER	#NAT	#ALI	#P	#FO	#OF	#0	CT	#LTS	#0-4	#4-10	#10-30	#>30
1	Т	57	49	30	0	3	54	48	9	2.	6	29	8		16	12		0	0
1	2	53	45	31	0	2	\$1	45	8	2	10	24	4	2	108	100	3	5	0
1	3	44	37	26	0	0	44	41	3	13	8	15	0	2	32	22	Э	5	2
1	4	55	41	30	t	0	54	51	4	10	13	18	0	2	56	51	2	3	0
1	5	32	26	22	0	0	32	28	4	7	6	9	2	2	43	27	10	5	1
1	6	38	32	25	0	0	38	37	I.	6	10	10	I.	3	29	15	7	7	0
1	7	41	33	26	0	0	41	38	3	7	12	10	1	3	57	39	14	3	1
I	8	44	34	25	0	0	44	41	3	11	II.	12	1	3	38	27	2	3	6
2	9	63	47	31	0	2	61	58	5	16	24	8	3	4	80	60	13	7	0
2	10	58	45	29	0	1	57	53	5	61	19	9	2	4	108	97	8	3	0
2	U I	56	44	28	0	I I	55	50	6	15	19	7	4	4	109	78	21	10	Û
2	12	59	41	28	Û	2	57	53	6	13	21	8	3	4	84	57	20	7	0
2	13	58	47	30	I I	1	56	53	5	16	22	9	1	3	89	70	13	6	0
2	14	62	47	31	0	2	60	54	8	18	22	12	3	4	94	74	13	7	O
2	15	37	30	24	0	0	37	34	3	11	10	6	1	3	78	68	4	6	0
2	16	57	42	29	0	1	56	52	4	16	21	7	5	4	100	72	26	2	0
3	17	42	31	24	I	0	41	39	2	9	18	9	0	3	168	160	2	5	1
3	j X	51	37	27	1	1	49	-18	3	13	16	11	2	3	35	32	I	0	2
3	19	49	34	28	1	0	48	47	2	10	20	L I	0	-1	55	52	3	0	0
3	20	36	29	23	I I	0	35	34	2	11	13	4	0	3	101	97	1	3	0
3	21	47	34	27	I I	1	45	43	4	17	12	9	۵	3	95	91	1	3	0
3	22	56	39	26	1	2	53	52	4	11	18	18	4	3	137	135	0	2	0
3	23	35	29	20	2	ì	31	32	2	12	13	ó	1	4	9	6	1	1	1
3	24	30	27	21	I .	1	28	29	I	12	11	4	0	З	24	13	6	5	0
4	25	107	77	39	4	4	99	84	22	11	24	-14	16	3	75	59	13	2	L
4	26	113	79	42	5	6	(0)	84	27	8	24	51	18	4	34	28	3	3	0
4	27	49	37	26	0	2	47	41	8	9	17	14	2	4	90	65	20	4	l
4	28	59	45	29	I I	0	58	53	6	15	21	11	3	4	79	62	8	8	4
4	29	37	32	23	3	0	36	34	3	9	12	7	1	3	115	96	61	2	Т
4	30	86	ól	34	4	2	80	75	10	12	24	33	8	4	-18	42	6	0	0

Appendix 9. Environmental data summary by quadrat (II).

PA#	Q#	#SPP	#GEN	∦FAM	#ANN	#BIE	#PER	#NAT	#ALI	#F	#FO	#OF	#O	СT	#1.TS	#0-4	#4-10	#10-30	#>30
4	31	76	54	31	4	4	67	64	9		22	23	8	4	54	38	14	2	0
4	32	72	50	28	3	I I	69	65	7	13	23	24	4	-4	78	51	21	6	0
5	33	52	38	22	1	1	\$0	45	7	14	11	15	4	3	63	35	17	L1	0
5	34	40	32	19	1	0	38	38	2	16	9	5	0	3	53	21	25	6	1
5	35	45	36	22	0	0	45	39	6	12	12	11	2	3	56	23	19	14	0
5	36	39	31	21	0	1	38	35	4	ł4	9	6	1	3	53	25	20	8	0
5	37	29	25	15	1	0	28	27	2	[]	7	5	0	3	46	11	26	9	0
5	38	47	37	22	I I	2	44	41	6	[]	11	13	3	2	77	55	11	11	0
5	39	31	26	20	0	0	31	29	2	13	6	3	1	3	43	14	17	12	0
5	40	38	31	20	0	Û	38	35	3	12	8	11	1	3	60	31	21	6	0
6	41	25	21	18	0	0	25	21	4	3	9	7	0	4	58	46	4	2	I.
6	42	50	43	32	0	1	49	45	5	14	14	10	0	-4	79	54	17	8	0
6	43	56	43	29	0	1	55	48	8	12	19	11	5	5	47	27	13	7	0
6	44	58	46	27	0	1	57	50	7	15	12	14	3	4	109	77	24	7	1
6	45	53	44	29	0	I I	52	48	5	13	15	11	l	4	93	63	25	5	0
6	46	44	35	24	U	0	-14	39	5	13	11	7	I	4	5.5	34	13	7	t
6	47	64	49	33	0	2	62	56	8	15	16	١j	4	4	93	60	22	11	Ð
6	-18	66	47	31	1	2	63	60	6	17	17	16	5	4	8 1	58	12	11	0
7	49	48	38	25	Û	0	48	44	4	13	10	15	0	2	44	30	5	7	2
7	50	59	43	24	0	I I	58	52	7	11	12	22	2	2	66	59	5	2	0
7	51	70	49	30	0	0	70	65	5	19	18	19	I I	2	99	95	2	2	0
7	52	33	27	22	0	0	33	32	I I	5	12	6	0	2	41	27	7	6	I I
7	53	46	37	25	0	0	46	-13	3	7	12	14	3	3	53	42	9	0	2
7	54	67	46	28	0	0	66	59	7	15	15	23	2	2	86	77	5	3	l
7	55	50	40	30	Û	0	50	46	3	11	16	12	D	2	82	73	5	2	2
7	56	48	34	21	Û	0	48	44	4	12	12	13	1	2	63	55	Э	1	4
8	57	30	25	19	0	0	30	29	1	8	9	3	0	3	131	126	4	0	1
8	58	16	15	12	0	0	16	14	2	3	4	2	0	2	66	52	9	3	2
8	59	14	t4	11	0	0	14	13	1	3	2	2	0	2	41	34	2	3	2
8	60	21	19	15	0	0	21	20	1	7	6	0	0	3	49	39	6	4	0

Appendix 9, Environmental data summary by quadrat (II),

₽A#	5	ddS#	#GEN	MEAM	NNN#	#1316	#PEK	T.VN#	ľ'IV#	ιF	βFΟ	:IO#	0 ⁴	5	¥LTS	#0-4	#4-10	#10-30	€>30
÷	19	17	- 2	Ξ	0	0	11	16	-	-	~	•	•	~	\$	6	Ξ	•7	A
8	62	5	6	5	0	c	21	20	_	5	9	ri	•	•0	\$	Ē	Ξ	4	0
-0	3	21	17	13	0	0	2	20	_	9	ir.	ĊI	0	ب	97 71	E]	•	ŝ	_
	64	25	61	15	_	0	17 17	5	r.	ç	Ś	c,	0	C1	50	30	7	_	-
6	65	33	30	ដ	0	0	33	31	•	ų	.	21	3	7	117	1D6	÷	1 271	2
6	66	Ä	56	24	_	0	EE.	33	Ч	r)	엄	2	_	- 1	58	<u>5</u>	m	ъ	_
6	67	35	38	21	C	Ð	÷۲.	35	0	+	۲.	2	C1	61	54	68	ŝ	~.	_
0	68	37	29	ដ	ç	0	37	36	-	7	5	2	0	c :	<u>8</u>	63	0	÷.	_
6	69	46	38	28	ç	0	·Ió	먂	ŝ	80	Ê	<u> </u>	ç	÷	67	6)	-	ŝ	0
¢	70	æ	27	53	¢	¢	ž	단	1 1	ъ	4	;+		ci	86	75	ŝ	Ó,	0
¢	71	37	0M	25	0	0	37	34	~	r-	۲	2	-	-1	112	98	ć	ন	ń
÷	ť.	4	35	24	0	0	2 1	90	ŕ	۲	<u> </u>	17	÷	r.	116	6 01	Ţ	e	0
2	73	53	4 6	33	ы	_	50	47	9	5	2	×	~;	ŝ	2	25	6 1	2	
01	74	48	1	33	C1	_	÷	4 3	ŝ	20	ĉ	Ŷ	•	ĉ	102	64	¢		_
10	75	28	<u>36</u>	5	-	0	27	27	_	ŝ	÷	r	-	r .	62	70	•	C 1	0
01	76	33	30	ន	-	ċ	5	ŧ	_	15	≘	ĸ,	0	•	7	58	.,	~	_
2	77	Ξ	ទ	7	L †	a	<u>ب</u>	1 1	0	8	×	~	÷	•	<u>3</u> 6	eļ,	r.	r.	_
01	78	99	₹	ы	6 1	_	57	57	~	21	6	~	0	r.	16	89	0	ç	••
10	62	47	30	62	CI	_	구	1 1 1	ŝ	20	Ŧ	5	0	rt.	2	76	ē	4	0
01	80	I .	ž	27	_	c	97	9	_	51	6	m	0	-	ñ	27		ŝ	_
Ξ	8	96	3	Ē	2	5	88	74	20	ŝ	F	£1		ŝ	66	56	6		0
Ħ	8 2	85	60	с Г	_	r h	82	3	5	+	17	37	ŝ	¥n.	30	26	÷	0	_
=	63	76	Ľ,	<u>6</u> 2	51	Ś	69	50	13	-	Ξ	38	5	w.	55	¥	۲	-	•
=	84	ž	6 5	31	_	-7	56	65	6	77		40	38	in i	104	\$	÷	_	•
=	85	62	5 6		c-1	~1	75	62	16	+	5	38	<u>:</u>	Ś	92	86	Ŧ	L 1	•
1	86	16	61	Ā	2	Ð	69	78	<u>.</u>	ŝ	50	0 1	ŗ	.	181	168	=	_	_
11	87	68	65	36		ŕ-1	8	77	-	ve.	61	45	01	.	5	137	<u>11</u>	-	•
Ŧ	88	88	54	ñ	2	0	86	74	13	Ŧ	5	46	~	Ś	114	<u>6</u>	*	2	÷
2	68	57	45	28	_	•	56	55	~	5	50	¢	a	Ţ	43	ŝ	'n	3	4
ដ	90	58	45	20	0	0	28	57	-	56	52	en.	ç	4	ñ	15	æ	6	61
	:													:			İ	:	

ij>30	_	<u> </u>	-	0	<u>د</u> ا	-	o	-	2		c 1	0	-	c i	-	_	÷	÷	-	¢	c i	-	P7	•	÷	9	c	¢,	•	-	ļ
#10-30	÷	\$	m	=	'n	9	o	61	0	-	ŝ	_	сı	0	4	er.	Ŀ	¢1	rı	ŝ	m	m	ŝ	m	'n	Ś	- -	÷	¢	4	
#4-10 	ы	•0	œ	인	2	Ŧ	r-	¢	ς,	•	-	2	Ģ	¢	≏	5	61	-	∝	¢	~	12	Ó,	cı.	s	0	_	~	Ð	Ś	ļ
7-0#	135	4	35	9	13	24	17	4	17	-	÷.	œ	16	96	œ	м. т	ur.	ιń.	ង	ភ	6	¢	Ŷ	16	먹	J	5	Ξ	2	ñ	
#LTS	17	8	ŝ	76	2	35	5	17		6	53	5	ŝ	9ß	$^{-56}$	57	R	20	Ъ4	66	32	52	99	 -	2	¢.	۳	5	7	53	
<u>1</u>		-	-7	न	-1	r*1	'n	÷	2	r	έ	دم ا	r I	Ċ)	r ~ .	r * .	æ.	m	÷		ب	~	-1	c i	~	r1	e,	c 1	m	r.	
ę	0	_	0	0	_	-	•	•	0	•	-	0	0	0	0	-	=	•	0	0	ð	0	-	•	¢	0	1	0	_	÷	
#OF	•	9	2	*	91	~		_	٢I	ŕ١	᠇	-	-	-1	-	٢	-	-	m	m	17	-	~	¢	ç	7	v.	v †:	01	6	
#FO	\$	17	61	17	8	8		¢	9	7	÷	5	œ	Ś	~	9	÷	'n	•	6	-	7	-	×	Ģ	Ŷ	7	Ŷ	6	\$	
4F	6	61	<u>•</u>	33	ភ	61	¢	w.	7	÷	Ś	Ś	ŝ	٢	•~	\$	¢	ŝ	÷	~ 7	Ŧ	Ŷ	ŝ	¤≎	ç	¢.	Ś	90	ŝ	ŗ	
1.1At	-7	ś	c)	Ś	m	¢	_	_	0	0		c	c	¢	C1	m	÷	=	_	_	_	0	_	cı	_	c	•	c	¢	c	
TAN#	20	53	<u>%</u>	64	57	6 1	2	20	20	17	5	17	18	되	₹	5	5	2	1	1	18	91	50	27	23	ĥ	Fi	36	30	25	
ИЛЕК	63	55	57	51	58	5	2	31	20	17	23	17	<u>8</u>	12	9	27	ຄ	1	15	1	61	16	30	29		25	គ	26	8	25	
#11H	0	_	0	-			0	0	0	0	÷	0	0	0	0	-		¢	0	0	0	0	0	0	0	0	0	0	÷	•	:
NNK	0	~	_	~	. →	2	G	•	0	0	0	•	a	0	ð	ç	0	0	•	0	0	c	0	0	0	0	c	0	•	0	
#FAM	2	56	50	5	ЗІ	5	15	1	2	=	15	5	2	5	13	61	5	Ξ	5	=	13	13	17	17	2	ŝ	ŝ	38	81	Lد اد	
#GEN	51	47	45	ç	47	50	8	61	16	5	52	۱¢	-	71	5	25	7	<u>•</u>	Ξ	n	16	Ŧ	25	26	E.	ន	21	ล	28	ន	!
ddS#	62	58	<u>.</u>	54	60	55	2	5	8	17	5	17	. 22	2	16	20	곱	r:	v :	- 	61	9†	30	ត	77	55	23	26	30	25	
ð	16	<u>7</u> 6	60	46	56	96	70	86	66	100	101	102	5	8	105	106	107	108	601	021	Ξ	211	Ê	17	115	911	117	811	119	120	
₽Λ#	 []	-1	12	12	12	12			12	1					<u>+</u>	4	4	٦ ا	ন	4	4	4	15	15	2	15	2	15	5	2	

#>30	-	Ċ	• •) r	4 -		1 6	1 -		• •	h¢	5	• <	= (- 1 -	•	•	c	¢	0	-	-	ç	¢	-	ç	Ģ	C	, c	•
#10-30	-	4				n r	ιr	ካና	46	4 4	ń -			_ ,	- e	-		<u>^</u>	⊅	6	÷	13	'n	Ś	œ	m	~1	s	÷			
#4-10	1	0		, -:	- 1		- -	•	<u> </u>	2 '			2 4		- ı	~ 1	-1	7	8	=	55	61	18	9	Ξ	~	\$	-1		<u>।</u> न	r -	t
#0-4	g	81		1 -		9 9	4	= =	2 8	9	n p	5,2	ទុទ	<u>-</u>	0	37	\$	71	ŝ	2	•	15	1	8	¢	34	68	97	02		72	ę.
#LTS	11	: :	15	4 8	? ?	1	2 :	27	1	5	<u>e</u> ;	ζ; :	; ;		2	5	¢	63	2	36	무	47	39	07	28	44	10	역	: ¥	;	32	2
5	~	n e	- -		. -	. -	. ,	- - ,	-, ,	- - 1	~ ^	·	-	·••	- - -	~	~	~	r,	r.	-,	-	C)	C 1	۴ı	~		. ~		- -	+ ·	Ŧ
Ç#	-			•	• •	- 1	•	0 4	0 4	0	0 (5	0	=	<i>v</i> .	÷C.	Ξ	÷	•	e	•	_	0	c	0	ð	0	; -	1		ц ,	-
#OF		2 1	•	Ŧ ,		5	- 1	r 1	-		- 1	<u>~</u> L ·	-	16	16	=	91	+	÷	4	5			\$	9	9	2 ר		5 4	<u>م</u>	<u>e</u> :	Z
04#	5	2:	- '	- :	2:	<u>न</u>	9	<u>च</u>	<u>-</u>	Ŧ	Ŷ	ন	ę	16	-	7	=	œ	w.	7		. .) -1	6		- 41	2 -	2	2 1	<u>-</u> ;	<u>ی</u>	×
14	15	2 9	2	•	: :	0	Ξ	[]	÷	×	디	F	9	9	건	Ŀ	۹.	Ξ	Ģ	9		9 9		¢	=	: =	2 2		1		9	<u>1</u>
		- '	<u>^</u>	- -	·••	Ś	4	7	и л ,	_	4	~1	_	91	16		٩ ا	_	_	_							• ^	-	n -	_	Ģ	ia.
∦NAT		P	ħ	ព	36	33	35	36	23	<u>-</u>	N,	<u>.</u>	61	Ŧ	47	2	4		Ē	5	1 7	10	9 F	i f	i é	90	3 7	2	£ 3	39	2	22
#PER		4]	37	24	37	38	37	38	33	Ξ	28	5	50	47	Р.	ę	48	F	F	12	ł ۲	19	í ž	2 9	35	1			2	Q	33	55
a BIE		•	_	_	_	_	_		_	ċ	0	c	ç	ç	÷	~			; :) c			> <	, ,	2 4	2 4		= ;	-	c		_
NNV#		-1	_	_	_	_			. →	0	Q	•	0		দ	•		. 5	2 4	2	2 4	2 4			> <	5.		•	0	•	-	۲ı
W V I		5	27	20	5	5	23	30	2 1	6	21	2	4	5		<u>•</u>	5	R 1	2 1	<u>1</u> -	<u>-</u> :	25	-	: :	<u>r</u> :		Ľ,	56		цч Г	36	33
W SEN		8	F	21	29	90 90	11	27	31	<u>11</u>	25	4	~	; Ç	5	12	1	1		2	<u>e</u> :	61 2	រា ។	2 2	9		ज न	-	7	8	9 <u>5</u>	48
anan	1104	4	39	26	39	9	30	40	9 6	1	80	<u>•</u>	20	ž	3	12	45		1	7			2	¢ :	3	5	57	41	5	0Þ	76	58
č	5	[2]	122	123	124	125	126	121	128	90	130	131	::	1 2	1.2				2	<u>s</u>	<u>8</u>	<u></u>	1	2	143	1	145	941	147	148	149	150
	"V"	91	16	16	16	9	9	2	2 9	1	: 5	-	- 1	22		2 2	2 5	2 9	z	2		18	<u>2</u>	×	8	18	5	61	61	61	51	6

489

.

...........

:

01.74	00~#	0	7	0	m	_	ה ו	ç	ŝ	~1	¢	0	_	¢	0	_	m	~1	-	-	-	0	-	0	-	0	-	¢		0	a
	HIU-3U	'n	Ś	_	Ŧ	Ģ	-	'n	Ξ	_	•	νn	•	Ģ	•	Ŷ	-	-1	_	ŝ	~	=	12	œ	4	r-	4	-1	~1	=	
	N-52	ç	ŝ	4	2	r,	Ģ	ч	ŝ	0	ŕι	_	_	-1	m	0	_	ç	ŕ1		1	~	r)	6	2	Ŀ	2	0	┯	m	ŝ
1. Ou	80-4	3	77	7	12	5	÷	শ	~	33	~	ŝ	33	\$	7 9	89	위	2	8	27	45	=	8	R	5	981	4 6	v'n	잌	01	80
	ר <u>י</u> י	36	68	2	2	ม	<u>ല</u>	6	<u>3</u> 6	36	3	35	35	<u> </u>	67	8	47	4	÷	#	65	47. M	48	γn T	Ę	120	63	9	ž	7	86
	5	न्त	7	Ś	Ŧ	Ŧ	Ś	Ţ	Ŧ	Ś	Ś	r,	-	m	+	ч	-7	-+	-+	Ċ	~1	r.	~1	~1	~ 1	e.	CI	<u>~</u> 1	~	0	r,
	5	m	-	+	0	r;	-	0	Ð	-	ð	0	ð	•	_	۲ı	0	÷	÷	-	-	-	~	C1	_	0	ei,	ç	-	÷	÷
		ŝ	51	2	n.	91	=	30	**	5	ŝ	~~	J	CI	<mark>ମ</mark>	=	w.	ç	17	2	¢¢	Ξ	=	39	01	¢	01	~	6	5	•
	044	61	5	n	ġ	5	17	=	¢	17	ç	6	r-	9	7	12	×	2	5	¢	Ξ	Ξ	0	<u>5</u>	Ξ	2	13	ŝ	'n	-1	0
	.	14	61	8	20	•	÷	80	Ś	m	_	=	-	=	œ	4	16	Ť	린	n	9	¢	ŝ	¢	÷	6	12	13	Ŷ	×	0
-	#VEI	Ŧ	4	~	ŝ	2	ŝ	ŝ	Ŷ	4	č 1	c	رب	0	••	×	¢1	ei	əc	¢1	-	-	-	_	_	-			ר ו	_	_
	1 Vu	54	62	33	ង	47	36	27	8	:	5	72	24	30	-	÷	30	5	Ŧ	₽₽	35	39	35	36	1	#	4	t.	러	57	E
	#PISK	55	64	33	27	53	36	30	24		=	27	77	30	45	÷8	5	ż	8÷	42	96	-10 1	36	37	ţ	35	45	35	26	č	27
		0	0	Ţ	_		0	_	_	0	0	0	0	0	~1	_	_	0	_	0	0	0	•	e	0	•	0	0	0	0	9
	NNAN		-1	ন	c		4	_	G	rī.	Ŀ,	0	c	0	r.	_	ċ	G	~	c	0	0	Đ	0	0	0	•	c	0	0	0
	WE AM	5		52	61	67	5	5	61	5	2	-	20		3	27	ព	2	5	5	×	54	50	2	57	20	2	2	51	17	50
	NHU	5	12	36	13	9		3	2	Q	4	Ŧ	T.	ŝ		50	62	£.	£	99	82	5	-	8	53	62			2	-	5
	AdS	*	9	=		5	2	2	5	5	4		5	0	0	0	2	-	-	2	9	9	2	11	4	2	5	5	9	123	
		51	52	5.	54	55	56	57		55	90	61	13	63	64	65	66	5	89	69	10	16	12	73	4 C	75	76	17	8	10	80
		-	-	-	-	_	-	-	_	_	_	-	_	_	_	_	-	_	_	-		-	-	-	-	_	-	-	_		

490

																· _··-			
PA#	Q#	#SPP	#GEN	#FAM	#ANN	#BIE	#PER	#NAT	#ALI	4F	#FO	#OF	#O	СТ	#LTS	#0-4	#4-10	#IQ-30	#>30
23	181	32	26	21	0	0	32	31	1	8	9	8	0	2	32	10	13	8	I.
23	182	18	17	12	0	0	18	18	0	6	3	4	CI I	3	49	45	0	3	1
23	183	26	21	15	0	0	26	25	t	9	6	6	0	2	23	15	2	5	1
23	184	30	27	20	0	0	30	29	l	10	8	5	0	2	31	25	1	3	2
24	185	24	17	16	0	l	23	21	3	6	8	2	0	3	54	39	10	4	1
24	186	58	40	25	ł	2	55	50	8	14	14	14	3	4	65	43	17	4	i
24	187	61	46	30	1	I	59	54	6	B	19	17	0	-1	103	95	3	3	2
24	188	44	30	20	1	1	42	37	6	11	9	16	2	\$	17	15	2	0	0
24	189	59	41	30	1	I	57	51	7	13	15	16	1	+	143	129	6	7	t
24	190	44	31	25	1	I	42	36	7	8	13	10	1	4	51	27	20	4	0
24	191	51	40	28	1	I	49	44	6	i I	17	12	0	4	92	85	3	1	3
24	192	47	35	27	1	1	45	40	6	٤I	13	10	1	4	47	39	3	4	l

II)))	l	•)))	1	•]	(ł	1	,		ł	d	1	ï	ſ	1	1	í	l	9	ξ	ŀ	l	l	l	ļ	(ſ	Ý)	3	ļ	1)	ł		ŕ	>	7	ľ	3	î	lí	1	ľ	Π	I	Ì	n	r	ł	1	l	L	1	5	5	1	I	Ì	1	1	ć	1	t	t	t	1	1	1	ź	1	1	C	5	I	l	1	8	ŀ	ţ	ļ	1	1	ľ	l	1	1	2	ç	Ę	ć	ŀ	1	ľ	1	1	ļ	Ì	1	Ì	I)	3	C	¢	1	ŋ	1	ľ	ľ	ľ	1		ļ	ļ	ļ	ļ	ļ	ļ	Ì
1		ļ	I))	ļ	ļ	ļ]	(ł	1	,		ł	d	1	ï	ſ	1	1	í	l	9	ξ	ŀ	l	l	l	ļ	(ſ	Ý)	3	ļ	1)	ł		ŕ	>	7	ľ	3	î	lí	1	ľ	Π	I	Ì	n	r	ł	1	l	L	1	5	5	1	I	Ì	1	1	ć	1	t	t	t	1	1	1	ź	1	1	C	5	I	l	1	8	ŀ	ţ	ļ	1	1	ľ	l	1	1	2	ç	Ę	ć	ŀ	1	ľ	1	1	ļ	Ì	1	Ì	I)	3	C	¢	1	ŋ	1	ľ	ľ	ľ	1		ļ	ļ	ļ	ļ	ļ	ļ	Ì
APPENDIX 10

ENVIRONMENTAL DATA SUMMARY BY QUADRAT (III)

<u>Legend</u>

Column Heading	Description	Unit/Label
PA#	patch number (label)	numeric label
Q#	quadrat number (label)	numeric label
#SPP	# species recorded in quadrat	number
#TRE	# tree species recorded in quadrat	number
#SHR	# shrub species recorded in quadrat	number
#VIN	# vine species recorded in quadrat	number
#FRN	# fern species recorded in quadrat	number
#FA	# fern ally species recorded in quadrat	number
#GRA	# grass species recorded in quadrat	number
#HRB	# herb species recorded in quadrat	number
#INH	# herb species dispersed by animal ingestion	number
#ADH	# herb species dispersed by animal adhesion	number
#ANH	# herb species dispersed by ants	number
#WIH	# herb species dispersed by wind	number
#PDH	# herb species dispersed by prolonged dormancy	number
#MEH	# herb species dispersed by mechanical expulsion	number
#NFH	# herb species dispersed by propagules with no	
	facilitating morphology	number
#MUH	# herb species dispersed by multiple modes	number
#VEH	# herb species dispersed by vegetative expansion	number
	-	

S# #0	¦₹	Ê	JTRE	#SHR	NIN#	#1.KN	#FA	#GRA	#JIRI3	#IN1	HCIAN	INAN	III/M#	HCI4#	#MEH	\$NFI-	HUM#	(VEH
1 27 12	57 12	2		9	س	Ē	Ē	\$	ន	. ~1	~	Ē	입	7		61	Ļ	먼
2 53 12 1	53 12 1	- ป		_	ŝ	C 1	_	m	21		m.	÷	¢	ç		-	÷	₫
3 44 8	44 B	8		**	Ŧ	r)	0	-	20	\$	-7	-	Ś	-	•	÷,		10
4 55 14	55 14	- -		≘	~	ר י	0	r.	ы. П	-	9	3	•	~1	÷	~	4	=
5 32 8	32 8	æ		و	- 1	പ	0			÷	_	~	eri)	. 1	•	C1	-	Ģ
6 38 11	38 11			J	J	ŝ	o	_	5	_	-1	-	v.	¢	0	ч с э		9
7 41 11	41	=		•	•	د ،	•	0	91	2	-1	4	ŝ	0	0	m	•	60
8 44 9]	44 9	•	-	0	CI	_	c	-1	50	4	4	'n	-7	_	Ð	9	ci	5
0 63 11	63	=		-		4	-	~	38	÷	w.	F-	9	►-	-7	÷	6	ŝ
10 58 10	58 10	2		•	-	ŝ	-	5	LŻ	•••	r")	5	¢	Ŧ	۳ı	ŝ	Ŷ	2
11 56 9	56 9	6		ν ή ι	_	Ģ	-	ŝ	20	4	ŝ	7	ç	¢	-	÷	5	뎊
12 59 13	59 IJ			. vo	Ċ1	ŝ	_	~	55	-	m		20	•	~1	×	Ŀ	<u>म</u>
13 58 10 10	28 10	9		. च	•••	r-	9		ŀ	~	~1	Ģ	9C	\$	7	~	9	9]
14 62 7	62 7	"		• ∩	ŝ	9	_	_	39	9	w.	**	œ	-	~	6	Ŀ	61
6 15 51	37 9				_	- P I	0	_	20	-	_	ň	~	-	a	9		Ξ
16 57 9	57 0	÷		J	e.	ŝ	-	~1	92	-	\$	9	-	r.	~	ъ¢	in.	<u>-</u>]
17 42 A	÷	ч с .		ŕ١	0	÷	0	r:	R	•1	ŝ	-	ŝ	21	-1	÷	m.	Ţ
18 51 9	51 0	÷		ç	_	-	ç		æ	-1	ç	~	ç	÷	<u>م</u>	æ	۰'n	9
8 61 61	8 67	÷		•71	61	5	0		30	~	ŝ	Ś	\$	_	r,	=	ι η	F
20 36 8	36 8	80		_	_	Ś	-	Ċ	50	+	ń	•	~	a	r,	cı	e	11
21 47 8	47 8	~		m	_	Ţ	0	o	ы	-+	ŝ	6	-	_	7	~	-1	15
22 56 4	56 +	4		m	~ 1	+	¢	ŝ	37	2	÷	6	Ξ	ć	Ś	2	÷	Ξ
23 15 2	15 2	۲ ۰۱		_	_	0	•	-1	20		r,	m.	¢1	C1	Ŧ	9	2	6
24 30 3	30 3	ŝ		_	_	ç	0	¢	5		7	4	-	-	Ś	J	-1	9
25 107 10	107 10	2		r.	ŕ'n	v.	~ 1	5	65	-	Ξ	ę	61	립	-	<u>5</u>	Ŧ	97
26 113 8	113 8	æ		o	÷	7	~1	ŝ	73	-	9	¢	33	33		R.	3	28
17 49 6	4 67	÷		m	ŕ'n	~1	0	¢1	22	4	-1	Ģ	r	×	¢	,	4	4
28 50 0	50 0	•		v:	~~;	2	0	÷	37	5	L 1	5	6	9	e.	ŗ	~	2
29 37 8	37 8	æ		•••	_	_	0	-	ព	ş	m	m	5	. 1	_	'n	ተነ	<u>51</u>
									!		:		!		:			ļ

Appendix 10. Environmental data summary by quadrat (111).

±94

Appendix 10. Environmental data summary by quadrat (III).

UH #VEH	51	5	6	91	Ξ	EI.	6	0	<u>.</u>	2	4	90	٦	17	<u>:</u>	=	0	দ	<u>8</u>	=	4	١¢	in.	6	8	1	2	
W#	•0	9	÷	•~	ŝ	'n	¢1	n,	6	r,	۴٦.	-	c	m.	сÌ	rı	-	⊸	'n	ы	'n	CI	¢	r-I	-	r)	•	
HJN#	4	ŝ	2	ŝ	ç	۴	r'i	ы	Ŧ	ţ-1	m	en,	v٦	96	9	Ŧ	÷	œ	2	Ţ	Ś	Ŷ		¥C.	¢	r n	4	
₿MEH	: י רו	~	-7	-	0		=	_	_	÷	_	•	_	_	_	_	э	~1	~ n	0	0		¢	_	_	ç	_	
HCId#	2	9	w.	9	~1	rris	er.	C1	÷	~	ŝ	_	r .(ŝ	÷	rı	- 1	Ś	ŝ	-1	ŝ	-	¢	r)	ŝ	2	-	
HIM#	6	17	=	15	9	30	ч г,	Ś	2	Ľ	20	-7	\$	0	20	9	w.	6	7	80	••	\$	m	×	ţ,	6	7	,
HNN#	- -		6	•	2	s	×	\$	×	9	9	F	Ŷ	–	÷	-	-	7	6	C 1	6 1	-	ب	-	.,	-7	÷	
#ADH		*	7	r h	ŝ	-	rt.	_	m	ŝ	ť	_	_	-	~ 1	_	0	6 1	.	c 1	r.	'n	_	_	~	ę.	e*.	
IINI#		÷	~	5	÷.	~ .	•	e	4	2	~	PT.	6	Ą	ŝ	4 7.	ę	7	7	Ţ	4	¢	6 1	~	2	6	-7	,
#HKB	- 81	49	1 3	Эd	26	25	53	17	50	20	54	1	54	22	27	5	50	31	37	20	39	30	10	5	33	51	20	,
kGK∧		4	7	ų	m	\$	~	-	vi	_	~	•	_		-	-	c 1		Ş	-	5	ŝ	_	_	~			
AFA	: • _	_	_	ċ	Ĵ	ð	0	C	0	0	0	_	0	-	_	-	•	_	_	-		-		0		_	0	
#FRN		•	2	_	_	_	_	_	_	_	_	_	-		-		·~	~	~	-	_	_ 	_	~	_	_	_	
N1V#		ب	۔ جا	ĉ	c	-	-	_	_		_	_	_	~	сі		_	~		_	~1	~	a	_		_	_	
#SHR	11	۰ n	<u>،</u>	-	-	~	2	0	¢1	-	•7	en	6	5		r	-		r~	0	2	15	- 	Ξ	1	1	Ξ	
WTRE	~	.	Ŀ	æ	\$	9C	6	9	÷	80	9	• ^	=	6	5	1	2	15	=	2		2	¢.	9	¢	0	01	
#SPP	86	76	72	52	01	45	66	29	47	16	38	2	50	<u>56</u>	58	8	Ŧ	64	99	48	50	70		46	67	20	48	
5	30	IE	32	E	Эd	35	36	17	38	39	-0F	41	ţ	()	÷	15	46	47	48	49	50	51	52	53	54	55	56	
μN	 	-+	+	~	~	5	6		~	•	~	9	2	9	5	5	æ	.0	<u>ر</u>		•	-		•	-	-	-	

. :

:

¥۲	ð	#SPI	#TRE	#SHR	۸IV	#FRN	₩FA	₿GRA	#IJRB	IINI	HGV#	IINV¢	HIM#	11014#	AMEH	₿NFH	#MUH	НЭЛ∄
 	58	9	٢	~1	. 0		: • •		. ~	. רו	¢	÷1	-1	-	0	_	. –	4
~	59	I	r	-	0	0	0	-	¢	<u>.</u>	•	-	2	0	0	_	0	4
20	09	5	œ	ν'n	Ċ	_	0	0	<i>۲</i> ۰	Ŧ	0	_		G	¢	_	2	-1
00	61	17	Ċ	ŕ"	¢	0	0	Ð	s	-	0	_	_	¢	¢	_	0	r,
80	62	2	×	ŕ١	¢	0	0	0	=	~	0	- 1	ŕι	_	¢	e i	-	6
80	63	2	•	÷	c	c	0	0	2	-	•	m	_	¢	ç	~	e	Ó
-	64	25	₫	•	c	_	Ċ	0	=	•	0	-1	r)	-	_	ŝ		ŝ
6	65	R	9	<u>e</u>	e	_	r.	c	=	-1	r:	_	r)	0	_	_	•	ŝ
•	66	ž	¢	×	4	-	¢1	0	10		_	ri	_	•	~1	m 3	0	4
•	67	35	¢	-	e	-1	_	G	<u>~</u>	4	_	-1	_	÷			÷	÷
•	68	37	=	÷	-1	٦	_	a	Ē	+	ŝ	¢1	_	÷	_	r-1	•	¢
•	69	1 6	Ξ	2	ŧ٩	•••	ĉ	6	15	m	~	_	~	_	=	s.	_	~
c	70	ų.	Ξ	2	_	-	_	ċ	≘	-	a	rı.	r,	÷	_	с1	•	•
•	71	37	2	<u>e</u>	_	_	0	8	15	ŝ	~1	~ 1	~	-1	_	न्त	_	æ
•	Ľ	Ş	'n	입	~	ŝ	0	0	ŝ	7	-	r:	~ .	÷	_	(1	0	Ξ
2	5	53	r	~ 1	~1	~1	=	۲	2	ب د.	۲ .	10	ь.	5	-1	r-	ч.	2
0	1 1	%	*	-7	¢1	~	0	r h	38	-	- †	=		'n	21	¥7i	-	¢
01	75	8 C	ন	4	_	6	•	e	1	r.	0	40	-	D	_	- T	0	ŗ
0	36	35	% 1	-1	_	÷	•	Ċ1	-	,	e.	×	-	-	_	ŝ	-1	۲.
0	77	34	ч г ,	~1	Ċ	ŝ	0	0	[]	~	e.	Ş	a	0	_	~	_	
0	78	() 9	≘	5	6 1	m	0	Ð	38	Ś	7	ŝ	¢.	-	_	=	6 1	13
01	<u>6</u>	47	vc,	Ċ.	ب ،	~	0	2	32	9	*0	ţ		¢1	~	~	~	2
0	80	- F	v.	œ.	_	-	Ð	2	26	è.	7	20	Ð	ç	_	¢	-	r-
=	8	96		Ξ	ה ו	_	_	•	6.1	-	5	~	25	16	+	30	2	5
=	숺	85	×	엳	сı	_	_	\$	54	۴ı	ŝ	ru	20	ŕ	0	50	•	5
=	83	76			rı	~1	_	25	17	_	v .	ተጎ	<u>년</u>	≞	-1	2	2	¢
-	8d	84	ñ	=	ŕ١	0	_	6	8 4	[5	~ 7	2	15		5	¢	61
][58	79	r-	•	~	_	_	5	5	_	~	~ !	5	5	~ n	17	÷	5
											:		:					

Appendix 10. Environmental data summary by quadrat (211).

96†

PA/r	5	#SPP	#TRJ:	#SILR	NIV#	HFRN	٨Чa	#GRA	#}IRB	±NI#	HOV	†INV#	‡WI1	HDH!	HEIM#	II:iN#	FMUH	#VEH
-		16	. 6	: =	:	; •	_		: . £	~1	t		8	2	i In	ខ	•	8
	5	68	5				-	•	53	r I	•**	-	19	믭	ن م	81	9	17
	3	5		2	. ~	· 广1	_	- 30	55	_	÷	-	61	9	-1	22	ŝ	ន
2	68	5	2	-	0	- 20	⇒		Ť	Ś	~	=	•	-	m	×	~ 1	20
::	5 9	5		· ~	. —		0	4	36	¢	m	9	÷	-	~	입	rı.	30
2 2			1	- 10	· rì	•	0	r.	50	Ŧ	ir.	•	•	1	¢.	88	m	16
2	. 6	3	: =	. –	; .	~	0	r.	Ę,	ν	~ 1	Ş	ŝ	Ŧ	v.	7	6	3
1 5	10	58	91	, in	_	ŝ		_	36	ą	÷.	2	-7	~	न	6	Ś	5
12	04	1	-	. –	· •	[~	•	- I	4£	, - 1	ţ.	ç	30	ন	ŝ	*	×.	2
1	56	(1 9			11	0	0	-1	ţ	9	<u>^</u>	ç	÷	4	4	r-	÷	2
12	36	\$	2		. c1	6	•	Ŧ	27	Ţ	÷	ŝ	. न	رم	- 7	ç	17	2
	16	17	. –	-	0	~1	0	¢	ŝ	m	r)	ç	_	Ċ	¢	ر م	0	₽
5	96	21	ç	=	_	~1	=	þ	Ը	ন	•	4		0	_	Ċ.)	•	5
	ŝ	02	. In	0	_	~	÷	_	9	⊸	Ģ	~ 5	rı		_	à	_	5
	100	17	~ 1		0	~1	=	0	2	.	_	ĸ.	÷	c	0	C 1	0	•
~		2	. ur.	т	-	_	÷	0	2	ю.	≏	¢	-	0	G	÷1	0	2
: "	201	17	-1	•	_	ŝ	ŧ	÷	÷	_	5	Ŷ	-	Ð	8	-	0	Ŀ
	6	×	- 1	-	0	~	÷	0	2	•71	•	Ŧ	¢I	0	Ċ	-	ð	Ċ
2 #	104	2		~ ~ 1	. 0	_	•	0	=	6 1	_	Ŧ	0	c	_	3	0	c
	105	91	Ś	Ļ	¢	_	à	=	7	m	_	Lt	_	0	•	0	0	~
1	106	% ^1	4	9	c	\$	0	=	<u> </u>	'n	-	m	7	m	۴ı	0	- - 1	\$
Ī	107	 1	Ś	4	0	m	Ġ	=	신	r 5		÷	-	_	_	C1	_	ŝ
4	108	13	· .	_	•	_	÷	•	~	᠇	_	C 1	0	÷	o	0	c	v
Ī	109	15	m	4	-		0	=	Ŷ	ri	_	~ 5	0	•	0	0	c	-7
4	011	آم ا	μ.	m	0	~	Ð	=	Ś	C 1	-	rı	•	0	C	¢	0	~
1		61	ŝ	гт,	0	÷	¢	÷	r-	m	-	21	-	=	D	0	G	-+
4	1	16	ŝ	rt,	0	0	0	÷	∞	m	-	- 1	=	c	8	^ 1	a	m
15	Ē	30	01	ŝ	¢1	_	¢	_	9	-7	0	-	_	0	0	7 1	D	æ
																	:	:

Appendix 10. Environmental data summary by quadrat (III).

101

.

:

(111).
quadral
ą
data summary
Environmental
10,
Appendix

					:									-		: :		
₽A#	¢#	4dS#	#'I'RE	#SHR	NIA	#FKN	ЧΓА	#GRA	âJ IRB	IINI <i>I</i>	HCIA*	HNN!	FII₩#	HCIA	#MEH	†IIN†	, #MUII	#VEI
≌	1	62	7	<i>د</i>		_	-	r.	=	Ma	¢	-1	~			-	-	
<u>5</u>	115	57	ş	ŝ	-	_	_	[6	-	_	~1	_	Ġ	÷	¢1	0	Ŷ
15	911	53	5	7	-	~ 1	a	-	9	÷	•	m	•	_	_	÷1	-	¢
15	117	33	*	9	0	-	¢	_	7	6 71	0	51	•	8	0	¢.	0	9
15	118	26	-	÷	_	r-ı	_	_	6	5	0	C1	_	¢	a	<u> </u>	0	~
15	611	Ř	~	30	_	_	0	.	9	5	0		•	2	_	_	_	¢
51	120	35	~	9	•	_	-	_	9	-	0		0	¢	G	_	0	\$
9t	121	ц Ю	\$	×	0	e.	0	_	5	Ŧ	7	~	-1	ri	ri	×	¢1	헙
9t	122	6E	-	7	G	ŝ	0	_	5	5	m	5	r-1		ŝ	(7	¢1	
16	123	8	ŝ	9	0	1 1	0	÷	2	_	m	~1	a	_	en.	'n	_	5
16	124	f.Ε	ŝ	~	0	~	c	_	5	۰.	÷	¢1	ጉነ	-	÷	Ŀ	_	2
16	125	7	ŝ	5	-1	m	0	_	타	~	v .	/ 1	5	Ŧ	r٩	m	Ŧ	Ξ
]6	126	¢۴	শ	~	¢	ŝ	•	_	51	ŝ	-	~	_	r)	ć	ч	6 1	-
91	127	4 0	vn.	×	0	15	c		F!	ъ.	ন	ŝ	_	2	Ē	Ŀ	CI	•
16	128	8 0 M	٢	Ş	0	'n	0	÷	17	4	÷.		.	'n	ŕ١	¢	r i n	=
17	129	#	-	_	₽	c 1	0	÷	0	-7	_	-7	Ċ	÷	=	_	0	ę
17	130	X 1	6 73	~	rı	_	Ċ	Ð	17	-	-1	7	¢.	¢.	•	~	6 1	
17	101	5	сı	ר 1	0	_	¢	•	0	ŕ~1	¢	ę	¢	•	0	2	¢	v.
13	132	20	ŝ	_	_	5	•	=	2	vŕ.	¢	-	_	0	0	1 .1	¢	ç
17	133	8 5	6 73	4	_	-1	¢	ش	4	¢1	=	ŝ	≏	1	cı	6	2	16
17	134	63	1	9		~1	ç		÷1:	÷.	×	ŝ	<u>.</u>	5	-	2	4	5
17	135	32	ci	Ŷ	-	_	Ċ	=	Ľ¦		Ŧ	Ś	¢	ſ.	-	r 1	c	÷
ŗ	901	57	ন	- 1	ru M	÷n	0	-	Ű	.,	\$	-+	9	2	-	\$	ŝ	Ŧ
18	LEI	5	9	20	-1	C1	¢	_	1	v,		+	~ 1	•	•	_	Ģ	ĝ
81	8£1	-1	Ŷ	-7		0	Ċ	_	6	.	¢	~ n	_	÷	•	_	ç	90
18	139	5	۲n	Ŷ	-	.	_	_	×	-+	¢	ζ.	-	•	0	_	0	~
81	070	얹	9	s.	_	_	c	_	×	-	¢	7 1	_	Ð	÷	e	a	ç
8	141	67	in.	۲	-1	_	Ċ	-	13	9		'n	_	•	•	сł	0	≙
•	!						:		•		:					:	-	

		-						4C D 4										
ΥΛ# 	Q#	#577	AIRC	#511K	#VIN	#1 KN	#PA	#GKA	PHRB	#1N1	#4191	#ANH	#WH1		#MEH	#NPE	FMOIT	# V L:H
18	142	18	5	3	1	Û	0	1	8	4	0	2	2	0	0	0	0	6
18	143	30	6	7	.2	1	2	2	10	4	I I	2	I	0	0	2	0	7
18	144	27	3	7	2	2	0	ŀ	12	5	1	3	I I	0	0	2	0	8
19	145	57	12	5	0	5	1	1	33	9	3	8	6	2	2	5	2	19
19	146	41	10	5	0	3	2	0	21	Ó	}	10	3	1	1	1	2	13
19	147	59	10	7	I I	5	0	3	33	7	3	9	5	2	2	7	2	19
19	148	40	7	7	0	3	I I	1	21	7	0	7	4	0	1	2	0	13
19	149	76	9	10	1	9	1	1	45	6	5	9	6	4	5	15	5	21
19	150	58	7	8	3	6	I I	2	31	4	3	4	4	4	4	11	3	4
19	151	58	6	ą.	I	8	[3	30	5	2	6	5	3	-I	8	3	15
19	152	66	7	12	0	6	I I	3	37	5	4	9	5	2	5	н	4	22
20	153	41	4	4	}	3	0	3	26	2	6	I	9	7	4	7	9	8
20	154	28	4	8	1	4	£1	I I	10	3	2	ļ	I	I	2	I.	1	4
20	155	55	7	7	2	6	1	3	29	3	ń	3	9	5	3	5	5	12
20	156	40	5	6	1	2	Û	2	24	I I	7	I I	6	5	3	7	6	5
20	157	32	4	6	2	6	0	1	13	ב	3	I I	I	I	3)	1	-1
20	158	25	2	7	1	3	0	1	12	2	2	3	3	3	2	0	3	7
20	159	37	3	6	3	3	0	3	19	1	5	0	7	-	1	6	5	6
20	160	14	2	0	1	2	0	I I	8	U	2	0	-1	2	1	2	3	1
2!	161	27	귀	4	0	3	0	0	16	2	1	6	3	I I	L	3	1	9
21	162	27	5	3	2	4	0	0	13	3	ì	6	0	0	0	3	0	8
21	163	30	7	3	0	3	0	0	17	4	I I	5	0	1	2	5	1	10
21	164	50	6	7	2	,ī	0	I	29	2	Ē	4	12	4	-	4	6	14
21	165	50	8	3	I	4	Q .	2	32	3	3	ú	8	4	ł	9	5	14
21	166	32	3	3	0	3	1	Ü	22	4	2	9	[]	1	2	5	1	12
21	167	34	6	5	2	ㅓ	0	Û	ł7	3	I I	4	3	I I	1	5	I I	11
21	168	51	7	5	3	4	0	I I	31	3	5	5	7	5	4	7	5	15
22	169	42	7	9	2	I.	Û	2	21	5	2	5	4	ł	2	3	L	15
··· - ·																		

Appendix 10.	Environmental data summary by quadrat (III).	
--------------	--	--

Appendix 10. Environmental data summary by quadrat (III).

₽A#	Ð	492₩	∦TRE	#SHR	NIV#	WRN	V.I <i>i</i>	#GRA	ALLER	NINEI	HOV	#ANII	HIW/	HD1I	4MEH	H.IN#	ΗΩΜ₿	₩ЛA#
2	170	. 97	9	6		6	0	 -	9 9		. =-1	Ś	~1	. 0	 _		 -	=
5	171	1 0	40	01	(m	_	0	¢1	9	vr.	_	4		0	_	2	0	ŝ
22	172	95	•	신	_	_	0	_	<u>+</u>	÷	_	4	~	0	0	r₁	0	10
22	173	37	9	6	r -1	_	_	_	17	+	r 1	стì	4	0	_	r")	ò	2
22	174	<u>ج</u>	=	13	CI	_	0	_	16 1	7	_	r .	r.	-	¢	12	c	=
22	175	35	٢	6	ŕ	-	⇒	_	4	6	~1	m	~1	_	_	_	•	=
22	176	45		Ŧ	CI	_	_	_	18	•	5	5	~	r1	_	Ē	~ :	<u>1</u>
23	177	35	5	5	6-1	_	0	- 1]6	,	_	÷.	-	_	_	m.	Ġ	¢
23	178	36	vс,	¥7	r -1	0	0		5	m	0	Ŧ	2	-1	_	e r;	5	ç
23	179	5	5	ŝ	r,	Т	•	~1	0	-	•	m	_	0	_	сı N	0	r~
33	180	ž	9	9	s	-1	ð		13	~	_	Ŧ	_	0	_	rn	0	6
3	181	32	7	٢	+	~:	0	.	0	-	0	Ŧ	_	_	_	•	_	æ
23	182	18	5	m	_	-	0	7 1	ŝ	-1	-	~1	-	a	•	_	0	÷
23	183	<u>7</u> 6	ır.	ç	÷	¢	0	6 1	₽	~	-	~	_	0	¢	c.I	c	ç
23	184	90	~	9	•		c	_	ñ	÷	_	-	_	0	_	~ 1	0	0
T.	185	저	r-	-+		-	D	=	¢	-	_	_	τı	_	_	C 1	_	÷
51	186	3	2	-	m	w.	5	r în	22 17	\$	F-	_	ę	Ś	~	7	ŝ	2
5	187	61	0	-	m.	9	0	~1	5	÷	Ŧ	Ţ	ó	÷	m	6	r,	2
77	188	1 T	च	*0	¢	ç	0	ŕ.	1	č 1	Ŧ	0	10	~	m .	,	2	2
44	681	65	2	e	m	ų	0	_	N CI	-7	Ŧ	C 1	•		-	9	-	12
4	190	6 4	<u>0</u>	Ľ	-		0	_	2	e.	J	_	\$		m	S	m .	6
24	161	5	¢	9	ŕ4	\$	0	m	36	.,	w.,	r.	~		.	ŝ	m	I
14	<u>192</u>	47	⁰	Ŀ	-	6	0	Ċ1	Ē	re.	n	_	9	Ē		\$	m	Ξ

.

:

.

.

.

....

-

APPENDIX 11

REPRESENTATIVE SEED DISPERSAL DISTANCES OF NATIVE AND ALIEN SPECIES IN NORTH AMERICA

Selected references to species from other continents are cited by country and enclosed in brackets. Methodologies vary: direct observation, experimental treatments (indoor, outdoor), estimated lateral dispersal distance based on terminal velocity values or mathematical models, and, measured parent-seedling distances (rarely). Legend: mtd = maximum trap distance. References: (1) Anderson 1988. (2) Anderson 1991. (3) Arnold 1981. (4) Antonovics and Ellstrand 1985. (5) Baker and O'Dowd 1982. (6) Bakker 1960, (7) Beattie and Lyons 1975, (8) Beer and Swaine 1977, (9) Berg 1966, (10) Berg 1969, (11) Boyd 1996, (12) Brodie 1951, (13) Brodie 1955, (14) Bullock 1989, (15) Bullock and Primack 1977, (16) Bulow-Olsen 1984, (17) Burton 1989, (18) Cain et al. 1998, (19) Campbell 1983, (20) Carey and Watkinson 1993, (21) Caspar 1987, (22) Clark et al. 1998, (23) Culver and Beattie 1978, (24) Davidson and Morton 1981, (25) Darley-Hill and Johnson 1981, (26) Evans et al. 1987, (27) Fenster 1991, (28) Gashwiler 1969, (29) Geiger 1950, (30) Greene and Johnson 1989. (31) Greene and Johnson 1995, (32) Greene and Johnson 1996. (33) Gross 1986, (34) Gross and Werner 1982, (35) Hinds and Hawksworth 1965, (36) Hoppes 1988, (37) Howard 1970, (38) Hughes and Westoby 1992ab. (39) Hughes et al. 1994. (40) Johnson 1996. (41) Johnson and Adkisson 1985. (42) Johnson and Patterson unpubl., (43) Jules 1996, (44) Kalisz unpubl., (45) Klinkenhamer et al. 1988, (46) Kjellsson 1985, (47) Kohlermann 1950, (48) Lee 1984. (49) Levin and Kerster 1969. (50) Levin and Kerster 1974, (51) Matlack 1987, (52) McCaughey et al. 1986. (53) McDonnell and Stiles 1983. (54) McEvoy and Cox 1987. (55) Meagher and Thompson 1987. (56) Mesler and Lu 1983. (57) Meyer and Schmid 1999. (58) Morse and Schmitt 1985. (59) Muller 1955. (60) O'Dowd and Hay 1980. (61) Ohara and Higashi 1987, (62) Okbo and Levin 1989, (63) Olivieri and Gouyon 1985. (64) Parra et al. 1993. (65) Pemberton 1988. (66) Platt 1976. (67) Platt and Weiss 1977. (68) Pudlo et al. 1980. (69) Rabinowitz and Rapp 1979. (70) Sacchi 1987. (71) Savile 1953. (72) Schaal 1980. (73) Schemske 1978. (74) Schmitt et al. 1985, (75) Sheldon and Burrows 1973, (76) Smith 1975, (77) Smith 1989, (78) Smith and Kok 1984, (79) Stamp 1989. (80) Stamp and Lucas 1983, (81) Stamp and Lucas 1990, (82) Stapanian and Smith 1978, (83) Stergios 1976. (84) Stratton 1994, (85) Thiede and Augspurger 1996, (86) Tomback and Linhart 1990, (87) Trapp. 1988, (\$8) Vander Wall and Balda 1977, (89) Venable and Levin 1985, (90) Vickery et al. 1986, (91) Waller 1980, (92)Watkinson 1978, (93) Webb and Willson 1985, (94) Weiblen and Thomson 1995, (95) Wemer 1975, (96) Wemer and Platt 1976. (97) Westlaken and Maun 1985. (98) Westoby and Rice 1981. (99) Willson et al. 1990.

Species		5	Seed Dispersal	Distance		Ref.#
	Mean (m)	Median (m)	Maximum (m)	Range (m)	Terminal Velocity (cm/s)	
ANIMAL DISPERSED						
Fleshy Fruits				-		
trees, shrubs, vines, herbs (n=13 spp)			170	-		53
Trees (n=1)						
Prunus serotina		E	25 35 9			76 53 36
Sarubs (n=3)		1				

Species		S	eed Dispersa	I Distance		Ref.#
	Mean (m)	Median (m)	Maximum (m)	Range (m)	Terminal Velocity (cm/s)	
Prunus virginiana			10 (mtd)			93
Prunus avium			10 (mtd)			93
Vines (n=2)						
Parthenocissus quinquefolia		9				36
Toxicodendron radicans			33 (mtd)			36
Vitis vulpina		0.5-6 (mode)				36
Herbs (n=1)						
Phytolacca americana			33 (mtd)			36
Nut Fruits						
Trees (n=3)			ŧ			
Fagus grandifolia			4 km			41
Juglans nigra	38.1		T			82
Quercus palustris	I.I.km		1.9 km 5 km	0.1-1.9 km		25 42 (25)
Adhesive Fruits						
Herbs (n=4)		<u> </u>				
Achyranihes uspera (Costa Rica)	24.7m - 2.4 km					15
Bidens sp. (Costa Rica)	12.4- 108.8					15
Daucus carota			15	1		34
Petiveria alliacea (Costa Rica)	32.9- 156.6					[5
Elaíosome Seeds/Fruits						
Trees (n=2)					<u> </u>	
Acacia suaveolens (Australia)	2.0-6.5	-	10.9	0.4-10.9	-+	1

Appendix11. Representative seed dispersal distances of native and alien species in N. America.

Species		S	ced Dispersa	Distance		Ref.#
	Mean (m)	Median (m)	Maximum (m)	Range (m)	Terminal Velocity (em/s)	
Acacia terminalis (Australia)	1.1	1	1.9	0.2-1.9		38
Shrubs (n=4)						
Dendromecon harfordii	3.6		5.1			9
Dendromacan rigida	2.7		4.8 > 0			9 14
Dillwynia retorta (Australia)	1.2		11.0	0.t-11.0		38
Fremontodendron decumbens			12	60% 6-8		ļι
Herbs (n=18)						
Asarum canadense	1.5		<u> </u>	0.1-35.0		18
Carex pilulifera (Denmark)				0-1.4		-16
Crotalaria rotundifolia	2.2					81
Datura discolor	2.3	1	39			60
Dicentra formosa			56			10
Erythronium grandiflorum		0.3	el (mtd)			94
Euphorbia esula			25 (mtd)	0-8		65
Sanquinaria candensis	0.2-3.1		12			68
Selerolaena dicantha (Australia)			77	4-77: 50-77 common		24
Trillium grandiflorum			10.0	<t-10.0< td=""><td></td><td>44</td></t-10.0<>		44
Trillium kamtschaticum (Japan)	0.6		3.3	0.2-3.3		6 l
Trillium ovatum			1.8 >30 (wasp)	0-1.8 ≥70% <0.6 ≥30 (wasp)		57 57 43
Trillium tschonoskii (Japan)	0.6		0.4	0.01-0.4		61
Viola odorata			50			77

Species	Seed Dispersal Distance					
	Mean (m)	Median (m)	Masimum (m)	Range (m)	Terminal Velocity (cm/s)	
Viola pedata	0.4	+ · · ·				. 23
Viola pensylvanica	0.6					23
Viola rostrata	0.6	1				23
Viola triloba	0.1					23
Other Fruits (n=3)						
Trees				ĺ		
Pinus spp			22 km	12-22 km		86
Pinus edulis			22 km	7.5-22 km		88
Herbs						
Mimulus guttatus			l km (deer)			90
WIND DISPERSED						
Trees (n=31)		1			1	
Abies alba	72		835 (mtd)			47 (39) 31
Acer griseum			29.2		124	51
Acer negundo			-+1,1		76 80 104	51 30 31
Acer platanoides	83		50.3		87 87	51 47 (30) 47 (39)
Acer pseudoplatanus	69		60 .1		104 95	51 47 (30) 47 (39)
Acer rubrum			98.7 835 (mtd)		76	51 31
Acer saccharum			170		127 124	17 (32) 31

		B		1 11 .		1 17		
Δ.	vnendiv i i	Kenrecents	tiive ceed disner	ol distances.	of notive and	i siren s	mentes in Ni	America
	pension i re	represente	mite accu diapor.	na diatanteea	or manye and		peeres min	Internets

Species	Seed Dispersal Distance					
	Mean (m)	Median (m)	Maximum (m)	Range (m)	Terminal Velocity (cm/s)	-
Albizzia julibrissum			16.2		154	51
Betula papyrifera			835 (mtd)		55	37 (32) 31
Betula populifolia			64.3		39	51
Catalpa bignoides			16.8		186	51
Carpinus betulus	73					47 (39)
Carpinus caroliniana			[9,4		129	51
Cedrus atlantica var. glauca			68.8		112	51
Fraxinus spp			>50	15% > 50	120-170	47 (29) 31
Fraxinus americana	[9,3		70.1 131		104 142 131	51 30 17 (32) 22
Fraxinus excelsior	53		40.3		217	51 47 (39)
Larix decidua	29		_		£19	47 (39) 32
Larix laricina			50	15-50	119	52 32
Liriodendron tulipifera	33.9		111.9		11	51 22
Piceu glauca			300		61	52 32
Picea mariana			165 (mtd) 100		_	37 52
Pinus bankstana			40	34-40		52
Pinus contorta			rarely >61 244 60		6[52 52 30
Pinus rigida	15.1			T		22

Species		Seed Dispersal Distance				
	Mean (m)	Median (m)	Maximum (m)	Range (m)	Terminal Velocity (cm/s)	
Pinus strohus	46		835 (mtd)			47 (39) 31
Plotanus occidentalis			62.8		166	51
Pseudotsuga menziesti			122 (mtd)	61% (5-30 26% 61-76 13% 107-122		28 28 28
Thuja plicata			122 (mtd)	79% 15-30 17% 61-76		28 28
Thuja plicata			122 (mtd) 201	4% 107-122		28 52
Tilia ameriçana	13.0		15.0		180	51 31 22
Tilia grandifolia	58				1	47 (39)
Tilia parviflora				 	103	47 (30)
Tsuga canadensis	19.7					12
Ulmus campestris	60				-	47 (39)
Shrubs (n=2)						
Almus crispa			6.7		94	51
Halesia monticola			18.5		338	51
Vines (n=1)						
Clematis virginiona			3.2		129	51
Grasses (n=5)						
Agrostis hiemalis					96	69
Andropogon gerardi					154	69
Andropogon glomeratus					20.1	19
Andropogon virginicus					32.3	19

Species		S	eed Dispersal	lispersal Distance				
	Mean (m)	Median (m)	Maximum (m)	Range (m)	Terminal Velocity (cm/s)			
Schizachyrium scoparium					91 101	69 19		
Herbs (n=51)				1		<u> </u>		
Apocynum cannabium			83.3		15	51		
Asclepias syriaca	13.8		45 150 (mmd) 18.1	4.2-25.3 2% >150	24.9	67 58 58 51		
Asclepias syriaca (cont'd)	15.6- 43.7	–				70		
Apocynum sibiricum	25.7		75		9.9	67		
Aster acuminatus		1	5.1		41	51		
Aster prenanthoides			. 3.1		66	51		
Carduus nutans			<3*s - 100 (mtd)	80% <40		78		
Carduus pyconcephalus			45	3-45		63		
Carduus tenuiflorus			2.1	0.7-2.1	78.6	75		
Carlina vulgaris			1.5	0.5-1.5	58.5	75		
Centauria scabiosa			1.6	0.5-1.6	219.8	75		
Cirsium arvense			2 km (mtd) 11.4	3.8-11.4	26 21.6	6 75		
Cirsium palustre		1	6.8	2.3-6.8	34.4	75		
Cirsium undulation	18.4	1	35		30.0	67		
Cirsium vulgare			11.6 32 (mtd) >>km	50% <1 16% 1-2 10% >32	36	51 45 45 45		
Cryptantha flava			31.3		1	21		

Species		Seed Dispersal Distance					
	Mean (m)	Median (m)	Maximum (m)	Range (m)	Terminal Velocity (cm/s)		
Erigeron acer			3,4	1.2-2.4	22.9	75	
Erigeron annus			2.5	[-25 <1% >3		84	
Eupatorium cannabinum			5.9 7.8	2.0-5.9	39.3 32	75 51	
Eupatorium rugosum			7.1		45	51	
Heterotheca latifolia	1.1-1.7		20 (mtd)			89	
Hieracium aurantiacium			2.0 (mtd)	82-95% ≤1.0		83	
Hieracium umbellatum			3.0		55	ΣI	
Hypochoeris glahra]			45	3	
Hypochoeris radicata			2.0	0.7-2.0	40.5	75	
Leontodon autumnalis			1.1	0.6-1.1	51.2	75	
Liatris aspera	2.5		9	-		19	
Mimulus guitatus		:	4.8	44% <0.3 8%n ≥ l		90	
Physalis subgiabrata			1.5		212	51	
Rumex obtusifalia			2.5		126	51	
Senecio jacobea disk achenes disk achenes	0.6-3.3		5.5 14	1.8-5.5 31% 1 58% 1-5	42.1	75 54 54	
Senecio squalidus			1.8 2.5	0.6-1.8	45.7 33	75 51	
Senecio viscosus			1.6	0.9-2.6	31.7	75	
Senecio vulgaris			2.9	1.0-2.9	28.0	75	
Solidago altissima			14.9 6-8 (mtd)	14.9	28	51 57	
Solidago conadensis			30.9	16.8-30.9		96	
Solidago gigantea			23.8	[96	

Species	Seed Dispersal Distance					
	Mean (m)	Median (m)	Maximum (m)	Range (m)	Terminal Velocity (cm/s)	
Solidago graminifolia	1		21.1	12.0-21.1		96
Solidago missouriensis			29,4	16.2-29.4		96
Solidago nemoralis		1	39.0	27.9-39.0		96
Solidago rigida	4.9	1	15	1	33.9	67
Solidago speciosa	1		41.4	37.6-41.4		96
Sonchus arvensis			10.0	3.3-10.0	24.1	73
Sonchus oleraceus		0.1-0.9 (mode)		1166	15.7	2
Taraxacum officinale			2.3	0.8-2.3	35.7 35.7 33.7 31 42	75 67 19 51
Tragopogon dubius		2 (mode)	>250		-	34 33 (62)
Tragopogon dubius					40.0	31
Tragopogon parrifolius			1.8 5.8 6.9	0.6-1.8	45.7 36 30	75 51 51
Trifolium arvense			0.8		125	51
Tusilago f arfar a		1	4 km (ntd) 4.4	1.5-4.4	15 19.2	6 75
Typha latifolia			46.9		13	51
BALLISTIC DISPERSED		1				
Shrubs (n=4)						1
Ceanothus cuneatus			10 (mtd)	[% = 9		26
Ceanothus leucodermis	1		10 (mtd)	[% = 9		26
Dendromecon harfordii	3.6	•	5.1		- ·	9
Dendromecon rigida	1.3	 	4.8	2.9-4.8		9

Appendix11. Representative seed dispersal distances of native and alien species in N. America.

Species	Seed Dispersal Distance					
	Mean (m)	Median (m)	Maximum (m)	Range (m)	Terminal Velocity (cm/s)	
Herbs (n=42)		-				
Ballechores s. I.			45			8
Amphicarpu bracteata: chasmogamous seeds aerial cleistogamous seeds		1.4 1.3	4.3 3.8	31% >2.0 25% >2.0		87 87
Arcenthobium cryptopodum			4.6			35
Cardamine hirsuta			1.4			39 (50)
Cassia fasciculata	1.4-2.1		4.8	1.6% >4		48
Chamaeorista fasciculata	0.3-0.7					27
Chrysosplenium americanum			0.4			71
Cnidascolus stimulosus	0.6		1.2			81
Crotaluria sp			4.3			49
Crotalaria rotundifolia	0.9		2.4			81
Erodium moschatum	0.6		0.9	0.0-0.9		79
Euphorbia marginata			4.0			60 (50)
Geranium carolinianum	3.3		4.2	1.8-4.2		80
Geranium columbinum			1.5			60 (50)
Geronium molle	1.8		2.8	1.0-2.8		80
Geranium maculatum	3.0		4.8	1.8-4.8		80
Impatiens copensis	0.2		1.6 2.0 2.1	0.1-1.6 1.5-2 most <0.2 15% >1.0		80 73 91 74 74
Impatiens pallida			2.0			73
Impatiens parviflora	ļ		3.4			60 (50)
Kalanchoe tubiflora			1.5			12
Lepidium campestre	0.3-0.8		2.0 (mid)			85

Species		Seed Dispersal Distance					
	Mean (m)	Median (m)	Maximum (m)	Range (m)	Terminal Velocity (cm/s)		
Lepidium vulgare			>1.9			99	
Lupinus lexensis	0.6		4			72	
Mitella diphylla	0.3		0.8	0.1-0.8		71	
Montia fontana			2.0			60 (50)	
Oenothera biennis	1.8	<1	5		120	67 34	
Oenothera rosea			1.0			12	
Oxalis acetosella			2.3			60 (50)	
Phlox drummondii	0.8		1.6	0.1-1 .6		80	
Phlox pilosa	1.2		3.6			49	
Sagina decumbens			0.5	1		12	
Salvia lyrata			2.0	0.5-2.0		13	
Stillingia sylvatica	1.1		3.0			81	
Verbascum thapsus		1	11			34	
<i>Viola blanda</i> chasmogamous seeds cleistogamous seeds	L.0 0.8		3.0 3.8 2.2	2.4-3.0		ו 6 7 7	
Viola cucullata	1.5		2.1			7	
Viola eriocarpa	1.2		5.4			7	
Viola fimbriatula						16	
Viola lanceolata						16	
Viola papilionacea chasmogamous seeds cleistogamous seeds	2.t 1.0		4.8 2.1			777	
Viola pedato	1.4		5.1			7	
Viola rostrata	1.2		4.2			7	

Species		Seed Dispersal Distance					
	Mean (m)	Median (m)	Maximum (m)	Range (m)	Terminal Velocity (cm/s)		
<i>Viola striata</i> chasmogamous seeds cleistogamous seeds	1.5 1.1 0.8		3.0 2.2 3.4	0.1-3.4		7 7 80	
UNASSISTED							
Grasses (n=3)							
Anthoxanthum odoratum			2	95% <1		+	
Vulpia ciliata	0.1					20	
Vulpia fasciculata			0.4	79% <0.1		92	
Herbs (n=11)							
Chamaelirium luteum	10.1		34			56	
Chaenarrhinum minus			1.4	75% <0.1		3	
Dipsacus sylvestris			1.5			95	
Echeveria gibbiflora	[.]					64	
Heterotheca latifolia ray achenes	0.5-0.9		12			89	
Lithospermum caroliniense			2.0 (mtd)	most <1		97	
Mimulus guittatus			4.8			90	
Mirabilis hirsuta	0.4		1.6 1.5		217	66 67	
Verbena stricta	1.0		3.5		166	67	

APPENDIX12

PRINCIPAL FRUGIVORES OF EASTERN NORTH AMERICA

Appendix 12. Principal Frugivores of Eastern North America. Legend: * Major frugivore (>10 stars. Martin *et al.* 1951, as calculated by Willson 1986): Sources: Birds: Willson 1986; Mammals: Hamilton 1941, Martin *et al.* 1951; Reptiles: Rust and Roth 1981. Willson 1986, Stiles 1992.

BIRDS (n=50)	
Eastern bluebird*	White-throated sparrow
Northern bobwhite	European starling*
Northern cardinal	Tree swallow
Gray catbird*	Summer tanager
Yellow-breasted chat	Brown thrasher*
American crow	Gray-cheeked thrush
Fish crow*	Hermit thrush
Purple finch	Swainson's thrush
Northern flicker	Wood thrush*
Great-crested flycatcher	Tufted titmouse
Evening grosbeak	Eastern towhee
Pine grosbeak	Wild turkey
Rose-breasted grosbeak	Veery
Ruffed-grouse*	Red-eyed vireo
Spruce grouse	Warbling vireo
Eastern kingbird	Bay-breasted warbler
Ruby-crowned kinglet	Chestnut-sided warbler
Northern mockingbird*	Tennessee warbler
Baltimore oriole	Yellow-rumped warbler
Orchard oriole	Bohemian waxwing
Ring-necked pheasant	Cedar waxwing*
American robin*	Downy woodpecker
Yellow-bellied sapsucker	Pilcated woodpecker
Fox sparrow	Red-bellied woodpecker

Appendix 12. Principal Frugivores of Eastern North America (cont'd).

MAMMALS (n=14)	
Black bear*	Moose
Eastern chipmunk	Deer mouse
Common cottontail	White-footed mouse
New England cottontail	Raccoon
White-tailed deer*	Eastern fox squirrel
Eastern red fox	Eastern gray squirrel
Gray fox	Eastern red squirrel
REPTILES (n=1)	
Box Turtle	

_

APPENDIX 13

KNOWN MYRMECOCHORES IN THE U.S. NORTHEAST

Appendix 13. Known Myrmecochores in the U.S. Northeast. Sources: (1) Beattie and Lyons 1975. (2) Beattie et al. 1979. (3) Beattie and Culver 1978. (4) Beattie and Culver 1981. (5) Culver and Beattie 1978. (6) Culver and Beattie 1980. (7) Gaddy 1986. (8) Gates 1941. (9) Gates 1942. (10) Gunther and Lanza 1989. (11) Handel 1976. (12) Handel 1978b. (13) Heithaus 1981. (14) Pudlo et al. 1980. (15) Ruhren and Dudash 1996

Name	Reference	Nesting Preference		
Aphaenogaster rudis.	4.5.6,7.11.12,13,14.15	majority of <i>Aphaenogaster</i> spp. nest in the soil; usually start nest beneath some covering object; if a log, ants may construct part of nest in it but main part of nest is subterranean (Creighton 1950, p.139)		
Aphaenogaster tennesseensis	12	usually nests on rotten stumps and fallen logs with few passages running into the soil (Creighton, 1950, p.139; Lynch <i>et al.</i> 1988)		
Aphaenogaster texuna	5	see A. rudis.		
Camponotus nearcticus	14	most species of <i>Cumponotus</i> nest in decaying wood, especially in soft and rotting parts: nests that occur in logs or trees rarely extend into the soil (Creighton 1950, p.365)		
Camponotus pennsylvanicus	8,14	see C. nearcticus.		
Crematogaster lineolata	7	nests in dead wood of standing or prostrate trunks (Wheeler 1960, p. 208)		
Formica fusca	1	nests in soil beneath a covering object (Creighton 1950, p.528)		
Formica integra	5.13	nests in stumps and in soil under cover of stones, logs, or branches (Wheeler 1960, pp. 204-206).		
Formica neogagales	8,9	nests in soil beneath stones or other covering objects (Creighton 1950, p 457)		
Formica subsericea	1.2,5.8,13.14	nests in soll where it forms mounds up to 30 cm high (Wheeler 1960, pp. 203-204)		

Name	Reference	Nesting Preference		
Lasius alienus	1,5.8.9,13,14	remarkably flexible; most specie prefer well drained soil that is not to dry: nest may be free in soil, unde stones or other covering objects, or i and under rotten logs and stump (Creighton 1950)		
Leptothorax curvispinosus	5,13	genus nests by choice in preformed cavities: e.g., in crannies under rock chips, or under bark, in hollow twigs, dried grass stems, old galls, or empty nut shells (Creighton 1950, p. 255)		
Leptothorax longispinosus	5.13	see L. curvispinosus.		
Myrmecina americana	5	usually nests in moist shady areas. often under small stones (Creighton 1950, p.248)		
Myrmica emeryana	9.14	most species in genus nest in soil under a covering object (Creighton 1950, p.91)		
Myrmica fracticornis	8	see M. emeryana.		
Myrmica punctiventris	1.2.3.10.13,14,	see M. emeryana.		
Myrmica spatulata	1	see M. emeryana.		
Phrenolepis imparis	13,14	often nests in damp soil in shady positions (Creighton 1950, p.135)		
Stenamma schmitti	5.13	habits little known: nests in wooded areas in leaf mould, under stones or logs or beneath thick, loose moss (Creighton 1950, p.135)		
Tapinoma sessile	1.5.13	not particular; in soil with or without a covering object, under bark, and in preformed cavities of various kinds (Creighton 1950, p.351)		

Appendix 13. Known Myrmecochores in the US Northeast (cont'd).

APPENDIX 14

SPECIES PREVALENT IN THE HERB LAYER IN THE MAPLE-BASSWOOD FOREST REGION IN SOUTHERN WISCONSIN AND PRESENT IN SUGAR MAPLE DOMINATED STANDS IN THE VICINITY OF PETERBOROUGH, ONTARIO

Appendix 14. Species prevalent in the herb layer in the Maple-Basswood Forest Region in southern Wisconsin and present in sugar maple dominated stands in the vicinity of Peterborough, Ontario. "•" denotes that species achieves maximum presence in this forest type in Wisconsin. W=Wisconsin, P=Peterborough; Flowering phenology at Peterborough: E=early spring flowering (before June 1st); ML=mid to late season flowering (after June 1st). Source: Curtis 1959, p.521.

Species	W	P	Species	w	6
Actaea pachypoda		E	Geum canadense		E
Adiantum pedatum		?	Hepatica acutiloba	•	E
Allium tricoccum	•	E	Hydrophyllum virginianum	•	E
Amphicarpaeu bructeatu		ML	Luportea canadensis		ML
Arisaema triphyllum		É	Matanthemum cunadense		E
Athyrium filix-femina		?	Osmorhiza claytonii		E
Botrychium virginianum		E	Parthenocissus inserta		ML
Brachyelytrum erectum		E	Phryma leptostachya		ML
Carex laxiflora		E	Podophyllum peltatum	•	Ë
Carex pensylvanica		Е	Polygonatum pubscens		C
Caulophyllum thalictroides	•	E	Prenanthes alba		ML
Celastrus scandens		ML	Ribes cynosbati		ML
Circaea lutiana		ML	Sanguinaria canadensis	•	E
Claytonia caroliniana		Е	Solidago flexicaulis		ML
Cryptotaenia canadensis		ML	Thalietrum dioicum		E
Erythronium americanum		Ē	Trillum grandiflorum	•	E.
Galium aparine	•	E	Uvularia grandiflora		E
Galium triflorum		ML	Viola cucullata		E
Geranium maculatum		E	Viola pubescens	•	E

Notes:

Species prevalent in Wisconsin but absent from stands at Peterborough: Erythronium albidum, Claytonia virginica, Galium concinnum, Sanicula gregaria, Smilax ecirrhata.